UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

FORM 10-Q

[X] QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the quarterly period ended **September 30, 2010**

or

[] TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES

Commission File Number	state of	For the transition period fromto me of registrant as specified in charter, f incorporation, address of principal ntive offices and telephone number		.S. Employe fication Nur	
001-32206	GREAT I	PLAINS ENERGY INCORPORATED (A Missouri Corporation) 1200 Main Street Kansas City, Missouri 64105 (816) 556-2200 www.greatplainsenergy.com	4	13-1916803	
000-51873	KANSAS	CITY POWER & LIGHT COMPANY (A Missouri Corporation) 1200 Main Street Kansas City, Missouri 64105 (816) 556-2200 www.kcpl.com	4	14-0308720	
	nonths (or for such she	all reports required to be filed by Section 13 or 15 orter period that the registrant was required to file Kansas City Power & Light Company	e such reports), and (2) has been	<u>X</u> No	_
		d electronically and posted on its corporate Web s ule 405 of Regulation S-T (§232.405 of this chap			
Data File required to be submitted and months (or for such shorter period that	t the registrant was rec Yes \underline{X} No	_ Kansas City Power & Light Company	Yes	_ No	_
Data File required to be submitted and months (or for such shorter period tha Great Plains Energy Incorporated Indicate by check mark whether the re	Yes X No egistrant is a large accelerated filer," "a		ed filer, or a smaller reporting	- - -	_

Kansas City Power & Light Company meets the conditions set forth in General Instruction (H)(1)(a) and (b) of Form 10-Q and is therefore filing this Form 10-Q with the reduced disclosure format.

Kansas City Power & Light Company had one share of common stock outstanding and held by Great Plains Energy Incorporated.

This combined Quarterly Report on Form 10-Q is being filed by Great Plains Energy Incorporated (Great Plains Energy) and Kansas City Power & Light Company (KCP&L). KCP&L is a wholly owned subsidiary of Great Plains Energy and represents a significant portion of its assets, liabilities, revenues, expenses and operations. Thus, all information contained in this report relates to, and is filed by, Great Plains Energy. Information that is specifically identified in this report as relating solely to Great Plains Energy, such as its financial statements and all information relating to Great Plains Energy's other operations, businesses and subsidiaries, including KCP&L Greater Missouri Operations Company (GMO), does not relate to, and is not filed by, KCP&L. KCP&L makes no representation as to that information. Neither Great Plains Energy nor its other subsidiaries have any obligation in respect of KCP&L's debt securities and holders of such securities should not consider Great Plains Energy's or its other subsidiaries' financial resources or results of operations in making a decision with respect to KCP&L's debt securities. Similarly, KCP&L has no obligation in respect of securities of Great Plains Energy or its other subsidiaries.

This report should be read in its entirety. No one section of the report deals with all aspects of the subject matter. It should be read in conjunction with the consolidated financial statements and related notes and with the management's discussion and analysis included in the 2009 Form 10-K for each of Great Plains Energy and KCP&L.

CAUTIONARY STATEMENTS REGARDING CERTAIN FORWARD-LOOKING INFORMATION

Statements made in this report that are not based on historical facts are forward-looking, may involve risks and uncertainties, and are intended to be as of the date when made. Forward-looking statements include, but are not limited to, the outcome of regulatory proceedings, cost estimates of the Comprehensive Energy Plan and other matters affecting future operations. In connection with the safe harbor provisions of the Private Securities Litigation Reform Act of 1995, the registrants are providing a number of important factors that could cause actual results to differ materially from the provided forward-looking information. These important factors include: future economic conditions in regional, national and international markets and their effects on sales, prices and costs, including but not limited to possible further deterioration in economic conditions and the timing and extent of any economic recovery; prices and availability of electricity in regional and national wholesale markets; market perception of the energy industry, Great Plains Energy and KCP&L; changes in business strategy, operations or development plans; effects of current or proposed state and federal legislative and regulatory actions or developments, including, but not limited to, deregulation, re-regulation and restructuring of the electric utility industry; decisions of regulators regarding rates the companies can charge for electricity; adverse changes in applicable laws, regulations, rules, principles or practices governing tax, accounting and environmental matters including, but not limited to, air and water quality; financial market conditions and performance including, but not limited to, changes in interest rates and credit spreads and in availability and cost of capital and the effects on nuclear decommissioning trust and p ension plan assets and costs; impairments of longlived assets or goodwill; credit ratings; inflation rates; effectiveness of risk management policies and procedures and the ability of counterparties to satisfy their contractual commitments; impact of terrorist acts; increased competition including, but not limited to, retail choice in the electric utility industry and the entry of new competitors; ability to carry out marketing and sales plans; weather conditions including, but not limited to, weather-related damage and their effects on sales, prices and costs; cost, availability, quality and deliverability of fuel; ability to achieve generation goals and the occurrence and duration of planned and unplanned generation outages; delays in the anticipated in-service dates and cost increases of additional generating capacity and environmental projects; nuclear operations; workforce risks, including, but not limited to, increased costs of retirement, healthcare and other benefits; the timing and amount of result ing synergy savings from the GMO acquisition; and other risks and uncertainties.

This list of factors is not all-inclusive because it is not possible to predict all factors. Part II Item 1A Risk Factors included in this report, together with the risk factors included in the 2009 Form 10-K for each of Great Plains Energy and KCP&L under Part I Item 1A, should be carefully read for further understanding of potential risks for each of Great Plains Energy and KCP&L. Other sections of this report and other periodic reports filed by each of Great Plains Energy and KCP&L with the Securities and Exchange Commission (SEC) should also be read for more information regarding risk factors. Each forward-looking statement speaks only as of the date of the particular statement. Great Plains Energy and KCP&L undertake no obligation to publicly update or revise any forward-loo king statement, whether as a result of new information, future events or otherwise.

GLOSSARY OF TERMS

The following is a glossary of frequently used abbreviations or acronyms that are found throughout this report.

Abbreviation or Acronym

AFUDC Allowance for Funds Used During Construction

ARO Asset Retirement Obligation
BART Best available retrofit technology
Board Great Plains Energy Board of Directors

CAIR Clean Air Interstate Rule
CAMR Clean Air Mercury Rule

Clean Air Act Clean Air Act Amendments of 1990

CO₂ Carbon dioxide

Collaboration Agreement Agreement among KCP&L, the Sierra Club and the Concerned

Citizens of Platte County

Company Great Plains Energy Incorporated and its subsidiaries

Companies Great Plains Energy Incorporated and its consolidated subsidiaries and KCP&L and its

Definition

consolidated subsidiaries

DOE Department of Energy ECA Energy Cost Adjustment

EIRR Environmental Improvement Revenue Refunding

EPA Environmental Protection Agency **EPS** Earnings per common share

ERISA Employee Retirement Income Security Act of 1974, as amended

FAC Fuel Adjustment Clause

FASB Financial Accounting Standards Board
FERC The Federal Energy Regulatory Commission
FGIC Financial Guaranty Insurance Company

FSS Forward Starting Swaps

GAAP Generally Accepted Accounting Principles

GMO KCP&L Greater Missouri Operations Company, a wholly owned subsidiary of

Great Plains Energy as of July 14, 2008

Great Plains Energy Great Plains Energy Incorporated and its subsidiaries

ISO Independent System Operator

KCC The State Corporation Commission of the State of Kansas
KCP&L Kansas City Power & Light Company, a wholly owned subsidiary

of Great Plains Energy

KDHE Kansas Department of Health and Environment

KLT Inc. KLT Inc., a wholly owned subsidiary of Great Plains Energy

KW Kilowatt **kWh** Kilowatt hour

MACT Maximum achievable control technology

MD&A Management's Discussion and Analysis of Financial Condition and

Results of Operations

MDNR Missouri Department of Natural Resources

MGP Manufactured gas plant

MISO Midwest Independent Transmission System Operator, Inc.

MPS Merchant Services, Inc., a wholly owned subsidiary of GMO

MPSC Public Service Commission of the State of Missouri

MW Megawatt
MWh Megawatt hour

NERC North American Electric Reliability Corporation

NEIL Nuclear Electric Insurance Limited

Abbreviation or Acronym

NO_v Nitrogen oxide

NPNS Normal purchases and normal sales NRC **Nuclear Regulatory Commission** NYMEX New York Mercantile Exchange Other Comprehensive Income OCI Polychlorinated biphenyls **PCB** Pension Protection Act of 2006 **PPA**

PRB Powder River Basin **QCA** Quarterly Cost Adjustment

Kansas City Power & Light Receivables Company, a wholly owned **Receivables Company**

subsidiary of KCP&L

Regional Transmission Organization RTO SCR Selective catalytic reduction SEC Securities and Exchange Commission

SERP Supplemental Executive Retirement Plan

Great Plains Energy Services Incorporated, a wholly owned subsidiary of Services

Great Plains Energy

Sulfur dioxide SO_2

SPP Southwest Power Pool, Inc. **Syncora** Syncora Guarantee Inc. T - Lock Treasury Lock

WCNOC Wolf Creek Nuclear Operating Corporation Westar Westar Energy, Inc., a Kansas utility company

Wolf Creek Wolf Creek Generating Station **Definition**

PART I – FINANCIAL INFORMATION ITEM 1. FINANCIAL STATEMENTS

Great Plains Energy Incorporated

Unaudited Consolidated Balance Sheets

Unaudited Consolidated Statements of Income

Unaudited Consolidated Statements of Cash Flows

Unaudited Consolidated Statements of Common Shareholders' Equity

Unaudited Consolidated Statements of Comprehensive Income

Kansas City Power & Light Company

Unaudited Consolidated Balance Sheets

Unaudited Consolidated Statements of Income

Unaudited Consolidated Statements of Cash Flows

Unaudited Consolidated Statements of Common Shareholder's Equity

Unaudited Consolidated Statements of Comprehensive Income

Combined Notes to Unaudited Consolidated Financial Statements for Great Plains Energy Incorporated and Kansas City Power & Light Company

Note 1:	Summary of	Significant	Accounting	Policies

Note 2: Supplemental Cash Flow Information

Note 3: Receivables

Note 4: Assets Held For Sale

Note 5: Nuclear Plant

Note 6: Regulatory Matters

Note 7: Pension Plans and Other Employee Benefits

Note 8: Equity Compensation

Note 9: Short-Term Borrowings and Short-Term Bank Lines of Credit

Note 10: Long-Term Debt

Note 11: Commitments and Contingencies

Note 12: Legal Proceedings

Note 13: Related Party Transactions and Relationships

Note 14: Derivative Instruments

Note 15: Fair Value Measurements

Note 16: Taxes

Note 17: Segments and Related Information

Note 18: Discontinued Operations

Note 19: Goodwill

GREAT PLAINS ENERGY INCORPORATED Consolidated Balance Sheets

(Unaudited)

	September 30 2010		December 31 2009	
ASSETS	(millions,	except	share an	nounts)
Current Assets				
Cash and cash equivalents	\$	7.6	\$	65.9
Funds on deposit		4.4		4.4
Receivables, net	27	74.6		230.5
Accounts receivable pledged as collateral	9	95.0		-
Fuel inventories, at average cost	8	30.1		85.0
Materials and supplies, at average cost	13	31.3		121.3
Deferred refueling outage costs		11.5		19.5
Refundable income taxes	1	15.3		13.5
Deferred income taxes	4	12.9		36.8
Assets held for sale (Note 4)	1	18.8		19.4
Derivative instruments		1.1		1.5
Prepaid expenses and other assets	1	10.7		14.7
Total	69	93.3		612.5
Utility Plant, at Original Cost				
Electric	10,35	52.8		8,849.0
Less-accumulated depreciation	3,95	59.2		3,774.5
Net utility plant in service	6,39	93.6		5,074.5
Construction work in progress	35	50.9		1,508.4
Nuclear fuel, net of amortization of \$125.7 and \$106.0	1	30.2		68.2
Total	6,82	24.7		6,651.1
Investments and Other Assets				
Affordable housing limited partnerships		11.5		13.2
Nuclear decommissioning trust fund	12	20.7		112.5
Regulatory assets	88	38.8		822.2
Goodwill	10	59.0		169.0
Derivative instruments		8.1		7.9
Other		33.0		94.4
Total	1,28	31.1		1,219.2
Total	\$ 8,79	99.1	\$	8,482.8

GREAT PLAINS ENERGY INCORPORATED Consolidated Balance Sheets

(Unaudited)

	September 30 2010	December 31 2009
LIABILITIES AND CAPITALIZATION	(millions, excep	t share amounts)
Current Liabilities		
Notes payable	\$ 22.0	\$ 252.0
Collateralized note payable	95.0	-
Commercial paper	209.5	186.6
Current maturities of long-term debt	335.7	1.3
Accounts payable	172.4	315.0
Accrued taxes	86.8	27.9
Accrued interest	70.5	72.5
Accrued compensation and benefits	44.1	45.1
Pension and post-retirement liability	4.7	4.6
Derivative instruments	25.1	0.3
Other	35.4	53.0
Total	1,101.2	958.3
Deferred Credits and Other Liabilities		
Deferred income taxes	544.3	381.9
Deferred tax credits	134.5	140.5
Asset retirement obligations	139.1	132.6
Pension and post-retirement liability	439.6	440.4
Regulatory liabilities	249.5	237.8
Derivative instruments	-	0.5
Other	135.3	145.1
Total	1,642.3	1,478.8
Capitalization		
Great Plains Energy common shareholders' equity		
Common stock-250,000,000 shares authorized without par value		
136,036,530 and 135,636,538 shares issued, stated value	2,322.5	2,313.7
Retained earnings	660.0	529.2
Treasury stock-385,419 and 213,423 shares, at cost	(8.6)	(5.5)
Accumulated other comprehensive loss	(59.8)	(44.9)
Total	2,914.1	2,792.5
Noncontrolling interest	1.2	1.2
Cumulative preferred stock \$100 par value		
3.80% - 100,000 shares issued	10.0	10.0
4.50% - 100,000 shares issued	10.0	10.0
4.20% - 70,000 shares issued	7.0	7.0
4.35% - 120,000 shares issued	12.0	12.0
Total	39.0	39.0
Long-term debt (Note 10)	3,101.3	3,213.0
Total	6,055.6	6,045.7
Commitments and Contingencies (Note 11)		
Total	\$ 8,799.1	\$ 8,482.8

GREAT PLAINS ENERGY INCORPORATED Consolidated Statements of Income

(Unaudited)

	Three Months Ended September 30		Year to I Septembe	
	2010	2009	2010	2009
Operating Revenues	(mi	llions, except per	r share amounts)	
Electric revenues	\$ 728.8	\$ 587.7	\$1,787.7	\$1,487.4
Operating Expenses			· · · · · · · · · · · · · · · · · · ·	
Fuel	127.3	118.1	333.2	302.5
Purchased power	68.0	46.1	171.4	140.9
Transmission of electricity by others	8.1	7.9	20.9	21.2
Utility operating and maintenance expenses	147.7	145.2	447.3	423.1
Depreciation and amortization	85.3	77.9	248.5	220.3
General taxes	43.6	38.4	119.2	106.8
Other	5.0	2.9	6.5	10.2
Total	485.0	436.5	1,347.0	1,225.0
Operating income	243.8	151.2	440.7	262.4
Non-operating income	7.5	12.7	34.8	37.4
Non-operating expenses	(3.4)	(0.7)	(7.1)	(2.9)
Interest charges	(45.5)	(49.0)	(138.7)	(133.2)
Income before income tax expense and loss from equity investments	202.4	114.2	329.7	163.7
Income tax expense	(70.4)	(35.6)	(112.1)	(26.3)
Loss from equity investments, net of income taxes	· -	(0.2)	(0.9)	(0.4)
Income from continuing operations	132.0	78.4	216.7	137.0
Income (loss) from discontinued operations, net of income taxes (Note 18)	-	0.8	-	(2.3)
Net income	132.0	79.2	216.7	134.7
Less: Net income attributable to noncontrolling interest	-	(0.1)	(0.1)	(0.2)
Net income attributable to Great Plains Energy	132.0	79.1	216.6	134.5
Preferred stock dividend requirements	0.4	0.4	1.2	1.2
Earnings available for common shareholders	\$ 131.6	\$ 78.7	\$ 215.4	\$ 133.3
Average number of basic common shares outstanding	135.2	134.6	135.1	127.5
Average number of diluted common shares outstanding	136.9	134.9	136.8	127.6
Basic earnings (loss) per common share				
Continuing operations	\$ 0.97	\$ 0.57	\$ 1.59	\$ 1.07
Discontinued operations	φ 0. 57	0.01	ψ 1. 55	(0.02)
Basic earnings per common share	\$ 0.97	\$ 0.58	\$ 1.59	\$ 1.05
Zaste carringo per common state	Ψ 0,0,	ψ 0.50	Ψ 1,00	Ψ 1.03
Diluted earnings (loss) per common share				
Continuing operations	\$ 0.96	\$ 0.57	\$ 1.57	\$ 1.06
Discontinued operations	-	0.01	-	(0.02)
Diluted earnings per common share	\$ 0.96	\$ 0.58	\$ 1.57	\$ 1.04
Cash dividends per common share	\$ 0.2075	\$ 0.2075	\$ 0.6225	\$ 0.6225
1 77 7 7 7	* * * * * * * * * * * * * * * * * * * *		*	

GREAT PLAINS ENERGY INCORPORATED Consolidated Statements of Cash Flows

(Unaudited)

Year to Date September 30	2010	2009
Cash Flows from Operating Activities	(millio	ons)
Net income	\$ 216.7	\$ 134.7
Adjustments to reconcile income to net cash from operating activities:		
Depreciation and amortization	248.5	220.3
Amortization of:		
Nuclear fuel	19.7	12.8
Other	(4.1)	(8.0)
Deferred income taxes, net	127.0	(4.8)
Investment tax credit amortization	(1.8)	(1.7)
Loss from equity investments, net of income taxes	0.9	0.4
Other operating activities (Note 2)	(199.2)	(91.9)
Net cash from operating activities	407.7	261.8
Cash Flows from Investing Activities		
Utility capital expenditures	(465.2)	(683.6)
Allowance for borrowed funds used during construction	(26.9)	(28.4)
Payment to Black Hills for asset sale working capital adjustment	-	(7.7)
Purchases of nuclear decommissioning trust investments	(78.3)	(36.1)
Proceeds from nuclear decommissioning trust investments	75.6	33.4
Other investing activities	(9.1)	(2.5)
Net cash from investing activities	(503.9)	(724.9)
Cash Flows from Financing Activities		
Issuance of common stock	4.7	217.9
Issuance of long-term debt	249.9	700.5
Issuance fees	(11.7)	(22.7)
Repayment of long-term debt	(1.3)	(2.0)
Net change in short-term borrowings	(207.1)	(390.2)
Net change in collateralized short-term borrowings	95.0	-
Dividends paid	(85.6)	(82.0)
Other financing activities	(6.0)	(3.0)
Net cash from financing activities	37.9	418.5
Net Change in Cash and Cash Equivalents	(58.3)	(44.6)
Cash and Cash Equivalents at Beginning of Year	65.9	61.1
Cash and Cash Equivalents at End of Period	\$ 7.6	\$ 16.5

GREAT PLAINS ENERGY INCORPORATED Consolidated Statements of Common Shareholders' Equity and Noncontrolling Interest (Unaudited)

Year to Date September 30	201	2010			2009			
	Shares	A	mount	Shares	A	nount		
Common Stock		(milli	ons, except sl					
Beginning balance	135,636,538	\$	2,313.7	119,375,923	\$	2,118.4		
Issuance of common stock	269,855		5.1	15,781,851		218.2		
Common stock issuance fees	-		-	-		(7.0)		
Issuance of restricted common stock	130,137		2.3	376,667		5.4		
Equity compensation expense, net of forfeitures			1.1			0.9		
Unearned Compensation								
Issuance of restricted common stock			(2.3)			(5.4)		
Forfeiture of restricted common stock			0.6			1.0		
Compensation expense recognized			1.9			2.7		
Equity Units allocated fees and expenses and the								
present value of contract adjustment payments			-			(22.5)		
Other			0.1			(0.5)		
Ending balance	136,036,530		2,322.5	135,534,441		2,311.2		
Retained Earnings								
Beginning balance			529.2			489.3		
Net income attributable to Great Plains Energy			216.6			134.5		
Dividends:								
Common stock			(84.4)			(80.8)		
Preferred stock - at required rates			(1.2)			(1.2)		
Performance shares			(0.2)			(0.1)		
Performance shares amendment			-			0.4		
Ending balance			660.0			542.1		
Treasury Stock								
Beginning balance	(213,423)		(5.5)	(120,677)		(3.6)		
Treasury shares acquired	(172,913)		(3.1)	(125,488)		(2.8)		
Treasury shares reissued	917		-	2,758		0.1		
Ending balance	(385,419)		(8.6)	(243,407)		(6.3)		
Accumulated Other Comprehensive Income (Loss)								
Beginning balance			(44.9)			(53.5)		
Derivative hedging activity, net of tax			(15.0)			4.2		
Change in unrecognized pension expense, net of tax			0.1			0.2		
Ending balance			(59.8)			(49.1)		
Total Great Plains Energy Common Shareholders' Equity		\$	2,914.1		\$	2,797.9		
Noncontrolling Interest								
Noncontrolling Interest Beginning balance		\$	1.2		\$	1.0		
		Þ	0.1		Ф			
Net income attributable to noncontrolling interest Distribution			(0.1)			0.2		
Ending balance			1.2		\$	1.2		
Ending balance		Φ	1,4		φ	1,4		

GREAT PLAINS ENERGY INCORPORATED Consolidated Statements of Comprehensive Income (Unaudited)

	Three Months Ended September 30		Year to Septemb	
	2010	2009	2010	2009
		(millio	ns)	
Net income	\$132.0	\$ 79.2	\$216.7	\$134.7
Other comprehensive income (loss)				
Gain (loss) on derivative hedging instruments	(10.3)	(0.6)	(32.4)	0.1
Income tax benefit	4.0	0.2	12.6	
Net gain (loss) on derivative hedging instruments	(6.3)	(0.4)	(19.8)	0.1
Reclassification to expenses, net of tax	2.0	2.1	4.8	4.1
Derivative hedging activity, net of tax	(4.3)	1.7	(15.0)	4.2
Defined benefit pension plans				
Amortization of net gains included in net periodic benefit costs	-	0.1	0.2	0.3
Income tax expense		-	(0.1)	(0.1)
Net change in unrecognized pension expense	-	0.1	0.1	0.2
Comprehensive income	127.7	81.0	201.8	139.1
Less: comprehensive income attributable to noncontrolling interest	-	(0.1)	(0.1)	(0.2)
Comprehensive income attributable to Great Plains Energy	\$127.7	\$ 80.9	\$ 201.7	\$138.9

KANSAS CITY POWER & LIGHT COMPANY Consolidated Balance Sheets

(Unaudited)

	September 30 2010	December 31 2009	
ASSETS			
ASSE1S Current Assets	(millions, except	snare amounts)	
Cash and cash equivalents	\$ 3.0	\$ 17.4	
•	0.2	5 17.4 0.1	
Funds on deposit			
Receivables, net	153.2	161.7	
Accounts receivable pledged as collateral	95.0	-	
Fuel inventories, at average cost	45.5	45.6	
Materials and supplies, at average cost	92.0	84.8	
Deferred refueling outage costs	11.5	19.5	
Refundable income taxes	41.8	-	
Deferred income taxes	4.9	0.3	
Derivative instruments	-	0.2	
Prepaid expenses and other assets	9.1	11.0	
Total	456.2	340.6	
Utility Plant, at Original Cost			
Electric	7,376.8	6,258.5	
Less-accumulated depreciation	3,046.9	2,899.0	
Net utility plant in service	4,329.9	3,359.5	
Construction work in progress	288.2	1,144.1	
Nuclear fuel, net of amortization of \$125.7 and \$106.0	80.2	68.2	
Total	4,698.3	4,571.8	
Investments and Other Assets			
Nuclear decommissioning trust fund	120.7	112.5	
Regulatory assets	645.6	612.1	
Other	38.8	65.3	
Total	805.1	789.9	
Total	\$ 5,959.6	\$ 5,702.3	

KANSAS CITY POWER & LIGHT COMPANY Consolidated Balance Sheets

(Unaudited)

	September 30 2010	December 31 2009		
LIABILITIES AND CAPITALIZATION	(millions, excep	t share amounts)		
Current Liabilities				
Collateralized note payable	\$ 95.0	\$ -		
Commercial paper	209.5	186.6		
Current maturities of long-term debt	0.3	0.2		
Accounts payable	141.5	237.9		
Accrued taxes	62.9	23.7		
Accrued interest	37.0	26.7		
Accrued compensation and benefits	44.1	45.1		
Pension and post-retirement liability	3.2	3.2		
Other	7.7	26.1		
Total	601.2	549.5		
Deferred Credits and Other Liabilities		<u>.</u>		
Deferred income taxes	666.2	559.4		
Deferred tax credits	130.3	135.7		
Asset retirement obligations	125.7	119.8		
Pension and post-retirement liability	420.7	421.2		
Regulatory liabilities	133.2	126.9		
Other	77.2	78.2		
Total	1,553.3	1,441.2		
Capitalization				
Common shareholder's equity				
Common stock-1,000 shares authorized without par value				
1 share issued, stated value	1,563.1	1,563.1		
Retained earnings	500.1	410.1		
Accumulated other comprehensive loss	(37.8)	(41.5)		
Total	2,025.4	1,931.7		
Long-term debt (Note 10)	1,779.7	1,779.9		
Total	3,805.1	3,711.6		
Commitments and Contingencies (Note 11)				
Total	\$ 5,959.6	\$ 5,702.3		

KANSAS CITY POWER & LIGHT COMPANY Consolidated Statements of Income

(Unaudited)

	Three Months Ended September 30				Year to Date September 30			
	2	010	2	009	2	2010	2	009
Operating Revenues				(millio	ns)			
Electric revenues	\$	486.5	\$	395.5	\$	1,194.7	\$	997.8
Operating Expenses								
Fuel		83.8		70.5		213.2		183.3
Purchased power		25.0		19.8		63.2		58.6
Transmission of electricity by others		4.4		3.2		11.3		9.1
Operating and maintenance expenses		102.9		104.2		323.5		301.6
Depreciation and amortization		66.4		59.3		192.2		166.1
General taxes		37.2		32.6		99.3		90.2
Other		3.2		-		3.2		(0.1)
Total		322.9		289.6		905.9		8.808
Operating income		163.6		105.9		288.8		189.0
Non-operating income		6.1		8.6		23.3		24.7
Non-operating expenses		(2.5)		(0.2)		(4.7)		(2.3)
Interest charges		(20.1)		(22.3)		(63.8)		(62.7)
Income before income tax expense		147.1		92.0		243.6		148.7
Income tax expense		(54.5)		(26.4)		(83.6)		(39.8)
Net income	\$	92.6	\$	65.6	\$	160.0	\$	108.9

KANSAS CITY POWER & LIGHT COMPANY Consolidated Statements of Cash Flows

(Unaudited)

Year to Date September 30	2010	2009
Cash Flows from Operating Activities	(million	s)
Net income	\$ 160.0	\$ 108.9
Adjustments to reconcile income to net cash from operating activities:		
Depreciation and amortization	192.2	166.1
Amortization of:		
Nuclear fuel	19.7	12.8
Other	18.1	13.9
Deferred income taxes, net	63.5	(49.6)
Investment tax credit amortization	(1.2)	(1.1)
Other operating activities (Note 2)	(171.1)	32.9
Net cash from operating activities	281.2	283.9
Cash Flows from Investing Activities		
Utility capital expenditures	(349.1)	(520.6)
Allowance for borrowed funds used during construction	(21.3)	(23.6)
Purchases of nuclear decommissioning trust investments	(78.3)	(36.1)
Proceeds from nuclear decommissioning trust investments	75.6	33.4
Net money pool lending	6.0	-
Other investing activities	(6.0)	1.5
Net cash from investing activities	(373.1)	(545.4)
Cash Flows from Financing Activities		
Issuance of long-term debt	-	413.0
Issuance fees	(5.0)	(4.0)
Repayment of long-term debt	(0.2)	-
Net change in short-term borrowings	22.9	(342.2)
Net change in collateralized short-term borrowings	95.0	-
Net money pool borrowings	34.8	-
Dividends paid to Great Plains Energy	(70.0)	(54.0)
Equity contribution from Great Plains Energy		247.5
Net cash from financing activities	77.5	260.3
Net Change in Cash and Cash Equivalents	(14.4)	(1.2)
Cash and Cash Equivalents at Beginning of Year	17.4	5.4
Cash and Cash Equivalents at End of Period	\$ 3.0	\$ 4.2

KANSAS CITY POWER & LIGHT COMPANY Consolidated Statements of Common Shareholder's Equity (Unaudited)

Year to Date September 30		2009		
	Shares	Amount	Shares	Amount
Common Stock		(millions, except s	hare amounts)	
Beginning balance	1	\$ 1,563.1	1	\$ 1,315.6
Equity contribution from Great Plains Energy		-		247.5
Ending balance	1	1,563.1	1	1,563.1
Retained Earnings				
Beginning balance		410.1		353.2
Net income		160.0		108.9
Dividends:				
Common stock held by Great Plains Energy		(70.0)		(54.0
Ending balance		500.1		408.1
Accumulated Other Comprehensive Income (Loss)				
Beginning balance		(41.5)		(46.9
Derivative hedging activity, net of tax		3.7		4.0
Ending balance		(37.8)		(42.9
Total Common Shareholder's Equity		\$ 2,025.4		\$ 1,928.3

KANSAS CITY POWER & LIGHT COMPANY Consolidated Statements of Comprehensive Income (Unaudited)

	Three Months Ended September 30					Year to Septemb		
	2010			009	20	2010		009
•	(millio							
Net income	\$	92.6	\$	65.6	\$	160.0	\$	108.9
Other comprehensive income (loss)								
Gain (loss) on derivative hedging instruments		(0.4)		(0.6)		(1.0)		0.1
Income tax benefit		0.2		0.2		0.4		-
Net gain (loss) on derivative hedging instruments		(0.2)		(0.4)		(0.6)		0.1
Reclassification to expenses, net of tax		1.6		2.0		4.3		3.9
Derivative hedging activity, net of tax		1.4		1.6		3.7		4.0
Comprehensive income	\$	94.0	\$	67.2	\$	163.7	\$	112.9

GREAT PLAINS ENERGY INCORPORATED KANSAS CITY POWER & LIGHT COMPANY

Notes to Unaudited Consolidated Financial Statements

The notes to unaudited consolidated financial statements that follow are a combined presentation for Great Plains Energy Incorporated and Kansas City Power & Light Company, both registrants under this filing. The terms "Great Plains Energy," "Company," "KCP&L," and "Companies" are used throughout this report. "Great Plains Energy" and the "Company" refer to Great Plains Energy Incorporated and its consolidated subsidiaries, unless otherwise indicated. "KCP&L" refers to Kansas City Power & Light Company and its consolidated subsidiaries. "Companies" refers to Great Plains Energy Incorporated and its consolidated subsidiaries and KCP&L and its consolidated subsidiaries. The Companies' interim financial statements reflect all adjustments (which include normal, recurring adjustments) that are necessary, in the opinion of management, for a fair presentation of the results for the interim periods presented.

1. SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES

Organization

Great Plains Energy, a Missouri corporation incorporated in 2001, is a public utility holding company and does not own or operate any significant assets other than the stock of its subsidiaries. Great Plains Energy's wholly owned direct subsidiaries with operations or active subsidiaries are as follows:

- · KCP&L is an integrated, regulated electric utility that provides electricity to customers primarily in the states of Missouri and Kansas. KCP&L has one active wholly owned subsidiary, Kansas City Power & Light Receivables Company (Receivables Company).
- · KCP&L Greater Missouri Operations Company (GMO) is an integrated, regulated electric utility that primarily provides electricity to customers in the state of Missouri. GMO also provides regulated steam service to certain customers in the St. Joseph, Missouri area. GMO wholly owns MPS Merchant Services, Inc. (MPS Merchant), which has certain long-term natural gas contracts remaining from its former non-regulated trading operations.
- · Great Plains Energy Services Incorporated (Services) obtains certain goods and third-party services for its affiliated companies.
- · KLT Inc. is an intermediate holding company that primarily holds investments in affordable housing limited partnerships.

Great Plains Energy's sole reportable business segment is electric utility. See Note 17 for additional information.

Basic and Diluted Earnings per Common Share Calculation

To determine basic EPS, preferred stock dividend requirements and net income attributable to noncontrolling interest are deducted from income from continuing operations and net income before dividing by the average number of common shares outstanding. The earnings (loss) per share impact of discontinued operations is determined by dividing income (loss) from discontinued operations, net of income taxes, by the average number of common shares outstanding. The effect of dilutive securities, calculated using the treasury stock method, assumes the issuance of common shares applicable to performance shares, restricted stock, stock options and Equity Units.

The following table reconciles Great Plains Energy's basic and diluted EPS.

	,	Three Months Ended September 30					to Date mber 30	
		2010 2009			:	2010		2009
Income	(millions, except per share amounts)							
Income from continuing operations	\$	132.0	\$	78.4	\$	216.7	\$	137.0
Less: net income attributable to noncontrolling interest		-		0.1		0.1		0.2
Less: preferred stock dividend requirements		0.4		0.4		1.2		1.2
Income from continuing operations available for common shareholders	\$	131.6	\$	77.9	\$	215.4	\$	135.6
Common Shares Outstanding								
Average number of common shares outstanding		135.2		134.6		135.1		127.5
Add: effect of dilutive securities		1.7		0.3		1.7		0.1
Diluted average number of common shares outstanding		136.9		134.9		136.8		127.6
Basic EPS from continuing operations	\$	0.97	\$	0.57	\$	1.59	\$	1.07
Diluted EPS from continuing operations	\$	0.96	\$	0.57	\$	1.57	\$	1.06

The computation of diluted EPS for the three months ended September 30, 2010, excludes anti-dilutive shares consisting of 107,958 performance shares, 103,114 restricted stock shares and 203,879 stock options. The computation of diluted EPS year to date September 30, 2010, excludes anti-dilutive shares consisting of 116,388 performance shares, 278,452 restricted stock shares and 209,837 stock options.

The computation of diluted EPS for the three months ended September 30, 2009, excludes anti-dilutive shares consisting of 68,310 performance shares, 267,685 restricted stock shares and 247,383 stock options. The computation of diluted EPS year to date September 30, 2009, excludes anti-dilutive shares consisting of 68,310 performance shares, 439,585 restricted stock shares, 247,383 stock options and 5.8 million Equity Units.

Dividends Declared

In October 2010, Great Plains Energy's Board of Directors (Board) declared a quarterly dividend of \$0.2075 per share on Great Plains Energy's common stock. The common dividend is payable on December 20, 2010, to shareholders of record as of November 29, 2010. The Board also declared regular dividends on Great Plains Energy's preferred stock, payable on March 1, 2011, to shareholders of record as of February 7, 2011.

In October 2010, KCP&L's Board of Directors declared a cash dividend payable to Great Plains Energy of \$25 million payable on December 17, 2010.

2. SUPPLEMENTAL CASH FLOW INFORMATION

Great Plains Energy Other Operating Activities

Year to Date September 30	2010	2009		
Cash flows affected by changes in:	(mill	ions)		
Receivables	\$ (45.9)	\$	7.9	
Accounts receivable pledged as collateral	(95.0)		-	
Fuel inventories	4.9		(3.1)	
Materials and supplies	(10.0)		(16.1)	
Accounts payable	(79.1)		(122.2)	
Accrued taxes	54.6	85.7		
Accrued interest	(2.0)		1.9	
Deferred refueling outage costs	8.0		4.3	
Accrued plant maintenance costs	1.9		2.2	
Fuel adjustment clauses	(2.3)		3.1	
Pension and post-retirement benefit obligations	15.5		43.0	
Allowance for equity funds used during construction	(25.5)		(29.6)	
Forward Starting Swaps settlement	(6.9)		(79.1)	
Other	(17.4)		10.1	
Total other operating activities	\$ (199.2)	\$	(91.9)	
Cash paid during the period:				
Interest	\$ 175.4	\$	162.8	
Income taxes	\$ 0.4	\$	4.7	
Non-cash investing activities:				
Liabilities assumed for capital expenditures	\$ 36.4	\$	56.8	

KCP&L Other Operating Activities

Year to Date September 30		2010		2009	
Cash flows affected by changes in:	· ·				
Receivables	\$	(0.1)	\$	6.7	
Accounts receivable pledged as collateral		(95.0)		-	
Fuel inventories		0.1		0.6	
Materials and supplies		(7.2)		(11.4)	
Accounts payable		(87.2)		(49.3)	
Accrued taxes		(6.3)		96.4	
Accrued interest		10.3		18.9	
Deferred refueling outage costs		8.0		4.3	
Pension and post-retirement benefit obligations		31.1		51.5	
Allowance for equity funds used during construction		(21.7)		(22.6)	
Kansas Energy Cost Adjustment		(5.4)		(0.2)	
Forward Starting Swaps settlement		-		(79.1)	
Other		2.3		17.1	
Total other operating activities	\$	(171.1)	\$	32.9	
Cash paid during the period:					
Interest	\$	60.3	\$	44.0	
Income taxes	\$	71.6	\$	-	
Non-cash investing activities:					
Liabilities assumed for capital expenditures	\$	31.6	\$	55.2	

On January 1, 2010, Great Plains Energy and KCP&L adopted new accounting guidance for transfers of financial assets, which resulted in the recognition of \$95.0 million of accounts receivables pledged as collateral and a corresponding short-term collateralized note payable on Great Plains Energy's and KCP&L's balance sheets at September 30, 2010. See Note 3 for additional information. As a result, cash flows from operating activities were reduced by \$95.0 million and cash flows from financing activities were raised by \$95.0 million with no impact to the net change in cash year to date September 30, 2010.

3. RECEIVABLES

Great Plains Energy's and KCP&L's receivables are detailed in the following table.

	Se	otember 30	Dec	cember 31	
		2010		2009	
Great Plains Energy		(millions)			
Customer accounts receivable - billed	\$	118.3	\$	47.3	
Customer accounts receivable - unbilled		80.3		77.9	
Allowance for doubtful accounts		(5.0)		(2.8)	
Other receivables		81.0		108.1	
Total	\$	274.6	\$	230.5	
KCP&L					
Customer accounts receivable - billed	\$	38.2	\$	-	
Customer accounts receivable - unbilled		52.0		44.6	
Allowance for doubtful accounts		(2.6)		(1.7)	
Intercompany receivables		12.7		42.4	
Other receivables		52.9		76.4	
Total	\$	153.2	\$	161.7	

Great Plains Energy's and KCP&L's other receivables at September 30, 2010, and December 31, 2009, consisted primarily of receivables from partners in jointly owned electric utility plants and wholesale sales receivables.

Sale of Accounts Receivable - KCP&L

KCP&L sells all of its retail electric accounts receivable to its wholly owned subsidiary, Receivables Company, which in turn sells an undivided percentage ownership interest in the accounts receivable to Victory Receivables Corporation, an independent outside investor. On January 1, 2010, Great Plains Energy and KCP&L adopted new accounting guidance for transfers of financial assets, which resulted in the sale of the undivided percentage ownership interest in accounts receivable by Receivables Company no longer meeting the criteria for derecognition and now being accounted for as a secured borrowing. As a result, \$95.0 million of accounts receivables pledged as collateral are recognized with a corresponding short-term collateralized note payable on Great Plains Energy's and KCP&L's balance sh eets at September 30, 2010.

KCP&L sells its receivables at a fixed price based upon the expected cost of funds and charge-offs. These costs comprise KCP&L's loss on the sale of accounts receivable. KCP&L services the receivables and receives an annual servicing fee of 1.5% to 2.5% of the outstanding principal amount of the receivables sold to Receivables Company. KCP&L does not recognize a servicing asset or liability because management determined the collection agent fee earned by KCP&L approximates market value. In May 2010, the term of the agreement was extended to May 2011.

Three Months Ended September 30, 2010	KCP&L			Receivables Company		lidated P&L
Time Frontis Ended September 50, 2010	IC	IGL		lions)	II.C.	- CL
Receivables (sold) purchased	\$	(442.0)	\$	442.0	\$	-
Gain (loss) on sale of accounts receivable (a)		(5.6)		5.4		(0.2)
Servicing fees		0.8		(0.8)		-
Fees to outside investor		-		(0.3)		(0.3)
Cash flows during the period						
Cash from customers transferred to Receivables Company		(430.9)		430.9		-
Cash paid to KCP&L for receivables purchased		425.5		(425.5)		-
Servicing fees		8.0		(8.0)		-
Interest on intercompany note		0.2		(0.2)		_

		Receivables	Consolidated	
Year to Date September 30, 2010	KCP&L	Company	KCP&L	
		(millions)		
Receivables (sold) purchased	\$ (1,067.7)	\$ 1,067.7	\$ -	
Gain (loss) on sale of accounts receivable (a)	(13.5)	12.9	(0.6)	
Servicing fees	1.9	(1.9)	-	
Fees to outside investor	-	(0.9)	(0.9)	
Cash flows during the period				
Cash from customers transferred to Receivables Company	(1,029.8)	1,029.8	-	
Cash paid to KCP&L for receivables purchased	1,016.9	(1,016.9)	-	
Servicing fees	1.9	(1.9)	-	
Interest on intercompany note	0.4	(0.4)	-	

			Receivables Company		Consolidated KCP&L				
Three Months Ended September 30, 2009	K	CP&L							
	(millions)								
Receivables (sold) purchased	\$	(354.2)	\$	354.2	\$	-			
Gain (loss) on sale of accounts receivable ^(a)		(4.5)		4.4		(0.1)			
Servicing fees		0.9		(0.9)		-			
Fees to outside investor		-		(0.4)		(0.4)			
Cash flows during the period									
Cash from customers transferred to Receivables Company		(352.5)		352.5		-			
Cash paid to KCP&L for receivables purchased		348.1		(348.1)		-			
Servicing fees		0.9		(0.9)		-			
Interest on intercompany note		0.1		(0.1)		-			

Year to Date September 30, 2009	KO	KCP&L		Receivables Company		lidated P&L
			(mil	lions)		
Receivables (sold) purchased	\$	(892.0)	\$	892.0	\$	-
Gain (loss) on sale of accounts receivable (a)		(11.3)		11.0		(0.3)
Servicing fees		2.4		(2.4)		-
Fees to outside investor		-		(0.9)		(0.9)
Cash flows during the period						
Cash from customers transferred to Receivables Company		(880.3)		880.3		-
Cash paid to KCP&L for receivables purchased		869.3		(869.3)		-
Servicing fees		2.4		(2.4)		-
Interest on intercompany note		0.3		(0.3)		-

⁽a) Any net gain (loss) is the result of the timing difference inherent in collecting receivables and over the life of the agreement will net to zero.

4. ASSETS HELD FOR SALE

Great Plains Energy has several real estate properties that will not be used. As a result, these real estate properties are available for immediate sale in their present condition and management is actively marketing these properties. The carrying amounts for these assets are presented at fair value less estimated selling cost and are included in assets held for sale on Great Plains Energy's balance sheets. In March 2010, one of the properties with a book value of \$0.6 million was sold resulting in an insignificant loss on the sale. Of the \$18.8 million of assets held for sale at September 30, 2010, \$14.4 million is included in the electric utility segment and the remaining \$4.4 million is included in the other category. In October 2010, one of the properties included in the electric utility segment with a book value of \$11.1 million was sold resulting in an insignificant gain on the sale.

5. NUCLEAR PLANT

KCP&L owns 47% of Wolf Creek, its only nuclear generating unit. Wolf Creek is regulated by the Nuclear Regulatory Commission (NRC), with respect to licensing, operations and safety-related requirements.

Spent Nuclear Fuel and High-Level Radioactive Waste

Under the Nuclear Waste Policy Act of 1982, the Department of Energy (DOE) is responsible for the permanent disposal of spent nuclear fuel. KCP&L pays the DOE a quarterly fee of one-tenth of a cent for each kWh of net nuclear generation delivered and sold for the future disposal of spent nuclear fuel. These disposal costs are charged to fuel expense. In March 2010, the DOE filed a motion to withdraw its application to the NRC to construct a national repository for the disposal of spent nuclear fuel and high-level radioactive waste at Yucca Mountain, Nevada, which would bring the licensing process to an end. An NRC board denied the DOE's motion to withdraw its application in June 2010, and the DOE appealed that decision to the full NRC in early July 2010. The NRC has not yet decided that appeal. The question of DOE's legal authority to withdraw its license application also is pending in multiple lawsuits filed with a federal appellate court. However, that court has put those cases on hold pending a final decision from the NRC. Wolf Creek has an on-site storage facility designed to hold all spent fuel generated at the plant through 2025, and believes it will be able to expand on-site storage as needed past 2025. Management cannot predict when, or if, an alternative disposal site will be available to receive Wolf Creek's spent nuclear fuel and will continue to monitor this activity. See Note 12 for a related legal proceeding.

Low-Level Radioactive Waste

Wolf Creek disposes of most of its low-level radioactive waste (Class A waste) at an existing third-party repository in Utah. Management expects that the site located in Utah will remain available to Wolf Creek for disposal of its Class A waste. Wolf Creek has contracted with a waste processor that will process, take title and

store in another state most of the remainder of Wolf Creek's low level radioactive waste (Classes B and C waste, which is higher in radioactivity but much lower in volume). Should on-site waste storage be needed in the future, Wolf Creek has current storage capacity on site for about four years' generation of Classes B and C waste and believes it will be able to expand that storage capacity as needed if it becomes necessary to do so.

Nuclear Decommissioning Trust Fund

The following table summarizes the change in Great Plains Energy's and KCP&L's decommissioning trust fund.

	Sep	Dec	cember 31 2009				
Decommissioning Trust		(millions)					
Beginning balance January 1	\$	112.5	\$	96.9			
Contributions		2.7		3.7			
Earned income, net of fees		1.2		2.8			
Net realized gains/(losses)		6.6		(5.5)			
Net unrealized gains/(losses)		(2.3)		14.6			
Ending balance	\$	120.7	\$	112.5			

The decommissioning trust is reported at fair value on the balance sheets and is invested in assets as detailed in the following table. At December 31, 2009, KCP&L was holding short-term investments in the decommissioning trust fund, which were invested in equity securities in early 2010 as a result of a change in the asset allocation of the trust to a higher proportion of equity securities given the 20-year extension of Wolf Creek's operating license approved by the NRC in November 2008.

	September 30 2010								December 31 2009								
	Cost asis	Unre	Gross Gross realized Unrealized Gains Losses		ealized		air alue		Cost Basis	Gross Unrealized Gains		Gross Unrealized Losses			Fair ⁄alue		
							(mil	lions)									
Equity securities	\$ 72.4	\$	6.4	\$	(2.2)	\$	76.6	\$	36.3	\$	8.9	\$	(0.7)	\$	44.5		
Debt securities	37.8		3.8		-		41.6		35.3		2.1		-		37.4		
Other	2.5		-		-		2.5		30.6		-		-		30.6		
Total	\$ 112.7	\$	10.2	\$	(2.2)	\$	120.7	\$	102.2	\$	11.0	\$	(0.7)	\$	112.5		

The weighted average maturity of debt securities held by the trust at September 30, 2010, was approximately 7.7 years. The costs of securities sold are determined on the basis of specific identification. The following table summarizes the realized gains and losses from the sale of securities by the nuclear decommissioning trust fund.

	T	Three Months Ended September 30				Year to Date			
					September 30				
	:	2010	2009		2010		2009		
				(millio	ns)				
Realized gains	\$	0.2	\$	0.5	\$	7.2	\$	1.2	
Realized losses		(0.1)		(0.9)		(0.6)		(7.6)	

6. REGULATORY MATTERS

Regulatory Proceedings

The following table summarizes the initial filing information in currently pending requests for retail rate increases with The State Corporation Commission of the State of Kansas (KCC) and the Public Service Commission of the State of Missouri (MPSC).

Rate Jurisdiction	File Date	Annual Revenue Increase	Return on Equity	Rate-Making Equity Ratio
		(millions)		
KCP&L - Kansas (a)	12/17/2009	\$ 55.2 (b)	11.25% (b)	46.17%
KCP&L - Missouri (c)	6/4/2010	92.1	11.00%	46.16%
GMO - Missouri Public Service division (c)	6/4/2010	75.8	11.00%	46.16%
GMO - St. Joseph Light & Power division (c)	6/4/2010	22.1	11.00%	46.16%

- (a) The request includes costs related to Iatan No. 2, a new coal-fired generation unit, upgrades to the transmission and distribution system to improve reliability and overall increased costs of service. Any authorized changes to retail rates are expected to be effective in December 2010.
- (b)The requested increase was adjusted by KCP&L during the hearing process to \$50.9 million as the net result of updates to the case, most notably a reduction in the requested return on equity. The requested return on equity was adjusted by KCP&L to 10.75% with the potential for a 0.25% adder (to 11.00%) if KCC adopts a particular rate design proposal by KCC Staff and other interveners.
- (c) The request includes costs related to Iatan No. 2, a new coal-fired generation unit, upgrades to the transmission and distribution system to improve reliability and overall increased costs of service. For KCP&L, it also includes increased coal transportation costs expected with the expiration in 2010 of the majority of KCP&L's current coal transportation contracts. Any authorized changes to retail rates are expected to be effective in May 2011 for KCP&L and June 2011 for GMO.

In June 2010, KCC staff and certain interveners filed their direct testimony in KCP&L's rate case. KCC staff recommended a rate reduction of \$9.1 million. The main differences between KCP&L's and KCC staff's positions are the staff's proposals for a return on equity of 9.70%, disallowance of \$231 million of the total project cost of Iatan No. 2 (\$58 million for KCP&L's Kansas jurisdictional portion of the plant) and depreciation rate differences. KCP&L supported a recommended disallowance of certain Iatan No. 2 construction costs in its post-hearing brief filed in September 2010. Great Plains Energy recorded a \$4.0 million pre-tax loss in the third quarter of 2010 for KCP&L's and GMO's combined share of these Iatan No. 2 construction costs. Management does not believe a further write-down of the Iatan No. 2 plant is appropriate under regulatory prudence standards. In the event of a further disallowance of certain Iatan No. 2 costs, KCP&L would recognize a loss with a corresponding write-down of utility plant for the amount of disallowance. The outcome of the KCP&L Kansas case will likely be different from either of the positions of KCP&L or KCC staff, though the decision of KCC cannot be predicted. Evidentiary hearings in the case were held August 16, 2010, through September 2, 2010. The case has been fully briefed and a KCC order is expected in November 2010 with new rates expected to be effective in December 2010.

In September 2010, GMO received an order from the MPSC approving construction accounting for the Iatan No. 2 project from the Iatan No. 2 in-service date to the effective date of new rates in the current rate case. The effect of the order is to defer GMO's share of Iatan No. 2 operating costs, depreciation expense and carrying costs (interest) to a regulatory asset rather than impacting the income statement until new rates are effective. KCP&L (Missouri jurisdiction only) was granted construction accounting as part of the Comprehensive Energy Plan. The next major milestone in the MPSC cases is November 2010, when the MPSC staff will file its Iatan No. 2 prudence report and the MPSC Staff and other interveners will file direct testimony. Hearings are scheduled for late January 2011 for KCP&L and early February 2011 for GMO. New rates are proposed to go into effect in May 2011 for KCP&L and June 2011 for GMO.

KCP&L's Comprehensive Energy Plan - Iatan No. 2

In April 2010, Great Plains Energy and KCP&L announced the results of a cost and schedule reforecast for Iatan No. 2. The current and previous cost estimate ranges are shown in the following table. The cost estimate ranges do not include allowance for funds used during construction or the cost of common facilities that were identified at the time of the start-up of the Iatan No. 1 environmental project that will be used by both Iatan No. 1 and Iatan No. 2.

		nt Es Rang	timate e		us Es Rang	stimate e	C	hang	e	
				(mil	lions)				_
Great Plains Energy's 73% share of Iatan No. 2	\$ 1,222	-	\$ 1,251	\$ 1,153	-	\$ 1,201	\$ 69	-	\$ 50)
KCP&L's 55% share of Iatan No. 2	919	-	941	868	-	904	51	-	37	7

The increase in the cost estimate ranges was primarily due to a shift in the expected in-service date, the impact of lower wholesale prices on expected test power revenues that offset construction cost, and a level of contingency management considered appropriate in light of recent start-up events encountered at other coal plants under construction. In August 2010, Iatan No. 2 successfully completed in-service testing, which was confirmed by KCC in October 2010, but is still subject to confirmation by the MPSC, which is expected during the current rate cases. The Companies estimate that the final cost will be within the current estimate range.

SPP and NERC Audits

In November 2009, the Southwest Power Pool, Inc. (SPP) and the North American Electric Reliability Corporation (NERC) conducted scheduled audits of KCP&L and GMO regarding compliance with NERC reliability and critical infrastructure protection standards. KCP&L and GMO have received the final audit report alleging violation of certain standards, which could result in penalties. The timing and amount of such penalties that may be proposed are unknown at this time. The SPP also conducted a compliance inquiry regarding a transmission system outage that occurred in the St. Joseph, Missouri area in the summer of 2009. NERC may also investigate the circumstances surrounding this transmission system outage. The outcome of the outage inquiries cannot be predicted at this time.

Energy Efficiency

Great Plains Energy and KCP&L have implemented various energy efficiency programs, and are evaluating expanded and new energy efficiency programs as one of the elements to meet future customer energy needs. KCP&L also agreed in the Collaboration Agreement to pursue initiatives, including energy efficiency, designed to offset CO₂ emissions. The Companies currently recover energy efficiency program expenses on a deferred basis. While there is an ongoing rulemaking proceeding in Missouri, and a pending KCC case filed by KCP&L to address recovery of and earnings on investments in energy efficiency programs, until these rules are set and programs are approved, the effects on Great Plains Energy's and KCP&L's plans and future results cannot be reasonably estimated.

MPSC Regulatory Approval of the GMO Acquisition

The MPSC order approving the GMO acquisition was received on July 1, 2008. Certain parties filed appeals and a motion to stay the order with the Cole County, Missouri, Circuit Court, which affirmed the order in June 2009. That decision was appealed and in August 2010, the Missouri Court of Appeals, Western District, upheld the MPSC order. In September 2010, certain parties filed motions for rehearing by the Missouri Court of Appeals and applications for transfer to the Missouri Supreme Court.

GMO Missouri 2007 Rate Case Appeal

Appeals of the May 2007 MPSC order approving an approximate \$59 million increase in annual revenues were filed in July and August of 2007 with the Circuit Court of Cole County, Missouri, by the Office of Public Counsel, AG Processing, Sedalia Industrial Energy Users' Association and AARP seeking to set aside or remand the order of the MPSC. In February 2009, the Circuit Court affirmed the MPSC order. The Circuit Court's

decision was affirmed by the Court of Appeals in August 2009. The case was transferred to the Missouri Supreme Court in August 2010. The order remains in effect unless reversed by the courts.

Regulatory Assets and Liabilities

Great Plains Energy's and KCP&L's regulatory assets and liabilities are detailed in the following tables.

			Great
September 30, 2010	KCP&L	GMO	Plains Energy
Regulatory Assets		(millions)	
Taxes recoverable through future rates	\$ 112.6	\$ 26.4	\$ 139.0
Loss on reacquired debt	5.1 (a)	0.8 (a)	5.9
Cost of removal	7.5	-	7.5
Asset retirement obligations	26.6	12.6	39.2
Pension settlements	10.1 (b)	-	10.1
Pension and post-retirement costs	366.9 (c)	103.0 (c)	469.9
Deferred customer programs	42.5 (d)	13.2	55.7
Rate case expenses	10.9 (e)	2.3 (e)	13.2
Skill set realignment costs	5.1 (f)	-	5.1
Fuel adjustment clauses	5.2 (e)	45.1 (e)	50.3
Acquisition transition costs	29.5 (g)	22.5 (g)	52.0
St. Joseph Light & Power acquisition	-	2.7 (h)	2.7
Storm damage	-	3.6 (i)	3.6
Derivative instruments	-	4.8 (j)	4.8
Iatan No. 1 and Common facilities depreciation and carrying costs	13.3 (k)	3.5 (k)	16.8
Iatan No. 2 construction accounting costs	6.4 (k)	2.2 (k)	8.6
Other	3.9 (l)	0.5 (l)	4.4
Total	\$ 645.6	\$ 243.2	\$ 888.8
Regulatory Liabilities			
Emission allowances	\$ 86.1	\$ 0.6	\$ 86.7
Asset retirement obligations	37.8	-	37.8
Pension	-	36.3	36.3
Cost of removal	-	63.2 (m)	63.2
Other	9.3	16.2	25.5
Total	\$ 133.2	\$ 116.3	\$ 249.5

- (a) Amortized over the life of the related new debt issuances or the remaining lives of the old debt issuances if no new debt was issued.
- (b) \$5.6 million not included in rate base and amortized through 2012.
- (c) Represents the funded status of the pension plans more than offset by related liabilities. Also represents financial and regulatory accounting method differences not included in rate base that will be eliminated over the life of the pension plans.
- (d) \$13.2 million not included in rate base and amortized over various periods.
- (e) Not included in rate base and amortized over various periods.
- (f) \$2.9 million not included in rate base and amortized through 2017.
- (g) Not included in rate base. The MPSC order provided for the deferral of transition costs to be amortized over a five-year period to the extent that synergy savings exceed transition cost amortization. The Company settled its first post-transaction rate cases and the settlement agreements did not address transition costs. The Company will continue to defer transition costs until amortization is ordered by the MPSC. KCC order approved the deferral of up to \$10.0 million of transition costs to be amortized over a five-year period beginning with rates expected to be effective in December 2010.
- (h) Not included in rate base and amortized through 2015.
- (i) Not included in rate base and amortized through 2012.
- (j) Represents the fair value of derivative instruments for commodity contracts. Settlements of the contracts are recognized in fuel expense and included in GMO's fuel adjustment clause (FAC).

- (k) Not included in rate base.
- (l) Certain insignificant items are not included in rate base and amortized over various periods.
- (m) Estimated cumulative net provision for future removal costs.

						Great
December 31, 2009	K	CP&L		GMO	Plai	ns Energy
Regulatory Assets			(n	nillions)		
Taxes recoverable through future rates	\$	77.6	\$	22.9	\$	100.5
Loss on reacquired debt		5.3		0.3		5.6
Cost of removal		7.9		-		7.9
Asset retirement obligations		23.8		11.9		35.7
Pension settlements		13.5		-		13.5
Pension and post-retirement costs		395.0		84.5		479.5
Deferred customer programs		35.6		7.1		42.7
Rate case expenses		7.4		1.5		8.9
Skill set realignment costs		6.1		-		6.1
Fuel adjustment clauses		0.7		47.5		48.2
Acquisition transition costs		29.3		22.2		51.5
St. Joseph Light & Power acquisition		-		3.1		3.1
Storm damage		-		4.8		4.8
Derivative instruments		-		2.1		2.1
Iatan No. 1 and Common facilities depreciation and carrying costs		4.6		1.4		6.0
Other		5.3		8.0		6.1
Total	\$	612.1	\$	210.1	\$	822.2
Regulatory Liabilities						
Emission allowances	\$	86.2	\$	8.0	\$	87.0
Asset retirement obligations		33.4		-		33.4
Pension		-		34.0		34.0
Cost of removal		-		62.5		62.5
Other		7.3		13.6		20.9
Total	\$	126.9	\$	110.9	\$	237.8

7. PENSION PLANS AND OTHER EMPLOYEE BENEFITS

Great Plains Energy maintains defined benefit pension plans for substantially all active and inactive employees, including officers, of KCP&L, GMO, and Wolf Creek Nuclear Operating Corporation (WCNOC) and incurs significant costs in providing the plans. Pension benefits under these plans reflect the employees' compensation, years of service and age at retirement.

KCP&L and GMO record pension expense in accordance with rate orders from the MPSC and KCC that allow the difference between pension costs under generally accepted accounting principles (GAAP) and pension costs for ratemaking to be recognized as a regulatory asset or liability. This difference between financial and regulatory accounting methods is due to timing and will be eliminated over the life of the pension plans.

In addition to providing pension benefits, Great Plains Energy provides certain post-retirement health care and life insurance benefits for substantially all retired employees of KCP&L, GMO, and WCNOC. The cost of post-retirement benefits charged to KCP&L and GMO are accrued during an employee's years of service and recovered through rates.

The following tables provide Great Plains Energy's components of net periodic benefit costs prior to the effects of capitalization and sharing with joint-owners of power plants.

	Pension 1	Benefits	Other E	Other Benefits		
Three Months Ended September 30	2010	2009	2010	2009		
Components of net periodic benefit costs		(mill	ions)			
Service cost	\$ 7.6	\$ 7.3	\$ 1.0	\$ 1.1		
Interest cost	12.3	11.8	2.2	2.0		
Expected return on plan assets	(9.2)	(8.1)	(0.5)	(0.4)		
Prior service cost	1.1	1.0	1.8	1.9		
Recognized net actuarial loss (gain)	9.4	9.1	(0.1)	(0.1)		
Transition obligation	-	-	0.3	0.3		
Net periodic benefit costs before						
regulatory adjustment	21.2	21.1	4.7	4.8		
Regulatory adjustment	(8.1)	(7.2)	-	-		
Net periodic benefit costs	\$ 13.1	\$ 13.9	\$ 4.7	\$ 4.8		

	Pension Benefits		Other Benefits		
Year to Date September 30	2010	2009	2010	2009	
Components of net periodic benefit costs		(milli	ions)		
Service cost	\$ 22.8	\$ 21.8	\$ 2.8	\$ 3.1	
Interest cost	36.9	35.4	6.6	6.2	
Expected return on plan assets	(27.5)	(24.1)	(1.6)	(1.2)	
Prior service cost	3.5	3.0	5.4	5.2	
Recognized net actuarial loss (gain)	28.1	27.3	(0.1)	(0.3)	
Transition obligation	-	-	1.0	1.0	
Settlement charge	-	0.1	-	-	
Net periodic benefit costs before					
regulatory adjustment	63.8	63.5	14.1	14.0	
Regulatory adjustment	(24.6)	(20.3)	-	(0.2)	
Net periodic benefit costs	\$ 39.2	\$ 43.2	\$ 14.1	\$ 13.8	

Year to date September 30, 2010, the Company contributed \$38.5 million to the pension plans and expects to contribute an additional \$29.8 million in 2010 to satisfy the ERISA minimum funding requirements and the MPSC and KCC rate orders, the majority of which is expected to be paid by KCP&L.

On March 23, 2010, the Patient Protection and Affordable Care Act, a comprehensive health care reform bill took effect. Management expects a minimal impact as a result of this new legislation in the short-term but will continue to monitor for any long-term impacts. Year to date September 30, 2010, Great Plains Energy and KCP&L recorded a \$2.8 million increase in income tax expense for the cumulative change in tax treatment of the Medicare Part D subsidy under this new legislation.

8. EQUITY COMPENSATION

Great Plains Energy's Long-Term Incentive Plan is an equity compensation plan approved by Great Plains Energy's shareholders. The Long-Term Incentive Plan permits the grant of restricted stock, stock options, limited stock appreciation rights, director shares, director deferred share units and performance shares to directors, officers and other employees of Great Plains Energy and KCP&L. Forfeiture rates are based on historical forfeitures and future expectations and are reevaluated annually.

The following table summarizes Great Plains Energy's and KCP&L's equity compensation expense and associated income tax benefits.

		Three Months Ended September 30			Year to Date September 3			
	20	10	20	09	20	10	20	09
Great Plains Energy				(millio	ons)			
Compensation expense	\$	8.0	\$	1.7	\$	3.4	\$	4.2
Income tax benefits		0.2		0.5		8.0		1.0
KCP&L								
Compensation expense		0.6		1.2		2.4		2.8
Income tax benefits		0.2		0.4		0.4		0.5

Performance Shares

Performance share activity year to date September 30, 2010, is summarized in the following table.

	Performance Shares	Grant Date Fair Value*
Beginning balance	294,641	\$ 13.62
Performance adjustment	(21,674)	
Granted	231,598	23.37
Earned	(8,433)	10.87
Forfeited	(44,168)	21.33
Ending balance	451,964	18.05

^{*} weighted-average

Performance adjustment represents the number of shares of common stock related to performance shares ultimately issued that can vary from the number of performance shares initially granted depending on Great Plains Energy's performance, based on internal and external measures, over stated performance periods.

The fair value of performance share awards is estimated using a Monte Carlo simulation technique that uses the closing stock price at the valuation date and incorporates assumptions for inputs of expected volatilities, dividend yield and risk-free rates. Expected volatility is based on daily stock price change during a historical period commensurate with the remaining term of the performance period of the grant. The risk-free rate is commensurate with the remaining life of the performance period of the grant based on the zero-coupon government bonds in effect at the time of the valuation. The dividend yield is based on the most recent dividends paid and the actual closing stock price on the valuation date. For shares granted year to date September 30, 2010, inputs for expected volatility, divide nd yield and risk-free rates were 31%, 4.65%, and 1.2%, respectively.

At September 30, 2010, the remaining weighted-average contractual term was 1.6 years. There were no performance shares granted for the three months ended September 30, 2010 and 2009. The weighted-average grant-date fair value of shares granted year to date September 30, 2010, was \$23.37. The weighted-average grant-date fair value of shares granted year to date September 30, 2009, was \$14.35. At September 30, 2010, there was \$3.7 million of total unrecognized compensation expense, net of forfeiture rates, related to performance shares granted under the Long-Term Incentive Plan, which will be recognized over the remaining weighted-average contractual term. The total fair value of performance shares earned and paid year to date September 30, 2010, was insignificant. There were no p erformance shares earned and paid for the three months ended and year to date September 30, 2009.

Restricted Stock

Restricted stock activity year to date September 30, 2010, is summarized in the following table.

	Nonvested Restricted Stock	Grant Date Fair Value*
Beginning balance	612,587	\$ 20.24
Granted and issued	130,137	17.80
Vested	(291,787)	25.00
Forfeited	(29,760)	19.05
Ending balance	421,177	16.21

^{*} weighted-average

At September 30, 2010, the remaining weighted-average contractual term was 1.5 years. The weighted-average grant-date fair value of shares granted for the three months ended and year to date September 30, 2010, was \$18.32 and \$17.80, respectively. The weighted-average grant-date fair value of shares granted for the three months ended and year to date September 30, 2009, was \$18.03 and \$14.50, respectively. At September 30, 2010, there was \$3.4 million of total unrecognized compensation expense, net of forfeiture rates, related to nonvested restricted stock granted under the Long-Term Incentive Plan, which will be recognized over the remaining weighted-average contractual term. The total fair value of shares vested for the three months ended and year to date September 30, 2010, was \$0.9 million and \$7.3 million, respectively. The total fair value of shares vested for the three months ended and year to date September 30, 2009, was insignificant and \$5.4 million, respectively.

Stock Options

Stock options were granted under the Long-Term Incentive Plan during 2001-2003 and GMO stock options outstanding on the July 14, 2008, acquisition date were converted to Great Plains Energy stock options. All stock options are fully vested at September 30, 2010. Stock option activity under all plans year to date September 30, 2010, is summarized in the following table.

Stock Options	Shares	Exercise Price*
Beginning balance	244,610	\$ 36.73
Exercised	(917)	9.21
Forfeited or expired	(29,681)	62.17
Outstanding and exercisable at September 30, 2010	214,012	33.32

^{*} weighted-average

There were no stock options exercised during the three months ended September 30, 2010. The weighted-average grant-date fair value of options exercised year to date September 30, 2010, was \$9.21. The weighted-average grant-date fair value of options exercised for the three months ended and year to date September 30, 2009, was \$10.65 and \$10.94 per share, respectively. The aggregate intrinsic value and cash received for options exercised for the three months ended and year to date September 30, 2010 and 2009, was insignificant. At September 30, 2010, there were no in-the-money outstanding and exercisable options.

The following table summarizes all outstanding and exercisable stock options as of September 30, 2010.

Outstanding and Exercisable Options

		Weighted Average	
		Remaining	Weighted
Exercise	Number of	Contractual Life	Average
Price Range	Shares	in Years	Exercise Price
\$23.91 - \$27.73	203,656	1.2	\$ 24.51
\$181.11	3,998	0.3	181.11
\$221.82 - \$251.86	6,358	0.6	222.33
Total	214,012	1.1	33.32

9. SHORT-TERM BORROWINGS AND SHORT-TERM BANK LINES OF CREDIT

Great Plains Energy's \$200 Million Revolving Credit Facility

In August 2010, Great Plains Energy entered into a new \$200 million revolving credit facility with a group of banks that expires in August 2013. The facility replaced a \$400 million revolving credit facility with a group of banks that would have expired in May 2011 and was sized based on expected short-term debt requirements over the facility's term. The facility's terms permit transfers of unused commitments between this facility and the KCP&L and GMO facilities discussed below, with the total amount of the facility not exceeding \$400 million at any one time. A default by Great Plains Energy or any of its significant subsidiaries on other indebtedness totaling more than \$50.0 million is a default under the facility. Under the terms of this facility, Great Plains Energy is requir ed to maintain a consolidated indebtedness to consolidated capitalization ratio, as defined in the facility, not greater than 0.65 to 1.00 at all times. At September 30, 2010, Great Plains Energy was in compliance with this covenant. At September 30, 2010, Great Plains Energy had \$22.0 million of outstanding cash borrowings with a weighted-average interest rate of 3.06% and had issued letters of credit totaling \$15.8 million under the credit facility. At December 31, 2009, Great Plains Energy had \$20.0 million of outstanding cash borrowings with a weighted-average interest rate of 0.68% and had issued letters of credit totaling \$25.4 million under the credit facility.

KCP&L's \$600 Million Revolving Credit Facility

In August 2010, KCP&L entered into a new \$600 million revolving credit facility with a group of banks to provide support for its issuance of commercial paper and other general corporate purposes that expires in August 2013. The facility replaced a \$600 million facility with a group of banks that would have expired in May 2011. Great Plains Energy and KCP&L may transfer up to \$200 million of unused commitments between Great Plains Energy's and KCP&L's facilities. A default by KCP&L on other indebtedness totaling more than \$50.0 million is a default under the facility. Under the terms of this facility, KCP&L is required to maintain a consolidated indebtedness to consolidated capitalization ratio, as defined in the facility, not greater than 0.65 to 1.00 at all times. At September 30, 2010, KCP&L was in compliance with this covenant. At September 30, 2010, KCP&L had \$209.5 million of commercial paper outstanding, at a weighted-average interest rate of 0.41%, \$24.4 million of letters of credit outstanding and no outstanding cash borrowings under the facility. At December 31, 2009, KCP&L had \$186.6 million of commercial paper outstanding, at a weighted-average interest rate of 0.58%, \$20.9 million of letters of credit outstanding and no outstanding cash borrowings under the facility.

GMO's \$450 Million Revolving Credit Facility

In August 2010, GMO entered into a new \$450 million revolving credit facility with a group of banks that expires in August 2013. The facility replaced a \$400 million revolving credit facility with a group of banks that would have expired in September 2011 and was sized based on expected short-term debt requirements over the facility's term. Great Plains Energy and GMO may transfer up to \$200 million of unused commitments between Great Plains Energy's and GMO's facilities. A default by GMO, Great Plains Energy or any of its significant subsidiaries on other indebtedness totaling more than \$50.0 million is a default under the facility. Under the terms of this facility, GMO is required to maintain a consolidated indebtedness to consolidated capitalization

ratio, as defined in the facility, not greater than 0.65 to 1.00 at all times. At September 30, 2010, GMO was in compliance with this covenant. At September 30, 2010, GMO had no outstanding cash borrowings and had issued letters of credit totaling \$13.2 million under the credit facility. At December 31, 2009, GMO had \$232.0 million of outstanding cash borrowings with a weighted-average interest rate of 1.50%, and had issued letters of credit totaling \$13.2 million under the credit facility.

10. LONG-TERM DEBT

Great Plains Energy's and KCP&L's long-term debt is detailed in the following table.

	Year Due	September 30 2010	December 31 2009	
KCP&L		(mil		
General Mortgage Bonds		()	
4.90%* EIRR bonds	2012-2035	\$ 158.8	\$ 158.8	
7.15% Series 2009A (8.59% rate**)	2019	400.0	400.0	
4.65% EIRR Series 2005	2035	50.0	50.0	
5.125% EIRR Series 2007A-1	2035	63.3	63.3	
2.625% EIRR Series 2007A-2	2035	10.0	10.0	
5.375% EIRR Series 2007B	2035	73.2	73.2	
Senior Notes				
6.50% Series	2011	150.0	150.0	
5.85% Series (5.72% rate**)	2017	250.0	250.0	
6.375% Series (7.49% rate**)	2018	350.0	350.0	
6.05% Series (5.78% rate**)	2035	250.0	250.0	
EIRR Bonds	2033	250.0	250.0	
4.90% Series 2008	2038	23.4	23.4	
Other	2011-2018	3.3	3.5	
Current maturities	2011-2010	(0.3)	(0.2)	
Unamortized discount		(2.0)	(2.1)	
Total KCP&L		1,779.7	1,779.9	
		1,//9./	1,//9.9	
Other Great Plains Energy				
GMO First Mortgage Bonds				
9.44% Series	2011-2021	12.4	13.5	
GMO Pollution Control Bonds				
5.85% SJLP Pollution Control	2013	5.6	5.6	
0.289%*** Wamego Series 1996	2026	7.3	7.3	
0.937%*** State Environmental 1993	2028	5.0	5.0	
GMO Senior Notes				
7.95% Series	2011	137.3	137.3	
7.75% Series	2011	197.0	197.0	
11.875% Series	2012	500.0	500.0	
8.27% Series	2021	80.9	80.9	
Fair Value Adjustment		58.6	84.5	
GMO Medium Term Notes				
7.16% Series	2013	6.0	6.0	
7.33% Series	2023	3.0	3.0	
7.17% Series	2023	7.0	7.0	
Great Plains Energy 2.75% Senior Notes (3.67% rate**)	2013	250.0	_	
Great Plains Energy 6.875% Senior Notes (7.33% rate**)	2017	100.0	100.0	
Great Plains Energy 10.00% Equity Units Subordinated Notes	2042	287.5	287.5	
Current maturities		(335.4)	(1.1)	
Unamortized discount		(0.6)	(0.4)	
Total Great Plains Energy excluding current maturities		\$ 3,101.3	\$ 3,213.0	
Weighted-average interest rates at Sentember 30, 2010		φ 3,101.3	φ 3,413	

^{*} Weighted-average interest rates at September 30, 2010

^{**} Rate after amortizing gains/losses recognized in OCI on settlements of interest rate hedging instruments

^{***} Variable rate

Fair Value of Long-Term Debt

Fair value of long-term debt is based on quoted market prices, with the incremental borrowing rate for similar debt used to determine fair value if quoted market prices were not available. At September 30, 2010, and December 31, 2009, the book value of Great Plains Energy's long-term debt, including current maturities, was \$3.4 billion and \$3.2 billion, respectively. At September 30, 2010, and December 31, 2009, the fair value of Great Plains Energy's long-term debt, including current maturities, was \$3.8 billion and \$3.4 billion, respectively. At September 30, 2010, and December 31, 2009, the book value of KCP&L's long-term debt, including current maturities, was \$1.8 billion. At September 30, 2010, and December 31, 2009, the fair value of KCP&L's long-term debt, including current maturities, was \$2.0 billion and \$1.9 billion, respectively.

Amortization of Debt Expense

Great Plains Energy's and KCP&L's amortization of debt expense is detailed in the following table.

	Three Months Ended September 30				Year to Date September 30					
	2010		20	9 20		10	2009			
	(millions)									
KCP&L	\$	0.7	\$	0.5	\$	1.8	\$	1.5		
Other Great Plains Energy		1.2		0.8		2.5		1.7		
Total Great Plains Energy	\$	1.9	\$	1.3	\$	4.3	\$	3.2		

KCP&L EIRR Bonds

In March 2010, KCP&L remarketed its 5.00% EIRR Series 2007A-2 general mortgage bonds maturing in 2035 totaling \$10.0 million to a new fixed rate of 2.625% from April 1, 2010, through March 31, 2011.

Great Plains Energy Senior Notes

In August 2010, Great Plains Energy issued \$250.0 million of 2.75% unsecured Senior Notes, maturing in 2013. As a result of amortizing the loss recognized in Other Comprehensive Income (OCI) on Great Plains Energy's Forward Starting Swaps (FSS), the effective interest rate is 3.67%.

11. COMMITMENTS AND CONTINGENCIES

Environmental Matters

Great Plains Energy and KCP&L are subject to extensive regulation by federal, state and local authorities with regard to environmental matters primarily through their utility operations. In addition to imposing extensive and continuing compliance obligations, laws, regulations and permits authorize the imposition of substantial penalties for noncompliance, including fines, injunctive relief and other sanctions. The cost of complying with current and future environmental requirements is expected to be material to Great Plains Energy and KCP&L. Failure to comply with environmental requirements or to timely recover environmental costs through rates could have a material adverse effect on Great Plains Energy and KCP&L.

The following discussion groups environmental and certain associated matters into the broad categories of air and climate change, water, solid waste and remediation.

Air and Climate Change

The Clean Air Act and associated regulations enacted by the Environmental Protection Agency (EPA) form a comprehensive program to preserve air quality. States are required to establish regulations and programs to address all requirements of the Clean Air Act and have the flexibility to enact more stringent requirements. All of Great Plains Energy's and KCP&L's generating facilities, and certain of their other facilities, are subject to the Clean Air Act.

Great Plains Energy's and KCP&L's current estimate of capital expenditures (exclusive of AFUDC and property taxes) to comply with the currently effective Clean Air Interstate Rule (CAIR) and with the best available retrofit technology (BART) rule is approximately \$1 billion. As discussed below, CAIR has been remanded to the EPA, but remains in effect until the EPA issues final rules consistent with the court's order or until the court takes further action. In July 2010, the EPA proposed regulations to replace CAIR. However, due to uncertainties regarding the proposal (discussed below), it is not possible to predict what the final rules may be, when the rules may be issued, or the costs associated with such rules. The actual cost of compliance with any future rules, and with BART, may be significantly different from the cost estimate provided.

The potential costs of the Collaboration Agreement provisions (discussed below) relating to NO_x , SO_2 and particulate emission limits at the LaCygne generating station are within the disclosed overall capital cost estimate. However, the estimated capital costs do not reflect potential costs relating to requirements enacted in the future, including potential requirements regarding climate change and control of mercury emissions (discussed below), and also do not reflect costs relating to additional wind generation, energy efficiency and other CO_2 emission offsets contemplated by the Collaboration Agreement or that may be required under the Missouri or Kansas renewable energy standards, which are discussed below. The estimate does not reflect the non-capital costs the Companies incur on an ongoing basis to comply with environmental laws, which may increase in the future due to the implementation of KCP&L's Comprehensive Energy Plan and the Companies' ongoing compliance with current or future environmental laws. KCP&L expects to seek recovery of the costs associated with the Collaboration Agreement and the Companies expect to seek recovery of the costs associated with environmental requirements through rate increases; however, there can be no assurance that such rate increases would be granted. The Companies may be subject to materially adverse rate treatment in response to competitive, economic, political, legislative or regulatory pressures and/or public perception of the Companies' environmental reputation.

Clean Air Interstate Rule (CAIR) and Transport Rule

The CAIR requires reductions in SO_2 and NO_x emissions in 28 states, including Missouri. The reduction in both SO_2 and NO_x emissions is set to be accomplished through establishment of permanent statewide caps for NO_x effective January 1, 2009, and SO_2 effective January 1, 2010. More restrictive caps are scheduled to become effective January 1, 2015. Great Plains Energy's and KCP&L's fossil fuel-fired plants located in Missouri are subject to CAIR, while their fossil fuel-fired plants in Kansas are not.

On July 11, 2008, the D.C. Circuit Court of Appeals vacated CAIR in its entirety and remanded the matter to the EPA to promulgate a new rule consistent with its opinion. On December 23, 2008, the Court issued an order remanding CAIR to the EPA to revise the rule consistent with its July 2008 order. The CAIR thus remains in effect pending future EPA or court action, including the proposed Transport Rule discussed below.

CAIR currently establishes a market-based cap-and-trade program with an emission allowance allocation. Facilities demonstrate compliance with CAIR by holding sufficient allowances for each ton of SO_2 and NO_x emitted in any given year. KCP&L and GMO are currently allowed to utilize unused SO_2 emission allowances that they have either accumulated during previous years of the Acid Rain Program or purchased to meet the more stringent CAIR requirements. At September 30, 2010, KCP&L had accumulated unused SO_2 emission allowances sufficient to support over 135,000 tons of SO_2 emissions (enough to support expected requirements under the current CAIR for the foreseeable future) under the provisions of the Acid Rain program, which are recorded in inventory at zero cost. KCP&L is permitted to sell excess SO_2 emission allowances in accordance with KCP&L's Comprehensive Energy Plan as approved by the MPSC and KCC. At September 30, 2010, GMO had accumulated unused SO_2 emission allowances sufficient to support just over 18,000 tons of SO_2 emissions (enough to support expected requirements under the current CAIR through 2011), which it has received under the Acid Rain Program or purchased, which are recorded in inventory at average cost. KCP&L and GMO purchase NO_x allowances as needed.

In 2009, KCP&L completed environmental upgrades at Iatan No. 1 for compliance with the current CAIR rule as part of its Comprehensive Energy Plan. Analysis of the current CAIR rule indicates that NO_x and SO_2 control may be required for KCP&L's Montrose Station and GMO's Sibley and Lake Road Stations in Missouri, and control may be achieved through a combination of pollution control equipment and the use or purchase of emission allowances as needed.

In July 2010, the EPA proposed the Transport Rule to replace the current CAIR. The Transport Rule, like CAIR, will require the states within its scope to reduce power plant SO_2 and NO_x emissions that contribute to ozone and fine particle nonattainment in other states. The geographical scope of the Transport Rule is broader than CAIR, and includes Kansas in addition to Missouri and other states. The Transport Rule would also impose more stringent emissions limitations than CAIR and, unlike CAIR, would not utilize Acid Rain Program allowances for compliance. The EPA is proposing a preferred approach and is taking comment on two alternatives. \Box 0;In the EPA's preferred approach, the EPA would set an emissions budget for each of the affected states and the District of Columbia. The preferred approach would allow limited interstate emissions allowance trading among power plants; however, it would not permit trading of SO_2 allowances between the Companies' Kansas and Missouri power plants. In the first alternative, the EPA is proposing to set an emissions budget for each state and allow emissions allowance trading only among power plants within a state. In the second alternative, the EPA is proposing to set an emissions budget for each state, specify the allowable emission limit for each power plant and allow some averaging. Compliance with the Transport Rule would begin in 2012, with additional reductions in SO_2 allowances allocable to the Companies' Kansas power plants.

In September and October 2010, the EPA supplemented the record supporting the proposed Transport Rule. The EPA made available additional information relevant to the rulemaking, including, among other things, an updated version of the power sector modeling that the EPA proposes to use to support the final rule.

The proposed Transport Rule is complex and, as noted, contains alternative approaches. Great Plains Energy and KCP&L are unable to predict when the Transport Rule (or other rule replacing CAIR) might be adopted, or the actual requirements of such rule. Preliminary analysis of the Transport Rule has raised various questions regarding the emission allowances allocation to, and the allowable emission rates for, the Companies' power plants pursuant to the preferred approach and alternatives, which the Companies will attempt to address during the rule's comment period. Regardless of the resolution of those questions, the Companies project that they may not be allocated sufficient SO₂ or NO_X emissions allowances to cover their currently expected operations starting in 2012 pursuant to the preferred approach. Any shortfall in allocated allowances would need to be addressed through permissible allowance trading, installing additional emission control equipment, changes in plant operation, purchasing additional power in the wholesale market, or a combination of these and other alternatives. While Great Plains Energy and KCP&L cannot reasonably predict at this time the impacts of the final Transport Rule, if it were finalized as currently proposed, the Companies expect that any required capital expenditures would not exceed the \$1 billion estimate of capital expenditures (exclusive of AFUDC and property taxes) to comply with the currently effective CAIR and BART rule disclosed above. Any final rule could have a significant adverse effect on Great Plains Energy's and KCP&L's results of operations, financial position and cash flows.

Best Available Retrofit Technology Rule (BART)

The EPA BART rule directs state air quality agencies to identify whether visibility-reducing emissions from sources subject to BART are below limits set by the state or whether retrofit measures are needed to reduce emissions. BART applies to specific eligible facilities including KCP&L's LaCygne Nos. 1 and 2 in Kansas, KCP&L's Iatan No. 1, in which GMO has an 18% interest, KCP&L's Montrose No. 3 in

Missouri, GMO's Sibley Unit No. 3 and Lake Road Unit No. 6 in Missouri and Westar Energy, Inc.'s (Westar) Jeffrey Unit Nos. 1 and 2 in Kansas, in which GMO has an 8% interest. Initially, in Missouri, compliance with CAIR will be compliance with BART for individual sources. Neither Missouri nor Kansas has received EPA approval for their BART plans.

Mercury Emissions

In January 2009, the EPA issued a memorandum stating that new electric steam generating units (EGUs) that began construction while the Clean Air Mercury Rule (CAMR) was effective are subject to a new source maximum achievable control technology (MACT) determination on a case-by-case basis.

In July 2009, the EPA sent letters notifying KCP&L that MACT determinations and schedules of compliance are required for coal and oil-fired EGUs that began actual construction or reconstruction after December 15, 2000, and identified Iatan No. 2 and Hawthorn No. 5 as affected EGUs. This was an outcome of the D.C. Court of Appeals' vacatur of both the CAMR and the contemporaneously promulgated rule removing EGUs from MACT requirements. KCP&L believes that Hawthorn No. 5 is not an affected EGU based on the reconstruction dates of the unit, and provided supporting documentation to the Missouri Department of Natural Resources (MDNR). It is not currently known how MACT determinations and schedules of compliance will impact the permitting or operating requirements for these two units, but it is po ssible a MACT determination may ultimately require additional emission control equipment and permit limits at Iatan No. 2, Hawthorn No. 5, or both.

In April 2010, the EPA, in a court approved settlement, agreed to develop MACT standards for mercury and potentially other hazardous air pollutant emissions. In the settlement agreement, the EPA agreed to propose MACT standards in March 2011 with final standards by November 2011. These MACT standards, if adopted, could impact KCP&L's and GMO's new and existing facilities.

Management cannot predict the outcome of further judicial, administrative or regulatory actions or their financial or operational effects on Great Plains Energy and KCP&L. Such actions could have a significant effect on Great Plains Energy's and KCP&L's results of operations, financial position and cash flows. Some of the control technology for SO_2 and NO_x could also aid in the control of mercury.

Industrial Boiler Rule

In April 2010, the EPA issued a proposed rule that would set MACT standards for hazardous air pollutants from industrial boilers. The proposed rule would establish emission limits for KCP&L's and GMO's new and existing units that produce steam other than for the generation of electricity. This proposed rule does not apply to KCP&L's and GMO's electricity generating boilers, but would apply to most of GMO's Lake Road boilers, which also serve steam customers, and to auxiliary boilers at other generating facilities. Until a rule is finalized, the financial and operational impacts to Great Plains Energy and KCP&L cannot be determined.

New Source Review

The Clean Air Act requires companies to obtain permits and, if necessary, install control equipment to reduce emissions when making a major modification or a change in operation if either is expected to cause a significant net increase in regulated emissions.

In January 2004, Westar received notification from the EPA alleging that it had violated new source review requirements and Kansas environmental regulations by making modifications to the Jeffrey Energy Center without obtaining the proper permits. The Jeffrey Energy Center consists of three coal-fired units located in Kansas that is 92% owned by Westar and operated exclusively by Westar. GMO has an 8% interest in the Jeffrey Energy Center and is generally responsible for its 8% share of the facility's operating costs and capital expenditures. In February 2009, the Attorney General of the United States filed a complaint against Westar alleging that it violated the Clean Air Act and related federal and state regulations by making major modifications to the Jeffrey Energy Center beginning in 1994 without

first obtaining appropriate permits authorizing this construction and without installing and operating best available control technology to control emissions. In January 2010, Westar entered into a settlement agreement, which was approved by the court in March 2010. The settlement agreement requires, among other things, the installation of a selective catalytic reduction (SCR) system at one of the Jeffrey Energy Center units by the end of 2014 and the payment of a \$3 million civil penalty. Westar has preliminarily estimated the cost of this SCR at approximately \$200 million. This amount could materially change depending on final engineering and design. Depending on the NO_x emission reductions attained by that SCR and attainable through the installation of other controls at the other two units, the settlement agreement requires the installation of a second SCR system on one of the other two units by the end of 2016. There is no assurance that GMO's share of these costs would be recovered in rates and failure to recover such costs could have a significant adverse effect on Great Plains Energy's results of operations, financial position and cash flows.

Collaboration Agreement

In March 2007, KCP&L, the Sierra Club and the Concerned Citizens of Platte County entered into a Collaboration Agreement under which KCP&L agreed to pursue a set of initiatives including energy efficiency, additional wind generation, lower emission permit levels at its Iatan and LaCygne generating stations and other initiatives designed to offset CO₂ emissions. Full implementation of the terms of the Collaboration Agreement will necessitate approval from the appropriate authorities, as some of the initiatives in the agreement require regulatory approval.

In 2006, KCP&L installed 100MW of wind generation at its Spearville wind site. KCP&L agreed in the Collaboration Agreement to pursue increasing its wind generation capacity to 500MW in total by the end of 2012 with 100MW to be added by the end of 2010 and the remainder added by the end of 2012, subject to regulatory approval. KCP&L is constructing a 48MW wind project adjacent to its existing Spearville wind site with wind turbines it currently owns, which is expected to be completed in December 2010. KCP&L also intends to secure 52MW of renewable energy credits. KCP&L is evaluating alternatives to meet the remaining wind generation capacity requirement, including the purchase of renewable energy credits, power purchase agreements, KCP&L-built installations or some combin ation thereof.

KCP&L agreed in the Collaboration Agreement to seek a consent agreement, which it has done, with the Kansas Department of Health and Environment (KDHE) incorporating limits for stack particulate matter emissions, as well as limits for NO_x and SO₂ emissions at its LaCygne Station that will be below the presumptive limits under BART. KCP&L further agreed to use its best efforts to install emission control technologies to reduce those emissions from the LaCygne Station prior to the required compliance date under BART, but in no event later than June 1, 2015. Also as provided for in the Collaboration Agreement, KCP&L issued, in 2008, requests for proposals for eq uipment required to comply with BART at the LaCygne Station. KCP&L is continuing to evaluate compliance options in light of developing potential legislative and regulatory environmental requirements.

In the Collaboration Agreement, KCP&L also agreed to offset an additional 711,000 tons of CO₂ by the end of 2012. KCP&L currently expects to achieve this offset through a number of alternatives, including improving the efficiency of its coal-fired units, equipping certain gas-fired units for winter operation and, if necessary, possibly reducing output of, or retiring, one or more coal-fired units.

Climate Change

The Companies are subject to existing greenhouse gas reporting regulations and, as discussed below, will be subject to certain greenhouse gas permitting requirements starting in 2011. Management believes it is likely that additional federal or relevant state or local laws or regulations could be enacted to address global climate change. At the international level, while the United States is not a current party to the Kyoto Protocol, it has agreed to undertake certain voluntary actions under the non-binding Copenhagen Accord, including the establishment of a goal to reduce greenhouse gas emissions. International

agreements legally binding on the United States may be reached in the future. Such new laws or regulations could mandate new or increased requirements to control or reduce the emission of greenhouse gases, such as CO_2 , which are created in the combustion of fossil fuels. The Companies' current generation capacity is primarily coal-fired and is estimated to produce about one ton of CO_2 per MWh, or approximately 23 million tons and 17 million tons per year for Great Plains Energy and KCP&L, respectively.

Laws have recently been passed in Missouri and Kansas, the states in which the Companies' retail electric businesses are operated, setting renewable energy standards, and management believes that national renewable energy standards are also likely. While management believes additional requirements addressing these matters will probably be enacted, the timing, provisions and impact of such requirements, including the cost to obtain and install new equipment to achieve compliance, cannot be reasonably estimated at this time. In addition, certain federal courts have held that state and local governments and private parties have standing to bring climate change tort suits seeking company-specific emission reductions and monetary or other damages. While the Companies are not a party to any climate change tort suit, there is no assurance that such suits may not be filed in the future or the outcome if such suits are filed. Such requirements or litigation outcomes could have the potential for a significant financial and operational impact on Great Plains Energy and KCP&L. The Companies would seek recovery of capital costs and expenses for compliance through rate increases; however, there can be no assurance that such rate increases would be granted.

Legislation concerning the reduction of emissions of greenhouse gases, including CO₂, is being considered at the federal and state levels. In June 2009, the U.S. House of Representatives passed the American Clean Energy and Security Act of 2009 (House Bill), which would establish a renewable electricity standard and a greenhouse gas cap and trade program that would require Great Plains Energy, KCP&L and other affected entities to surrender allowances or offsets for each ton of greenhouse gas emitted, and that would reduce the available quantity of emission allowances over time. The U.S. Senate has not yet enacted companion legislation. Legislation pro posed or enacted in the future, however, may include greenhouse gas reduction measures, including those contained in the House Bill. The timing and effects of any such legislation cannot be determined at this time.

In the absence of new Congressional mandates, the EPA is proceeding with the regulation of greenhouse gases under the existing Clean Air Act. In April 2010, the EPA finalized greenhouse gas emission standards for light-duty vehicles. These are the first-ever national greenhouse gas emission standards under the Clean Air Act.

In May 2010, the EPA issued a final rule addressing greenhouse gas emissions from stationary sources under the Clean Air Act permitting programs. This final rule sets thresholds for GHG emissions that define when permits under the PSD and Title V Operating Permit programs are required for new and existing industrial facilities. The EPA will phase in the Clean Air Act permitting requirements for greenhouse gas emissions in two initial steps. In step 1, starting January 2, 2011, only sources currently subject to the PSD permitting program (i.e., those that are newlyconstructed or modified in a way that significantly increases emissions of a pollutant other than greenhouse gas) would be subject to Title V or PSD permitting requirements, respectively, for their greenhouse gas emissions. For these projects, only projects with new or increases of greenhouse gas emissions of 75,000 tons per year or more of total greenhouse gases, on a CO₂ equivalent basis, would need to determine the best available control technology for their greenhouse gas emissions. In addition, sources subject to the Title V Operating Permit Program would need to address greenhouse gas emissions as those permits are applied for or renewed. In step 2, starting July 1, 2011, Title V and PSD permitting requirements will cover, for the first time, new construction projects that emit greenhouse gas emissions of at least 100,000 tons per year even if they do not exceed the permitting thresholds for any other pollutant. In addition, modifications at such existing facilities that increase greenhouse gas emissions by at least 75,000 tons per year will be

subject to permitting requirements, even if they do not significantly increase emissions of any other pollutant. Great Plains Energy's and KCP&L's generating facilities that trigger these thresholds for new installations, modifications or Title V operating permits will be subject to this rule.

At the state level, a Kansas law enacted in May 2009 requires Kansas public electric utilities, including KCP&L, to have renewable energy generation capacity equal to at least 10% of their three-year average Kansas peak retail demand by 2011. The percentage increases to 15% by 2016 and 20% by 2020. A Missouri law enacted in November 2008 requires at least 2% of the electricity provided by Missouri investor-owned utilities (including KCP&L and GMO) to their Missouri retail customers to come from renewable resources, including wind, solar, biomass and hydropower, by 2011, increasing to 5% in 2014, 10% in 2018, and 15% in 2021, with a small portion (estimated to be about 2MW in 2011 for each of KCP&L and GMO) required to come from solar resources. Regulations implementing these laws are being drafted by the MPSC and KCC, and the ultimate impacts on the Companies cannot be reasonably estimated at this time. Subject to the terms of the final MPSC regulations, KCP&L and GMO project that their current renewable resources (including accumulated renewable energy credits), combined with an expected purchase of solar renewable energy credits, will be sufficient for compliance with the Missouri requirements through 2013. KCP&L also projects, subject to the terms of the final KCC regulations, that its current renewable resources (including accumulated renewable energy credits) combined with the 48MW wind project and 52MW of renewable energy credits discussed above will be sufficient for compliance with the 2011 Kansas requirements.

Additionally, in November 2007, governors from six Midwestern states, including Kansas, signed the Midwestern Greenhouse Gas Reduction Accord, which has established the goal of reducing member states' greenhouse gas emissions to 15% to 20% below 2005 levels by 2020, and 60% to 80% below 2005 levels by 2050.

Greenhouse gas legislation or regulation has the potential of having significant financial and operational impacts on Great Plains Energy and KCP&L, including the potential costs and impacts of achieving compliance with limits that may be established. However, the ultimate financial and operational consequences to Great Plains Energy and KCP&L cannot be determined until such legislation is passed, regulations are issued or, with respect to those regulations are issued, additional guidance is provided. Management will continue to monitor the progress of relevant legislation and regulations.

Ozone NAAQS

In June 2007, monitor data indicated that the Kansas City area violated the 1997 primary eight-hour ozone national ambient air quality standard (NAAQS). Missouri and Kansas have implemented the responses established in the maintenance plans for control of ozone. The responses in both states do not require additional controls at Great Plains Energy's and KCP&L's generation facilities beyond the currently proposed controls for CAIR and BART. The EPA has various options over and above the implementation of the maintenance plans for control of ozone to address the violation but has not yet acted. At this time, management is unable to predict how the EPA will respond or how that response will impact Great Plains Energy's and KCP&L's operations. However, the EPA's response could have a significant effect on Great Plains Energy's and KCP&L's results of operations, financial position and cash flows.

In March 2008, the EPA significantly strengthened its NAAQS for ground-level ozone. The EPA revised the primary eight-hour ozone standard, designed to protect public health, to a level of 0.075 parts per million (ppm). The EPA also strengthened the secondary eight-hour ozone standard to the level of 0.075 ppm making it identical to the revised primary standard. The previous primary and secondary standards, set in 1997, were effectively 0.084 ppm.

In March 2009, the MDNR and KDHE submitted to the EPA their determinations that the Kansas City area is a nonattainment area under the 2008 primary eight-hour ozone standard. The EPA will make final designations of attainment and nonattainment areas. By 2013, states must submit state implementation plans outlining how states will reduce ozone to meet the standards in nonattainment areas. Although the impact on Great Plains Energy's and KCP&L's operations will not be known until after the final nonattainment designations and the state implementation plans are submitted, it could have a significant effect on Great Plains Energy's and KCP&L's results of operations, financial position and cash flows.

In January 2010, the EPA proposed to reconsider and further strengthen the 2008 NAAQS for ground-level ozone. The EPA proposed to strengthen the primary eight-hour ozone standard to a level within the range of 0.060-0.070 ppm. The EPA also proposed to establish a distinct cumulative, seasonal secondary standard, designed to protect sensitive vegetation and ecosystems, to within the range of 7-15 ppm-hours.

SO2 NAAQS

In June 2010, the EPA strengthened the primary NAAQS for SO_2 . The EPA revised the primary SO_2 standard by establishing a new 1-hour standard at a level of 0.075 ppm. The EPA revoked the two existing primary standards of 0.140 ppm evaluated over 24-hours and 0.030 ppm evaluated over an entire year. Although the impact on Great Plains Energy's and KCP&L's operations will not be known until after the nonattainment designations are approved and the state implementation plans submitted, it could have a significant effect on Great Plains Energy's and KCP&L's results of operat ions, financial position and cash flows.

Montrose Station Notice of Violation

In June 2009, KCP&L received notification from the MDNR alleging that its Montrose Station had excess particulate matter emissions in 2008. KCP&L is working with the MDNR to resolve this issue and management believes the outcome will have an insignificant impact to Great Plains Energy's and KCP&L's results of operations, financial position and cash flows.

Water

The Clean Water Act and associated regulations enacted by the EPA form a comprehensive program to preserve water quality. Like the Clean Air Act, states are required to establish regulations and programs to address all requirements of the Clean Water Act, and have the flexibility to enact more stringent requirements. All of Great Plains Energy's and KCP&L's generating facilities, and certain of their other facilities, are subject to the Clean Water Act.

Section 316(b) of the Clean Water Act is designed to protect aquatic life from being killed or injured by cooling water intake structures. The EPA had previously issued regulations pursuant to Section 316(b) of the Clean Water Act regarding cooling water intake structures. Subsequent to an appellate court ruling, the EPA suspended the regulations and is engaged in further rulemaking on this matter. At this time, management is unable to predict how the EPA will respond or how that response will impact Great Plains Energy's and KCP&L's operations.

KCP&L holds a permit from the MDNR covering water discharge from its Hawthorn Station. The permit authorizes KCP&L, among other things, to withdraw water from the Missouri river for cooling purposes and return the heated water to the Missouri river. KCP&L has applied for a renewal of this permit and the EPA has submitted an interim objection letter regarding the allowable amount of heat that can be contained in the returned water. Until this matter is resolved, KCP&L continues to operate under its current permit. KCP&L cannot predict the outcome of this matter; however, while less significant outcomes are possible, this matter may require KCP&L to reduce its generation at Hawthorn Station, install cooling towers or both, any of which could have a significant impact on KCP&L. The outcome could also affect the terms of water permit renewals at KCP&L's Iatan Station and at GMO's Sibley and Lake Road Stations.

Additionally, in September 2009, the EPA announced plans to revise the existing standards for water discharges from coal-fired power plants. Until a rule is proposed and finalized, the financial and operational impacts to Great Plains Energy and KCP&L cannot be determined.

Solid Waste

Solid and hazardous waste generation, storage, transportation, treatment and disposal is regulated at the federal and state levels under various laws and regulations. In May 2010, the EPA proposed to regulate coal combustion residuals (CCRs) under the Resource Conservation and Recovery Act (RCRA) to address the risks from the disposal of CCRs generated from the combustion of coal at electric generating facilities. The EPA is considering two options in this proposal. Under the first proposal, the EPA would regulate CCRs as special wastes subject to regulation under subtitle C of RCRA, when they are destined for disposal in landfills or surface impoundments. Under the second proposal, the EPA would regulate disposal of CCRs under subtitle D of RCRA. The Companies principally use coal in generating electricity and dispose of the combustion products in both on-site facilities and facilities owned by third parties. The proposed CCR rule has the potential of having a significant financial and operational impact on Great Plains Energy and KCP&L in connection with achieving compliance with the proposed requirements. However, the financial and operational consequences to Great Plains Energy and KCP&L cannot be determined until an option is selected by the EPA and the final regulation is enacted.

Remediation

Certain federal and state laws, including the Comprehensive Environmental Response, Compensation and Liability Act (CERCLA) hold current and previous owners or operators of real property, and any person who arranges for the disposal or treatment of hazardous substances at a property, liable on a joint and several basis for the costs of cleaning up contamination at or migrating from such real property, even if they did not know of and were not responsible for such contamination. CERCLA and other laws also authorize the EPA and other agencies to issue orders compelling potentially responsible parties to clean up sites that are determined to present an actual or potential threat to human health or the environment. GMO is named as a potentially responsible party at two disposal sites for polychlorinated biphenyls (PCBs), and retains some environmental liability for several operations and investments it no longer owns. In addition, GMO also owns, or has acquired liabilities from companies that once owned or operated, former manufactured gas plant (MGP) sites, which are subject to the supervision of the EPA and various state environmental agencies.

At September 30, 2010, and December 31, 2009, KCP&L had \$0.3 million accrued for environmental remediation expenses, which covers ground water monitoring at a former MGP site. At September 30, 2010, and December 31, 2009, Great Plains Energy had \$0.4 million accrued for environmental remediation expenses, which includes the \$0.3 million at KCP&L, and additional potential remediation and ground water monitoring costs relating to two GMO sites. The amounts accrued were established on an undiscounted basis and Great Plains Energy and KCP&L do not currently have an estimated time frame over which the accrued amounts may be paid.

In addition to the \$0.4 million accrual above, at September 30, 2010, Great Plains Energy had \$2.0 million accrued for the future investigation and remediation of certain additional GMO identified MGP sites, PCB sites and retained liabilities. This estimate was based upon review of the potential costs associated with conducting investigative and remedial actions at identified sites, as well as the likelihood of whether such actions will be necessary. This estimate could change materially after further investigation, and could also be affected by the actions of environmental agencies and the financial viability of other potentially responsible parties.

GMO has pursued recovery of remediation costs from insurance carriers and other potentially responsible parties. As a result of a settlement with an insurance carrier, approximately \$2.2 million in insurance proceeds less an annual deductible is available to GMO to recover qualified MGP remediation expenses. GMO would seek recovery of additional remediation costs and expenses through rate increases; however, there can be no assurance that such rate increases would be granted.

In January 2010, the EPA announced an advance notice of proposed rulemaking under CERCLA identifying classes of facilities for which the EPA will develop financial assurance requirements, including the electric power generation, transmission and distribution industry. The CERCLA financial assurance would be for risks associated with Great Plains Energy's and KCP&L's production, transportation, treatment, storage or disposal of CERCLA hazardous substances. The impact on Great Plains Energy and KCP&L cannot be determined until the regulations are finalized.

In April 2010, the EPA announced an advance notice of proposed rulemaking for the use and distribution in commerce of certain PCBs, PCB items and certain other areas of the PCB regulations. The EPA is reassessing the use, distribution in commerce, marking, and storage for reuse of liquid PCBs in electric and non-electric equipment and the use of the 50 ppm level for excluded PCB products among other things. The impact on Great Plains Energy and KCP&L cannot be determined until the regulations are finalized.

12. LEGAL PROCEEDINGS

KCP&L Hawthorn No. 5 Litigation

KCP&L received reimbursement for the 1999 Hawthorn No. 5 boiler explosion under a property damage insurance policy with Travelers Property Casualty Company of America (Travelers). Travelers filed suit in the U.S. District Court for the Eastern District of Missouri in November 2005, against National Union Fire Insurance Company of Pittsburgh, Pennsylvania, (National Union) and KCP&L was added as a defendant in June 2006. The case was subsequently transferred to the U.S. District Court for the Western District of Missouri. Travelers sought recovery of \$10 million that KCP&L recovered through subrogation litigation. On July 24, 2008, the Court held that Travelers is not entitled to any recovery from KCP&L. Travelers appealed this decision on March 11, 2009, to the Court of Appeals for the Eighth Circuit. In September 2010, the Court of Appeals affirmed the District Court's decision.

KCP&L Spent Nuclear Fuel and Radioactive Waste

KCP&L and the other two Wolf Creek owners have a lawsuit pending against the United States in the U.S. Court of Federal Claims seeking \$14.1 million of damages resulting from the government's failure to begin accepting spent nuclear fuel for disposal in January 1998, as the government was required to do by the Nuclear Waste Policy Act of 1982. Approximately seventy other similar cases were filed with that court, a few of which have settled. To date, the court has rendered final decisions in several of the cases, most of which are on appeal now. Another Federal appellate court has already determined that the government breached its obligation to begin accepting spent fuel for disposal. The questions now before the court in the pending cases are whether and to what extent the utilities are en titled to monetary damages for that breach. The Wolf Creek case was tried before a Court of Federal Claims judge in June 2010 and the parties expect a decision in late 2010.

KCP&L Advanced Coal Credit Arbitration

In 2009, KCP&L was served a notice to arbitrate by Empire District Electric Company (Empire), Kansas Electric Cooperative, Inc. (KEPCO) and Missouri Joint Municipal Electric Utility Commission (MJMEUC), the non-Company joint owners of Iatan No. 2. These joint owners asserted that they were entitled to receive proportionate shares (or the monetary equivalent) of approximately \$125 million of qualifying advance coal project credits for Iatan No. 2. As independent entities, the joint owners are taxed separately and the non-Company joint owners do not dispute that they did not, in fact, apply for the credits themselves. Notwithstanding this, they contended that they should receive proportional shares of the credit. On December 30, 2009, an arbitration panel issued its order denying the KEPCO and MJMEUC claims but ordering KCP&L and Empire to jointly seek a reallocation of the tax credit from the IRS giving Empire its representative percentage of the total tax credit, worth approximately \$17.7 million. The order further specified that if the IRS denies the parties' reallocation request or if Empire is allocated less than its proportionate share of the tax credits, KCP&L will be responsible for paying Empire the full value of its representative percentage of the tax credits (less the amount of tax credits, if any, Empire ultimately receives) in cash. In September 2010, the IRS issued an amended memorandum of understanding to reallocate \$17.7 million of the original \$125 million of the advanced coal project credits to Empire, meeting the requirements of the arbitration order issued on December 30, 2009. KCP&L subsequently dismissed its March

31, 2010, appeal of the arbitration order. KCP&L reversed a \$17.7 million liability previously recorded in other current liabilities for this matter.

Iatan Levee Litigation

On May 22, 2009, several farmers filed suit against Great Plains Energy and KCP&L in the Circuit Court of Platte County, Missouri, alleging negligence, private nuisance, trespass and violations of the Missouri Crop Protection Act and seeking unspecified compensatory and punitive damages. These allegations stem from flooding at or near the Iatan Station in 2007 and 2008. The farmers allege the flooding was a result of maintenance of a nearby levee. The petition seeks class certification from the courts. Written discovery and depositions are underway. Management cannot predict the outcome of this matter.

GMO Price Reporting Litigation

In response to complaints of manipulation of the California energy market, in 2002 FERC issued an order requiring net sellers of power in the California markets from October 2, 2000, through June 20, 2001, at prices above a FERC determined competitive market clearing price to make refunds to net purchasers of power in the California market during that time period. Because MPS Merchant was a net purchaser of power during the refund period, it has received approximately \$8 million in refunds through settlements with certain sellers of power. MPS Merchant estimates that it is entitled to approximately \$12 million in additional refunds under the standards FERC has used in this case. FERC has stated that interest will be applied to the refunds but the amount of interest has not yet been determined. 60; However, various parties appealed the FERC order to the United States Court of Appeals for the Ninth Circuit seeking review of a number of issues, including changing the refund period to include periods prior to October 2, 2000. MPS Merchant was a net seller of power during the period prior to October 2, 2000. On August 2, 2006, the U.S. Court of Appeals for the Ninth Circuit issued an order finding, among other things, that FERC did not provide a sufficient justification for refusing to exercise its remedial authority under the Federal Power Act to determine whether market participants violated FERC-approved tariffs during the period prior to October 2, 2000, and imposing a remedy for any such violations. The court remanded the matter to FERC to determine whether tariff violations occurred and, if so, the appropriate remedy. In March 2008, FERC issued an order declining to order refunds for the period prior to October 2, 2000. That order h as been appealed to the U.S. Court of Appeals for the Ninth Circuit. If FERC ultimately includes the period prior to October 2, 2000, MPS Merchant could be found to owe refunds.

FERC initiated a separate docket, generally referred to as the Pacific Northwest refund proceeding, to determine if any refunds were warranted related to the potential impact of the California market issues on buyers in the Pacific Northwest between December 25, 2000, and June 20, 2001. FERC rejected the refund requests, but its decision was remanded by the Court of Appeals for FERC to consider whether any acts of market manipulation support the imposition of refunds. Claims against MPS Merchant total \$5.1 million for the period addressed under the Pacific Northwest refund proceedings.

In October 2006, the MPSC filed suit in the Circuit Court of Jackson County, Missouri against 18 companies, including GMO and MPS Merchant alleging that the companies manipulated natural gas prices through the misreporting of natural gas trade data and, therefore, violated Missouri antitrust laws. The suit does not specify alleged damages and was filed on behalf of all local distribution gas companies in Missouri who bought and sold natural gas from June 2000 to October 2002. The defendants' motions to dismiss the case were granted in January 2009. The MPSC has appealed the dismissal to the Missouri Court of Appeals for the Western District of Missouri. In December 2009, the court affirmed the dismissal and the MPSC filed a request for rehearing or, in the alternative, transfer to the Miss ouri Supreme Court. The Supreme Court accepted the transfer in April 2010, but in September 2010, transferred the case back to the Court of Appeals.

The ultimate outcome of these matters cannot be predicted.

13. RELATED PARTY TRANSACTIONS AND RELATIONSHIPS

KCP&L employees manage GMO's business and operate its facilities at cost. These costs totaled \$26.3 million and \$73.5 million, respectively, for the three months ended and year to date September 30, 2010. These costs totaled \$24.6 million and \$75.2 million, respectively, for the same periods in 2009. Additionally, KCP&L and GMO engage in wholesale electricity transactions with each other. KCP&L and GMO are also authorized to participate in the Great Plains Energy money pool, an internal financing arrangement in which funds may be lent on a short-term basis to KCP&L and GMO. The following table summarizes KCP&L's related party receivables and payables.

	Sep	tember 30 2010	Decem 20	
		(millio	ons)	
Receivable (payable) from/to GMO	\$	(19.0)	\$	26.4
Payable to Services		-		(0.2)
Receivable from Great Plains Energy		12.7		15.1
Receivable from MPS Merchant		-		0.9

14. DERIVATIVE INSTRUMENTS

Great Plains Energy and KCP&L are exposed to a variety of market risks including interest rates and commodity prices. Management has established risk management policies and strategies to reduce the potentially adverse effects that the volatility of the markets may have on Great Plains Energy's and KCP&L's operating results. Commodity risk management activities, including the use of certain derivative instruments, are subject to the management, direction and control of an internal risk management committee. Management's interest rate risk management strategy uses derivative instruments to adjust Great Plains Energy's and KCP&L's liability portfolio to optimize the mix of fixed and floating rate debt within an established range. In addition, Great Plains Energy and KCP&L use derivative instruments to hedge against future interest rate fluctuations on anticipated debt issuances. Management maintains commodity price risk management strategies that use derivative instruments to reduce the effects of fluctuations in fuel expense caused by commodity price volatility. Counterparties to commodity derivatives and interest rate swap agreements expose Great Plains Energy and KCP&L to credit loss in the event of nonperformance. This credit loss is limited to the cost of replacing these contracts at current market rates. Derivative instruments, excluding those instruments that qualify for the normal purchase normal sale election, which are accounted for by accrual accounting, are recorded on the balance sheet at fair value as an asset or liability. Changes in the fair value of derivative instruments are recognized currently in net income unless specific hedge accounting criteria are met, except GMO utility o perations hedges that are recorded to a regulatory asset or liability consistent with MPSC regulatory orders, as discussed below.

Great Plains Energy and KCP&L post collateral, in the normal course of business, for the aggregate fair value of all derivative instruments with credit risk-related contingent features that are in a liability position. If the credit risk-related contingent features underlying these agreements were triggered, Great Plains Energy and KCP&L would be required to post an insignificant amount of collateral to its counterparties.

The Wall Street Reform and Consumer Protection Act, signed into law in July 2010, includes provisions related to the swaps and over-the-counter derivative markets. The Companies currently expect that their commodity and interest rate hedges will be exempt from mandatory clearing and exchange trading requirements. Capital and margin requirements for these hedges are expected to be determined over the next year as regulatory agencies implement rules. While the Companies currently do not anticipate this law and the associated regulatory rules to have a material impact on their financial condition, the ultimate impact cannot be reasonably determined until the final rules are issued.

Interest Rate Risk Management

In August 2010, Great Plains Energy issued \$250.0 million of long-term debt and settled two Forward Starting Swaps (FSS) simultaneously with the issuance of the long-term fixed rate debt. Great Plains Energy had entered into the two FSS with notional amounts of \$125.0 million to hedge against interest rate fluctuations on a portion of the August 2010 debt issuance. The two FSS were treated as cash flow hedges with no ineffectiveness recorded in 2010 or 2009. A pre-tax loss of \$6.9 million was recorded to OCI and is being reclassified to interest expense over the life of the three-year debt. At September 30, 2010, \$0.3 million of the loss has been reclassified from OCI to interest expense.

Since December 2009, Great Plains Energy has entered into five FSS with total notional amounts of \$350.0 million to hedge against interest rate fluctuations on debt anticipated to be issued in 2011. The five FSS remove a portion of the interest rate variability on \$350.0 million of the debt expected to be issued thereby enabling Great Plains Energy to predict with greater assurance its future interest costs on that debt. The five FSS are treated as cash flow hedges with no ineffectiveness for the three months ended and year to date September 30, 2010. At September 30, 2010, a \$25.1 million loss was recorded in OCI for the five FSS. The FSS will settle simultaneously with the issuance of the underlying long-term debt expected to be issued. Any gain or loss on the settlement will be rec orded to OCI and reclassified to interest expense over the life of the debt.

Commodity Risk Management

KCP&L's risk management policy is to use derivative instruments to mitigate its exposure to market price fluctuations on a portion of its projected natural gas purchases to meet generation requirements for retail and firm wholesale sales. At September 30, 2010, KCP&L has hedged 7% of the 2011 projected natural gas usage for retail load and firm MWh sales, primarily by utilizing futures contracts and financial instruments. The fair values of these instruments are recorded as derivative assets or liabilities with an offsetting entry to OCI for the effective portion of the hedge. To the extent the hedges are not effective, any ineffective portion of the change in fair market value would be recorded currently in fuel expense. KCP&L has not recorded any ineffectiveness on natural gas hedges for the three months ended and year to date September 30, 2010 and 2009.

GMO's risk management policy is to use derivative instruments to mitigate exposure to natural gas price volatility in the market. The fair value of the portfolio relates to financial contracts that will settle against actual purchases of natural gas and purchased power. At September 30, 2010, GMO had financial contracts in place to hedge all for the remainder of 2010, 44% for 2011 and 7% for 2012 of the expected on-peak natural gas and natural gas equivalent purchased power price exposure. In connection with GMO's 2005 Missouri electric rate case, it was agreed that the settlement costs of these contracts would be recognized in fuel expense. The settlement costs are included in GMO's FAC. A regulatory asset has been recorded to reflect the change in the timing of recognition authorized by the MPSC. To the extent that recovery of actual costs incurred is allowed, amounts will not impact earnings, but will impact cash flows due to the timing of the recovery mechanism.

MPS Merchant manages the daily delivery of its remaining contractual commitments with economic hedges (non-hedging derivatives) to reduce its exposure to changes in market prices. Within the trading portfolio, MPS Merchant takes certain positions to hedge physical sale or purchase contracts. MPS Merchant records the fair value of trading energy contracts, both physical and financial, as derivative assets or liabilities with an offsetting entry to Great Plains Energy's consolidated statements of income.

The notional and recorded fair values of open positions for derivative instruments are summarized in the following table. The fair values of these derivatives are recorded on the consolidated balance sheets. The fair values below are gross values before netting agreements and netting of cash collateral.

		September 30 2010		oer 31 9
	Notional Contract Amount	Fair Value	Notional Contract Amount	Fair Value
Great Plains Energy		(millio	ons)	
Futures contracts				
Cash flow hedges	\$ 0.3	\$ (0.1)	\$ 3.2	\$ -
Non-hedging derivatives	17.9	(4.2)	29.8	(0.9)
Forward contracts				
Non-hedging derivatives	206.5	9.1	234.4	9.1
Option contracts				
Cash flow hedges	-	-	2.3	0.2
Non-hedging derivatives	0.5	0.1	-	-
Anticipated debt issuance				
Forward starting swaps	350.0	(25.1)	362.5	(0.7)
KCP&L				
Future contracts				
Cash flow hedges	0.3	(0.1)	3.2	-
Option contracts				
Cash flow hedges	-	-	2.3	0.2

The fair value of Great Plains Energy's and KCP&L's open derivative positions are summarized in the following tables. The tables contain derivative instruments designated as hedging instruments as well as derivative instruments not designated as hedging instruments (non-hedging derivatives) under GAAP. The fair values below are gross values before netting agreements and netting of cash collateral.

Great Plains Energy

Control 20, 2010	Balance Sheet	Asset Derivatives	Liability Derivatives
September 30, 2010	Classification	Fair Value	Fair Value
Derivatives Designated as Hedging Instruments		(mi	illions)
Commodity contracts	Derivative instruments	\$ -	\$ 0.1
Interest rate contracts	Derivative instruments	-	25.1
Derivatives Not Designated as Hedging Instruments			
Commodity contracts	Derivative instruments	9.2	4.2
Total Derivatives		\$ 9.2	\$ 29.4
December 31, 2009			
Derivatives Designated as Hedging Instruments			
Commodity contracts	Derivative instruments	\$ 0.4	\$ 0.2
Interest rate contracts	Derivative instruments	-	0.7
Derivatives Not Designated as Hedging Instruments			
Commodity contracts	Derivative instruments	9.9	1.7
Total Derivatives		\$ 10.3	\$ 2.6

KCP&L

September 30, 2010	Balance Sheet Classification	Asset Derivatives Fair Value	Liability Derivatives Fair Value
*	Classification		
Derivatives Designated as Hedging Instruments		(mil	lions)
Commodity contracts	Derivative instruments	\$ -	\$ 0.1
December 31, 2009			
Derivatives Designated as Hedging Instruments			
Commodity contracts	Derivative instruments	\$ 0.4	\$ 0.2

The following tables summarize the amount of gain (loss) recognized in OCI or earnings for interest rate and commodity hedges.

Great Plains Energy

Derivatives in Cash Flow Hedging Relationship		Gain (Loss) Reclassifi Accumulated OCI into (Effective Portio	Income
	Amount of Gain (Loss) Recognized in OCI on Derivatives (Effective Portion)	Income Statement Classification	Amount
Three Months Ended September 30, 2010	(millions)		(millions)
Interest rate contracts	\$ (9.9)	Interest charges	\$ (2.6)
Commodity contracts	(0.4)	Fuel	(0.5)
Income tax benefit (expense)	4.0	Income tax benefit (expense)	1.1
Total	\$ (6.3)	Total	\$ (2.0)
Year to Date September 30, 2010			
Interest rate contracts	\$ (31.4)	Interest charges	\$ (7.2)
Commodity contracts	(1.0)	Fuel	(0.5)
Income tax benefit (expense)	12.6	Income tax benefit (expense)	2.9
Total	\$ (19.8)	Total	\$ (4.8)
Three Months Ended September 30, 2009			
Interest rate contracts	\$ -	Interest charges	\$ (2.1)
Commodity contracts	(0.6)	Fuel	(1.1)
Income tax benefit (expense)	0.2	Income tax benefit (expense)	1.1
Total	\$ (0.4)	Total	\$ (2.1)
Year to Date September 30, 2009			
Interest rate contracts	\$ 1.0	Interest charges	\$ (5.6)
Commodity contracts	(0.9)	Fuel	(1.1)
Income tax benefit (expense)	-	Income tax benefit (expense)	2.6
Total	\$ 0.1	Total	\$ (4.1)

Gain (Loss) Reclassified from Accumulated OCI into Income (Effective Portion)

		(Effective Portion	n)
	Amount of Gain (Loss) Recognized in OCI on Derivatives (Effective Portion)	Income Statement Classification	Amount
Three Months Ended September 30, 2010	(millions)		(millions)
Interest rate contracts	\$ -	Interest charges	\$ (2.2)
Commodity contracts	(0.4)	Fuel	(0.5)
Income tax benefit (expense)	0.2	Income tax benefit (expense)	1.1
Total	\$ (0.2)	Total	\$ (1.6)
Year to Date September 30, 2010			
Interest rate contracts	\$ -	Interest charges	\$ (6.6)
Commodity contracts	(1.0)	Fuel	(0.5)
Income tax benefit (expense)	0.4	Income tax benefit (expense)	2.8
Total	\$ (0.6)	Total	\$ (4.3)
Three Months Ended September 30, 2009			
Interest rate contracts	\$ -	Interest charges	\$ (2.2)
Commodity contracts	(0.6)	Fuel	(1.1)
Income tax benefit (expense)	0.2	Income tax benefit (expense)	1.3
Total	\$ (0.4)	Total	\$ (2.0)
Year to Date September 30, 2009			
Interest rate contracts	\$ 1.0	Interest charges	\$ (5.3)
Commodity contracts	(0.9)	Fuel	(1.1)
Income tax benefit (expense)	-	Income tax benefit (expense)	2.5
Total	\$ 0.1	Total	\$ (3.9)

The following table summarizes the amount of gain (loss) recognized in a regulatory balance sheet account or earnings for GMO utility commodity hedges. GMO utility commodity derivatives fair value changes are recorded to either a regulatory asset or liability consistent with MPSC regulatory orders.

Great Plains Energy

			Gain (Loss) Reclassified from Regulatory Account				
	Recognized Account o	of Gain (Loss) I on Regulatory on Derivatives ive Portion)	Income Statement Classification	Amo			
Three Months Ended September 30, 2010	(m	illions)		(m	illions)		
Commodity contracts	\$	(2.8)	Fuel	\$	(1.6)		
Total	\$	(2.8)	Total	\$	(1.6)		
Year to Date September 30, 2010 Commodity contracts	\$	(8.7)	Fuel	\$	(5.0)		
Total	\$	(8.7)	Total	\$	(5.9) (5.9)		
Three Months Ended September 30, 2009							
Commodity contracts	\$	(0.2)	Fuel	\$	(10.9)		
Total	\$	(0.2)	Total		(10.9)		
Year to Date September 30, 2009							
Commodity contracts	\$	(11.1)	Fuel	\$	(17.0)		
Total	\$	(11.1)	Total	\$	(17.0)		

Great Plains Energy's income statement reflects gains (losses) for the change in fair value of the MPS Merchant commodity contract derivatives not designated as hedging instruments of \$(1.6) million and an insignificant amount for the three months ended and year to date September 30, 2010, respectively, and \$1.4 million and \$2.0 million, respectively, for the three months ended and year to date September 30, 2009.

The amounts recorded in accumulated OCI related to the cash flow hedges are summarized in the following table.

		Great Plains Energy				KCP&L			
	Sep	September 30 - I 2010		ember 31 2009				ember 31 2009	
				(millio	ns)				
Current assets	\$	12.2	\$	13.3	\$	12.2	\$	13.3	
Current liabilities		(109.0)		(84.9)		(74.1)		(81.2)	
Noncurrent liabilities		-		(0.5)		-		-	
Deferred income taxes		37.7		28.0		24.1		26.4	
Total	\$	(59.1)	\$	(44.1)	\$	(37.8)	\$	(41.5)	

Great Plains Energy's accumulated OCI in the table above at September 30, 2010, includes \$14.4 million that is expected to be reclassified to expense over the next twelve months. KCP&L's accumulated OCI includes \$8.8 million that is expected to be reclassified to expense over the next twelve months.

15. FAIR VALUE MEASUREMENTS

GAAP defines fair value as the price that would be received to sell an asset or paid to transfer a liability in an orderly transaction between market participants at the measurement date. GAAP establishes a fair value hierarchy, which prioritizes the inputs to valuation techniques used to measure fair value into three broad categories, giving the highest priority to quoted prices in active markets for identical assets or liabilities and lowest priority to unobservable inputs. A definition of the various levels, as well as discussion of the various measurements within the levels, is as follows:

Level 1 – Unadjusted quoted prices for identical assets or liabilities in active markets that Great Plains Energy and KCP&L have access to at the measurement date. Assets categorized within this level consist of Great Plains Energy's and KCP&L's various exchange traded derivative instruments and equity and U.S. Treasury securities that are actively traded within KCP&L's decommissioning trust fund and GMO's SERP rabbi trust fund.

Level 2 – Market-based inputs for assets or liabilities that are observable (either directly or indirectly) or inputs that are not observable but are corroborated by market data. Assets and liabilities categorized within this level consist of Great Plains Energy's and KCP&L's various non-exchange traded derivative instruments traded in over-the-counter markets and certain debt securities within KCP&L's decommissioning trust fund and GMO's SERP rabbi trust fund.

Level 3 – Unobservable inputs, reflecting Great Plains Energy's and KCP&L's own assumptions about the assumptions market participants would use in pricing the asset or liability. Assets categorized within this level consist of Great Plains Energy's various non-exchange traded derivative instruments traded in over-the-counter markets and certain debt securities within KCP&L's decommissioning trust fund for which sufficiently observable market data is not available to corroborate the valuation inputs.

					Fair Value Measurements Using						
Description		nber 30 110	Netti	ing ^(d)	in A Mark Iden Ass (Lev	l Prices ctive ets for itical sets rel 1)	Signit Otl	ficant her vable outs	Signif Unobse Inp (Lev	rvable uts	
KCP&L					(millio	ns)					
Assets	ф		ф		ф		¢.		¢.		
Derivative instruments (a)	\$	-	\$	-	\$	-	\$	-	\$	-	
Nuclear decommissioning trust ^(b) Equity securities		76.6				76.6					
Debt securities		70.0		-		70.0		-		-	
U.S. Treasury		8.4				8.4					
U.S. Agency		5.3		_		0.4		5.3		_	
State and local obligations		3.3 2.4		-		-		2.4		-	
Corporate bonds		24.8		_		_		24.8		-	
Foreign governments		0.7		_		_		0.7		_	
Other		(0.3)		_		_		(0.3)		_	
Total nuclear decommissioning trust		117.9				85.0		32.9			
Total Tuctear decommissioning trust	-	117.9				85.0		32.9			
Liabilities		117.9				05.0		32.9			
Derivative instruments ^(a)		_		(0.1)		0.1					
	\$		\$		\$	0.1	\$		\$		
Total	3		Ъ	(0.1)	Þ	0.1	Э	-	Þ		
Other Great Plains Energy											
Assets Derivative instruments (a)	\$	9.2	\$		\$	0.1	\$	5.4	\$	3.7	
SERP rabbi trust ^(c)	Ф	9.2	Ф	-	Ф	0.1	Þ	5.4	Ф	3./	
Equity securities		0.2				0.2					
Debt securities		7.0		_		0.2		7.0		_	
		7.0				0.2		7.0			
Total SERP rabbi trust				-		0.2				-	
Total		16.4		-		0.3		12.4		3.7	
Liabilities											
Derivative instruments (a)		25.1		(4.2)		4.2		25.1		-	
Total	\$	25.1	\$	(4.2)	\$	4.2	\$	25.1	\$	-	
Great Plains Energy											
Assets											
Derivative instruments (a)	\$	9.2	\$	-	\$	0.1	\$	5.4	\$	3.7	
Nuclear decommissioning trust (b)		117.9		-		85.0		32.9		-	
SERP rabbi trust (c)		7.2		-		0.2		7.0			
Total		134.3		-		85.3		45.3		3.7	
Liabilities											
Derivative instruments (a)		25.1		(4.3)		4.3		25.1		-	
Total	\$	25.1	\$	(4.3)	\$	4.3	\$	25.1	\$	-	

					Fair Value Measurements Using					
Description	Decem 20	ber 31 09	Netti	Quoted Prices in Active Markets for Identical Assets Netting ^(d) (Level 1)		Significant Other Observable Inputs (Level 2)		Significant Unobservable Inputs (Level 3)		
KCP&L					(millio	ns)				
Assets				(0.5)			4			
Derivative instruments (a)	\$	0.2	\$	(0.2)	\$	0.2	\$	0.2	\$	-
Nuclear decommissioning trust (b)										
Equity securities		44.5		-		44.5		-		-
Debt securities		11.0				44.0				
U.S. Treasury		11.2		-		11.2		-		-
U.S. Agency		3.5		-		-		3.5		- 0.2
State and local obligations		3.1 18.9		-		-		2.9		0.2
Corporate bonds		18.9 0.7		-		-		18.9		-
Foreign governments		1.2		-		-		0.7 1.2		-
Other						-				- 0.2
Total nuclear decommissioning trust		83.1		- (0.0)		55.7		27.2		0.2
Total		83.3		(0.2)		55.9		27.4		0.2
Liabilities										
Derivative instruments (a)		-		(0.2)		-		0.2		-
Total	\$	-	\$	(0.2)	\$	-	\$	0.2	\$	-
Other Great Plains Energy										
Assets										
Derivative instruments (a)	\$	9.2	\$	(0.7)	\$	0.7	\$	5.1	\$	4.1
SERP rabbi trust ^(c)										
Equity securities		0.2		-		0.2		-		-
Debt securities		6.9		-		-		6.9		-
Total SERP rabbi trust		7.1		-		0.2		6.9		-
Total		16.3		(0.7)		0.9		12.0		4.1
Liabilities										
Derivative instruments (a)		8.0		(1.6)		1.6		0.8		-
Total	\$	8.0	\$	(1.6)	\$	1.6	\$	0.8	\$	-
Great Plains Energy										
Assets										
Derivative instruments (a)	\$	9.4	\$	(0.9)	\$	0.9	\$	5.3	\$	4.1
Nuclear decommissioning trust (b)		83.1		-		55.7		27.2		0.2
SERP rabbi trust (c)		7.1		-		0.2		6.9		-
Total		99.6		(0.9)		56.8		39.4		4.3
Liabilities										
Derivative instruments (a)		8.0		(1.8)		1.6		1.0		-
Total	\$	0.8	\$	(1.8)	\$	1.6	\$	1.0	\$	_

- (a) The fair value of derivative instruments is estimated using market quotes, over-the-counter forward priced and volatility curves and correlations among fuel prices, net of estimated credit risk.
- (b) Fair value is based on quoted market prices of the investments held by the fund and/or valuation models. The total does not include \$2.8 million and \$29.4 million at September 30, 2010, and December 31, 2009, respectively, of cash and cash equivalents, which are not subject to the fair value requirements.
- (c) Fair value is based on quoted market prices of the investments held by the fund and/or valuation models. The total does not include \$15.0 million and \$16.2 million at September 30, 2010, and December 31, 2009, respectively, of cash and cash equivalents, which are not subject to the fair value requirements.
- (d) Represents the difference between derivative contracts in an asset or liability position presented on a net basis by counterparty on the consolidated balance sheet where a master netting agreement exists between the Company and the counterparty. At September 30, 2010, and December 31, 2009, Great Plains Energy netted \$4.3 million and \$0.9 million, respectively, of cash collateral posted with counterparties.

The following tables reconcile the beginning and ending balances for all level 3 assets and liabilities, net, measured at fair value on a recurring basis for the three months ended and year to date September 30, 2010 and 2009.

Fair Value Measurements Using Significant Unobservable Inputs (Level 3)		
	- (Other
	(Great
	J	Plains
	E	Energy
	De	rivative
	Inst	truments
	(n	nillions)
Balance July 1, 2010	\$	5.2
Total realized/unrealized gains or (losses)		
Included in non-operating income		(3.6)
Settlements		2.1
Balance September 30, 2010	\$	3.7
Total unrealized gains and (losses) included in non-operating		
income relating to assets and liabilities still on the		
consolidated balance sheet at September 30, 2010	\$	(1.3)

	KCP&L		P	Other Great Plains nergy	P	Great lains nergy
		State & Local Obligations				Total (
			(milli	ions)		
Balance January 1, 2010	\$	0.2	\$	4.1	\$	4.3
Total realized/unrealized gains or (losses)						
Included in non-operating income		-		(9.5)		(9.5)
Sales		(0.2)		-		(0.2)
Settlements		-		9.1		9.1
Balance September 30, 2010	\$	_	\$	3.7	\$	3.7

			КСР&	·L				Gi Pl	ther reat ains ergy	P	Freat lains nergy
Description		J.S. ency	 & Local gations	Ba	tgage cked irities	7	Total		vative uments	1	Total
					(milli	,					
Balance July 1, 2009	\$	2.9	\$ 0.5	\$	3.5	\$	6.9	\$	3.4	\$	10.3
Total realized/unrealized gains or (losses)											
Included in regulatory liability		(0.1)	-		0.6		0.5		-		0.5
Included in non-operating income		-	-		-		-		1.0		1.0
Purchase, issuances, and settlements		-	-		1.3		1.3		(0.1)		1.2
Transfers in and/or out of Level 3		-	(0.1)		-		(0.1)		-		(0.1)
Balance September 30, 2009	\$	2.8	\$ 0.4	\$	5.4	\$	8.6	\$	4.3	\$	12.9

			КСР	&L			Other Great Plains Energy	Great Plains Energy
Description	U.S. Agency	State & Lo		Mortga Backe Securit	eď	Total	Derivative Instruments	Total
					(million	s)		
Balance January 1, 2009	\$ 3.9	\$	-	\$	2.9	\$ 6.8	\$ 3.8	\$10.6
Total realized/unrealized gains or (losses)								
Included in regulatory liability	0.1		-		0.8	0.9	-	0.9
Included in non-operating income	-		-		-	-	1.2	1.2
Purchase, issuances, and settlements	(1.2)		-		1.7	0.5	(0.7)	(0.2)
Transfers in and/or out of Level 3	-	0	.4		-	0.4	-	0.4
Balance September 30, 2009	\$ 2.8	\$ 0	.4	\$	5.4	\$ 8.6	\$ 4.3	\$12.9
Total unrealized gains and (losses) included in non-operating income relating to assets and liabilities still on the consolidated balance sheet at September 30, 2009	\$ -	\$	_	\$	_	\$ -	\$ 0.9	\$ 0.9

16. TAXES

Components of income tax expense are detailed in the following tables.

	Three Mo Septo	onths End ember 30	Year to Date September 30				
Great Plains Energy	2010		2009		2010		2009
Current income taxes			(milli	ons)			
Federal	\$ (7.2)	\$	16.1	\$	(8.1)	\$	(1.3)
State	(2.0)		1.1		0.3		(2.6)
Foreign	 (0.2)		-		0.2		1.2
Total	 (9.4)		17.2		(7.6)		(2.7)
Deferred income taxes							
Federal	73.5		(5.8)		108.2		(13.6)
State	 14.6		4.0		18.8		8.8
Total	 88.1		(1.8)		127.0		(4.8)
Noncurrent income taxes							
Federal	(6.8)		7.7		(1.3)		6.8
State	(1.0)		0.9		(0.3)		8.0
Foreign	 0.2		0.6		0.2		(1.9)
Total	 (7.6)		9.2		(1.4)		5.7
Investment tax credit							
Deferral	-		12.2		(4.1)		28.4
Amortization	(0.7)		(0.6)		(1.8)		(1.7)
Total	(0.7)		11.6		(5.9)		26.7
Total income tax expense	70.4		36.2		112.1		24.9
Less: taxes on discontinued operations							
Deferred tax expense (benefit)	-		0.6		-		(1.4)
Income tax expense on continuing operations	\$ 70.4	\$	35.6	\$	112.1	\$	26.3

	Three Month Septembe		Year to Date September 30			
KCP&L	2010	2009	2010	2009		
Current income taxes		(million	ıs)			
Federal	\$ (15.3)	\$ 41.4	\$ 22.5	\$ 51.9		
State	(2.6)	5.0	4.6	6.9		
Total	(17.9)	46.4	27.1	58.8		
Deferred income taxes						
Federal	63.9	(34.8)	53.9	(49.6)		
State	11.7	(1.0)	9.5	-		
Total	75.6	(35.8)	63.4	(49.6)		
Noncurrent income taxes						
Federal	(2.4)	3.6	(1.5)	2.9		
State	(0.3)	0.4	(0.1)	0.4		
Total	(2.7)	4.0	(1.6)	3.3		
Investment tax credit						
Deferral	-	12.2	(4.1)	28.4		
Amortization	(0.5)	(0.4)	(1.2)	(1.1)		
Total	(0.5)	11.8	(5.3)	27.3		
Total	\$ 54.5	\$ 26.4	\$ 83.6	\$ 39.8		

Great Plains Energy's income tax expense increased \$34.8 million and \$85.8 million for the three months ended and year to date September 30, 2010, respectively, compared to the same periods in 2009 due to increased pre-tax income. Year to date September 30, 2009, reflects a \$16.0 million tax benefit due to the settlement of GMO's 2003-2004 tax audit. A significant portion of total income tax expense in 2010 is reflected as deferred income tax expense due to the utilization of net operating losses and accelerated tax deductions, including bonus depreciation, which reduce the current tax liability of Great Plains Energy.

KCP&L's income tax expense increased \$28.1 million and \$43.8 million for the three months ended and year to date September 30, 2010, respectively, compared to the same periods in 2009 due to increased pre-tax income. A significant portion of total income tax expense in 2010 is reflected as deferred income tax expense due to accelerated tax deductions, including bonus depreciation, which reduce the current tax liability of KCP&L.

Income Tax Expense and Effective Income Tax Rates

Income tax expense and the effective income tax rates reflected in the financial statements and the reasons for their differences from the statutory federal rates are detailed in the following tables.

Great Plains Energy	Iı	ncome Ta	х Ехр	ense	Income Tax	x Rate
Three Months Ended September 30	2	010	2009		2010	2009
		(milli	ons)			
Federal statutory income tax	\$	70.8	\$	39.9	35.0 %	35.0 %
Differences between book and tax						
depreciation not normalized		(0.9)		(2.4)	(0.4)	(2.1)
Amortization of investment tax credits		(0.7)		(0.6)	(0.4)	(0.5)
Federal income tax credits		(2.0)		(1.9)	(0.9)	(1.7)
State income taxes		7.5		4.0	3.7	3.5
Changes in uncertain tax positions, net		0.3		8.0	0.2	0.6
Valuation allowance		(2.9)		-	(1.4)	-
Other		(1.7)		(4.2)	(1.0)	(3.5)
Total	\$	70.4	\$	35.6	34.8%	31.3%

Great Plains Energy	Iı	ıcome Tax	Expe	ense	Income Tax Rate			
Year to Date September 30	2	2010		009	2010	2009		
		(millio	ons)					
Federal statutory income tax	\$	115.0	\$	57.1	35.0 %	35.0 %		
Differences between book and tax								
depreciation not normalized		(6.0)		(8.6)	(1.8)	(5.3)		
Amortization of investment tax credits		(1.8)		(1.7)	(0.5)	(1.0)		
Federal income tax credits		(6.1)		(6.5)	(1.8)	(4.0)		
State income taxes		12.0		5.4	3.7	3.3		
Medicare Part D subsidy legislation		2.8		-	0.8	-		
Changes in uncertain tax positions, net		0.3		(72.5)	0.1	(44.5)		
Valuation allowance		(2.9)		56.5	(0.9)	34.6		
Other		(1.2)		(3.4)	(0.5)	(2.0)		
Total	\$	112.1	\$	26.3	34.1 %	16.1 %		

KCP&L	In	come Tax	Expe	nse	Income Tax	x Rate
Three Months Ended September 30	20	010	2009		2010	2009
		(millio	ons)			
Federal statutory income tax	\$	51.5	\$	32.2	35.0 %	35.0 %
Differences between book and tax						
depreciation not normalized		(1.1)		(3.0)	(8.0)	(3.2)
Amortization of investment tax credits		(0.5)		(0.4)	(0.4)	(0.4)
Federal income tax credits		(1.9)		(1.6)	(1.3)	(1.8)
State income taxes		5.7		3.0	3.9	3.2
Changes in uncertain tax positions, net		0.1		(0.1)	0.1	(0.1)
Other		0.7		(3.7)	0.6	(4.0)
Total	\$	54.5	\$	26.4	37.1 %	28.7 %

KCP&L	In	come Tax	Expe	nse	Income Tax Rate			
Year to Date September 30	20	10	20	009	2010	2009		
		(millio	ns)					
Federal statutory income tax	\$	85.3	\$	52.0	35.0 %	35.0 %		
Differences between book and tax								
depreciation not normalized		(5.2)		(6.8)	(2.2)	(4.6)		
Amortization of investment tax credits		(1.2)		(1.1)	(0.5)	(0.7)		
Federal income tax credits		(6.0)		(5.8)	(2.5)	(3.9)		
State income taxes		8.8		4.8	3.6	3.2		
Medicare Part D subsidy legislation		2.8		-	1.2	-		
Changes in uncertain tax positions, net		0.1		-	0.1	-		
Other		(1.0)		(3.3)	(0.4)	(2.2)		
Total	\$	83.6	\$	39.8	34.3 %	26.8 %		

Advanced Coal Credit

In April 2008, KCP&L was notified that its application filed in 2007 for \$125.0 million in advanced coal investment tax credits (ITC) was approved by the IRS. The credit is based on the amount of expenses incurred on the construction of Iatan No. 2. Additionally, in order to meet the advanced clean coal standards and avoid forfeiture and/or the recapture of tax credits in the future, KCP&L must meet or exceed certain environmental performance standards for at least five years once the plant is placed in service.

In September 2010, the IRS issued an amended memorandum of understanding to reallocate \$17.7 million of the original \$125 million of the advanced coal project credits to Empire, meeting the requirements of an arbitration order issued on December 30, 2009. See Note 12 for the related legal proceeding. As a result, Great Plains Energy and KCP&L reduced the amount of advanced coal credit previously recognized. The amount of deferred federal tax expense associated with the reduction year to date September 30, 2010, was \$4.1 million. Since the tax laws require KCP&L to reduce income tax expense for ratemaking and financial statement purposes ratably over the life of the plant, Great Plains Energy and KCP&L concurrently recognized a separate deferred advanced coal ITC benefit to offset the current and deferred federal tax expense. Great Plains Energy and KCP&L recognized an insignificant amount of tax benefits of the ITC in the third quarter of 2010 when the plant was placed in service and will continue to recognize the tax benefits over the life of the plant. At September 30, 2010, Great Plains Energy and KCP&L had \$107.1 million of deferred advanced coal ITC.

Uncertain Tax Positions

At September 30, 2010, and December 31, 2009, Great Plains Energy had \$52.4 million and \$51.4 million, respectively, of liabilities related to unrecognized tax benefits. Of these amounts, \$17.5 million and \$17.3 million, respectively, at September 30, 2010 and December 31, 2009, are expected to impact the effective tax rate, if recognized.

At September 30, 2010, and December 31, 2009, KCP&L had \$19.3 million and \$20.9 million, respectively, of liabilities related to unrecognized tax benefits. Of these amounts, \$0.4 million at September 30, 2010, and December 31, 2009, are expected to impact the effective tax rate, if recognized.

The following table reflects activity for Great Plains Energy and KCP&L related to the liability for unrecognized tax benefits..

	Great Plains Energy				KCP&L				
	•	ptember 30 2010		ember 31 2009	Septem 20			ember 31 2009	
				(millio	ons)				
Beginning balance	\$	51.4	\$	97.3	\$	20.9	\$	17.6	
Additions for current year tax positions		2.0		13.2		1.0		3.9	
Additions for prior year tax positions		3.5		8.2		1.6		3.0	
Additions for GMO prior year tax positions		-		11.6		-		-	
Reductions for prior year tax positions		(1.8)		(1.3)		(1.3)		(8.0)	
Settlements		(2.9)		(76.7)		(2.9)		(2.2)	
Statute expirations		-		(0.7)		-		(0.6)	
Foreign currency translation adjustments		0.2		(0.2)		-		-	
Ending balance	\$	52.4	\$	51.4	\$	19.3	\$	20.9	

Great Plains Energy and KCP&L recognize interest accrued related to unrecognized tax benefits in interest expense and recognize penalties related to unrecognized tax benefits in non-operating expenses. At September 30, 2010, and December 31, 2009, accrued interest related to unrecognized tax benefits for Great Plains Energy was \$6.5 million and \$5.9 million, respectively. Amounts accrued for penalties related to unrecognized tax benefits for Great Plains Energy were \$1.1 million at September 30, 2010, and December 31, 2009. KCP&L had accrued interest related to unrecognized tax benefits of \$1.5 million and \$1.7 million at September 30, 2010, and December 31, 2009, respectively. Amounts accrued for penalties related to unrecognized tax benefits for KCP&L are insignificant.

In July 2010, the Joint Committee on Taxation approved the settlement of the IRS audit of Great Plains Energy's 2005 tax year. Great Plains Energy and KCP&L recognized \$2.9 million of unrecognized tax benefits in the third quarter of 2010 associated with this settlement. The IRS is currently auditing Great Plains Energy and its subsidiaries for the 2006-2008 tax years and the Company is protesting an audit assessment by the Canada Revenue Authority (CRA) against a former GMO subsidiary for the 2002 tax year. The Company is unable to estimate the amount of unrecognized tax benefits that may be recognized in the next twelve months.

17. SEGMENTS AND RELATED INFORMATION

Great Plains Energy

Great Plains Energy has one reportable segment based on its method of internal reporting, which generally segregates reportable segments based on products and services, management responsibility and regulation. The one reportable business segment is electric utility, consisting of KCP&L and GMO's regulated utility operations. Other includes GMO activity other than its regulated utility operations, Services, KLT Inc., unallocated corporate charges, consolidating entries and intercompany eliminations. Intercompany eliminations include insignificant amounts of intercompany financing-related activities. The summary of significant accounting policies applies to the reportable segment. Segment performance is evaluated based on net income attributable to Great Plains Energy.

Three Months Ended	Ele	Great Plains				
September 30, 2010	Ut	Utility Other			Energy	
			(mill	ions)		
Operating revenues	\$	728.8	\$	-	\$	728.8
Depreciation and amortization		(85.3)		-		(85.3)
Interest charges		(34.8)	((10.7)		(45.5)
Income tax (expense) benefit		(77.0)		6.6		(70.4)
Net income (loss) attributable to Great Plains Energy		136.2		(4.2)		132.0

Year to Date		Electric			Great Plains		
September 30, 2010	U	Utility Other			Other Ener		
			(m	illions)			
Operating revenues	\$	1,787.7	\$	-	\$	1,787.7	
Depreciation and amortization		(248.5)		-		(248.5)	
Interest charges		(107.4)		(31.3)		(138.7)	
Income tax (expense) benefit		(125.4)		13.3		(112.1)	
Loss from equity investments		-		(0.9)		(0.9)	
Net income (loss) attributable to Great Plains Energy		232.8		(16.2)		216.6	

Three Months Ended	Ele	ectric				Great Plains		
September 30, 2009	Ut	tility	Other		En	ergy		
		(millions)						
Operating revenues	\$	587.7	\$	-	\$	587.7		
Depreciation and amortization		(77.9)		-		(77.9)		
Interest charges		(38.6)	((10.4)		(49.0)		
Income tax (expense) benefit		(42.2)		6.6		(35.6)		
Loss from equity investments		-		(0.2)		(0.2)		
Discontinued operations		-		8.0		8.0		
Net income (loss) attributable to Great Plains Energy		83.9		(4.8)		79.1		

Year to Date	Electric	G	Great Plains				
September 30, 2009	Utility	y Other		Energy			
	(millions)						
Operating revenues	\$ 1,487.4	\$	- \$	1,487.4			
Depreciation and amortization	(220.3)	,	-	(220.3)			
Interest charges	(113.6)	(19.6	5)	(133.2)			
Income tax (expense) benefit	(55.5)	29.2	2	(26.3)			
Loss from equity investments	-	(0.4	1)	(0.4)			
Discontinued operations	-	(2.3	3)	(2.3)			
Net income attributable to Great Plains Energy	134.1	0.4	1	134.5			

	Electric Utility		Other		inations		eat Plains Energy	
September 30, 2010	(millions)							
Assets	\$ 9,115.2	\$	97.1	\$	(413.2)	\$	8,799.1	
Capital expenditures (a)	465.3		-		-		465.3	
December 31, 2009								
Assets	\$ 8,765.3	\$	152.5	\$	(435.0)	\$	8,482.8	
Capital expenditures (a)	841.3		-		-		841.3	

⁽a) Capital expenditures reflect year to date amounts for the periods presented.

18. DISCONTINUED OPERATIONS

In 2008, Great Plains Energy sold Strategic Energy, L.L.C. to Direct Energy Services, LLC (Direct Energy), a subsidiary of Centrica. In the second quarter of 2009, Great Plains Energy recorded \$5.1 million of gross receipts taxes for periods prior to the sale for which Great Plains Energy indemnified Direct Energy. In the third quarter of 2009, Great Plains Energy reduced its previously recorded reserve of \$2.0 million related to indemnification obligations by \$1.4 million. Corresponding income tax expense of \$0.6 million and a \$1.4 million tax benefit for the three months ended and year to date September 30, 2009, respectively, resulted in \$0.8 million income and \$2.3 million loss from discontinued operations, net of income taxes for the three months ended and year to date September 30, 2009, respectively.

19. GOODWILL

Goodwill is required to be tested for impairment at least annually and more frequently when indicators of impairment exist. The annual impairment test for the GMO acquisition goodwill was conducted on September 1, 2010. The goodwill impairment test is a two step process, the first step of which is the comparison of the fair value of a reporting unit to its carrying amount, including goodwill, to identify potential impairment. If the carrying amount exceeds the fair value of the reporting unit, the second step of the test is performed, consisting of assignment of the reporting unit's fair value to its assets and liabilities to determine an implied fair value of goodwill which is compared to the carrying amount of goodwill to determine the impairment loss, if any, to be recognized in the financial stat ements. Great Plains Energy's regulated electric utility operations are considered one reporting unit for assessment of impairment, as they are included within the same operating segment and have similar economic characteristics. The determination of fair value of the reporting unit consisted of two valuation techniques: an income approach consisting of a discounted cash flow analysis and a market approach consisting of a determination of reporting unit invested capital using market multiples derived from the historical revenue, EBITDA and net utility asset values and market prices of stock of electric and gas company regulated peers. The results of the two techniques were evaluated and weighted to determine a point within the range that management considered representative of fair value for the reporting unit. Fair value of the reporting unit exceeded the carrying amount, including goodwill; therefore, there was no impairment of goodwill.

ITEM 2. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

GREAT PLAINS ENERGY INCORPORATED

EXECUTIVE SUMMARY

Description of Business

Great Plains Energy is a public utility holding company and does not own or operate any significant assets other than the stock of its subsidiaries. Great Plains Energy's direct subsidiaries are KCP&L, GMO, KLT Inc. and Services. Great Plains Energy's sole reportable business segment is electric utility.

Electric utility consists of KCP&L, a regulated utility, and GMO's regulated utility operations, which include its Missouri Public Service and St. Joseph Light & Power divisions. Electric utility has over 6,000 MWs of generating capacity and engages in the generation, transmission, distribution and sale of electricity to over 820,000 customers in the states of Missouri and Kansas. Electric utility's retail electricity rates are below the national average of investor-owned utilities.

Earnings Overview

Great Plains Energy's earnings available for common shareholders for the three months ended September 30, 2010, were \$131.6 million or \$0.96 per share compared to \$78.7 million, or \$0.58 per share, for the same period in 2009. Electric utility's net income increased \$52.3 million for the three months ended September 30, 2010, compared to the same period in 2009 primarily driven by an increase in gross margin due to new retail rates and favorable weather. Gross margin is a financial measure that is not calculated in accordance with GAAP. See the explanation of gross margin and the reconciliation to GAAP operating revenues under Great Plains Energy's Results of Operations for further information. Partially offsetting the increase in gross margin was increased depreciation and amo rtization expense due to additional regulatory amortization pursuant to KCP&L's 2009 rate cases and depreciation from placing Iatan No. 2 in service in the third quarter of 2010 and increased general taxes. KCP&L supported a recommended disallowance of certain Iatan No. 2 construction costs in its Kansas rate case post-hearing brief filed in September 2010. Great Plains Energy recorded a \$4.0 million pre-tax loss in the third quarter of 2010 for KCP&L's and GMO's combined share of these Iatan No. 2 construction costs.

Great Plains Energy's earnings available for common shareholders year to date September 30, 2010, were \$215.4 million or \$1.57 per share compared to \$133.3 million, or \$1.04 per share, for the same period in 2009. Electric utility's net income increased \$98.7 million year to date September 30, 2010, compared to the same period in 2009 primarily driven by an increase in gross margin due to new retail rates and favorable weather. Partially offsetting the increase in gross margin was higher operations and maintenance expense driven by planned plant outages, increased depreciation and amortization expense due to additional regulatory amortization pursuant to KCP&L's 2009 rate cases and depreciation from placing in service the Iatan No. 1 environmental equipment in April 2009 and Iatan No. 2 in the t hird quarter of 2010 and increased general taxes. KCP&L supported a recommended disallowance of certain Iatan No. 2 construction costs in its Kansas rate case post-hearing brief filed in September 2010. Great Plains Energy recorded a \$4.0 million pre-tax loss in the third quarter of 2010 for KCP&L's and GMO's combined share of these Iatan No. 2 construction costs. Great Plains Energy's corporate and other activities had a loss of \$17.4 million year to date September 30, 2010, compared to income of \$1.5 million for the same period in 2009. Year to date September 30, 2010, reflects an additional \$6.9 million of after-tax interest expense for Equity Units issued in May 2009 and year to date September 30, 2009, reflects a \$16.0 million tax benefit due to the settlement of GMO's 2003-2004 tax audit. Additionally, year to date September 30, 2009, includes a \$2.3 million loss from the discontinued operations of Strategic Energy.

Comprehensive Energy Plan - Iatan No. 2

In April 2010, Great Plains Energy and KCP&L announced the results of a cost and schedule reforecast for Iatan No. 2. The current and previous cost estimate ranges are shown in the following table. The cost estimate ranges do not include allowance for funds used during construction or the cost of common facilities that were identified at the time of the start-up of the Iatan No. 1 environmental project that will be used by both Iatan No. 1 and Iatan No. 2.

	Currei F	timate e	Previo	Change						
		J		(mil	lions)				
Great Plains Energy's 73% share of Iatan No. 2	\$ 1,222	-	\$ 1,251	\$ 1,153	-	\$ 1,201	\$	69	-	\$ 50
KCP&L's 55% share of Iatan No. 2	919	-	941	868	-	904		51	-	37

The increase in the cost estimate ranges was primarily due to a shift in the expected in-service date, the impact of lower wholesale prices on expected test power revenues that offset construction cost, and a level of contingency management considered appropriate in light of recent start-up events encountered at other coal plants under construction. In August 2010, Iatan No. 2 successfully completed in-service testing, which was confirmed by KCC in October 2010, but is still subject to confirmation by the MPSC, which is expected during the current rate cases. The Companies estimate that the final cost will be within the current estimate range.

Regulatory Proceedings

The following table summarizes the initial filing information in currently pending requests for retail rate increases with KCC and the MPSC.

Rate Jurisdiction	File Date	Annual Revenue Increase	Return on Equity	Rate-Making Equity Ratio						
		(millions)								
KCP&L - Kansas (a)	12/17/2009	\$ 55.2 (b)	11.25% (b)	46.17%						
KCP&L - Missouri (c)	6/4/2010	92.1	11.00%	46.16%						
GMO - Missouri Public Service division (c)	6/4/2010	75.8	11.00%	46.16%						
GMO - St. Joseph Light & Power division (c)	6/4/2010	22.1	11.00%	46.16%						

- (a) The request includes costs related to Iatan No. 2, a new coal-fired generation unit, upgrades to the transmission and distribution system to improve reliability and overall increased costs of service. Any authorized changes to retail rates are expected to be effective in December 2010.
- (b) The requested increase was adjusted by KCP&L during the hearing process to \$50.9 million as the net result of updates to the case, most notably a reduction in the requested return on equity. The requested return on equity was adjusted by KCP&L to 10.75% with the potential for a 0.25% adder (to 11.00%) if KCC adopts a particular rate design proposal by KCC Staff and other interveners.
- (c) The request includes costs related to Iatan No. 2, a new coal-fired generation unit, upgrades to the transmission and distribution system to improve reliability and overall increased costs of service. For KCP&L, it also includes increased coal transportation costs expected with the expiration in 2010 of the majority of KCP&L's current coal transportation contracts. Any authorized changes to retail rates are expected to be effective in May 2011 for KCP&L and June 2011 for GMO.

In June 2010, KCC staff and certain interveners filed their direct testimony in KCP&L's rate case. KCC staff recommended a rate reduction of \$9.1 million. The main differences between KCP&L's and KCC staff's positions are the staff's proposals for a return on equity of 9.70%, disallowance of \$231 million of the total project cost of Iatan No. 2 (\$58 million for KCP&L's Kansas jurisdictional portion of the plant) and depreciation rate differences. KCP&L supported a recommended disallowance of certain Iatan No. 2 construction costs in its

post-hearing brief filed in September 2010. Great Plains Energy recorded a \$4.0 million pre-tax loss in the third quarter of 2010 for KCP&L's and GMO's combined share of these Iatan No. 2 construction costs. Management does not believe a further write-down of the Iatan No. 2 plant is appropriate under regulatory prudence standards. In the event of a further disallowance of certain Iatan No. 2 costs, KCP&L would recognize a loss with a corresponding write-down of utility plant for the amount of disallowance. The outcome of the KCP&L Kansas case will likely be different from either of the positions of KCP&L or KCC staff, though the decision of KCC cannot be predicted. Evidentiary hearings in the case were held August 16, 2010, through September 2, 2010. The case has been fully briefed and a KCC order is expected in November 2010 with new rates expected to be effective in December 2010.

In September 2010, GMO received an order from the MPSC approving construction accounting for the Iatan No. 2 project from the Iatan No. 2 in-service date to the effective date of new rates in the current rate case. The effect of the order is to defer GMO's share of Iatan No. 2 operating costs, depreciation expense and carrying costs (interest) to a regulatory asset rather than impacting the income statement until new rates are effective. KCP&L (Missouri jurisdiction only) was granted construction accounting as part of the Comprehensive Energy Plan. The next major milestone in the MPSC cases is November 2010, when the MPSC staff will file its Iatan No. 2 prudence report and the MPSC Staff and other interveners will file direct testimony. Hearings are scheduled for late January 2011 for KCP&L and early February 2011 for GMO. New rates are proposed to go into effect in May 2011 for KCP&L and June 2011 for GMO.

Transmission Investment Opportunities

In September 2010, GMO accepted a Notification to Construct from SPP for the Missouri portion of a 175-mile, 345kV transmission line in GMO's service territory from Sibley, Missouri to Nebraska City, Nebraska. Construction is expected to occur over 2012 to 2017, with an estimated cost of about \$380 million for GMO's portion of the line. This line is one of a number of priority projects that the SPP has developed as part of its transmission expansion plans for the region. In June 2010, FERC approved the SPP's proposed cost allocation method for these projects. KCP&L has also accepted a Notification to Construct from SPP for a 30-mile, 345kV transmission line, with estimated construction costs of \$54 million and 2015 in-service date, from KCP&L's Iatan generating station to KCP&mam p;L's Nashua substation. GMO and KCP&L will have the obligation to build their separate lines, which may be done solely or with other entities, unless the obligation is transferred to another qualified transmission owner. GMO has not determined which of these alternative courses of action to pursue. SPP retains the authority to revise or withdraw existing Notifications to Construct for transmission projects based upon emerging transmission plans and the associated needs for specific projects.

Rail Transportation Contracts

KCP&L and GMO have rail transportation contracts with various railroads to transport coal to their generating units. The majority of KCP&L's and GMO's rail transportation contracts expire at the end of 2010. KCP&L and GMO anticipate entering into new rail transportation contracts in the fourth quarter of 2010. After 2010, rail transportation costs are anticipated to be significantly higher as a result of the new contracts.

ENVIRONMENTAL MATTERS

See Note 11 to the consolidated financial statements for information regarding environmental matters.

RELATED PARTY TRANSACTIONS

See Note 13 to the consolidated financial statements for information regarding related party transactions and relationships.

GREAT PLAINS ENERGY RESULTS OF OPERATIONS

The following table summarizes Great Plains Energy's comparative results of operations.

		Three Months Ended September 30				
	2010	2009	2010	2009		
		(mill	ions)			
Operating revenues	\$ 728.8	\$ 587.7	\$ 1,787.7	\$ 1,487.4		
Fuel	(127.3)	(118.1)	(333.2)	(302.5)		
Purchased power	(68.0)	(46.1)	(171.4)	(140.9)		
Transmission of electricity by others	(8.1)	(7.9)	(20.9)	(21.2)		
Gross margin (a)	525.4	415.6	1,262.2	1,022.8		
Other operating expenses	(196.3)	(186.5)	(573.0)	(540.1)		
Depreciation and amortization	(85.3)	(77.9)	(248.5)	(220.3)		
Operating income	243.8	151.2	440.7	262.4		
Non-operating income and expenses	4.1	12.0	27.7	34.5		
Interest charges	(45.5)	(49.0)	(138.7)	(133.2)		
Income tax expense	(70.4)	(35.6)	(112.1)	(26.3)		
Loss from equity investments	-	(0.2)	(0.9)	(0.4)		
Income from continuing operations	132.0	78.4	216.7	137.0		
Income (loss) from discontinued operations	-	0.8	-	(2.3)		
Net income	132.0	79.2	216.7	134.7		
Less: Net income attributable to noncontrolling interest	-	(0.1)	(0.1)	(0.2)		
Net income attributable to Great Plains Energy	132.0	79.1	216.6	134.5		
Preferred dividends	(0.4)	(0.4)	(1.2)	(1.2)		
Earnings available for common shareholders	\$ 131.6	\$ 78.7	\$ 215.4	\$ 133.3		

⁽a) Gross margin is a non-GAAP financial measure. See explanation of gross margin below.

Three Months Ended September 30, 2010 Compared to September 30, 2009

Great Plains Energy's earnings available for common shareholders for the three months ended September 30, 2010, increased to \$131.6 million, or \$0.96 per share, from \$78.7 million, or \$0.58 per share for the same period in 2009.

Electric utility's net income increased \$52.3 million for the three months ended September 30, 2010, compared to the same period in 2009 primarily driven by an increase in gross margin due to new retail rates and favorable weather. Partially offsetting the increase in gross margin was increased depreciation and amortization expense due to additional regulatory amortization pursuant to KCP&L's 2009 rate cases and depreciation from placing in service Iatan No. 2 in the third quarter of 2010 and increased general taxes. KCP&L supported a recommended disallowance of certain Iatan No. 2 construction costs in its Kansas rate case post-hearing brief filed in September 2010. Great Plains Energy recorded a \$4.0 million pre-tax loss in the third quarter of 2010 for KCP&L's and GMO'; s combined share of these Iatan No. 2 construction costs.

Year to Date September 30, 2010 Compared to September 30, 2009

Great Plains Energy's earnings available for common shareholders year to date September 30, 2010, increased to \$215.4 million, or \$1.57 per share, from \$133.3 million, or \$1.04 per share for the same period in 2009.

Electric utility's net income increased \$98.7 million year to date September 30, 2010, compared to the same period in 2009 primarily driven by an increase in gross margin due to new retail rates and favorable weather. Partially offsetting the increase in gross margin was higher operations and maintenance expense driven by planned plant outages, increased depreciation and amortization expense due to additional regulatory amortization pursuant to KCP&L's 2009 rate cases and depreciation from placing in service the Iatan No. 1 environmental equipment in April 2009 and Iatan No. 2 in the third quarter of 2010 and increased general taxes. KCP&L supported a recommended disallowance of certain Iatan No. 2 construction costs in its Kansas rate case post-hearing brief filed in September 2010. Gr eat Plains Energy recorded a \$4.0 million pre-tax loss in the third quarter of 2010 for KCP&L's and GMO's combined share of these Iatan No. 2 construction costs.

Great Plains Energy's corporate and other activities had a loss of \$17.4 million year to date September 30, 2010, compared to income of \$1.5 million for the same period in 2009. Year to date September 30, 2010, reflects an additional \$6.9 million of after-tax interest expense for Equity Units issued in May 2009 and year to date September 30, 2009, reflects a \$16.0 million tax benefit due to the settlement of GMO's 2003-2004 tax audit.

Gross Margin

Gross margin is a financial measure that is not calculated in accordance with GAAP. Gross margin, as used by Great Plains Energy and KCP&L, is defined as operating revenues less fuel, purchased power and transmission of electricity by others. Expenses for fuel, purchased power and transmission of electricity by others, offset by wholesale sales margin, are subject to recovery through cost adjustment mechanisms, except for KCP&L's Missouri retail operations. As a result, operating revenues increase or decrease in relation to a significant portion of these expenses. Management believes that gross margin provides a more meaningful basis for evaluating electric utility's operations across periods than operating revenues because gross margin excludes the revenue effect of fluctuat ions in these expenses. Gross margin is used internally to measure performance against budget and in reports for management and the Board. The Companies' definition of gross margin may differ from similar terms used by other companies.

ELECTRIC UTILITY RESULTS OF OPERATIONS

The following table summarizes the electric utility segment results of operations.

	Three Months Ended September 30					te 30
	2010	2009		2010		2009
		(1	nillions	lions)		
Operating revenues	\$ 728.8	\$ 587.7	\$	1,787.7	\$	1,487.4
Fuel	(127.3)	(118.1)		(333.2)		(302.5)
Purchased power	(68.0)	(46.1)		(171.4)		(140.9)
Transmission of electricity by others	(8.1)	(7.9)		(20.9)		(21.2)
Gross margin (a)	525.4	415.6		1,262.2		1,022.8
Other operating expenses	(195.3)	(183.2)		(570.0)		(529.3)
Depreciation and amortization	(85.3)	(77.9)		(248.5)		(220.3)
Operating income	244.8	154.5		443.7		273.2
Non-operating income and expenses	3.2	10.2		21.9		30.0
Interest charges	(34.8)	(38.6)		(107.4)		(113.6)
Income tax expense	(77.0)	(42.2)		(125.4)		(55.5)
Net income	\$ 136.2	\$ 83.9	\$	232.8	\$	134.1

⁽a) Gross margin is a non-GAAP financial measure. See explanation of gross margin under Great Plains Energy's Results of Operations.

The following tables summarize electric utility's gross margins and MWhs sold.

		Revenue	es and	Costs	%	MWhs	%	
Three Months Ended September 30		2010		2009	Change	2010	2009	Change
Retail revenues		(millions)				(thous	ands)	
Residential	\$	331.9	\$	243.3	36	3,010	2,401	25
Commercial		255.5		219.0	17	2,986	2,780	7
Industrial		54.6		48.9	11	847	824	3
Other retail revenues		3.9		3.9	1	26	26	(1)
Provision for rate refund (excess Missouri								
wholesale margin)		(2.2)		-	NA	NA	NA	NA
Fuel recovery mechanism under recovery		17.5		12.3	42	NA	NA	NA
Total retail		661.2		527.4	25	6,869	6,031	14
Wholesale revenues		55.0		49.4	11	1,756	1,708	3
Other revenues		12.6		10.9	16	NA	NA	NA
Operating revenues		728.8		587.7	24	8,625	7,739	11
Fuel		(127.3)		(118.1)	8			
Purchased power		(68.0)		(46.1)	48			
Transmission of electricity by others		(8.1)		(7.9)	3			
Gross margin (a)	\$	525.4	\$	415.6	26			

⁽a) Gross margin is a non-GAAP financial measure. See explanation of gross margin under Great Plains Energy's Results of Operations.

		Revenue	s and	Costs	%	MWhs Sold				%	
Year to Date September 30		2010	2009		Change	2	010	20	009	Change	e
Retail revenues		(millions)				(thousands)					
Residential	\$	748.5	\$	594.2	26		7,532	(6,621	1	L4
Commercial		655.9		569.4	15	8	3,334	8	8,035		4
Industrial		148.4		128.5	15	2	2,473	2	2,348		5
Other retail revenues		12.9		12.3	7		84		86	((3)
Provision for rate refund (excess Missouri											
wholesale margin)		(3.7)		-	NM		NA		NA	N	A
Fuel recovery mechanism under recovery		38.3		26.8	43		NA		NA	N	A
Total retail	1	1,600.3		1,331.2	20	18	3,423	17	7,090	- !	8
Wholesale revenues		151.7		123.1	23	4	4,658	3	3,962	1	18
Other revenues		35.7		33.1	8		NA		NA	N	A
Operating revenues	1	L,787.7		1,487.4	20	23	3,081	22	1,052	1	LO
Fuel		(333.2)		(302.5)	10						
Purchased power		(171.4)		(140.9)	22						
Transmission of electricity by others		(20.9)		(21.2)	(1)						
Gross margin (a)	\$ 1	1,262.2	\$	1,022.8	23						

⁽a) Gross margin is a non-GAAP financial measure. See explanation of gross margin under Great Plains Energy's Results of Operations.

Electric utility's gross margin increased \$109.8 million for the three months ended September 30, 2010, compared to the same period in 2009 primarily due to the increase in retail revenues driven by new retail rates effective August 1, 2009, and September 1, 2009, for Kansas and Missouri, respectively, and favorable summer weather with a 73% increase in cooling degree days.

Electric utility's gross margin increased \$239.4 million year to date September 30, 2010, compared to the same period in 2009 primarily due to the increase in retail revenues driven by new retail rates effective August 1, 2009, and September 1, 2009, for Kansas and Missouri, respectively, and favorable weather.

Retail MWhs sold year to date September 30, 2010, increased due to favorable weather, with a 15% increase in heating degree days and a 55% increase in cooling degree days. Wholesale MWhs sold increased due to a 10% increase in generation resulting in more MWhs available for sale, partially offset by the higher retail load requirements. The increase in generation was primarily a result of Iatan No. 1 being off-line from January though mid-April 2009 to complete an environmental upgrade and unit overhaul, with the expenditures being capitalized, and therefore, not impacting operating and maintenance expenses. The coal base load equivalent availability factor increased to 82% year to date September 30, 2010, compared to 77% for the same period in 2009.

Electric Utility Other Operating Expenses (including utility operating and maintenance expenses, general taxes and other)

Electric utility's other operating expenses increased \$12.1 million and \$40.7 million for the three months ended and year to date September 30, 2010, respectively, compared to the same periods in 2009. Operating and maintenance expenses increased \$2.5 million and \$24.2 million for the three months ended and year to date September 30, 2010, respectively, compared to the same periods in 2009 primarily driven by planned plant outages, amortization of deferred costs related to customer programs and an increase in spending for vegetation management, partially offset by \$7.5 million expensed in September 2009 after KCP&L exercised its option to terminate an agreement for the construction of a wind project. General taxes increased \$5.2 million and \$12.4 million for the three months ended and year to date September 30, 2010, respectively, compared to the same periods in 2009 driven by increased gross receipts taxes on increased retail revenues and increased property taxes. KCP&L supported a recommended disallowance of certain Iatan No. 2 construction costs in its Kansas rate case

post-hearing brief filed in September 2010. Great Plains Energy recorded a \$4.0 million pre-tax loss in the third quarter of 2010 for KCP&L's and GMO's combined share of these Iatan No. 2 construction costs.

Electric Utility Depreciation and Amortization

Electric utility's depreciation and amortization costs increased \$7.4 million and \$28.2 million for the three months ended and year to date September 30, 2010, respectively, compared to the same periods in 2009. Additional regulatory amortization pursuant to KCP&L's 2009 rate cases increased depreciation and amortization \$3.2 million and \$17.2 million for the three months ended and year to date September 30, 2010, respectively. The remaining increases were due to placing in service the Iatan No. 1 environmental equipment in April 2009, which only impacted the comparative year to date period, and the KCP&L Kansas portion of Iatan No. 2 depreciation expense in the third quarter of 2010, as well as normal depreciation activity for other capital additions.

Electric Utility Non-Operating Income and Expenses

Electric utility's non-operating income and expenses decreased \$7.0 million and \$8.1 million for the three months ended and year to date September 30, 2010, respectively, compared to the same periods in 2009 primarily due to a decrease in the equity component of AFUDC resulting from a lower average construction work in progress balance due to KCP&L's Comprehensive Energy Plan projects being placed in service.

Electric Utility Interest Charges

Electric utility's interest charges decreased \$3.8 million for the three months ended September 30, 2010, compared to the same period in 2009 primarily due to the deferral to a regulatory asset of construction accounting carrying costs for Iatan No. 1, Iatan No. 2 and common facilities and the maturity of \$68.5 million of GMO's 7.625% Senior Notes in December 2009, partially offset by a decrease in the debt component of AFUDC resulting from a lower average construction work in progress balance due to KCP&L's Comprehensive Energy Plan projects being placed in service and interest on an intercompany note from Great Plains Energy to GMO.

Electric utility's interest charges decreased \$6.2 million year to date September 30, 2010, compared to the same period in 2009 primarily due to the deferral to a regulatory asset of construction accounting carrying costs for Iatan No. 1, Iatan No. 2 and common facilities and the maturity of \$68.5 million of GMO's 7.625% Senior Notes in December 2009, partially offset by interest on KCP&L's \$400.0 million of 7.15% Mortgage Bonds Series 2009A issued in March 2009 and interest on an intercompany note from Great Plains Energy to GMO.

Electric Utility Income Tax Expense

Electric utility's income tax expense increased \$34.8 million for the three months ended September 30, 2010, compared to the same period in 2009 due to increased pre-tax income. Electric utility's income tax expense increased \$69.9 million year to date September 30, 2010, compared to the same period in 2009 due to increased pre-tax income and a \$2.8 million increase in income tax expense for the cumulative change in tax treatment of the Medicare Part D subsidy under the Federal health care reform legislation signed into law in the first quarter of 2010.

GREAT PLAINS ENERGY SIGNIFICANT BALANCE SHEET CHANGES (September 30, 2010 compared to December 31, 2009)

- · Great Plains Energy's receivables, net increased \$44.1 million primarily due to a \$73.4 million increase in customer accounts receivable resulting from seasonal increases partially offset by a \$16.3 million decrease in receivables from joint owners primarily related to the Iatan No. 2 project.
- · Great Plains Energy's accounts receivable pledged as collateral and collateralized note payable of \$95.0 million reflects the adoption on January 1, 2010, of new accounting guidance for transfers of financial assets. See Note 3 to the consolidated financial statements for additional information.
- · Great Plains Energy's electric utility plant increased \$1.5 billion primarily due to \$1.3 billion and \$80.5 million placed in service for Iatan No. 2 and the Iatan No. 1 environmental project and certain Iatan facility common costs, respectively, in addition to normal plant activity.

- · Great Plains Energy's construction work in progress decreased \$1.2 billion primarily due to projects placed in service as described above, in addition to normal plant activity.
- · Great Plains Energy's notes payable decreased \$230.0 million primarily due to repayment with proceeds from the issuance of \$250.0 million of 2.75% Senior Notes, partially offset by a \$6.9 million payment for the settlement of FSS and additional borrowings to support other normal operating activities.
- · Great Plains Energy's accounts payable decreased \$142.6 million primarily due to the timing of cash payments, including payments related to KCP&L's Comprehensive Energy Plan projects.
- · Great Plains Energy's accrued taxes increased \$58.9 million primarily due to the timing of property tax payments.
- · Great Plains Energy's derivative instruments current liabilities increased \$24.8 million primarily due to mark-to-market losses on Great Plains Energy's FSS, which is offset in OCI.
- · Great Plains Energy's deferred income taxes long-term increased \$162.4 million primarily due to an \$85.2 million increase in temporary differences mostly as a result of bonus depreciation and utilizing \$40.7 million of net operating loss benefits.
- · Great Plains Energy's long-term debt decreased \$111.7 million primarily to reflect GMO's \$137.3 million 7.95% Senior Notes and \$197.0 million 7.75% Senior Notes as current maturities. Current maturities of long-term debt increased similarly. Partially offsetting the decrease in long-term debt was Great Plains Energy's issuance of \$250.0 million of 2.75% Senior Notes in August 2010.

CAPITAL REQUIREMENTS AND LIQUIDITY

Great Plains Energy operates through its subsidiaries and has no material assets other than the stock of its subsidiaries. Great Plains Energy's ability to make payments on its debt securities and its ability to pay dividends is dependent on its receipt of dividends or other distributions from its subsidiaries, proceeds from the issuance of its securities and borrowings under its revolving credit facility.

Great Plains Energy's capital requirements are principally comprised of debt maturities and electric utility's construction and other capital expenditures. These items as well as additional cash and capital requirements are discussed below.

Great Plains Energy's liquid resources at September 30, 2010, consisted of \$7.6 million of cash and cash equivalents on hand and \$965.1 million of unused bank lines of credit. The unused lines consisted of \$162.2 million from Great Plains Energy's revolving credit facility, \$366.1 million from KCP&L's credit facilities and \$436.8 million from GMO's revolving credit facility. See Note 9 to the consolidated financial statements for more information on these credit facilities. Generally, Great Plains Energy uses these liquid resources to meet its day-to-day cash flow requirements, and from time to time issues equity and/or long-term debt to repay short-term debt or increase cash balances.

Great Plains Energy intends to meet day-to-day cash flow requirements including interest payments, retirement of maturing debt, construction requirements, dividends and pension benefit plan funding requirements, with a combination of internally generated funds and proceeds from the issuance of equity securities, equity-linked securities and/or short-term and long-term debt. Great Plains Energy's intention to meet a portion of these requirements with internally generated funds may be impacted by the effect of inflation on operating expenses, the level of retail MWh sales, regulatory actions, compliance with environmental regulations and the availability of generating units. In addition, Great Plains Energy may issue equity, equity-linked securities and/or debt to finance growth.

Cash Flows from Operating Activities

Great Plains Energy generated positive cash flows from operating activities for the periods presented. Great Plains Energy's cash flows from operating activities year to date September 30, 2010, increased primarily due to an increase in net income and an increase in deferred income taxes from utilizing net operating loss benefits and

bonus depreciation. On January 1, 2010, Great Plains Energy adopted new accounting guidance for transfers of financial assets, which resulted in the recognition of \$95.0 million of accounts receivables pledged as collateral and a corresponding short-term collateralized note payable on Great Plains Energy's balance sheet at September 30, 2010. See Note 3 for additional information. As a result, cash flows from operating activities were reduced by \$95.0 million and cash flow from financing activities were raised by \$95.0 million with no impact to the net change in cash year to date September 30, 2010. Additionally, cash flows from operating activities year to date September 30, 2009, reflect the payment of \$79.1 million for the settlement of FSS upon the issuance of \$400.0 million of 7.15% M ortgage Bonds Series 2009A in 2009. Other changes in working capital are detailed in Note 2 to the consolidated financial statements. The individual components of working capital vary with normal business cycles and operations.

Cash Flows from Investing Activities

Great Plains Energy's cash used for investing activities varies with the timing of utility capital expenditures and purchases of investments and nonutility property. Investing activities are offset by the proceeds from the sale of properties and insurance recoveries.

Great Plains Energy's utility capital expenditures decreased \$218.4 million year to date September 30, 2010, compared to the same period in 2009 due to a decrease in KCP&L's cash utility expenditures primarily related to the Iatan No. 1 environmental project, Iatan No. 2 and \$46.0 million of expenditures in 2009 related to a wind project.

Cash Flows from Financing Activities

Great Plains Energy's cash flows from financing activities year to date September 30, 2010, reflect the issuance, at a discount, of \$250.0 million of 2.75% Senior Notes that mature in 2013. Great Plains Energy used the proceeds to make a three-year intercompany loan to GMO with GMO using the proceeds to repay short-term borrowings. Also reflected is the \$95.0 million impact of the short-term collateralized note payable described above under cash flows from operating activities.

Great Plains Energy's cash flows from financing activities year to date September 30, 2009, reflect proceeds of \$161.0 million from the issuance of 11.5 million shares of common stock at \$14 per share, proceeds of \$287.5 million from the issuance of 5.8 million Equity Units in May 2009 and KCP&L's issuance, at a discount, of \$400.0 million of 7.15% Mortgage Bonds Series 2009A that mature in 2019. Additionally, in the first quarter of 2009, Great Plains Energy sold 3.8 million shares of common stock for \$49.5 million in net proceeds under a Sales Agency Financing Agreement with BNY Mellon Capital Markets, LLC. Great Plains Energy paid \$22.7 million year to date September 30, 2009, for fees related to all issuances of debt and common stock. The proceeds from these issuances were used primarily to repay short-term borrowings.

Debt Financing

The ratings of Great Plains Energy's, KCP&L's and GMO's securities by the credit rating agencies impact the cost of borrowings under their revolving credit agreements and in the capital markets. The Companies view maintenance of strong credit ratings as extremely important to their ability to obtain debt financing and to the cost of debt financing and to that end an active and ongoing dialogue is maintained with the agencies with respect to results of operations, financial position, and future prospects. While a decrease in these credit ratings would not cause any acceleration of Great Plains Energy's, KCP&L's or GMO's debt, it could increase interest charges under Great Plains Energy's 6.875% Senior Notes due 2017, GMO's 11.875% Senior Notes due 2012, GMO'; s 7.95% Senior Notes due 2011 and Great Plains Energy's, KCP&L's and GMO's revolving credit agreements. A decrease in credit ratings could also have, among other things, an adverse impact, which could be material, on Great Plains Energy's, KCP&L's and GMO's access to capital, the cost of funds, the ability to recover actual interest costs in state regulatory proceedings, the amounts of collateral required under supply agreements and Great Plains Energy's ability to provide credit support for its subsidiaries.

On March 12, 2010, Moody's Investors Service, Inc. lowered the senior unsecured rating for KCP&L to "Baa2" from "Baa1", and affirmed KCP&L's "A3" senior secured rating and "Prime-2" short-term commercial paper rating. Moody's affirmed Great Plains Energy's and GMO's senior unsecured rating at "Baa3", and changed the outlook for Great Plains Energy, KCP&L and GMO to "stable" from "negative".

On April 9, 2010, Standard & Poor's changed the outlook for Great Plains Energy, KCP&L and GMO to "stable" from "negative". Standard & Poor's also raised KCP&L's short-term commercial paper rating to "A-2" from "A-3" and affirmed all of Great Plains Energy's, KCP&L's and GMO's long-term ratings, including Great Plains Energy's "BBB" corporate credit rating.

Financing Authorization

Under stipulations with the MPSC and KCC, Great Plains Energy and KCP&L maintain common equity at not less than 30% and 35%, respectively, of total capitalization (including only the amount of short-term debt in excess of the amount of construction work in progress). KCP&L's long-term financing activities are subject to the authorization of the MPSC. In March 2010, the MPSC authorized KCP&L to issue up to \$450.0 million of long-term debt and to enter into interest rate hedging instruments in connection with such debt through December 31, 2011. KCP&L had not utilized any of this authorized amount as of September 30, 2010.

In December 2008, FERC authorized KCP&L to have outstanding at any time up to a total of \$1.1 billion in short-term debt instruments through December 2010. The authorization is subject to four restrictions: (i) proceeds of debt backed by utility assets must be used for utility purposes; (ii) if any utility assets that secure authorized debt are divested or spun off, the debt must follow the assets and also be divested or spun off; (iii) if any proceeds of the authorized debt are used for non-utility purposes, the debt must follow the non-utility assets (specifically, if the non-utility assets are divested or spun off, then a proportionate share of the debt must follow the divested or spun off non-utility assets); and (iv) if utility assets financed by the authorized short-term debt are divested or spun off to another entity, a proportionate share of the debt must also be divested or spun off. At September 30, 2010, there was \$890.5 million available under this authorization. In the third quarter of 2010, KCP&L filed an application with FERC requesting \$1.0 billion of short-term debt authorization through December 2012, which is pending FERC decision.

In March 2010, and modified in April 2010, FERC authorized GMO to have outstanding at any time up to a total of \$500.0 million of short-term debt authorization through April 2012, conditioned on interest rates not exceeding 4.3% over LIBOR, the prime rate or federal funds rate, as applicable, and subject to the same four restrictions as the KCP&L FERC short-term authorization discussed in the preceding paragraph. At September 30, 2010, there was \$500.0 million available under this authorization. On July 1, 2010, FERC authorized GMO to issue up to a total of \$850.0 million of long-term debt, including intercompany debt, through July 31, 2012. At September 30, 2010, there was \$601.2 million available under this authorization.

KCP&L and GMO are also authorized by FERC to participate in the Great Plains Energy money pool, an internal financing arrangement in which funds may be lent on a short-term basis to KCP&L and GMO. At September 30, 2010, GMO had no outstanding payables and KCP&L had outstanding payables under the money pool of \$2.5 million to Great Plains Energy and \$33.1 million to GMO.

Debt Agreements

See Note 9 to the consolidated financial statements for discussion of revolving credit facilities.

Pensions

The Company maintains defined benefit plans for substantially all active and inactive employees of KCP&L, GMO and WCNOC and incurs significant costs in providing the plans. Funding of the plans follows legal and regulatory requirements with funding equaling or exceeding the minimum requirements of the Employee Retirement Income Security Act of 1974, as amended (ERISA). Year to date September 30, 2010, the Company contributed \$38.5 million to the plans and expects to contribute an additional \$29.8 million in 2010 to satisfy the

ERISA minimum funding requirements and the MPSC and KCC rate orders, the majority of which is expected to be paid by KCP&L.

Additionally, the Company provides post-retirement health and life insurance benefits for certain retired employees and expects to make benefit contributions of \$18.4 million under the provisions of these plans in 2010, with the majority paid by KCP&L.

Management believes the Company has adequate access to capital resources through cash flows from operations or through existing lines of credit to support these funding requirements.

KANSAS CITY POWER & LIGHT COMPANY

MANAGEMENT'S NARRATIVE ANALYSIS OF RESULTS OF OPERATIONS

The following table summarizes KCP&L's consolidated comparative results of operations.

	Three Months Ended September 30				te		
	2010	iiibei	2009		September 30		
	2010			1.	2010		2009
			`	lion	S)		
Operating revenues	\$ 486.5	\$	395.5	\$	1,194.7	\$	997.8
Fuel	(83.8)		(70.5)		(213.2)		(183.3)
Purchased power	(25.0)		(19.8)		(63.2)		(58.6)
Transmission of electricity by others	(4.4)		(3.2)		(11.3)		(9.1)
Gross margin (a)	373.3		302.0		907.0		746.8
Other operating expenses	(143.3)		(136.8)		(426.0)		(391.7)
Depreciation and amortization	(66.4)		(59.3)		(192.2)		(166.1)
Operating income	163.6		105.9		288.8		189.0
Non-operating income and expenses	3.6		8.4		18.6		22.4
Interest charges	(20.1)		(22.3)		(63.8)		(62.7)
Income tax expense	(54.5)		(26.4)		(83.6)		(39.8)
Net income	\$ 92.6	\$	65.6	\$	160.0	\$	108.9

⁽a) Gross margin is a non-GAAP financial measure. See explanation of gross margin under Great Plains Energy's Results of Operations.

	Revenue	s and	Costs	%		M	Whs So	ld	%
Three Months Ended September 30	2010		2009	Chan	ge	2010		2009	Change
Retail revenues	(mi	llions	5)			(th	ousand	s)	
Residential	\$ 209.8	\$	152.8		37	1,872		1,486	26
Commercial	183.0		156.9		17	2,107		1,971	7
Industrial	34.0		31.0		10	508		508	-
Other retail revenues	2.9		2.7		7	22		20	-
Provision for rate refund (excess									
Missouri wholesale margin)	(2.2)		-	N	ΙA	NA		NA	NA
Kansas ECA (over) under recovery	2.6		(0.3)	N	M	NA		NA	NA
Total retail	430.1		343.1		25	4,509		3,985	13
Wholesale revenues	50.2		47.3		6	1,615		1,642	(2)
Other revenues	6.2		5.1		23	NA		NA	NA
Operating revenues	486.5		395.5	_	23	6,124		5,627	9
Fuel	(83.8)		(70.5)		19				
Purchased power	(25.0)		(19.8)		26				
Transmission of electricity by others	(4.4)		(3.2)		39				
Gross margin (a)	\$ 373.3	\$	302.0	_	24				

⁽a) Gross margin is a non-GAAP financial measure. See explanation of gross margin under Great Plains Energy's Results of Operations.

	Revenues	and	Costs		%		MWh	s Sold		%	1
Year to Date September 30	2010		2009	(Change	2	010	2	009	Chai	nge
Retail revenues	(mil	lions)				(thou	sands)			
Residential	\$ 464.4	\$	365.5		27	4	1,573	4	1,001		14
Commercial	472.2		409.8		15		5,875	5	5,687		3
Industrial	93.5		81.2		15	1	l,481	1	l,418		4
Other retail revenues	8.8		7.9		13		65		64		-
Provision for rate refund (excess											
Missouri wholesale margin)	(3.7)		-		NA		NA		NA		NA
Kansas ECA under recovery	5.5		1.2		NM		NA		NA		NA
Total retail	1,040.7		865.6		20	11	L,994	11	l,170		7
Wholesale revenues	138.0		117.8		17	4	1,288	3	3,816		12
Other revenues	16.0		14.4		11		NA		NA		NA
Operating revenues	1,194.7		997.8		20	16	5,282	14	1,986		9
Fuel	(213.2)		(183.3)		16						
Purchased power	(63.2)		(58.6)		8						
Transmission of electricity by others	(11.3)		(9.1)		25						
Gross margin ^(a)	\$ 907.0	\$	746.8		21						

⁽a) Gross margin is a non-GAAP financial measure. See explanation of gross margin under Great Plains Energy's Results of Operations.

KCP&L's gross margin increased \$71.3 million for the three months ended September 30, 2010, compared to the same period in 2009 primarily due to the increase in retail revenues driven by new retail rates effective August 1, 2009, and September 1, 2009, for Kansas and Missouri, respectively, and favorable summer weather with a 73% increase in cooling degree days.

KCP&L's gross margin increased \$160.2 million year to date September 30, 2010, compared to the same period in 2009 primarily due to the increase in retail revenues driven by new retail rates effective August 1, 2009, and September 1, 2009, for Kansas and Missouri, respectively, and favorable weather.

Retail MWhs sold year to date September 30, 2010, increased due to favorable weather, with a 15% increase in heating degree days and a 55% increase in cooling degree days and an increase in weather-normalized customer demand reflecting an improving economic climate. Wholesale MWhs sold increased due to a 11% increase in generation resulting in more MWhs available for sale, partially offset by the higher retail load requirements. The increase in generation was a result of Iatan No. 1 being off-line from January though mid-April 2009 to complete an environmental upgrade and unit overhaul, with the expenditures being capitalized, and therefore, not impacting operating and maintenance expenses. As a result, coal base load equivalent availability factor increased to 81% year to date September 30, 2010, compared to 75% for the same period in 2009.

KCP&L Other Operating Expenses (including operating and maintenance expenses, general taxes and other)

KCP&L's other operating expenses increased \$6.5 million for the three months ended September 30, 2010, compared to the same period in 2009. Operating and maintenance expenses decreased \$1.3 million for the three months ended September 30, 2010, compared to the same period in 2009 primarily driven by \$7.5 million expensed in September 2009 after KCP&L exercised its option to terminate an agreement for the construction of a wind project, mostly offset by plant outages and amortization of deferred costs related to customer programs. General taxes increased \$4.6 million for the three months ended September 30, 2010, compared to the same period in 2009 primarily driven by increased gross receipts taxes on increased retail revenues and increased property taxes. KCP&L supported a recommended dis allowance of certain Iatan No. 2 construction costs in its Kansas rate case post-hearing brief filed in September 2010. KCP&L recorded a \$3.0 million pre-tax loss in the third quarter of 2010 for KCP&L's share of these Iatan No. 2 construction costs.

KCP&L's other operating expenses increased \$34.3 million year to date September 30, 2010, compared to the same period in 2009. Operating and maintenance expenses increased \$21.9 million year to date September 30, 2010, compared to the same period in 2009 primarily driven by plant outages, amortization of deferred costs related to customer programs and an increase in spending for vegetation management, partially offset by \$7.5 million expensed in September 2009 after KCP&L exercised its option to terminate an agreement for the construction of a wind project. General taxes increased \$9.1 million year to date September 30, 2010, compared to the same period in 2009 primarily driven by increased gross receipts taxes on increased retail revenues and increased property taxes. KCP&L supported a r ecommended disallowance of certain Iatan No. 2 construction costs in its Kansas rate case post-hearing brief filed in September 2010. KCP&L recorded a \$3.0 million pre-tax loss in the third quarter of 2010 for KCP&L's share of these Iatan No. 2 construction costs.

KCP&L Depreciation and Amortization

KCP&L's depreciation and amortization costs increased \$7.1 million and \$26.1 million for the three months ended and year to date September 30, 2010, respectively, compared to the same periods in 2009. Additional regulatory amortization pursuant to KCP&L's 2009 rate cases increased depreciation and amortization \$3.2 million and \$17.2 million for the three months ended and year to date September 30, 2010, respectively. The remaining increases were due to placing in service the Iatan No. 1 environmental equipment in April 2009, which only impacted the comparative year to date period, and the KCP&L Kansas portion of Iatan No. 2 depreciation expense in the third quarter of 2010, as well as normal depreciation activity for other capital additions.

KCP&L Non-operating Income and Expenses

KCP&L's non-operating income and expenses decreased \$4.8 million and \$3.8 million for the three months ended and year to date September 30, 2010, respectively, compared to the same periods in 2009 primarily due to a decrease in the equity component of AFUDC resulting from a lower average construction work in progress balance due to KCP&L's Comprehensive Energy Plan projects being placed in service.

KCP&L Interest Charges

KCP&L's interest charges decreased \$2.2 million for the three months ended September 30, 2010, compared to the same period in 2009 primarily due to the deferral to a regulatory asset of construction accounting carrying costs for Iatan No. 1, Iatan No. 2 and common facilities, mostly offset by a decrease in the debt component of AFUDC resulting from a lower average construction work in progress balance due to KCP&L's Comprehensive Energy Plan projects.

KCP&L's interest charges increased \$1.1 million year to date September 30, 2010, compared to the same period in 2009 primarily due to interest on \$400.0 million of 7.15% Mortgage Bonds Series 2009A issued in March 2009 and a decrease in the debt component of AFUDC resulting from a lower average construction work in progress balance due to KCP&L's Comprehensive Energy Plan projects, mostly offset by the deferral to a regulatory asset of construction accounting carrying costs for Iatan No. 1, Iatan No. 2 and common facilities.

KCP&L Income Tax Expense

KCP&L's income tax expense increased \$28.1 million for the three months ended September 30, 2010, compared to the same period in 2009 primarily due to increased pre-tax income. KCP&L's income tax expense increased \$43.8 million year to date September 30, 2010, compared to the same period in 2009 primarily due to increased pre-tax income and a \$2.8 million increase in income tax expense for the cumulative change in tax treatment of the Medicare Part D subsidy under the Federal health care reform legislation signed into law in the first quarter of 2010.

ITEM 3. QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK

Great Plains Energy and KCP&L are exposed to market risks associated with commodity price and supply, interest rates and equity prices. Market risks are handled in accordance with established policies, which may include entering into various derivative transactions. In the normal course of business, Great Plains Energy and KCP&L also face risks that are either non-financial or non-quantifiable. Such risks principally include business, legal, regulatory, operational and credit risks and are discussed elsewhere in this document as well as in the 2009 Form 10-K and therefore are not represented here.

Great Plains Energy and KCP&L interim period disclosures about market risk included in quarterly reports on Form 10-Q address material changes, if any, from the most recently filed annual report on Form 10-K. Therefore, these interim period disclosures should be read in connection with Item 7A. Quantitative and Qualitative Disclosures About Market Risk, included in the 2009 Form 10-K of each of Great Plains Energy and KCP&L, incorporated herein by reference.

MPS Merchant is exposed to credit risk. Credit risk is measured by the loss that would be recorded if counterparties failed to perform pursuant to the terms of the contractual obligations less the value of any collateral held. MPS Merchant's counterparties are not externally rated. Credit exposure to counterparties at September 30, 2010, was \$40.0 million, net of \$2.0 million of collateral.

ITEM 4. CONTROLS AND PROCEDURES

GREAT PLAINS ENERGY

Disclosure Controls and Procedures

Great Plains Energy carried out an evaluation of its disclosure controls and procedures (as defined in Rules 13a-15(e) or 15d-15(e) under the Securities Exchange Act of 1934, as amended (the Exchange Act)). This evaluation was conducted under the supervision, and with the participation, of Great Plains Energy's management, including the chief executive officer and chief financial officer, and Great Plains Energy's disclosure committee. Based upon this evaluation, the chief executive officer and chief financial officer of Great Plains Energy have concluded as of the end of the period covered by this report that the disclosure controls and procedures of Great Plains Energy were effective at a reasonable assurance level.

Changes in Internal Control Over Financial Reporting

There has been no change in Great Plains Energy's internal control over financial reporting (as defined in Rule 13a-15(f) and 15d-15(f) of the Exchange Act) that occurred during the quarterly period ended September 30, 2010, that has materially affected, or is reasonably likely to materially affect, its internal control over financial reporting.

KCP&L

Disclosure Controls and Procedures

KCP&L carried out an evaluation of its disclosure controls and procedures (as defined in Rules 13a-15(e) or 15d-15(e) under the Securities Exchange Act of 1934, as amended). This evaluation was conducted under the supervision, and with the participation, of KCP&L's management, including the chief executive officer and chief financial officer, and KCP&L's disclosure committee. Based upon this evaluation, the chief executive officer and chief financial officer of KCP&L have concluded as of the end of the period covered by this report that the disclosure controls and procedures of KCP&L were effective at a reasonable assurance level.

Changes in Internal Control Over Financial Reporting

There has been no change in KCP&L's internal control over financial reporting (as defined in Rule 13a-15(f) and 15d-15(f) of the Exchange Act) that occurred during the quarterly period ended September 30, 2010, that has materially affected, or is reasonably likely to materially affect, its internal control over financial reporting.

PART II - OTHER INFORMATION

ITEM 1. LEGAL PROCEEDINGS

Other Proceedings

The Companies are parties to various lawsuits and regulatory proceedings in the ordinary course of their respective businesses. For information regarding material lawsuits and proceedings, see Notes 6, 11 and 12 to the consolidated financial statements. Such descriptions are incorporated herein by reference.

ITEM 1A. RISK FACTORS

Actual results in future periods for Great Plains Energy and KCP&L could differ materially from historical results and the forward looking statements contained in this report. The business of Great Plains Energy and KCP&L is influenced by many factors that are difficult to predict, involve uncertainties that may materially affect actual results and are often beyond their control. Factors that might cause or contribute to such differences include, but are not limited to, those discussed in Item 1A. Risk Factors included in the 2009 Form 10-K for each of Great Plains Energy and KCP&L. There have been no material changes with regard to those risk factors. Those risk factors, as well as the information included in this report and in the other documents filed with the SEC, should be carefully considered before making an investment in the securities of Great Plains Energy or KCP&L. Risk factors of KCP&L are also risk factors of Great Plains Energy.

ITEM 2. UNREGISTERED SALES OF EQUITY SECURITIES AND USE OF PROCEEDS

The following table provides information regarding purchases by Great Plains Energy of its equity securities during the three months ended September 30, 2010.

Issuer Purchases of Equity Securities					
Month	Total Number of Shares (or Units) Purchased	Average Price Paid per Share (or Unit)	Total Number of Shares (or Units) Purchased as Part of Publicly Announced Plans or Programs	Maximum Number (or Approximate Dollar Value) of Shares (or Units) that May Yet Be Purchased Under the Plans or Programs	
July 1 - 31	13,780 ⁽¹⁾	\$ 17.97	-	N/A	
August 1 - 31	1,826 ⁽¹⁾	18.05	-	N/A	
September 1 - 30	-	-	-	N/A	
Total	15,606	\$ 17.98	-	N/A	

⁽¹⁾ Represents common stock shares surrendered to the Company by certain officers to pay taxes related to the vesting of restricted common shares.

ITEM 3. DEFAULTS UPON SENIOR SECURITIES

None.

ITEM 4. (REMOVED AND RESERVED)

ITEM 5. OTHER INFORMATION

None.

ITEM 6. EXHIBITS

The documents listed below are being, or have been, filed, furnished or submitted on behalf of Great Plains Energy and KCP&L.

Exhibit <u>Number</u> 4.1	*	<u>Description of Document</u> Third Supplemental Indenture dated as of August 13, 2010 between Great Plains Energy Incorporated and The Bank of New York Mellon Trust Company, N.A., as trustee (Exhibit 4.1 to Form 8-K filed on August 13, 2010).	Registrant(s) Great Plains Energy
10.1		Credit Agreement dated as of August 9, 2010 among Great Plains Energy Incorporated, Certain Lenders, Bank of America, N.A., as Administrative Agent, and Union Bank, N.A. and Wells Fargo Bank, National Association, as Syndication Agents, Barclays Bank PLC and U.S. Bank National Association, as Documentation Agents, Banc of America Securities LLC, Union Bank, N.A. and Wells Fargo Securities, LLC as Joint Lead Arrangers and Joint Book Managers.	Great Plains Energy
10.2		Credit Agreement dated as of August 9, 2010 among Kansas City Power & Light Company, Certain Lenders, Bank of America, N.A., as Administrative Agent, and Union Bank, N.A. and Wells Fargo Bank, National Association, as Syndication Agents, JPMorgan Chase Bank, N.A. and The Bank of Nova Scotia, as Documentation Agents, Banc of America Securities LLC, Union Bank, N.A. and Wells Fargo Securities LLC as Joint Lead Arrangers and Joint Book Managers.	Great Plains Energy KCP&L
10.3		Credit Agreement dated as of August 9, 2010 among KCP&L Greater Missouri Operations Company, Certain Lenders, Bank of America, N.A., as Administrative Agent, and Union Bank, N.A. and Wells Fargo Bank, National Association, as Syndication Agents, The Royal Bank of Scotland PLC and BNP Paribas, as Documentation Agents, Banc of America Securities LLC, Union Bank, N.A. and Wells Fargo Securities, LLC as Joint Lead Arrangers and Joint Book Managers.	
10.4	+ *	Consulting Services Assignment and Assumption Agreement between John R. Marshall and Coastal Partners Inc. (Exhibit 10.2 to Form 10-Q for the quarter ended June 30, 2010).	Great Plains Energy KCP&L
12.1		Computation of Ratio of Earnings to Fixed Charges.	Great Plains Energy
12.2		Computation of Ratio of Earnings to Fixed Charges.	KCP&L
31.1		Rule 13a-14(a)/15d-14(a) Certifications of Michael J. Chesser.	Great Plains Energy
31.2		Rule 13a-14(a)/15d-14(a) Certifications of James C. Shay.	Great Plains Energy
31.3		Rule 13a-14(a)/15d-14(a) Certifications of Michael J. Chesser.	KCP&L
31.4		Rule 13a-14(a)/15d-14(a) Certifications of James C. Shay.	KCP&L
32.1	**	Section 1350 Certifications.	Great Plains Energy
32.2	**	Section 1350 Certifications.	KCP&L

101,INS ** XBRL Instance Document.	Great Plains Energy KCP&L
101.SCH ** XBRL Taxonomy Extension Schema Document.	Great Plains Energy KCP&L
101.CAL ** XBRL Taxonomy Extension Calculation Linkbase Document.	Great Plains Energy KCP&L
101.DEF ** XBRL Taxonomy Extension Definition Linkbase Document.	Great Plains Energy KCP&L
101.LAB ** XBRL Taxonomy Extension Labels Linkbase Document.	Great Plains Energy KCP&L
101.PRE ** XBRL Taxonomy Extension Presentation Linkbase Document.	Great Plains Energy

^{*}Filed with the SEC as an exhibit to prior SEC filings and is incorporated herein by reference and made a part hereof. The SEC filing and the exhibit number of the document so filed, and incorporated herein by reference, is stated in parenthesis in the description of such exhibit.

KCP&L

Copies of any of the exhibits filed with the SEC in connection with this document may be obtained from the registrants upon written request.

The registrants agree to furnish to the SEC upon request any instrument with respect to long-term debt as to which the total amount of securities authorized does not exceed 10% of total assets of such registrant and its subsidiaries on a consolidated basis.

^{**}Furnished and shall not be deemed filed for the purpose of Section 18 of the Securities Exchange Act of 1934, as amended (the "Exchange Act"). Such document shall not be incorporated by reference into any registration statement or other document pursuant to the Exchange Act or the Securities Act of 1933, as amended, unless otherwise indicated in such registration statement or other document.

⁺ Indicates management contract or compensatory plan or arrangement.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, Great Plains Energy Incorporated and Kansas City Power & Light Company have duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

GREAT PLAINS ENERGY INCORPORATED

Dated: October 28, 2010 By: /s/ Michael J. Chesser

(Michael J. Chesser) (Chief Executive Officer)

Dated: October 28, 2010 By: /s/ Lori A. Wright

(Lori A. Wright)

(Principal Accounting Officer)

KANSAS CITY POWER & LIGHT COMPANY

Dated: October 28, 2010 By: /s/ Michael J. Chesser

(Michael J. Chesser) (Chief Executive Officer)

Dated: October 28, 2010 By: <u>/s/ Lori A. Wright</u>

(Lori A. Wright)

(Principal Accounting Officer)

Dana

Published CUSIP Number: 39116FAF2

CREDIT AGREEMENT

Dated as of August 9, 2010

among

GREAT PLAINS ENERGY INCORPORATED,

CERTAIN LENDERS,

BANK OF AMERICA, N.A., as Administrative Agent

and

UNION BANK, N.A. and WELLS FARGO BANK, NATIONAL ASSOCIATION, as Syndication Agents

BARCLAYS BANK PLC and U.S. BANK NATIONAL ASSOCIATION, as Documentation Agents

BANC OF AMERICA SECURITIES LLC, UNION BANK, N.A. and WELLS FARGO SECURITIES, LLC, as Joint Lead Arrangers and Joint Book Managers

TABLE OF CONTENTS

		1 agc
ARTICI	LE I <u>DEFINITIONS</u>	
1.1	<u>Definitions</u> .	1
1.2	Accounting Principles.	19
1.3	Letter of Credit Amounts.	20
1.4	<u>Times of Day.</u>	20
ARTICI	LE II THE CREDITS	
2.1	Commitment.	20
2.2	Required Payments; Termination.	21
2.3	Ratable Loans.	21
2.4	Types of Advances; Minimum Amount.	21
2.5	Commitment Fee.	21
2.6	Changes in Aggregate Commitment.	21
2.7	Optional Prepayments.	27
2.8	Method of Selecting Types and Interest Periods for New Advances.	28
2.9	Conversion and Continuation of Outstanding Advances.	28
2.10	Interest Rates.	29
2.11	Rates Applicable After Default.	29
2.12	Method of Payment.	30
2.13	Noteless Agreement; Evidence of Indebtedness.	30
2.14	Telephonic Notices.	31
2.15	Interest Payment Dates; Interest and Fee Basis.	31

2.16	Notification of Advances, Interest Rates, Prepayments and Commitment Reductions.	31
2.17	Lending Installations.	32
2.18	Non-Receipt of Funds by the Administrative Agent.	32
2.19	<u>Letters of Credit</u> .	32
2.20	Swing Line Loans.	40
2.21	Extension of Facility Termination Date.	43
2.22	Cash Collateral.	44
2.23	Defaulting Lenders.	45
A RTICI	LE III YIELD PROTECTION; TAXES	
3.1	Yield Protection.	47
3.2	Changes in Capital Adequacy Regulations.	48
3.3	Availability of Types of Advances.	48
3.4	Funding Indemnification.	49
3.5	<u>Taxes</u> .	49
3.6	<u>Lender Statements; Survival of Indemnity</u> .	53
ARTICI	LE IV <u>CONDITIONS PRECEDENT</u>	
4.1	Initial Credit Extension.	53
4.2	Each Credit Extension.	54
4.2	Lacir Creuit Extension.	54
A DELCI	A DAY DEDDECTORATIONS AND WARD ANTICO	
	LE V_REPRESENTATIONS AND WARRANTIES	
5.1	Existence and Standing.	55
5.2	Authorization and Validity.	55
	i	
5.3	No Conflict Coveryment Concert	56
	No Conflict; Government Consent.	
5.4	<u>Financial Statements</u> .	56
5.5	Material Adverse Change.	56
5.6	Taxes.	57
5.7	Litigation; etc.	57
5.8	ERISA.	57
5.9	Accuracy of Information.	57
5.10	Regulation U.	58
5.11	Material Agreements.	58
5.12	Compliance With Laws.	58
5.13	Ownership of Properties.	58
5.14	<u>Plan Assets; Prohibited Transactions.</u>	58
5.15	Environmental Matters.	59
5.16	Investment Company Act.	59
5.17	Pari Passu Indebtedness.	59
5.18	Solvency.	59
5.19	No Defaults.	59
	LE VI <u>COVENANTS</u>	
6.1	Financial Reporting and Notices.	60
6.2	Permits, Etc.	62
6.3	Use of Proceeds.	62
6.4	Notice of Default.	62
6.5	Conduct of Business.	62
6.6	<u>Taxes</u> .	63
6.7	<u>Insurance</u> .	63
6.8	Compliance with Laws.	63
6.9	Maintenance of Properties; Books of Record.	63
6.10	<u>Inspection</u> .	64
6.11	Consolidations, Mergers and Sale of Assets.	64
6.12	Liens.	65
6.13	Affiliates.	68
6.14	ERISA.	69
6.15	Total Indebtedness to Total Capitalization.	69
6.16	Restrictions on Subsidiary Dividends.	69
6.17	Organization Documents.	69
ARTICI	LE VII <u>DEFAULTS</u>	
ARTICI	LE VIII <u>ACCELERATION, WAIVERS, AMENDMENTS AND REMEDIES</u>	
8.1	Acceleration; Letter of Credit Account.	72
8.2	Amendments.	72
8.3	Preservation of Rights.	74
	LE IX GENERAL PROVISIONS	
9.1	Survival of Representations.	74
9.2	Governmental Regulation.	74

9.5	Several Obligations; Benefits of this Agreement.	75
	ii	
9.6	Expenses; Indemnification.	75
9.7	Numbers of Documents.	76
9.8	Accounting.	76
9.9	Severability of Provisions.	76
9.10	Nonliability of Lenders.	77
9.11	<u>Limited Disclosure</u> .	77
9.12	<u>USA PATRIOT ACT NOTIFICATION</u> .	78
9.13	Nonreliance.	78
9.14	No Advisory or Fiduciary Responsibility.	78
ARTIC	LE X THE ADMINISTRATIVE AGENT	
10.1	Appointment and Authority.	79
10.2	Rights as a Lender.	79
10.3	Exculpatory Provisions.	80
10.4	Reliance by Administrative Agent.	81
10.5	Delegation of Duties.	81
10.6	Resignation of Administrative Agent.	81
10.7	Non-Reliance on Administrative Agent and Other Lenders.	82
10.8	No Other Duties, Etc.	82
10.9	Administrative Agent May File Proofs of Claim.	82
A DTIC	LE XI_SETOFF; RATABLE PAYMENTS; PAYMENTS SET ASIDE	
11.1	Setoff, RAIABLE PATMENTS, PATMENTS SET ASIDE	83
11.1	Ratable Payments.	84
11.3	Payments Set Aside.	84
11.5	rayments set Aside.	04
ARTIC	LE XII BENEFIT OF AGREEMENT; ASSIGNMENTS; PARTICIPATIONS	
12.1	Successors and Assigns.	85
12.2	Replacement of Lenders.	89
ARTIC	LE XIII <u>NOTICES</u>	
13.1	Notices.	90
13.2	Change of Address.	90
13.3	The Platform.	90
Δ PTIC	LE XIV COUNTERPARTS	
ANTIC	LL AIV COUNTEMAKIS	
ARTIC	LE XV OTHER AGENTS	
ARTIC	LE XVI CHOICE OF LAW; CONSENT TO JURISDICTION; WAIVER OF JURY TRIAL.	
16.1	CHOICE OF LAW.	91
16.2	CONSENT TO JURISDICTION.	92
16.3	WAIVER OF JURY TRIAL.	92

74

74

ARTICLE XVII TERMINATION OF EXISTING CREDIT FACILITY

iii

SCHEDULES

9.3

9.4

Headings.

Entire Agreement.

- Commitments and Letter of Credit Commitments Existing Letters of Credit
- II
- III Pricing Schedule
- IV Certain Addresses for Notices

- A Form of Compliance Certificate
 B Form of Assignment and Assumption
 C Form of Wire Transfer Instructions
 - D Form of Revolving NoteE Form of Swing Line Note

iv

CREDIT AGREEMENT

This Credit Agreement dated as of August 9, 2010 is among Great Plains Energy Incorporated, a Missouri corporation, the Lenders, Union Bank, N.A. and Wells Fargo Bank, National Association, as Syndication Agents, and Bank of America, N.A., as Administrative Agent. The parties hereto agree as follows:

ARTICLE I

DEFINITIONS

1.1 <u>Definitions</u>.

As used in this Agreement, the following terms have the following meanings (such meanings to be equally applicable to both the singular and plural forms of such terms):

- "Additional Commitment Lender" is defined in Section 2.21(d).
- "<u>Administrative Agent</u>" means Bank of America in its capacity as administrative agent under any of the Loan Documents, or any successor administrative agent.
 - "Administrative Agent's Fee Letter" means certain fee letter dated June 2, 2010 among Bank of America, the Borrower, KCPL and KCPL GMO.
- "<u>Administrative Agent's Office</u>" means the Administrative Agent's address and, as appropriate, account as set forth on <u>Schedule IV</u> or such other address or account as the Administrative Agent may from time to time notify to the Borrower and the Lenders.
 - "Administrative Questionnaire" means an Administrative Questionnaire in a form supplied by the Administrative Agent.
- "Advance" means a borrowing hereunder (or conversion or continuation thereof) consisting of the aggregate amount of the several Loans, including the Swing Line Loans, made on the same Borrowing Date (or date of conversion or continuation) by the Lenders to the Borrower of the same Type and, in the

case of Eurodollar Advances, for the same Interest Period.

"Affiliate" of any Person means any other Person directly or indirectly controlling, controlled by or under common control with such Person. For purposes of this definition, "control" (including, with correlative meanings, the terms "controlled by" and "under common control with") shall mean the possession, directly or indirectly, of the power to direct or cause the direction of the management or policies of a Person, whether through the ownership of voting securities or by contract or otherwise.

"Agent Parties" is defined in Section 13.3.

- "Agents" means, collectively, the Administrative Agent and the Syndication Agents, and "Agent" means any one of them.
- "Aggregate Commitment" means the aggregate of the Commitments of all Lenders, as changed from time to time pursuant to the terms hereof. The amount of the Aggregate Commitment in effect as of the Closing Date is TWO HUNDRED MILLION DOLLARS (\$200,000,000).
 - "Aggregate Outstanding Credit Exposure" means, at any time, the aggregate of the Outstanding Credit Exposure of all Lenders.
 - "Agreement" means this credit agreement, as it may be amended or modified and in effect from time to time.
- "Alternate Base Rate" means for any day a fluctuating rate per annum equal to the highest of (a) the Federal Funds Rate plus 1/2 of 1%, (b) the rate of interest in effect for such day as publicly announced from time to time by Bank of America as its "prime rate" and (c) Eurodollar Base Rate plus 1.0%. The "prime rate" is a rate set by Bank of America based upon various factors including Bank of America's costs and desired return, general economic conditions and other factors, and is used as a reference point for pricing some loans, which may be priced at, above, or below such announced rate. Any change in such prime rate announced by Bank of America shall take effect at the opening of business on the& #160;day specified in the public announcement of such change.
- "Applicable Margin" means, with respect to Advances of any Type at any time, the percentage rate per annum which is applicable at such time with respect to Advances of such Type as set forth in the Pricing Schedule.
- "Approved Fund" means any Fund that is administered or managed by (a) a Lender, (b) an Affiliate of a Lender or (c) an entity or an Affiliate of an entity that administers or manages a Lender.
 - "Approving Lenders" is defined in Section 2.21(e).
 - "Arrangers" means BAS, Union Bank and Wells Fargo Securities, LLC and "Arranger" means any one of them.
 - "Article" means an article of this Agreement unless another document is specifically referenced.
- "Assignment Agreement" means an assignment and assumption entered into by a Lender and an Eligible Assignee (with the consent of any party whose consent is required by Section 12.1(b)), and accepted by the Administrative Agent, in substantially the form of Exhibit B or any other form approved by the Administrative Agent.

2

- "Attributable Indebtedness" means, on any date, (a) in respect of any Capitalized Lease Obligation of any Person, the capitalized amount thereof that would appear on a balance sheet of such Person prepared as of such date in accordance with GAAP, and (b) in respect of any Synthetic Lease Obligation, the capitalized amount of the remaining lease payments under the relevant lease that would appear on a balance sheet of such Person prepared as of such date in accordance with GAAP if such lease were accounted for as a Capitalized Lease.
- "<u>Authorized Officer</u>" means any of the Chief Executive Officer, the President, any Vice President, the Chief Financial Officer or the Treasurer of the Borrower, in each case acting singly.
 - "Bank of America" means Bank of America, N.A. in its individual capacity and its successors.
 - "BAS" means Banc of America Securities LLC.
 - "Borrower" means Great Plains Energy Incorporated, a Missouri corporation, and its permitted successors and assigns.
 - "Borrower Materials" is defined in Section 6.1.
 - "Borrowing Date" means a date on which an Advance is made hereunder.
 - "Borrowing Notice" is defined in Section 2.8.
 - "BTMU" means The Bank of Tokyo-Mitsubishi UFJ, Ltd., in its individual capacity and its successors.
- "Business Day" means any day other than a Saturday, Sunday or other day on which commercial banks are authorized to close under the Laws of, or are in fact closed in, New York City, New York and, if such day relates to any Eurodollar Loan or any Floating Rate Loan bearing interest at a rate based on the Eurodollar Rate, means any such day that is also a London Banking Day.

"Capitalized Lease" of a Person means any lease of Property by such Person as lessee which would be capitalized on a balance sheet of such Person prepared in accordance with GAAP.

"Capitalized Lease Obligations" of a Person means the amount of the obligations of such Person under Capitalized Leases which would be shown as a liability on a balance sheet of such Person prepared in accordance with GAAP.

"Cash Collateralize" means to pledge and deposit with or deliver to the Administrative Agent, for the benefit of the Administrative Agent, the Issuers or the Swing Line Lender (as applicable) and the Lenders, as collateral for Letter of Credit Obligations, Obligations in respect of Swing Line Loans, or obligations of Lenders to fund participations in respect of either thereof

3

(as the context may require), cash or deposit account balances or, if the applicable Issuer or the Swing Line Lender benefitting from such collateral shall agree in its sole discretion, other credit support, in each case pursuant to documentation in form and substance reasonably satisfactory to (a) the Administrative Agent and (b) the applicable Issuer or the Swing Line Lender (as applicable). "Cash Collateral" shall have a meaning correlative to the foregoing and shall include the proceeds of such cash collateral and other credit support.

"Change" is defined in Section 3.2.

"Change in Law" means the occurrence, after the date of this Agreement, of any of the following: (a) the adoption or taking effect of any law, rule, regulation or treaty, (b) any change in any law, rule, regulation or treaty or in the administration, interpretation or application thereof by any Governmental Authority or (c) the making or issuance of any request, guideline or directive (whether or not having the force of law) by any Governmental Authority.

"Change of Control" means an event or series of events by which:

- (i) any "person" or "group" (as such terms are used in Sections 13(d) and 14(d) of the Securities Exchange Act of 1934, but excluding any employee benefit plan of the Borrower or its Subsidiaries, or any Person acting in its capacity as trustee, agent or other fiduciary or administrator of any such plan) becomes the "beneficial owner" (as defined in Rules 13d-3 and 13d-5 under the Securities Exchange Act of 1934), directly or indirectly, of thirty-three and one-third percent (33 1/3%) or more of the "voting equity interests" (meaning for this purpose the power under ordinary circumstances to vote for the election of members of the board of directors) of the Borrower; or
- during any period of twelve (12) consecutive months (or such lesser period of time as shall have elapsed since the formation of the Borrower), a majority of the members of the board of directors or other equivalent governing body of the Borrower ceases to be composed of individuals (x) who were members of that board or equivalent governing body on the first day of such period, (y) whose election or nomination to that board or equivalent governing body was approved by individuals referred to in $\underline{\text{clause}}(x)$ above constituting at the time of such election or nomination at least a majority of that board or equivalent governing body or (z) whose election or nomination to that board or other equivalent governing body was approved by individuals referred to in $\underline{\text{clauses}}(x)$ and $\underline{\text{(y)}}$ above constituting at the time of such election or nomination at least a majority of that board or equivalent governing body.

"Closing Date" means August 9, 2010.

"Code" means the Internal Revenue Code of 1986, as amended, reformed or otherwise modified from time to time.

"Collateral Shortfall Amount" is defined in Section 8.1.

4

"Commitment" means, for each Lender, the obligation of such Lender to make Loans and to participate in Letters of Credit and Swing Line Loans in an aggregate amount not exceeding the amount set forth on Schedule I hereto or as set forth in any Assignment Agreement relating to any assignment that has become effective pursuant to Section 12.1(b), as such amount may be modified from time to time pursuant to the terms hereof.

"Commitment Fee Rate" means, at any time, the percentage rate per annum at which commitment fees are accruing at such time as set forth in the Pricing Schedule.

"Consolidated Net Income" means, for any period, for the Borrower and its Consolidated Subsidiaries, the net income of the Borrower and its Consolidated Subsidiaries from continuing operations, excluding extraordinary items for that period.

"Consolidated Subsidiaries" means all Subsidiaries of the Borrower that are (or should be) included when preparing the consolidated financial statements of the Borrower.

"Consolidated Tangible Net Worth" means, as of any date of determination, for the Borrower and its Consolidated Subsidiaries, Shareholders' Equity of the Borrower and its Consolidated Subsidiaries on that date minus the Intangible Assets of the Borrower and its Consolidated Subsidiaries on that date.

"Contingent Obligation" of a Person means any agreement, undertaking or arrangement by which such Person assumes, guarantees, endorses, contingently agrees to purchase or provide funds for the payment of, or otherwise becomes or is contingently liable upon, the obligation or liability of any

other Person, or agrees to maintain the net worth or working capital or other financial condition of any other Person, or otherwise assures any creditor of such other Person against loss.

"Controlled Group" means all members of a controlled group of corporations or other business entities and all trades or businesses (whether or not incorporated) under common control which, together with the Borrower, are treated as a single employer under Section 414 of the Code.

"Conversion/Continuation Notice" is defined in Section 2.9.

"Credit Extension" means the making of an Advance or the issuance of a Letter of Credit.

"<u>Debtor Relief Law</u>" means the Bankruptcy Code of the United States, and all other liquidation, conservatorship, bankruptcy, assignment for the benefit of creditors, moratorium, rearrangement, receivership, insolvency, reorganization, or similar debtor relief Laws of the United States or other applicable jurisdictions from time to time in effect and affecting the rights of creditors generally.

"Default" means an event described in Article VII.

5

"Defaulting Lender" means, subject to Section 2.23(b), any Lender, as reasonably determined by the Administrative Agent, that, at any time during the term of this Agreement, (a) has failed to perform any of its funding obligations hereunder, including in respect of its Loans or participations in respect of Letters of Credit or Swing Line Loans, within three (3) Business Days of the date required to be funded by it hereunder, unless such obligation is the subject of a good faith dispute, (b) has notified the Borrower, the Administrative Agent or any Lender in writing, or has otherwise indicated through a public statement, that it does not intend to comply with any of its funding obligations under this Agreement or generally under other agreements in which it commits to extend credit, (c) has failed, within five (5) Business Days after written request by the Administrative Agent (based on the reasonable belief that such Lender may not fulfill any of its funding obligations hereunder or after any reasonable request by the Borrower to do so), to confirm in a manner reasonably satisfactory to the Administrative Agent and the Borrower that it will comply with its funding obligations, unless such obligation is the subject of a good faith dispute, provided that any such Lender shall cease to be a Defaulting Lender under this clause (c) upon receipt of such confirmation by the Administrative Agent in a manner reasonably satisfactory to the Administrative Agent, (d) otherwise failed to pay over to the Administrative Agent or any other Lender any other amount (other than a de minimis amount) required to be paid by it hereunder within three (3) Business Days of the date when due, unless the subject of a good-faith dispute, or (e) has, or has a direct or indirect parent company that has, (i) become the subject of a proceeding under any Debtor Relief Law, (ii) had a receiver, conservator, trustee, administrator, assignee for the benefit of creditors or similar Person charged with reorganization or liquidation of its business or a custodian appointed for it or for any substantial part of its assets, or (iii) taken any action in furtherance of, or indicated its consent to, approval of or acquiescence in any such proceeding or appointment; provided that a Lender shall not be a Defaulting Lender solely by virtue of the ownership or acquisition of any equity interest in that Lender or any direct or indirect parent company thereof, or the exercise of control over such Lender or any direct or indirect parent company thereof controlling such Lender, by a Governmental Authority.

"Dollar" and "\$" mean lawful money of the United States.

"<u>Eligible Assignee</u>" means (a) a Lender; (b) an Affiliate of a Lender; (c) an Approved Fund; and (d) any other Person (other than a natural person) approved by (i) the Administrative Agent, the Swing Line Lender and the Issuers, and (ii) unless a Default has occurred and is continuing, the Borrower (each such approval not to be unreasonably withheld or delayed); <u>provided</u> that notwithstanding the foregoing, "Eligible Assignee" shall not include the Borrower or any of the Borrower's Affiliates or Subsidiaries.

"Environmental Laws" means any and all federal, state, local and foreign statutes, laws, judicial decisions, regulations, ordinances, rules, judgments, orders, decrees, plans, injunctions, permits, concessions, grants, franchises, licenses, agreements and other governmental restrictions relating to (a) the protection of the environment, (b) the effect of the environment on human health, (c) emissions, discharges or releases of pollutants, contaminants, hazardous substances or wastes into surface water, ground water or land, or (d) the manufacture, processing, distribution, use, treatment, storage, disposal, transport or handling of pollutants, contaminants, hazardous substances or wastes or the clean-up or other remediation thereof.

6

"Equity-Linked Securities" means (a) all securities issued by the Borrower (including without limitation the May 2009 issuance of 5.75 million equity units by the Borrower) or any Subsidiary that contain two distinct components: (i) medium-term debt and (ii) a forward contract for the issuance of common stock of the Borrower or such Subsidiary prior to the maturity of, and in an amount not less than, such debt, including the securities commonly referred to by the tradenames "FELINE PRIDES", "PEPS", "HITS" and "DECS" and generally referred to as "equity units"; provided that such securities shall not contain any provision permitting them to be put to the Borrower or any Subsidiary prior to the settlement of the related purchase contract and (b) all other securities issued by the Borrower or any Subsidiary that are similar to those described in clause (a).

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended from time to time, and any rule or regulation issued thereunder.

"Eurodollar Advance" means an Advance which bears interest at the applicable Eurodollar Rate.

"Eurodollar Rate" means:

(a) for any Interest Period with respect to a Eurodollar Loan, a rate per annum (rounded to the nearest multiple of 1/16 of 1%) determined by the Administrative Agent pursuant to the following formula:

Eurodollar Rate = <u>Eurodollar Base Rate</u>

1.00 – Eurodollar Reserve Percentage

and (b) for any day with respect to a Floating Rate Loan bearing interest at a rate based on the Eurodollar Base Rate, a rate per annum (rounded to the nearest multiple of 1/16 of 1%) determined by the Administrative Agent pursuant to the following formula:

 $Eurodollar\ Rate \ = \underbrace{\frac{Eurodollar\ Base\ Rate}{1.00-Eurodollar\ Reserve\ Percentage}}.$

"Eurodollar Base Rate" means:

(a) for such Interest Period with respect to any Eurodollar Loan, the rate per annum equal to (i) the British Bankers Association LIBOR Rate ("BBA LIBOR"), as published by Reuters (or such other commercially available source providing quotations of BBA LIBOR as may be designated by the Administrative Agent from time to time) at approximately 11:00 a.m., London time, two (2) London Banking Days prior to the commencement of such Interest Period, for Dollar deposits (for delivery on the first day of such Interest Period) with a term equivalent to such Interest Period or (ii) if such published rate is not available at such time for any reason, the rate determined by the Administrative Agent to be the rate at which deposits in Dollars for delivery on the first day of such Interest Period in same day funds in the approximate amount of the Eurodollar Loan being made, continued or converted and with a term equivalent to such

7

Interest Period would be offered by Bank of America's London Branch to major banks in the London interbank eurodollar market at their request at approximately 11:00 a.m. (London time) two (2) London Banking Days prior to the commencement of such Interest Period and

(b) for any interest rate calculation with respect to any Floating Rate Loan, the rate per annum equal to (i) BBA LIBOR, at approximately 11:00 a.m., London time determined two (2) London Banking Days prior to such date for Dollar deposits being delivered in the London interbank market for a term of one (1) month commencing that day or (ii) if such published rate is not available at such time for any reason, the rate per annum determined by the Administrative Agent to be the rate at which deposits in Dollars for delivery on the date of determination in same day funds in the approximate amount of the Floating Rate Loan being made or maintained and with a term equal to one (1) month would be offered by Bank of America's London Branch to major banks in the London interbank Eurodollar market at their request at the date and time of det ermination.

"Eurodollar Reserve Percentage" means the reserve percentage (expressed as a decimal, carried out to five decimal places) in effect on such day, whether or not applicable to any Lender, under regulations issued from time to time by the FRB for determining the maximum reserve requirement (including any emergency, supplemental or other marginal reserve requirement) with respect to Eurocurrency funding (currently referred to as "Eurocurrency liabilities"). The Eurodollar Rate for each outstanding Eurodollar Loan or Floating Rate Loan bearing interest at a rate based on the Eurodollar Rate shall be adjusted automatically as of the effective date of any change in the Eurodollar Reserve Percentage.

"Eurodollar Loan" means a Loan which bears interest at the applicable Eurodollar Rate.

"Excluded Taxes" means, with respect to the Administrative Agent, any Lender, or any other recipient of any payment to be made by or on account of any Obligation of the Borrower hereunder or under any other Loan Document, (a) taxes imposed on or measured by its overall net income (however denominated), and franchise taxes imposed on it (in lieu of net income taxes), by the jurisdiction (or any political subdivision thereof) under the laws of which such recipient is organized or in which its principal office is located or, in the case of any Lender, in which its applicable Lending Installation is located, (b) any branch profits taxes imposed by the United States or any similar tax imposed by any other jurisdiction, (c) any backup withholding tax that is required by the Code to be withheld from amounts payable to a Lender, (d) in the case of a Foreign Lender (other than an assignee pursuant to a request by the Borrower under Section 12.2), any United States withholding tax that (i) is imposed on amounts payable to such Foreign Lender pursuant to the laws in force at the time such Foreign Lender becomes a party hereto (or designates a new Lending Installation) or (ii) is attributable to such Foreign Lender's failure or inability (other than solely as a result of a Change in Law occurring after such Foreign Lender becomes a party to this Agreement or designates a new Lending Installation) to comply with Section 3.5(e), except to the extent that such Foreign Lender (or its assignor, if any) was entitled, immediately prior to the time of designation of a new Lending Installation (or assignment), to receive additional amounts from the Bor rower with respect to such withholding tax pursuant to Section 3.5(a)(ii), and (e) any taxes imposed with respect to the requirements of FATCA.

8

"Exhibit" refers to an exhibit to this Agreement, unless another document is specifically referenced.

"Existing Credit Facility" means the credit agreement among the Borrower, Bank of America, N.A., as administrative agent and the other lenders party thereto, dated as of May 11, 2006, as amended or modified from time to time.

"Existing Letters of Credit" means those letters of credit identified on Schedule II.

"<u>Facility Termination Date</u>" means (a) the later of (i) August 9, 2013 and (ii) with respect to some or all of the Lenders if the facility termination date is extended pursuant to <u>Section 2.21</u>, such extended facility termination date or (b) any earlier date on which the Aggregate Commitment is reduced to zero or otherwise terminated pursuant to the terms hereof.

"FATCA" means sections 1471 through 1474 of the Code and any regulations that are issued thereunder or official governmental interpretations thereof.

"Federal Funds Rate" means, for any day, the rate per annum equal to the weighted average of the rates on overnight Federal funds transactions with members of the Federal Reserve System arranged by Federal funds brokers on such day, as published by the Federal Reserve Bank of New York on the Business Day next succeeding such day; provided that (a) if such day is not a Business Day, the Federal Funds Rate for such day shall be such rate on such transactions on the next preceding Business Day as so published on the next succeeding Business Day, and (b) if no such rate is so published on such next succeeding Business Day, the Federal Funds Rate for such day shall be the average rate (rounded upward, i f necessary, to a whole multiple of 1/100 of 1%) charged to Bank of America on such day on such transactions as determined by the Administrative Agent.

"Fee Letter" means that certain fee letter dated June 2, 2010 among the Agents, the Arrangers, the Borrower, KCPL and KCPL GMO.

"Floating Rate Advance" means an Advance which bears interest at the Alternate Base Rate.

"Floating Rate Loan" means a Loan which bears interest at the Alternate Base Rate.

"Foreign Lender" means any Lender that is not a "United States person" within the meaning of Section 7701(a)(30) of the Code.

"FRB" means the Board of Governors of the Federal Reserve System.

"Fronting Exposure" means, at any time there is a Defaulting Lender, (a) with respect to an Issuer, such Defaulting Lender's Pro Rata Share of the outstanding Letter of Credit Obligations other than Letter of Credit Obligations as to which such Defaulting Lender's participation obligation has been reallocated to other Lenders or Cash Collateralized in accordance with the terms hereof, and (b) with respect to the Swing Line Lender, such

9

Defaulting Lender's Pro Rata Share of Swing Line Loans other than Swing Line Loans as to which such Defaulting Lender's participation obligation has been reallocated to other Lenders or Cash Collateralized in accordance with the terms hereof.

"Fund" means any Person (other than a natural person) that is (or will be) engaged in making, purchasing, holding or otherwise investing in commercial loans and similar extensions of credit in the ordinary course of its activities.

"GAAP" means generally accepted accounting principles set forth from time to time in the opinions and pronouncements of the Accounting Principles Board and the American Institute of Certified Public Accountants and statements of the Financial Accounting Standards Board.

"GMO-GPE Re-Transfer" is defined in Section 2.6(f)(ii).

"GMO-GPE Transfer" is defined in Section 2.6(f)(i).

"Governmental Authority" means the government of the United States or any other nation, or of any political subdivision thereof, whether state or local, and any agency, authority, instrumentality, regulatory body, court, central bank or other entity exercising executive, legislative, judicial, taxing, regulatory or administrative powers or functions of or pertaining to government.

"GPE-KCPL Re-Transfer" is defined in Section 2.6(c)(ii).

"GPE-KCPL Transfer" is defined in Section 2.6(c)(i).

"GPE-GMO Re-Transfer" is defined in Section 2.6(d)(ii).

"GPE-GMO Transfer" is defined in Section 2.6(d)(i).

"Honor Date" is defined in Section 2.19(f).

"including" means "including without limiting the generality of the following".

"Indebtedness" means, as to any Person at a particular time, all of the following, without duplication, to the extent recourse may be had to the assets or properties of such Person in respect thereof: (a) all obligations of such Person for borrowed money and all obligations of such Person evidenced by bonds, debentures, notes, loan agreements or other similar instruments; (b) any direct or contingent obligations of such Person in the aggregate in excess of \$2,000,000 arising under letters of credit (including standby and commercial), banker's acceptances, bank guaranties, surety bonds and similar instruments; (c) net obligations of such Person under Swap Contracts; (d) all obligations of such Person to pay the deferred purchase price of property or service s (except trade accounts payable arising, and accrued expenses incurred, in the ordinary course of business), and indebtedness (excluding prepaid interest thereon) secured by a Lien on property owned or being purchased by such Person (including indebtedness arising under conditional sales or other title retention agreements), whether or not such indebtedness shall have

10

been assumed by such Person or is limited in recourse; (e) Capitalized Lease Obligations and Synthetic Lease Obligations of such Person; and (f) all Contingent Obligations of such Person in respect of any of the foregoing.

For all purposes hereof, the Indebtedness of any Person shall include the Indebtedness of any partnership or joint venture in which such Person is a general partner or a joint venturer, unless such Indebtedness is non-recourse to such Person. It is understood and agreed that Indebtedness (including

Contingent Obligations) shall not include any obligations of any Person with respect to (i) subordinated, deferrable interest debt securities, and any related securities issued by a trust or other special purpose entity in connection therewith, or any similar securities that are classified at the time of issuance, as possessing a minimum of "intermediate equity content" by S&P and "Basket C equity credit" by Moody's (or the equivalent classification then in effect by such agencies), as long as the m aturity date of such debt is subsequent to the Facility Termination Date; provided that the amount of mandatory principal amortization or defeasance of such debt prior to the Facility Termination Date shall be included in this definition of Indebtedness; (ii) Equity-Linked Securities until the mandatory redemption date therefor, provided that the principal amount of all outstanding Equity-Linked Securities in excess of twenty percent (20%) of Total Capitalization shall constitute Indebtedness; or (iii) utility "rate reduction" bonds, for the payment of which legislatively authorized charges are imposed on customers. The amount of any Capitalized Lease Obligation or Synthetic Lease Obligation as of any date shall be deemed to be the amount of Attributable Indebtedness in respect thereof as of such date.

"Indemnified Taxes" means Taxes other than Excluded Taxes.

"Intangible Assets" means, assets that are considered to be intangible assets under GAAP, including, but not limited to, customer lists, goodwill, computer software, copyrights, trade names, trademarks, patents, franchises and licenses.

"Interest Period" means, with respect to a Eurodollar Advance, a period of one (1), two (2), three (3) or six (6) months commencing on a Business Day selected by the Borrower pursuant to this Agreement. Such Interest Period shall end on the day which corresponds numerically to such date one (1), two (2), three (3) or six (6) months thereafter; provided that if there is no such numerically corresponding day in such next, second, third or sixth succeeding month, such Interest Period shall end on the last Business Day of such next, second, third or sixth succeeding month. If an Interest Period would otherwise end on a day which is not a Business Day, such Interest Period shall end on the next succeeding Business Day; provided that if said next succeeding Business Day falls in a new calendar month, such Interest Period shall end on the immediately preceding Business Day.

"ISP" means, with respect to any Letter of Credit, the "International Standby Practices 1998" published by the Institute of International Banking Law & Practice, Inc. (or such later version thereof as may be in effect at the time of issuance of such Letter of Credit).

"Issuer" means each of Bank of America, Wells Fargo, BTMU and any other Lender selected by the Borrower with the approval of the Administrative Agent (except with respect to

11

any selection for which such approval is not required pursuant to Section 2.19(n)), in each case, in its capacity as an issuer of Letters of Credit hereunder and includes, without limitation, any Lender appointed by the Borrower (with the consent of the Administrative Agent, except with respect to any selection for which such approval is not required pursuant to Section 2.19(n)) as such by notice to the Lenders as a replacement for any Issuer who at the time of such appointment is a Defaulting Lender. For the avoidance of doubt, any Lender (other than Bank of America, Wells Fargo and BTMU) approached to become an issuer of Letters of Credit may elect or decline, in its sole discretion, to become an issuer.

"<u>Issuer Documents</u>" means with respect to any Letter of Credit, any Letter of Credit Application and any other document, agreement and instrument entered into by the applicable Issuer and the Borrower or in favor of the applicable Issuer and relating to such Letter of Credit.

"KCPL" means Kansas City Power & Light Company, a Missouri corporation.

"KCPL Credit Agreement" means that certain Credit Agreement dated as of the Closing Date among KCPL, the financial institutions party thereto, Union Bank and Wells Fargo, as syndication agents and Bank of America, N.A., as administrative agent, as amended or modified from time to time.

"KCPL GMO" means KCP&L Greater Missouri Operations Company, a Delaware corporation.

"KCPL GMO Credit Agreement" means that certain Credit Agreement dated as of the Closing Date among KCPL GMO, the financial institutions party thereto, Union Bank and Wells Fargo, as syndication agents and Bank of America, N.A., as administrative agent, as amended or modified from time to time.

"KCPL-GPE Re-Transfer" is defined in Section 2.6(e)(ii).

"KCPL-GPE Transfer" is defined in Section 2.6(e)(i).

"<u>Laws</u>" means, collectively, all international, foreign, federal, state and local statutes, treaties, rules, guidelines, regulations, ordinances, codes and administrative or judicial precedents or authorities, including the interpretation or administration thereof by any Governmental Authority charged with the enforcement, interpretation or administration thereof, and all applicable administrative orders, directed duties, requests, licenses, authorizations and permits of, and agreements with, any Governmental Authority, in each case as in effect from time to time.

"LC Collateral Accounts" is defined in Section 2.22(b).

"<u>Lenders</u>" means the lending institutions listed on the signature pages of this Agreement and their respective successors and assigns, and any other Person that shall become a party hereto pursuant to an Assignment Agreement or a joinder agreement provided in <u>Section 2.6(b)(iv</u>), and as the context requires, includes the Swing Line Lender.

"<u>Lending Installation</u>" means, with respect to a Lender or the Administrative Agent, the office, branch, subsidiary or affiliate of such Lender or the Administrative Agent listed on the signature pages hereof or on a Schedule or otherwise selected by such Lender or the Administrative Agent pursuant to <u>Section 2.17</u>.

"Letter of Credit" means any standby letter of credit issued pursuant to Section 2.19 and any Existing Letter of Credit.

"Letter of Credit Application" is defined in Section 2.19(c).

"<u>Letter of Credit Commitment</u>" means, for each Issuer, the obligation of such Issuer to issue Letters of Credit in an aggregate amount not exceeding the amount set forth on <u>Schedule I</u> hereto or as set forth in any commitment agreement entered into by any Lender appointed as an Issuer by the Borrower in accordance with the terms hereof.

"Letter of Credit Fee" is defined in Section 2.19(d).

"<u>Letter of Credit Fee Rate</u>" means, at any time, the percentage rate per annum applicable to Letter of Credit Fees at such time as set forth in the Pricing Schedule.

"<u>Letter of Credit Obligations</u>" means, at any time, the sum, without duplication, of (i) the aggregate undrawn stated amount of all outstanding Letters of Credit at such time plus (ii) the aggregate unpaid amount of all Reimbursement Obligations at such time. For purposes of computing the amount available to be drawn under any Letter of Credit, the amount of such Letter of Credit shall be determined in accordance with <u>Section 1.3</u>. For all purposes of this Agreement, if on any date of determination a Letter of Credit has expired by its terms but any amount may still be drawn thereunder by reason of the operation of Rule 3.14 of the ISP, such Letter of Credit shall be deemed to be "outstanding" in the amount so remaining available to be drawn.

"Letter of Credit Payment Date" is defined in Section 2.19(e).

"<u>Letter of Credit Sublimit</u>" means an amount equal to the lesser of (a) \$50,000,000 and (b) the Aggregate Commitments. The Letter of Credit Sublimit is part of, and not in addition to, the Aggregate Commitments.

"<u>Lien</u>" means any lien (statutory or other), mortgage, pledge, hypothecation, assignment, deposit arrangement, encumbrance or preference, priority or other security agreement or preferential arrangement of any kind or nature whatsoever (including the interest of a vendor or lessor under any conditional sale, Capitalized Lease or other title retention agreement).

"Loan" means, with respect to a Lender, such Lender's loans made pursuant to <u>Article II</u> (or any conversion or continuation thereof), including the Swing Line Loans.

13

"Loan Documents" means this Agreement, each Note, any agreement creating or perfecting rights in Cash Collateral pursuant to the provisions of Section 2.22, each Letter of Credit, each Issuer Document, the Administrative Agent's Fee Letter and the Fee Letter.

"<u>London Banking Day</u>" means any day on which dealings in Dollar deposits are conducted by and between banks in the London interbank eurodollar market.

"Material Adverse Effect" means a material adverse effect on (i) the business, Property, financial condition, or results of operations of the Borrower and its Subsidiaries taken as a whole, (ii) the ability of the Borrower to perform its obligations under the Loan Documents or (iii) the validity or enforceability of any of the Loan Documents or the rights or remedies of the Agents, the Lenders or the Issuers thereunder.

"Material Indebtedness" is defined in Section 7.6.

"Modification" and "Modify" are defined in Section 2.19(a).

"Moody's" means Moody's Investors Service, Inc and any successor thereto.

"Multiemployer Plan" means "multiemployer plan" as defined in Section 3(37) of ERISA to which the Borrower or any member of the Controlled Group is obligated to make contributions.

"Non-Extending Lender" is defined in Section 2.21(b).

"Note" or "Notes" means the Revolving Notes and/or the Swing Line Note, as applicable.

"Notice Date" is defined in Section 2.21(b).

"Obligations" means all unpaid principal of and accrued and unpaid interest on the Loans, all Reimbursement Obligations and accrued and unpaid interest thereon, all accrued and unpaid fees and all expenses, reimbursements, indemnities and other obligations of the Borrower to any Lender, any Issuer, either Agent or any indemnified party arising under any Loan Document, including interest and fees after commencement of, by or against the Borrower of any proceeding under any the federal bankruptcy laws or any comparable provisions of any applicable state law naming such Person as the debtor in such proceeding, regardless of whether such interest and fees are allowed claims in such proceeding.

"Organization Documents" means, (a) with respect to any corporation, the certificate or articles of incorporation and the bylaws (or equivalent or comparable constitutive documents with respect to any non-U.S. jurisdiction); (b) with respect to any limited liability company, the certificate or articles of formation or organization and operating agreement; and (c) with respect to any partnership, joint venture, trust or other form of business entity, the

14

organization and, if applicable, any certificate or articles of formation or organization of such entity.

"Other Taxes" means all present or future stamp or documentary taxes or any other excise or property taxes, charges or similar levies (other than Excluded Taxes) arising from any payment made hereunder or under any other Loan Document or from the execution, delivery or enforcement of, or otherwise with respect to, this Agreement or any other Loan Document.

"Outstanding Credit Exposure" means, as to any Lender at any time, the sum of (a) the aggregate principal amount of its Loans outstanding at such time, plus (b) its Pro Rata Share of the Letter of Credit Obligations at such time, plus (c) its Pro Rata Share of Swing Line Loans at such time.

"Participant" is defined in Section 12.1(d).

"Payment Date" means the last Business Day of each March, June, September and December.

"PBGC" means the Pension Benefit Guaranty Corporation, or any successor thereto.

"<u>Person</u>" means any natural person, corporation, firm, joint venture, partnership, limited liability company, association, enterprise, trust or other entity or organization, or any government or political subdivision or any agency, department or instrumentality thereof.

"<u>Plan</u>" means an employee pension benefit plan which is covered by Title IV of ERISA or subject to the minimum funding standards under Section 412 or Section 430 of the Code as to which the Borrower or any member of the Controlled Group may have any liability.

"Platform" is defined in Section 6.1.

"Pricing Schedule" means Schedule III attached hereto identified as such.

"<u>Project Finance Subsidiary</u>" means any Subsidiary that meets the following requirements: (a) it is primarily engaged, directly or indirectly, in the ownership, operation and/or financing of independent power production and related facilities and assets; and (b) neither the Borrower nor any other Subsidiary (other than another Project Finance Subsidiary) has any liability, contingent or otherwise, for the Indebtedness or other obligations of such Subsidiary (other than non-recourse liability resulting from the pledge of stock of such Subsidiary).

"Property" of a Person means any and all property, whether real, personal, tangible, intangible, or mixed, of such Person, or other assets owned or leased by such Person.

"Pro Rata Share" means, with respect to any Lender on any date of determination, the percentage which the amount of such Lender's Commitment is of the Aggregate Commitment (or, if the Commitments have terminated, which such Lender's Outstanding Credit Exposure is

15

of the Aggregate Outstanding Credit Exposure) as of such date, subject to adjustment as provided in Section 2.23. For purposes of determining liability for any indemnity obligation under Section 9.6(c), each Lender's Pro Rata Share shall be determined as of the date the applicable Issuer, the Swing Line Lender or the Administrative Agent notifies the Lenders of such indemnity obligation (or, if such notice is given after termination of this Agreement, as of the date of such termination).

"Public Lender" is defined in Section 6.1.

"Register" is defined in Section 12.1(c).

"Regulation U" means Regulation U of the FRB as from time to time in effect and any successor or other regulation or official interpretation of the FRB relating to the extension of credit by banks for the purpose of purchasing or carrying margin stocks applicable to member banks of the Federal Reserve System.

"Reimbursement Obligations" means, at any time, the aggregate of all obligations of the Borrower then outstanding under <u>Section 2.19</u> to reimburse (i) the Issuers for amounts paid by the Issuers in respect of any one or more drawings under Letters of Credit and/or (ii) the Lenders for amounts paid by the Lenders to reimburse the Issuers pursuant to <u>Section 2.19</u>.

"Related Parties" means, with respect to any Person, such Person's Affiliates and the partners, directors, officers, employees, agents, trustees and advisors of such Person and of such Person's Affiliates.

"Reportable Event" means a reportable event as defined in Section 4043 of ERISA and the regulations issued under such section, with respect to a Plan, excluding, however, such events as to which the PBGC has by regulation waived the requirement of Section 4043(a) of ERISA that it be notified within thirty (30) days of the occurrence of such event; provided that a failure to meet the minimum funding standard of Section 412 or Section 430 of the Code and of Section 302 of ERISA shall be a Reportable Event regardless of the issuance of any such waiver of the notice requirement in accordance with either Section 4043(a) of ERISA or Section 412(d) of the Code.

"Required Lenders" means Lenders in the aggregate having more than fifty percent (50%) of the Aggregate Commitment or, if the Aggregate Commitment has been terminated, Lenders in the aggregate holding more than fifty percent (50%) of the Aggregate Outstanding Credit Exposure. The unfunded Commitments of, and the outstanding Loans, Letter of Credit Obligations and participations therein held or deemed held by, any Defaulting Lender shall be excluded for purposes of making a determination of Required Lenders.

"Revolving Note" is defined in Section 2.13(d).

"S&P" means Standard and Poor's Ratings Services, a division of The McGraw Hill Companies, Inc and any successor thereto.

16

"Schedule" refers to a specific schedule to this Agreement, unless another document is specifically referenced.

"SEC" means the Securities and Exchange Commission or any other U.S. federal governmental authority succeeding to any or all of the functions of the Securities and Exchange Commission.

"Section" means a numbered section of this Agreement, unless another document is specifically referenced.

"Shareholders' Equity" means, as of any date of determination for the Borrower and its Consolidated Subsidiaries on a consolidated basis, shareholders' equity as of that date determined in accordance with GAAP.

"Significant Subsidiary" means, at any time, KCPL and each other Subsidiary which (a) as of the date of determination, owns consolidated assets equal to or greater than fifteen percent (15%) of the consolidated assets of the Borrower and its Subsidiaries or (b) which had consolidated net income from continuing operations (excluding extraordinary items) during the four (4) most recently ended fiscal quarters equal to or greater than fifteen percent (15%) of Consolidated Net Income (excluding extraordinary items) during such period.

"Single Employer Plan" means a Plan, other than a Multiemployer Plan, maintained by the Borrower or any member of the Controlled Group for employees of the Borrower or any member of the Controlled Group.

"Specified Information" is defined in Section 9.11(a).

"Subsidiary" of a Person means (a) any corporation more than fifty percent (50%) of the outstanding securities having ordinary voting power of which shall at the time be owned or controlled, directly or indirectly, by such Person or by one or more of its Subsidiaries or by such Person and one or more of its Subsidiaries, (b) any partnership, limited liability company, association, joint venture or similar business organization more than fifty percent (50%) of the ownership interests having ordinary voting power of which shall at the time be so owned or controlled; or (c) any other Person the operations and/or financial results of which are required to be consolidated with those of such first Person in accordance with GAAP. Unless otherwise expressly s tated, all references herein to a "Subsidiary" shall mean a Subsidiary of the Borrower.

"Substantial Portion" means, with respect to the Property of the Borrower and its Subsidiaries, Property which represents more than fifteen percent (15%) of the consolidated assets of the Borrower and its Consolidated Subsidiaries as would be shown in the consolidated financial statements of the Borrower and its Consolidated Subsidiaries as at the beginning of the twelve-month period ending with the month in which such determination is made, or of the Consolidated Net Income (excluding extraordinary items) of the Borrower and its Consolidated Subsidiaries as reflected in the financial statements referred to above.

17

"Swap Contract" means (a) any and all rate swap transactions, basis swaps, credit derivative transactions, forward rate transactions, commodity swaps, commodity options, forward commodity contracts, equity or equity index swaps or options, bond or bond price or bond index swaps or options or forward bond or forward bond price or forward bond index transactions, interest rate options, forward foreign exchange transactions, cap transaction, floor transactions, collar transactions, currency swap transactions, cross-currency rate swap transactions, currency options, spot contracts, or any other similar transactions or any combination of any of the foregoing (including any options to enter into any of the foregoing), whether or not any such transaction is governed by or subject to any master agreement, and (b) any and all transactions of any kind, and the related confirmations, which are subject to the terms and conditions of, or governed by, any form of master agreement published by the International Swaps and Derivatives Association, Inc., any International Foreign Exchange Master Agreement, or any other master agreement (any such master agreement, together with any related schedules, a "Master Agreement"), including any such obligations or liabilities under any Master Agreement.

"Swing Line Lender" means Bank of America in its capacity as provider of Swing Line Loans, or any successor swing line lender hereunder.

"Swing Line Loan" is defined in Section 2.20(a).

"Swing Line Note" is defined in Section 2.13(d).

"Swing Line Loan Notice" is defined in Section 2.20(b).

"Swing Line Sublimit" means an amount equal to the lesser of (a) \$25,000,000 and (b) the Aggregate Commitments. The Swing Line Sublimit is part of, and not in addition to, the Aggregate Commitments.

"Syndication Agents" means Union Bank and Wells Fargo, each in its capacity as syndication agent hereunder, and not in its individual capacity as a Lender, and any successor thereto.

"Synthetic Lease Obligation" means the monetary obligation of a Person under (a) a so-called synthetic or tax retention lease, pursuant to which notwithstanding the off-balance sheet treatment of the lease obligation the assets are deemed owned by the lessee for U.S. federal income tax purposes, or (b) an agreement for the use or possession of property creating obligations that do not appear on the balance sheet of such Person but which, upon the insolvency or bankruptcy of such Person, would be characterized as the indebtedness of such Person (without regard to accounting treatment).

"<u>Taxes</u>" means all present or future taxes, levies, imposts, duties, deductions, withholdings (including backup withholding), assessments, fees or other charges imposed by any Governmental Authority, including any interest, additions to tax or penalties applicable thereto.

18

"34 Act Reports" means the periodic reports of the Borrower filed with the SEC on Forms 10-K, 10-Q and 8-K (or any successor forms thereto).

"<u>Total Capitalization</u>" means Total Indebtedness of the Borrower and its Consolidated Subsidiaries plus the sum of (a) Shareholders' Equity (without giving effect to the application of ASC Topic 815) and (b) to the extent not otherwise included in Total Indebtedness or Shareholders' Equity, preferred and preference stock and securities of the Borrower and its Subsidiaries included in a consolidated balance sheet of the Borrower and its Consolidated Subsidiaries in accordance with GAAP.

"<u>Total Indebtedness</u>" means all Indebtedness of the Borrower and its Consolidated Subsidiaries on a consolidated basis (and without duplication) but without giving effect to the application of ASC Topic 860 with respect to transfers of accounts receivable by KCPL, KCPL GMO or one or more of their respective Subsidiaries to a non-Subsidiary, excluding (a) Indebtedness arising under Swap Contracts entered into in the ordinary course of business to hedge bona fide transactions and business risks and not for speculation, (b) Indebtedness of Project Finance Subsidiaries, and (c) Indebtedness of KLT Investments Inc. incurred in connection with the acquisition and maintenance of its interests (whether direct o r indirect) in low income housing projects.

"Type" means, with respect to any Advance, its nature as a Floating Rate Advance or a Eurodollar Advance.

"Union Bank" means Union Bank, N.A. in its individual capacity and its successors.

"Unmatured Default" means an event which but for the lapse of time or the giving of notice, or both, would constitute a Default.

"Unreimbursed Amount" is defined in Section 2.19(f).

"Wells Fargo" means Wells Fargo Bank, National Association in its individual capacity and its successors.

"Wholly-Owned Subsidiary" of a Person means (i) any Subsidiary all of the outstanding voting securities of which (except directors' qualifying shares) shall at the time be owned or controlled, directly or indirectly, by such Person or one or more Wholly-Owned Subsidiaries of such Person, or by such Person and one or more Wholly-Owned Subsidiaries of such Person, or (ii) any partnership, limited liability company, association, joint venture or similar business organization one hundred percent (100%) of the ownership interests having ordinary voting power of which (except directors' qualifying shares) shall at the time be so owned or controlled.

1.2 <u>Accounting Principles</u>.

Unless the context otherwise clearly requires, all accounting terms not expressly defined herein shall be construed, and all financial computations required under this Agreement shall be made, in accordance with GAAP, consistently applied; <u>provided</u> that if the Borrower notifies the

19

Administrative Agent that the Borrower wishes to amend any covenant in <u>Article VI</u> to eliminate the effect of any change in GAAP on the operation of such covenant (or if the Administrative Agent notifies the Borrower that the Required Lenders wish to amend any covenant in <u>Article VI</u> for such purpose), then the Borrower's compliance with such covenant shall be determined on the basis of GAAP in effect immediately before the relevant change in GAAP became effective, until either such notice is withdrawn or such covenant is amended in a manner satisfactory to the Borrower and the Required Lenders.

1.3 <u>Letter of Credit Amounts</u>.

Unless otherwise specified herein, the amount of a Letter of Credit at any time shall be deemed to be the stated amount of such Letter of Credit in effect at such time; <u>provided</u> that with respect to any Letter of Credit that, by its terms or the terms of any Issuer Document related thereto, provides for one or more automatic increases in the stated amount thereof, the amount of such Letter of Credit shall be deemed to be the maximum stated amount of such Letter of Credit after giving effect to all such increases, whether or not such maximum stated amount is in effect at such time.

1.4 <u>Times of Day.</u>

Unless otherwise specified, all references herein to times of day shall be references to Eastern time (daylight or standard, as applicable).

THE CREDITS

2.1 <u>Commitment</u>.

From and including the date of this Agreement and prior to the Facility Termination Date, subject to the terms and conditions set forth in this Agreement, (a) each Lender severally agrees to make Loans to the Borrower from time to time in amounts not to exceed in the aggregate at any one time outstanding the amount of its Commitment, and (b) each Issuer agrees to issue Letters of Credit for the account of the Borrower from time to time in amounts not to exceed in the aggregate at any one time outstanding the amount of its Letter of Credit Commitment (and each Lender severally agrees to participate in each such Letter of Credit as more fully set forth in Section 2.19); provided (i) that the Aggregate Outstanding Credit Exposure shall not at any time exceed the Aggregate Commitment; (ii) the Outstanding Credit Exposure of any Lender shall not at any time exceed the amount of such Lender's Commitment; and (iii) the outstanding amount of Letter of Credit Obligations shall not exceed the Letter of Credit Sublimit. Subject to the terms of this Agreement, the Borrower may borrow, repay and reborrow Loans at any time prior to the Facility Termination Date. The Commitments shall expire on the Facility Termination Date.

20

2.2 Required Payments; Termination.

The Borrower shall (a) repay the principal amount of all Advances made to it on the Facility Termination Date and (b) deposit into the LC Collateral Accounts on the Facility Termination Date an amount in immediately available funds equal to the aggregate stated amount of all Letters of Credit that will remain outstanding after the Facility Termination Date.

2.3 Ratable Loans.

Each Advance hereunder shall consist of Loans made from the several Lenders ratably in proportion to their respective Pro Rata Shares.

2.4 Types of Advances; Minimum Amount.

The Advances may be Floating Rate Advances or Eurodollar Advances, or a combination thereof, selected by the Borrower in accordance with Sections 2.8 and 2.9. Each Eurodollar Advance shall be in the amount of \$5,000,000 or a higher integral multiple of \$1,000,000, and each Floating Rate Advance shall be in the amount of \$1,000,000 or an integral multiple thereof.

2.5 <u>Commitment Fee</u>.

The Borrower agrees to pay to the Administrative Agent for the account of each Lender which is not a Defaulting Lender in accordance with its Pro Rata Share, a commitment fee equal to the product of (i) the Commitment Fee Rate times (ii) the actual daily amount by which the Aggregate Commitments exceed the sum of (x) the outstanding principal amount of Loans (other than Swing Line Loans) and (y) the outstanding amount of Letter of Credit Obligations, subject to adjustment as provided in Section 2.23. The Commitment Fee shall be due and payable quarterly in arrears on each Payment Date, commencing with the first Payment Date to occur after the Closing Date, and on the Facility Termination Date. The Commitment Fee shall be calculated quarterly in arrears. For purposes of clarification, Swing Line Loans shall not be considered outstanding for purposes of determining the unused portion of the Aggregate Commitments.

2.6 <u>Changes in Aggregate Commitment.</u>

(a) Optional Commitment Reductions. The Borrower may at any time and from time to time, without penalty, permanently reduce the Aggregate Commitment in whole or in part ratably among the Lenders (according to their respective Pro Rata Shares) in a minimum principal amount of \$5,000,000 and integral multiples of \$1,000,000 in excess thereof, upon at least three (3) Business Days' prior written notice to the Administrative Agent, which notice shall specify the amount of any such reduction; provided that the amount of the Aggregate Commitment may not be reduced below the Aggregate Outstanding Credit Exposure. All accrued commitment fees shall be payable on the effective date of any termination of the obligations of the Lenders to make Loans hereunder.

21

- (b) Optional Increase of Commitments. The Borrower may at any time and from time to time, upon five (5) days' prior written notice to the Administrative Agent, increase the Commitments (but not the Letter of Credit Sublimit or the Swing Line Sublimit) with additional Commitments from any existing Lender with a Commitment or new Commitments from any other Person selected by the Borrower and reasonably acceptable to the Administrative Agent, the Swing Line Lender and the Issuers; provided that:
 - (i) any such increase shall be in a minimum principal amount of \$10,000,000 and in integral multiples of \$1,000,000 in excess thereof;
 - (ii) no Default or Unmatured Default shall exist and be continuing at the time of any such increase;
 - (iii) no existing Lender shall be under any obligation to increase its Commitment and any such decision whether to increase its Commitment shall be in such Lender's sole and absolute discretion;

- (iv) (A) any new Lender shall join this Agreement by executing such joinder documents required by the Administrative Agent and/or (B) any existing Lender electing to increase its Commitment shall have executed a commitment agreement reasonably satisfactory to the Administrative Agent;
- (v) as a condition precedent to such increase, the Borrower shall deliver to the Administrative Agent (A) a certificate of each of the Borrower, KCPL and KCPL GMO dated as of the date of such increase (in sufficient copies for each Lender) signed by an Authorized Officer of such Person certifying and attaching the resolutions adopted by such Person approving or consenting to such increase, and (B) a certificate of the Borrower certifying that, before and after giving effect to such increase, (x) the representations and warranties contained in Article V and the other Loan Documents are true and correct in all material respects on and as of the date of such increase, except to the extent that such representations and warranties specifically refer to an earlier date, in which case they are true and correct in all material respects as of such earlier date, and (y) no Default or Unmatured Default exists as of such date; and
- (vi) after giving effect to the increase in the Commitments (but without giving effect to any increase in the Commitments pursuant to $\underline{\text{Sections 2.6(c)}}$, $\underline{\text{(d)}}$, $\underline{\text{(e)}}$, or $\underline{\text{(f)}}$), the Aggregate Commitment shall not be greater than FOUR HUNDRED MILLION DOLLARS (\$400,000,000).

The Borrower shall prepay any Loans owing by it and outstanding on the date of any such increase with the proceeds required to be advanced by any Lender increasing its, or any Person providing a new, Commitment under this Section (and pay any additional amounts required pursuant to Article III) to the extent necessary to keep the outstanding Loans ratable with any revised Commitments arising from any nonratable increase in the Commitments under this Section.

22

- (c) <u>Transfer and Re-Transfer of the Commitments to KCPL</u>. (i) Subject to Section 4.2 of the KCPL Credit Agreement, the Borrower and KCPL may, by joint election in a written notice to the Administrative Agent (which shall promptly provide a copy of such notice to the Lenders) and the "Administrative Agent" under the KCPL Credit Agreement, transfer from time to time up to TWO HUNDRED MILLION DOLLARS (\$200,000,000) of the unused Commitments to the Commitments (as such term is defined in the KCPL Credit Agreement) under the KCPL Credit Agreement (any such reduction, a "GPE-KCPL Transfer").
- (ii) Subject to <u>Section 4.2</u>, the Borrower and KCPL may, by joint election in a written notice to the Administrative Agent (which shall promptly provide a copy of such notice to the Lenders) and the "Administrative Agent" under the KCPL Credit Agreement, re-transfer from time to time any portion (up to \$200,000,000) of the unused Commitments (as such term is defined in the KCPL Credit Agreement) previously transferred from this Agreement to the KCPL Credit Agreement pursuant to <u>subclause (c)(i)</u> above back to the Commitments hereunder (any such addition, a "<u>GPE-KCPL Re-Transfer</u>").
- (iii) On the effective date of a GPE-KCPL Transfer, which shall be specified in the notice delivered pursuant to Section 2.6(c)(i) and which shall not be less than five (5) Business Days subsequent to the date of giving of such notice, then subject to the satisfaction of the conditions precedent specified in Section 4.2 of the KCPL Credit Agreement, (i) the Commitments hereunder shall be ratably decreased by the aggregate amount specified in such notice and (ii) the aggregate amount of the "Commitments" under and as defined in the KCPL Credit Agreement shall be ratably increased by such amount. Such GPE-KCPL Transfer and the consequent de creases and increases shall be irrevocable subject, however, to subsequent permissible GPE-KCPL Re-Transfers in accordance with the terms hereof.
- (iv) On the effective date of a GPE-KCPL Re-Transfer, which shall be specified in the notice delivered pursuant to Section 2.6(c)(ii) and which shall not be less than five (5) Business Days subsequent to the date of giving of such notice, then subject to the satisfaction of the conditions precedent specified in Section 4.2, (i) the Commitments hereunder shall be ratably increased by the aggregate amount specified in such notice and (ii) the aggregate amount of the "Commitments" under and as defined in the KCPL Credit Agreement shall be ratably decreased by such amount. Such GPE-KCPL Re-Transfer and the consequent decreases and increases shall be irrevocable, subject, however, to subsequent permissible GPE-KCPL Transfers in accordance with the terms hereof.
- (d) <u>Transfer and Re-Transfer of the Commitments to KCPL GMO</u>. (i) Subject to Section 5.2 of the KCPL GMO Credit Agreement, the Borrower and KCPL GMO may, by joint election in a written notice to the Administrative Agent (which shall promptly provide a copy of such notice to the Lenders) and the "Administrative Agent" under the KCPL GMO Credit Agreement, transfer from time to time up to TWO HUNDRED MILLION DOLLARS (\$200,000,000) of the unused Commitments to the Commitments (as such term is defined in the KCPL GMO Credit Agreement) under the KCPL GMO Credit Agreement (any such reduction, a "GPE-GMO Transfer").

23

- (ii) Subject to Section 4.2, the Borrower and KCPL GMO may, by joint election in a written notice to the Administrative Agent (which shall promptly provide a copy of such notice to the Lenders) and the "Administrative Agent" under the KCPL GMO Credit Agreement, re-transfer from time to time any portion (up to \$200,000,000) of the unused Commitments (as such term is defined in the KCPL GMO Credit Agreement) previously transferred from this Agreement to the KCPL GMO Credit Agreement pursuant to subclause (d)(i) above back to the Commitments hereunder (any such addition, a "GPE-GMO Re-Transfer").
- (iii) On the effective date of a GPE-GMO Transfer, which shall be specified in the notice delivered pursuant to Section 2.6(d)(i) and which shall not be less than five (5) Business Days subsequent to the date of giving of such notice, then subject to the satisfaction of the conditions precedent specified in Section 5.2 of the KCPL GMO Credit Agreement, (i) the Commitments hereunder shall be ratably decreased by the aggregate amount specified in such notice and (ii) the aggregate amount of the "Commitments" under and as defined in the KCPL GMO Credit Agreement shall be ratably increased by such

amount. Such GPE-GMO Transfer and the consequ ent decreases and increases shall be irrevocable subject, however, to subsequent permissible GPE-GMO Re-Transfers in accordance with the terms hereof.

- (iv) On the effective date of a GPE-GMO Re-Transfer, which shall be specified in the notice delivered pursuant to Section 2.6(d)(ii) and which shall not be less than five (5) Business Days subsequent to the date of giving of such notice, then subject to the satisfaction of the conditions precedent specified in Section 4.2, (i) the Commitments hereunder shall be ratably increased by the aggregate amount specified in such notice and (ii) the aggregate amount of the "Commitments" under and as defined in the KCPL GMO Credit Agreement shall be ratably decreased by such amount. Such GPE-GMO Re-Transfer and the consequent decrease s and increases shall be irrevocable, subject, however, to subsequent permissible GPE-GMO Transfers in accordance with the terms hereof.
- (e) <u>Transfer and Re-Transfer of the "Commitments" of KCPL to the Borrower</u>. (i) Subject to <u>Section 4.2</u>, the Borrower and KCPL may, by joint election in a written notice to the Administrative Agent (which shall promptly provide a copy of such notice to the Lenders) and the "Administrative Agent" under the KCPL Credit Agreement, transfer from time to time up to TWO HUNDRED MILLION DOLLARS (\$200,000,000) of the unused Commitments (as such term is defined in the KCPL Credit Agreement) under the KCPL Credit Agreement to the Commitments hereunder (any such addition, a "KCPL-GPE Transfer").
- (ii) Subject to Section 4.2 of the KCPL Credit Agreement, the Borrower and KCPL may, by joint election in a written notice to the Administrative Agent (which shall promptly provide a copy of such notice to the Lenders) and the "Administrative Agent" under the KCPL Credit Agreement, re-transfer from time to time any portion (up to \$200,000,000) of the unused Commitments previously transferred from the KCPL Credit Agreement to this Agreement pursuant to subclause (e)(i) above back to the Commitments (as such term is defined in the KCPL Credit Agreement) under the KCPL Credit Agreement (any such decrease, a "KCPL-GPE Re-Transfe r").

24

- (iii) On the effective date of a KCPL-GPE Transfer, which shall be specified in the notice delivered pursuant to Section 2.6(e)(i) and which shall not be less than five (5) Business Days subsequent to the date of giving of such notice, then subject to (A) the satisfaction of the conditions precedent specified in Section 4.2 and (B) if necessary in the reasonable discretion of the Administrative Agent, receipt by the Administrative Agent of a certificate of the Borrower dated as of the date of such KCPL-GPE Transfer (in sufficient copies for each Lender) signed by an Authorized Officer of the Borrower certifying and attaching resolutions adopted by the Borrower approving or consenting to borrowings up to at least the amount of the Commitments after giving effect to such KCPL-GPE Transfer, (i) the Commitments hereunder shall be ratably increased by the aggregate amount specified in such notice and (ii) the aggregate amount of the "Commitments" under and as defined in the KCPL Credit Agreement shall be ratably decreased by such amount. Such KCPL-GPE Transfer and the consequent increases and decreases shall be irrevocable subject, however, to subsequent permissible KCPL-GPE Re-Transfers in accordance with the terms hereof.
- (iv) On the effective date of a KCPL-GPE Re-Transfer, which shall be specified in the notice delivered pursuant to Section 2.6(e)(ii) and which shall not be less than five (5) Business Days subsequent to the date of giving of such notice, then subject to the satisfaction of the conditions precedent specified in Section 4.2 of the KCPL Credit Agreement, (i) the "Commitments" under and as defined in the KCPL Credit Agreement shall be ratably increased by the aggregate amount specified in such notice and (ii) the aggregate amount of the Commitments hereunder shall be ratably decreased by such amount. Such KCPL-GPE Re-Transfer and the consequent increases and decreases shall be irrevocable, subject, however, to subsequent permissible KCPL-GPE Transfers in accordance with the terms hereof.
- (f) <u>Transfer and Re-Transfer of the "Commitments" of KCPL GMO to the Borrower</u>. (i) Subject to <u>Section 4.2</u>, the Borrower and KCPL GMO may, by joint election in a written notice to the Administrative Agent (which shall promptly provide a copy of such notice to the Lenders) and the "Administrative Agent" under the KCPL GMO Credit Agreement, transfer from time to time up to TWO HUNDRED MILLION DOLLARS (\$200,000,000) of the unused Commitments (as such term is defined in the KCPL GMO Credit Agreement) under the KCPL GMO Credit Agreement to the Commitments hereunder (any such addition, a "<u>GMO-GPE Transfer</u>").
- (ii) Subject to Section 5.2 of the KCPL GMO Credit Agreement, the Borrower and KCPL GMO may, by joint election in a written notice to the Administrative Agent (which shall promptly provide a copy of such notice to the Lenders) and the "Administrative Agent" under the KCPL GMO Credit Agreement, re-transfer from time to time any portion (up to \$200,000,000) of the unused Commitments previously transferred from the KCPL GMO Credit Agreement to this Agreement pursuant to <u>subclause (f)(i)</u> above back to the Commitments (as such term is defined in the KCPL GMO Credit Agreement) under the KCPL GMO Credit Agreement (any such decrease, a "GMO-GPE Re-Transfer").
- (iii) On the effective date of a GMO-GPE Transfer, which shall be specified in the notice delivered pursuant to <u>Section 2.6(f)(i)</u> and which shall not be less than five (5) Business Days subsequent to the date of giving of such notice, then subject to (A) the satisfaction of the

25

conditions precedent specified in Section 4.2 and (B) if necessary in the reasonable discretion of the Administrative Agent, receipt by the Administrative Agent of a certificate of the Borrower dated as of the date of such GMO-GPE Transfer (in sufficient copies for each Lender) signed by an Authorized Officer of the Borrower certifying and attaching resolutions adopted by the Borrower approving or consenting to borrowings up to at least the amount of the Commitments after giving effect to such GMO-GPE Transfer, (i) the Commitments hereunder shall be ratably increased by the aggregate amount specified in such notice and (ii) the aggregate amount of the "Commitments" under and as defined in the KCPL GMO Credit Agreement shall be ratably decreased by such amount. [] 60; Such GMO-GPE Transfer and the consequent increases and decreases shall be irrevocable subject, however, to subsequent permissible GMO-GPE Re-Transfers in accordance with the terms hereof.

(iv) On the effective date of a GMO-GPE Re-Transfer, which shall be specified in the notice delivered pursuant to Section 2.6(f)(ii) and which shall not be less than five (5) Business Days subsequent to the date of giving of such notice, then subject to the satisfaction of the conditions precedent

specified in Section 5.2 of the KCPL GMO Credit Agreement, (i) the "Commitments" under and as defined in the KCPL GMO Credit Agreement shall be ratably increased by the aggregate amount specified in such notice and (ii) the aggregate amount of the Commitments hereunder shall be ratably decreased by such amount. Such GMO-GPE Re-Transfer and the consequent increases and decreases shall be irrevoc able, subject, however, to subsequent permissible GMO-GPE Transfers in accordance with the terms hereof.

Notwithstanding the foregoing, the parties acknowledge and agree that after giving effect to any increase in the Commitments pursuant to any GPE-KCPL Transfer or GPE-KCPL Re-Transfer contemplated in subclauses (i) and (ii) of Section 2.6(g), any GPE-GMO Transfer or GPE-GMO Re-Transfer contemplated in subclauses (i) and (ii) of Section 2.6(g), and any GMO-GPE Transfer or GMO-GPE Re-Transfer contemplated in subclauses (i) and (ii) of Section 2.6(g), (a) the aggregate Commitments her eunder shall not exceed \$400,000,000, (b) the aggregate Commitments under and as defined in the KCPL Credit Agreement shall not exceed \$800,000,000, (c) the aggregate Commitments under and as defined in the KCPL GMO Credit Agreement shall not exceed \$650,000,000 and (d) the aggregate commitments under this Agreement, the KCPL Credit Agreement and the KCPL GMO Credit Agreement shall not exceed \$1,250,000,000; provided, however, that after giving effect to such GPE-KCPL Transfers, GPE-KCPL Re-Transfers, GPE-GMO Transfers, GPE-GMO Re-Transfers, KCPL-GPE Transfers, KCPL-GPE Re-Transfers, GMO-GPE Transfers, GMO-GPE Re-Transfers, Section 2.6(b), Section 2.6(b) of the KCPL Credit Agreement, and Section 2.6(b) of the KCPL GMO Credit Agreement, (w) the aggregate Commitments hereunder shall not exceed \$600,000,000, (x) the aggregate Commitments under and as defined in the KCPL Credit Agreement shall not exceed \$1,000,000,000, (y) the aggregate Commitments under and as defined in the KCPL GMO Credit Agreement shall not exceed \$850,000,000 and (z) the aggregate commitments under this Agreement, the KCPL Credit Agreement and the KCPL GMO Credit Agreement shall not exceed \$850,000,000,000.

26

2.7 Optional Prepayments.

- (a) The Borrower may at any time and from time to time, in whole or in part, prepay Floating Rate Advances upon one (1) Business Day's prior written notice to the Administrative Agent, without penalty or premium. Each partial prepayment of Floating Rate Advances shall be in an aggregate amount of \$1,000,000 or an integral multiple thereof. Each such notice shall specify the date and amount of such prepayment, and the payment amount specified in such notice shall be due and payable on the date specified therein; provided, however, subject to Section 3.4, t hat a notice of prepayment may state that such notice is conditioned upon the effectiveness of other credit facilities or the closing of a capital markets transaction, in which case such notice may be revoked by the Borrower (by notice to the Administrative Agent on or prior to the specified prepayment date) if such condition is not satisfied.
- (b) The Borrower may at any time and from time to time, in whole or in part, prepay Eurodollar Advances (subject to the payment of any funding indemnification amounts required by Section 3.4) upon three (3) Business Days' prior written notice to the Administrative Agent, without penalty or premium. Each partial prepayment of Eurodollar Advances shall be in an aggregate amount of \$5,000,000 or a higher integral multiple of \$1,000,000. Each such notice shall specify the date and amount of such prepayment, and the payment amount specified in such notice shall be due and payable on the date specified therein; provided, however, subject to Section 3.4, that a notice of prepayment may state that such notice is conditioned upon the effectiveness of other credit facilities or the closing of a capital markets transaction, in which case such notice may be revoked by the Borrower (by notice to the Administrative Agent on or prior to the specified prepayment date) if such condition is not satisfied. Any notice to prepay Eurodollar Advances shall indicate the Interest Period for such Eurodollar Advances.
- (c) Subject to Section 2.23, all prepayments of Advances shall be applied ratably to the Loans of the Lenders in accordance with their respective Pro Rata Shares.
- (d) The Borrower may at any time and from time to time prepay Swing Line Loans upon one (1) Business Day's prior notice to the Swing Line Lender (with a copy to the Administrative Agent), without penalty or premium; provided that (i) such notice must be received by the Swing Line Lender and the Administrative Agent not later than 1:00 p.m. on the date of the prepayment, and (ii) any such prepayment shall be in a minimum principal amount of \$500,000 or a whole multiple of \$100,000 in excess thereof (or, if less, the entire principal thereof then outstanding). Each such notice shall specify the date and amount of such prepayment. If such notice is given by the Borrower, the Borrower's hall make such prepayment and the payment amount specified in such notice shall be due and payable on the date specified therein; provided, however, subject to Section 3.4, that a notice of prepayment may state that such notice is conditioned upon the effectiveness of other credit facilities or the closing of a capital markets transaction, in which case such notice may be revoked by the Borrower (by notice to the Administrative Agent on or prior to the specified prepayment date) if such condition is not satisfied.

27

2.8 <u>Method of Selecting Types and Interest Periods for New Advances.</u>

The Borrower shall select the Type of Advance and, in the case of each Eurodollar Advance, the Interest Period applicable thereto from time to time. The Borrower shall give the Administrative Agent irrevocable notice (a "Borrowing Notice") not later than noon on the Borrowing Date of each Floating Rate Advance and not later than noon three (3) Business Days before the Borrowing Date for each Eurodollar Advance, specifying:

- (i) the Borrowing Date, which shall be a Business Day, of such Advance,
- (ii) the aggregate amount of such Advance,
- (iii) the Type of Advance selected, and
- (iv) in the case of each Eurodollar Advance, the Interest Period applicable thereto.

Not later than 1:00 p.m. on each Borrowing Date, each Lender shall make available its Loan or Loans in funds immediately available to the Administrative Agent at its address specified pursuant to <u>Article XIII</u>. The Administrative Agent will make the funds so received from the Lenders available to the Borrower at the Administrative Agent's aforesaid address.

2.9 <u>Conversion and Continuation of Outstanding Advances.</u>

Floating Rate Advances shall continue as Floating Rate Advances unless and until such Floating Rate Advances are converted into Eurodollar Advances pursuant to this Section 2.9 or are repaid in accordance with Section 2.7. Each Eurodollar Advance shall continue as a Eurodollar Advance until the end of the then applicable Interest Period therefor, at which time such Eurodollar Advance shall be automatically converted into a Floating Rate Advance unless (x) such Eurodollar Advance is or was repaid in accordance with Section 2.7 or (y) the Borrower shall have given the Administrative Agent a Conversion/Continuation Notice (as defined below) requesting that, at the end of such Interest Period, such Eurodollar Advance continue as a Eurodollar Advance for the same or another Interest Period. Subject to the terms of Section 2.4, the Borrower may elect from time to time to convert all or any part of a Floating Rate Advance into a Eurodollar Advance. The Borrower shall give the Administrative Agent irrevocable notice (a "Conversion/Continuation Notice") of each conversion of a Floating Rate Advance into a Eurodollar Advance or continuation of a Eurodollar Advance not later than 11:00 a.m. at least three (3) Business Days prior to the date of the requested conversion or continuation, specifying:

- (i) the requested date, which shall be a Business Day, of such conversion or continuation,
- (ii) the aggregate amount and Type of the Advance which is to be converted or continued, and
- (iii) the amount of such Advance which is to be converted into or continued as a Eurodollar Advance and the duration of the Interest Period applicable thereto.

28

2.10 <u>Interest Rates</u>.

Each Floating Rate Advance shall bear interest on the outstanding principal amount thereof, for each day from and including the date such Advance is made or is automatically converted from a Eurodollar Advance into a Floating Rate Advance pursuant to Section 2.9, to but excluding the date it is paid or is converted into a Eurodollar Advance pursuant to Section 2.9 hereof, at a rate per annum equal to the Alternate Base Rate plus the Applicable Margin for such day. Changes in the rate of interest on that portion of any Advance maintained as a Floating Rate Advance will take effect simultaneously with each change in the Alternate Base Rate. Each Eurodollar Advance shall bear inter est on the outstanding principal amount thereof from and including the first day of the Interest Period applicable thereto to (but not including) the last day of such Interest Period at the Eurodollar Rate plus the Applicable Margin based upon the Borrower's selections under Sections 2.8 and 2.9 and otherwise in accordance with the terms hereof. No Interest Period may end after the Facility Termination Date. Each Swing Line Loan shall bear interest on the outstanding principal amount thereof from the applicable borrowing date at a rate per annum equal to the Alternate Base Rate plus the Applicable Margin for each day from and including the date such Swing Line Loan is made to but excluding the date it is paid. Changes in the rate of interest on that portion of any Swing Line Loan will take effect simultaneously with each change in the Alternate Base Rate.

2.11 Rates Applicable After Default.

Notwithstanding anything to the contrary contained in Section 2.8 or 2.9, during the continuance of a Default or Unmatured Default the Required Lenders may, at their option, by notice to the Borrower (which notice may be revoked at the option of the Required Lenders notwithstanding any provision of Section 8.2 requiring unanimous consent of the Lenders to changes in interest rates), declare that no Advance may be made as, converted into or continued as a Eurodollar Advance. During the continuance of a Default the Required Lenders may, at their option, by notice to the Borrower (which notice may be revoked at the option of the Required Lenders notwithstanding any provision of Section 8.2 requiring unanimous consent of the Lenders to changes in interest rates), declare that (i) each Eurodollar Advance shall bear interest for the remainder of the applicable Interest Period at the rate otherwise applicable to such Interest Period plus two percent (2%) per annum, (ii) each Floating Rate Advance shall bear interest at a rate per annum equal to the Alternate Base Rate in effect from time to time plus the Applicable Margin plus two percent (2%) per annum, (iii) the Letter of Credit Fee Rate shall be increased by two percent (2%) per annum and (iv) each Swing Line Loan shall bear interest at a rate per annum equal to the Alternate Base Rate in effect from time to time plus the Applicable Margin plus two percent (2%) per annum; provided that, during the continuance of a Default under Sections 7.7 or 7.8, the interest rates set forth in clauses (i), (ii) and (iv) above and the increase in the Letter of Credit Fee Rate set forth in clause (iii) above shall be applicable to all applicable Credit Extensions without any election or action on the part of the Administrative Agent or any Lender.

29

2.12 <u>Method of Payment.</u>

Except to the extent that any Tax is required to be withheld or deducted under applicable Law, but subject to the provisions of Article III, all payments of the Obligations hereunder shall be made, without setoff, deduction, or counterclaim, in immediately available funds to the Administrative Agent at the Administrative Agent among the Lenders in accordance with their respective Pro Rata Shares. Each payment delivered to the Administrative Agent for the account of any Lender shall be delivered promptly by the Administrative Agent to such Lender in the same type of funds that the Administrative Agent received at the Administrative Agent's Office from such Lender.

2.13 Noteless Agreement; Evidence of Indebtedness.

- (a) Each Lender shall maintain in accordance with its usual practice an account or accounts evidencing the indebtedness of the Borrower to such Lender resulting from each Loan made by such Lender from time to time, including the amounts of principal and interest payable and paid to such Lender from time to time hereunder.
- (b) The Administrative Agent shall also maintain accounts in which it will record (i) the amount of each Loan (including any Swing Line Loan) made hereunder, the Type thereof and the Interest Period with respect thereto, (ii) the amount of any principal or interest due and payable or to become due and payable from the Borrower to each Lender hereunder, (iii) the original stated amount of each Letter of Credit and the amount of Letter of Credit Obligations outstanding at any time and (iv) the amount of any sum received by the Administrative Agent hereunder from the Borrower and each Lender's share thereof.
- (c) The entries maintained in the accounts maintained pursuant to <u>clauses (a)</u> and <u>(b)</u> above shall be *prima facie* evidence of the existence and amounts of the Obligations therein recorded; <u>provided</u> that the failure of the Administrative Agent or any Lender to maintain such accounts or any error therein shall not in any manner affect the obligation of the Borrower to repay the Obligations in accordance with their terms.
- (d) Any Lender may request that its Loans be evidenced by a promissory note substantially in the form of Exhibit D (a "Revolving Note"). In such event, the Borrower shall prepare, execute and deliver to such Lender a Note payable to the order of such Lender. Thereafter, the Loans evidenced by such Note and interest thereon shall at all times (including after any assignment pursuant to Section 12.1(b)) be represented by one or more Notes payable to the order of the payee named therein or any assignee pursuant to Section 12.1(b), except to the extent that any such Lender or assignee subsequently returns any such Note for cancellation and requests that such Loans once again be evidenced as described in clauses (a) and (b) above. The Swing Line Loans shall be evidenced by a promissory note substantially in the form of Exhibit E (the "Swing Line Note").

30

2.14 <u>Telephonic Notices</u>.

The Borrower hereby authorizes the Lenders and the Administrative Agent to extend, convert or continue Advances, effect selections of Types of Advances and to transfer funds based on telephonic notices made by any person or persons the Administrative Agent or any Lender in good faith believes to be acting on behalf of the Borrower. The Borrower agrees to deliver promptly to the Administrative Agent a written confirmation, if such confirmation is requested by the Administrative Agent or any Lender, of each telephonic notice signed by an Authorized Officer. If the written confirmation differs in any material respect from the action taken by the Administrative Agent and the Lenders, the records of the Administrative Agent and the Lenders shall govern absent manifest error.

2.15 <u>Interest Payment Dates; Interest and Fee Basis.</u>

Interest accrued on each Floating Rate Advance and each Swing Line Loan shall be payable on each Payment Date, commencing with the first such date to occur after the date hereof, and at maturity. Interest accrued on each Eurodollar Advance shall be payable on the last day of its applicable Interest Period, on any date on which such Eurodollar Advance is prepaid, whether by acceleration or otherwise, and at maturity. Interest accrued on each Eurodollar Advance having an Interest Period longer than three (3) months shall also be payable on the last day of each three (3)-month interval during such Interest Period. All computations of interest for Floating Rate Loans and Swing Line Loans (including Floating Rate Loans determined by reference to the Eurodollar Rate) shall be made on the basis of a year of 365 or 366 days, as the case may be, and actual days elapsed. All other computations of interest and fees shall be calculated for actual days elapsed on the basis of a 360-day year. Interest shall be payable for the day an Advance is made but not for the day of any payment on the amount paid if payment is received prior to 1:00 p.m. at the place of payment (it being understood that the Administrative Agent shall be deemed to have received a payment prior to 1:00 p.m. if (x) the Borrower has provided the Administrative Agent with evidence satisfactory to the Administrative Agent that the Borrower has initiated a wire transfer of such payment prior to such time and (y) the Administrative Agent actually receives such payment on the same Business Day on which such wire transfer was initiated). If any payment of principal of or interest on an Advance shall become due on a day which is not a Business Day, such payment shall be made on the next succeeding Business Day and, in the case of a principal payment, such extension of time shall be included in computing interest in connection with such payment.

2.16 <u>Notification of Advances, Interest Rates, Prepayments and Commitment Reductions.</u>

Promptly after receipt thereof, the Administrative Agent will notify each Lender of the contents of each Aggregate Commitment reduction notice, Borrowing Notice, Conversion/Continuation Notice, and repayment notice received by it hereunder. The Administrative Agent will notify each Lender of the interest rate applicable to each Eurodollar Advance promptly upon determination of such interest rate and will give each Lender prompt notice of each change in the Alternate Base Rate. The Administrative Agent will also promptly notify each Lender of any increase or reduction of the Aggregate Commitments pursuant to the

31

terms hereof. The Administrative Agent will also promptly notify the Borrower of the interest rate applicable to each Eurodollar Advance and Floating Rate Advance promptly upon determination of such interest rates and will give the Borrower prompt notice of each change in the Alternate Base Rate.

2.17 <u>Lending Installations</u>.

Each Lender may book its Loans at any Lending Installation selected by such Lender and may change its Lending Installation from time to time. All terms of this Agreement shall apply to any such Lending Installation and the Loans and any Notes issued hereunder shall be deemed held by each Lender for the benefit of such Lending Installation. Each Lender may, by written notice to the Administrative Agent and the Borrower in accordance with Article XIII, designate replacement or additional Lending Installations through which Loans will be made by it and for whose account Loan payments are to be made;

<u>provided</u>, <u>however</u>, that such designation shall be permitted by applicable Laws and that such designation shall not result in the imposition of increased costs or obligations on the Borrower.

2.18 Non-Receipt of Funds by the Administrative Agent.

Unless a Lender notifies the Administrative Agent, prior to the proposed date of any Eurodollar Advance (or, in the case of any Floating Rate Advance, prior to noon on the date of such Advance), that such Lender will not make available to the Administrative Agent such Lender's share of such Advance, the Administrative Agent may assume that such payment has been made. Unless the Borrower notifies the Administrative Agent, prior to the date on which it is scheduled to make a payment to the Administrative Agent of principal, interest or fees for the account of the Lenders, that it does not intend to make such payment, the Administrative Agent may assume that such payment has been made. The Administrative Agent may, but shall not be obligated to, make the amount of such payment available to the intended recipient in reliance upon such assumption. If such Lender or the Borrower, as the case may be, has not in fact made such payment to the Administrative Agent, the recipient of such payment shall, on demand by the Administrative Agent, repay to the Administrative Agent the amount so made available together with interest thereon in respect of each day during the period commencing on the date such amount was so made available by the Administrative Agent until the date the Administrative Agent recovers such amount at a rate per annum equal to (x) in the case of payment by a Lender, the Federal Funds Rate for such day or (y) in the case of payment by the Borrower, the interest rate applicable to the relevant Loan.

2.19 Letters of Credit.

(a) <u>Issuance</u>. Each Issuer hereby agrees, on the terms and conditions set forth in this Agreement and in reliance upon the agreements of the Lenders set forth in this <u>Section 2.19</u>, to issue Letters of Credit and to extend, increase, decrease or otherwise modify Letters of Credit ("<u>Modify</u>," and each such action a "<u>Modification</u>") in amounts not to exceed in the aggregate at any one time outstanding the amount of its Letter of Credit Commitment from time to time on any Business Day from a nd including the date of this Agreement and prior to the Facility Termination Date upon the request of the Borrower; <u>provided</u> that immediately after each such

32

Letter of Credit is issued or Modified, (x) the Aggregate Outstanding Credit Exposure shall not exceed the Aggregate Commitment and (y) the outstanding amount of Letters of Credit Obligations shall not exceed the Letter of Credit Sublimit. No Letter of Credit shall have an expiry date later than the date that is five (5) days prior to the scheduled Facility Termination Date. All Existing Letters of Credit shall be deemed to have been issued pursuant hereto, and from and after the Closing Date shall be subject to and governed by the terms and conditions hereof.

- (b) <u>Participations</u>. Upon the issuance or Modification by any Issuer of a Letter of Credit in accordance with this <u>Section 2.19</u>, such Issuer shall be deemed, without further action by any Person, to have unconditionally and irrevocably sold to each Lender, and each Lender shall be deemed, without further action by any Person, to have unconditionally and irrevocably purchased from such Issuer, a participation in such Letter of Credit (and each Modification thereof) and the related Letter of Credit Obligations in proportion to its Pro Rata Share.
- Notice. Subject to Section 2.19(a), the Borrower shall give the applicable Issuer and the Administrative Agent notice prior to 11:00 a.m. at least three (3) Business Days (or such lesser period of time as the Administrative Agent and such Issuer may agree in their sole discretion) prior to the proposed date of issuance or Modification of each Letter of Credit, specifying, in form and detail reasonably satisfactory to the applicable Issuer, (A) the proposed issuance date of the requested Letter of Credit (which shall be a Business Day) and the expiry date of such Letter of Credit; (B) the amount thereof; (C) the name and a ddress of the beneficiary thereof; (D) the documents to be presented by such beneficiary in case of any drawing thereunder; (E) the full text of any certificate to be presented by such beneficiary in case of any drawing thereunder; (F) the purpose and nature of the requested Letter of Credit; and (G) such other matters as the applicable Issuer may reasonably require. In the case of a request for a Modification of any outstanding Letter of Credit, such notice shall specify in form and detail reasonably satisfactory to the applicable Issuer (A) the Letter of Credit to be amended; (B) the proposed date of amendment thereof (which shall be a Business Day); (C) the nature of the proposed amendment; and (D) such other matters as the applicable Issuer may reasonably require. Upon receipt of such notice, the applicable Issuer shall promptly notify the Administrative Agent, and the Administrative Agent shall promptly notify each Lender, of the contents thereof and of the amount of such Lender 217;s participation in such proposed Letter of Credit. The issuance or Modification by an Issuer of any Letter of Credit shall, in addition to the conditions precedent set forth in Article IV (the satisfaction of which such Issuer shall have no duty to ascertain, it being understood, however, that such Issuer shall not issue any Letter of Credit if it has received written notice from the Borrower, the Administrative Agent or any Lender one (1) day prior to the proposed date of issuance, that any such condition precedent has not been satisfied), be subject to the conditions precedent that such Letter of Credit shall be satisfactory to such Issuer and that an Authorized Officer of the Borrower shall have executed and delivered such application agreement and/or such other instruments and agreements relating to such Letter of Credit as such Issuer shall have reasonably requested (each a "Letter of Credit Application"). Promptly after its delivery of any Letter of Credit or any amendment to a Letter of Credit to an advising bank with respect thereto or to the beneficiary thereof, the applicable Issuer will also deliver to the Borrower and the Administrative Agent a true and complete copy of such

33

Letter of Credit or amendment. In the event of any conflict between the terms of this Agreement and the terms of any Letter of Credit Application, the terms of this Agreement shall control.

(d) Letter of Credit Fees. Upon the issuance of each Letter of Credit and until termination, cancellation or expiration of such Letter of Credit, the Borrower agrees to pay to the Administrative Agent, for the account of the Lenders ratably in accordance with their respective Pro Rata Shares, with respect to such Letter of Credit, a letter of credit fee (the "Letter of Credit Fee") at a per annum rate equal to the Letter of Credit Fee Rate in effect from time to time multiplied by the daily maximum amount available under such Letter of Credit, such fee to be payable in arrears on each Payment Date, on the Facility Terminati on Date and, if applicable, thereafter on demand; provided, however, any Letter of Credit Fees otherwise payable for the account of a Defaulting Lender with respect to any Letter of Credit as to which such Defaulting Lender has not provided Cash Collateral satisfactory to the applicable

Issuer pursuant to this Section 2.19 shall be payable, to the maximum extent permitted by applicable Law, to the other Lenders in accordance with the upward adjustments in their respective Pro Rata Shares allocable to such Letter of Credit pursuant to Section 2.23(a)(iv), with the balance of such fee, if any, payable to the applicable Issuer for its own account. The Borrower shall also pay to each Issuer for its own account (x) a fronting fee of 0.20% per annum on the aggregate maximum stated amount for each Letter of Credit issued by such Issuer and currently outstanding, with such fee to be payable in arrears on the first Business Day following the end of each March, June, September and December, and (y) documentary and processing charges in connection with the issuance or Modification of and draws under Letters of Credit in accordance with such Issuer's standard schedule for such charges as in effect from time to time.

(e) Administration; Reimbursement by Lenders. Upon receipt from the beneficiary of any Letter of Credit of any demand for payment under such Letter of Credit, the applicable Issuer shall notify the Administrative Agent and the Borrower and the Administrative Agent shall promptly notify each Lender of the amount to be paid by such Issuer as a result of such demand and the proposed payment date (the "Letter of Credit Payment Date"). After honoring any demand for payment under any Letter of Credit, the applicable Issuer shall, upon the request of the Borrower, send to the Borrower the beneficiary statement or an acc urate copy of such beneficiary statement pursuant to which the applicable beneficiary requested payment. The responsibility of any Issuer to the Borrower and each Lender shall be only to determine that the documents delivered under each Letter of Credit issued by such Issuer in connection with a demand for payment are in conformity in all material respects with such Letter of Credit. Each Issuer shall endeavor to exercise the same care in its issuance and administration of Letters of Credit as it does with respect to letters of credit in which no participations are granted, it being understood that in the absence of any gross negligence or willful misconduct by such Issuer, each Lender shall be unconditionally and irrevocably obligated, without regard to the occurrence of any Default or any condition precedent whatsoever, to reimburse (and the Administrative Agent may apply Cash Collateral for this purpose) such Issuer on demand for (i) such Lender's Pro Rata Share of the amount of each payment made by such Issuer under each Letter of Credit to the extent such amount is not reimbursed by the Borrower pursuant to Section 2.19(f) below, plus, without limiting the other provisions of this Agreement, (ii) interest on the foregoing amount, for each day from the date of the applicable payment by such Issuer to the date on which such Issuer is reimbursed by such Lender for its Pro Rata Share thereof, at a rat

34

greater of the Federal Funds Rate and a rate determined by the applicable Issuer in accordance with banking industry rules on interbank compensation, plus (iii) any administrative, processing or similar fees customarily charged by the applicable Issuer in connection with the foregoing.

Reimbursement by Borrower. (i) The Borrower shall be irrevocably and unconditionally obligated to reimburse each Issuer through the Administrative Agent, for any amount to be paid by such Issuer upon any drawing under any Letter of Credit without presentment, demand, protest or other formalities of any kind, not later than (i) the applicable Letter of Credit Payment Date, if the Borrower shall have received such notice on or prior to 11:00 a.m. on such Letter of Credit Payment Date or (ii) on the Business Day immediately following the applicable Letter of Credit Payment Date, if the Borrower shall have received such notice after 11:00 a.m. (either of such dates, the "Honor Date"); provided that the Borrower shall not be precluded from asserting any claim for direct (but not consequential) damages suffered by the Borrower which the Borrower proves were caused by (i) the willful misconduct or gross negligence of such Issuer in determining whether a request presented under any Letter of Credit complied with the terms of such Letter of Credit or (ii) such Issuer's failure to pay under any Letter of Credit after the presentation to it of a request strictly complying with the terms and conditions of such Letter of Credit. The Administrative Agent will pay to each Lender ratably in accordance with its Pro Rata Share all amounts received by it from the Borrower for application in payment, in whole or in part, of the Reimbursement Obligation in respect of any Letter of Credit, but only to the extent such Lender made payment to the applicable Issuer in respect of such Letter of Credit pursuant to Section 2.19(e). If the Borrower fails to so reimburse any Issuer for any amounts paid by any Issuer by such applicable time, the Administrative Agent shall promptly notify each Lender of the Honor Date, the amount of the unreimbursed drawing (the "Unreimbursed Amount"), and the amount of such Lender's Pro Rata Share thereof. In such event, the Borrower shall be deemed to have requested a Floating Rate Advance to be disbursed on the Honor Date in an amount equal to the Unreimbursed Amount and, so long as no Unmatured Default or Default has occurred and is continuing, such disbursement shall be deemed to occur automatically without further act and without regard to the minimum and multiples specified in Section 2.4 for the principal amount of Floating Rate Advance but subject to the conditions set forth in Section 4.2 (other than delivery by the Borrower of a Borrowing Notice) and provided that, after giving effect to such Floating Rate Advance, the Aggregate Outstanding Credit Exposure shall not exceed the Aggregate Commitments. To the extent that any amounts paid by an Issuer are not reimbursed by the Borrower pursuant to the terms of this Section 2.19(f), all such amounts shall bear interest, payable on demand, for each day until paid at a rate per annum equal to the sum of two percent (2%) plus the rate applicable to Floating Rate Advances. Any notice given by any Issuer or the Administrative Agent pursuant to this Section 2.19(f) (i) may be given by telephone if immediately confirmed in writing; provided that the lack of such an immediate confirmation shall not affect the conclusiveness or binding effect of such notice. Until each Lender funds its Floating Rate Loan to reimburse the applicable Issuer for any amount drawn under any Letter of Credit, interest in respect of such Lender's Pro Rata Share of such amount shall be solely for the account of such Issuer. Each Lender shall upon any notice pursuant to this Section 2.19(f) make funds available (and the Administrative Agent may apply Cash Collateral provided for this purpose) for the account of the applicable Issuer at the Administrative Agent's Office in an amount equal to its Pro Rata Share of the Unreimbursed Amount not later than 1:00 p.m. on the Business Day specified in such notice by the

35

Administrative Agent. The Administrative Agent shall remit the funds so received to the applicable Issuer.

(ii) Each Lender's obligation to make Floating Rate Loans to reimburse an Issuer for amounts drawn under Letters of Credit, as contemplated by this Section 2.19(f), shall be absolute and unconditional and shall not be affected by any circumstance, including (A) any setoff, counterclaim, recoupment, defense or other right which such Lender may have against the applicable Issuer, the Borrower or any other Person for any reason whatsoever; (B) the occurrence or continuance of a Default, or (C) any other occurrence, event or condition, whether or not similar to any of the foregoing; provided, however, that each Lender's obligation to make Floating Rate Loans is subject to the conditions set forth in Section 4.2 (other than delivery by the Borrower of a Borrowing Notice).

If any payment received by the Administrative Agent for the account of any Issuer pursuant to <u>Section 2.19(f)</u> is required to be returned under any of the circumstances described in <u>Section 11.3</u> (including pursuant to any settlement entered into by an Issuer in its discretion), each Lender shall pay to the

Administrative Agent for the account of the applicable Issuer its Pro Rata Share thereof on demand of the Administrative Agent, plus interest thereon from the date of such demand to the date such amount is returned by such Lender, at a rate per annum equal to the Federal Funds Rate from time to time in effect. The obligations of the Lenders under this paragraph shall survive the payment in full of the Obligations and the termination of this Agreement.

Obligations Absolute. Except to the extent that any Tax is required to be withheld or deducted under applicable Law, but subject to the provisions of Article III, the Borrower's obligations under this Section 2.19 shall be absolute, unconditional and irrevocable under any and all circumstances and irrespective of any setoff, counterclaim or defense to payment which the Borrower or any Subsidiary may have or have had against any Issuer, any Lender or any beneficiary or transferee of a Letter of Credit. The Borrower further agrees with the Issuers and the Lenders that neither any Issuer nor any Lender shall be responsible for, and the Borrower's Reimbursement Obligation in respect of any Letter of Credit shall not be affected by, among other things, (i) the validity, enforceability or genuineness of documents or of any endorsements thereon, even if such documents should in fact prove to be in any or all respects invalid, fraudulent or forged or any statement therein untrue or inaccurate, (ii) any dispute between or among the Borrower, any of its Affiliates, the beneficiary of any Letter of Credit or any financing institution or other party to whom any Letter of Credit may be transferred or any claims or defenses whatsoever of the Borrower or of any of its Affiliates against the beneficiary of any Letter of Credit or any such transferee, (iii) any payment by an Issuer under such Letter of Credit against presentation of a draft or certificate that does not strictly comply with the terms of such Letter of Credit, (iv) any payment made by an Issuer under such Letter of Credit to any Person purporting to be a trustee in bankruptcy, debtor-in-possession, assignee for the benefit of creditors, liquidator, receiver or other representative of or successor to any beneficiary or any transferee of such Letter of Credit, including any arising in connection with any proceeding under any Debtor Relief Law or (v) any other circumstance or happening whatsoever, whether or not similar to any of the foregoing, including any other c

36

shall be liable for any error, omission, interruption or delay in transmission, dispatch or delivery of any message or advice, however transmitted, in connection with any Letter of Credit other than, with respect to any Issuer, any such error, omission, interruption or delay in transmission, dispatch or delivery of any message or advice resulting from the gross negligence or willful misconduct of such Issuer. The Borrower agrees that any action taken or omitted by any Issuer or any Lender under or in connection with any Letter of Credit and the related drafts and documents, if done without gross negligence or willful misconduct, shall be binding upon the Borrower and shall not put any Issuer or any Lender under any liability to the Borrower. The Borrower shall promptly examine a copy of each Letter of Credit and each amendment thereto that is delivered to it and, in the event of any claim of noncompliance with the Borrower's instructions or other irregularity, the Borrower will immediately notify the applicable Issuer. The Borrower shall be conclusively deemed to have waived any such claim against the applicable Issuer and its correspondents unless such notice is given as aforesaid. Nothing in this Section 2.19(g) is intended to limit the right of the Borrower to make a claim against any Issuer for damages as contemplated by the proviso to the first sentence of Section 2.19(f).

Actions of the Issuers. Each Issuer shall be entitled to rely, and shall be fully protected in relying, upon any Letter of Credit, draft, writing, resolution, notice, consent, certificate, affidavit, letter, cablegram, telegram, facsimile, telex or teletype message, statement, order or other document believed by it to be genuine and correct and to have been signed, sent or made by the proper Person or Persons, and upon advice and statements of legal counsel, independent accountants and other experts selected by such Issuer. Each Lender and the Borrower agree that, in paying any drawing under a Letter of Credit, the Issuers shall not have any responsibility to obtain any document (other than any sight draft, certificates and documents expressly required by such Letter of Credit) or to ascertain or inquire as to the validity or accuracy of any such document or the authority of the Person executing or delivering any such document. Each Issuer shall be fully justified in failing or refusing to take any action under this Agreement unless it shall first have received such advice or concurrence of the Required Lenders as it reasonably deems appropriate or it shall first be indemnified to its reasonable satisfaction by the Lenders against any and all liability and expense which may be incurred by it by reason of taking or continuing to take any such action. Notwithstanding any other provision of this Section 2.19, each Issuer shall in all cases be fully protected in acting, or in refraining from acting, under this Agreement in accordance with a request of the Required Lenders, and such request and any action taken or failure to act pursuant thereto shall be binding upon the Lenders and any future holder of a participation in any Letter of Credit issued by such Issuer. The Borrower hereby assumes all risks of the acts or omissions of any beneficiary or transferee with respect to its use of any Letter of Credit; provided, however, that this assumption is not intended to, and shall not, preclude the Borrower's pursuing such rights and remedies as it may have against the beneficiary or transferee at law or under any other agreement. None of the Issuers, the Administrative Agent, any of their respective Related Parties nor any correspondent, participant or assignee of any of the Issuers shall be liable or responsible for any of the matters described in clauses (i) through (v) of Section 2.19(g) except, with respect to any Issuer, for any such matters that are determined by a court of competent jurisdiction by final and non-appealable judgment to have resulted from the gross negligence or willful misconduct of such Person. In furtherance and not in limitation of the foregoing, the Issuers may accept documents that appear on their face to be in order, without responsibility for further investigation, regardless of any

37

notice or information to the contrary, and the Issuers shall not be responsible for the validity or sufficiency of any instrument transferring or assigning or purporting to transfer or assign a Letter of Credit or the rights or benefits thereunder or proceeds thereof, in whole or in part, which may prove to be invalid or ineffective for any reason.

(i) <u>Indemnification</u>. The Borrower agrees to indemnify and hold harmless each Lender, each Issuer and the Administrative Agent, and their respective directors, officers, agents and employees, from and against any and all claims and damages, losses, liabilities, costs or expenses which such Person may incur (or which may be claimed against such Person by any other Person whatsoever) by reason of or in connection with the issuance, execution and delivery or transfer of or payment or failure to pay under any Letter of Credit or any actual or proposed use of any Letter of Credit, including any claims, damages, losses, liabilities, costs or expenses which any Issuer may incur by reason of or in connection with (i) the failure of any other Lender to fulfill or comply with its obligations to such Issuer hereunder (but nothing herein contained shall affect any right the Borrower may have against any defaulting Lender) or (ii) by reason of or on account of such Issuer issuing any Letter of Credit which specifies that the term "Beneficiary" therein includes any successor by operation of law of the named Beneficiary, but which Letter of Credit does not require that any drawing by any such successor Beneficiary be accompanied by a copy of a legal document, satisfactory to such Issuer, evidencing the appointment of such successor Beneficiary; <u>provided</u> that the Borrower shall not be required to indemnify any Person for any claims, damages, losses, liabilities, costs or expenses to the extent, but only to the extent,

caused by (x) the willful misconduct or gross negligence of any Issuer in determining whether a request presented under any Let ter of Credit issued by such Issuer complied with the terms of such Letter of Credit or (y) any Issuer's failure to pay under any Letter of Credit issued by it after the presentation to it of a request strictly complying with the terms and conditions of such Letter of Credit. Nothing in this Section 2.19(i) is intended to limit the obligations of the Borrower under any other provision of this Agreement.

- (j) The Issuers' Obligation to Issue Letters of Credit. No Issuer shall be under any obligation to issue any Letter of Credit if:
- (i) any order, judgment or decree of any Governmental Authority or arbitrator shall by its terms purport to enjoin or restrain such Issuer from issuing such Letter of Credit, or any law applicable to such Issuer or any request or directive (whether or not having the force of law) from any Governmental Authority with jurisdiction over such Issuer shall prohibit, or request that such Issuer refrain from, the issuance of letters of credit generally or such Letter of Credit in particular or shall impose upon such Issuer with respect to such Letter of Credit any restriction, reserve or capital requirement (for which such Issuer is not otherwise compensated hereunder) not in effect on the Closing Date, or shall impose upon such Issuer any unreimbursed loss, cost or expense which was not applicable on the Closing Date and which such Issuer in good faith reasonably deems material to it; provided, however, that in the event a Lender participating in the Letters of Credit is not affected by any such restriction, requirement or imposition, and is able to issue such Letter of Credit and expressly agrees in its sole discretion to issue such Letter of Credit, such Lender, subject to the consent of the Administrative Agent, such consent not to be unreasonably withheld, conditioned or delayed, shall be added as an Issuer

38

pursuant to $\underline{\text{Section 2.19(n)}}$ and shall issue such Letter of Credit and shall be deemed the Issuer with regard to such Letter of Credit for all purposes of this Agreement;

- (ii) the issuance of such Letter of Credit would violate one or more policies of such Issuer applicable to letters of credit generally;
- (iii) except as otherwise agreed by the Administrative Agent and the applicable Issuer, such Letter of Credit is in an initial stated amount less than \$250,000;
 - (iv) such Letter of Credit is to be denominated in a currency other than Dollars; or
- (v) any Lender is at that time a Defaulting Lender, unless each Issuer has entered into arrangements, including the delivery of Cash Collateral, satisfactory to the Issuers (in their sole discretion) with the Borrower or such Lender to eliminate the Issuers' actual or potential Fronting Exposure (after giving effect to Section 2.23(a)(iv)) with respect to the Defaulting Lender arising from either the Letter of Credit then proposed to be issued or that Letter of Credit and all other Letter of Credit Obligations as to which the Issuers have actual or potential Fronting Exposure, as it may elect in their sole discretion.
- (k) <u>Rights as a Lender</u>. In its capacity as a Lender, each Issuer shall have the same rights and obligations as any other Lender.
- (l) <u>Letters of Credit Issued for Subsidiaries</u>. Notwithstanding that a Letter of Credit issued or outstanding hereunder is in support of any obligations of, or is for the account of, a Subsidiary, the Borrower shall be obligated to reimburse the Issuers hereunder for any and all drawings under such Letter of Credit. The Borrower hereby acknowledges that the issuance of Letters of Credit for the account of Subsidiaries inures to the benefit of the Borrower, and that the Borrower's business derives substantial benefits from the businesses of such Subsidiaries.
- (m) <u>Applicability of ISP</u>. Unless otherwise expressly agreed by each Issuer and the Borrower when a Letter of Credit is issued, the rules of the ISP shall apply to each Letter of Credit.
- (n) Replacement or Addition of Issuer. Any Issuer may be replaced or added at any time by written agreement among the Borrower, the Administrative Agent (unless, in the case of the replacement of any Issuer, the successor Issuer is a Lender and, if applicable, such agreement not to be unreasonably withheld, conditioned or delayed) and the successor or additional Issuer, as applicable; provided that such new Issuer shall be assigned all or a portion, as applicable, of the Letter of Credit Commitment of another Issuer. The Administrative Agent shall notify the Lenders of any such replacement or addition, as applicable, of any Issuer. Where any Issuer is replaced, at the time such replacement shall become effective, the Borrower shall pay all unpaid fees accrued for account of the replaced Issuer. Furthermore, from and after the effective date of such replacement, the successor Issuer, shall have all the rights and obligations of the replaced Issuer under this Agreement with respect to Letters of Credit to be issued thereafter. References

39

herein to the term "Issuer" shall be deemed to refer to any successor or additional Issuer, as applicable, or to any previous Issuer, or to any successor or additional Issuer, as applicable, and all previous Issuers, as the context shall require. After the replacement of any Issuer hereunder, the replaced Issuer shall remain a party hereto and shall continue to have all the rights and obligations of an Issuer under this Agreement with respect to Letters of Credit issued by it prior to such replacement, but shall not be required to issue additional Letters of Credit.

2.20 Swing Line Loans.

(a) <u>Swing Line Facility</u>. From and including the date of this Agreement and prior to the Facility Termination Date, subject to the terms and conditions set forth in this Agreement, the Swing Line Lender, in reliance upon the agreements of the other Lenders set forth in this <u>Section 2.20</u>, shall, unless

- (i) any Lender at such time is a Defaulting Lender and (ii) the Swing Line Lender has not entered into arrangements satisfactory to it with the Borrower or such Defaulting Lender to eliminate the Swing Line Lender's risk with respect to such Defaulting Lender, in which case the Swing Line Lender may in its discretion, make loans (eac h such loan, a "Swing Line Loan") to the Borrower in Dollars from time to time in an aggregate amount not to exceed at any time outstanding the amount of the Swing Line Sublimit; provided, however, that after giving effect to any Swing Line Loan, (i) the Aggregate Outstanding Credit Exposure shall not exceed the Aggregate Commitments, and (ii) the Outstanding Credit Exposure of any Lender (other than the Swing Line Lender) shall not exceed such Lender's Commitment, and provided, further, that the Borrower shall not use the proceeds of any Swing Line Loan to refinance any outstanding Swing Line Loan. Within the foregoing limits, and subject to the other terms and conditions hereof, the Borrower may borrow Swing Line Loans under this Section 2.20, prepay Swing Line Loans under Section 2.7, and reborrow Swing Line Loans under this Section 2.20. Each Swing Line Loan shall bear interest at the Alternate Base Rate plus the Applicable Margin. Immediately upon the making of a Swing Line Loan, each Lender shall be deemed to, and hereby irrevocably and unconditionally agrees to, purchase from the Swing Line Lender a risk participation in such Swing Line Loan in an amount equal to such Lender's Pro Rata Share times the amount of such Swing Line Loan.
- (b) Borrowing Procedures. Each Swing Line Loan shall be made upon the Borrower's irrevocable notice to the Swing Line Lender and the Administrative Agent, which may be given by telephone. Each such notice must be received by the Swing Line Lender and the Administrative Agent not later than 1:00 p.m. on the requested borrowing date, and shall specify (i) the amount to be borrowed, which shall be a minimum principal amount of \$500,000 and integral multiples of \$100,000 in excess thereof, and (ii) the requested borrowing date, which shall be a Business Day. Each such telephonic notice must be confirmed promptly by delivery to the Swing Line Lender and the Administrative Agent of a written notice (the "Swing Line Loan Notice"). Promptly after receipt by the Swing Line Lender of any telephonic Swing Line Loan Notice, the Swing Line Lender will confirm with the Administrative Agent (by telephone or in writing) that the Administrative Agent has also received such Swing Line Loan Notice and, if not, the Swing Line Lender will notify the Administrative Agent (by telephone or in writing) of the contents thereof. Unless the Swing Line Lender has received notice (by telephone or in writing) from the Administrative Agent (including at the request of any Lender) prior to 2:00 p.m. on the date of the proposed Advance of Swing Line Loans (A) directing the

40

Swing Line Lender not to make such Swing Line Loan as a result of the limitations set forth in the first proviso to the first sentence of Section 2.20(a), or (B) that one or more of the applicable conditions specified in Article IV is not then satisfied, then, subject to the terms and conditions hereof, the Swing Line Lender will, not later than 3:00 p.m. on the borrowing date specified in such Swing Line Loan Notice, make the amount of its Swing Line Loan available to the Borrower by means of a credit to the general deposit account of the Borrower with the Swing Line Lender or such other account specified by the Borrower to the Swing Line Lender.

(c) <u>Refinancing of Swing Line Loans</u>.

- (i) The Swing Line Lender at any time in its sole and absolute discretion may request, on behalf of the Borrower (which hereby irrevocably requests and authorizes the Swing Line Lender to so request on its behalf), that each Lender make a Floating Rate Loan in an amount equal to such Lender's Pro Rata Share of the amount of Swing Line Loans then outstanding. Such request shall be made in writing (which written request shall be deemed to be a Borrowing Notice for purposes hereof) and in accordance with the requirements of Section 2.8 without regard to the minimum and multiples specified in Section 2.4 and provided that, after giving effect to such Advance, the Aggregate Outstanding Credit Exposure shall not exceed the Aggregate Commitments. The Swing Line Lender shall furnish the Borrower with a copy of the applicable Loan notice promptly after delivering such notice to the Administrative Agent. Each Lender shall make an amount equal to its Pro Rata Share of the amount specified in such Loan notice available to the Administrative Agent in immediately available funds (and the Administrative Agent may apply Cash Collateral available with respect to the applicable Swing Line Loan) for the account of the Swing Line Lender at the Administrative Agent's Office not later than 1:00 p.m. on the day specified in such Loan notice, whereupon, subject to Section 2.20(c)(ii), each Lender that so makes funds available shall be deemed to have made a Floating Rate Loan to the Borrower in such a mount. The Administrative Agent shall remit the funds so received to the Swing Line Lender.
- (ii) If for any reason any Swing Line Loan cannot be refinanced by such a Advance of Loans in accordance with $\underline{Section\ 2.20(c)(i)}$, the request for Floating Rate Loans submitted by the Swing Line Lender as set forth herein shall be deemed to be a request by the Swing Line Lender that each of the Lenders fund its risk participation in the relevant Swing Line Loan and each Lender's payment to the Administrative Agent for the account of the Swing Line Lender pursuant to $\underline{Section\ 2.20(c)(i)}$ shall be deemed payment in respect of such participation.
- (iii) If any Lender fails to make available to the Administrative Agent for the account of the Swing Line Lender any amount required to be paid by such Lender pursuant to the foregoing provisions of this Section 2.20(c) by the time specified in Section 2.20(c)(i), the Swing Line Lender shall be entitled to recover from such Lender (acting through the Administrative Agent), on demand, such amount with interest thereon for the period from the date such payment is required to the date on which such payment is immediately available to the Swing Line Lender at a rate per annum equal to the greater of the Federal Funds Rate and a rate determined by the Swing Line Lender in

41

accordance with banking industry rules on interbank compensation. A certificate of the Swing Line Lender submitted to any Lender (through the Administrative Agent) with respect to any amounts owing under this <u>clause (iii)</u> shall be conclusive absent manifest error.

(iv) Each Lender's obligation to make Loans or to purchase and fund risk participations in Swing Line Loans pursuant to this Section 2.20(c) shall be absolute and unconditional and shall not be affected by any circumstance, including (A) any setoff, counterclaim, recoupment, defense or other right that such Lender may have against the Swing Line Lender, the Borrower or any other Person for any reason whatsoever, (B) the occurrence or continuance of a Default or Unmatured Default, or (C) any other occurrence, event or condition, whether or not similar to any of the foregoing; provided, however, that each Lender's obligation to make Loans pursuant to this Section 2.20(c) is subject to the conditions set forth in Section 4.2. No such purchase or funding of risk participations shall relieve or otherwise impair the obligation of the Borrower to repay Swing Line Loans, together with interest as provided herein.

(d) Repayment of Participations.

- (i) At any time after any Lender has purchased and funded a risk participation in a Swing Line Loan, if the Swing Line Lender receives any payment on account of such Swing Line Loan, the Swing Line Lender will distribute to such Lender its Pro Rata Share of such payment (appropriately adjusted, in the case of interest payments, to reflect the period of time during which such Lender's risk participation was funded) in the same funds as those received by the Swing Line Lender.
- (ii) If any payment received by the Swing Line Lender in respect of principal or interest on any Swing Line Loan is required to be returned by the Swing Line Lender under any of the circumstances described in Section 11.3 (including pursuant to any settlement entered into by the Swing Line Lender in its discretion), each Lender shall pay to the Swing Line Lender its Pro Rata Share thereof on demand of the Administrative Agent, plus interest thereon from the date of such demand to the date such amount is returned, at a rate per annum equal to the Federal Funds Rate. The Administrative Agent will make such demand upon the request of the Swing Line Lender. The obligations of the Lenders und er this clause shall survive the payment in full of the Obligations and the termination of this Agreement.
- (e) <u>Interest for Account of Swing Line Lender</u>. The Swing Line Lender shall be responsible for invoicing the Borrower for interest on the Swing Line Loans. Until each Lender funds its Floating Rate Loans or risk participation pursuant to this <u>Section 2.20</u> to refinance such Lender's Pro Rata Share of any Swing Line Loan, interest in respect of such Applicable Percentage shall be solely for the account of the Swing Line Lender.
- (f) <u>Payments Directly to Swing Line Lender</u>. The Borrower shall make all payments of principal and interest in respect of the Swing Line Loans directly to the Swing Line Lender.

42

2.21 <u>Extension of Facility Termination Date</u>.

- (a) Request for Extension. The Borrower may by written notice to the Administrative Agent (who shall promptly notify the Lenders) given not more than sixty (60) days and not less than forty-five (45) days prior to any anniversary of the Closing Date, request that each Lender extend the Facility Termination Date for an additional one (1) year from the then existing Facility Termination Date; provided, that the Borrower shall only be permitted to exercise this extension option two (2) times during the term of the Agreement.
- (b) <u>Lenders Election to Extend</u>. Each Lender, acting in its sole and individual discretion, shall, by notice to the Administrative Agent given not later than fifteen (15) days following the receipt of notice of such request from the Administrative Agent (the "Notice Date"), advise the Administrative Agent in writing whether or not such Lender agrees to such extension (and each Lender that determines not to so extend its Facility Termination Date (a "Non-Extending Lender") shall notify the Administrative Agent of such fact promptly after such determination (but in any event no later than the Notice Date) and any Lender that does not so advise the Administrative Agent on or before the Notice Date shall be deemed to be a Non-Extending Lender. The election of any Lender to agree to such extension shall not obligate any other Lender to so agree.
- (c) <u>Notification by Administrative Agent</u>. The Administrative Agent shall notify the Borrower of each Lender's determination under this Section promptly and in any event no later than the date fifteen (15) days after the Notice Date (or, if such date is not a Business Day, on the next preceding Business Day).
- (d) Additional Commitment Lenders. The Borrower shall have the right on or before the applicable anniversary of the Closing Date to replace each Non-Extending Lender with, and add as "Lenders" under this Agreement in place thereof, one or more Eligible Assignees (each, an "Additional Commitment Lender") as provided in Section 12.2, each of which Additional Commitment Lenders shall have entered into an Assignment Agreement pursuant to which such Additional Commitment Lender shall, undertake a Commitment (and, if any such Additional Commitment Lender is already a Lender, its Commitment shall be in addition to such Lender's Commitment hereunder on such date) and shall be a "Lender" for all purposes of this Agreement.
- (e) <u>Minimum Extension Requirement.</u> If all of the Lenders agree to any such request for extension of the Facility Termination Date then the Facility Termination for all Lenders shall be extended for the additional one (1) year, as applicable. If there exists any Non-Extending Lenders then the Borrower shall (i) withdraw its extension request and the Facility Termination Date will remain unchanged or (ii) provided that the Required Lenders (but for the avoidance of doubt, not including any Additional Commitment Lenders) have agreed to the extension request (such Lenders agreeing to such extension, the "<u>Approving Lenders</u>& #8221;) no later than fifteen (15) days prior to such anniversary of the Closing Date, then the Borrower may extend the Facility Termination Date solely as to the Approving Lenders and the Additional Commitment Lenders with a reduced amount of Aggregate Commitments during such extension period equal to the aggregate Commitments of the Approving Lenders and the Additional Commitment Lenders so long as the aggregate amount of the Commitments of the Approving Lenders and the Additional Commitment Lenders equals or

43

exceeds the Letter of Credit Sublimit (as it may be amended as of the extension date with the consent of such Approving Lenders and Additional Commitment Lenders); it being understood that (A) the Facility Termination Date relating to any Non-Extending Lenders not replaced by an Additional Commitment Lender shall not be extended and the repayment of all obligations owed to them and the termination of their Commitments shall occur on the already existing Facility Termination Date and (B) the Facility Termination Date relating to the Approving Lenders and the Additional Commitment Lenders shall be extended for an additional year, as applicable.

- (f) <u>Conditions to Effectiveness of Extensions</u>. Notwithstanding the foregoing, any extension of the Facility Termination Date pursuant to this Section shall not be effective with respect to any Lender unless:
 - (i) on the date of such extension the conditions for a Credit Extension provided in $\underline{Section 4.2(a)}$ and $\underline{Section 4.2(b)}$ shall be satisfied; and
 - (ii) on any Facility Termination Date, the Borrower shall prepay any Loans outstanding on such date (and pay any additional amounts required pursuant to Section 3.4) to the extent necessary to keep outstanding Loans ratable with any revised Pro Rata Shares of the respective Lenders effective as of such date.

2.22 <u>Cash Collateral</u>.

- (a) <u>Certain Credit Support Events</u>. Upon the request of the Administrative Agent or any of the Issuers (i) if any Issuer has honored any full or partial drawing request under any Letter of Credit and such drawing has not been reimbursed by the Borrower on the Letter of Credit Payment Date or (ii) if, as of the Facility Termination Date, any Letter of Credit Obligation for any reason remains outstanding and partially or wholly drawn, the Borrower shall, in each case, immediately Cash Collateralize the then outstanding amount of all Letter of Credit Obligations. At any time that there shall exist a Defaulting Lender, immediately upon the request of the Administrative Agent, the Issuers or the Swing Line Lender, the Borrower shall deliver to the Administrative Agent Cash Collateral in an amount sufficient to cover all Fronting Exposure, if any (after giving effect to Section 2.23(a)(iv) and any Cash Collateral provided by the Defaulting Lender).
- (b) <u>Grant of Security Interest</u>. All Cash Collateral (other than credit support not constituting funds subject to deposit) shall be maintained in blocked, interest bearing deposit accounts at Bank of America (the "<u>LC Collateral Accounts</u>"). The Borrower, and to the extent provided by any Lender, such Lender, hereby grants to (and subjects to the control of) the Administrative Agent, for the benefit of the Administrative Agent, the Issuers and the Lenders (including the Swing Line Lender), and agrees to maintain, a first priority security interest in all such cash, dep osit accounts and all balances therein, and all other property so provided as collateral pursuant hereto, and in all proceeds of the foregoing, all as security for the obligations to which such Cash Collateral may be applied pursuant to <u>Section 2.22(c)</u>. If at any time the Administrative Agent determines that Cash Collateral is subject to any right or claim of any Person other than the Administrative Agent as herein provided, or that the total amount of such Cash Collateral is less than the applicable Fronting Exposure and other obligations secured

44

thereby, the Borrower or the relevant Defaulting Lender will, promptly upon written demand by the Administrative Agent, pay or provide to the Administrative Agent additional Cash Collateral in an amount sufficient to eliminate such deficiency.

- (c) <u>Application</u>. Notwithstanding anything to the contrary contained in this Agreement, Cash Collateral provided under any of this <u>Section 2.22</u> or <u>Sections 2.19, 2.20, 2.23</u> or <u>8.1</u> in respect of Letters of Credit or Swing Line Loans shall be held and applied to the satisfaction of the specific Letter of Credit Obligations, Swing Line L oans, obligations to fund participations therein (including, as to Cash Collateral provided by a Defaulting Lender, any interest accrued on such obligation) and other obligations for which the Cash Collateral was so provided, prior to any other application of such property as may be provided for herein.
- (d) Release. Cash Collateral (or the appropriate portion thereof) provided to reduce Fronting Exposure or other obligations, together with all interest accrued thereon, shall be released promptly following (i) the elimination of the applicable Fronting Exposure or other obligations giving rise thereto (including by the termination of Defaulting Lender status of the applicable Lender (or, as appropriate, its assignee following compliance with Section 12.1(b) (v))) or (ii) the Administrative Agent's good faith determination that there exists excess Cash Collateral; provided, however, (x) that Cash Collateral furnished by or on behalf of the Borrower shall not be released during the continuance of an Unmatured Default or a Default, and (y) the Person providing Cash Collateral and the Issuers or Swing Line Lender, as applicable, may agree, in their sole discretion, that Cash Collateral shall not be released but instead held to support future anticipated Fronting Exposure or other obligations.

2.23 <u>Defaulting Lenders</u>.

- (a) <u>Adjustments</u>. Notwithstanding anything to the contrary contained in this Agreement, if any Lender becomes a Defaulting Lender, then, until such time as that Lender is no longer a Defaulting Lender, to the extent permitted by applicable Law:
 - (i) <u>Waivers and Amendments</u>. That Defaulting Lender's right to approve or disapprove any amendment, waiver or consent with respect to this Agreement shall be restricted as set forth in Section 8.2.
 - (ii) Reallocation of Payments. Any payment of principal, interest, fees or other amounts received by the Administrative Agent for the account of that Defaulting Lender under the Loan Documents (whether voluntary or mandatory, at maturity, pursuant to Article VIII or otherwise, and including any amounts made available to the Administrative Agent by that Defaulting Lender pursuant to Section 11.1), shall be applied at such time or times as may be reasonably determined by the Administrative Agent as follows: first, to the payment of any amounts owing by that Defaulting Lender to the Administrative Agent hereunder; second, to the payment on a pro rata basis of any amounts owing by that Defaulting Lender to the Issuers or Swing Line Lender hereunder; third, if so reasonably determined by the Administrative Agent or requested by the Issuers or Swing Line Lender, to be held as Cash Collateral for future funding obligations of that Defaulting Lender of any participation in any Swing Line Loan or Letter of

Credit; fourth, as the Borrower may request (so long as no Default exists), to the funding of any Loan in respect of which that Defaulting Lender has failed to fund its portion thereof as required by this Agreement, as determined by the Administrative Agent; fifth, if so reasonably determined by the Administrative Agent and the Borrower, to be held in a non-interest bearing deposit account and released in order to satisfy obligations of that Defaulting Lender to fund Loans under this Agreement; sixth, to the payment of any amounts owing to the Lenders, the Issuers or Swing Line Lender as a result of any judgment of a court of competent jurisdicti on obtained by any Lender, the Issuers or Swing Line Lender against that Defaulting Lender as a result of that Defaulting Lender's breach of its obligations under this Agreement; seventh, so long as no Default exists, to the payment of any amounts owing to the Borrower as a result of any judgment of a court of competent jurisdiction obtained by the Borrower against that Defaulting Lender as a result of that Defaulting Lender's breach of its obligations under this Agreement; and eighth, to that Defaulting Lender or as otherwise directed by a court of competent jurisdiction; provided that if (x) such payment is a payment of the principal amount of any Loans or a reimbursement of any payment on a Letter of Credit, in respect of which that Defaulting Lender has not fully funded its appropriate share and (y) s uch Loans or reimbursement of payment on a Letter of Credit were made at a time when the conditions set forth in Section 4.2 were satisfied or waived, such payment shall be applied solely to pay the Loans of, and such Letter of Credit reimbursements owed to, all non-Defaulting Lenders on a pro rata basis prior to being applied to the payment of any Loans of, or such Letter of Credit reimbursements owed to, that Defaulting Lender. Any payments, prepayments or other amounts paid or payable to a Defaulting Lender that are applied (or held) to pay amounts owed by a Defaulting Lender or to post Cash Collateral pursuant to this Section 2.23(a)(ii) shall be deemed paid to and redirected by that Defaulting Lender, and each Lender irrevocably consents hereto.

- (iii) <u>Certain Fees</u>. That Defaulting Lender (x) shall not be entitled to receive any commitment fee pursuant to <u>Section 2.5</u> for any period during which that Lender is a Defaulting Lender (and the Borrower shall not be required to pay any such fee that otherwise would have been required to have been paid to that Defaulting Lender) and (y) shall be limited in its right to receive Letter of Credit Fees as provided in <u>Section 2.19(d)</u>.
- (iv) Reallocation of Pro Rata Shares to Reduce Fronting Exposure. During any period in which there is a Defaulting Lender, for purposes of computing the amount of the obligation of each non-Defaulting Lender to acquire, refinance or fund participations in Letters of Credit or Swing Line Loans pursuant to Sections 2.19 and 2.20, the "Pro Rata Share" of each non-Defaulting Lender shall be computed without giving effect to the Commitment of that Defaulting Lender; provided, that, (i) each such reallocation shall be given effect only if, at the date the applicable Lender becomes a Defaulting Lender, no Default exists; and (ii) the aggregate obligation of each non-Defaulting Lender to acquire, refinance or fund participations in Letters of Credit and Swing Line Loans shall not exceed the positive difference, if any, of (1) the Commitment of that non-Defaulting Lender minus (2) the Outstanding Credit Exposure of that Lender.

46

(b) <u>Defaulting Lender Cure</u>. If the Borrower, the Administrative Agent, the Swing Line Lender and the Issuers agree in writing in their reasonable discretion that a Defaulting Lender should no longer be deemed to be a Defaulting Lender, the Administrative Agent will so notify the parties hereto, whereupon as of the effective date specified in such notice and subject to any conditions set forth therein (which may include arrangements with respect to any Cash Collateral), that Lender will, to the extent applicable, purchase that portion of outstanding Loans of the other Lenders or take such other actions as the Administrative Agent may determine to be nec essary to cause the Loans and funded and unfunded participations in Letters of Credit and Swing Line Loans to be held on a pro rata basis by the Lenders in accordance with their Pro Rata Shares (without giving effect to Section 2.23(a)(iv)), whereupon that Lender will cease to be a Defaulting Lender; provided that no adjustments will be made retroactively with respect to fees accrued or payments made by or on behalf of the Borrower while that Lender was a Defaulting Lender; and provided, further, that except to the extent otherwise expressly agreed by the affected parties, no change hereunder from Defaulting Lender to Lender will constitute a waiver or release of any claim of any party hereunder arising from that Lender's having been a Defaulting Lender.

ARTICLE III

YIELD PROTECTION; TAXES

3.1 <u>Yield Protection</u>.

If, on or after the date of this Agreement, the adoption of any law or any governmental or quasi-governmental rule, regulation, policy, guideline or directive (whether or not having the force of law), or any change which occurs after the date of this Agreement in the interpretation or administration thereof by any governmental or quasi-governmental authority, central bank or comparable agency charged with the interpretation or administration thereof, or compliance by any Lender, any applicable Lending Installation or any Issuer with any request or directive (whether or not having the force of law) of any such authority, central bank or comparable agency imposed after the date of this Agreement:

- (i) subjects any Lender to any tax of any kind whatsoever with respect to this Agreement, any Letter of Credit, any participation in a Letter of Credit or any Eurodollar Loan made by it, or change the basis of taxation of payments to such Lender in respect thereof (except for Indemnified Taxes or Other Taxes covered by Section 3.5 and the imposition of, or any change in the rate of, any Excluded Tax), or
- (ii) imposes or increases or deems applicable any reserve, assessment, insurance charge, special deposit or similar requirement against assets of, deposits with or for the account of, or credit extended by, any Lender, any applicable Lending Installation or any Issuer (other than reserves and assessments taken into account in determining the interest rate applicable to Eurodollar Advances), or
- (iii) imposes any other condition the result of which is to increase the cost to any Lender, any applicable Lending Installation or any Issuer of making, funding or

maintaining its Eurodollar Loans or of issuing or participating in Letters of Credit or reduces any amount receivable by any Lender, any applicable Lending Installation or any Issuer in connection with its Eurodollar Loans or Letters of Credit, or requires any Lender, any applicable Lending Installation or any Issuer to make any payment calculated by reference to the amount of Eurodollar Loans or Letters of Credit held or interest received by it, by an amount deemed material by such Lender or such Issuer, as the case may be,

and the result of any of the foregoing is to increase the cost to such Lender, the applicable Lending Installation or such Issuer of making or maintaining its Eurodollar Loans, Letters of Credit or Commitment or to reduce the return received by such Lender, the applicable Lending Installation or such Issuer in connection with such Eurodollar Loans, Letters of Credit or Commitment, then, within fifteen (15) Business Days of written demand by such Lender or such Issuer, the Borrower shall pay such Lender or such Issuer such additional amount or amounts as will compensate such Lender or such Issuer for such increased cost or reduction in amount received.

3.2 Changes in Capital Adequacy Regulations.

If a Lender or an Issuer determines the amount of capital required or expected to be maintained by such Lender, any Lending Installation of such Lender, such Issuer or any corporation controlling such Lender or such Issuer is increased as a result of a Change, then, within thirty (30) days of written demand by such Lender or such Issuer, the Borrower shall pay such Lender or such Issuer the amount necessary to compensate for any shortfall in the rate of return on the portion of such increased capital which such Lender or such Issuer determines is attributable to this Agreement, its Outstanding Credit Exposure or its Commitment to make Loans or to issue or participate in Letters of Credit hereunder (after taking into account such Lender's policies as to capital adequacy). "Change" means (i) any change after the date of this Agreement in (or in the interpretation of) the Risk-Based Capital Guidelines or (ii) any adoption of or change in (or any change in the interpretation of) any other law, governmental or quasi-governmental rule, regulation, policy, guideline, interpretation, or directive (whether or not having the force of law) after the date of this Agreement which affects the amount of capital required or expected to be maintained by any Lender, any Lending Installation, any Issuer or any corporation controlling any Lender or any Issuer. "Risk-Based Capital Guidelines" means (x) the risk-based capital guidelines in effect in the United States on the date of this Agreement, including transition rules, and (y) the corresponding capital regulations promulgated by regulatory authorities outside the United States implementing the July 1988 report of the Basle Committee on Banking Regulation and Supervisory Practices Entitled "International Convergence of Capital Measurements and Capital Standards," including transition rules, and any amendments to such regulations adopted prior to the date of this Agreement.

3.3 **Availability of Types of Advances.**

If (i) any Lender determines that maintenance of its Eurodollar Loans at a suitable Lending Installation would violate any applicable law, rule, regulation, or directive, whether or not having the force of law, enacted or imposed after the date of this Agreement, (ii) the

48

Required Lenders determine that (a) deposits of a type and maturity appropriate to match fund Eurodollar Advances are not available or (b) the interest rate applicable to a Type of Advance does not accurately reflect the cost of making or maintaining such Advance or (iii) the Administrative Agent determines that adequate and reasonable means do not exist for determining the Eurodollar Base Rate, then the Administrative Agent shall suspend the availability of the affected Type of Advance and, in the case of clause(i), require any affected Eurodollar Advances to be repaid or converted to Floating Rate Advances, subject to the payment of any funding indemnification amounts required by Section 3.4.

3.4 <u>Funding Indemnification</u>.

If any conversion, prepayment or payment of a Eurodollar Advance occurs on a date which is not the last day of the applicable Interest Period, whether because of acceleration, prepayment or otherwise, or a Eurodollar Advance is not made, paid, continued or converted on the date or in the amount specified by the Borrower for any reason other than default by the Lenders, the Borrower will, within fifteen (15) Business Days of written demand by any Lender, indemnify such Lender for any loss or cost incurred by it resulting therefrom, including any loss or cost in liquidating or employing deposits acquired to fund or maintain such Eurodollar Advance; <u>provided</u> that each such Lender shall have delivered to the Borrower a certificate as to the amount of such loss or cost, setting forth in reasonable detail the calculation thereof, which certificate shall be conclusive in the absence of manifest error.

3.5 <u>Taxes</u>.

- (a) Payments Free of Taxes; Obligation to Withhold; Payments on Account of Taxes. (i) Any and all payments by or on account of any Obligation of the Borrower hereunder or under any other Loan Document shall to the extent permitted by applicable Laws be made free and clear of and without reduction or withholding for any Taxes. If, however, applicable Laws require the Borrower or the Administrative Agent to withhold or deduct any Tax, such Tax shall be withheld or deducted in accordance with such Laws as determined by the Borrower or the Administrative Agent, as the case may be, upon the basis of the information and documentation to be delivered pursuant to subsection (e) below.
- (ii) If the Borrower or the Administrative Agent shall be required by the Code to withhold or deduct any Taxes, including both United States Federal backup withholding and withholding taxes, from any payment, then (A) the Borrower or the Administrative Agent, as the case may be, shall withhold or make such deductions as are determined by the Borrower or the Administrative Agent, as the case may be, to be required based upon the information and documentation it has received pursuant to subsection (e) below, (B) the Borrower or the Administrative Agent, as the case may be, shall timely pay the full amount withheld or deducted to the relevant Governmental Authority in accordance with the Code, and (C) to the exte nt that the withholding or deduction is made on account of Indemnified Taxes or Other Taxes, the sum payable by the Borrower shall be increased as necessary so that after any required withholding or the making of all required deductions (including deductions applicable to additional sums payable under this Section) the Administrative Agent or Lender, as the case may be, receives an amount equal to the sum it would have received had no such withholding or deduction of Indemnified Taxes or Other Taxes been made.

- (b) <u>Payment of Other Taxes by the Borrower</u>. Without limiting the provisions of <u>subsection (a)</u> above, the Borrower shall timely pay any Other Taxes to the relevant Governmental Authority in accordance with applicable Laws.
- (c) Tax Indemnifications. (i) Without limiting the provisions of subsection (a) or (b) above, the Borrower shall, and does hereby, indemnify the Administrative Agent and each Lender, and shall make payment in respect thereof within ten (10) days after demand therefor, for the full amount of any Indemnified Taxes or Other Taxes (including Indemnified Taxes or Other Taxes imposed or asserted on or attributable to amounts payable under this Section but excluding Indemnified Taxes or Other Taxes paid by the Borrower to the Administrative Agent pursuant to Section 3.5(a)(ii)(C)) withheld or deducted by the Borrower or the Administrative Agent or paid by the Administrative Agent or such Lender, as the case may be, and any penalties, interest and reasonable expenses arising therefrom or with respect thereto, whether or not such Indemnified Taxes or Other Taxes were correctly or legally imposed or asserted by the relevant Governmental Authority. The Borrower shall also, and does hereby, indemnify the Administrative Agent, and shall make payment in respect thereof within ten (10) days after demand therefor, for any amount which a Lender for any reason fails to pay indefeasibly to the Administrative Agent as required by clause (ii) of this subsection. A certificate as to the amount of any such payment or liability delivered to the Borrower by a Lender (with a copy to the Administrative Agent), or by the Administrative Agent on its own behalf or on behalf of a Lender, shall be conclusive absent manifest error.
 - (ii) Without limiting the provisions of <u>subsection (a)</u> or (b) above, each Lender shall, and does hereby, indemnify the Borrower and the Administrative Agent, and shall make payment in respect thereof within ten (10) days after demand therefor, against any and all Taxes and any and all related losses, claims, liabilities, penalties, interest and expenses (including the fees, charges and disbursements of any counsel for the Borrower or the Administrative Agent) incurred by or asserted against the Borrower or the Administrative Agent by any Governmental Authority as a result of the failure by such Lender to deliver, or as a result of the inaccuracy, inadequacy or deficiency of, any documentation required to be delivered by such Lender to the Borrower or the Administrative Agent pursuant to <u>subsection (e)</u>. Each Lender hereby authorizes the Administrative Agent to set off and apply any and all amounts at any time owing to such Lender under this Agreement or any other Loan Document against any amount due to the Administrative Agent under this <u>clause (ii)</u>. The agreements in this <u>clause (ii)</u> shall survive the resignation and/or replacement of the Administrative Agent, any assignment of rights by, or the replacement of, a Lender, the termination of the Aggregate Commitments and the repayment, satisfaction or discharge of all other Obligations.
- (d) <u>Evidence of Payments</u>. Upon request by the Borrower or the Administrative Agent, as the case may be, after any payment of Taxes by the Borrower or the Administrative Agent to a Governmental Authority as provided in this <u>Section 3.5</u>, the Borrower shall deliver to the Administrative Agent or the Administrative Agent shall deliver to the Borrower, as the case may be, the original or a certified copy of a receipt issued by such Governmental Authority evidencing such payment or other evidence of such payment reasonably satisfactory to the Borrower or the Administrative Agent, as the case may be.

50

- (e) <u>Status of Lenders; Tax Documentation.</u> (i) Each Lender shall deliver to the Borrower and to the Administrative Agent, at the time or times prescribed by applicable Laws or when reasonably requested by the Borrower or the Administrative Agent, such properly completed and executed documentation prescribed by applicable Laws or by the taxing authorities of any jurisdiction and such other reasonably requested information as will permit the Borrower or the Administrative Agent, as the case may be, to determine (A) whether or not payments made hereunder or under any other Loan Document are subject to Taxes, (B) if applicable, the required rate of withholding or deduction, and (C) such Lender's entitlement to any available exemption from, or reduction of, applicable Taxes in respect of all payments to be made to such Lender by the Borrower pursuant to this Agreement or otherwise to establish such Lender's status for withholding tax purposes in the applicable jurisdiction.
 - (ii) Without limiting the generality of the foregoing, if the Borrower is resident for tax purposes in the United States,
 - (A) any Lender that is a "United States person" within the meaning of Section 7701(a)(30) of the Code shall deliver to the Borrower and the Administrative Agent executed originals of Internal Revenue Service Form W-9 or such other documentation or information prescribed by applicable Laws or reasonably requested by the Borrower or the Administrative Agent as will enable the Borrower or the Administrative Agent, as the case may be, to determine whether or not such Lender is subject to backup withholding or information reporting requirements; and
 - (B) each Foreign Lender that is entitled under the Code or any applicable treaty to an exemption from or reduction of withholding tax with respect to payments hereunder or under any other Loan Document shall deliver to the Borrower and the Administrative Agent (in such number of copies as shall be requested by the recipient) on or prior to the date on which such Foreign Lender becomes a Lender under this Agreement (and from time to time thereafter upon the request of the Borrower or the Administrative Agent, but only if such Foreign Lender is legally entitled to do so), whichever of the following is applicable:
 - (I) duly completed and executed originals of Internal Revenue Service Form W-8BEN claiming eligibility for benefits of an income tax treaty to which the United States is a party,
 - (II) duly completed and executed originals of Internal Revenue Service Form W-8ECI,
 - (III) duly completed and executed originals of Internal Revenue Service Form W-8IMY and all required supporting documentation,
 - (IV) in the case of a Foreign Lender claiming the benefits of the exemption for portfolio interest under section 881(c) of the Code, (x) a certificate to the effect that such Foreign Lender is not (A) a "bank" within the meaning of section 881(c)(3)(A) of the Code, (B) a "10 percent shareholder" of the Borrower within the meaning of section 881(c)(3)(B) of the Code, or (C) a "controlled"

foreign corporation" described in section 881(c)(3)(C) of the Code and (y) duly completed and executed originals of Internal Revenue Service Form W-8BEN, or

- (V) duly completed and executed originals of any other form prescribed by applicable Laws as a basis for claiming exemption from or a reduction in United States Federal withholding tax together with such supplementary documentation as may be prescribed by applicable Laws to permit the Borrower or the Administrative Agent to determine the withholding or deduction required to be made.
- (iii) Each Lender shall promptly (A) notify the Borrower and the Administrative Agent of any change in circumstances which would modify or render invalid any claimed exemption or reduction, and (B) take such steps as shall not be materially disadvantageous to it, in the reasonable judgment of such Lender, and as may be reasonably necessary (including the re-designation of its lending office) to avoid any requirement of applicable Laws of any jurisdiction that the Borrower or the Administrative Agent make any withholding or deduction for Taxes from amounts payable to such Lender.
- (iv) Each Foreign Lender shall provide such documentation or information prescribed by applicable Laws or reasonably requested by the Borrower or the Administrative Agent as will enable the Borrower or the Administrative Agent, as the case may be, to determine whether or not such Foreign Lender is subject to withholding of tax under FATCA.
- (f) Treatment of Certain Refunds. Unless required by applicable Laws, at no time shall the Administrative Agent have any obligation to file for or otherwise pursue on behalf of a Lender, or have any obligation to pay to any Lender, any refund of Taxes withheld or deducted from funds paid for the account of such Lender. If the Administrative Agent or any Lender determines, in its sole discretion, that it has received a refund of any Indemnified Taxes or Other Taxes as to which it has been indemnified by the Borrower or with respect to which the Borrower has paid additional amounts pursuant to this Section, it shall pay to the Borrower an amount equal to such refund (but only to the extent of indemnity payments made, or additional amounts paid, by the Borrower under this Section with respect to the Indemnified Taxes or Other Taxes giving rise to such refund), net of all out-of-pocket expenses incurred by the Administrative Agent or such Lender and without interest (other than any interest paid by the relevant Governmental Authority with respect to such refund), provided that the Borrower, upon the request of the Administrative Agent or such Lender, agrees to repay the amount paid over to the Borrower (plus any penalties, interest or other charges imposed by the relevant Governmental Authority) to the Administrative Agent or such Lender in the event the Administrative Agent or such Lender is required to repay such refund to such Governmental Authority. This subsection shall not be construed to require the Borrower, the Administrative Agent or any Lender to make available its tax returns (or any other information relatin g to its Taxes that it deems confidential) to the Borrower or any other Person.

52

3.6 <u>Lender Statements; Survival of Indemnity.</u>

To the extent reasonably possible and upon the request of the Borrower, each Lender or Issuer shall designate an alternate Lending Installation to the extent it would reduce any liability of the Borrower to such Lender or Issuer under Sections 3.1, 3.2 and 3.5 or to avoid the unavailability of Eurodollar Advances under Section 3.3, so long as such designation is not, in the judgment of such Lender or Issuer, disadvantageous to such Lender or Issuer. Each Lender or each Issuer, as applicable, shall deliver a written statement of such Lender or such Issuer to the Borrower (with a copy to the Administrative Agent) as to any amount due under Section 3.1, 3.2, 3.4 or 3.5 and containing the other information contemplated by such Sections. Such written statement shall set forth in reasonable detail the calculations upon which such Lender or such Issuer determined such amount and shall be final, conclusive and binding on the Borrower in the absence of manifest error. Determination of amounts payable under such Sections in connection with a Eurodollar Loan shall be calculated as though each Lender funded its Eurodollar Loan through the purchase of a deposit of the type and maturity corresponding to the deposit used as a reference in determining the Eurodollar Rate applicable to such Loan , whether in fact that is the case or not. Unless otherwise provided herein, the amount specified in the written statement of any Lender or any Issuer shall be payable on demand after receipt by the Borrower of such written statement. The obligations of the Borrower under Sections 3.1, 3.2, 3.4 and 3.5 shall survive payment of the Obligations and termination of this Agreement.

ARTICLE IV

CONDITIONS PRECEDENT

4.1 <u>Initial Credit Extension</u>.

This Agreement shall become effective upon and the obligation of the Lenders and the Issuers to make the initial Credit Extension hereunder is subject to the Administrative Agent's receipt of (a) all fees required to be paid to the Administrative Agent, the Arrangers and the Lenders on the date hereof (including the reasonable fees and expenses of counsel to the Administrative Agent for which reasonably detailed invoices have been presented on or prior to the date hereof), (b) evidence that, prior to or concurrently with the effectiveness of this Agreement, all obligations under the Existing Credit Facility have been paid in full and all commitments to lend thereunder have been terminated, and (c) all of the following, in form and substance satisfactory to each Agent and each Lender, and in sufficient copies for each Lender:

(i) Copies of the articles or certificate of incorporation of the Borrower, together with all amendments, certified by the Secretary or an Assistant Secretary of the Borrower, and a certificate of good standing, certified by the appropriate governmental officer in its jurisdiction of incorporation, as well as any other information that any Lender may request that is required by Section 326 of the USA PATRIOT ACT or necessary for the Administrative Agent or any Lender to verify the identity of the Borrower as required by Section 326 of the USA PATRIOT ACT.

- (ii) Copies, certified by the Secretary or an Assistant Secretary of the Borrower, of its by-laws and of its Board of Directors' resolutions and of resolutions or actions of any other body authorizing the execution of the Loan Documents to which the Borrower is a party.
- (iii) An incumbency certificate, executed by the Secretary or an Assistant Secretary of the Borrower, which shall identify by name and title and bear the signatures of the Authorized Officers and any other officers of the Borrower authorized to sign the Loan Documents to which the Borrower is a party, upon which certificate the Administrative Agent and the Lenders shall be entitled to rely until informed of any change in writing by the Borrower.
- (iv) A certificate, signed by an Authorized Officer of the Borrower, stating that (A) on the date hereof, no Default or Unmatured Default has occurred and is continuing and (B) the representations and warranties contained in Article V are true and correct in all material respects as of the date hereof.
- (v) A written opinion of the Borrower's counsel, addressed to the Administrative Agent and the Lenders in a form reasonably satisfactory to the Administrative Agent and its counsel.
 - (vi) Executed counterparts of this Agreement executed by the Borrower, the Administrative Agent, each Issuer and each Lender.
- (vii) Any Revolving Notes requested by a Lender pursuant to <u>Section 2.13</u> payable to the order of each such requesting Lender and the Swing Line Note.
 - (viii) If the initial Credit Extension will be the issuance of a Letter of Credit, a properly completed Letter of Credit Application.
 - (ix) Evidence of the effectiveness of the KCPL Credit Agreement and the KCPL GMO Credit Agreement.
- (x) Written money transfer instructions, in substantially the form of Exhibit C, addressed to the Administrative Agent and signed by an Authorized Officer who has executed and delivered an incumbency certificate in accordance with the terms hereof, together with such other related money transfer authorizations as the Administrative Agent may have reasonably requested.
 - (xi) Such other documents as any Lender or its counsel may have reasonably requested.

4.2 Each Credit Extension.

The Lenders shall not be required to make any Credit Extension (other than a Credit Extension that, after giving effect thereto and to the application of the proceeds thereof, does not

54

increase the aggregate amount of outstanding Credit Extensions) or increase their Commitments pursuant to any GPE-KCPL Re-Transfer, GPE-GMO Re-Transfer, KCPL-GPE Transfer or GMO-GPE Transfer, unless on the date of such Credit Extension, GPE-KCPL Re-Transfer, GPE-GMO Re-Transfer, KCPL-GPE Transfer or GMO-GPE Transfer:

- (a) No Default or Unmatured Default exists or would result from such Credit Extension.
- (b) The representations and warranties contained in <u>Article V</u> are true and correct in all material respects as of the date of such Credit Extension except to the extent any such representation or warranty is stated to relate solely to an earlier date, in which case such representation or warranty shall have been true and correct in all material respects on and as of such earlier date; <u>provided</u> that this <u>clause (b)</u> shall not apply to the representations and warranties set forth in <u>Section 5.5</u> (as it relates to <u>clause (i)</u> or <u>(ii)</u> of the definition of "Material Adverse Effect"), <u>clause (a)</u> of the first sentence of <u>Section 5.7</u> and the second sentence of <u>Section 5.7</u> with respect to any borrowing hereunder which is not part of the Initial Credit Extension.

Each delivery of a Borrowing Notice and each request for the issuance of a Letter of Credit shall constitute a representation and warranty by the Borrower that the conditions contained in <u>Sections 4.2(a)</u> and (b) have been satisfied.

ARTICLE V

REPRESENTATIONS AND WARRANTIES

The Borrower represents and warrants to the Lenders that:

5.1 Existence and Standing.

Each of the Borrower and its Significant Subsidiaries (a) is a corporation, partnership (in the case of Subsidiaries only) or limited liability company duly and properly incorporated or organized, as the case may be, validly existing and (to the extent such concept applies to such entity) in good standing under the laws of its jurisdiction of incorporation or organization and (b) has all corporate power and authority and all governmental licenses, authorizations, consents and approvals required to carry on its business as now conducted, except where the failure to do so could not be reasonably expected to have a Material Adverse Effect.

5.2 <u>Authorization and Validity.</u>

The Borrower has the requisite corporate power and authority and legal right to execute and deliver the Loan Documents and to perform its obligations thereunder. The execution and delivery by the Borrower of the Loan Documents and the performance of its obligations thereunder have been duly authorized by all necessary corporate action, and the Loan Documents constitute legal, valid and binding obligations of the Borrower enforceable against the Borrower

55

in accordance with their terms, except as enforceability may be limited by bankruptcy, moratorium, reorganization or similar laws affecting the enforcement of creditors' rights generally and by general principles of equity (regardless of whether such enforceability is considered in a proceeding in equity or law).

5.3 <u>No Conflict; Government Consent.</u>

Neither the execution and delivery by the Borrower of the Loan Documents, nor the consummation of the transactions therein contemplated, nor compliance with the provisions thereof will (a) violate (i) in any material respect any law, statute, rule, regulation, order, writ, judgment, injunction, decree or award binding on the Borrower, (ii) the Borrower's certificate of incorporation or by-laws or (iii) the provisions of any indenture, instrument or agreement to which the Borrower is a party or is subject, or by which it, or its Property, is bound, except to the extent any such violation could not reasonably be expected to have a Material Adverse Effect (b) conflict with or constitute a default under any such indenture, instrument or other agreement in a manner which could reasonably be expected to have a Material Adverse Effect, or (c) result in, or require, the creation or imposition of any Lien in, of or on the Property of the Borrower pursuant to the terms of any such indenture, instrument or other agreement. No order, consent, adjudication, approval, license, authorization, or validation of, or filing, recording or registration with, or exemption by, or other action in respect of any governmental or public body or authority, or any subdivision thereof, which has not been obtained by the Borrower, is required to be obtained by the Borrower in connection with the execution and delivery of the Loan Documents, the borrowings under this Agreement, the payment and performance by the Borrower of the Obligations or the legality, validity, binding effect or enforceability of any of the Loan Documents excepting, however, '34 Act Reports due after the date hereof disclosing the Loan Documents and related information as required pursuant to applicable '34 Act Report filing requirements.

5.4 **Financial Statements.**

The December 31, 2009 and June 30, 2010 consolidated financial statements of the Borrower and its Subsidiaries heretofore delivered to the Lenders were prepared in accordance with GAAP and fairly present the consolidated financial condition and results of operations of the Borrower and its Subsidiaries at such dates and the consolidated results of their operations for the periods then ended subject, in the case of the June 30, 2010 financial statements, to normal year-end adjustments.

5.5 <u>Material Adverse Change</u>.

Since December 31, 2009 and except as disclosed in the Borrower's '34 Act Reports, there has been no change in the business, Property, financial condition or results of operations of the Borrower and its Subsidiaries, taken as a whole, which could reasonably be expected to have a Material Adverse Effect.

56

5.6 <u>Taxes</u>.

The Borrower and its Significant Subsidiaries have filed all United States federal income tax returns and all other material tax returns which are required to be filed by them and have paid all taxes due and payable pursuant to said returns or pursuant to any assessment received by the Borrower or any of its Significant Subsidiaries, except such taxes, if any, as are being contested in good faith and as to which adequate reserves have been provided in accordance with GAAP and as to which no Lien exists or except to the extent that the failure to pay any such tax, or to file any such tax return, could not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect. No tax liens have been filed and no material claims are being asserted against the Borrower or any Significant Subsidiary with respect to any such taxes except to the extent that such tax liens or claims could not reasonably be expected to have a Material Adverse Effect.

5.7 <u>Litigation; etc</u>.

Except as disclosed in the Borrower's '34 Act Reports, there is no litigation, arbitration, governmental investigation, proceeding or inquiry pending or, to the knowledge of any of their officers, threatened against or affecting the Borrower or any of its Subsidiaries in which there is a reasonable possibility of an adverse decision and which (a) could reasonably be expected to have a Material Adverse Effect or (b) seeks to prevent, enjoin or delay the making of any Credit Extension. Other than any liability incident to any litigation, arbitration or proceeding which could not reasonably be expected to have a Material Adverse Effect, the Borrower has no material contingent obligations not provided for or disclosed in the financial statements referred to in S ection 5.4.

5.8 <u>ERISA</u>.

The Borrower and each other member of the Controlled Group has fulfilled its obligations under the minimum funding standards of ERISA and the Code with respect to each Plan and is in compliance with the presently applicable provisions of ERISA and the Code with respect to each Plan, except to the extent that noncompliance, individually or in the aggregate, has not resulted in and could not reasonably be expected to result in a Material Adverse Effect. Neither the Borrower nor any other member of the Controlled Group has (i) failed to make any required contribution or payment to any Plan or Multiemployer Plan, or made any amendment to any Plan which has resulted or could result in the imposition of a Lien or the posting of a bond or other security under ERISA or the Code or (ii) incurred any liability under Title IV of ERI SA other than a liability to the PBGC for premiums under Section 4007 of ERISA.

5.9 <u>Accuracy of Information</u>.

No information, exhibit or report furnished by the Borrower or any of its Subsidiaries to the Administrative Agent, any Issuer or any Lender in connection with the negotiation of, or compliance with, the Loan Documents contained, as of the date of delivery thereof and taken as a whole, any untrue statement of a material fact or, when considered together with all reports filed with the SEC and furnished or deemed furnished pursuant to the terms hereof, omitted to state

57

any material fact necessary to make the statements contained therein, in light of the circumstances under which they were made, not misleading; <u>provided</u> that, with respect to projected financial information, forecasts and other forward-looking information, the Borrower represents only that such information was prepared in good faith based upon assumptions and estimates developed by management of the Borrower in good faith and believed to be reasonable at the time (it being understood that such information is not a guarantee of future performance and that actual results during the period or periods covered by such information may materially differ from the projected results therein).

5.10 Regulation U.

The Borrower is not engaged and will not engage, principally or as one of its important activities, in the business of purchasing or carrying margin stock (as defined in Regulation U), or extending credit for the purpose of purchasing or carrying margin stock. Margin stock constitutes less than twenty-five percent (25%) of the value of those assets of the Borrower and its Subsidiaries which are subject to any limitation on sale, pledge or other restriction hereunder.

5.11 <u>Material Agreements</u>.

Neither the Borrower nor any Subsidiary is a party to any agreement or instrument or subject to any charter or other corporate restriction which is reasonably likely to have a Material Adverse Effect. Neither the Borrower nor any Subsidiary is in default in the performance, observance or fulfillment of any of the obligations, covenants or conditions contained in any agreement to which it is a party, which default could reasonably be expected to have a Material Adverse Effect.

5.12 Compliance With Laws.

The Borrower and its Subsidiaries have complied with all applicable statutes, rules, regulations, orders and restrictions of any domestic or foreign government or any instrumentality or agency thereof having jurisdiction over the conduct of their respective businesses or the ownership of their respective Property except for any failure to comply with any of the foregoing which could not reasonably be expected to have a Material Adverse Effect.

5.13 Ownership of Properties.

On the date of this Agreement, the Borrower and its Significant Subsidiaries have good title, free of all Liens other than those permitted by <u>Section 6.12</u>, to all of the Property and assets reflected as owned by the Borrower and its Significant Subsidiaries in the Borrower's most recent consolidated financial statements provided to the Administrative Agent.

5.14 Plan Assets; Prohibited Transactions.

To the Borrower's knowledge, (i) the Borrower is not an entity deemed to hold "plan assets" within the meaning of 29 C.F.R. § 2510.3-101 of another entity's employee benefit plan (as defined in Section 3(3) of ERISA) which is subject to Title I of ERISA or any plan (within

58

the meaning of Section 4975 of the Code), and (ii) neither the execution of this Agreement nor the making of Loans hereunder gives rise to a prohibited transaction within the meaning of Section 406 of ERISA or Section 4975 of the Code.

5.15 <u>Environmental Matters.</u>

Except as set forth in the Borrower's '34 Act Reports, there are no known risks and liabilities accruing to the Borrower or any of its Subsidiaries due to Environmental Laws that could reasonably be expected to have a Material Adverse Effect.

5.16 <u>Investment Company Act</u>.

Neither the Borrower nor any Subsidiary is or is required to be registered as an "investment company" within the meaning of the Investment Company Act of 1940, as amended.

5.17 Pari Passu Indebtedness.

The Indebtedness under the Loan Documents ranks at least pari passu with all other unsecured Indebtedness of the Borrower.

5.18 <u>Solvency</u>.

As of the date hereof and after giving effect to the consummation of the transactions contemplated by the Loan Documents, the Borrower and each Significant Subsidiary is solvent. For purposes of the preceding sentence, solvent means (a) the fair saleable value (on a going concern basis) of the Borrower's assets or a Significant Subsidiary's assets, as applicable, exceed its liabilities, contingent or otherwise, fairly valued, (b) such Person will be able to pay its debts as they become due and (c) such Person will not be left with unreasonably small capital as is necessary to satisfy all of its current and reasonably anticipated obligations giving due consideration to the prevailing practice in the industry in which such Person is engaged. In computing the amount of contingent liabilities at any time, it is intended that such liabilities will be computed at the amount which, in light of all the facts and circumstances existing at such time, represents the amount that can reasonably be expected to become an actual or matured liability. The Borrower is not entering into the Loan Documents with the actual intent to hinder, delay or defraud its current or future creditors, nor does the Borrower intend to or believe that it will incur, as a result of entering into this Agreement and the other Loan Documents, debts beyond its ability to repay.

5.19 No Defaults.

No Unmatured Default or Default has occurred and is continuing.

59

ARTICLE VI

COVENANTS

During the term of this Agreement, unless the Required Lenders shall otherwise consent in writing:

6.1 Financial Reporting and Notices.

The Borrower will maintain, for itself and each Subsidiary, a system of accounting established and administered in accordance with generally accepted accounting principles, and (subject to the penultimate paragraph of this <u>Section 6.1</u>) furnish to the Administrative Agent for distribution to the Lenders:

- (a) Within ninety (90) days after the close of each of its fiscal years, an unqualified audit report certified by an independent registered public accounting firm which is a member of the "Big Four," prepared in accordance with GAAP on a consolidated basis for itself and its Consolidated Subsidiaries, including balance sheets as of the end of such period and related statements of income, common shareholders' equity and cash flows, accompanied by any management letter prepared by said accountants.
- (b) Within forty-five (45) days after the close of the first three (3) quarterly periods of each of its fiscal years, for itself and its Consolidated Subsidiaries, either (i) consolidated and consolidating unaudited balance sheets as at the close of each such period and consolidated and consolidating profit and loss and reconciliation of surplus statements and a statement of cash flows for the period from the beginning of such fiscal year to the end of such quarter, all certified by its Chief Accounting Officer or Chief Financial Officer or (ii) if the Borrower is then a "registrant" within the meaning of Rule 1-01 of Regulation S-X of the SEC and required to file a report on Form 10-Q with the SEC, a copy of the Borrower's report on Form&# 160;10-Q for such quarterly period.
- (c) Together with the financial statements required under <u>Sections 6.1(a)</u> and (b), a compliance certificate in substantially the form of <u>Exhibit A</u> signed by its Chief Accounting Officer or Chief Financial Officer setting forth calculations of the financial covenants contained in <u>Article VI</u> and stating that no Default or Unmatured Default exists, or if any Default or Unmatured Default exists, stating the nature and status thereof.
- (d) As soon as possible and in any event within thirty (30) days after the Borrower or any member of the Controlled Group knows that any Reportable Event has occurred with respect to any Plan that could reasonably be expected to result in liability in an aggregate amount in excess of \$50,000,000, a statement, signed by the Chief Accounting Officer or Chief Financial Officer of the Borrower, describing said Reportable Event and the action which the Borrower or member of the Controlled Group proposes to take with respect thereto.

60

- (e) Promptly after receipt of notice by the Borrower or any member of the Controlled Group of the PBGC's intention to terminate any Plan or to have a trustee appointed to administer any Plan, a copy of such notice.
- (f) Promptly upon the furnishing thereof to the shareholders of the Borrower, copies of all financial statements, reports and proxy statements so furnished.
- (g) Promptly upon the filing thereof, copies of all registration statements and annual, quarterly, monthly or other regular reports which the Borrower files with the SEC.
- (h) Promptly after the public announcement of, or promptly after receiving a written notice of, any change (whether an increase or decrease) by Moody's or S&P in the senior unsecured debt rating of the Borrower.
- (i) Such other information (including non-financial information) as the Administrative Agent or any Lender may from time to time reasonably request.

The statements and reports required to be furnished by the Borrower pursuant to <u>clauses (a)</u>, <u>(b)</u>, <u>(f)</u> and <u>(g)</u> above shall be deemed furnished for such purpose upon becoming publicly available on the SEC's EDGAR web page.

The Borrower hereby acknowledges that (a) the Administrative Agent and/or BAS will make available to the Lenders and Issuers materials and/or information provided by or on behalf of the Borrower hereunder (collectively, "Borrower Materials") by posting the Borrower Materials on IntraLinks or another similar restricted access electronic system (the "Platform") and (b) certain of the Lenders may be "public-side" Lenders (i.e., Lenders that do not wish to receive material non-public information with respect to the Borrower or its securities) (each, a "Public Lender"). The Borrower hereby agre es that (w) all Borrower Materials that are to be made available to Public Lenders shall be clearly and conspicuously marked "PUBLIC" which, at a minimum, shall mean that the word "PUBLIC" shall appear prominently on the first page thereof; (x) by marking Borrower Materials "PUBLIC," the Borrower shall be deemed to have authorized the Agents, the Arrangers, the Issuers and the Lenders to treat such Borrower Materials as not containing any material non-public information with respect to the Borrower or its securities for purposes of United States Federal and state securities laws (provided, however, that to the extent such Borrower Materials constitute Specified Information, they shall be treated as set forth in Section 9.11(a)); (y) all Borrower Materials marked "PUBLIC& #8221; are permitted to be made available through a portion of the Platform designated "Public Investor;" and (z) the Administrative Agent and BAS shall be entitled to treat any Borrower Materials that are not marked "PUBLIC" as being suitable only for posting on a portion of the Platform not designated "Public Investor." Notwithstanding the foregoing, the Borrower shall be under no obligation to mark any Borrower Materials "PUBLIC."

61

6.2 **Permits, Etc.**

The Borrower will, and will cause each Significant Subsidiary to, take all reasonable action to maintain all rights, privileges, permits, licenses and franchises necessary or desirable in the normal conduct of its business, except to the extent failure to do so could not reasonably be expected to have a Material Adverse Effect; and preserve or renew all of its registered patents, trademarks, trade names and service marks, the non-preservation of which could reasonably be expected to have a Material Adverse Effect.

6.3 <u>Use of Proceeds</u>.

The Borrower will use the proceeds of the Credit Extensions (i) to repay the Existing Credit Facility and (ii) for general corporate purposes, including the refinancing of any commercial paper program of the Borrower and its Subsidiaries and use for working capital, capital expenditures and other lawful purposes of the Borrower and its Subsidiaries. The Borrower will not use any of the proceeds of the Credit Extensions to purchase or carry any margin stock (as defined in Regulation U) or to extend credit for the purpose of purchasing or carrying margin stock; provided that the Borrower may repurchase its own stock or Equity-Linked Securities (or components thereof) so long as such stock or Equity-Linked Securities are not reissued. The Borrower will not permit margin stock to constitute twenty-five percent (25%) or more of the value of those assets of the Borrower and its Subsidiaries which are subject to any limitation on sale, pledge or other restriction hereunder.

6.4 Notice of Default.

The Borrower will give prompt notice in writing to the Administrative Agent and the Lenders of the occurrence of any Default or Unmatured Default and of any other development, financial or otherwise, which could reasonably be expected to have a Material Adverse Effect.

6.5 <u>Conduct of Business</u>.

Except as otherwise permitted under this Agreement, the Borrower will, and will cause each Significant Subsidiary to, carry on and conduct its business in substantially the same manner and in substantially the same general fields of enterprise as it is presently conducted and do all things necessary to remain duly incorporated or organized, validly existing and (to the extent such concept applies to such entity) in good standing as a domestic corporation, partnership or limited liability company in its jurisdiction of incorporation or organization, as the case may be, and except to the extent the failure to do so could not reasonably be expected to have a Material Adverse Effect, keep in full force and effect its respective rights, privileges and franchises necessary or desirable in the normal conduct of business; provided, however, that nothing in this Section 6.5 shall prohibit the termination of the corporate existence of a Subsidiary of the Borrower or a component of its business if the Borrower determines in good faith that such termination is in the best interest of the Borrower and is not materially disadvantageous to the Lenders.

62

6.6 <u>Taxes</u>.

The Borrower will, and will cause each Significant Subsidiary to, timely file United States federal and applicable foreign, state and local tax returns required by law and pay when due all taxes, assessments and governmental charges and levies upon it or its income, profits or Property, except those which are being contested in good faith by appropriate proceedings and with respect to which adequate reserves have been set aside in accordance with GAAP.

6.7 <u>Insurance</u>.

The Borrower will, and will cause each Significant Subsidiary to, maintain (either in such Borrower's name or in such Significant Subsidiary's own name) with financially sound and reputable insurance companies that are not Affiliates of the Borrower or its Subsidiaries (other than any captive insurance company) insurance on all their respective Properties and business against loss or damage of the kinds customarily insured against by Persons engaged in the same or similar business, of such types and in such amounts as are customarily carried under similar circumstances by such other Persons, and the Borrower will furnish to the Lenders upon request by the Administrative Agent, full information presented in reasonable detail as to the insurance carried. Such insurance may be subject to co-insurance, deductibility or similar clauses which, in effect, result in self-insurance of certain losses; provided that such self-

insurance is in accord with the customary industry practices for Persons in the same or similar businesses and adequate insurance reserves are maintained in connection with such self-insurance to the extent required by GAAP.

6.8 <u>Compliance with Laws</u>.

The Borrower will, and will cause each Significant Subsidiary to, comply with all laws (including, without limitation, Environmental Laws), rules, regulations, orders, writs, judgments, injunctions, decrees or awards to which it may be subject, except where the necessity of compliance therewith is contested in good faith by appropriate proceedings or except where the failure to comply could not reasonably be expected to have a Material Adverse Effect.

6.9 Maintenance of Properties; Books of Record.

The Borrower will, and will cause each Significant Subsidiary to, (i) do all things necessary to maintain, preserve, protect and keep its material Property in good repair, working order and condition, ordinary wear and tear excepted, and make all necessary and proper repairs, renewals and replacements so that its business carried on in connection therewith may be properly conducted at all times and (ii) keep proper books of record and account, in which full and correct entries shall be made of all material financial transactions and the assets and business of the Borrower and each Significant Subsidiary in accordance with GAAP; provided that nothing in this Section 6.9 shall prevent the Borrower or any Si gnificant Subsidiary from selling, abandoning, disposing of, or otherwise discontinuing the operation or maintenance of any of its Property (including the Capital Stock of any Subsidiary that is not a Significant Subsidiary) or equipment the retention of which in the good faith judgment of the Borrower or such Significant

63

Subsidiary is inadvisable or unnecessary to the Borrower and its Subsidiaries, taken as a whole, or the failure to could not reasonably be expected to have a Material Adverse Effect.

6.10 <u>Inspection</u>.

The Borrower will, and if a Default or Unmatured Default exists, will cause each Significant Subsidiary to, permit the Administrative Agent and the Lenders, by their respective representatives and agents designated by the Required Lenders, upon not less than five (5) Business Days notice, to inspect any of the Property, books and financial records of such Person, to examine and make copies of the books of accounts and other financial records of such Person, and to discuss the affairs, finances and accounts of such Person with, and to be advised as to the same by, such Person's officers at such reasonable times and intervals as the Administrative Agent or any Lender may designate, subject, however, in all cases to the imposition of such conditions as the Borrower and such Significant Subsidiaries shall deem necessary based on reason able considerations of safety and security; provided, however, that neither the Borrower nor any of its Significant Subsidiaries shall be required to disclose to the Administrative Agent, and any Lender, or any representatives thereof any information which is the subject of attorney-client privilege or attorney work-product privilege properly asserted by the applicable Person to prevent the loss of such privilege in connection with such information or which is prevented from disclosure pursuant to a confidentiality agreement with third parties. After the occurrence and during the continuance of a Default, any such inspection shall be at the Borrower's expense; at all other times, the Borrower shall not be liable to pay the expenses of the Administrative Agent or any Lender in connection with such inspections.

6.11 <u>Consolidations, Mergers and Sale of Assets.</u>

The Borrower will not, nor will it permit any Significant Subsidiary (other than any Project Finance Subsidiary) to, sell, lease, transfer, or otherwise dispose of all or substantially all of its assets (whether by a single transaction or a number of related transactions and whether at one time or over a period of time) or consolidate with or merge into any Person or permit any Person to merge into it, except

- (a) A Wholly-Owned Subsidiary may be merged into the Borrower.
- (b) Any Significant Subsidiary may sell all or substantially all of its assets to, or consolidate or merge into, another Significant Subsidiary; <u>provided</u> that, immediately before and after such merger, consolidation or sale, no Default or Unmatured Default shall exist.
 - (c) KCPL or KCPL GMO may sell or transfer accounts receivable, in each case pursuant to one or more securitization transactions.
- (d) The Borrower may sell all or substantially all of its assets to, or consolidate with or merge into, any other Person, or permit another Person to merge into it; <u>provided</u> that (i) the surviving Person, if such surviving Person is not the Borrower, or the transferee Person in the case of a sale of all or substantially all of the Borrower's

64

assets (A) shall be a Person organized and existing under the laws of the United States of America or a state thereof or the District of Columbia, (B) shall expressly assume, in a writing reasonably satisfactory to the Administrative Agent, the due and punctual payment of the Obligations and the due and punctual performance of and compliance with all of the terms of this Agreement and the other Loan Documents to be performed or complied with by the Borrower and (C) shall deliver all documents required to be delivered pursuant to Sections 4.1(c)(i), (c)(ii), (c)(iii), (c)(iii), (c)(iii), (c)(iii) and (c)(ix), (ii) immediately before and after such merger, consolidation or sale, there shall not exist any Default or Unmatured Default and (iii) the surviving Person of such merger or consolidation, or the transferee Person of the assets of the Borrower, as applicable, has, both immediately before and after such merger, consolidation or sale, a Moody's Rating of Baa3 or better or an S&P Rating of BBB - or better.

Notwithstanding the foregoing, the Borrower and its Consolidated Subsidiaries (excluding Project Finance Subsidiaries) will not convey, transfer, lease or otherwise dispose of (whether in one transaction or a series of transactions, but excluding (i) sales of inventory in the ordinary course of business, (ii) transactions permitted by clauses (a) through (d) above, (iii) transfers by KCPL or KCPL GMO of assets related to, or ownership interests in, Iatan 2 to co-

owners of Iatan 2 pursuant to the co-ownership, co-operating or other similar agreements of the co-owners of Iatan 2 and (iv) sales of the capital stock or assets of KLT Gas Inc. and Subsidiaries thereof) in the aggregate within any twelve (12)-month period, more than twenty percent (20%) of the aggregate book value of the assets of the Borrower and its Consolidated Subsidiaries (excluding Project Finance Subsidiaries) as calculated as of the end of the most recent fiscal quarter.

6.12 <u>Liens</u>.

The Borrower will not, nor will it permit any Significant Subsidiary (other than any Project Finance Subsidiary) to, create, incur, or suffer to exist any Lien in, of or on the Property of the Borrower or any of its Significant Subsidiaries (other than any Project Finance Subsidiary), except:

- (a) Liens for taxes, assessments or governmental charges or levies on its Property if the same shall not at the time be delinquent or thereafter can be paid without penalty, or are being contested in good faith and by appropriate proceedings and for which adequate reserves in accordance with GAAP shall have been set aside on its books.
- (b) Liens imposed by law, such as carriers', warehousemen's, mechanics' and landlords' liens and other similar liens arising in the ordinary course of business which secure payment of obligations not more than sixty (60) days past due or which are being contested in good faith by appropriate proceedings and for which adequate reserves shall have been set aside on its books.
- (c) Liens arising out of pledges or deposits in the ordinary course of business under worker's compensation laws, unemployment insurance, old age pensions, or other

65

social security or retirement benefits, or similar legislation, other than any Lien imposed under ERISA.

- (d) Liens incidental to the normal conduct of the Borrower or any Significant Subsidiary or the ownership or leasing of its Property or the conduct of the ordinary course of its business, including (i) zoning restrictions, easements, building restrictions, rights of way, reservations, restrictions on the use of real property and such other encumbrances or charges against real property as are of a nature generally existing with respect to properties of a similar character and which are not substantial in amount and do not in any material way affect the marketability of the same, (ii) rights of lessees and lessors under leases, (iii) rights of collecting banks having rights of setoff, revocation, refund or chargeback with respect to money or instruments of the Borrower or any Significant Subsidiary on deposit with or in the possession of such banks, (iv) Liens or deposits to secure the performance of statutory obligations, tenders, bids, contracts, leases, progress payments, performance or return-of-money bonds, surety and appeal bonds, performance or other similar bonds, letters of credit, or other obligations of a similar nature incurred in the ordinary course of business, (v) Liens required by any contract or statute in order to permit the Borrower or Significant Subsidiary to perform any contract or subcontract made by it with or pursuant to the requirements of a governmental entity, in each case which are not incurred in connection with the borrowing of money, the obtaining of advances of credit or the payment of the deferred purchase price of Property and which do not in the aggregate impair the use of Property in the operation of the business of the Borrower and its Significant Subsidiaries taken as a whole and (vi) rights reserved to or vested in any Governmental Author ity by the terms of any right, power, franchise, grant, license or permit, or by any Laws, to terminate such right, power, franchise, license or permit or to purchase, condemn, expropriate or recapture or to designate a purchaser of any of
 - (e) Liens arising under the General Mortgage Indenture and Deed of Trust Dated December 1, 1986 from KCPL to UMB, N.A.
- (f) Liens on Property of the Borrower, KCPL or KCPL GMO existing on the date hereof and any renewal or extension thereof; provided that the Property covered thereby is not increased and any renewal or extension of the obligations secured or benefited thereby is permitted by this Agreement.
 - (g) Judgment Liens which secure payment of legal obligations that would not constitute a Default under <u>Section 7.9</u>.
- (h) Liens on Property acquired by the Borrower or a Significant Subsidiary after the date hereof, existing on such Property at the time of acquisition thereof (and not created in anticipation thereof); <u>provided</u> that in any such case no such Lien shall extend to or cover any other Property of the Borrower or such Significant Subsidiary, as the case may be.

66

- (i) Liens on the Property, revenues and/or assets of any Person that exist at the time such Person becomes a Significant Subsidiary and the continuation of such Liens in connection with any refinancing or restructuring of the obligations secured by such Liens.
- (j) Liens on Property securing Indebtedness incurred or assumed at the time of, or within twelve (12) months after, the acquisition of such Property for the purpose of financing all or any part of the cost of acquiring such Property; provided that (i) such Lien attaches to such Property concurrently with or within twelve (12) months after the acquisition thereof, (ii) such Lien attaches solely to the Property so acquired in such transaction and (iii) the principal amount of the Indebtedness secured thereby does not exceed the cost or fair market value determined at the date of incurrence, whichever is lower, of the Property being acquired on the date of acquisition.
- (k) Liens on any improvements to Property securing Indebtedness incurred to provide funds for all or part of the cost of such improvements in a principal amount not exceeding the cost of construction of such improvements and incurred within twelve (12) months after completion of such improvements or construction, <u>provided</u> that such Liens do not extend to or cover any property of the Borrower or any Significant Subsidiary other than such improvements.

- (l) Liens to government entities granted to secure pollution control or industrial revenue bond financings, which Liens in each financing transaction cover only Property the acquisition or construction of which was financed by such financings and Property related thereto.
- (m) Liens on or over gas, oil, coal, fissionable material, or other fuel or fuel products as security for any obligations incurred by such Person (or any special purpose entity formed by such Person) for the sole purpose of financing the acquisition or storage of such fuel or fuel products or, with respect to nuclear fuel, the processing, reprocessing, sorting, storage and disposal thereof.
- (n) Liens on (including Liens arising out of the sale of) accounts receivable and/or contracts which will give rise to accounts receivable of KCPL and KCPL GMO; and other Liens on (including Liens arising out of the sale of) accounts receivable and/or contracts which will give rise to accounts receivable of the Borrower or any Significant Subsidiary in an aggregate amount not at any time exceeding \$10,000,000.
- (o) Liens on Property or assets of a Significant Subsidiary securing obligations owing to the Borrower or any Significant Subsidiary (other than a Project Finance Subsidiary).
- (p) Liens on the stock or other equity interests of any Project Finance Subsidiary to secure obligations of such Project Finance Subsidiary (provided that the agreement under which any such Lien is created shall expressly state that it is non-recourse to the pledgor).

67

- (q) Liens on Property of KCPL and KCPL GMO arising in connection with utility co-ownership, co-operating and similar agreements that are consistent with the utilities business and ancillary operations.
- (r) Liens on assets held by entities which are deemed Subsidiaries solely due to the application of clause (c) of the definition of "Subsidiary" and which are required to be included in the Borrower's consolidated financial statements solely as a result of the application of ASC Topic 810, as it may be amended or supplemented.
 - (s) Liens securing Swap Contracts permitted to be incurred under this Agreement.
- (t) Liens securing any extension, renewal, replacement or refinancing of Indebtedness secured by any Lien referred to in the foregoing <u>clauses (h), (i), (j), (k), (l), (m)</u> and (q); provided that (A) such new Lien shall be limited to all or part of the same Property that secured the original Lien (plus improvements on such Property) and (B) the amount secured by such Lien at such time is not increased to any amount greater than the amount outstanding at the time of such renewal, replacement or refinancing.
- (u) Liens in favor of the Issuers or the Swing Line Lender on cash collateral securing the obligations of a Defaulting Lender to fund risk participations hereunder.
- (v) Liens which would otherwise not be permitted by <u>clauses (a)</u> through <u>(u)</u> securing additional Indebtedness of the Borrower or a Significant Subsidiary (other than a Project Finance Subsidiary); <u>provided</u> that after giving effect thereto the aggregate unpaid principal amount of Indebtedness (including Capitalized Lease Obligations) of the Borrower and its Significant Subsidiaries (other than any Project Finance Subsidiary) (including prepayment premiums and penalties) secured by Liens permitted by this <u>clause (y)</u> sha ll not exceed the greater of (a) \$50,000,000 and (b) ten percent (10%) of Consolidated Tangible Net Worth.
- (w) Liens on the stock or assets of Subsidiaries created pursuant to "rate reduction" bonds, for the payment of which legislatively authorized charges are imposed on customers.

6.13 <u>Affiliates</u>.

Except to the extent required by applicable Law or regulation, the Borrower will not, and will not permit any Subsidiary (other than any Project Finance Subsidiary) to, enter into any transaction (including the purchase or sale of any Property or service) with, or make any payment or transfer to, any Affiliate except in the ordinary course of business and pursuant to the reasonable requirements of the Borrower's or such Subsidiary's business and upon fair and

68

reasonable terms no less favorable to the Borrower or such Subsidiary than the Borrower or such Subsidiary would obtain in a comparable arms-length transaction.

6.14 ERISA.

The Borrower will not, nor will it permit any Significant Subsidiary to, (i) voluntarily terminate any Plan, so as to result in any material liability of the Borrower or any Significant Subsidiary to the PBGC or (ii) enter into any transaction that the Borrower or such Significant Subsidiary knows or should know is a Prohibited Transaction (as defined in Section 4975 of the Code and in Section 406 of ERISA) involving any Plan which results in any liability of the Borrower or any Significant Subsidiary that could reasonably be expected, individually or in the aggregate, to cause a Material Adverse Effect or (iii) cause any occurrence of any Reportable Event which results in any liability of the Borrower or any Significant Subsidiary to the PBGC that could reasonably be expected, individually or in the aggregate, to cause a Material Adverse Effect or (iv) allow or suffer to exist any other event or condition known to the Borrower which results in any material liability of the Borrower or any Significant Subsidiary to the PBGC.

6.15 <u>Total Indebtedness to Total Capitalization.</u>

The Borrower shall at all times cause the ratio of (i) Total Indebtedness to (ii) Total Capitalization to be less than or equal to 0.65 to 1.0.

6.16 <u>Restrictions on Subsidiary Dividends.</u>

The Borrower will not, nor will it permit any Significant Subsidiary (other than any Project Finance Subsidiary) to, be a party to any agreement prohibiting or restricting the ability of such Significant Subsidiary to declare or pay dividends to the Borrower; <u>provided</u>, that the foregoing provisions of this <u>Section 6.16</u> shall not prohibit the Borrower or any Significant Subsidiary from entering into any debt instrument containing a total debt to capitalization covenant.

6.17 <u>Organization Documents</u>.

The Borrower and its Subsidiaries shall not amend, modify or change their Organization Documents in a manner materially adverse to the Lenders.

ARTICLE VII

DEFAULTS

The occurrence of any one or more of the following events shall constitute a Default:

7.1 Any representation or warranty made or deemed made by or on behalf of the Borrower to the Lenders or the Administrative Agent under or in connection with this Agreement, any Loan, or any certificate or information delivered in connection with this

69

Agreement or any other Loan Document shall prove to have been incorrect in any material respect when made or deemed made.

- 7.2 (a) Nonpayment of principal of any Loan when due, nonpayment of any Reimbursement Obligations within one (1) Business Day after the same becomes due, or (b) nonpayment of interest upon any Loan or of any fee or other obligation under any of the Loan Documents within five (5) Business Days after the same becomes due.
- **7.3** The breach by the Borrower of any of the terms or provisions of <u>Section 6.3</u>, <u>6.10</u> (with respect to the Borrower and its Significant Subsidiaries only), <u>6.11</u>, <u>6.15</u> or <u>6.16</u>.
- 7.4 The breach by the Borrower of any of the terms or provisions of Section 6.12 which is not remedied within thirty (30) days after the earlier of (a) an Authorized Officer of the Borrower becoming aware of such breach and (b) receipt by the Borrower of written notice from the Administrative Agent or any Lender.
- 7.5 The breach by the Borrower (other than a breach which constitutes a Default under another Section of this Article VII) of any of the terms or provisions of this Agreement which is not remedied within thirty (30) days after the earlier of (a) an Authorized Officer of the Borrower becoming aware of such breach and (b) receipt by the Borrower of written notice from the Administrative Agent or any Lender; provided that if such breach is capable of cure but (i) cannot be c ured by payment of money and (ii) cannot be cured by diligent efforts within such thirty (30)-day period, but such diligent efforts shall be properly commenced within such thirty (30)-day period and the Borrower is diligently pursuing, and shall continue to pursue diligently, remedy of such failure, the cure period shall be extended for an additional ninety (90) days, but in no event beyond the Facility Termination Date.
- 7.6 Failure of the Borrower or any of its Significant Subsidiaries to pay when due any Indebtedness aggregating in excess of \$50,000,000 ("Material Indebtedness"); or the default by the Borrower or any of its Significant Subsidiaries in the performance of any term, provision or condition contained in any agreement under which any such Material Indebtedness was created or is governed, or any other event shall occur or condition exist, the effect of which default or event is to cause, or to permit the holder or holders of such Material Indebtedness to cause, such Material Indebtedness to become due prior to its stated maturity; or any Material Indebtedness of the Borrower or any of its Significant Subsidiaries shall be declared to be due and payable or required to be prepaid or repurchased (other than by a regularly scheduled payment) prior to the stated maturity thereof; or the Borrower or any of its Significant Subsidiaries shall not pay, or admit in writing its inability to pay, its debts generally as they become due.
- 7.7 The Borrower or any of its Significant Subsidiaries shall (a) have an order for relief entered with respect to it under the Federal bankruptcy laws as now or hereafter in effect, (b) make an assignment for the benefit of creditors, (c) apply for, seek, consent to, or acquiesce in, the appointment of a receiver, custodian, trustee, examiner, liquidator or similar official for it or any Substantial Portion of its Property, (d) institute any proceeding seeking an order for relief under the Federal bankruptcy laws as now or hereafter in effect or seeking to adjudicate it a bankrupt or insolvent, or seeking di ssolution, winding up, liquidation, reorganization,

70

- **7.8** Without the application, approval or consent of the Borrower or any of its Subsidiaries, a receiver, trustee, examiner, liquidator or similar official shall be appointed for the Borrower or any of its Subsidiaries or any Substantial Portion of its Property, or a proceeding described in Section 7.7(d) shall be instituted against the Borrower or any of its Subsidiaries and such appointment continues undischarged or such proceeding continues undismissed or unstayed for a period of sixty (60) consecutive days.
- **7.9** The Borrower or any of its Significant Subsidiaries shall fail within thirty (30) days to pay, bond or otherwise discharge (a) any judgment or order for the payment of money not covered by insurance in excess of \$50,000,000 (either singly or in the aggregate with other such judgments) or (b) any non-monetary final judgment that has, or could reasonably be expected to have, a Material Adverse Effect, in either case which is not stayed on appeal or otherwise being appropriately contested in good faith.
 - **7.10** A Change of Control shall occur.
- **7.11** A Reportable Event shall have occurred which results in liability of the Borrower or any Significant Subsidiary in excess of \$50,000,000 and, thirty (30) days after written notice thereof shall have been given to the Borrower by the Administrative Agent or any Lender, the amount of such liability shall still exceed \$50,000,000.
- 7.12 Any authorization or approval or other action by any Governmental Authority or regulatory body required for the execution, delivery or performance of this Agreement or any other Loan Document by the Borrower shall fail to have been obtained at the time required or be terminated, revoked or rescinded or shall otherwise no longer be in full force and effect, and such occurrence shall (i) materially adversely affect the enforceability of the Loan Documents against the Borrower and (ii) to the extent that such occurrence can be cured, shall continue for five (5) days (unless being contested in good faith by the Borrower and the pendency of such contest shall suspend such adverse effect).
- **7.13** The Borrower shall fail to own, directly or indirectly, all of the outstanding stock of (a) KCPL which, in the absence of any contingency, has the right to vote in an election of directors of KCPL or (b) KCPL GMO which, in the absence of any contingency, has the right to vote in an election of directors of KCPL GMO.
- **7.14** Any Loan Document, at any time after its execution and delivery and for any reason other than as expressly permitted hereunder or thereunder or satisfaction in full of all the Obligations, ceases to be in full force and effect in any material respect; or the Borrower contests in any manner the validity or enforceability of any Loan Document; or the Borrower denies that

71

it has any or further liability or obligation under any Loan Document, or purports to revoke, terminate or rescind any Loan Document.

ARTICLE VIII

ACCELERATION, WAIVERS, AMENDMENTS AND REMEDIES

8.1 <u>Acceleration; Letter of Credit Account.</u>

(a) If any Default described in Section 7.7 or 7.8 occurs with respect to the Borrower, the obligations of the Lenders to make Loans hereunder and the obligation and power of the Issuers to issue Letters of Credit and the Swing Line Lender to make Swing Line Loans shall automatically terminate and the Obligations shall immediately become due and payable without any election or action on the part of the Administrative Agent, any Lender or any Issuer and the Borrower will be and become thereby unconditionally obligated, without any further notice, act or demand, to pay to the Administrative Agent an amount in immediately available funds, which funds shall be held in the LC Collateral Accounts, equal to the excess of (i) the amount of Letter of Credit Obligations at such time over (ii) the amount on deposit in the LC Collateral Account at such time which is free and clear of all rights and claims of third parties and has not been applied against the Obligations (such difference, the "Collateral Shortfall Amount"). If any other Default occurs, the Administrative Agent may with the consent, or shall at the request, of the Required Lenders, (x) terminate or suspend the obligations of the Lenders to make Loans hereunder and the obligation and power of the Issuers to issue Letters of Credit, or declare the Obligations to be due and payable, or both, whereupon the Obligations shall become immediately due and payable, without presentment, demand, protest or notice of any kind, all of which the Borrower hereby expressly waives, and (y) upon notice to the Borrower and i n addition to the continuing right to demand payment of all amounts payable under this Agreement, make demand on the Borrower to pay, and the Borrower will, forthwith upon such demand and without any further notice or act, pay to the Administrative Agent in immediately available funds the Collateral Shortfall Amount, which funds shall be deposited in the LC Collateral Accounts.

If (a) within thirty (30) days after acceleration of the maturity of the Obligations or termination of the obligations of the Lenders to make Loans hereunder as a result of any Default (other than any Default as described in <u>Section 7.7</u> or <u>7.8</u> with respect to the Borrower) and (b) before any judgment or decree for the payment of the Obligations due shall have been obtained or entered, the Required Lenders (in their sole discretion) shall so direct, the Administrative Agent shall, by notice to the Borrower, rescind and annul such acceleration and/or termination.

8.2 <u>Amendments</u>.

Subject to the provisions of this <u>Article VIII</u>, the Required Lenders (or the Administrative Agent with the consent in writing of the Required Lenders) and the Borrower may enter into agreements supplemental hereto for the purpose of adding or modifying any provisions to the

Loan Documents or changing in any manner the rights of the Lenders or the Borrower hereunder or waiving any Default hereunder; <u>provided</u> that no such supplemental agreement shall:

- (a) Extend the final maturity of any Loan or the expiry date of any Letter of Credit to a date after the Facility Termination Date or forgive all or any portion of the principal amount thereof, or reduce the rate or extend the time of payment of interest (other than default interest) or fees thereon, without the consent of each Lender directly affected thereby;
- (b) Reduce the percentage specified in the definition of Required Lenders, without the consent of each Lender directly affected thereby;
- (c) Increase the amount of the Commitment of any Lender without the consent of such Lender (except as provided for in <u>Section 2.6</u>), or extend the Facility Termination Date (except as provided for in <u>Section 2.21</u>), reduce the amount or extend the payment date for, the mandatory payments required under <u>Section 2.2</u> or permit the Borrower to assign its rights under this Agreement, without the consent of the Administrative Agent, the Issuers, the Swing Line Lender, and each Lender directly affected thereby;
 - (d) Amend this <u>Section 8.2</u> without the consent of each Lender directly affected thereby;
- (e) Release any funds from the LC Collateral Accounts, except to the extent such release is expressly permitted hereunder without the consent of each Lender directly affected thereby; or
- (f) Alter the pro rata sharing of payments, the order of application of payments required thereby or the pro rata treatment of the Lenders without the consent of each Lender directly affected thereby.

Notwithstanding the foregoing, any amendment, waiver, or consent which by its terms requires the consent of all Lenders or each affected Lender may be effected with the consent of the applicable Lenders other than the Defaulting Lenders except that (x) the Commitment of any Defaulting Lender may not be increased or extended without the consent of each Lender and (y) any waiver, amendment or modification requiring the consent of all Lenders or each affected Lender that (i) by its terms affects any Defaulting Lender more adversely than other affected Lenders, (ii) reduces the principal of or the rate of interest for Loans of such Defaulting Lender, or fees or other amounts payable hereunder or under any other Loan Document to such Defaulting Lender or (iii) extends the time of the payment of interest or fees owing to such Defaulting Lender shall require the consent of such Defaulting Lender. No amendment of any provision of this Agreement relating to the Administrative Agent shall be effective without the written consent of such Issuer and no amendment of any provision of this Agreement relating to the Swing Line Lender shall be effective without the written consent of the Swing Line Lender. The Administrative Agent may

73

waive payment of the fee required under Section 12.1(b) without obtaining the consent of any other party to this Agreement.

8.3 Preservation of Rights.

No delay or omission of the Lenders, the Swing Line Lender, the Issuers or the Administrative Agent to exercise any right under the Loan Documents shall impair such right or be construed to be a waiver of any Default or an acquiescence therein, and the making of a Credit Extension notwithstanding the existence of a Default or the inability of the Borrower to satisfy the conditions precedent to such Credit Extension shall not constitute any waiver or acquiescence. Any single or partial exercise of any such right shall not preclude other or further exercise thereof or the exercise of any other right, and no waiver, amendment or other variation of the terms, conditions or provisions of the Loan Documents whatsoever shall be valid unless in writing signed by the Lenders required pursuant to Section 8.2, and then only to the extent in such writing specifically set forth. All remedies contained in the Loan Documents or by law afforded shall be cumulative and all shall be available to the Administrative Agent, the Swing Line Lender, the Lenders and the Issuers until the Obligations have been paid in full.

ARTICLE IX

GENERAL PROVISIONS

9.1 <u>Survival of Representations.</u>

All representations and warranties of the Borrower contained in this Agreement shall survive the making of the Credit Extensions herein contemplated.

9.2 <u>Governmental Regulation</u>.

Anything contained in this Agreement to the contrary notwithstanding, no Lender shall be obligated to extend credit to the Borrower in violation of any limitation or prohibition provided by any applicable statute or regulation.

9.3 <u>Headings</u>.

Section headings in the Loan Documents are for convenience of reference only, and shall not govern the interpretation of any of the provisions of the Loan Documents.

9.4 Entire Agreement.

THIS AGREEMENT AND THE OTHER LOAN DOCUMENTS REPRESENT THE FINAL AGREEMENT AMONG THE PARTIES AND MAY NOT BE CONTRADICTED BY EVIDENCE OF PRIOR, CONTEMPORANEOUS, OR SUBSEQUENT ORAL AGREEMENTS OF THE PARTIES. THERE ARE NO UNWRITTEN ORAL AGREEMENTS AMONG THE PARTIES.

74

9.5 <u>Several Obligations; Benefits of this Agreement.</u>

The respective obligations of the Lenders hereunder are several and not joint and no Lender shall be the partner or agent of any other (except to the extent to which the Administrative Agent is authorized to act as such). The failure of any Lender to perform any of its obligations hereunder shall not relieve any other Lender from any of its obligations hereunder. This Agreement shall not be construed so as to confer any right or benefit upon any Person other than the parties to this Agreement and their respective successors and assigns; provided that the parties hereto expressly agree that each Arranger shall enjoy the benefits of the provisions of Sections 9.6, 9.10 and 10.7 to the extent specifically set forth therein and shall have the right to enforce such provisions on its own behalf and in its own name to the same extent as if it were a party to this Agreement.

9.6 Expenses; Indemnification.

- (a) The Borrower shall reimburse (i) the Agents and the Arrangers for any reasonable costs and expenses (including the reasonable fees and charges of outside counsel for the Agents) paid or incurred by the Agents or the Arrangers in connection with the preparation, negotiation, execution, delivery, syndication, distribution (including via the internet), review, amendment, modification, and administration of the Loan Documents and (ii) the Issuers for all reasonable out-of-pocket expenses incurred by the Issuers in connection with the issuance, amendment, renewal or extension of any Letter of Credit or any demand for payment thereunder. The Borrower also agrees to reimburse each Agent, each Arranger, the Swing Line Lender, each Lender and each Issuer for any reasonable costs, internal charges and expenses (including the reasonable fees and charges of attorneys for such Agent, such Arranger, the Swing Line Lender, such Lender or such Issuer) paid or incurred by either Agent, either Arranger, the Swing Line Lender, any Lender or any Issuer in connection with the collection and enforcement, attempted enforcement, and preservation of rights and remedies under, any of the Loan Documents (including all such costs and expenses incurred during any "workout" or restructuring in respect of the Obligations and during any legal proceeding).
- (b) The Borrower hereby further agrees to indemnify each Agent, each Arranger, the Swing Line Lender, each Lender, each Issuer, their respective affiliates and the directors, partners, officers, employees, agents and advisors of the foregoing against all losses, claims, damages, penalties, judgments, liabilities and expenses (including all reasonable expenses of litigation or preparation therefor whether or not either Agent, either Arranger, the Swing Line Lender, any Lender or any Issuer or any affiliate is a party thereto) which any of them may pay or incur arising out of or relating to this Agreement, the other Loan Documents, the transactions contemplated hereby or the direct or indirect application or proposed application of the proceeds of any Credit Extension hereund er except to the extent that they are determined in a final non-appealable judgment by a court of competent jurisdiction to have resulted from the gross negligence or willful misconduct of the party seeking indemnification. In the case of any investigation, litigation or proceeding to which the indemnity in this Section applies, such indemnity shall be effective whether or not such investigation, litigation or proceeding is brought

75

by a third party, by the Borrower or by any affiliate of the Borrower. The obligations of the Borrower under this <u>Section 9.6</u> shall survive the payment of the Obligations and termination of this Agreement.

- (c) To the extent that the Borrower for any reason fails to indefeasibly pay any amount required under <u>subsection (a)</u> or (b) of this Section to be paid by it to the Administrative Agent (or any sub-agent thereof), the Swing Line Lender, any Issuer or any Related Party of any of the foregoing (and without limiting its obligation to do so), each Lender severally agrees to pay to the Administrative Agent (or any such sub-agent), the Swing Line Lender, such Issuer or such Related Party, as the case may be, such Lender's Pro Rata Share (determined as of the time that the applicable unreimbursed expense or indemnity payment is sought but without giving effect to the reallocation of Pro Rata Shares contemplated by <u>Section 2.23(a)(iv)</u>) of such unpaid amount, <u>provided</u> that the unreimbursed expense or indemnified loss, claim, damage, liability or related expense, as the case may be, was incurred by or asserted against the Administrative Agent (or any such sub-agent), the Swing Line Lender or any Issuer in its capacity as such, or against any Related Party of any of the foregoing acting for the Administrative Agent (or any such sub-agent), the Swing Line Lender or any Issuer in connection with such capacity.
- (d) Notwithstanding any provision to the contrary, this <u>Section 9.6</u> shall not apply with respect to any matters, liabilities, or obligations relating to Taxes; all such matters are governed exclusively by <u>Section 3.5</u>.

9.7 <u>Numbers of Documents</u>.

All statements, notices, closing documents, and requests hereunder shall be furnished to the Administrative Agent with sufficient counterparts so that the Administrative Agent may furnish one to each of the Lenders.

9.8 Accounting.

Except as provided to the contrary herein, all accounting terms used herein shall be interpreted and all accounting determinations hereunder shall be made in accordance with GAAP.

9.9 <u>Severability of Provisions</u>.

Any provision in any Loan Document that is held to be inoperative, unenforceable, or invalid in any jurisdiction shall, as to that jurisdiction, be inoperative, unenforceable, or invalid without affecting the remaining provisions in that jurisdiction or the operation, enforceability, or validity of that

provision in any other jurisdiction, and to this end the provisions of all Loan Documents are declared to be severable. Without limiting the foregoing provisions of this Section 9.9, if and to the extent that the enforceability of any provisions in this Agreement relating to Defaulting Lenders shall be limited by Debtor Relief Laws, as determined in good faith by the Administrative Agent, the Issuers or the Swing Line Lender, as applicable, then such provisions shall be deemed to be in effect only to the extent not so limited.

76

9.10 Nonliability of Lenders.

The relationship between the Borrower on the one hand and the Lenders, the Swing Line Lender, the Issuers and the Agents on the other hand shall be solely that of borrower and lender. None of either Agent, either Arranger, the Swing Line Lender, any Lender or any Issuer shall have any fiduciary responsibilities to the Borrower. None of either Agent, either Arranger, the Swing Line Lender, any Lender or any Issuer undertakes any responsibility to the Borrower to review or inform the Borrower of any matter in connection with any phase of the Borrower's business or operations. The Borrower agrees that none of either Agent, either Arranger, the Swing Line Lender, any Lender or any Issuer shall have liability to the Borrower (whether sounding in tort, contract or otherwise) for losses suffered by the Bor rower in connection with, arising out of, or in any way related to, the transactions contemplated and the relationship established by the Loan Documents, or any act, omission or event occurring in connection therewith, unless it is determined in a final non-appealable judgment by a court of competent jurisdiction that such losses resulted from the gross negligence or willful misconduct of the party from which recovery is sought. None of either Agent, either Arranger, the Swing Line Lender, any Lender, any Issuer or any Related Party of any of the foregoing Persons shall have any liability with respect to, and the Borrower hereby waives, releases and agrees not to sue for, any special, indirect, consequential or punitive damages suffered by the Borrower in connection with, arising out of, or in any way related to the Loan Documents or the transactions contemplated thereby. None of either Agent, either Arranger, the Swing Line Lender, any Lender, any Issuer or any Related Party of any of the foregoing Persons shall be liable for any damages arising from the use by unintended recipients of any information or other materials distributed by it through telecommunications, electronic or other information transmission syste

9.11 <u>Limited Disclosure</u>.

- (a) Neither the Administrative Agent nor any Lender may disclose to any Person any Specified Information (as defined below) except (i) to its, and its Affiliates', officers, employees, agents, accountants, legal counsel, advisors and other representatives who have a need to know such Specified Information or any direct, indirect, actual or prospective counterparty (and its advisor) to any swap, derivative or securitization transaction related to the Obligations or to any credit insurance provider relating to the Obligations (it being understood that the Persons to whom such disclosure is made will be informed of the confidential nature of such Specified Information and instructed to keep such Specified Information confidential, and shall agree to do so) or (ii) with the Borrower's prior consent. "Specified Information" means information that the Borrower furnishes to the Administrative Agent or any Lender that is designated in writing as confidential, but does not include any such information that is or becomes generally available to the public or that is or becomes available to the Administrative Agent or such Lender from a source other than the Borrower.
- (b) The provisions of <u>clause (a)</u> above shall not apply to Specified Information (i) that is a matter of general public knowledge or has heretofore been or is hereafter published in any

77

source generally available to the public, (ii) that is required to be disclosed by law, regulation or subpoena, judicial order or other legal process, (iii) that is requested by any regulatory body (including any self-regulatory authority) with jurisdiction over the Administrative Agent or any Lender, or (iv) that is disclosed (A) to legal counsel, accountants and other professional advisors to such Lender, (B) in connection with the exercise of any right or remedy hereunder or under any Note or any suit or other litigation or proceeding relating to this Agreement or any Note, (C) to a rating agency if required by such agency in connection with a rating relating to Credit Extensions hereunder or (D) to assignees or participants or potential assignees or participants who agree to be bound by the provisions of this Section 9.11.

9.12 <u>USA PATRIOT ACT NOTIFICATION</u>.

The following notification is provided to the Borrower pursuant to Section 326 of the USA Patriot Act of 2001, 31 U.S.C. Section 5318: IMPORTANT INFORMATION ABOUT PROCEDURES FOR OPENING A NEW ACCOUNT. To help the government fight the funding of terrorism and money laundering activities, Federal law requires all financial institutions to obtain, verify and record information that identifies each person or entity that opens an account, including any deposit account, treasury management account, loan, other extension of credit or other financial services product. What this means for the Borrower: When the Borrower opens an account, the Lenders will ask for the Borrower's name, tax identification number, business address and other information that will allow the Administrative Agent and the Lenders to identify the Borrower. The Administrative Agent and the Lenders may also ask to see the Borrower's legal organizational documents or other identifying documents.

9.13 <u>Nonreliance</u>.

Each Lender hereby represents that it is not relying on or looking to any margin stock (as defined in Regulation U of the FRB) for the repayment of the Loans provided for herein.

9.14 No Advisory or Fiduciary Responsibility.

In connection with all aspects of each transaction contemplated hereby, the Borrower acknowledges and agrees, and acknowledges its Affiliates' understanding, that: (a) the credit facility provided for hereunder and any related arranging or other services in connection therewith (including in connection with any amendment, waiver or other modification hereof or of any other Loan Document) are an arm's-length commercial transaction between the Borrower

and its Affiliates, on the one hand, and the Agents and the Arrangers, on the other hand, and the Borrower is capable of evaluating and understanding and understands and accepts the terms, risks and conditions of the transactions contemplated hereby and by the other Loan Documents (including any amendment, waiver or other modification hereof or thereof); (b) in connection with the process leading to such transaction, each Agent and Arranger is and has been acting solely as a principal and is not the financial advisor, agent or fiduciary, for the Borrower or any of its Affiliates, stockholders, creditors or employees or any other Person; (c) no Agent or Arranger has assumed or will assume an advisory, agency or fiduciary responsibility in favor of the Borrower with respect to any of the transactions contemplated hereby or the process leading thereto, including with respect to any amendment, waiver or other modification hereof or of any

78

other Loan Document (irrespective of whether any Agent or Arranger has advised or is currently advising the Borrower or any of its Affiliates on other matters) and no Agent or Arranger has any obligation to the Borrower or any of its Affiliates with respect to the transactions contemplated hereby except those obligations expressly set forth herein and in the other Loan Documents; (d) the Agents and the Arrangers and their respective Affiliates may be engaged in a broad range of transactions that involve interests that differ from those of the Borrower and its Affiliates, and no Agent or Arranger has any obligation to disclose any of such interests by virtue of any advisory, agency or fiduciary relationship; and (e) the Agents and the Arrangers have not provided and will not provide any legal, accounting, regulatory or tax advice with respect to any of the transactions contemplated hereby (including any amendment, waiver or other modification hereof or of any other Loan Document) and the Borrower has consulted its own legal, accounting, regulatory and tax advisors to the extent it has deemed appropriate. The Borrower hereby waives and releases, to the fullest extent permitted by law, any claims that it may have against any Agent and any Arranger with respect to any breach or alleged breach of agency or fiduciary duty.

ARTICLE X

THE ADMINISTRATIVE AGENT

10.1 Appointment and Authority.

Each of the Lenders and the Issuers hereby irrevocably appoints Bank of America to act on its behalf as the Administrative Agent hereunder and under the other Loan Documents and authorizes the Administrative Agent to take such actions on its behalf and to exercise such powers as are delegated to the Administrative Agent by the terms hereof or thereof, together with such actions and powers as are reasonably incidental thereto. The provisions of this Article are solely for the benefit of the Administrative Agent, the Lenders and the Issuers, and the Borrower shall have no rights as a third party beneficiary of any of such provisions.

10.2 Rights as a Lender.

The Person serving as the Administrative Agent hereunder shall have the same rights and powers in its capacity as a Lender as any other Lender and may exercise the same as though it were not the Administrative Agent and the term "Lender" or "Lenders" shall, unless otherwise expressly indicated or unless the context otherwise requires, include the Person serving as the Administrative Agent hereunder in its individual capacity. Such Person and its Affiliates may accept deposits from, lend money to, act as the financial advisor or in any other advisory capacity for and generally engage in any kind of business with the Borrower or any Subsidiary or other Affiliate thereof as if such Person were not the Administrative Agent hereunder and without any duty to account therefor to the Lenders.

79

10.3 <u>Exculpatory Provisions</u>.

The Administrative Agent shall not have any duties or obligations except those expressly set forth herein and in the other Loan Documents. Without limiting the generality of the foregoing, the Administrative Agent:

- (a) shall not be subject to any fiduciary or other implied duties, regardless of whether a Default or Unmatured Default has occurred and is continuing;
- (b) shall not have any duty to take any discretionary action or exercise any discretionary powers, except discretionary rights and powers expressly contemplated hereby or by the other Loan Documents that the Administrative Agent is required to exercise as directed in writing by the Required Lenders (or such other number or percentage of the Lenders as shall be expressly provided for herein or in the other Loan Documents), provided that the Administrative Agent shall not be required to take any action that, in its opinion or the opinion of its counsel, may expose the Administrative Agent to liability or that is contrary to any Loan Document or applicable law; and
- (c) shall not, except as expressly set forth herein and in the other Loan Documents, have any duty to disclose, and shall not be liable for the failure to disclose, any information relating to the Borrower or any of its Affiliates that is communicated to or obtained by the Person serving as the Administrative Agent or any of its Affiliates in any capacity.

The Administrative Agent shall not be liable for any action taken or not taken by it (i) with the consent or at the request of the Required Lenders (or such other number or percentage of the Lenders as shall be necessary, or as the Administrative Agent shall believe in good faith shall be necessary, under the circumstances as provided in <u>Sections 8.1</u> and <u>8.2</u>) or (ii) in the absence of its own gross negligence or willful misconduct. The Administrative Agent shall be deemed not to have knowledge of any Default or Unmatured Default unless and until notice describing such Default or Unmatured Default is given to the Administrative Agent by the Borrower, a Lender or an Issuer.

The Administrative Agent shall not be responsible for or have any duty to ascertain or inquire into (i) any statement, warranty or representation made in or in connection with this Agreement or any other Loan Document, (ii) the contents of any certificate, report or other document delivered hereunder or thereunder or in connection herewith or therewith, (iii) the performance or observance of any of the covenants, agreements or other terms or conditions set forth herein or therein or the occurrence of any Default or Unmatured Default, (iv) the validity, enforceability, effectiveness or genuineness of this Agreement, any other Loan Document or any other agreement, instrument or document or (v) the satisfaction of any condition set forth in Article IV or elsewhere he rein, other than to confirm receipt of items expressly required to be delivered to the Administrative Agent.

80

10.4 Reliance by Administrative Agent.

The Administrative Agent shall be entitled to rely upon, and shall not incur any liability for relying upon, any notice, request, certificate, consent, statement, instrument, document or other writing (including any electronic message, Internet or intranet website posting or other distribution) believed by it to be genuine and to have been signed, sent or otherwise authenticated by the proper Person. The Administrative Agent also may rely upon any statement made to it orally or by telephone and believed by it to have been made by the proper Person, and shall not incur any liability for relying thereon. In determining compliance with any condition hereunder to the making of a Loan, or the issuance of a Letter of Credit, that by its terms must be fulfilled to the satisfaction of a Lender or an Issuer, the Administrative Agent may presume that such condition is satisfactory to such Lender or such Issuer unless the Administrative Agent shall have received notice to the contrary from such Lender or such Issuer prior to the making of such Loan or the issuance of such Letter of Credit. The Administrative Agent may consult with legal counsel (who may be counsel for the Borrower), independent accountants and other experts selected by it, and shall not be liable for any action taken or not taken by it in accordance with the advice of any such counsel, accountants or experts.

10.5 <u>Delegation of Duties</u>.

The Administrative Agent may perform any and all of its duties and exercise its rights and powers hereunder or under any other Loan Document by or through any one or more sub-agents appointed by the Administrative Agent. The Administrative Agent and any such sub-agent may perform any and all of its duties and exercise its rights and powers by or through their respective Related Parties. The exculpatory provisions of this Article shall apply to any such sub-agent and to the Related Parties of the Administrative Agent and any such sub-agent, and shall apply to their respective activities in connection with the syndication of the credit facilities provided for herein as well as activities as Administrative Agent.

10.6 Resignation of Administrative Agent.

The Administrative Agent may, upon giving five (5) Business Days prior written notice to the Borrower, resign at any time by giving notice of its resignation to the Lenders, the Issuers and the Borrower. Upon receipt of any such notice of resignation, the Borrower shall have the right, with the consent of the Required Lenders, such consent not to be unreasonably withheld, conditioned or delayed, to appoint a successor, which successor shall be a Lender. If no such successor shall have been so appointed by the Borrower and shall have accepted such appointment within thirty (30) days after the retiring Administrative Agent gives notice of its resignation, then the retiring Administrative Agent may on behalf of the Lenders and the Issuers, appoint a successor Administrative Agent from among the Lenders meeting the qualifications set forth above; provided that if the Administrative Agent shall notify the Borrower and the Lenders that no qualifying Person has accepted such appointment, then such resignation shall nonetheless become effective in accordance with such notice and (1) the retiring Administrative Agent shall be discharged from its duties and obligations hereunder and under the other Loan Documents and (2) all payments, communications and determinations provided to be made by, to or through the Administrative Agent shall instead be made by or to each Lender and each Issuer directly, until such time as the Borrower appoints a successor Administrative Agent as provided for above in

81

this Section. Upon the acceptance of a successor's appointment as Administrative Agent hereunder, such successor shall succeed to and become vested with all of the rights, powers, privileges and duties of the retiring (or retired) Administrative Agent, and the retiring Administrative Agent shall be discharged from all of its duties and obligations hereunder or under the other Loan Documents (if not already discharged therefrom as provided above in this Section). The fees payable by the Borrower to a successor Administrative Agent shall be the same as those payable to its predecessor unless otherwise agreed between the Borrower and such successor. After the retiring Administrative Agent's resignation hereunder and under the other Loan Documents, the provisions of this Article and Section 9.6 shall continue in effect for the benefit of such retiring Administrative Agent, its sub-agents and their respective Related Parties in respect of any actions taken or omitted to be taken by any of them while the retiring Administrative Agent was acting as Administrative Agent.

Any resignation by Bank of America as Administrative Agent pursuant to this Section shall also constitute its resignation as an Issuer and as the Swing Line Lender. Upon the acceptance of a successor's appointment as Administrative Agent hereunder, (a) such successor shall succeed to and become vested with all of the rights, powers, privileges and duties of the retiring Issuer and the retiring Swing Line Lender, (b) the retiring Issuer and the retiring Swing Line Lender shall be discharged from all of their respective duties and obligations hereunder or under the other Loan Documents, and (c) the successor Issuer shall issue letters of credit in substitution for the Letters of Credit, if any, outstanding at the time of such succession or make other arrangements satisfactory to the retiring Issuer to effectively assu me the obligations of the retiring Issuer with respect to such Letters of Credit.

10.7 <u>Non-Reliance on Administrative Agent and Other Lenders.</u>

Each Lender, the Swing Line Lender and each Issuer acknowledges that it has, independently and without reliance upon the Administrative Agent or any other Lender or any of their Related Parties and based on such documents and information as it has deemed appropriate, made its own credit analysis and decision to enter into this Agreement. Each Lender, the Swing Line Lender and each Issuer also acknowledges that it will, independently and without reliance upon the Administrative Agent or any other Lender or any of their Related Parties and based on such documents and information as it shall from time to time deem appropriate, continue to make its own decisions in taking or not taking action under or based upon this Agreement, any other Loan Document or any related agreement or any document furnished hereunder or thereunder.

10.8 No Other Duties, Etc.

Anything herein to the contrary notwithstanding, none of the Bookrunners or Arrangers listed on the cover page hereof shall have any powers, duties or responsibilities under this Agreement or any of the other Loan Documents, except in its capacity, as applicable, as the Administrative Agent, the Swing Line Lender, a Lender or an Issuer hereunder.

10.9 Administrative Agent May File Proofs of Claim.

In case of the pendency of any proceeding under any Debtor Relief Law or other judicial proceeding relative to the Borrower, the Administrative Agent (irrespective of whether the

82

principal of any Loan or Letter of Credit Obligation shall then be due and payable as herein expressed or by declaration or otherwise and irrespective of whether the Administrative Agent shall have made any demand on any Borrower) shall be entitled and empowered, by intervention in such proceeding or otherwise

- (a) to file and prove a claim for the whole amount of the principal and interest owing and unpaid in respect of the Loans, Letter of Credit Obligations and all other Obligations that are owing and unpaid and to file such other documents as may be necessary or advisable in order to have the claims of the Lenders, the Swing Line Lender, the Issuers and the Administrative Agent (including any claim for the reasonable compensation, expenses, disbursements and advances of the Lenders, the Swing Line Lender, the Issuers and the Administrative Agent and their respective agents and counsel and all other amounts due the Lenders, the Issuers and the Administrative Agent under Sections 2.5, 2.19(d), and 9.6) allowed in such judicial proceeding; and
 - (b) to collect and receive any monies or other property payable or deliverable on any such claims and to distribute the same;

and any custodian, receiver, assignee, trustee, liquidator, sequestrator or other similar official in any such judicial proceeding is hereby authorized by each Lender, the Swing Line Lender and each Issuer to make such payments to the Administrative Agent and, in the event that the Administrative Agent shall consent to the making of such payments directly to the Swing Line Lender, the Lenders and the Issuers, to pay to the Administrative Agent any amount due for the reasonable compensation, expenses, disbursements and advances of the Administrative Agent and its agents and counsel, and any other amounts due the Administrative Agent under Sections 2.5 and 9.6.

Nothing contained herein shall be deemed to authorize the Administrative Agent to authorize or consent to or accept or adopt on behalf of any Lender, the Swing Line Lender or any Issuer any plan of reorganization, arrangement, adjustment or composition affecting the Obligations or the rights of any Lender or to authorize the Administrative Agent to vote in respect of the claim of any Lender in any such proceeding.

ARTICLE XI

SETOFF; RATABLE PAYMENTS; PAYMENTS SET ASIDE

11.1 <u>Setoff</u>.

In addition to, and without limitation of, any rights of the Lenders under applicable law, if the Borrower becomes insolvent, however evidenced, or any Default occurs, after obtaining the prior written consent of the Administrative Agent, any and all deposits (including all account balances, whether provisional or final and whether or not collected or available) and any other Indebtedness at any time held or owing by any Lender, any Issuer, the Swing Line Lender or any such Affiliate to or for the credit or account of the Borrower may be offset and applied toward the payment of the Obligations owing to such Lender, the Issuer or the Swing Line Lender

83

whether or not the Obligations, or any part hereof, shall then be due; <u>provided</u>, that in the event that any Defaulting Lender shall exercise any such right of setoff, (x) all amounts so set off shall be paid over immediately to the Administrative Agent for further application in accordance with the provisions of <u>Section 2.23</u> and, pending such payment, shall be segregated by such Defaulting Lender from its other funds and deemed held in trust for the benefit of the Administrative Agent and the Lenders, and (y) the Defaulting Lender shall provide promptly to the Administrative Agent a statement describing in reasonable detail the Obligations owing to such Defaulting Lender as to which it exercised such right of s etoff. Each Lender, each Issuer and the Swing Line Lender agree to notify the Borrower and the Administrative Agent promptly after any such setoff and application; <u>provided</u> that the failure to give such notice shall not affect the validity of such setoff and application.

11.2 Ratable Payments.

If any Lender, whether by setoff or otherwise, has payment made to it upon its Outstanding Credit Exposure (other than payments received pursuant to Section 3.1, 3.2, 3.4 or 3.5 and payments made to any Issuer in respect of Reimbursement Obligations so long as the Lenders have not funded their participations therein) in a greater proportion than that received by any other Lender, such Lender agrees, promptly upon demand, to purchase a portion of the Aggregate Outstanding Credit Exposure held by the other Lenders so that after such purchase each Lender will hold its Pro Rata Sha re of the Aggregate Outstanding Credit Exposure. If any Lender, whether in connection with setoff or amounts which might be subject to setoff or otherwise, receives collateral or other protection for its Obligations or such amounts which may be subject to setoff, such Lender agrees, promptly upon demand, to take such action necessary such that all Lenders share in the benefits of such collateral ratably in accordance with their respective Pro Rata Shares. In case any such payment

is disturbed by legal process, or otherwise, appropriate further adjustments shall be made. For the avoidance of doubt, the provisions of this Section shall not be construed to apply to (A) any payment made by or on behalf of the Borrower pursuant to and in accordance with the express terms of this Agreement (including the application of funds arising from the existence of a Defaulting Lender), (B) the application of Cash Collateral provided for in Section 2.22, or (C) any payment obtained by a Lender as consideration for the assignment of or sale of a participation in any of its Loans or subparticipations in Letter of Credit Obligations or Swing Line Loans to any assignee or participant, other than an assignment to the Borrower or any Subsidiary thereof (as to which the provisions of this Section shall apply).

11.3 Payments Set Aside.

To the extent that any payment by or on behalf of the Borrower is made to the Administrative Agent, any Issuer or any Lender, or the Administrative Agent, any Issuer or any Lender exercises its right of setoff, and such payment or the proceeds of such setoff or any part thereof is subsequently invalidated, declared to be fraudulent or preferential, set aside or required (including pursuant to any settlement entered into by the Administrative Agent, such Issuer or such Lender in its discretion) to be repaid to a trustee, receiver or any other party, in connection with any proceeding under any Debtor Relief Law or otherwise, then (a) to the extent of such recovery, the obligation or part thereof originally intended to be satisfied shall be revived and

84

continued in full force and effect as if such payment had not been made or such setoff had not occurred, and (b) each Lender and each Issuer severally agrees to pay to the Administrative Agent upon demand its applicable share (without duplication) of any amount so recovered from or repaid by the Administrative Agent, plus interest thereon from the date of such demand to the date such payment is made at a rate per annum equal to the Federal Funds Rate from time to time in effect. The obligations of the Lenders and the Issuers under <u>clause (b)</u> of the preceding sentence shall survive the payment in full of the Obligations and the termination of this Agreement.

ARTICLE XII

BENEFIT OF AGREEMENT; ASSIGNMENTS; PARTICIPATIONS

12.1 Successors and Assigns.

- (a) The provisions of this Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns permitted hereby, except that the Borrower may not assign or otherwise transfer any of its rights or obligations hereunder without the prior written consent of the Administrative Agent, the Swing Line Lender and each Lender and no Lender may assign or otherwise transfer any of its rights or obligations hereunder except (i) to an Eligible Assignee in accordance with the provisions of subsection (b) of this Section, (ii) by way of participation in accordance with the provisions of subsection (d) of this Section, or (iii) by way of pledge or assignment of a security interest subject to the restrictions of subsection (f) of this Section (and any other attempted assignment or transfer by any party hereto shall be null and void). Nothing in this Agreement, expressed or implied, shall be construed to confer upon any Person (other than the parties hereto, their respective successors and assigns permitted hereby, Participants to the extent provided in subsection (d) of this Section and, to the extent expressly contemplated hereby, the Related Parties of each of the Administrative Agent, the Issuers and the Lenders) any legal or equitable right, remedy or claim under or by reason of this Agreement.
- (b) Any Lender may at any time assign to one or more Eligible Assignees all or a portion of its rights and obligations under this Agreement (including all or a portion of its Commitment and the Loans (including for purposes of this <u>subsection (b)</u> participations in Letter of Credit Obligations) at the time owing to it); <u>provided</u> that
 - (i) except in the case of an assignment of the entire remaining amount of the assigning Lender's Commitment and the Loans at the time owing to it or in the case of an assignment to a Lender or an Affiliate of a Lender or an Approved Fund with respect to a Lender, the aggregate amount of the Commitment (which for this purpose includes Loans outstanding thereunder) or, if the Commitment is not then in effect, the principal outstanding balance of the Loans of the assigning Lender subject to each such assignment, determined as of the date the Assignment Agreement with respect to such assignment is delivered to the Administrative Agent or, if "Trade Date" is specified in the Assignment Agreement, as of the Trade Date, shall not be less than \$5,000,000 unless

85

each of the Administrative Agent and, so long as no Default has occurred and is continuing, the Borrower otherwise consents (each such consent not to be unreasonably withheld or delayed); <u>provided</u>, <u>however</u>, that concurrent assignments to members of an Assignee Group and concurrent assignments from members of an Assignee Group to a single Eligible Assignee (or to an Eligible Assignee and members of its Assignee Group) will be treated as a single assignment for purposes of determining whether such minimum amount has been met; <u>provided</u>, <u>further</u>, that the Borrower shall be deemed to have consented to any such assignment unless it shall object thereto by written notice to the Administrative Agent within ten (10) Business Days after having received notice thereof.

- (ii) each partial assignment shall be made as an assignment of a proportionate part of all the assigning Lender's rights and obligations under this Agreement with respect to the Loans or the Commitment assigned;
- (iii) any assignment of a Commitment must be approved by the Administrative Agent, the Swing Line Lender and the Issuers (each such approval not to be unreasonably withheld or delayed) unless the Person that is the proposed assignee is itself a Lender, an Affiliate of such Lender or an Approved Fund with respect to such Lender;

- (iv) (A) such Lender shall at the same time enter into an Assignment Agreement (as defined in the KCPL Credit Agreement) with the same Eligible Assignee(s) in an amount representing an equal proportion of such Lender's Commitment (as defined in the KCPL Credit Agreement) under the KCPL Credit Agreement and (B) such Lender shall at the same time enter into an Assignment Agreement (as defined in the KCPL GMO Credit Agreement) with the same Eligible Assignee(s) in an amount representing an equal proportion of such Lender's Commitment (as defined in the KCPL GMO Credit Agreement) under the KCPL GMO Credit Agreement;
- (v) the parties to each assignment shall execute and deliver to the Administrative Agent an Assignment Agreement, together with a processing and recordation fee of \$3,500; provided, however, that (A) such processing and recordation fee shall not apply to assignments to Affiliates of the assigning Lender, (B) such processing and recordation fee shall, with respect to a particular assignment, be payable only once by such assigning Lender in connection with the concurrent assignments of Commitments of such assigning Lender required by Section 12.1(b)(iv) and (C) the Administrative Agent may, in its sole discretion, elect to waive such processing and recordation fee in the case of any assignment. The Eligible Assignee, if it shall not be a Lender, shall deliver to the Administrative Agent an Administrative Questionnaire;
- (vi) No such assignment shall be made (A) to the Borrower or any of the Borrower's Affiliates or Subsidiaries, (B) to any Defaulting Lender or any of its Subsidiaries, or any Person who upon becoming a Lender hereunder would constitute any of the foregoing Persons described in this <u>clause (B)</u>, or (C) to a natural person; and
- (vii) In connection with any assignment of rights and obligations of any Defaulting Lender hereunder, no such assignment shall be effective unless and until, in

86

addition to the other conditions thereto set forth herein, the parties to the assignment shall make such additional payments to the Administrative Agent in an aggregate amount sufficient, upon distribution thereof as appropriate (which may be outright payment, purchases by the assignee of participations or subparticipations, or other compensating actions, including funding, with the consent of the Borrower and the Administrative Agent, the applicable pro rata share of Loans previously requested but not funded by the Defaulting Lender, to each of which the applicable assignee and assignor hereby irrevocably consent), to (x) pay and satisfy in full all payment liabilities then owed by such Defaulting Lender to the Administrative Agent or any Lender hereunder (and interest accrued thereon) and (y) acquire (and fund as appropriate) its full pro rata share of all Loans and participations in Letters of Credit and Swing Line Loans in accordance with its Pro Rata Share. Notwithstanding the foregoing, in the event that any assignment of rights and obligations of any Defaulting Lender hereunder shall become effective under applicable Law without compliance with the provisions of this paragraph, then the assignee of such interest shall be deemed to be a Defaulting Lender for all purposes of this Agreement until such compliance occurs.

Subject to acceptance and recording thereof by the Administrative Agent pursuant to <u>subsection (c)</u> of this Section, from and after the effective date specified in each Assignment Agreement, the Eligible Assignee thereunder shall be a party to this Agreement and, to the extent of the interest assigned by such Assignment Agreement, have the rights and obligations of a Lender under this Agreement, and the assigning Lender thereunder shall, to the extent of the interest assigned by such Assignment Agreement, be released from its obligations under this Agreement (and, in the case of an Assignment Agreement covering all of the assigning Lender's rights and obligations under this Agreement, such Lender shall cease to be a party hereto) but shall continue to be entitled to the benefits of Sections 3.1, 3.4, 3.5, and 9.6 with respect to facts and circumstances occurring prior to the effective date of such assignment. Upon request, the Borrower (at its expense) shall execute and deliver a Note to the assignee Lender. Any assignment or transfer by a Lender of rights or obligations under this Agreement that does not comply with this subsection shall be treated for purposes of this Agreement as a sale by such Lender of a participation in such rights and obligations in accordance with <u>subsection (d)</u> of this Section.

(c) The Administrative Agent, acting solely for this purpose as an agent of the Borrower, shall maintain at the Administrative Agent's Office a copy of each Assignment Agreement delivered to it and a register for the recordation of the names and addresses of the Lenders, and the Commitments of, and principal amounts of the Loans and Letter of Credit Obligations owing to, each Lender pursuant to the terms hereof from time to time (the "Register"). The entries in the Register shall be conclusive, and the Borrower, the Administrative Agent and the Lenders may treat each Person whose name is recorded in the Register pursuant to the terms hereof as a Lender hereunder for all purposes of this Agreement, notwithstanding notice to the contrary. In addition, the Administrative Agent shall maintain on the Register information regarding the designation, and revocation of designation, of any Lender as a Defaulting Lender. The Register shall be available for inspection by each of the Borrower and the Issuers at any reasonable time and from time to time upon reasonable prior notice. In addition, at any time that a request for a consent for a material or substantive change to the Loan

87

Documents is pending, any Lender may request and receive from the Administrative Agent a copy of the Register.

(d) Any Lender may at any time, without the consent of, or notice to, the Borrower or the Administrative Agent, sell participations to any Person (other than a natural person, a Defaulting Lender, or the Borrower or any of the Borrower's Affiliates or Subsidiaries) (each, a "Participant") in all or a portion of such Lender's rights and/or obligations under this Agreement (including all or a portion of its Commitment and/or the Loans (including such Lender's participations in Letter of Credit Obligations) owing to it); provided that (i) such Lender's obligations under this Agreement shall remain unchanged, (ii) such Lender shall remain solely responsible to the other parties hereto for the performance of such obligations and (iii) the Borrower, the Administrative Agent, the Lenders and the Issuers shall continue to deal solely and directly with such Lender in connection with such Lender's rights and obligations under this Agreement.

Any agreement or instrument pursuant to which a Lender sells such a participation shall provide that such Lender shall retain the sole right to enforce this Agreement and to approve any amendment, modification or waiver of any provision of this Agreement; <u>provided</u> that such agreement or instrument may provide that such Lender will not, without the consent of the Participant, agree to any amendment, waiver or other modification described in the proviso to <u>Section 8.2</u> that affects such Participant. Subject to <u>subsection (e)</u> of this Section, the Borrower agrees that each Participant shall be entitled to

the benefits of <u>Sections 3.1, 3.4</u> and <u>3.5</u> to the same extent as if it were a Lender and had acquired its interest by assignment pursuant to <u>subsection (b)</u> of this Section. To the extent permitted by law, each Participant also shall be entitled to the benefits of <u>Section 11.1</u> as though it were a Lender, <u>provided</u> such Participant agrees to be subject to <u>Section 11.2</u> as though it were a Lender.

- (e) A Participant shall not be entitled to receive any greater payment under <u>Section 3.1</u> or <u>3.5</u> than the applicable Lender would have been entitled to receive with respect to the participation sold to such Participant, unless the sale of the participation to such Participant is made with the Borrower's prior written consent. A Participant that would be a Foreign Lender if it were a Lender shall not be entitled to the benefits of <u>Section 3.5</u> unless the Borrower is notified of the participation so ld to such Participant and such Participant agrees, for the benefit of the Borrower, to comply with <u>Section 3.5(e)</u> as though it were a Lender.
- (f) Any Lender may at any time pledge or assign a security interest in all or any portion of its rights under this Agreement (including under its Note, if any) to secure obligations of such Lender, including any pledge or assignment to secure obligations to a Federal Reserve Bank; provided that no such pledge or assignment shall release such Lender from any of its obligations hereunder or substitute any such pledgee or assignee for such Lender as a party hereto.
- (g) The words "execution," "signed," "signature," and words of like import in any Assignment Agreement shall be deemed to include electronic signatures or the keeping of records in electronic form, each of which shall be of the same legal effect, validity or enforceability as a manually executed signature or the use of a paper-based recordkeeping

88

system, as the case may be, to the extent and as provided for in any applicable law, including the Federal Electronic Signatures in Global and National Commerce Act, the New York State Electronic Signatures and Records Act, or any other similar state laws based on the Uniform Electronic Transactions Act.

(h) Notwithstanding anything to the contrary contained herein, if at any time Bank of America assigns all of its Commitment and Loans pursuant to <u>subsection (b)</u> above, Bank of America may, upon thirty (30) days' notice to the Borrower and the Lenders, resign as an Issuer. In the event of any such resignation as an Issuer, the Borrower shall be entitled to appoint from among the Lenders another Issuer hereunder who agrees to such appointment; <u>provided</u>, <u>however</u>, that no failure by the Borrower to appoint any such Issuer, or the failure of any Lender to accept su ch appointment, shall affect the resignation of Bank of America as an Issuer. If Bank of America resigns as an Issuer, it shall retain all the rights, powers, privileges and duties of an Issuer hereunder with respect to all Letters of Credit outstanding as of the effective date of its resignation as Issuer and all Letter of Credit Obligations with respect thereto (including the right to require the Lenders to fund risk participations pursuant to <u>Section 2.19(e)</u>).

12.2 Replacement of Lenders.

If (i) any Lender requests compensation under Section 3.1 or Section 3.2, (ii) suspends the availability of Eurodollar Loans under Section 3.3, (iii) the Borrower is required to pay any additional amount to any Lender or any Governmental Authority for the account of any Lender pursuant to Section 3.4 or Section 3.5, (iv) any Lender is a Non-Extending Lender pursuant to Section 2.21(b) or (v) any Lender is a Defaulting Lender, then the Borrower may, at its sole expense and effort, upon notice to such Lender and the Administrative Agent, require such Lender to assign and delegate, without recourse (in accordance with and subject to the restrictions contained in, and consents required by, Section 12.1(b)), all of its interests, rights and obligations under this Agreement and the related Loan Documents to an Eligible Assignee that shall assume such obligations (which Eligible Assignee may be another Lender, if a Lender accepts such assignment), provided that:

- (a) the Borrower shall have paid to the Administrative Agent the assignment fee specified in <u>Section 12.1(b)</u>;
- (b) such Lender shall have received payment of an amount equal to the outstanding principal of its Loans and participations, accrued interest thereon, accrued fees and all other amounts payable to it hereunder and under the other Loan Documents (including any amounts payable under Section 3.4) from such Eligible Assignee (to the extent of such outstanding principal and accrued interest and fees) or the Borrower (in the case of all other amounts);
- (c) in the case of any such assignment resulting from a claim for compensation under <u>Section 3.1</u> or payments required to be made pursuant to <u>Section 3.4</u>, such assignment will result in a reduction in such compensation or payments thereafter; and

89

(d) such assignment does not conflict with applicable Laws.

A Lender shall not be required to make any such assignment or delegation if, prior thereto, as a result of a waiver by such Lender or otherwise, the circumstances entitling the Borrower to require such assignment and delegation cease to apply.

ARTICLE XIII

NOTICES

13.1 Notices.

Except as otherwise permitted by <u>Section 2.14</u> with respect to borrowing notices, all notices, requests and other communications to any party hereunder shall be in writing (including electronic transmission, facsimile transmission or similar writing) and shall be given to such party: (i) in the case of

the Borrower or the Administrative Agent, at its address or facsimile number set forth on Schedule IV, (ii) in the case of any Lender, at its address or facsimile number specified in its Administrative Questionnaire or (iii) in the case of any party, at such other address or facsimile number as such party may hereafter specify for the purpose by notice to the Administrative Agent and the Borrower in accordance with the provisions of this Section 13.1. Each such notice, request or other communication shall be effective (a) if given by facsimile transmission, when transmitted to the facsimile number specified in this Section and confirmation of receipt is received, (b) if given by mail, when received, or (c) if given by any other means, when delivered (or, in the case of electronic transmission, received) at the address specified in this Section; provided that notices to the Administrative Agent under Article II shall not be effective until received.

13.2 Change of Address.

The Borrower, the Administrative Agent and any Lender may each change the address for service of notice upon it by a notice in writing to the other parties hereto.

13.3 The Platform.

THE PLATFORM IS PROVIDED "AS IS" AND "AS AVAILABLE." THE AGENT PARTIES (AS DEFINED BELOW) DO NOT WARRANT THE ACCURACY OR COMPLETENESS OF THE BORROWER MATERIALS OR THE ADEQUACY OF THE PLATFORM, AND EXPRESSLY DISCLAIM LIABILITY FOR ERRORS IN OR OMISSIONS FROM THE BORROWER MATERIALS. NO WARRANTY OF ANY KIND, EXPRESS, IMPLIED OR STATUTORY, INCLUDING ANY WARRANTY OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE, NON INFRINGEMENT OF THIRD PARTY RIGHTS OR FREEDOM FROM VIRUSES OR OTHER CODE DEFECTS, IS MADE BY ANY AGENT PARTY IN CONNECTION WITH THE BORROWER MATERIALS OR THE PLATFORM. In no event shall the Administrative Agent or any of its Related Parties (collectively, the "Agent Parties") have any liability to Borrower,

90

any Lender, any Issuer or any other Person for losses, claims, damages, liabilities or expenses of any kind (whether in tort, contract or otherwise) arising out of Borrower's or the Administrative Agent's transmission of Borrower Materials through the Internet, except to the extent that such losses, claims, damages, liabilities or expenses are determined by a court of competent jurisdiction by a final and nonappealable judgment to have resulted from the gross negligence or willful misconduct of such Agent Party; <u>provided</u>, <u>however</u>, that in no event shall any Agent Party have any liability to Borrower, any Lender, any Issuer or any other Person for indirect, special, incidental, consequential or punit ive damages (as opposed to direct or actual damages).

ARTICLE XIV

COUNTERPARTS

This Agreement may be executed in any number of counterparts, all of which taken together shall constitute one agreement, and any of the parties hereto may execute this Agreement by signing any such counterpart. This Agreement shall be effective when it has been executed by the Borrower, the Administrative Agent and the Lenders and each party has notified the Administrative Agent by facsimile transmission or telephone that it has taken such action.

ARTICLE XV

OTHER AGENTS

No Lender identified on the cover page, the signature pages or otherwise in this Agreement, or in any document related hereto, as being the "Syndication Agent" or a "Co-Documentation Agent" shall have any right, power, obligation, liability, responsibility or duty under this Agreement in such capacity other than those applicable to all Lenders. Each Lender acknowledges that it has not relied, and will not rely, on the Syndication Agent or any Co-Documentation Agent in deciding to enter into this Agreement or in taking or refraining from taking any action hereunder or pursuant hereto.

ARTICLE XVI

CHOICE OF LAW; CONSENT TO JURISDICTION; WAIVER OF JURY TRIAL.

16.1 <u>CHOICE OF LAW</u>.

THIS AGREEMENT SHALL BE CONSTRUED IN ACCORDANCE WITH THE INTERNAL LAWS (AND NOT THE LAW OF CONFLICTS) OF THE STATE OF NEW YORK, BUT GIVING EFFECT TO FEDERAL LAWS APPLICABLE TO NATIONAL BANKS.

91

16.2 <u>CONSENT TO JURISDICTION</u>.

THE BORROWER, THE ADMINISTRATIVE AGENT, THE SWING LINE LENDER, EACH LENDER AND EACH ISSUER HEREBY IRREVOCABLY SUBMIT TO THE NON-EXCLUSIVE JURISDICTION OF ANY UNITED STATES FEDERAL OR NEW YORK STATE COURT

SITTING IN NEW YORK CITY, NEW YORK IN ANY ACTION OR PROCEEDING ARISING OUT OF OR RELATING TO THIS AGREEMENT AND THE BORROWER, THE ADMINISTRATIVE AGENT, THE SWING LINE LENDER, EACH LENDER AND EACH ISSUER HEREBY IRREVOCABLY AGREE THAT ALL CLAIMS IN RESPECT OF SUCH ACTION OR PROCEEDING MAY BE HEARD AND DETERMINED IN ANY SUCH COURT AND IRREVOCABLY WAIVE ANY OBJECTION THEY MAY NOW OR HEREAFTER HAVE AS TO THE VENUE OF ANY SUCH SUIT, ACTION OR PROCEEDING BROUGHT IN SUCH A COURT OR THAT SUCH COURT IS AN INCONVENIENT FORUM. NOTHING HEREIN SHALL LIMIT THE RIGHT OF ANY PARTY TO BRING PROCEEDINGS AGAINST THE OTHER PARTIES IN THE COURTS OF ANY OTHER JURISDICTION. ANY JUDICIAL PROCEEDING INVOLVING, DIRECTLY OR INDIRECTLY, ANY MATTER IN ANY WAY ARISING OUT OF, RELATED TO, OR CONNECTED WITH THIS AGREEMENT SHALL BE BROUGHT ONLY IN A COURT IN NEW YORK CITY, NEW YORK.

16.3 WAIVER OF JURY TRIAL.

THE BORROWER, THE ADMINISTRATIVE AGENT, THE SWING LINE LENDER, EACH LENDER AND EACH ISSUER HEREBY WAIVE TRIAL BY JURY IN ANY JUDICIAL PROCEEDING INVOLVING, DIRECTLY OR INDIRECTLY, ANY MATTER (WHETHER SOUNDING IN TORT, CONTRACT OR OTHERWISE) IN ANY WAY ARISING OUT OF, RELATED TO, OR CONNECTED WITH ANY LOAN DOCUMENT OR THE RELATIONSHIP ESTABLISHED THEREUNDER.

ARTICLE XVII

TERMINATION OF EXISTING CREDIT FACILITY

Lenders which are parties to the Existing Credit Facility (and which constitute "Required Lenders" under and as defined in the Existing Credit Facility) hereby waive any advance notice requirement for terminating the commitments under the Existing Credit Facility, and the Borrower and the applicable Lenders agree that the Existing Credit Facility and the commitments thereunder shall be terminated on the date hereof (except for any provisions thereof which by their terms survive termination thereof).

92

IN WITNESS WHEREOF, the Borrower, the Lenders, the Swing Line Lender, the Issuers and the Agents have executed this Agreement as of the date first above written.

GREAT PLAINS ENERGY INCORPORATED, as the Borrower

By: /s/ Michael W. Cline Name: Michael W. Cline

Title: Vice President – Investor Relations and

Treasurer and Corporate Secretary

BANK OF AMERICA, N.A., as Administrative Agent

By: /s/ Kimberly Williams
Name: Kimberly Williams
Title: Vice President

BANK OF AMERICA, N.A., as an Issuer, as Swing Line Lender and as a Lender

By: /s/ Kevin P. Bertelsen
Name: Kevin P. Bertelsen
Title: Senior Vice President

THE BANK OF TOKYO-MITSUBISHI UFJ, LTD., as an Issuer and as a Lender

By: /s/ Bradford Joyce
Name: Bradford Joyce
Title: Authorized Signatory

WELLS FARGO BANK, NATIONAL ASSOCIATION, as Syndication Agent, as an Issuer and as a Lender

By: /s/ Allison Newman Name: Allison Newman Title: Vice President

UNION BANK, N.A., as Syndication Agent and as a Lender

By: /s/ Susan K. Johnson Name: Susan K. Johnson Title: Vice President

JPMORGAN CHASE BANK, N.A., as a Lender

By: /s/ Nancy R. Barwig
Name: Nancy R. Barwig
Title: Senior Vice President

BNP PARIBAS, as a Lender

By: /s/ Pasquale A. Perraglia IV Name: Pasquale A. Perraglia IV

Title: Vice President

By: /s/ Denis O'Meara Name: Denis O'Meara Title: Managing Director

THE BANK OF NOVA SCOTIA, as a Lender

By: /s/ Thane Rattew
Name: Thane Rattew
Title: Managing Director

THE ROYAL BANK OF SCOTLAND PLC, as a Lender

By: /s/ Tyler J. McCarthy Name: Tyler J. McCarthy

Title: Director

BARCLAYS BANK PLC, as a Lender

By: /s/ David Barton Name: David Barton Title: Director

U.S. BANK NATIONAL ASSOCIATION as a Lender

By: /s/ Eric J. Cosgrove Name: Eric J. Cosgrove Title: Vice President

THE BANK OF NEW YORK MELLON, as a Lender

/s/ Hussam S. Alsahlani By: Hussam S. Alsahlani Name: Title: Senior Associate

KEYBANK NATIONAL ASSOCIATION, as a Lender

By: /s/ Keven D. Smith Name: Keven D. Smith Title: Senior Vice President

SUNTRUST BANK, as a Lender

By: /s/ Andrew Johnson Andrew Johnson Name:

Title: Director

DEUTSCHE BANK AG NEW YORK BRANCH, as a Lender

By: /s/ Frederick Laird Name: Frederick Laird Title: Managing Director

By: /s/ Heidi Sandquist Name: Heidi Sandquist Title: Director

UMB BANK, N.A., as a Lender

By: /s/ Robert P. Elbert Name: Robert P. Elbert Title: Senior Vice President COMMERCE BANK, N.A., as a Lender

By: /s/ Aaron M. Siders Name: Title: Aaron M. Siders Commercial Banking Officer

SCHEDULE I

COMMITMENTS

Lender	Commitment
Bank of America, N.A.	\$19,200,000.00
Wells Fargo Bank, National Association	\$19,200,000.00
The Bank of Tokyo-Mitsubishi UFJ, Ltd.	\$9,600,000.00
Union Bank, N.A.	\$9,600,000.00
JPMorgan Chase Bank, N.A.	\$13,600,000.00
BNP Paribas	\$13,600,000.00
The Bank of Nova Scotia	\$13,600,000.00
The Royal Bank of Scotland plc	\$13,600,000.00
Barclays Bank PLC	\$13,600,000.00
U.S. Bank National Association	\$13,600,000.00
The Bank of New York Mellon	\$12,000,000.00
KeyBank National Association	\$12,000,000.00
SunTrust Bank	\$12,000,000.00
Deutsche Bank AG New York Branch	\$12,000,000.00
UMB Bank, n.a.	\$8,000,000.00
Commerce Bank, N.A.	\$4,800,000.00
Total	\$200,000,000.00

LETTER OF CREDIT COMMITMENTS

Issuer	Letter of Credit Commitment
Bank of America, N.A.	\$16,666,666.67
Wells Fargo Bank, National Association	\$16,666,666.67
The Bank of Tokyo-Mitsubishi UFJ, Ltd.	\$16,666,666.66
Total	\$50,000,000.00

SCHEDULE II

EXISTING LETTERS OF CREDIT

Issuer	Beneficiary	Amount	Expiry Date	Letter of Credit Number
The Bank of Tokyo-Mitsubishi UFJ, Ltd.	Lumbermens Mutual Casualty Company	\$1,840,000.00	08/11/10	S-015179
The Bank of Tokyo-Mitsubishi UFJ, Ltd.	ACE American Insurance Company	\$ 7,210,102.00	08/11/10	S-015180
The Bank of Tokyo-Mitsubishi UFJ, Ltd.	Fidelity and Deposit Company of Maryland	\$ 6,730,000.00	08/11/10	S-015229

SCHEDULE III

PRICING SCHEDULE

	>A-/A3	>BBB+/Baa1	>BBB/Baa2	>BBB-/Baa3	>BB+/Ba1	<bb ba2<="" th=""></bb>
	LEVEL I STATUS	Level II Status	Level III Status	LEVEL IV STATUS	Level V	LEVEL VI STATUS
Pricing					Status	
Applicable Margin for						
Eurodollar Advances and Letter	1.75%	2.00%	2.25%	2.75%	3.00%	3.25%
of Credit Fee Rate						
Commitment Fee Rate	0.225%	0.30%	0.375%	0.50%	0.625%	0.625%
Applicable Margin for Floating Rate Advances	0.75%	1.00%	1.25%	1.75%	2.00%	2.25%

"Level I Status" exists at any date if, on such date, the Borrower's Moody's Rating is A3 or better or the Borrower's S&P Rating is A- or better.

"Level II Status" exists at any date if, on such date, (i) the Borrower has not qualified for Level I Status and (ii) the Borrower's Moody's Rating is Baa1 or better *or* the Borrower's S&P Rating is BBB+ or better.

"Level III Status" exists at any date if, on such date, (i) the Borrower has not qualified for Level I Status or Level II Status and (ii) the Borrower's Moody's Rating is Baa2 or better *or* the Borrower's S&P Rating is BBB or better.

"Level IV Status" exists at any date if, on such date, (i) the Borrower has not qualified for Level I Status, Level II Status or Level III Status and (ii) the Borrower's Moody's Rating is Baa3 or better *or* the Borrower's S&P Rating is BBB- or better.

"Level V Status" exists at any date if, on such date, (i) the Borrower has not qualified for Level I Status, Level II Status, Level III Status or Level IV Status and (ii) the Borrower's Moody's Rating is BB+ or better *or* the Borrower's S&P Rating is Ba1 or better.

"Level VI Status" exists at any date if, on such date, the Borrower has not qualified for Level I Status, Level II Status, Level III Status, Level IV Status or Level V Status.

"Moody's Rating" means, at any time, the rating issued by Moody's and then in effect with respect to the Borrower's senior unsecured long-term debt securities without third-party credit enhancement (or, if there is no such debt outstanding, the Borrower's issuer rating issued by Moody's then in effect for the Borrower).

"S&P Rating" means, at any time, the rating issued by S&P and then in effect with respect to the Borrower's senior unsecured long-term debt securities without third-party credit enhancement (or, if there is no such debt outstanding, the Borrower's issuer rating issued by S&P then in effect for the Borrower).

"Status" means Level I Status, Level II Status, Level III Status, Level IV Status, Level V Status or Level VI Status.

The Applicable Margin for Eurodollar Advances, the Letter of Credit Fee Rate, the Applicable Margin for Floating Rate Advances and the Commitment Fee Rate shall be determined in accordance with the foregoing pricing grid based on the Borrower's Status as determined from its then-current Moody's and S&P Ratings. The credit rating in effect on any date for purposes of this Schedule is that in effect at the close of business on such date. If at any time the Borrower ceases to have a Moody's Rating or an S&P Rating, Level VI Status shall exist.

Notwithstanding the foregoing, (a) if the Borrower is split-rated and the ratings differential is one level, the higher rating will apply; and (b) if the Borrower is split-rated and the ratings differential is two levels or more, the intermediate rating at the midpoint will apply. If there is no midpoint, the higher of the two intermediate ratings will apply.

CERTAIN ADDRESSES FOR NOTICES

BORROWER:

1200 Main Street

Kansas City, MO 64105

Attention: Michael W. Cline, Vice-President – Investor Relations, Treasurer and Corporate Secretary

Telephone: (816) 556-2622 Facsimile: (816) 556-2992

Electronic Mail: michael.cline@kcpl.com

ADMINISTRATIVE AGENT:

Administrative Agent's Office

(for payments and Requests for Credit Extensions):

Bank of America, N.A. Credit Services Rep. II Mail Code: TX1-492-14-12 Bank of America Plaza 901 Main Street Dallas, TX 75202-3714

Attention: Karen Puente Telephone: 214-209-4108 Telecopier: 214-290-8378

Electronic Mail: Karen.r.puente@baml.com

Bank of America New York, NY

Account No.: 1292000883

Account Name: Credit Services

Ref: Great Plains ABA# 026009593

Other Notices as Administrative Agent:

Bank of America, N.A. Agency Management Bank of America Plaza 101 S. Tryon Street

Charlotte, NC 28255-0001 Mail Code: NC1-002-15-36

Attention: Kelly Weaver Telephone: (980) 387-5452 Facsimile: (704) 208-2871

Electronic Mail: kelly.weaver@baml.com

ISSUERS:

Bank of America, N.A. Trade Operations 1000 West Temple Street, 7th Floor Mail Code: CA9-705-07-05 Los Angeles, CA 90012 1514

Attention: Sandra Leon Telephone: (213) 580-8369 Telecopier: (213) 457-8841

Electronic Mail: sandra.leon@baml.com

The Bank of Tokyo-Mitsubishi UFJ, Ltd. 1251 Avenue of the Americas, 12th Floor

New York, NY 10020-1104

Attention:Jaya AngaraAttention:Antonina BondiTelephone:(201) 413-8843Telephone:(201) 413-8823

	Street, 15th Floor 8288 MAC: D1053-150
Attention: Telephone: Telecopier: Electronic Mail:	Elaine Shue (704) 715-3133 (704) 715-0358 Elaine.Shue@wachovia.com
	EXHIBIT A
	COMPLIANCE CERTIFICATE
	enders parties to the nt Described Below
from time to tim Union Bank, N	ompliance Certificate is furnished pursuant to the Credit Agreement dated as of August 9, 2010 (as amended, modified, renewed or extended ne, the "Agreement") among Great Plains Energy Incorporated (the "Borrower"), the lenders from time to time party thereto (the "Lenders"). A. and Wells Fargo Bank, National Association, as Syndication Agents, and Bank of America, N.A., as Administrative Agent for the sotherwise defined herein, capitalized terms used in this Compliance Certificate have the meanings ascribed thereto in the Agreement.
THE U	INDERSIGNED HEREBY CERTIFIES THAT:
1.	I am the duly electedof the Borrower;
	I have reviewed the terms of the Agreement and I have made, or have caused to be made under my supervision, an appropriate review of the conditions of the Borrower and, to the extent relevant to the Agreement, its Subsidiaries during the accounting period covered by the ents attached as Schedule I;
Certificate, exce	The examinations described in paragraph 2 did not disclose, and I have no knowledge of, the existence of any condition or event which fault or Unmatured Default during or at the end of the accounting period covered by the attached financial statements or as of the date of this ept as set forth on Schedule II which lists, in detail, the nature of the condition or event, the period during which it has existed and the action over has taken, is taking, or proposes to take with respect to each such condition or event; and
4.	Set forth on <u>Schedule III</u> are true and correct computations of the financial ratios and restrictions set forth in Section 6.15 of the Agreement.
The for	regoing certifications, together with the financial statements delivered with this Certificate in support hereof, are made and delivered this _
	GREAT PLAINS ENERGY INCORPORATED
Title:	By: Print Name:
	SCHEDIU E I

TO COMPLIANCE CERTIFICATE

Reports and Deliveries Currently Due

Telecopier:

(201) 413-8817

Electronic Mail: abondi@us.mufg.jp

Telecopier: (201) 413-8817 Electronic Mail: jangara@us.mufg.jp

Wells Fargo Bank, National Association

SCHEDULE II

TO COMPLIANCE CERTIFICATE

Conditions or Events which Constitute a Default or Unmatured Default

SCHEDULE III

TO COMPLIANCE CERTIFICATE

Computation of Financial Covenants

EXHIBIT B

ASSIGNMENT AGREEMENT

This Assignment and Assumption (the "Assignment and Assumption") is dated as of the Effective Date set forth below and is entered into by and between [Insert name of Assignor] (the "Assignor") and [Insert name of Assignee"). Capitalized terms used but not defined herein shall have the respective meanings given to them in the Credit Agreement identified below (as amended, the "Credit Agreement"), receipt of a copy of which is here by acknowledged by the Assignee. The Terms and Conditions set forth in Annex 1 attached hereto (the "Terms and Conditions") are hereby agreed to and incorporated herein by reference and made a part of this Assignment and Assumption as if set forth herein in full.

For an agreed consideration, the Assignor hereby irrevocably sells and assigns to the Assignee, and the Assignee hereby irrevocably purchases and assumes from the Assignor, subject to and in accordance with the Terms and Conditions and the Credit Agreement, as of the Effective Date inserted by the Administrative Agent as contemplated below, the interest in and to all of the Assignor's rights and obligations in its capacity as a Lender under the Credit Agreement and any other documents or instruments delivered pursuant thereto that represents the amount and percentage interest identified below of all of the Assignor's outstanding rights and obligations under the respective facilities identified below (including any letters of credit, swing line loans and guaranties included in such facilities and, to the extent permitted to be assigned under applicable law, all claims (including contract claims, tort claims, malpractice claims, statutory claims and all other claims at law or in equity), suits, causes of action and any other right of the Assignor against any Person whether known or unknown arising under or in connection with the Credit Agreement, any other documents or instruments delivered pursuant thereto or the loan transactions governed thereby) (the "Assigned Interest"). Such sale and assignment is without recourse to the Assignor and, except as expressly provided in this Assignment and Assumption, without representation or warranty by the Assignor.

1.	Assignor:			
2.	Assignee:			
3.	Borrower:	Great Plains Energy Incorporated		
4.	Administrative A	Agent: Bank of America, N.A., as the Administrative Agent under the Credit Agreement		
5.	Credit Agreement:	t Agreement: Credit Agreement dated as of August 9, 2010 among the Borrower, the Lenders from time to time party thereto and Bank of America, N.A., as Administrative Agent.		

6.	Assigned Interest:

Aggregate Amount of Commitment/Loans for Amount of Commitment/Loans Assigned*
all Lenders*

Percentage Assigned of Commitment/Loans¹

\$ _____%

\$ \$		\$ \$		% %		
7.	Trade Date:	U	<u>2</u>	^		
, ·	Hade Date.		_			
Effecti	Effective Date:, 20 [TO BE INSERTED BY AGENT AND WHICH SHALL BE THE EFFECTIVE DATE OF RECORDATION OF TRANSFER BY THE					
ADMI	NISTRATIVE AGENT.]	THE WHICH SHALL BE THE	LITECTIVE DITTE	Of RECORDATION OF TRANSFER BY THE		
	The terms set forth in this Assign	ment and Assumption are hereby a	agreed to:			
		ASSIGNOR [NAME OF ASSIGNOR]				
		By: Title:				
		Tiue.				
		ASSIGNEE [NAME OF ASSIGNEE]				
		By: Title:				
	ented to and] ³ Accepted: COF AMERICA, N.A., as Administ	votivo Agost				
By:	OF AMERICA, N.A., ds Administ	rative Agent				
Title:						
				nade between the Trade Date and the Effective Date.		
	orth, to at least 9 decimals, as a perc			nder.		
	t if satisfaction of minimum amount					
³ To be	e added only if the consent of the Ac	lministrative Agent is required by	the terms of the Credi	it Agreement.		
[Conse	ented to:] ⁴					
GREA	T PLAINS ENERGY INCORPORA	ATED				
By: Title:						
[Consented to:] $\frac{5}{2}$						
[ISSUERS]						
By: Title:						
1106.						
[Conse	ented to:] ⁶					
[SWING LINE LENDER]						
By: Title:						

- ⁴ To be added only if the consent of the Borrower is required by the terms of the Credit Agreement.
- ⁵ To be added only if the consent of the Issuers is required by the terms of the Credit Agreement.
- ⁶ To be added only if the consent of the Swing Line Lender is required by the terms of the Credit Agreement.

ANNEX 1

TERMS AND CONDITIONS FOR ASSIGNMENT AND ASSUMPTION

1. Representations and Warranties.

- Assignor. The Assignor (a) represents and warrants that (i) it is the legal and beneficial owner of the Assigned Interest, (ii) the Assigned Interest is free and clear of any lien, encumbrance or other adverse claim and (iii) it has full power and authority, and has taken all action necessary, to execute and deliver this Assignment and to consummate the transactions contemplated hereby; and (b) assumes no responsibility with respect to (i) any statements, warranties or representations made in or in connection with the Credit Agreement or any other Loan Document, (ii) the execution, legality, validity, enforceability, genuineness, sufficiency or value of the Loan Documents or any collateral th ereunder, (iii) the financial condition of the Borrower, any of its Subsidiaries or Affiliates or any other Person obligated in respect of any Loan Document or (iv) the performance or observance by the Borrower, any of its Subsidiaries or Affiliates or any other Person of any of their respective obligations under any Loan Document.
- Assignee. The Assignee (a) represents and warrants that (i) it has full power and authority, and has taken all action necessary, to execute and deliver this Assignment and to consummate the transactions contemplated hereby and to become a Lender under the Credit Agreement, (ii) it meets all requirements of an Eligible Assignee under the Credit Agreement (subject to receipt of such consents as may be required under the Credit Agreement), (iii) from and after the Effective Date, it shall be bound by the provisions of the Credit Agreement as a Lender thereunder and, to the extent of the Assigned Interest, shall have the obligations of a Lender thereunder, (iv) it has received a copy of the Credit Agreement, together with copies of the most recent financial statements delivered pursuant to Section 6.1(a) and (b) thereof, as applicable, and such other documents and information as it has deemed appropriate to make its own credit analysis and decision to enter into this Assignment and to purchase the Assigned Interest on the basis of which it has made such analysis and decision independently and without reliance on the Administrative Agent or any other Lender and (v) if it is a Foreign Lender, attached hereto is any documentation required to be delivered by it pursuant to the terms of the Credit Agreement, duly completed and executed by the Assignee; and (b) agrees that (i) it will, independently and without reliance on the Administrative Agent, the Assignor or any other Lender, and based on such documents and information as it shall deem appropriate at the time, continue to make its own credit decisions in taking or not taking action under the Loan Documents, and (ii) it will perform in accordance with their terms all of the obligations which by the terms of the Loan Documents are required to be performed by it as a Lender.
- 2. <u>Payments</u>. From and after the Effective Date, the Administrative Agent shall make all payments in respect of the Assigned Interest (including payments of principal, interest, fees and other amounts) to the Assignor for amounts which have accrued to but excluding the

Effective Date and to the assignee for amounts which have accrued from and after the Effective Date.

3. <u>General Provisions</u>. This Assignment shall be binding upon, and inure to the benefit of, the parties hereto and their respective successors and assigns. This Assignment may be executed in any number of counterparts, which together shall constitute one instrument. Delivery of an executed counterpart of a signature page of this Assignment by telecopy shall be effective as delivery of a manually executed counterpart of this Assignment. This Assignment shall be governed by, and construed in accordance with, the law of the State of New York.

EXHIBIT C

To: Bank of America, N.A., as Administrative Agent (the "Administrative Agent") under the Credit Agreement described below.

Re: Credit Agreement dated as of August 9, 2010 (as the same may be amended or modified, the "Credit Agreement"), among Great Plains Energy Incorporated (the "Borrower"), the lenders from time to time party thereto, Union Bank, N.A. and Wells Fargo Bank, National Association, as Syndication Agents, and the Administrative Agent. Capitalized terms used herein and not otherwise defined herein shall have the meanings assigned thereto in the Credit Agreement. The Administrative Agent is specifically authorized and directed to act upon the following standing money transfer instructions with respect to the proceeds of Credit Extensions or other extensions of credit from time to time until receipt by the Administrative Agent of a specific written revocation of such instructions by the Borrower; provided that the Administrative Agent may otherwise transfer funds as hereafter directed in writing by the Borrower in accordance with Section 13.1 of the Credit Agreement or based on any telephonic notice made in accordance with Section 2.14 of the Credit Agreement. Facility Identification Number(s) Customer/Account Name Transfer Funds To For Account No. Reference/Attention To Authorized Officer (Customer Representative) Date (Please Print) Signature Bank Officer Name Date (Please Print) Signature (Deliver Completed Form to Credit Support Staff For Immediate Processing) **EXHIBIT D** REVOLVING NOTE 20 GREAT PLAINS ENERGY INCORPORATED, a Missouri corporation (the "Borrower"), promises to pay to the order of (the "Lender") the aggregate unpaid principal amount of all Loans made by the Lender to the Borrower pursuant to Article II of the Agreement (as hereinafter defined), in immediately available funds at the Administrative Agent's Office, together with interest on the unpaid principal amount hereof at the rates and on the dates set forth in the Agreement. The Borrower shall pay the principal of and accrued and unpaid interest on the Loans in full on the Facility Termination Date. The Lender shall, and is hereby authorized to, record on a schedule attached hereto, or to otherwise record in accordance with its usual practice, the date and amount of each Loan and the date and amount of each principal payment hereunder. This Revolving Note is one of the Revolving Notes issued pursuant to, and is entitled to the benefits of, the Credit Agreement dated as of August 9, 2010 (which, as it may be amended or modified and in effect from time to time, is herein called the "Agreement"), among the Borrower, the lenders from time to time party thereto, including the Lender, Union Bank, N.A. and Wells Fargo Bank, National Association, as Syndication Agents, and Bank of America, N.A., as Administrative Agent, to which Agreement reference is hereby made for a statement of the terms and conditions governing this Revolving Note, including the terms and conditions under which this Revolving Note may be prepaid or its maturity date accelerated. Capitalized terms used herein and not otherwise defined herein are used with the meanings attributed t o them in the Agreement. THIS REVOLVING NOTE SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF NEW YORK.

[Remainder of page intentionally left blank]

GREAT PLAINS ENERGY INCORPORATED

By: Print Title:	t Name:	
	EXHIBIT E	
	SWING LINE NOTE	
	, 20	
accordance with the provisions of the Credit Agreement (the Swing Line Lender to the Borrower under that certain from time to time, the "Credit Agreement") among the E	"Borrower"), hereby promises to pay to BANK OF AMERICA, N.A. (the "Swing Line Lender"), in as hereinafter defined), the principal amount of each Swing Line Loan from time to time made by a Credit Agreement dated as of August 9, 2010 (as amended, modified, supplemented or extended Borrower, the Lenders from time to time party thereto, Union Bank, N.A. and Wells Fargo Bank, f America, N.A., as Administrative Agent and Swing Line Lender. Capitalized terms used but not be Credit Agr eement.	
The Borrower promises to pay interest on the unpaid principal amount of each Swing Line Loan from the date of such Swing Line Loan until such principal amount is paid in full, at such interest rates and at such times as provided in the Credit Agreement. All payments of principal and interest shall be made to the Administrative Agent for the account of the Swing Line Lender in dollars in immediately available funds at the Administrative Agent's Office. If any amount is not paid in full when due hereunder, such unpaid amount shall bear interest, to be paid upon demand, from the due date thereof until the date of actual payment (and before as well as after judgment) computed at the per annum rate set forth in the Credit Agreement.		
in part subject to the terms and conditions provided there Agreement, all amounts then remaining unpaid on this Sw in the Credit Agreement. Swing Line Loans made by the	ferred to in the Credit Agreement, is entitled to the benefits thereof and may be prepaid in whole or ein. Upon the occurrence and continuation of one or more of the Defaults specified in the Credit ing Line Note shall become, or may be declared to be, immediately due and payable all as provided Swing Line Lender shall be evidenced by one or more loan accounts or records maintained by the Lender may also attach schedules to this Swing Line Note and endorse thereon the date, amount espect there to.	
The Borrower, for itself, its successors and ass dishonor and nonpayment of this Swing Line Note.	igns, hereby waives diligence, presentment, protest and demand and notice of protest, demand,	
THIS SWING LINE NOTE SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF NEW YORK.		
[Remainder of page intentionally left blank]		
GRE	EAT PLAINS ENERGY INCORPORATED	
By: Print Title:	t Name:	

Published CUSIP Number: 48513MAC4

CREDIT AGREEMENT

Dated as of August 9, 2010

among

KANSAS CITY POWER & LIGHT COMPANY,

CERTAIN LENDERS,

BANK OF AMERICA, N.A., as Administrative Agent

and

UNION BANK, N.A. and WELLS FARGO BANK, NATIONAL ASSOCIATION, as Syndication Agents

> JPMORGAN CHASE BANK, N.A. and THE BANK OF NOVA SCOTIA, as Documentation Agents

BANC OF AMERICA SECURITIES LLC,
UNION BANK, N.A.
and
WELLS FARGO SECURITIES, LLC,
as Joint Lead Arrangers and Joint Book Managers

TABLE OF CONTENTS

	Page
ARTICLE I DEFINITIONS	1
1.1 Definitions.	1
1.2 Accounting Principles.	19
1.3 Letter of Credit Amounts.	19
1.4 Times of Day.	19
ARTICLE II THE CREDITS	19
2.1 Commitment.	19
2.2 Required Payments; Termination.	20
2.3 Ratable Loans.	20
2.4 Types of Advances; Minimum Amount.	20
2.5 Commitment Fee.	20
2.6 Changes in Aggregate Commitment.	21
2.7 Optional Prepayments.	24
2.8 Method of Selecting Types and Interest Periods for New Advances.	25
2.9 Conversion and Continuation of Outstanding Advances.	25
2.10 Interest Rates.	26
2.11 Rates Applicable After Default.	26
2.12 Method of Payment.	27
2.13 Noteless Agreement; Evidence of Indebtedness.	27

2.14 Telephonic Nouces.	40
2.15 Interest Payment Dates; Interest and Fee Basis.	28
2.16 Notification of Advances, Interest Rates, Prepayments and Commitment Reductions.	29
2.17 Lending Installations.	29
2.18 Non-Receipt of Funds by the Administrative Agent.	29
2.19 Letters of Credit.	30 37
2.20 Swing Line Loans. 2.21 Extension of Facility Termination Date.	40
2.22 Cash Collateral.	41
2.23 Defaulting Lenders.	42
ARTICLE III YIELD PROTECTION; TAXES	44
3.1 Yield Protection.	44
3.2 Changes in Capital Adequacy Regulations.	45
3.3 Availability of Types of Advances.	46
3.4 Funding Indemnification.	46
3.5 Taxes.	46
3.6 Lender Statements; Survival of Indemnity. ARTICLE IV CONDITIONS PRECEDENT	50 50
4.1 Initial Credit Extension.	50
4.2 Each Credit Extension.	52
ARTICLE V REPRESENTATIONS AND WARRANTIES	53
5.1 Existence and Standing.	53
5.2 Authorization and Validity.	53
5.3 No Conflict; Government Consent.	53
i	
5.4 Financial Statements.	54
5.5 Material Adverse Change.	54
5.6 Taxes.	54
5.7 Litigation; etc.	54
5.8 ERISA.	55 55
5.9 Accuracy of Information. 5.10 Regulation U.	55
5.11 Material Agreements.	55
5.12 Compliance With Laws.	56
5.13 Ownership of Properties.	56
5.14 Plan Assets; Prohibited Transactions.	56
5.15 Environmental Matters.	56
5.16 Investment Company Act.	56
5.17 Pari Passu Indebtedness.	56
5.18 Solvency.	57
5.19 No Defaults. ARTICLE VI COVENANTS	57 57
6.1 Financial Reporting and Notices.	57
6.2 Permits, Etc.	59
6.3 Use of Proceeds.	59
6.4 Notice of Default.	59
6.5 Conduct of Business.	60
6.6 Taxes.	60
6.7 Insurance.	60
6.8 Compliance with Laws.	60
6.9 Maintenance of Properties; Books of Record.	61
6.10 Inspection.	61
6.11 Consolidations, Mergers and Sale of Assets. 6.12 Liens.	61 63
6.13 Affiliates.	66
6.14 ERISA.	66
6.15 Total Indebtedness to Total Capitalization.	66
6.16 Restrictions on Subsidiary Dividends.	66
6.17 Organization Documents.	67
ARTICLE VII DEFAULTS	67
ARTICLE VIII ACCELERATION, WAIVERS, AMENDMENTS AND REMEDIES	69
8.1 Acceleration; Letter of Credit Account.	69
8.2 Amendments.	70
8.3 Preservation of Rights.	71
ARTICLE IX GENERAL PROVISIONS	71
9.1 Survival of Representations.	71
9.2 Governmental Regulation.9.3 Headings.	71 72
9.4 Entire Agreement.	72
9.5 Several Obligations; Benefits of this Agreement.	72
	72

0.7 N. okaza (Dagara)	70
9.7 Numbers of Documents.	73
9.8 Accounting.	73
9.9 Severability of Provisions.	74
9.10 Nonliability of Lenders.	74
9.11 Limited Disclosure.	74
9.12 USA PATRIOT ACT NOTIFICATION.	75
9.13 Nonreliance.	75
9.14 No Advisory or Fiduciary Responsibility.	75
ARTICLE X THE ADMINISTRATIVE AGENT	76
10.1 Appointment and Authority.	76
10.2 Rights as a Lender.	77
10.3 Exculpatory Provisions.	77
10.4 Reliance by Administrative Agent.	78
10.5 Delegation of Duties.	78
10.6 Resignation of Administrative Agent.	78
10.7 Non-Reliance on Administrative Agent and Other Lenders.	79
10.8 No Other Duties, Etc.	80
10.9 Administrative Agent May File Proofs of Claim.	80
ARTICLE XI SETOFF; RATABLE PAYMENTS; PAYMENTS SET ASIDE	81
11.1 Setoff.	81
11.2 Ratable Payments.	81
11.3 Payments Set Aside.	82
ARTICLE XII BENEFIT OF AGREEMENT; ASSIGNMENTS; PARTICIPATIONS	82
12.1 Successors and Assigns.	82
12.2 Replacement of Lenders.	86
ARTICLE XIII NOTICES	87
13.1 Notices.	87
13.2 Change of Address.	88
13.3 The Platform.	88
ARTICLE XIV COUNTERPARTS	88
ARTICLE XV OTHER AGENTS	88
ARTICLE XVI CHOICE OF LAW; CONSENT TO JURISDICTION; WAIVER OF JURY TRIAL.	89
16.1 CHOICE OF LAW.	89
16.2 CONSENT TO JURISDICTION.	89
16.3 WAIVER OF JURY TRIAL.	89
ARTICLE XVII TERMINATION OF EXISTING CREDIT FACILITY	90

iii

SCHEDULES

- Commitments and Letter of Credit Commitments
- II **Existing Letters of Credit**
- III Pricing Schedule
- IV Certain Addresses for Notices

EXHIBITS

- Form of Compliance Certificate
- Form of Assignment and Assumption В C
- Form of Wire Transfer Instructions
 - D Form of Revolving Note
 - E Form of Swing Line Note

iv

CREDIT AGREEMENT

This Credit Agreement dated as of August 9, 2010 is among Kansas City Power & Light Company, a Missouri corporation, the Lenders, Union Bank, N.A. and Wells Fargo Bank, National Association, as Syndication Agents, and Bank of America, N.A., as Administrative Agent. The parties hereto agree as follows:

ARTICLE I

DEFINITIONS

1.1 Definitions.

As used in this Agreement, the following terms have the following meanings (such meanings to be equally applicable to both the singular and plural forms of such terms):

- "Additional Commitment Lender" is defined in Section 2.21(d).
- "Administrative Agent" means Bank of America in its capacity as administrative agent under any of the Loan Documents, or any successor administrative agent.
- "Administrative Agent's Fee Letter" means certain fee letter dated June 2, 2010 among Bank of America, the Borrower, Great Plains and KCPL GMO.
- "<u>Administrative Agent's Office</u>" means the Administrative Agent's address and, as appropriate, account as set forth on <u>Schedule IV</u> or such other address or account as the Administrative Agent may from time to time notify to the Borrower and the Lenders.
 - "Administrative Questionnaire" means an Administrative Questionnaire in a form supplied by the Administrative Agent.
- "Advance" means a borrowing hereunder (or conversion or continuation thereof) consisting of the aggregate amount of the several Loans, including the Swing Line Loans, made on the same Borrowing Date (or date of conversion or continuation) by the Lenders to the Borrower of the same Type and, in the case of Eurodollar Advances, for the same Interest Period.
- "Affiliate" of any Person means any other Person directly or indirectly controlling, controlled by or under common control with such Person. For purposes of this definition, "control" (including, with correlative meanings, the terms "controlled by" and "under common control with") shall mean the possession, directly or indirectly, of the power to direct or cause the direction of the management or policies of a Person, whether through the ownership of voting securities or by contract or otherwise.
 - "Agent Parties" is defined in Section 13.3.
 - "Agents" means, collectively, the Administrative Agent and the Syndication Agents, and "Agent" means any one of them.
- "Aggregate Commitment" means the aggregate of the Commitments of all Lenders, as changed from time to time pursuant to the terms hereof. The amount of the Aggregate Commitment in effect as of the Closing Date is SIX HUNDRED MILLION DOLLARS (\$600,000,000).
 - "Aggregate Outstanding Credit Exposure" means, at any time, the aggregate of the Outstanding Credit Exposure of all Lenders.
 - "Agreement" means this credit agreement, as it may be amended or modified and in effect from time to time.
- "Alternate Base Rate" means for any day a fluctuating rate per annum equal to the highest of (a) the Federal Funds Rate plus 1/2 of 1%, (b) the rate of interest in effect for such day as publicly announced from time to time by Bank of America as its "prime rate" and (c) Eurodollar Base Rate plus 1.0%. The "prime rate" is a rate set by Bank of America based upon various factors including Bank of America's costs and desired return, general economic conditions and other factors, and is used as a reference point for pricing some loans, which may be priced at, above, or below such announced rate. Any change in such prime rate announced by Bank of America shall take effect at the opening of business on the& #160;day specified in the public announcement of such change.
- "Applicable Margin" means, with respect to Advances of any Type at any time, the percentage rate per annum which is applicable at such time with respect to Advances of such Type as set forth in the Pricing Schedule.
- "Approved Fund" means any Fund that is administered or managed by (a) a Lender, (b) an Affiliate of a Lender or (c) an entity or an Affiliate of an entity that administers or manages a Lender.
 - "Approving Lenders" is defined in Section 2.21(e).
 - "Arrangers" means BAS, Union Bank and Wells Fargo Securities, LLC and "Arranger" means any one of them.
 - "Article" means an article of this Agreement unless another document is specifically referenced.
- "Assignment Agreement" means an assignment and assumption entered into by a Lender and an Eligible Assignee (with the consent of any party whose consent is required by Section 12.1(b)), and accepted by the Administrative Agent, in substantially the form of Exhibit B or any other form approved by the Administrative Agent.

"Attributable Indebtedness" means, on any date, (a) in respect of any Capitalized Lease Obligation of any Person, the capitalized amount thereof that would appear on a balance sheet of such Person prepared as of such date in accordance with GAAP, and (b) in respect of any Synthetic Lease Obligation, the capitalized amount of the remaining lease payments under the relevant lease that would appear on a balance sheet of such Person prepared as of such date in accordance with GAAP if such lease were accounted for as a Capitalized Lease.

"Authorized Officer" means any of the Chief Executive Officer, the President, any Vice President, the Chief Financial Officer or the Treasurer of the Borrower, in each case acting singly.

"Bank of America" means Bank of America, N.A. in its individual capacity and its successors.

"BAS" means Banc of America Securities LLC.

"Borrower" means Kansas City Power & Light Company, a Missouri corporation, and its permitted successors and assigns.

"Borrower Materials" is defined in Section 6.1.

"Borrowing Date" means a date on which an Advance is made hereunder.

"Borrowing Notice" is defined in Section 2.8.

"BTMU" means The Bank of Tokyo-Mitsubishi UFJ, Ltd., in its individual capacity and its successors.

"Business Day" means any day other than a Saturday, Sunday or other day on which commercial banks are authorized to close under the Laws of, or are in fact closed in, New York City, New York and, if such day relates to any Eurodollar Loan or any Floating Rate Loan bearing interest at a rate based on the Eurodollar Rate, means any such day that is also a London Banking Day.

"Capitalized Lease" of a Person means any lease of Property by such Person as lessee which would be capitalized on a balance sheet of such Person prepared in accordance with GAAP.

"Capitalized Lease Obligations" of a Person means the amount of the obligations of such Person under Capitalized Leases which would be shown as a liability on a balance sheet of such Person prepared in accordance with GAAP.

"Cash Collateralize" means to pledge and deposit with or deliver to the Administrative Agent, for the benefit of the Administrative Agent, the Issuers or the Swing Line Lender (as applicable) and the Lenders, as collateral for Letter of Credit Obligations, Obligations in respect of Swing Line Loans, or obligations of Lenders to fund participations in respect of either thereof

3

(as the context may require), cash or deposit account balances or, if the applicable Issuer or the Swing Line Lender benefitting from such collateral shall agree in its sole discretion, other credit support, in each case pursuant to documentation in form and substance reasonably satisfactory to (a) the Administrative Agent and (b) the applicable Issuer or the Swing Line Lender (as applicable). "Cash Collateral" shall have a meaning correlative to the foregoing and shall include the proceeds of such cash collateral and other credit support.

"Change" is defined in Section 3.2.

"Change in Law" means the occurrence, after the date of this Agreement, of any of the following: (a) the adoption or taking effect of any law, rule, regulation or treaty, (b) any change in any law, rule, regulation or treaty or in the administration, interpretation or application thereof by any Governmental Authority or (c) the making or issuance of any request, guideline or directive (whether or not having the force of law) by any Governmental Authority.

"Change of Control" means an event or series of events by which:

- (i) any "person" or "group" (as such terms are used in Sections 13(d) and 14(d) of the Securities Exchange Act of 1934, but excluding any employee benefit plan of Great Plains or its Subsidiaries, or any Person acting in its capacity as trustee, agent or other fiduciary or administrator of any such plan) becomes the "beneficial owner" (as defined in Rules 13d-3 and 13d-5 under the Securities Exchange Act of 1934), directly or indirectly, of thirty-three and one-third percent (33 1/3%) or more of the "voting equity interests" (meaning for this purpose the power under ordinary circumstances to vote for the election of members of the board of directors) of Great Plains; or
- (ii) during any period of twelve (12) consecutive months (or such lesser period of time as shall have elapsed since the formation of Great Plains), a majority of the members of the board of directors or other equivalent governing body of Great Plains ceases to be composed of individuals (x) who were members of that board or equivalent governing body on the first day of such period, (y) whose election or nomination to that board or equivalent governing body was approved by individuals referred to in clause (x) above constituting at the time of such election or nomination at least a majority of that board or equivalent governing body or (z) whose election or nomination to that board or other equivalent governing body was approved by individuals referred to in clauses (x) and (y) above constituting at the time of such election or nomination at least a majority of that board or equivalent governing body.

"Closing Date" means August 9, 2010.

"Code" means the Internal Revenue Code of 1986, as amended, reformed or otherwise modified from time to time.

"Collateral Shortfall Amount" is defined in Section 8.1.

"Commitment" means, for each Lender, the obligation of such Lender to make Loans and to participate in Letters of Credit and Swing Line Loans in an aggregate amount not exceeding the amount set forth on Schedule I hereto or as set forth in any Assignment Agreement relating to any assignment that has become effective pursuant to Section 12.1(b), as such amount may be modified from time to time pursuant to the terms hereof.

"Commitment Fee Rate" means, at any time, the percentage rate per annum at which commitment fees are accruing at such time as set forth in the Pricing Schedule.

"Consolidated Net Income" means, for any period, for the Borrower and its Consolidated Subsidiaries, the net income of the Borrower and its Consolidated Subsidiaries from continuing operations, excluding extraordinary items for that period.

"Consolidated Subsidiaries" means all Subsidiaries of the Borrower that are (or should be) included when preparing the consolidated financial statements of the Borrower.

"Consolidated Tangible Net Worth" means, as of any date of determination, for the Borrower and its Consolidated Subsidiaries, Shareholders' Equity of the Borrower and its Consolidated Subsidiaries on that date minus the Intangible Assets of the Borrower and its Consolidated Subsidiaries on that date.

"Contingent Obligation" of a Person means any agreement, undertaking or arrangement by which such Person assumes, guarantees, endorses, contingently agrees to purchase or provide funds for the payment of, or otherwise becomes or is contingently liable upon, the obligation or liability of any other Person, or agrees to maintain the net worth or working capital or other financial condition of any other Person, or otherwise assures any creditor of such other Person against loss.

"Controlled Group" means all members of a controlled group of corporations or other business entities and all trades or businesses (whether or not incorporated) under common control which, together with the Borrower, are treated as a single employer under Section 414 of the Code.

"Conversion/Continuation Notice" is defined in Section 2.9.

"Credit Extension" means the making of an Advance or the issuance of a Letter of Credit.

"<u>Debtor Relief Law</u>" means the Bankruptcy Code of the United States, and all other liquidation, conservatorship, bankruptcy, assignment for the benefit of creditors, moratorium, rearrangement, receivership, insolvency, reorganization, or similar debtor relief Laws of the United States or other applicable jurisdictions from time to time in effect and affecting the rights of creditors generally.

"Default" means an event described in Article VII.

5

"Defaulting Lender" means, subject to Section 2.23(b), any Lender, as reasonably determined by the Administrative Agent, that, at any time during the term of this Agreement, (a) has failed to perform any of its funding obligations hereunder, including in respect of its Loans or participations in respect of Letters of Credit or Swing Line Loans, within three (3) Business Days of the date required to be funded by it hereunder, unless such obligation is the subject of a good faith dispute, (b) has notified the Borrower, the Administrative Agent or any Lender in writing, or has otherwise indicated through a public statement, that it does not intend to comply with any of its funding obligations under this Agreement or generally under other agreements in which it commits to extend credit, (c) has failed, within five (5) Business Days after written request by the Administrative Agent (based on the reasonable belief that such Lender may not fulfill any of its funding obligations hereunder or after any reasonable request by the Borrower to do so), to confirm in a manner reasonably satisfactory to the Administrative Agent and the Borrower that it will comply with its funding obligations, unless such obligation is the subject of a good faith dispute, provided that any such Lender shall cease to be a Defaulting Lender under this clause (c) upon receipt of such confirmation by the Administrative Agent in a manner reasonably satisfactory to the Administrative Agent, (d) otherwise failed to pay over to the Administrative Agent or any other Lender any other amount (other than a de minimis amount) required to be paid by it hereunder within three (3) Business Days of the date when due, unless the subject of a good-faith dispute, or (e) has, or has a direct or indirect parent company that has, (i) become the subject of a proceeding under any Debtor Relief Law, (ii) had a receiver, conservator, trustee, administrator, assignee for the benefit of creditors or similar Person charged with reorganization or liquidation of its business or a custodian appointed for it or for any substantial part of its assets, or (iii) taken any action in furtherance of, or indicated its consent to, approval of or acquiescence in any such proceeding or appointment; provided that a Lender shall not be a Defaulting Lender solely by virtue of the ownership or acquisition of any equity interest in that Lender or any direct or indirect parent company thereof, or the exercise of control over such Lender or any direct or indirect parent company thereof controlling such Lender, by a Governmental Authority.

"Dollar" and "\$" mean lawful money of the United States.

"Eligible Assignee" means (a) a Lender; (b) an Affiliate of a Lender; (c) an Approved Fund; and (d) any other Person (other than a natural person) approved by (i) the Administrative Agent, the Swing Line Lender and the Issuers, and (ii) unless a Default has occurred and is continuing, the Borrower (each such approval not to be unreasonably withheld or delayed); provided that notwithstanding the foregoing, "Eligible Assignee" shall not include the Borrower or any of the Borrower's Affiliates or Subsidiaries.

"Environmental Laws" means any and all federal, state, local and foreign statutes, laws, judicial decisions, regulations, ordinances, rules, judgments, orders, decrees, plans, injunctions, permits, concessions, grants, franchises, licenses, agreements and other governmental restrictions relating to (a) the protection of the environment, (b) the effect of the environment on human health, (c) emissions, discharges or releases of pollutants, contaminants, hazardous

6

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended from time to time, and any rule or regulation issued thereunder.

"Eurodollar Advance" means an Advance which bears interest at the applicable Eurodollar Rate.

"Eurodollar Rate" means:

(a) for any Interest Period with respect to a Eurodollar Loan, a rate per annum (rounded to the nearest multiple of 1/16 of 1%) determined by the Administrative Agent pursuant to the following formula:

Eurodollar Rate = $\frac{\text{Eurodollar Base Rate}}{1.00 - \text{Eurodollar Reserve Percentage}}$

and (b) for any day with respect to a Floating Rate Loan bearing interest at a rate based on the Eurodollar Base Rate, a rate per annum (rounded to the nearest multiple of 1/16 of 1%) determined by the Administrative Agent pursuant to the following formula:

 $Eurodollar\ Rate = \underbrace{\frac{Eurodollar\ Base\ Rate}{1.00-Eurodollar\ Reserve\ Percentage}}.$

"Eurodollar Base Rate" means:

(a) for such Interest Period with respect to any Eurodollar Loan, the rate per annum equal to (i) the British Bankers Association LIBOR Rate ("BBA LIBOR"), as published by Reuters (or such other commercially available source providing quotations of BBA LIBOR as may be designated by the Administrative Agent from time to time) at approximately 11:00 a.m., London time, two (2) London Banking Days prior to the commencement of such Interest Period, for Dollar deposits (for delivery on the first day of such Interest Period) with a term equivalent to such Interest Period or (ii) if such published rate is not available at such time for any reason, the rate determined by the Administrative Agent to be the rate at which deposits in Dollars for delivery on the first day of such Interest Period in same day funds in the approximate amount of the Eurodollar Loan being made, continued or converted and with a term equivalent to such Interest Period would be offered by Bank of America's London Branch to major banks in the London interbank eurodollar market at their request at approximately 11:00 a.m. (London time) two (2) London Banking Days prior to the commencement of such Interest Period and

(b) for any interest rate calculation with respect to any Floating Rate Loan, the rate per annum equal to (i) BBA LIBOR, at approximately 11:00 a.m., London time determined two (2) London Banking Days prior to such date for Dollar deposits being delivered in the London interbank market for a term of one (1) month commencing that day or (ii) if such published rate is not available at such time for any reason, the rate per annum determined by the Administrative Agent to be the rate at which deposits in Dollars for delivery on the date of determination in same day funds in the approximate amount of the Floating Rate Loan being made or maintained

7

and with a term equal to one (1) month would be offered by Bank of America's London Branch to major banks in the London interbank Eurodollar market at their request at the date and time of determination.

"Eurodollar Reserve Percentage" means the reserve percentage (expressed as a decimal, carried out to five decimal places) in effect on such day, whether or not applicable to any Lender, under regulations issued from time to time by the FRB for determining the maximum reserve requirement (including any emergency, supplemental or other marginal reserve requirement) with respect to Eurocurrency funding (currently referred to as "Eurocurrency liabilities"). The Eurodollar Rate for each outstanding Eurodollar Loan or Floating Rate Loan bearing interest at a rate based on the Eurodollar Rate shall be adjusted automatically as of the effective date of any change in the Eurodollar Reserve Percentage.

"Eurodollar Loan" means a Loan which bears interest at the applicable Eurodollar Rate.

"Excluded Taxes" means, with respect to the Administrative Agent, any Lender, or any other recipient of any payment to be made by or on account of any Obligation of the Borrower hereunder or under any other Loan Document, (a) taxes imposed on or measured by its overall net income (however denominated), and franchise taxes imposed on it (in lieu of net income taxes), by the jurisdiction (or any political subdivision thereof) under the laws of which such recipient is organized or in which its principal office is located or, in the case of any Lender, in which its applicable Lending Installation is located, (b) any branch profits taxes imposed by the United States or any similar tax imposed by any other jurisdiction, (c) any backup withholding tax that is required by the Code to be withheld from amounts payable to a Lender, (d) in the case of a Foreign Lender (other than an assignee pursuant to a request by the Borrower under Section 12.2), any United States withholding tax that (i) is imposed on amounts payable to such Foreign Lender pursuant to the laws in force at the time such Foreign Lender becomes a party hereto (or designates a new Lending Installation) or (ii) is attributable to such Foreign Lender's failure or inability (other than solely as a result of a Change in Law occurring after such Foreign Lender becomes a party to this Agreement or designates a new Lending Installation) to comply with Section 3.5(e), except to the extent that such Foreign Lender (or its assignor, if any) was entitled, immediately prior to the time of designation of a new Lending Installation (or assignment), to receive additional amounts from the Bor rower with respect to such withholding tax pursuant to Section 3.5(a)(ii), and (e) any taxes imposed with respect to the requirements of FATCA.

"Exhibit" refers to an exhibit to this Agreement, unless another document is specifically referenced.

"Existing Credit Facility" means the credit agreement among the Borrower, Bank of America, N.A., as administrative agent and the other lenders party thereto, dated as of May 11, 2006, as amended or modified from time to time.

"Existing Letters of Credit" means those letters of credit identified on Schedule II.

"Facility Termination Date" means (a) the later of (i) August 9, 2013 and (ii) with respect to some or all of the Lenders if the facility termination date is extended pursuant to Section 2.21,

8

such extended facility termination date or (b) any earlier date on which the Aggregate Commitment is reduced to zero or otherwise terminated pursuant to the terms hereof.

"FATCA" means sections 1471 through 1474 of the Code and any regulations that are issued thereunder or official governmental interpretations thereof.

"Federal Funds Rate" means, for any day, the rate per annum equal to the weighted average of the rates on overnight Federal funds transactions with members of the Federal Reserve System arranged by Federal funds brokers on such day, as published by the Federal Reserve Bank of New York on the Business Day next succeeding such day; provided that (a) if such day is not a Business Day, the Federal Funds Rate for such day shall be such rate on such transactions on the next preceding Business Day as so published on the next succeeding Business Day, and (b) if no such rate is so published on such next succeeding Business Day, the Federal Funds Rate for such day shall be the average rate (rounded upward, i f necessary, to a whole multiple of 1/100 of 1%) charged to Bank of America on such day on such transactions as determined by the Administrative Agent.

"Fee Letter" means that certain fee letter dated June 2, 2010 among the Agents, the Arrangers, the Borrower, Great Plains and KCPL GMO.

"Floating Rate Advance" means an Advance which bears interest at the Alternate Base Rate.

"Floating Rate Loan" means a Loan which bears interest at the Alternate Base Rate.

"Foreign Lender" means any Lender that is not a "United States person" within the meaning of Section 7701(a)(30) of the Code.

"FRB" means the Board of Governors of the Federal Reserve System.

"Fronting Exposure" means, at any time there is a Defaulting Lender, (a) with respect to an Issuer, such Defaulting Lender's Pro Rata Share of the outstanding Letter of Credit Obligations other than Letter of Credit Obligations as to which such Defaulting Lender's participation obligation has been reallocated to other Lenders or Cash Collateralized in accordance with the terms hereof, and (b) with respect to the Swing Line Lender, such Defaulting Lender's Pro Rata Share of Swing Line Loans other than Swing Line Loans as to which such Defaulting Lender's participation obligation has been reallocated to other Lenders or Cash Collateralized in accordance with the terms hereof.

"Fund" means any Person (other than a natural person) that is (or will be) engaged in making, purchasing, holding or otherwise investing in commercial loans and similar extensions of credit in the ordinary course of its activities.

"GAAP" means generally accepted accounting principles set forth from time to time in the opinions and pronouncements of the Accounting Principles Board and the American Institute of Certified Public Accountants and statements of the Financial Accounting Standards Board.

9

"Governmental Authority" means the government of the United States or any other nation, or of any political subdivision thereof, whether state or local, and any agency, authority, instrumentality, regulatory body, court, central bank or other entity exercising executive, legislative, judicial, taxing, regulatory or administrative powers or functions of or pertaining to government.

"GPE-KCPL Re-Transfer" is defined in Section 2.6(d)(ii).

"GPE-KCPL Transfer" is defined in Section 2.6(d)(i).

"Great Plains" means Great Plains Energy Incorporated, a Missouri corporation.

"Great Plains Credit Agreement" means that certain Credit Agreement dated as of the Closing Date among Great Plains, the financial institutions party thereto, Union Bank and Wells Fargo, as syndication agents and Bank of America, N.A., as administrative agent, as amended or modified from time to time.

"Honor Date" is defined in Section 2.19(f).

"including" means "including without limiting the generality of the following".

"Indebtedness" means, as to any Person at a particular time, all of the following, without duplication, to the extent recourse may be had to the assets or properties of such Person in respect thereof: (a) all obligations of such Person for borrowed money and all obligations of such Person evidenced by bonds,

debentures, notes, loan agreements or other similar instruments; (b) any direct or contingent obligations of such Person in the aggregate in excess of \$2,000,000 arising under letters of credit (including standby and commercial), banker's acceptances, bank guaranties, surety bonds and similar instruments; (c) net obligations of such Person under Swap Contracts; (d) all obligations of such Person to pay the deferred purchase price of property or service s (except trade accounts payable arising, and accrued expenses incurred, in the ordinary course of business), and indebtedness (excluding prepaid interest thereon) secured by a Lien on property owned or being purchased by such Person (including indebtedness arising under conditional sales or other title retention agreements), whether or not such indebtedness shall have been assumed by such Person or is limited in recourse; (e) Capitalized Lease Obligations and Synthetic Lease Obligations of such Person; and (f) all Contingent Obligations of such Person in respect of any of the foregoing.

For all purposes hereof, the Indebtedness of any Person shall include the Indebtedness of any partnership or joint venture in which such Person is a general partner or a joint venturer, unless such Indebtedness is non-recourse to such Person. It is understood and agreed that Indebtedness (including Contingent Obligations) shall not include any obligations of any Person with respect to (i) subordinated, deferrable interest debt securities, and any related securities issued by a trust or other special purpose entity in connection therewith, or any similar securities that are classified at the time of issuance, as possessing a minimum of "intermediate equity content" by S&P and "Basket C equity credit" by Moody's (or the equivalent classification then

10

in effect by such agencies), as long as the maturity date of such debt is subsequent to the Facility Termination Date; <u>provided</u> that the amount of mandatory principal amortization or defeasance of such debt prior to the Facility Termination Date shall be included in this definition of Indebtedness; or (ii) utility "rate reduction" bonds, for the payment of which legislatively authorized charges are imposed on customers. The amount of any Capitalized Lease Obligation or Synthetic Lease Obligation as of any date shall be deemed to be the amount of Attributable Indebtedness in respect thereof as of such date.

"Indemnified Taxes" means Taxes other than Excluded Taxes.

"<u>Intangible Assets</u>" means, assets that are considered to be intangible assets under GAAP, including, but not limited to, customer lists, goodwill, computer software, copyrights, trade names, trademarks, patents, franchises and licenses.

"Interest Period" means, with respect to a Eurodollar Advance, a period of one (1), two (2), three (3) or six (6) months commencing on a Business Day selected by the Borrower pursuant to this Agreement. Such Interest Period shall end on the day which corresponds numerically to such date one (1), two (2), three (3) or six (6) months thereafter; provided that if there is no such numerically corresponding day in such next, second, third or sixth succeeding month, such Interest Period shall end on the last Business Day of such next, second, third or sixth succeeding month. If an Interest Period would otherwise end on a day which is not a Business Day, such Interest Period shall end on the next succeeding Business Day; provided that if said next succeeding Business Day falls in a new calendar month, such Interest Period shall end on the immediately preceding Business Day.

"ISP" means, with respect to any Letter of Credit, the "International Standby Practices 1998" published by the Institute of International Banking Law & Practice, Inc. (or such later version thereof as may be in effect at the time of issuance of such Letter of Credit).

"Issuer" means each of Bank of America, Wells Fargo, BTMU and any other Lender selected by the Borrower with the approval of the Administrative Agent (except with respect to any selection for which such approval is not required pursuant to Section 2.19(n)), in each case, in its capacity as an issuer of Letters of Credit hereunder and includes, without limitation, any Lender appointed by the Borrower (with the consent of the Administrative Agent, except with respect to any selection for which such approval is not required pursuant to Section 2.19(n)) as such by notice to the Lenders as a replacement for any Issuer who at the time of su ch appointment is a Defaulting Lender. For the avoidance of doubt, any Lender (other than Bank of America, Wells Fargo and BTMU) approached to become an issuer of Letters of Credit may elect or decline, in its sole discretion, to become an issuer. Notwithstanding the foregoing, JPMorgan Chase Bank, N.A. shall be the Issuer with respect to those Existing Letters of Credit identified on Part A of Schedule II.

"Issuer Documents" means with respect to any Letter of Credit, any Letter of Credit Application and any other document, agreement and instrument entered into by the applicable Issuer and the Borrower or in favor of the applicable Issuer and relating to such Letter of Credit.

11

"KCPL GMO" means KCP&L Greater Missouri Operations Company, a Delaware corporation.

"KCPL GMO Credit Agreement" means that certain Credit Agreement dated as of the Closing Date among KCPL GMO, the financial institutions party thereto, Union Bank and Wells Fargo, as syndication agents and Bank of America, N.A., as administrative agent, as amended or modified from time to time.

"KCPL-GPE Re-Transfer" is defined in Section 2.6(c)(ii).

"KCPL-GPE Transfer" is defined in Section 2.6(c)(i).

"<u>Laws</u>" means, collectively, all international, foreign, federal, state and local statutes, treaties, rules, guidelines, regulations, ordinances, codes and administrative or judicial precedents or authorities, including the interpretation or administration thereof by any Governmental Authority charged with the enforcement, interpretation or administration thereof, and all applicable administrative orders, directed duties, requests, licenses, authorizations and permits of, and agreements with, any Governmental Authority, in each case as in effect from time to time.

"LC Collateral Accounts" is defined in Section 2.22(b).

"<u>Lenders</u>" means the lending institutions listed on the signature pages of this Agreement and their respective successors and assigns, and any other Person that shall become a party hereto pursuant to an Assignment Agreement or a joinder agreement provided in <u>Section 2.6(b)(iv)</u>, and as the context requires, includes the Swing Line Lender.

"<u>Lending Installation</u>" means, with respect to a Lender or the Administrative Agent, the office, branch, subsidiary or affiliate of such Lender or the Administrative Agent listed on the signature pages hereof or on a Schedule or otherwise selected by such Lender or the Administrative Agent pursuant to Section 2.17.

"Letter of Credit" means any standby letter of credit issued pursuant to Section 2.19 and any Existing Letter of Credit.

"Letter of Credit Application" is defined in Section 2.19(c).

"<u>Letter of Credit Commitment</u>" means, for each Issuer, the obligation of such Issuer to issue Letters of Credit in an aggregate amount not exceeding the amount set forth on <u>Schedule I</u> hereto or as set forth in any commitment agreement entered into by any Lender appointed as an Issuer by the Borrower in accordance with the terms hereof.

"Letter of Credit Fee" is defined in Section 2.19(d).

"<u>Letter of Credit Fee Rate</u>" means, at any time, the percentage rate per annum applicable to Letter of Credit Fees at such time as set forth in the Pricing Schedule.

12

"<u>Letter of Credit Obligations</u>" means, at any time, the sum, without duplication, of (i) the aggregate undrawn stated amount of all outstanding Letters of Credit at such time plus (ii) the aggregate unpaid amount of all Reimbursement Obligations at such time. For purposes of computing the amount available to be drawn under any Letter of Credit, the amount of such Letter of Credit shall be determined in accordance with <u>Section 1.3</u>. For all purposes of this Agreement, if on any date of determination a Letter of Credit has expired by its terms but any amount may still be drawn thereunder by reason of the operation of Rule 3.14 of the ISP, such Letter of Credit shall be deemed to be "outstanding" in the amount so remaining available to be drawn.

"Letter of Credit Payment Date" is defined in Section 2.19(e).

"<u>Letter of Credit Sublimit</u>" means an amount equal to the lesser of (a) \$75,000,000 and (b) the Aggregate Commitments. The Letter of Credit Sublimit is part of, and not in addition to, the Aggregate Commitments.

"<u>Lien</u>" means any lien (statutory or other), mortgage, pledge, hypothecation, assignment, deposit arrangement, encumbrance or preference, priority or other security agreement or preferential arrangement of any kind or nature whatsoever (including the interest of a vendor or lessor under any conditional sale, Capitalized Lease or other title retention agreement).

"Loan" means, with respect to a Lender, such Lender's loans made pursuant to <u>Article II</u> (or any conversion or continuation thereof), including the Swing Line Loans.

"Loan Documents" means this Agreement, each Note, any agreement creating or perfecting rights in Cash Collateral pursuant to the provisions of Section 2.22, each Letter of Credit, each Issuer Document, the Administrative Agent's Fee Letter and the Fee Letter.

"London Banking Day." means any day on which dealings in Dollar deposits are conducted by and between banks in the London interbank eurodollar market.

"<u>Material Adverse Effect</u>" means a material adverse effect on (i) the business, Property, financial condition, or results of operations of the Borrower and its Subsidiaries taken as a whole, (ii) the ability of the Borrower to perform its obligations under the Loan Documents or (iii) the validity or enforceability of any of the Loan Documents or the rights or remedies of the Agents, the Lenders or the Issuers thereunder.

"Material Indebtedness" is defined in Section 7.6.

"Modification" and "Modify" are defined in Section 2.19(a).

"Moody's" means Moody's Investors Service, Inc and any successor thereto.

13

"<u>Multiemployer Plan</u>" means "multiemployer plan" as defined in Section 3(37) of ERISA to which the Borrower or any member of the Controlled Group is obligated to make contributions.

"Non-Extending Lender" is defined in Section 2.21(b).

"Note" or "Notes" means the Revolving Notes and/or the Swing Line Note, as applicable.

"Notice Date" is defined in Section 2.21(b).

"Obligations" means all unpaid principal of and accrued and unpaid interest on the Loans, all Reimbursement Obligations and accrued and unpaid interest thereon, all accrued and unpaid fees and all expenses, reimbursements, indemnities and other obligations of the Borrower to any Lender, any Issuer, either Agent or any indemnified party arising under any Loan Document, including interest and fees after commencement of, by or against the Borrower of any proceeding under any the federal bankruptcy laws or any comparable provisions of any applicable state law naming such Person as the debtor in such proceeding, regardless of whether such interest and fees are allowed claims in such proceeding.

"Organization Documents" means, (a) with respect to any corporation, the certificate or articles of incorporation and the bylaws (or equivalent or comparable constitutive documents with respect to any non-U.S. jurisdiction); (b) with respect to any limited liability company, the certificate or articles of formation or organization and operating agreement; and (c) with respect to any partnership, joint venture, trust or other form of business entity, the partnership, joint venture or other applicable agreement of formation or organization and any agreement, instrument, filing or notice with respect thereto filed in connection with its formation or organization with the applicable Governmental Authority in the jurisdiction of its formation or organization and, if applicable, any certificate or articles of formation or organization of such entity.

"Other Taxes" means all present or future stamp or documentary taxes or any other excise or property taxes, charges or similar levies (other than Excluded Taxes) arising from any payment made hereunder or under any other Loan Document or from the execution, delivery or enforcement of, or otherwise with respect to, this Agreement or any other Loan Document.

"Outstanding Credit Exposure" means, as to any Lender at any time, the sum of (a) the aggregate principal amount of its Loans outstanding at such time, plus (b) its Pro Rata Share of the Letter of Credit Obligations at such time, plus (c) its Pro Rata Share of Swing Line Loans at such time.

"Participant" is defined in Section 12.1(d).

"Payment Date" means the last Business Day of each March, June, September and December.

"PBGC" means the Pension Benefit Guaranty Corporation, or any successor thereto.

14

"Person" means any natural person, corporation, firm, joint venture, partnership, limited liability company, association, enterprise, trust or other entity or organization, or any government or political subdivision or any agency, department or instrumentality thereof.

"<u>Plan</u>" means an employee pension benefit plan which is covered by Title IV of ERISA or subject to the minimum funding standards under Section 412 or Section 430 of the Code as to which the Borrower or any member of the Controlled Group may have any liability.

"Platform" is defined in Section 6.1.

"Pricing Schedule" means Schedule III attached hereto identified as such.

"Property" of a Person means any and all property, whether real, personal, tangible, intangible, or mixed, of such Person, or other assets owned or leased by such Person.

"Pro Rata Share" means, with respect to any Lender on any date of determination, the percentage which the amount of such Lender's Commitment is of the Aggregate Commitment (or, if the Commitments have terminated, which such Lender's Outstanding Credit Exposure is of the Aggregate Outstanding Credit Exposure) as of such date, subject to adjustment as provided in Section 2.23. For purposes of determining liability for any indemnity obligation under Section 9.6(c), each Lender's Pro Rata Share shall be determined as of the date the applicable Issuer, the Swing Line Lender or the Administrative Agent notifies the Lenders of such indemnity obligation (or, if such notice is given after termination of this Agreement, as of the date of such termination).

"Public Lender" is defined in Section 6.1.

"Register" is defined in Section 12.1(c).

"Regulation U" means Regulation U of the FRB as from time to time in effect and any successor or other regulation or official interpretation of the FRB relating to the extension of credit by banks for the purpose of purchasing or carrying margin stocks applicable to member banks of the Federal Reserve System.

"Reimbursement Obligations" means, at any time, the aggregate of all obligations of the Borrower then outstanding under Section 2.19 to reimburse (i) the Issuers for amounts paid by the Issuers in respect of any one or more drawings under Letters of Credit and/or (ii) the Lenders for amounts paid by the Lenders to reimburse the Issuers pursuant to Section 2.19.

"Related Parties" means, with respect to any Person, such Person's Affiliates and the partners, directors, officers, employees, agents, trustees and advisors of such Person and of such Person's Affiliates.

"Reportable Event" means a reportable event as defined in Section 4043 of ERISA and the regulations issued under such section, with respect to a Plan, excluding, however, such events

as to which the PBGC has by regulation waived the requirement of Section 4043(a) of ERISA that it be notified within thirty (30) days of the occurrence of such event; <u>provided</u> that a failure to meet the minimum funding standard of Section 412 or Section 430 of the Code and of Section 302 of ERISA shall be a Reportable Event regardless of the issuance of any such waiver of the notice requirement in accordance with either Section 4043(a) of ERISA or Section 412(d) of the Code.

"Required Lenders" means Lenders in the aggregate having more than fifty percent (50%) of the Aggregate Commitment or, if the Aggregate Commitment has been terminated, Lenders in the aggregate holding more than fifty percent (50%) of the Aggregate Outstanding Credit Exposure. The unfunded Commitments of, and the outstanding Loans, Letter of Credit Obligations and participations therein held or deemed held by, any Defaulting Lender shall be excluded for purposes of making a determination of Required Lenders.

"Revolving Note" is defined in Section 2.13(d).

"S&P" means Standard and Poor's Ratings Services, a division of The McGraw Hill Companies, Inc and any successor thereto.

"Schedule" refers to a specific schedule to this Agreement, unless another document is specifically referenced.

"SEC" means the Securities and Exchange Commission or any other U.S. federal governmental authority succeeding to any or all of the functions of the Securities and Exchange Commission.

"Section" means a numbered section of this Agreement, unless another document is specifically referenced.

"Shareholders' Equity" means, as of any date of determination for the Borrower and its Consolidated Subsidiaries on a consolidated basis, shareholders' equity as of that date determined in accordance with GAAP.

"Significant Subsidiary" means, at any time, each Subsidiary which (a) as of the date of determination, owns consolidated assets equal to or greater than fifteen percent (15%) of the consolidated assets of the Borrower and its Subsidiaries or (b) which had consolidated net income from continuing operations (excluding extraordinary items) during the four (4) most recently ended fiscal quarters equal to or greater than fifteen percent (15%) of Consolidated Net Income (excluding extraordinary items) during such period.

"Single Employer Plan" means a Plan, other than a Multiemployer Plan, maintained by the Borrower or any member of the Controlled Group for employees of the Borrower or any member of the Controlled Group.

"Specified Information" is defined in Section 9.11(a).

16

"Subsidiary." of a Person means (a) any corporation more than fifty percent (50%) of the outstanding securities having ordinary voting power of which shall at the time be owned or controlled, directly or indirectly, by such Person or by one or more of its Subsidiaries or by such Person and one or more of its Subsidiaries, (b) any partnership, limited liability company, association, joint venture or similar business organization more than fifty percent (50%) of the ownership interests having ordinary voting power of which shall at the time be so owned or controlled; or (c) any other Person the operations and/or financial results of which are required to be consolidated with those of such first Person in accordance with GAAP. Unless otherwise expressly s tated, all references herein to a "Subsidiary" shall mean a Subsidiary of the Borrower.

"Substantial Portion" means, with respect to the Property of the Borrower and its Subsidiaries, Property which represents more than fifteen percent (15%) of the consolidated assets of the Borrower and its Consolidated Subsidiaries as would be shown in the consolidated financial statements of the Borrower and its Consolidated Subsidiaries as at the beginning of the twelve-month period ending with the month in which such determination is made, or of the Consolidated Net Income (excluding extraordinary items) of the Borrower and its Consolidated Subsidiaries as reflected in the financial statements referred to above.

"Swap Contract" means (a) any and all rate swap transactions, basis swaps, credit derivative transactions, forward rate transactions, commodity swaps, commodity options, forward commodity contracts, equity or equity index swaps or options, bond or bond price or bond index swaps or options or forward bond or forward bond price or forward bond index transactions, interest rate options, forward foreign exchange transactions, cap transaction, floor transactions, collar transactions, currency swap transactions, cross-currency rate swap transactions, currency options, spot contracts, or any other similar transactions or any combination of any of the foregoing (including any options to enter into any of the foregoing), whether or not any such transaction is governed by or subject to any master agreement, and (b) any and all transactions of any kind, and the related confirmations, which are subject to the terms and conditions of, or governed by, any form of master agreement published by the International Swaps and Derivatives Association, Inc., any International Foreign Exchange Master Agreement, or any other master agreement (any such master agreement, together with any related schedules, a "Master Agreement"), including any such obligations or liabilities under any Master Agreement.

"Swing Line Lender" means Bank of America in its capacity as provider of Swing Line Loans, or any successor swing line lender hereunder.

"Swing Line Loan" is defined in Section 2.20(a).

"Swing Line Note" is defined in Section 2.13(d).

"Swing Line Loan Notice" is defined in Section 2.20(b).

"Swing Line Sublimit" means an amount equal to the lesser of (a) \$25,000,000 and (b) the Aggregate Commitments. The Swing Line Sublimit is part of, and not in addition to, the Aggregate Commitments.

"Syndication Agents" means Union Bank and Wells Fargo, each in its capacity as syndication agent hereunder, and not in its individual capacity as a Lender, and any successor thereto.

"Synthetic Lease Obligation" means the monetary obligation of a Person under (a) a so-called synthetic or tax retention lease, pursuant to which notwithstanding the off-balance sheet treatment of the lease obligation the assets are deemed owned by the lessee for U.S. federal income tax purposes, or (b) an agreement for the use or possession of property creating obligations that do not appear on the balance sheet of such Person but which, upon the insolvency or bankruptcy of such Person, would be characterized as the indebtedness of such Person (without regard to accounting treatment).

"Taxes" means all present or future taxes, levies, imposts, duties, deductions, withholdings (including backup withholding), assessments, fees or other charges imposed by any Governmental Authority, including any interest, additions to tax or penalties applicable thereto.

"34 Act Reports" means the periodic reports of the Borrower filed with the SEC on Forms 10-K, 10-Q and 8-K (or any successor forms thereto).

"<u>Total Capitalization</u>" means Total Indebtedness of the Borrower and its Consolidated Subsidiaries plus the sum of (a) Shareholders' Equity (without giving effect to the application of ASC Topic 815) and (b) to the extent not otherwise included in Total Indebtedness or Shareholders' Equity, preferred and preference stock and securities of the Borrower and its Subsidiaries included in a consolidated balance sheet of the Borrower and its Consolidated Subsidiaries in accordance with GAAP.

"<u>Total Indebtedness</u>" means all Indebtedness of the Borrower and its Consolidated Subsidiaries on a consolidated basis (and without duplication) but without giving effect to the application of ASC Topic 860 with respect to transfers of accounts receivable by the Borrower or one or more of its Subsidiaries to a non-Subsidiary, excluding Indebtedness arising under Swap Contracts entered into in the ordinary course of business to hedge bona fide transactions and business risks and not for speculation.

"Type" means, with respect to any Advance, its nature as a Floating Rate Advance or a Eurodollar Advance.

"Union Bank" means Union Bank, N.A. in its individual capacity and its successors.

"Unmatured Default" means an event which but for the lapse of time or the giving of notice, or both, would constitute a Default.

"Unreimbursed Amount" is defined in Section 2.19(f).

"Wells Fargo" means Wells Fargo Bank, National Association in its individual capacity and its successors.

18

"Wholly-Owned Subsidiary" of a Person means (i) any Subsidiary all of the outstanding voting securities of which (except directors' qualifying shares) shall at the time be owned or controlled, directly or indirectly, by such Person or one or more Wholly-Owned Subsidiaries of such Person, or by such Person and one or more Wholly-Owned Subsidiaries of such Person, or (ii) any partnership, limited liability company, association, joint venture or similar business organization one hundred percent (100%) of the ownership interests having ordinary voting power of which (except directors' qualifying shares) shall at the time be so owned or controlled.

1.2 Accounting Principles.

Unless the context otherwise clearly requires, all accounting terms not expressly defined herein shall be construed, and all financial computations required under this Agreement shall be made, in accordance with GAAP, consistently applied; provided that if the Borrower notifies the Administrative Agent that the Borrower wishes to amend any covenant in Article VI to eliminate the effect of any change in GAAP on the operation of such covenant (or if the Administrative Agent notifies the Borrower that the Required Lenders wish to amend any covenant in Article VI for such purpose), then the Borrower's compliance with such covenant shall be determined on the basis of GAAP in effect immediately before the relevant change in GAAP became effective, until either such notice is withdrawn or such covenant is amended in a manner satisfactory to the Borrower and the Required Lenders.

1.3 Letter of Credit Amounts.

Unless otherwise specified herein, the amount of a Letter of Credit at any time shall be deemed to be the stated amount of such Letter of Credit in effect at such time; <u>provided</u> that with respect to any Letter of Credit that, by its terms or the terms of any Issuer Document related thereto, provides for one or more automatic increases in the stated amount thereof, the amount of such Letter of Credit shall be deemed to be the maximum stated amount of such Letter of Credit after giving effect to all such increases, whether or not such maximum stated amount is in effect at such time.

1.4 Times of Day.

Unless otherwise specified, all references herein to times of day shall be references to Eastern time (daylight or standard, as applicable).

ARTICLE II

THE CREDITS

2.1 Commitment.

agrees to make Loans to the Borrower from time to time in amounts not to exceed in the aggregate at any one time outstanding the amount of its Commitment, and (b) each Issuer agrees to issue Letters of Credit for the account of the Borrower from time to time in amounts not to exceed in the aggregate at any one time outstanding the amount of its Letter of Credit Commitment (and each Lender severally agrees to participate in each such Letter of Credit as more fully set forth in Section 2.19); provided (i) that the Aggregate Outstanding Credit Exposure shall not at any time exceed the Aggregate Commitment; (ii) the Outstanding Credit Exposure of any Lender shall not at any time exceed the amount of such Lender's Commitment; and (iii) the outstanding amount of Letter of Credit Obligations shall not exceed the Letter of Credit Sublimit. Subject to the terms of this Agreement, the Borrower may borrow, repay and reborrow Loans at any time prior to the Facility Termination Date. The Commitments shall expire on the Facility Termination Date.

2.2 <u>Required Payments; Termination</u>.

The Borrower shall (a) repay the principal amount of all Advances made to it on the Facility Termination Date and (b) deposit into the LC Collateral Accounts on the Facility Termination Date an amount in immediately available funds equal to the aggregate stated amount of all Letters of Credit that will remain outstanding after the Facility Termination Date.

2.3 Ratable Loans.

Each Advance hereunder shall consist of Loans made from the several Lenders ratably in proportion to their respective Pro Rata Shares.

2.4 Types of Advances; Minimum Amount.

The Advances may be Floating Rate Advances or Eurodollar Advances, or a combination thereof, selected by the Borrower in accordance with Sections 2.8 and 2.9. Each Eurodollar Advance shall be in the amount of \$5,000,000 or a higher integral multiple of \$1,000,000, and each Floating Rate Advance shall be in the amount of \$1,000,000 or an integral multiple thereof.

2.5 Commitment Fee.

The Borrower agrees to pay to the Administrative Agent for the account of each Lender which is not a Defaulting Lender in accordance with its Pro Rata Share, a commitment fee equal to the product of (i) the Commitment Fee Rate times (ii) the actual daily amount by which the Aggregate Commitments exceed the sum of (x) the outstanding principal amount of Loans (other than Swing Line Loans) and (y) the outstanding amount of Letter of Credit Obligations, subject to adjustment as provided in Section 2.23. The Commitment Fee shall be due and payable quarterly in arrears on each Payment Date, commencing with the first Payment Date to occur after the Closing Date, and on the Facility Termination Date. The Commitment Fee shall be calculated quarterly in arrears. For purposes of clarification, Swing Line Loans shall not be considered outstanding for purposes of determining the unused portion of the Aggregate Commitments.

20

2.6 <u>Changes in Aggregate Commitment.</u>

- (a) Optional Commitment Reductions. The Borrower may at any time and from time to time, without penalty, permanently reduce the Aggregate Commitment in whole or in part ratably among the Lenders (according to their respective Pro Rata Shares) in a minimum principal amount of \$5,000,000 and integral multiples of \$1,000,000 in excess thereof, upon at least three (3) Business Days' prior written notice to the Administrative Agent, which notice shall specify the amount of any such reduction; provided that the amount of the Aggregate Commitment may not be reduced below the Aggregate Outstanding Credit Exposure. All accrued commitment fees shall be payable on the effective date of any termination of the obligations of the Lenders to make Loans hereunder.
- (b) <u>Optional Increase of Commitments</u>. The Borrower may at any time and from time to time, upon five (5) days' prior written notice to the Administrative Agent, increase the Commitments (but not the Letter of Credit Sublimit or the Swing Line Sublimit) with additional Commitments from any existing Lender with a Commitment or new Commitments from any other Person selected by the Borrower and reasonably acceptable to the Administrative Agent, the Swing Line Lender and the Issuers; <u>provided</u> that:
 - (i) any such increase shall be in a minimum principal amount of \$10,000,000 and in integral multiples of \$1,000,000 in excess thereof;
 - (ii) no Default or Unmatured Default shall exist and be continuing at the time of any such increase;
 - (iii) no existing Lender shall be under any obligation to increase its Commitment and any such decision whether to increase its Commitment shall be in such Lender's sole and absolute discretion;
 - (iv) (A) any new Lender shall join this Agreement by executing such joinder documents required by the Administrative Agent and/or (B) any existing Lender electing to increase its Commitment shall have executed a commitment agreement reasonably satisfactory to the Administrative Agent;

(v) as a condition precedent to such increase, the Borrower shall deliver to the Administrative Agent (A) a certificate of each of the Borrower, Great Plains and KCPL GMO dated as of the date of such increase (in sufficient copies for each Lender) signed by an Authorized Officer of such Person certifying and attaching the resolutions adopted by such Person approving or consenting to such increase, and (B) a certificate of the Borrower certifying that, before and after giving effect to such increase, (x) the representations and warranties contained in Article V and the other Loan Documents are true and correct in all material respects on and as of the date of such increase, except to the extent that such representations and warranties specifically refer to an earlier date, in which case they are true and correct in all material respects as of such earlier date, and (y) no Default or Unmatured Default exists as of such date; and

21

(vi) after giving effect to the increase in the Commitments (but without giving effect to any increase in the Commitments pursuant to $\underline{\text{Sections 2.6(c)}}$ or $\underline{\text{(d)}}$), the Aggregate Commitment shall not be greater than EIGHT HUNDRED MILLION DOLLARS (\$800,000,000).

The Borrower shall prepay any Loans owing by it and outstanding on the date of any such increase with the proceeds required to be advanced by any Lender increasing its, or any Person providing a new, Commitment under this Section (and pay any additional amounts required pursuant to Article III) to the extent necessary to keep the outstanding Loans ratable with any revised Commitments arising from any nonratable increase in the Commitments under this Section.

- (c) <u>Transfer and Re-Transfer of the Commitments to Great Plains</u>. (i) Subject to Section 4.2 of the Great Plains Credit Agreement, the Borrower and Great Plains may, by joint election in a written notice to the Administrative Agent (which shall promptly provide a copy of such notice to the Lenders) and the "Administrative Agent" under the Great Plains Credit Agreement, transfer from time to time up to TWO HUNDRED MILLION DOLLARS (\$200,000,000) of the unused Commitments to the Commitments (as such term is defined in the Great Plains Credit Agreement) under the Great Plains Credit Agreement (any such reduction, a "<u>KCPL-GPE Transfer"</u>).
- (ii) Subject to Section 4.2, the Borrower and Great Plains may, by joint election in a written notice to the Administrative Agent (which shall promptly provide a copy of such notice to the Lenders) and the "Administrative Agent" under the Great Plains Credit Agreement, re-transfer from time to time any portion (up to \$200,000,000) of the unused Commitments (as such term is defined in the Great Plains Credit Agreement) previously transferred from this Agreement to the Great Plains Credit Agreement pursuant to subclause (c)(i) above back to the Commitments hereunder (any such addition, a "KCPL-GPE Re-Transfer").
- (iii) On the effective date of a KCPL-GPE Transfer, which shall be specified in the notice delivered pursuant to Section 2.6(c)(i) and which shall not be less than five (5) Business Days subsequent to the date of giving of such notice, then subject to the satisfaction of the conditions precedent specified in Section 4.2 of the Great Plains Credit Agreement, (i) the Commitments hereunder shall be ratably decreased by the aggregate amount specified in such notice and (ii) the aggregate amount of the "Commitments" under and as defined in the Great Plains Credit Agreement shall be ratably increased by such amount. Such KCPL-GPE Transfer and t he consequent decreases and increases shall be irrevocable subject, however, to subsequent permissible KCPL-GPE Re-Transfers in accordance with the terms hereof.
- (iv) On the effective date of a KCPL-GPE Re-Transfer, which shall be specified in the notice delivered pursuant to Section 2.6(c)(ii) and which shall not be less than five (5) Business Days subsequent to the date of giving of such notice, then subject to the satisfaction of the conditions precedent specified in Section 4.2, (i) the Commitments hereunder shall be ratably increased by the aggregate amount specified in such notice and (ii) the aggregate amount of the "Commitments" under and as defined in the Great Plains Credit Agreement shall be ratably

22

decreased by such amount. Such KCPL-GPE Re-Transfer and the consequent decreases and increases shall be irrevocable, subject, however, to subsequent permissible KCPL-GPE Transfers in accordance with the terms hereof.

- (d) Transfer and Re-Transfer of the "Commitments" of Great Plains to the Borrower. (i) Subject to Section 4.2, the Borrower and Great Plains may, by joint election in a written notice to the Administrative Agent (which shall promptly provide a copy of such notice to the Lenders) and the "Administrative Agent" under the Great Plains Credit Agreement, transfer from time to time up to TWO HUNDRED MILLION DOLLARS (\$200,000,000) of the unused Commitments (as such term is defined in the Great Plains Credit Agreement) under the Great Plains Credit Agreement to the Commitments hereunder (any such addition, a "GPE-KCPL Transfer").
- (ii) Subject to Section 4.2 of the Great Plains Credit Agreement, the Borrower and Great Plains may, by joint election in a written notice to the Administrative Agent (which shall promptly provide a copy of such notice to the Lenders) and the "Administrative Agent" under the Great Plains Credit Agreement, re-transfer from time to time any portion (up to \$200,000,000) of the unused Commitments previously transferred from the Great Plains Credit Agreement to this Agreement pursuant to <u>subclause (d)(i)</u> above back to the Commitments (as such term is defined in the Great Plains Credit Agreement) under the Great Plains Credit Agreement (any such decrease, a "GPE-KCPL Re-Transfer").
- (iii) On the effective date of a GPE-KCPL Transfer, which shall be specified in the notice delivered pursuant to Section 2.6(d)(i) and which shall not be less than five (5) Business Days subsequent to the date of giving of such notice, then subject to (A) the satisfaction of the conditions precedent specified in Section 4.2 and (B) if necessary in the reasonable discretion of the Administrative Agent, receipt by the Administrative Agent of a certificate of the Borrower dated as of the date of such GPE-KCPL Transfer (in sufficient copies for each Lender) signed by an Authorized Officer of the Borro wer certifying and attaching the resolutions adopted by the Borrower approving or consenting to borrowings up to at least the amount of the Commitments after giving effect to such GPE-KCPL Transfer, (i) the Commitments hereunder shall be ratably increased by the aggregate amount specified in such notice and (ii) the aggregate amount of the "Commitments" under and as defined in the Great Plains Credit Agreement shall be ratably decreased by such amount. Such GPE-KCPL

Transfer and the consequent increases and decreases shall be irrevocable subject, however, to subsequent permissible GPE-KCPL Re-Transfers in accordance with the terms hereof.

(iv) On the effective date of a GPE-KCPL Re-Transfer, which shall be specified in the notice delivered pursuant to Section 2.6(d)(ii) and which shall not be less than five (5) Business Days subsequent to the date of giving of such notice, then subject to the satisfaction of the conditions precedent specified in Section 4.2 of the Great Plains Credit Agreement, (i) the "Commitments" under and as defined in the Great Plains Credit Agreement shall be ratably increased by the aggregate amount specified in such notice and (ii) the aggregate amount of the Commitments hereunder shall be ratably decreased by such amount. Such GPE-K CPL Re-Transfer and the consequent increases and decreases shall be irrevocable, subject, however, to subsequent permissible GPE-KCPL Transfers in accordance with the terms hereof.

23

Notwithstanding the foregoing, the parties acknowledge and agree that after giving effect to any increase in the Commitments pursuant to any KCPL-GPE Transfer or KCPL-GPE Re-Transfer contemplated in subclauses (i) and (ii) of Section 2.6(c) and any GPE-KCPL Transfer or GPE-KCPL Re-Transfer contemplated in subclauses (i) and (ii) of Section 2.6(d), (a) the aggregate Commitments hereunder shall not exceed \$800,000,000, (b) the aggregate Commitments under and as defined in the Great Plains Credit Agreement shall not exceed \$400,000,000 and (c) the aggregate commitments under this Agreement, the Great Plains Credit Agreement and the KCPL GMO Credit Agreement shall not exceed \$1,250,000,000; provided, however, that after giving effect to such KCPL-GPE Transfers, KCPL-GPE Re-Transfers, GPE-KCPL Re-Transfers, Section 2.6(b) and Section 2.6(b) of the Great Plains Credit Agreement, (x) the aggregate Commitments hereunder shall not exceed \$1,000,000,000, (y) the aggregate Commitments under and as defined in the Great Plains Credit Agreement shall not exceed \$600,000,000 and (z) the aggregate commitments under this Agreement, the Great Plains Credit Agreement and the KCPL GMO Credit Agreement shall not exceed \$1,850,000,000.

2.7 Optional Prepayments.

- (a) The Borrower may at any time and from time to time, in whole or in part, prepay Floating Rate Advances upon one (1) Business Day's prior written notice to the Administrative Agent, without penalty or premium. Each partial prepayment of Floating Rate Advances shall be in an aggregate amount of \$1,000,000 or an integral multiple thereof. Each such notice shall specify the date and amount of such prepayment, and the payment amount specified in such notice shall be due and payable on the date specified therein; provided, however, subject to Section 3.4, t hat a notice of prepayment may state that such notice is conditioned upon the effectiveness of other credit facilities or the closing of a capital markets transaction, in which case such notice may be revoked by the Borrower (by notice to the Administrative Agent on or prior to the specified prepayment date) if such condition is not satisfied.
- (b) The Borrower may at any time and from time to time, in whole or in part, prepay Eurodollar Advances (subject to the payment of any funding indemnification amounts required by Section 3.4) upon three (3) Business Days' prior written notice to the Administrative Agent, without penalty or premium. Each partial prepayment of Eurodollar Advances shall be in an aggregate amount of \$5,000,000 or a higher integral multiple of \$1,000,000. Each such notice shall specify the date and amount of such prepayment, and the payment amount specified in such notice shall be due and payable on the date specified therein; provided, however, subject to Section 3.4, that a notice of prepayment may state that such notice is conditioned upon the effectiveness of other credit facilities or the closing of a capital markets transaction, in which case such notice may be revoked by the Borrower (by notice to the Administrative Agent on or prior to the specified prepayment date) if such condition is not satisfied. Any notice to prepay Eurodollar Advances shall indicate the Interest Period for such Eurodollar Advances.
- (c) Subject to Section 2.23, all prepayments of Advances shall be applied ratably to the Loans of the Lenders in accordance with their respective Pro Rata Shares.

24

(d) The Borrower may at any time and from time to time prepay Swing Line Loans upon one (1) Business Day's prior notice to the Swing Line Lender (with a copy to the Administrative Agent), without penalty or premium; provided that (i) such notice must be received by the Swing Line Lender and the Administrative Agent not later than 1:00 p.m. on the date of the prepayment, and (ii) any such prepayment shall be in a minimum principal amount of \$500,000 or a whole multiple of \$100,000 in excess thereof (or, if less, the entire principal thereof then outstanding). Each such notice shall specify the date and amount of such prepayment. If such notice is given by the Borrower, the Borrower s hall make such prepayment and the payment amount specified in such notice shall be due and payable on the date specified therein; provided, however, subject to Section 3.4, that a notice of prepayment may state that such notice is conditioned upon the effectiveness of other credit facilities or the closing of a capital markets transaction, in which case such notice may be revoked by the Borrower (by notice to the Administrative Agent on or prior to the specified prepayment date) if such condition is not satisfied.

2.8 <u>Method of Selecting Types and Interest Periods for New Advances.</u>

The Borrower shall select the Type of Advance and, in the case of each Eurodollar Advance, the Interest Period applicable thereto from time to time. The Borrower shall give the Administrative Agent irrevocable notice (a "Borrowing Notice") not later than noon on the Borrowing Date of each Floating Rate Advance and not later than noon three (3) Business Days before the Borrowing Date for each Eurodollar Advance, specifying:

- (i) the Borrowing Date, which shall be a Business Day, of such Advance,
- (ii) the aggregate amount of such Advance,
- (iii) the Type of Advance selected, and
- (iv) in the case of each Eurodollar Advance, the Interest Period applicable thereto.

Not later than 1:00 p.m. on each Borrowing Date, each Lender shall make available its Loan or Loans in funds immediately available to the Administrative Agent at its address specified pursuant to Article XIII. The Administrative Agent will make the funds so received from the Lenders available to the Borrower at the Administrative Agent's aforesaid address.

2.9 <u>Conversion and Continuation of Outstanding Advances.</u>

Floating Rate Advances shall continue as Floating Rate Advances unless and until such Floating Rate Advances are converted into Eurodollar Advances pursuant to this <u>Section 2.9</u> or are repaid in accordance with <u>Section 2.7</u>. Each Eurodollar Advance shall continue as a Eurodollar Advance until the end of the then applicable Interest Period therefor, at which time such Eurodollar Advance shall be automatically converted into a Floating Rate Advance unless (x) such Eurodollar Advance is or was repaid in accordance with <u>Section 2.7</u> or (y) the Borrower shall have given the Administrative Agent a Conversion/Continuation Notice (as defined below) requesting that, at the end of such Interest Period, such Eurodollar Advance continue as a

25

Eurodollar Advance for the same or another Interest Period. Subject to the terms of <u>Section 2.4</u>, the Borrower may elect from time to time to convert all or any part of a Floating Rate Advance into a Eurodollar Advance. The Borrower shall give the Administrative Agent irrevocable notice (a "<u>Conversion/Continuation Notice</u>") of each conversion of a Floating Rate Advance into a Eurodollar Advance or continuation of a Eurodollar Advance not later than 11:00 a.m. at least three (3) Business Days prior to the date of the requested conversion or continuation, specifying:

- (i) the requested date, which shall be a Business Day, of such conversion or continuation,
- (ii) the aggregate amount and Type of the Advance which is to be converted or continued, and
- (iii) the amount of such Advance which is to be converted into or continued as a Eurodollar Advance and the duration of the Interest Period applicable thereto.

2.10 Interest Rates.

Each Floating Rate Advance shall bear interest on the outstanding principal amount thereof, for each day from and including the date such Advance is made or is automatically converted from a Eurodollar Advance into a Floating Rate Advance pursuant to Section 2.9, to but excluding the date it is paid or is converted into a Eurodollar Advance pursuant to Section 2.9 hereof, at a rate per annum equal to the Alternate Base Rate plus the Applicable Margin for such day. Changes in the rate of interest on that portion of any Advance maintained as a Floating Rate Advance will take effect simultaneously with each change in the Alternate Base Rate. Each Eurodollar Advance shall bear inter est on the outstanding principal amount thereof from and including the first day of the Interest Period applicable thereto to (but not including) the last day of such Interest Period at the Eurodollar Rate plus the Applicable Margin based upon the Borrower's selections under Sections 2.8 and 2.9 and otherwise in accordance with the terms hereof. No Interest Period may end after the Facility Termination Date. Each Swing Line Loan shall bear interest on the outstanding principal amount thereof from the applicable borrowing date at a rate per annum equal to the Alternate Base Rate plus the Applicable Margin for each day from and including the date such Swing Line Loan is made to but excluding the date it is paid. Changes in the rate of interest on that portion of any Swing Line Loan will take effect simultaneously with each change in the Alternate Base Rate.

2.11 Rates Applicable After Default.

Notwithstanding anything to the contrary contained in <u>Section 2.8</u> or <u>2.9</u>, during the continuance of a Default or Unmatured Default the Required Lenders may, at their option, by notice to the Borrower (which notice may be revoked at the option of the Required Lenders notwithstanding any provision of <u>Section 8.2</u> requiring unanimous consent of the Lenders to changes in interest rates), declare that no Advance may be made as, converted into or continued as a Eurodollar Advance. During the continuance of a Default the Required Lenders may, at their option, by notice to the Borrower (which notice may be revoked at the option of the Required Lenders notwithstanding any provision of <u>Section 8.2</u> requiring unanimous consent of

26

the Lenders to changes in interest rates), declare that (i) each Eurodollar Advance shall bear interest for the remainder of the applicable Interest Period at the rate otherwise applicable to such Interest Period plus two percent (2%) per annum, (ii) each Floating Rate Advance shall bear interest at a rate per annum equal to the Alternate Base Rate in effect from time to time plus the Applicable Margin plus two percent (2%) per annum, (iii) the Letter of Credit Fee Rate shall be increased by two percent (2%) per annum and (iv) each Swing Line Loan shall bear interest at a rate per annum equal to the Alternate Base Rate in effect from time to time plus the Applicable Margin plus two percent (2%) per annum; provided that, during the continuance of a Default under Secti ons 7.7 or 7.8, the interest rates set forth in clauses (i), (ii) and (iv) above and the increase in the Letter of Credit Fee Rate set forth in clause (iii) above shall be applicable to all applicable Credit Extensions without any election or action on the part of the Administrative Agent or any Lender.

2.12 Method of Payment.

Except to the extent that any Tax is required to be withheld or deducted under applicable Law, but subject to the provisions of Article III, all payments of the Obligations hereunder shall be made, without setoff, deduction, or counterclaim, in immediately available funds to the Administrative Agent at the Administrative Agent among the Lenders in accordance with their respective Pro Rata Shares. Each payment delivered to the Administrative Agent for the account of any Lender shall be delivered promptly by the Administrative Agent to such Lender in the same type of funds that the Administrative Agent received at the Administrative Agent's Office from such Lender.

2.13 <u>Noteless Agreement; Evidence of Indebtedness.</u>

- (a) Each Lender shall maintain in accordance with its usual practice an account or accounts evidencing the indebtedness of the Borrower to such Lender resulting from each Loan made by such Lender from time to time, including the amounts of principal and interest payable and paid to such Lender from time to time hereunder.
- (b) The Administrative Agent shall also maintain accounts in which it will record (i) the amount of each Loan (including any Swing Line Loan) made hereunder, the Type thereof and the Interest Period with respect thereto, (ii) the amount of any principal or interest due and payable or to become due and payable from the Borrower to each Lender hereunder, (iii) the original stated amount of each Letter of Credit and the amount of Letter of Credit Obligations outstanding at any time and (iv) the amount of any sum received by the Administrative Agent hereunder from the Borrower and each Lender's share thereof.
- (c) The entries maintained in the accounts maintained pursuant to <u>clauses (a)</u> and <u>(b)</u> above shall be *prima facie* evidence of the existence and amounts of the Obligations therein recorded; <u>provided</u> that the failure of the Administrative Agent or any Lender to maintain such accounts or any error therein shall not in any manner affect the obligation of the Borrower to repay the Obligations in accordance with their terms.

27

(d) Any Lender may request that its Loans be evidenced by a promissory note substantially in the form of Exhibit D (a "Revolving Note"). In such event, the Borrower shall prepare, execute and deliver to such Lender a Note payable to the order of such Lender. Thereafter, the Loans evidenced by such Note and interest thereon shall at all times (including after any assignment pursuant to Section 12.1(b)) be represented by one or more Notes payable to the order of the payee named therein or any assignee pursuant to Section 12.1(b), except to the extent that any such Lender or assignee subsequently returns any such Note for cancellation and requests that such Loans once again be evidenced as described in clauses (a) and (b) above. The Swing Line Loans shall be evidenced by a promissory note substantially in the form of Exhibit E (the "Swing Line Note").

2.14 <u>Telephonic Notices</u>.

The Borrower hereby authorizes the Lenders and the Administrative Agent to extend, convert or continue Advances, effect selections of Types of Advances and to transfer funds based on telephonic notices made by any person or persons the Administrative Agent or any Lender in good faith believes to be acting on behalf of the Borrower. The Borrower agrees to deliver promptly to the Administrative Agent a written confirmation, if such confirmation is requested by the Administrative Agent or any Lender, of each telephonic notice signed by an Authorized Officer. If the written confirmation differs in any material respect from the action taken by the Administrative Agent and the Lenders, the records of the Administrative Agent and the Lenders shall govern absent manifest error.

2.15 <u>Interest Payment Dates; Interest and Fee Basis.</u>

Interest accrued on each Floating Rate Advance and each Swing Line Loan shall be payable on each Payment Date, commencing with the first such date to occur after the date hereof, and at maturity. Interest accrued on each Eurodollar Advance shall be payable on the last day of its applicable Interest Period, on any date on which such Eurodollar Advance is prepaid, whether by acceleration or otherwise, and at maturity. Interest accrued on each Eurodollar Advance having an Interest Period longer than three (3) months shall also be payable on the last day of each three (3)-month interval during such Interest Period. All computations of interest for Floating Rate Loans and Swing Line Loans (including Floating Rate Loans determined by reference to the Eurodollar Rate) shall be made on the basis of a year of 365 or 366 days, as the case may be, and actual days elapsed. All other computations of interest and fees shall be calculated for actual days elapsed on the basis of a 360-day year. Interest shall be payable for the day an Advance is made but not for the day of any payment on the amount paid if payment is received prior to 1:00 p.m. at the place of payment (it being understood that the Administrative Agent shall be deemed to have received a payment prior to 1:00 p.m. if (x) the Borrower has provided the Administrative Agent with evidence satisfactory to the Administrative Agent that the Borrower has initiated a wire transfer of such payment prior to such time and (y) the Administrative Agent actually receives such payment on the same Business Day on which such wire transfer was initiated). If any payment of principal of or interest on an Advance shall become due on a day which is not a Business Day, such payment shall be made on the next succeeding Business Day and, in the case of a principal payment, such extension of time shall be included in computing interest in connection with such payment.

28

2.16 <u>Notification of Advances, Interest Rates, Prepayments and Commitment Reductions.</u>

Promptly after receipt thereof, the Administrative Agent will notify each Lender of the contents of each Aggregate Commitment reduction notice, Borrowing Notice, Conversion/Continuation Notice, and repayment notice received by it hereunder. The Administrative Agent will notify each Lender of the interest rate applicable to each Eurodollar Advance promptly upon determination of such interest rate and will give each Lender prompt notice of each change in the Alternate Base Rate. The Administrative Agent will also promptly notify each Lender of any increase or reduction of the Aggregate Commitments pursuant to the terms hereof. The Administrative Agent will also promptly notify the Borrower of the interest rate applicable to each Eurodollar Advance and Floating Rate Advance promptly upon determination of such interest rates and will give the Borrower prompt notice of each change in the Alternate Base Rate.

2.17 <u>Lending Installations</u>.

Each Lender may book its Loans at any Lending Installation selected by such Lender and may change its Lending Installation from time to time. All terms of this Agreement shall apply to any such Lending Installation and the Loans and any Notes issued hereunder shall be deemed held by each Lender for the benefit of such Lending Installation. Each Lender may, by written notice to the Administrative Agent and the Borrower in accordance with Article XIII, designate replacement or additional Lending Installations through which Loans will be made by it and for whose account Loan payments are to be made;

<u>provided</u>, <u>however</u>, that such designation shall be permitted by applicable Laws and that such designation shall not result in the imposition of increased costs or obligations on the Borrower.

2.18 Non-Receipt of Funds by the Administrative Agent.

Unless a Lender notifies the Administrative Agent, prior to the proposed date of any Eurodollar Advance (or, in the case of any Floating Rate Advance, prior to noon on the date of such Advance), that such Lender will not make available to the Administrative Agent such Lender's share of such Advance, the Administrative Agent may assume that such payment has been made. Unless the Borrower notifies the Administrative Agent, prior to the date on which it is scheduled to make a payment to the Administrative Agent of principal, interest or fees for the account of the Lenders, that it does not intend to make such payment, the Administrative Agent may assume that such payment has been made. The Administrative Agent may, but shall not be obligated to, make the amount of such payment available to the intended recipient in reliance upon such assumption. If such Lender or the Borrower, as the case may be, has not in fact made such payment to the Administrative Agent, the recipient of such payment shall, on demand by the Administrative Agent, repay to the Administrative Agent the amount so made available together with interest thereon in respect of each day during the period commencing on the date such amount was so made available by the Administrative Agent until the date the Administrative Agent recovers such amount at a rate per annum equal to (x) in the case of

29

payment by a Lender, the Federal Funds Rate for such day or (y) in the case of payment by the Borrower, the interest rate applicable to the relevant Loan.

2.19 Letters of Credit.

- (a) <u>Issuance</u>. Each Issuer hereby agrees, on the terms and conditions set forth in this Agreement and in reliance upon the agreements of the Lenders set forth in this <u>Section 2.19</u>, to issue Letters of Credit and to extend, increase, decrease or otherwise modify Letters of Credit ("<u>Modify</u>," and each such action a "<u>Modification</u>") in amounts not to exceed in the aggregate at any one time outstanding the amount of its Letter of Credit Commitment from time to time on any Business Day from a nd including the date of this Agreement and prior to the Facility Termination Date upon the request of the Borrower; <u>provided</u> that immediately after each such Letter of Credit is issued or Modified, (x) the Aggregate Outstanding Credit Exposure shall not exceed the Aggregate Commitment and (y) the outstanding amount of Letters of Credit Obligations shall not exceed the Letter of Credit Sublimit. No Letter of Credit shall have an expiry date later than the date that is five (5) days prior to the scheduled Facility Termination Date. All Existing Letters of Credit shall be deemed to have been issued pursuant hereto, and from and after the Closing Date shall be subject to and governed by the terms and conditions hereof.
- (b) <u>Participations</u>. Upon the issuance or Modification by any Issuer of a Letter of Credit in accordance with this <u>Section 2.19</u>, such Issuer shall be deemed, without further action by any Person, to have unconditionally and irrevocably sold to each Lender, and each Lender shall be deemed, without further action by any Person, to have unconditionally and irrevocably purchased from such Issuer, a participation in such Letter of Credit (and each Modification thereof) and the related Letter of Credit Obligations in proportion to its Pro Rata Share.
- (c) Notice. Subject to Section 2.19(a), the Borrower shall give the applicable Issuer and the Administrative Agent notice prior to 11:00 a.m. at least three (3) Business Days (or such lesser period of time as the Administrative Agent and such Issuer may agree in their sole discretion) prior to the proposed date of issuance or Modification of each Letter of Credit, specifying, in form and detail reasonably satisfactory to the applicable Issuer, (A) the proposed issuance date of the requested Letter of Credit (which shall be a Business Day) and the expiry date of such Letter of Credit; (B) the amount thereof; (C) the name and a ddress of the beneficiary thereof; (D) the documents to be presented by such beneficiary in case of any drawing thereunder; (F) the purpose and nature of the requested Letter of Credit; and (G) such other matters as the applicable Issuer may reasonably require. In the case of a request for a Modification of any outstanding Letter of Credit, such notice shall specify in form and detail reasonably satisfactory to the applicable Issuer (A) the Letter of Credit to be amended; (B) the proposed date of amendment thereof (which shall be a Business Day); (C) the nature of the proposed amendment; and (D) such other matters as the applicable Issuer may reasonably require. Upon receipt of such notice, the applicable Issuer shall promptly notify the Administrative Agent, and the Administrative Agent shall promptly notify each Lender, of the contents thereof and of the amount of such Lender 217;s participation in such proposed Letter of Credit. The issuance or Modification by an Issuer of any Letter of Credit shall, in addition to the

30

conditions precedent set forth in Article IV (the satisfaction of which such Issuer shall have no duty to ascertain, it being understood, however, that such Issuer shall not issue any Letter of Credit if it has received written notice from the Borrower, the Administrative Agent or any Lender one (1) day prior to the proposed date of issuance, that any such condition precedent has not been satisfied), be subject to the conditions precedent that such Letter of Credit shall be satisfactory to such Issuer and that an Authorized Officer of the Borrower shall have executed and delivered such application agreement and/or such other instruments and agreements relating to such Letter of Credit as such Issuer shall have reasonably requested (each a "Letter of Credit Application"). Promptly after its delivery of any Letter of Credit or any amendment to a Letter of Credit to an advising bank with respect thereto or to the beneficiary thereof, the applicable Issuer will also deliver to the Borrower and the Administrative Agent a true and complete copy of such Letter of Credit or amendment. In the event of any conflict between the terms of this Agreement and the terms of any Letter of Credit Application, the terms of this Agreement shall control.

(d) <u>Letter of Credit Fees</u>. Upon the issuance of each Letter of Credit and until termination, cancellation or expiration of such Letter of Credit, the Borrower agrees to pay to the Administrative Agent, for the account of the Lenders ratably in accordance with their respective Pro Rata Shares, with respect to such Letter of Credit, a letter of credit fee (the "<u>Letter of Credit Fee</u>") at a per annum rate equal to the Letter of Credit Fee Rate in effect from time to time multiplied by the daily maximum amount available under such Letter of Credit, such fee to be payable in arrears on each Payment Date, on the Facility Terminati on Date and, if applicable, thereafter on demand; <u>provided</u>, however, any Letter of Credit Fees otherwise payable for the account of a Defaulting Lender with respect to any Letter of Credit as to which such Defaulting Lender has not provided Cash Collateral satisfactory to the applicable Issuer pursuant to this <u>Section 2.19</u> shall be payable, to the maximum extent permitted by applicable Law, to the other Lenders in accordance with the upward adjustments in their respective Pro Rata Shares allocable to such Letter of Credit pursuant to <u>Section 2.23(a)(iv)</u>, with the balance of such fee, if any, payable

to the applicable Issuer for its own account. The Borrower shall also pay to each Issuer for its own account (x) a fronting fee of 0.20% per annum on the aggregate maximum stated amount for each Letter of Credi t issued by such Issuer and currently outstanding, with such fee to be payable in arrears on the first Business Day following the end of each March, June, September and December, and (y) documentary and processing charges in connection with the issuance or Modification of and draws under Letters of Credit in accordance with such Issuer's standard schedule for such charges as in effect from time to time.

(e) Administration; Reimbursement by Lenders. Upon receipt from the beneficiary of any Letter of Credit of any demand for payment under such Letter of Credit, the applicable Issuer shall notify the Administrative Agent and the Borrower and the Administrative Agent shall promptly notify each Lender of the amount to be paid by such Issuer as a result of such demand and the proposed payment date (the "Letter of Credit Payment Date"). After honoring any demand for payment under any Letter of Credit, the applicable Issuer shall, upon the request of the Borrower, send to the Borrower the beneficiary statement or an accurate copy of such beneficiary statement pursuant to which the applicable beneficiary requested payment. The responsibility of any Issuer to the Borrower and each Lender shall be only to determine that the documents delivered under each Letter of Credit issued by such Issuer in connection with a demand for payment are in conformity in all material respects with such Letter of Credit. Each

31

Issuer shall endeavor to exercise the same care in its issuance and administration of Letters of Credit as it does with respect to letters of credit in which no participations are granted, it being understood that in the absence of any gross negligence or willful misconduct by such Issuer, each Lender shall be unconditionally and irrevocably obligated, without regard to the occurrence of any Default or any condition precedent whatsoever, to reimburse (and the Administrative Agent may apply Cash Collateral for this purpose) such Issuer on demand for (i) such Lender's Pro Rata Share of the amount of each payment made by such Issuer under each Letter of Credit to the extent such amount is not reimbursed by the Borrower pursuant to Section 2.19(f) below, plus, without limiting the other provisions of this Agreement, (ii) interest on the foregoing amount, for each day from the date of the applicable payment by such Issuer to the date on which such Issuer is reimbursed by such Lender for its Pro Rata Share thereof, at a rate per annum equal to the greater of the Federal Funds Rate and a rate determined by the applicable Issuer in accordance with banking industry rules on interbank compensation, plus (iii) any administrative, processing or similar fees customarily charged by the applicable Issuer in connection with the foregoing.

Reimbursement by Borrower. (i) The Borrower shall be irrevocably and unconditionally obligated to reimburse each Issuer through the Administrative Agent, for any amount to be paid by such Issuer upon any drawing under any Letter of Credit without presentment, demand, protest or other formalities of any kind, not later than (i) the applicable Letter of Credit Payment Date, if the Borrower shall have received such notice on or prior to 11:00 a.m. on such Letter of Credit Payment Date or (ii) on the Business Day immediately following the applicable Letter of Credit Payment Date, if the Borrower shall have received such notice after 11:00 a.m. (either of such dates, the "Honor Date"); provided that the Borrower shall not be precluded from asserting any claim for direct (but not consequential) damages suffered by the Borrower which the Borrower proves were caused by (i) the willful misconduct or gross negligence of such Issuer in determining whether a request presented under any Letter of Credit complied with the terms of such Letter of Credit or (ii) such Issuer's failure to pay under any Letter of Credit after the presentation to it of a request strictly complying with the terms and conditions of such Letter of Credit. The Administrative Agent will pay to each Lender ratably in accordance with its Pro Rata Share all amounts received by it from the Borrower for application in payment, in whole or in part, of the Reimbursement Obligation in respect of any Letter of Credit, but only to the extent such Lender made payment to the applicable Issuer in respect of such Letter of Credit pursuant to Section 2.19(e). If the Borrower fails to so reimburse any Issuer for any amounts paid by any Issuer by such applicable time, the Administrative Agent shall promptly notify each Lender of the Honor Date, the amount of the unreimbursed drawing (the "Unreimbursed Amount"), and the amount of such Lender's Pro Rata Share thereof. In such event, the Borrower shall be deemed to have requested a Floating Rate Advance to be disbursed on the Honor Date in an amount equal to the Unreimbursed Amount and, so long as no Unmatured Default or Default has occurred and is continuing, such disbursement shall be deemed to occur automatically without further act and without regard to the minimum and multiples specified in Section 2.4 for the principal amount of Floating Rate Advance but subject to the conditions set forth in Section 4.2 (other than delivery by the Borrower of a Borrowing Notice) and provided that, after giving effect to such Floating Rate Advance, the Aggregate Outstanding Credit Exposure shall not exceed the Aggregate Commitments. To the extent that any amounts paid by an Issuer are not reimbursed by the Borrower pursuant to the terms of this Section 2.19(f), all such amounts shall bear interest, payable on demand, for each day until paid

32

at a rate per annum equal to the sum of two percent (2%) plus the rate applicable to Floating Rate Advances. Any notice given by any Issuer or the Administrative Agent pursuant to this Section 2.19(f)(i) may be given by telephone if immediately confirmed in writing; provided that the lack of such an immediate confirmation shall not affect the conclusiveness or binding effect of such notice. Until each Lender funds its Floating Rate Loan to reimburse the applicable Issuer for any amount drawn under any Letter of Credit, interest in respect of such Lender's Pro Rata Share of such amount shall be solely for the account of such Issuer. Each Lender shall upon any notice pursuan t to this Section 2.19(f) make funds available (and the Administrative Agent may apply Cash Collateral provided for this purpose) for the account of the applicable Issuer at the Administrative Agent's Office in an amount equal to its Pro Rata Share of the Unreimbursed Amount not later than 1:00 p.m. on the Business Day specified in such notice by the Administrative Agent. The Administrative Agent shall remit the funds so received to the applicable Issuer.

(ii) Each Lender's obligation to make Floating Rate Loans to reimburse an Issuer for amounts drawn under Letters of Credit, as contemplated by this Section 2.19(f), shall be absolute and unconditional and shall not be affected by any circumstance, including (A) any setoff, counterclaim, recoupment, defense or other right which such Lender may have against the applicable Issuer, the Borrower or any other Person for any reason whatsoever; (B) the occurrence or continuance of a Default, or (C) any other occurrence, event or condition, whether or not similar to any of the foregoing; provided, however, that each Lender's obligation to make Floating Rate Loans is subject to the conditions set forth in Section 4.2 (other than delivery by the Borrower of a Borrowing Notice).

If any payment received by the Administrative Agent for the account of any Issuer pursuant to Section 2.19(f) is required to be returned under any of the circumstances described in Section 11.3 (including pursuant to any settlement entered into by an Issuer in its discretion), each Lender shall pay to the Administrative Agent for the account of the applicable Issuer its Pro Rata Share thereof on demand of the Administrative Agent, plus interest thereon from

the date of such demand to the date such amount is returned by such Lender, at a rate per annum equal to the Federal Funds Rate from time to time in effect. The obligations of the Lenders under this paragraph shall survive the payment in full of the Obligations and the termination of this Agreement.

Obligations Absolute. Except to the extent that any Tax is required to be withheld or deducted under applicable Law, but subject to the provisions of Article III, the Borrower's obligations under this Section 2.19 shall be absolute, unconditional and irrevocable under any and all circumstances and irrespective of any setoff, counterclaim or defense to payment which the Borrower or any Subsidiary may have or have had against any Issuer, any Lender or any beneficiary or transferee of a Letter of Credit. The Borrower further agrees with the Issuers and the Len ders that neither any Issuer nor any Lender shall be responsible for, and the Borrower's Reimbursement Obligation in respect of any Letter of Credit shall not be affected by, among other things, (i) the validity, enforceability or genuineness of documents or of any endorsements thereon, even if such documents should in fact prove to be in any or all respects invalid, fraudulent or forged or any statement therein untrue or inaccurate, (ii) any dispute between or among the Borrower, any of its Affiliates, the beneficiary of any Letter of Credit or any

33

financing institution or other party to whom any Letter of Credit may be transferred or any claims or defenses whatsoever of the Borrower or of any of its Affiliates against the beneficiary of any Letter of Credit or any such transferee, (iii) any payment by an Issuer under such Letter of Credit against presentation of a draft or certificate that does not strictly comply with the terms of such Letter of Credit, (iv) any payment made by an Issuer under such Letter of Credit to any Person purporting to be a trustee in bankruptcy, debtor-in-possession, assignee for the benefit of creditors, liquidator, receiver or other representative of or successor to any beneficiary or any transferee of such Letter of Credit, including any arising in connection with any proceeding under any Debtor Relief Law or (v) any other circumstance or happening what soever, whether or not similar to any of the foregoing, including any other circumstance that might otherwise constitute a defense available to, or a discharge of, the Borrower or any Subsidiary. No Issuer shall be liable for any error, omission, interruption or delay in transmission, dispatch or delivery of any message or advice, however transmitted, in connection with any Letter of Credit other than, with respect to any Issuer, any such error, omission, interruption or delay in transmission, dispatch or delivery of any message or advice resulting from the gross negligence or willful misconduct of such Issuer. The Borrower agrees that any action taken or omitted by any Issuer or any Lender under or in connection with any Letter of Credit and the related drafts and documents, if done without gross negligence or willful misconduct, shall be binding upon the Borrower and shall not put any Issuer or any Lender under any liability to the Bor rower. The Borrower shall promptly examine a copy of each Letter of Credit and each amendment thereto that is delivered to it and, in the event of any claim of noncompliance with the Borrower's instructions or other irregularity, the Borrower will immediately notify the applicable Issuer. The Borrower shall be conclusively deemed to have waived any such claim against the applicable Issuer and its correspondents unless such notice is given as aforesaid. Nothing in this Section 2.19(g) is intended to limit the right of the Borrower to make a claim against any Issuer for damages as contemplated by the proviso to the first sentence of Section 2.19(f).

(h) Actions of the Issuers. Each Issuer shall be entitled to rely, and shall be fully protected in relying, upon any Letter of Credit, draft, writing, resolution, notice, consent, certificate, affidavit, letter, cablegram, telegram, facsimile, telex or teletype message, statement, order or other document believed by it to be genuine and correct and to have been signed, sent or made by the proper Person or Persons, and upon advice and statements of legal counsel, independent accountants and other experts selected by such Issuer. Each Lender and the Borrower agree that, in paying any drawing under a Letter of Credit, the Issuers shall not have any responsibility to obtain any document (other than any sight draft, certificates and documents expressly required by such Letter of Credit) or to ascertain or inquire as to the validity or accuracy of any such document or the authority of the Person executing or delivering any such document. Each Issuer shall be fully justified in failing or refusing to take any action under this Agreement unless it shall first have received such advice or concurrence of the Required Lenders as it reasonably deems appropriate or it shall first be indemnified to its reasonable satisfaction by the Lenders against any and all liability and expense which may be incurred by it by reason of taking or continuing to take any such action. Notwithstanding any other provision of this Section 2.19, each Issuer shall in all cases be fully protected in acting, or in refraining from acting, under this Agreement in accordance with a request of the Required Lenders, and such request and any action taken o r failure to act pursuant thereto shall be binding upon the Lenders and any future holder of a participation in any Letter of Credit issued by such Issuer. The

34

Borrower hereby assumes all risks of the acts or omissions of any beneficiary or transferee with respect to its use of any Letter of Credit; provided, however, that this assumption is not intended to, and shall not, preclude the Borrower's pursuing such rights and remedies as it may have against the beneficiary or transferee at law or under any other agreement. None of the Issuers, the Administrative Agent, any of their respective Related Parties nor any correspondent, participant or assignee of any of the Issuers shall be liable or responsible for any of the matters described in clauses (i) through (v) of Section 2.19(g) except, with respect to any Issuer, for any such matters that are determined by a court of competent jurisdiction by final and non-appealable judgment to have resulted from the gross negligence or willful misconduct of such Person. In furtherance and not in limitation of the foregoing, the Issuers may accept documents that appear on their face to be in order, without responsibility for further investigation, regardless of any notice or information to the contrary, and the Issuers shall not be responsible for the validity or sufficiency of any instrument transferring or assigning or purporting to transfer or assign a Letter of Credit or the rights or benefits thereunder or proceeds thereof, in whole or in part, which may prove to be invalid or ineffective for any reason.

(i) Indemnification. The Borrower agrees to indemnify and hold harmless each Lender, each Issuer and the Administrative Agent, and their respective directors, officers, agents and employees, from and against any and all claims and damages, losses, liabilities, costs or expenses which such Person may incur (or which may be claimed against such Person by any other Person whatsoever) by reason of or in connection with the issuance, execution and delivery or transfer of or payment or failure to pay under any Letter of Credit or any actual or proposed use of any Letter of Credit, including any claims, damages, losses, liabilities, costs or expenses which any Issuer may incur by reason of or in connection with (i) the failure of any other Lender to fulfill or comply with its obligations to such Issuer hereunder (but nothing herein contained shall affect any right the Borrower may have against any defaulting Lender) or (ii) by reason of or on account of such Issuer issuing any Letter of Credit which specifies that the term "Beneficiary" therein includes any successor by operation of law of the named Beneficiary, but which Letter of Credit does not require that any drawing by any such successor Beneficiary be accompanied by a copy of a legal document, satisfactory to such Issuer, evidencing the appointment of such successor Beneficiary; provided that the Borrower shall not be required to indemnify any Person for any claims, damages, losses, liabilities, costs or expenses to the extent, but only to the extent, caused by (x) the willful misconduct or gross negligence of any Issuer in determining whether a request presented under any Let ter of Credit issued by such Issuer complied with the terms of such Letter of Credit or (y) any Issuer's failure to pay under any Letter of Credit issued by it after the presentation to it of a

request strictly complying with the terms and conditions of such Letter of Credit. Nothing in this <u>Section 2.19(i)</u> is intended to limit the obligations of the Borrower under any other provision of this Agreement.

- (j) The Issuers' Obligation to Issue Letters of Credit. No Issuer shall be under any obligation to issue any Letter of Credit if:
- (i) any order, judgment or decree of any Governmental Authority or arbitrator shall by its terms purport to enjoin or restrain such Issuer from issuing such Letter of Credit, or any law applicable to such Issuer or any request or directive (whether or not having the force of law) from any Governmental Authority with jurisdiction over such Issuer shall prohibit, or request that such Issuer refrain from, the issuance of letters of

35

credit generally or such Letter of Credit in particular or shall impose upon such Issuer with respect to such Letter of Credit any restriction, reserve or capital requirement (for which such Issuer is not otherwise compensated hereunder) not in effect on the Closing Date, or shall impose upon such Issuer any unreimbursed loss, cost or expense which was not applicable on the Closing Date and which such Issuer in good faith reasonably deems material to it; <u>provided</u>, <u>however</u>, that in the event a Lender participating in the Letters of Credit is not affected by any such restriction, requirement or imposition, and is able to issue such Letter of Credit and expressly agrees in its sole discretion to issue such Letter of Credit, such Lender, subject to the consent of the Administrative Agent, such consent not to be unreasonably withheld, conditioned or delayed, shall be added as an Issuer pursuant to <u>Section 2.19(n)</u> and shall issue such Letter of Credit and shall be deemed the Issuer with regard to such Letter of Credit for all purposes of this Agreement;

- (ii) the issuance of such Letter of Credit would violate one or more policies of such Issuer applicable to letters of credit generally;
- (iii) except as otherwise agreed by the Administrative Agent and the applicable Issuer, such Letter of Credit is in an initial stated amount less than \$250,000;
 - (iv) such Letter of Credit is to be denominated in a currency other than Dollars; or
- (v) any Lender is at that time a Defaulting Lender, unless each Issuer has entered into arrangements, including the delivery of Cash Collateral, satisfactory to the Issuers (in their sole discretion) with the Borrower or such Lender to eliminate the Issuers' actual or potential Fronting Exposure (after giving effect to Section 2.23(a)(iv)) with respect to the Defaulting Lender arising from either the Letter of Credit then proposed to be issued or that Letter of Credit and all other Letter of Credit Obligations as to which the Issuers have actual or potential Fronting Exposure, as it may elect in their sole discretion.
- (k) <u>Rights as a Lender</u>. In its capacity as a Lender, each Issuer shall have the same rights and obligations as any other Lender.
- (l) <u>Letters of Credit Issued for Subsidiaries</u>. Notwithstanding that a Letter of Credit issued or outstanding hereunder is in support of any obligations of, or is for the account of, a Subsidiary, the Borrower shall be obligated to reimburse the Issuers hereunder for any and all drawings under such Letter of Credit. The Borrower hereby acknowledges that the issuance of Letters of Credit for the account of Subsidiaries inures to the benefit of the Borrower, and that the Borrower's business derives substantial benefits from the businesses of such Subsidiaries.
- (m) <u>Applicability of ISP</u>. Unless otherwise expressly agreed by each Issuer and the Borrower when a Letter of Credit is issued, the rules of the ISP shall apply to each Letter of Credit.

36

(n) Replacement or Addition of Issuer. Any Issuer may be replaced or added at any time by written agreement among the Borrower, the Administrative Agent (unless, in the case of the replacement of any Issuer, the successor Issuer is a Lender and, if applicable, such agreement not to be unreasonably withheld, conditioned or delayed) and the successor or additional Issuer, as applicable; provided that such new Issuer shall be assigned all or a portion, as applicable, of the Letter of Credit Commitment of another Issuer. The Administrative Agent shall notify the Lenders of any such replacement or addition, as applicable, of any Issuer. Where any Issuer is replaced, at the time such replacement shall become effective, the Borrower shall pay all unpaid fees accrued for account of the replaced Issuer. Furthermore, from and after the effective date of such replacement, the successor Issuer, shall have all the rights and obligations of the replaced Issuer under this Agreement with respect to Letters of Credit to be issued thereafter. References herein to the term "Issuer" shall be deemed to refer to any successor or additional Issuer, as applicable, and all previous Issuers, as the context shall require. After the replacement of any Issuer hereunder, the replaced Issuer shall remain a party hereto and shall continue to have all the rights and obligations of an Issuer under this Agreement with respect to Letters of Credit issued by it prior to such replacement, but shall not be required to issue additional Letters of Credit.

2.20 Swing Line Loans.

(a) <u>Swing Line Facility.</u> From and including the date of this Agreement and prior to the Facility Termination Date, subject to the terms and conditions set forth in this Agreement, the Swing Line Lender, in reliance upon the agreements of the other Lenders set forth in this <u>Section 2.20</u>, shall, unless (i) any Lender at such time is a Defaulting Lender and (ii) the Swing Line Lender has not entered into arrangements satisfactory to it with the Borrower or such Defaulting Lender to eliminate the Swing Line Lender's risk with respect to such Defaulting Lender, in which case the Swing Line Lender may in its

discretion, make loans (eac h such loan, a "Swing Line Loan") to the Borrower in Dollars from time to time in an aggregate amount not to exceed at any time outstanding the amount of the Swing Line Sublimit; provided, however, that after giving effect to any Swing Line Loan, (i) the Aggregate Outstanding Credit Exposure shall not exceed the Aggregate Commitments, and (ii) the Outstanding Credit Exposure of any Lender (other than the Swing Line Lender) shall not exceed such Lender's Commitment, and provided, further, that the Borrower shall not use the proceeds of any Swing Line Loan to refinance any outstanding Swing Line Loan. Within the foregoing limits, and subject to the other terms and conditions hereof, the Borrower may borrow Swing Line Loans under this Section 2.20, prepay Swing Line Loans under Section 2.7, and reborrow Swing Line Loans under this Section 2.20. Each Swing Line Loan shall bear interest at the Alternate Base Rate plus the Applicable Margin. Immediately upon the making of a Swing Line Loan, each Lender shall be deemed to, and hereby irrevocably and unconditionally agrees to, purchase from the Swing Line Lender a risk participation in such Swing Line Loan in an amount equal to such Lender's Pro Rata Share times the amount of such Swing Line Loan.

(b) <u>Borrowing Procedures</u>. Each Swing Line Loan shall be made upon the Borrower's irrevocable notice to the Swing Line Lender and the Administrative Agent, which may be given by telephone. Each such notice must be received by the Swing Line Lender and the Administrative Agent not later than 1:00 p.m. on the requested borrowing date, and shall

37

specify (i) the amount to be borrowed, which shall be a minimum principal amount of \$500,000 and integral multiples of \$100,000 in excess thereof, and (ii) the requested borrowing date, which shall be a Business Day. Each such telephonic notice must be confirmed promptly by delivery to the Swing Line Lender and the Administrative Agent of a written notice (the "Swing Line Loan Notice"). Promptly after receipt by the Swing Line Lender of any telephonic Swing Line Loan Notice, the Swing Line Lender will confirm with the Administrative Agent (by telephone or in writing) that the Administrative Agent has also received such Swing Line Loan Notice and, if not, the Swing Line Lender will notify the Administrative Agent (by telephone or in writing) of the contents thereof. Unless the Swing Line Lender has received notice (by telephone or in writing) from the Administrative Agent (including at the request of any Lender) prior to 2:00 p.m. on the date of the proposed Advance of Swing Line Loans (A) directing the Swing Line Lender not to make such Swing Line Loan as a result of the limitations set forth in the first proviso to the first sentence of Section 2.20(a), or (B) that one or more of the applicable conditions specified in Article IV is not then satisfied, then, subject to the terms and conditions hereof, the Swing Line Lender will, not later than 3:00 p.m. on the borrowing date specified in such Swing Line Loan Notice, make the amount of its Swing Line Loan available to the Borrower by means of a credit to the general deposit account of the Borrower with the Swing Line Lender or such other account specified by the Borrower to the Swing Line Lender.

(c) <u>Refinancing of Swing Line Loans.</u>

- (i) The Swing Line Lender at any time in its sole and absolute discretion may request, on behalf of the Borrower (which hereby irrevocably requests and authorizes the Swing Line Lender to so request on its behalf), that each Lender make a Floating Rate Loan in an amount equal to such Lender's Pro Rata Share of the amount of Swing Line Loans then outstanding. Such request shall be made in writing (which written request shall be deemed to be a Borrowing Notice for purposes hereof) and in accordance with the requirements of Section 2.8 without regard to the minimum and multiples specified in Section 2.4 and provided that, after giving effect to such Advance, the Aggregate Outstanding Credit Exposure shall not exceed the Aggregate Commitments. The Swing Line Lender shall furnish the Borrower with a copy of the applicable Loan notice promptly after delivering such notice to the Administrative Agent. Each Lender shall make an amount equal to its Pro Rata Share of the amount specified in such Loan notice available to the Administrative Agent in immediately available funds (and the Administrative Agent may apply Cash Collateral available with respect to the applicable Swing Line Loan) for the account of the Swing Line Lender at the Administrative Agent's Office not later than 1:00 p.m. on the day specified in such Loan notice, whereupon, subject to Section 2.20(c)(ii), each Lender that so makes funds available shall be deemed to have made a Floating Rate Loan to the Borrower in such amount. The Administrative Agent shall remit the funds so received to the Swing Line Lender.
- (ii) If for any reason any Swing Line Loan cannot be refinanced by such a Advance of Loans in accordance with <u>Section 2.20(c)(i)</u>, the request for Floating Rate Loans submitted by the Swing Line Lender as set forth herein shall be deemed to be a request by the Swing Line Lender that each of the Lender's payment to the Administrative Agent

38

for the account of the Swing Line Lender pursuant to Section 2.20(c)(i) shall be deemed payment in respect of such participation.

- (iii) If any Lender fails to make available to the Administrative Agent for the account of the Swing Line Lender any amount required to be paid by such Lender pursuant to the foregoing provisions of this Section 2.20(c) by the time specified in Section 2.20(c)(i), the Swing Line Lender shall be entitled to recover from such Lender (acting through the Administrative Agent), on demand, such amount with interest thereon for the period from the date such payment is required to the date on which such payment is immediately available to the Swing Line Lender at a rate per annum equal to the greater of the Federal Funds Rate and a rate determined by the Swing Line Lender in accordance with banking industry rules on interbank compensation. A certificate of the Swing Line Lender submitted to any Lender (through the Administrative Agent) with respect to any amounts owing under this clause (iii) shall be conclusive absent manifest error.
- (iv) Each Lender's obligation to make Loans or to purchase and fund risk participations in Swing Line Loans pursuant to this Section 2.20(c) shall be absolute and unconditional and shall not be affected by any circumstance, including (A) any setoff, counterclaim, recoupment, defense or other right that such Lender may have against the Swing Line Lender, the Borrower or any other Person for any reason whatsoever, (B) the occurrence or continuance of a Default or Unmatured Default, or (C) any other occurrence, event or condition, whether or not similar to any of the foregoing; provided, however, that each Lender's obligation to make Loans pursuant to this Section 2.20(c) is subject to the conditions set forth in Section 4.2. No such purchase or funding of risk participations shall relieve or otherwise impair the obligation of the Borrower to repay Swing Line Loans, together with interest as provided herein.

(d) Repayment of Participations.

(i)	At any time after any Lender has purchased and funded a risk participation in a Swing Line Loan, if the Swing Line Lend
receives any payn	ment on account of such Swing Line Loan, the Swing Line Lender will distribute to such Lender its Pro Rata Share of such
payment (appropri	riately adjusted, in the case of interest payments, to reflect the period of time during which such Lender's risk participation w
funded) in the sam	me funds as those received by the Swing Line Lender.

(ii) If any payment received by the Swing Line Lender in respect of principal or interest on any Swing Line Loan is required to be returned by the Swing Line Lender under any of the circumstances described in <u>Section 11.3</u> (including pursuant to any settlement entered into by the Swing Line Lender in its discretion), each Lender shall pay to the Swing Line Lender its Pro Rata Share thereof on demand of the Administrative Agent, plus interest thereon from the date of such demand to the date such amount is returned, at a rate per annum equal to the Federal Funds Rate. The Administrative Agent will make such demand upon the request of the Swing Line Lender. The obligations of

39

the Lenders under this clause shall survive the payment in full of the Obligations and the termination of this Agreement.

- (e) <u>Interest for Account of Swing Line Lender</u>. The Swing Line Lender shall be responsible for invoicing the Borrower for interest on the Swing Line Loans. Until each Lender funds its Floating Rate Loans or risk participation pursuant to this <u>Section 2.20</u> to refinance such Lender's Pro Rata Share of any Swing Line Loan, interest in respect of such Applicable Percentage shall be solely for the account of the Swing Line Lender.
- (f) <u>Payments Directly to Swing Line Lender</u>. The Borrower shall make all payments of principal and interest in respect of the Swing Line Loans directly to the Swing Line Lender.

2.21 <u>Extension of Facility Termination Date</u>.

- (a) Request for Extension. The Borrower may by written notice to the Administrative Agent (who shall promptly notify the Lenders) given not more than sixty (60) days and not less than forty-five (45) days prior to any anniversary of the Closing Date, request that each Lender extend the Facility Termination Date for an additional one (1) year from the then existing Facility Termination Date; provided, that the Borrower shall only be permitted to exercise this extension option two (2) times during the term of the Agreement.
- (b) <u>Lenders Election to Extend</u>. Each Lender, acting in its sole and individual discretion, shall, by notice to the Administrative Agent given not later than fifteen (15) days following the receipt of notice of such request from the Administrative Agent (the "<u>Notice Date</u>"), advise the Administrative Agent in writing whether or not such Lender agrees to such extension (and each Lender that determines not to so extend its Facility Termination Date (a "<u>Non-Extending Lender</u>") shall notify the Administrative Agent of such fact promptly after such determination (but in any event no later than the Notice Date) and any Lender that does not so advise the Administrative Agent on or before the Notice Date shall be deemed to be a Non-Extending Lender. The election of any Lender to agree to such extension shall not obligate any other Lender to so agree.
- (c) <u>Notification by Administrative Agent</u>. The Administrative Agent shall notify the Borrower of each Lender's determination under this Section promptly and in any event no later than the date fifteen (15) days after the Notice Date (or, if such date is not a Business Day, on the next preceding Business Day).
- (d) <u>Additional Commitment Lenders</u>. The Borrower shall have the right on or before the applicable anniversary of the Closing Date to replace each Non-Extending Lender with, and add as "Lenders" under this Agreement in place thereof, one or more Eligible Assignees (each, an "<u>Additional Commitment Lender</u>") as provided in <u>Section 12.2</u>, each of which Additional Commitment Lenders shall have entered into an Assignment Agreement pursuant to which such Additional Commitment Lender shall, undertake a Commitment (and, if any such Additional Commitment Lender is already a Lender, its Commitment shall be in addition to such Lender's Commitment hereunder on such date) and shall be a "Lender" for all purposes of this Agreement.

40

- (e) Minimum Extension Requirement. If all of the Lenders agree to any such request for extension of the Facility Termination Date then the Facility Termination for all Lenders shall be extended for the additional one (1) year, as applicable. If there exists any Non-Extending Lenders then the Borrower shall (i) withdraw its extension request and the Facility Termination Date will remain unchanged or (ii) provided that the Required Lenders (but for the avoidance of doubt, not including any Additional Commitment Lenders) have agreed to the extension request (such Lenders ag reeing to such extension, the "Approving Lenders") no later than fifteen (15) days prior to such anniversary of the Closing Date, then the Borrower may extend the Facility Termination Date solely as to the Approving Lenders and the Additional Commitment Lenders with a reduced amount of Aggregate Commitments during such extension period equal to the aggregate Commitments of the Approving Lenders and the Additional Commitment Lenders so long as the aggregate amount of the Commitments of the Approving Lenders and the Additional Commitment Lenders be Letter of Credit Sublimit (as it may be amended as of the extension date with the consent of such Approving Lenders and Additional Commitment Lenders); it being understood that (A) the Facility Termination Date relating to any Non-Extending Lenders not replaced by an Additional Commitment Lender shall not be extended and the repayment of all obligations owed to them and t he termination of their Commitments shall occur on the already existing Facility Termination Date and (B) the Facility Termination Date relating to the Approving Lenders and the Additional Commitment Lenders shall be extended for an additional year, as applicable.
- (f) <u>Conditions to Effectiveness of Extensions.</u> Notwithstanding the foregoing, any extension of the Facility Termination Date pursuant to this Section shall not be effective with respect to any Lender unless:
 - (i) on the date of such extension the conditions for a Credit Extension provided in <u>Section 4.2(a)</u>, <u>Section 4.2(b)</u> and <u>Section 4.2(c)</u> shall be satisfied; and

(ii) on any Facility Termination Date, the Borrower shall prepay any Loans outstanding on such date (and pay any additional amounts required pursuant to Section 3.4) to the extent necessary to keep outstanding Loans ratable with any revised Pro Rata Shares of the respective Lenders effective as of such date.

2.22 <u>Cash Collateral</u>.

(a) <u>Certain Credit Support Events</u>. Upon the request of the Administrative Agent or any of the Issuers (i) if any Issuer has honored any full or partial drawing request under any Letter of Credit and such drawing has not been reimbursed by the Borrower on the Letter of Credit Payment Date or (ii) if, as of the Facility Termination Date, any Letter of Credit Obligation for any reason remains outstanding and partially or wholly drawn, the Borrower shall, in each case, immediately Cash Collateralize the then outstanding amount of all Letter of Credit Obligations. At any time that there shall exist a Defaulting Lender, immediately upon the request of the Administrative Agent, the Issuers or the Swing Line Lender, the Borrower shall deliver to the Administrative Agent Cash Collateral in an amount sufficient to cover all Fronting Exposure, if any (after giving effect to <u>Section 2.23(a)(iv)</u> and any Cash Collateral provided by the Defaulting Lender).

41

- (b) Grant of Security Interest. All Cash Collateral (other than credit support not constituting funds subject to deposit) shall be maintained in blocked, interest bearing deposit accounts at Bank of America (the "LC Collateral Accounts"). The Borrower, and to the extent provided by any Lender, such Lender, hereby grants to (and subjects to the control of) the Administrative Agent, for the benefit of the Administrative Agent, the Issuers and the Lenders (including the Swing Line Lender), and agrees to maintain, a first priority security interest in all such cash, deposit accounts and all balances therein, and all other property so provided as collateral pursuant hereto, and in all proceeds of the foregoing, all as security for the obligations to which such Cash Collateral may be applied pursuant to Section 2.22(c). If at any time the Administrative Agent determines that Cash Collateral is less than the applicable Fronting Exposure and other obligations secured thereby, the Borrower or the relevant Defaulting Lender will, promptly upon written demand by the Administrative Agent, pay or provide to the Administrative Agent additional Cash Collateral in an amount sufficient to eliminate such deficiency.
- (c) <u>Application</u>. Notwithstanding anything to the contrary contained in this Agreement, Cash Collateral provided under any of this <u>Section 2.22</u> or <u>Sections 2.19</u>, <u>2.20</u>, <u>2.23</u> or <u>8.1</u> in respect of Letters of Credit or Swing Line Loans shall be held and applied to the satisfaction of the specific Letter of Credit Obligations, Swing Lin e Loans, obligations to fund participations therein (including, as to Cash Collateral provided by a Defaulting Lender, any interest accrued on such obligation) and other obligations for which the Cash Collateral was so provided, prior to any other application of such property as may be provided for herein.
- (d) <u>Release</u>. Cash Collateral (or the appropriate portion thereof) provided to reduce Fronting Exposure or other obligations, together with all interest accrued thereon, shall be released promptly following (i) the elimination of the applicable Fronting Exposure or other obligations giving rise thereto (including by the termination of Defaulting Lender status of the applicable Lender (or, as appropriate, its assignee following compliance with <u>Section 12.1(b)</u> (v))) or (ii) the Administrative Agent's good faith determination that there exists excess Cash Collateral; <u>provided</u>, <u>however</u>, (x) that Cash Collateral furnished by or on behalf of the Borrower shall not be released during the continuance of an Unmatured Default or a Default, and (y) the Person providing Cash Collateral and the Issuers or Swing Line Lender, as applicable, may agree, in their sole discretion, that Cash Collateral shall not be released but instead held to support future anticipated Fronting Exposure or other obligations.

2.23 <u>Defaulting Lenders</u>.

- (a) <u>Adjustments</u>. Notwithstanding anything to the contrary contained in this Agreement, if any Lender becomes a Defaulting Lender, then, until such time as that Lender is no longer a Defaulting Lender, to the extent permitted by applicable Law:
 - (i) <u>Waivers and Amendments</u>. That Defaulting Lender's right to approve or disapprove any amendment, waiver or consent with respect to this Agreement shall be restricted as set forth in <u>Section 8.2</u>.

42

Reallocation of Payments. Any payment of principal, interest, fees or other amounts received by the Administrative Agent for the account of that Defaulting Lender under the Loan Documents (whether voluntary or mandatory, at maturity, pursuant to Article VIII or otherwise, and including any amounts made available to the Administrative Agent by that Defaulting Lender pursuant to Section 11.1), shall be applied at such time or times as may be reasonably determined by the Administrative Agent as follows: first, to the payment of any amounts owing by that Defaulting Lender to the Administrative Agent hereunder; second, to the payment on a pro rata basis of any amounts owing by that Defaulting Lender to the Issuers or Swing Line Lender hereunder; third, if so reasonably determined by the Administrative Agent or requested by the Issuers or Swing Line Lender, to be held as Cash Collateral for future funding obligations of that Defaulting Lender of any participation in any Swing Line Loan or Letter of Credit; fourth, as the Borrower may request (so long as no Default exists), to the funding of any Loan in respect of which that Defaulting Lender has failed to fund its portion thereof as required by this Agreement, as determined by the Administrative Agent; fifth, if so reasonably determined by the Administrative Agent and the Borrower, to be held in a non-interest bearing deposit account and released in order to satisfy obligations of that Defaulting Lender to fund Loans under this Agreement; sixth, to the payment of any amounts owing to the Lenders, the Issuers or Swing Line Lender as a result of any judgment of a court of competent jurisdiction obtained by any Lender, the Issuers or Swing Line Lender against that Defaulting Lender as a result of that Defaulting Lender's breach of its obligations under this Agreement; seventh, so long as no Default exists, to the payment of any amounts owing to the Borrower as a result of any judgment of a court of competent jurisdiction obtained by the Borrower against that Defaulting Lender as a result of that Defaulting Lender's breach of its obligations under this Agreement; and eighth, to that Defaulting Lender or as otherwise directed by a court of competent jurisdiction; provided that if (x) such payment is a payment of the principal amount of any Loans or a reimbursement of any payment on a Letter of Credit, in respect of which that Defaulting Lender has not fully funded its appropriate share and (y) such Loans or reimbursement of payment on a Letter of Credit were made at a time when the conditions set forth in Section 4.2 were satisfied or

waived, such payment shall be applied solely to pay the Loans of, and such Letter of Credit reimbursements owed to, all non-Defaulting Lenders on a pro rata basis prior to being applied to the payment of any Loans of, or such Letter of Credit reimbursements owed to, that Defaulting Lender. Any p ayments, prepayments or other amounts paid or payable to a Defaulting Lender that are applied (or held) to pay amounts owed by a Defaulting Lender or to post Cash Collateral pursuant to this Section 2.23(a)(ii) shall be deemed paid to and redirected by that Defaulting Lender, and each Lender irrevocably consents hereto.

(iii) <u>Certain Fees</u>. That Defaulting Lender (x) shall not be entitled to receive any commitment fee pursuant to <u>Section 2.5</u> for any period during which that Lender is a Defaulting Lender (and the Borrower shall not be required to pay any such fee that otherwise would have been required to have been paid to that Defaulting Lender) and (y) shall be limited in its right to receive Letter of Credit Fees as provided in <u>Section 2.19(d)</u>.

43

- (iv) Reallocation of Pro Rata Shares to Reduce Fronting Exposure. During any period in which there is a Defaulting Lender, for purposes of computing the amount of the obligation of each non-Defaulting Lender to acquire, refinance or fund participations in Letters of Credit or Swing Line Loans pursuant to Sections 2.19 and 2.20, the "Pro Rata Share" of each non-Defaulting Lender shall be computed without giving effect to the Commitment of that Defaulting Lender; provided, that, (i) each such reallocation shall be given effect only if, at the date the applicable Lender becomes a Defaulting Lender, no Default exists; and (ii) the aggregate obligation of each non-Defaulting Lender to acquire, refinance or fund participations in Letters of Credit and Swing Line Loans shall not exceed the positive difference, if any, of (1) the Commitment of that non-Defaulting Lender minus (2) the Outstanding Credit Exposure of that Lender.
- (b) <u>Defaulting Lender Cure</u>. If the Borrower, the Administrative Agent, the Swing Line Lender and the Issuers agree in writing in their reasonable discretion that a Defaulting Lender should no longer be deemed to be a Defaulting Lender, the Administrative Agent will so notify the parties hereto, whereupon as of the effective date specified in such notice and subject to any conditions set forth therein (which may include arrangements with respect to any Cash Collateral), that Lender will, to the extent applicable, purchase that portion of outstanding Loans of the other Lenders or take such other actions as the Administrative Agent may determine to be necessary to cause the Loans and funded and unfunded participations in Letters of Credit and Swing Line Loans to be held on a pro rata basis by the Lenders in accordance with their Pro Rata Shares (without giving effect to Section 2.23(a)(iv)), whereupon that Lender will cease to be a Defaulting Lender; provided that no adjustments will be made retroactively with respect to fees accrued or payments made by or on behalf of the Borrower while that Lender was a Defaulting Lender; and provided, further, that except to the extent otherwise expressly agreed by the affected parties, no change hereunder from Defaulting Lender to Lender will constitute a waiver or release of any claim of any party hereunder arising from that Lender's having been a Defaulti ng Lender.

ARTICLE III

YIELD PROTECTION; TAXES

3.1 <u>Yield Protection</u>.

If, on or after the date of this Agreement, the adoption of any law or any governmental or quasi-governmental rule, regulation, policy, guideline or directive (whether or not having the force of law), or any change which occurs after the date of this Agreement in the interpretation or administration thereof by any governmental or quasi-governmental authority, central bank or comparable agency charged with the interpretation or administration thereof, or compliance by any Lender, any applicable Lending Installation or any Issuer with any request or directive (whether or not having the force of law) of any such authority, central bank or comparable agency imposed after the date of this Agreement:

(i) subjects any Lender to any tax of any kind whatsoever with respect to this Agreement, any Letter of Credit, any participation in a Letter of Credit or any Eurodollar

11

Loan made by it, or change the basis of taxation of payments to such Lender in respect thereof (except for Indemnified Taxes or Other Taxes covered by Section 3.5 and the imposition of, or any change in the rate of, any Excluded Tax), or

- (ii) imposes or increases or deems applicable any reserve, assessment, insurance charge, special deposit or similar requirement against assets of, deposits with or for the account of, or credit extended by, any Lender, any applicable Lending Installation or any Issuer (other than reserves and assessments taken into account in determining the interest rate applicable to Eurodollar Advances), or
- (iii) imposes any other condition the result of which is to increase the cost to any Lender, any applicable Lending Installation or any Issuer of making, funding or maintaining its Eurodollar Loans or of issuing or participating in Letters of Credit or reduces any amount receivable by any Lender, any applicable Lending Installation or any Issuer in connection with its Eurodollar Loans or Letters of Credit, or requires any Lender, any applicable Lending Installation or any Issuer to make any payment calculated by reference to the amount of Eurodollar Loans or Letters of Credit held or interest received by it, by an amount deemed material by such Lender or such Issuer, as the case may be,

and the result of any of the foregoing is to increase the cost to such Lender, the applicable Lending Installation or such Issuer of making or maintaining its Eurodollar Loans, Letters of Credit or Commitment or to reduce the return received by such Lender, the applicable Lending Installation or such Issuer in connection with such Eurodollar Loans, Letters of Credit or Commitment, then, within fifteen (15) Business Days of written demand by such Lender or such Issuer, the Borrower shall pay such Lender or such Issuer such additional amount or amounts as will compensate such Lender or such Issuer for such increased cost or reduction in amount received.

3.2 <u>Changes in Capital Adequacy Regulations.</u>

If a Lender or an Issuer determines the amount of capital required or expected to be maintained by such Lender, any Lending Installation of such Lender, such Issuer or any corporation controlling such Lender or such Issuer is increased as a result of a Change, then, within thirty (30) days of written demand by such Lender or such Issuer, the Borrower shall pay such Lender or such Issuer the amount necessary to compensate for any shortfall in the rate of return on the portion of such increased capital which such Lender or such Issuer determines is attributable to this Agreement, its Outstanding Credit Exposure or its Commitment to make Loans or to issue or participate in Letters of Credit hereunder (after taking into account such Lender's policies as to capital adequacy). "Change" means (i) any change after the date of this Agreement in (or in the interpretation of) the Risk-Based Capital Guidelines or (ii) any adoption of or change in (or any change in the interpretation of) any other law, governmental or quasi-governmental rule, regulation, policy, guideline, interpretation, or directive (whether or not having the force of law) after the date of this Agreement which affects the amount of capital required or expected to be maintained by any Lender, any Lending Installation, any Issuer or any corporation controlling any Lender or any Issuer. "Risk-Based Capital Guidelines" means (x)

45

the risk-based capital guidelines in effect in the United States on the date of this Agreement, including transition rules, and (y) the corresponding capital regulations promulgated by regulatory authorities outside the United States implementing the July 1988 report of the Basle Committee on Banking Regulation and Supervisory Practices Entitled "International Convergence of Capital Measurements and Capital Standards," including transition rules, and any amendments to such regulations adopted prior to the date of this Agreement.

3.3 **Availability of Types of Advances.**

If (i) any Lender determines that maintenance of its Eurodollar Loans at a suitable Lending Installation would violate any applicable law, rule, regulation, or directive, whether or not having the force of law, enacted or imposed after the date of this Agreement, (ii) the Required Lenders determine that (a) deposits of a type and maturity appropriate to match fund Eurodollar Advances are not available or (b) the interest rate applicable to a Type of Advance does not accurately reflect the cost of making or maintaining such Advance or (iii) the Administrative Agent determines that adequate and reasonable means do not exist for determining the Eurodollar Base Rate, then the Administrative Agent shall suspend the availability of the affected Type of Advance and, in the case of clause (i), require any affected Eurodollar Advances to be repaid or converted to Floating Rate Advances, subject to the payment of any funding indemnification amounts required by Section 3.4.

3.4 Funding Indemnification.

If any conversion, prepayment or payment of a Eurodollar Advance occurs on a date which is not the last day of the applicable Interest Period, whether because of acceleration, prepayment or otherwise, or a Eurodollar Advance is not made, paid, continued or converted on the date or in the amount specified by the Borrower for any reason other than default by the Lenders, the Borrower will, within fifteen (15) Business Days of written demand by any Lender, indemnify such Lender for any loss or cost incurred by it resulting therefrom, including any loss or cost in liquidating or employing deposits acquired to fund or maintain such Eurodollar Advance; provided that each such Lender shall have delivered to the Borrower a certificate as to the amount of such loss or cost, setting forth in reasonable detail the calculation thereof, which certificate shall be conclusive in the absence of manifest error.

3.5 <u>Taxes</u>.

- (a) Payments Free of Taxes; Obligation to Withhold; Payments on Account of Taxes. (i) Any and all payments by or on account of any Obligation of the Borrower hereunder or under any other Loan Document shall to the extent permitted by applicable Laws be made free and clear of and without reduction or withholding for any Taxes. If, however, applicable Laws require the Borrower or the Administrative Agent to withhold or deduct any Tax, such Tax shall be withheld or deducted in accordance with such Laws as determined by the Borrower or the Administrative Agent, as the case may be, upon the basis of the information and documentation to be delivered pursuant to subsection (e) below.
- (ii) If the Borrower or the Administrative Agent shall be required by the Code to withhold or deduct any Taxes, including both United States Federal backup withholding and

46

withholding taxes, from any payment, then (A) the Borrower or the Administrative Agent, as the case may be, shall withhold or make such deductions as are determined by the Borrower or the Administrative Agent, as the case may be, to be required based upon the information and documentation it has received pursuant to <u>subsection (e)</u> below, (B) the Borrower or the Administrative Agent, as the case may be, shall timely pay the full amount withheld or deducted to the relevant Governmental Authority in accordance with the Code, and (C) to the extent that the withholding or deduction is made on account of Indemnified Taxes or Other Taxes, the sum payable by the Borrower shall be increased as necessary so that after any required withholding or the making of all required deductions (including deductions applicable to additional sums payable under this Section) the Administrative Agent or Lender, as the case may be, receives an amount equal to the sum it would have received had no such withholding or deduction of Indemnified Taxes or Other Taxes been made.

- (b) <u>Payment of Other Taxes by the Borrower</u>. Without limiting the provisions of <u>subsection (a)</u> above, the Borrower shall timely pay any Other Taxes to the relevant Governmental Authority in accordance with applicable Laws.
- (c) <u>Tax Indemnifications</u>. (i) Without limiting the provisions of <u>subsection (a)</u> or (<u>b</u>) above, the Borrower shall, and does hereby, indemnify the Administrative Agent and each Lender, and shall make payment in respect thereof within ten (10) days after demand therefor, for the full amount of any Indemnified Taxes or Other Taxes (including Indemnified Taxes or Other Taxes imposed or asserted on or attributable to amounts payable under this Section but excluding Indemnified Taxes or Other Taxes paid by the Borrower to the Administrative Agent pursuant to <u>Section 3.5(a)(ii)(C)</u>) withheld or deducted by the Borrower or the Administrative Agent or such Lender, as the case may be, and any penalties, interest and reasonable

expenses arising therefrom or with respect thereto, whether or not such Indemnified Taxes or Other Taxes were correctly or legally imposed or asserted by the relevant Governmental Authority. The Borrower shall also, and does hereby, indemnify the Administrative Agent, and shall make payment in respect thereof within ten (10) days after demand therefor, for any amount which a Lender for any reason fails to pay indefeasibly to the Administrative Agent as required by clause (ii) of this subsection. A certificate as to the amount of any such payment or liability delivered to the Borrower by a Lender (with a copy to the Administrative Agent), or by the Administrative Agent on its own behalf or on behalf of a Lender, shall be conclusive absent manifest error.

(ii) Without limiting the provisions of <u>subsection (a)</u> or (<u>b)</u> above, each Lender shall, and does hereby, indemnify the Borrower and the Administrative Agent, and shall make payment in respect thereof within ten (10) days after demand therefor, against any and all Taxes and any and all related losses, claims, liabilities, penalties, interest and expenses (including the fees, charges and disbursements of any counsel for the Borrower or the Administrative Agent) incurred by or asserted against the Borrower or the Administrative Agent by any Governmental Authority as a result of the failure by such Lender to deliver, or as a result of the inaccuracy, inadequacy or deficiency of, any documentation required to be delivered by such Lender to the Borrower or the Administrative Agent pursuant to <u>subsection (e)</u>. Each Lender hereby authorizes the Administrative Agent to set off and apply any and all amounts at any time owing to such Lender under this Agreement or any other Loan Document against any amount due to the Administrative Agent under this

47

<u>clause (ii)</u>. The agreements in this <u>clause (ii)</u> shall survive the resignation and/or replacement of the Administrative Agent, any assignment of rights by, or the replacement of, a Lender, the termination of the Aggregate Commitments and the repayment, satisfaction or discharge of all other Obligations.

- (d) <u>Evidence of Payments</u>. Upon request by the Borrower or the Administrative Agent, as the case may be, after any payment of Taxes by the Borrower or the Administrative Agent to a Governmental Authority as provided in this <u>Section 3.5</u>, the Borrower shall deliver to the Administrative Agent or the Administrative Agent shall deliver to the Borrower, as the case may be, the original or a certified copy of a receipt issued by such Governmental Authority evidencing such payment or other evidence of such payment reasonably satisfactory to the Borrower or the Administrative Agent, as the case may be.
- (e) <u>Status of Lenders; Tax Documentation.</u> (i) Each Lender shall deliver to the Borrower and to the Administrative Agent, at the time or times prescribed by applicable Laws or when reasonably requested by the Borrower or the Administrative Agent, such properly completed and executed documentation prescribed by applicable Laws or by the taxing authorities of any jurisdiction and such other reasonably requested information as will permit the Borrower or the Administrative Agent, as the case may be, to determine (A) whether or not payments made hereunder or under any other Loan Document are subject to Taxes, (B) if applicable, the required rate of withholding or deduction, and (C) such Lender's entitlement to any available exemption from, or reduction of, applicable Taxes in respect of all payments to be made to such Lender by the Borrower pursuant to this Agreement or otherwise to establish such Lender's status for withholding tax purposes in the applicable jurisdiction.
 - (ii) Without limiting the generality of the foregoing, if the Borrower is resident for tax purposes in the United States,
 - (A) any Lender that is a "United States person" within the meaning of Section 7701(a)(30) of the Code shall deliver to the Borrower and the Administrative Agent executed originals of Internal Revenue Service Form W-9 or such other documentation or information prescribed by applicable Laws or reasonably requested by the Borrower or the Administrative Agent as will enable the Borrower or the Administrative Agent, as the case may be, to determine whether or not such Lender is subject to backup withholding or information reporting requirements; and
 - (B) each Foreign Lender that is entitled under the Code or any applicable treaty to an exemption from or reduction of withholding tax with respect to payments hereunder or under any other Loan Document shall deliver to the Borrower and the Administrative Agent (in such number of copies as shall be requested by the recipient) on or prior to the date on which such Foreign Lender becomes a Lender under this Agreement (and from time to time thereafter upon the request of the Borrower or the Administrative Agent, but only if such Foreign Lender is legally entitled to do so), whichever of the following is applicable:

48

- (I) duly completed and executed originals of Internal Revenue Service Form W-8BEN claiming eligibility for benefits of an income tax treaty to which the United States is a party,
 - (II) duly completed and executed originals of Internal Revenue Service Form W-8ECI,
- (III) duly completed and executed originals of Internal Revenue Service Form W-8IMY and all required supporting documentation,
- (IV) in the case of a Foreign Lender claiming the benefits of the exemption for portfolio interest under section 881(c) of the Code, (x) a certificate to the effect that such Foreign Lender is not (A) a "bank" within the meaning of section 881(c)(3)(A) of the Code, (B) a "10 percent shareholder" of the Borrower within the meaning of section 881(c)(3)(B) of the Code, or (C) a "controlled foreign corporation" described in section 881(c)(3)(C) of the Code and (y) duly completed and executed originals of Internal Revenue Service Form W-8BEN, or
- (V) duly completed and executed originals of any other form prescribed by applicable Laws as a basis for claiming exemption from or a reduction in United States Federal withholding tax together with such supplementary documentation as may be

prescribed by applicable Laws to permit the Borrower or the Administrative Agent to determine the withholding or deduction required to be made.

- (iii) Each Lender shall promptly (A) notify the Borrower and the Administrative Agent of any change in circumstances which would modify or render invalid any claimed exemption or reduction, and (B) take such steps as shall not be materially disadvantageous to it, in the reasonable judgment of such Lender, and as may be reasonably necessary (including the re-designation of its lending office) to avoid any requirement of applicable Laws of any jurisdiction that the Borrower or the Administrative Agent make any withholding or deduction for Taxes from amounts payable to such Lender.
- (iv) Each Foreign Lender shall provide such documentation or information prescribed by applicable Laws or reasonably requested by the Borrower or the Administrative Agent as will enable the Borrower or the Administrative Agent, as the case may be, to determine whether or not such Foreign Lender is subject to withholding of tax under FATCA.
- (f) Treatment of Certain Refunds. Unless required by applicable Laws, at no time shall the Administrative Agent have any obligation to file for or otherwise pursue on behalf of a Lender, or have any obligation to pay to any Lender, any refund of Taxes withheld or deducted from funds paid for the account of such Lender. If the Administrative Agent or any Lender determines, in its sole discretion, that it has received a refund of any Indemnified Taxes or Other Taxes as to which it has been indemnified by the Borrower or with respect to which the Borrower has paid additional amounts pursuant to this Section, it shall pay to the Borrower an

49

amount equal to such refund (but only to the extent of indemnity payments made, or additional amounts paid, by the Borrower under this Section with respect to the Indemnified Taxes or Other Taxes giving rise to such refund), net of all out-of-pocket expenses incurred by the Administrative Agent or such Lender and without interest (other than any interest paid by the relevant Governmental Authority with respect to such refund), provided that the Borrower, upon the request of the Administrative Agent or such Lender, agrees to repay the amount paid over to the Borrower (plus any penalties, interest or other charges imposed by the relevant Governmental Authority) to the Administrative Agent or such Lender in the event the Administrative Agent or such Lender is required to repay such refund to such Governmental Authority. This subsection shall not be construed to require the Borrower, the Administrative Agent or any Lender to make available its tax returns (or any other information relating to its Taxes that it deems confidential) to the Borrower or any other Person.

3.6 <u>Lender Statements; Survival of Indemnity.</u>

To the extent reasonably possible and upon the request of the Borrower, each Lender or Issuer shall designate an alternate Lending Installation to the extent it would reduce any liability of the Borrower to such Lender or Issuer under Sections 3.1, 3.2 and 3.5 or to avoid the unavailability of Eurodollar Advances under Section 3.3, so long as such designation is not, in the judgment of such Lender or Issuer, disadvantageous to such Lender or Issuer. Each Lender or each Issuer, as applicable, shall deliver a written statement of such Lender or such Issuer to the Borrower (with a copy to the Administrative Agent) as to any amount due under Section 3.1, 3.2, 3.4 or 3.5 and containing the other information contemplated by such Sections. Such written statement shall set forth in reasonable detail the calculations upon which such Lender or such Issuer determined such amount and shall be final, conclusive and binding on the Borrower in the absence of manifest error. Determination of amounts payable under such Sections in connection with a Eurodollar Loan shall be calculated as though each Lender funded its Eurodollar Loan through the purchase of a deposit of the type and maturity corresponding to the deposit used as a reference in determining the Eurodollar Rate applicable to such Loan , whether in fact that is the case or not. Unless otherwise provided herein, the amount specified in the written statement of any Lender or any Issuer shall be payable on demand after receipt by the Borrower of such written statement. The obligations of the Borrower under Sections 3.1, 3.2, 3.4 and 3.5 shall survive payment of the Obligations and termination of this Agreement.

ARTICLE IV

CONDITIONS PRECEDENT

4.1 <u>Initial Credit Extension</u>.

This Agreement shall become effective upon and the obligation of the Lenders and the Issuers to make the initial Credit Extension hereunder is subject to the Administrative Agent's receipt of (a) all fees required to be paid to the Administrative Agent, the Arrangers and the Lenders on the date hereof (including the reasonable fees and expenses of counsel to the Administrative Agent for which reasonably detailed invoices have been presented on or prior to

50

the date hereof), (b) evidence that, prior to or concurrently with the effectiveness of this Agreement, all obligations under the Existing Credit Facility have been paid in full and all commitments to lend thereunder have been terminated, and (c) all of the following, in form and substance satisfactory to each Agent and each Lender, and in sufficient copies for each Lender:

- (i) Copies of the articles or certificate of incorporation of the Borrower, together with all amendments, certified by the Secretary or an Assistant Secretary of the Borrower, and a certificate of good standing, certified by the appropriate governmental officer in its jurisdiction of incorporation, as well as any other information that any Lender may request that is required by Section 326 of the USA PATRIOT ACT or necessary for the Administrative Agent or any Lender to verify the identity of the Borrower as required by Section 326 of the USA PATRIOT ACT.
- (ii) Copies, certified by the Secretary or an Assistant Secretary of the Borrower, of its by-laws and of its Board of Directors' resolutions and of resolutions or actions of any other body authorizing the execution of the Loan Documents to which the Borrower is a party.

- (iii) An incumbency certificate, executed by the Secretary or an Assistant Secretary of the Borrower, which shall identify by name and title and bear the signatures of the Authorized Officers and any other officers of the Borrower authorized to sign the Loan Documents to which the Borrower is a party, upon which certificate the Administrative Agent and the Lenders shall be entitled to rely until informed of any change in writing by the Borrower.
- (iv) A certificate, signed by an Authorized Officer of the Borrower, stating that (A) on the date hereof, no Default or Unmatured Default has occurred and is continuing and (B) the representations and warranties contained in Article V are true and correct in all material respects as of the date hereof.
- (v) A written opinion of the Borrower's counsel, addressed to the Administrative Agent and the Lenders in a form reasonably satisfactory to the Administrative Agent and its counsel.
 - (vi) Executed counterparts of this Agreement executed by the Borrower, the Administrative Agent, each Issuer and each Lender.
- (vii) Any Revolving Notes requested by a Lender pursuant to <u>Section 2.13</u> payable to the order of each such requesting Lender and the Swing Line Note.
 - (viii) If the initial Credit Extension will be the issuance of a Letter of Credit, a properly completed Letter of Credit Application.
 - (ix) Evidence of the effectiveness of the Great Plains Credit Agreement and the KCPL GMO Credit Agreement.

51

- (x) Written money transfer instructions, in substantially the form of <u>Exhibit C</u>, addressed to the Administrative Agent and signed by an Authorized Officer who has executed and delivered an incumbency certificate in accordance with the terms hereof, together with such other related money transfer authorizations as the Administrative Agent may have reasonably requested.
 - (xi) Such other documents as any Lender or its counsel may have reasonably requested.

4.2 <u>Each Credit Extension</u>.

The Lenders shall not be required to make any Credit Extension (other than a Credit Extension that, after giving effect thereto and to the application of the proceeds thereof, does not increase the aggregate amount of outstanding Credit Extensions) or increase their Commitments pursuant to any KCPL-GPE Re-Transfer or GPE-KCPL Transfer, unless on the date of such Credit Extension, KCPL-GPE Re-Transfer or GPE-KCPL Transfer:

- (a) No Default or Unmatured Default exists or would result from such Credit Extension.
- (b) The representations and warranties contained in <u>Article V</u> are true and correct in all material respects as of the date of such Credit Extension except to the extent any such representation or warranty is stated to relate solely to an earlier date, in which case such representation or warranty shall have been true and correct in all material respects on and as of such earlier date; <u>provided</u> that this <u>clause (b)</u> shall not apply to the representations and warranties set forth in <u>Section 5.5</u> (as it relates to <u>clause (i)</u> or <u>(ii)</u> of the definition of "Material Adverse Effect"), <u>clause (a)</u> of the first sentence of <u>Section 5.7</u> and the second sentence of <u>Section 5.7</u> with respect to any borrowing hereunder which is not part of the Initial Credit Extension.
- (c) With respect to a Credit Extension, no authorization or approval of any Governmental Authority, other than any such authorizations and approvals as have already been obtained and remain in full force and effect, will be required for the making of such Credit Extension.

Each delivery of a Borrowing Notice and each request for the issuance of a Letter of Credit shall constitute a representation and warranty by the Borrower that the conditions contained in Sections 4.2(a), (b) and (c) have been satisfied.

52

ARTICLE V

REPRESENTATIONS AND WARRANTIES

The Borrower represents and warrants to the Lenders that:

5.1 <u>Existence and Standing.</u>

Each of the Borrower and its Significant Subsidiaries (a) is a corporation, partnership (in the case of Subsidiaries only) or limited liability company duly and properly incorporated or organized, as the case may be, validly existing and (to the extent such concept applies to such entity) in good standing under the laws of its jurisdiction of incorporation or organization and (b) has all corporate power and authority and all governmental licenses, authorizations, consents and approvals required to carry on its business as now conducted, except where the failure to do so could not be reasonably expected to have a Material Adverse Effect.

5.2 <u>Authorization and Validity</u>.

The Borrower has the requisite corporate power and authority and legal right to execute and deliver the Loan Documents and to perform its obligations thereunder. The execution and delivery by the Borrower of the Loan Documents and the performance of its obligations thereunder have been duly authorized by all necessary corporate action, and the Loan Documents constitute legal, valid and binding obligations of the Borrower enforceable against the Borrower in accordance with their terms, except as enforceability may be limited by bankruptcy, moratorium, reorganization or similar laws affecting the enforcement of creditors' rights generally and by general principles of equity (regardless of whether such enforceability is considered in a proceeding in equity or law).

5.3 <u>No Conflict; Government Consent.</u>

Neither the execution and delivery by the Borrower of the Loan Documents, nor the consummation of the transactions therein contemplated, nor compliance with the provisions thereof will (a) violate (i) in any material respect any law, statute, rule, regulation, order, writ, judgment, injunction, decree or award binding on the Borrower, (ii) the Borrower's certificate of incorporation or by-laws or (iii) the provisions of any indenture, instrument or agreement to which the Borrower is a party or is subject, or by which it, or its Property, is bound, except to the extent any such violation could not reasonably be expected to have a Material Adverse Effect (b) conflict with or constitute a default under any such indenture, instrument or other agreement in a manner which could reasonably be expected to have a Material Adverse Effect, or (c) result in, or require, the creation or imposition of any Lien in, of or on the Property of the Borrower pursuant to the terms of any such indenture, instrument or other agreement. No order, consent, adjudication, approval, license, authorization, or validation of, or filing, recording or registration with, or exemption by, or other action in respect of any governmental or public body or authority, or any subdivision thereof, which has not been obtained by the Borrower, is required to be obtained by the Borrower in connection with the execution and delivery of the Loan Documents, the borrowings under this Agreement, the payment and performance by the

53

Borrower of the Obligations or the legality, validity, binding effect or enforceability of any of the Loan Documents excepting, however, (a) '34 Act Reports due after the date hereof disclosing the Loan Documents and related information as required pursuant to applicable '34 Act Report filing requirements, and (b) Borrower's short-term borrowing authorization pursuant to the Federal Power Act, as in effect from time to time; <u>provided</u> that the Borrower has such short-term borrowing authorization pursuant to the Federal Power Act at the date of this Agreement and through December 15, 2010.

5.4 Financial Statements.

The December 31, 2009 and June 30, 2010 consolidated financial statements of the Borrower and its Subsidiaries heretofore delivered to the Lenders were prepared in accordance with GAAP and fairly present the consolidated financial condition and results of operations of the Borrower and its Subsidiaries at such dates and the consolidated results of their operations for the periods then ended subject, in the case of the June 30, 2010 financial statements, to normal year-end adjustments.

5.5 <u>Material Adverse Change</u>.

Since December 31, 2009 and except as disclosed in the Borrower's '34 Act Reports, there has been no change in the business, Property, financial condition or results of operations of the Borrower and its Subsidiaries, taken as a whole, which could reasonably be expected to have a Material Adverse Effect.

5.6 <u>Taxes</u>.

The Borrower and its Significant Subsidiaries have filed all United States federal income tax returns and all other material tax returns which are required to be filed by them and have paid all taxes due and payable pursuant to said returns or pursuant to any assessment received by the Borrower or any of its Significant Subsidiaries, except such taxes, if any, as are being contested in good faith and as to which adequate reserves have been provided in accordance with GAAP and as to which no Lien exists or except to the extent that the failure to pay any such tax, or to file any such tax return, could not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect. No tax liens have been filed and no material claims are being asserted against the Borrower or any Significant Subsidiary with respect to any such taxes except to the extent that such tax liens or claims could not reasonably be expected to have a Material Adverse Effect.

5.7 <u>Litigation; etc.</u>

Except as disclosed in the Borrower's '34 Act Reports, there is no litigation, arbitration, governmental investigation, proceeding or inquiry pending or, to the knowledge of any of their officers, threatened against or affecting the Borrower or any of its Subsidiaries in which there is a reasonable possibility of an adverse decision and which (a) could reasonably be expected to have a Material Adverse Effect or (b) seeks to prevent, enjoin or delay the making of any Credit Extension. Other than any liability incident to any litigation, arbitration or proceeding which

54

could not reasonably be expected to have a Material Adverse Effect, the Borrower has no material contingent obligations not provided for or disclosed in the financial statements referred to in Section 5.4.

5.8 <u>ERISA</u>.

The Borrower and each other member of the Controlled Group has fulfilled its obligations under the minimum funding standards of ERISA and the Code with respect to each Plan and is in compliance with the presently applicable provisions of ERISA and the Code with respect to each Plan, except to the extent that noncompliance, individually or in the aggregate, has not resulted in and could not reasonably be expected to result in a Material Adverse Effect. Neither the Borrower nor any other member of the Controlled Group has (i) failed to make any required contribution or payment to any Plan or Multiemployer Plan, or made any amendment to any Plan which has resulted or could result in the imposition of a Lien or the posting of a bond or other

security under ERISA or the Code or (ii) incurred any liability under Title IV of ERI SA other than a liability to the PBGC for premiums under Section 4007 of ERISA.

5.9 Accuracy of Information.

No information, exhibit or report furnished by the Borrower or any of its Subsidiaries to the Administrative Agent, any Issuer or any Lender in connection with the negotiation of, or compliance with, the Loan Documents contained, as of the date of delivery thereof and taken as a whole, any untrue statement of a material fact or, when considered together with all reports filed with the SEC and furnished or deemed furnished pursuant to the terms hereof, omitted to state any material fact necessary to make the statements contained therein, in light of the circumstances under which they were made, not misleading; provided that, with respect to projected financial information, forecasts and other forward-looking information, the Borrower represents only that such information was prepared in good faith based upon assumptions and estimates developed by management of the Borrower in good faith and believed to be reasonable at the time (it being understood that such information is not a guarantee of future performance and that actual results during the period or periods covered by such information may materially differ from the projected results therein).

5.10 Regulation U.

The Borrower is not engaged and will not engage, principally or as one of its important activities, in the business of purchasing or carrying margin stock (as defined in Regulation U), or extending credit for the purpose of purchasing or carrying margin stock. Margin stock constitutes less than twenty-five percent (25%) of the value of those assets of the Borrower and its Subsidiaries which are subject to any limitation on sale, pledge or other restriction hereunder.

5.11 <u>Material Agreements</u>.

Neither the Borrower nor any Subsidiary is a party to any agreement or instrument or subject to any charter or other corporate restriction which is reasonably likely to have a Material Adverse Effect. Neither the Borrower nor any Subsidiary is in default in the performance,

55

observance or fulfillment of any of the obligations, covenants or conditions contained in any agreement to which it is a party, which default could reasonably be expected to have a Material Adverse Effect.

5.12 <u>Compliance With Laws</u>.

The Borrower and its Subsidiaries have complied with all applicable statutes, rules, regulations, orders and restrictions of any domestic or foreign government or any instrumentality or agency thereof having jurisdiction over the conduct of their respective businesses or the ownership of their respective Property except for any failure to comply with any of the foregoing which could not reasonably be expected to have a Material Adverse Effect.

5.13 Ownership of Properties.

On the date of this Agreement, the Borrower and its Significant Subsidiaries have good title, free of all Liens other than those permitted by Section 6.12, to all of the Property and assets reflected as owned by the Borrower and its Significant Subsidiaries in the Borrower's most recent consolidated financial statements provided to the Administrative Agent.

5.14 <u>Plan Assets; Prohibited Transactions.</u>

To the Borrower's knowledge, (i) the Borrower is not an entity deemed to hold "plan assets" within the meaning of 29 C.F.R. § 2510.3-101 of another entity's employee benefit plan (as defined in Section 3(3) of ERISA) which is subject to Title I of ERISA or any plan (within the meaning of Section 4975 of the Code), and (ii) neither the execution of this Agreement nor the making of Loans hereunder gives rise to a prohibited transaction within the meaning of Section 406 of ERISA or Section 4975 of the Code.

5.15 Environmental Matters.

Except as set forth in the Borrower's '34 Act Reports, there are no known risks and liabilities accruing to the Borrower or any of its Subsidiaries due to Environmental Laws that could reasonably be expected to have a Material Adverse Effect.

5.16 <u>Investment Company Act.</u>

Neither the Borrower nor any Subsidiary is or is required to be registered as an "investment company" within the meaning of the Investment Company Act of 1940, as amended.

5.17 Pari Passu Indebtedness.

The Indebtedness under the Loan Documents ranks at least <u>pari passu</u> with all other unsecured Indebtedness of the Borrower.

As of the date hereof and after giving effect to the consummation of the transactions contemplated by the Loan Documents, the Borrower and each Significant Subsidiary is solvent. For purposes of the preceding sentence, solvent means (a) the fair saleable value (on a going concern basis) of the Borrower's assets or a Significant Subsidiary's assets, as applicable, exceed its liabilities, contingent or otherwise, fairly valued, (b) such Person will be able to pay its debts as they become due and (c) such Person will not be left with unreasonably small capital as is necessary to satisfy all of its current and reasonably anticipated obligations giving due consideration to the prevailing practice in the industry in which such Person is engaged. In computing the amount of contingent liabilities at any time, it is intended that such liabilities will be computed at the amount which, in light of all the facts and circumstances existing at such time, represents the amount that can reasonably be expected to become an actual or matured liability. The Borrower is not entering into the Loan Documents with the actual intent to hinder, delay or defraud its current or future creditors, nor does the Borrower intend to or believe that it will incur, as a result of entering into this Agreement and the other Loan Documents, debts beyond its ability to repay.

5.19 No Defaults.

No Unmatured Default or Default has occurred and is continuing.

ARTICLE VI

COVENANTS

During the term of this Agreement, unless the Required Lenders shall otherwise consent in writing:

6.1 <u>Financial Reporting and Notices</u>.

The Borrower will maintain, for itself and each Subsidiary, a system of accounting established and administered in accordance with generally accepted accounting principles, and (subject to the penultimate paragraph of this <u>Section 6.1</u>) furnish to the Administrative Agent for distribution to the Lenders:

- (a) Within ninety (90) days after the close of each of its fiscal years, an unqualified audit report certified by an independent registered public accounting firm which is a member of the "Big Four," prepared in accordance with GAAP on a consolidated basis for itself and its Consolidated Subsidiaries, including balance sheets as of the end of such period and related statements of income, common shareholders' equity and cash flows, accompanied by any management letter prepared by said accountants.
- (b) Within forty-five (45) days after the close of the first three (3) quarterly periods of each of its fiscal years, for itself and its Consolidated Subsidiaries, either (i)

57

consolidated and consolidating unaudited balance sheets as at the close of each such period and consolidated and consolidating profit and loss and reconciliation of surplus statements and a statement of cash flows for the period from the beginning of such fiscal year to the end of such quarter, all certified by its Chief Accounting Officer or Chief Financial Officer or (ii) if the Borrower is then a "registrant" within the meaning of Rule 1-01 of Regulation S-X of the SEC and required to file a report on Form 10-Q with the SEC, a copy of the Borrower's report on Form 10-Q for such quarterly period.

- (c) Together with the financial statements required under <u>Sections 6.1(a)</u> and (<u>b)</u>, a compliance certificate in substantially the form of <u>Exhibit A</u> signed by its Chief Accounting Officer or Chief Financial Officer setting forth calculations of the financial covenants contained in <u>Article VI</u> and stating that no Default or Unmatured Default exists, or if any Default or Unmatured Default exists, stating the nature and status thereof.
- (d) As soon as possible and in any event within thirty (30) days after the Borrower or any member of the Controlled Group knows that any Reportable Event has occurred with respect to any Plan that could reasonably be expected to result in liability in an aggregate amount in excess of \$50,000,000, a statement, signed by the Chief Accounting Officer or Chief Financial Officer of the Borrower, describing said Reportable Event and the action which the Borrower or member of the Controlled Group proposes to take with respect thereto.
- (e) Promptly after receipt of notice by the Borrower or any member of the Controlled Group of the PBGC's intention to terminate any Plan or to have a trustee appointed to administer any Plan, a copy of such notice.
- (f) Promptly upon the furnishing thereof to the shareholders of the Borrower, copies of all financial statements, reports and proxy statements so furnished.
- (g) Promptly upon the filing thereof, copies of all registration statements and annual, quarterly, monthly or other regular reports which the Borrower files with the SEC.
- (h) Promptly after the public announcement of, or promptly after receiving a written notice of, any change (whether an increase or decrease) by Moody's or S&P in the senior unsecured debt rating of the Borrower.
- (i) Such other information (including non-financial information) as the Administrative Agent or any Lender may from time to time reasonably request.

The statements and reports required to be furnished by the Borrower pursuant to <u>clauses (a)</u>, <u>(b)</u>, <u>(f)</u> and <u>(g)</u> above shall be deemed furnished for such purpose upon becoming publicly available on the SEC's EDGAR web page.

The Borrower hereby acknowledges that (a) the Administrative Agent and/or BAS will make available to the Lenders and Issuers materials and/or information provided by or on behalf of the

Borrower hereunder (collectively, "Borrower Materials") by posting the Borrower Materials on IntraLinks or another similar restricted access electronic system (the "Platform") and (b) certain of the Lenders may be "public-side" Lenders (i.e., Lenders that do not wish to receive material non-public information with respect to the Borrower or its securities) (each, a "Public Lender"). The Borrower hereby agrees that (w) all Borrower Materials that are to be made available to Public Lenders shall be clearly and conspicuously marked "PUBLIC" which, at a minimum, shall mean that the word "PUBLIC" shall appear prominently on the first page thereof; (x) by marking Borrower Materials "PUBLIC," the Borrower shall be deemed to have authorized the Agents, the Arrangers, the Issuers and the Lenders to treat such Borrower Materials as not containing any material non-public information with respect to the Borrower or its securities for purposes of United States Federal and state securities laws (provided, however, that to the extent such Borrower Materials constitute Specified Information, they shall be treated as set forth in Section 9.11(a)); (y) all Borrower Materials marked "PUBLIC" are permitted to be made available through a portion of the Platform designated "Public Investor;" and (z) the Administrative Agent and BAS shall be entitled to treat any Borrower Materials that are not marked "PUBLIC" as being suitable only for posting on a portion of the Platform not designated "Public Investor." Notwithstanding the foregoing, the Borrower shall be under no obligation to mark any Borrower Materials "PUBLIC."

6.2 Permits, Etc.

The Borrower will, and will cause each Significant Subsidiary to, take all reasonable action to maintain all rights, privileges, permits, licenses and franchises necessary or desirable in the normal conduct of its business, except to the extent failure to do so could not reasonably be expected to have a Material Adverse Effect; and preserve or renew all of its registered patents, trademarks, trade names and service marks, the non-preservation of which could reasonably be expected to have a Material Adverse Effect.

6.3 Use of Proceeds.

The Borrower will use the proceeds of the Credit Extensions (i) to repay the Existing Credit Facility and (ii) for general corporate purposes, including the refinancing of any commercial paper program of the Borrower and its Subsidiaries and use for working capital, capital expenditures and other lawful purposes of the Borrower and its Subsidiaries. The Borrower will not use any of the proceeds of the Credit Extensions to purchase or carry any margin stock (as defined in Regulation U) or to extend credit for the purpose of purchasing or carrying margin stock; provided that the Borrower may repurchase its own stock (or components thereof) so long as such stock is not reissued. The Borrower will not permit margin stock to constitute twenty-five perc ent (25%) or more of the value of those assets of the Borrower and its Subsidiaries which are subject to any limitation on sale, pledge or other restriction hereunder.

6.4 Notice of Default.

The Borrower will give prompt notice in writing to the Administrative Agent and the Lenders of the occurrence of any Default or Unmatured Default and of any other development, financial or otherwise, which could reasonably be expected to have a Material Adverse Effect.

59

6.5 <u>Conduct of Business</u>.

Except as otherwise permitted under this Agreement, the Borrower will, and will cause each Significant Subsidiary to, carry on and conduct its business in substantially the same manner and in substantially the same general fields of enterprise as it is presently conducted and do all things necessary to remain duly incorporated or organized, validly existing and (to the extent such concept applies to such entity) in good standing as a domestic corporation, partnership or limited liability company in its jurisdiction of incorporation or organization, as the case may be, and except to the extent the failure to do so could not reasonably be expected to have a Material Adverse Effect, keep in full force and effect its respective rights, privileges and franchises necessary or desirable in the normal conduct of business; provided, however, that nothing in this Section 6.5 shall prohibit the termination of the corporate existence of a Subsidiary of the Borrower or a component of its business if the Borrower determines in good faith that such termination is in the best interest of the Borrower and is not materially disadvantageous to the Lenders.

6.6 Taxes.

The Borrower will, and will cause each Significant Subsidiary to, timely file United States federal and applicable foreign, state and local tax returns required by law and pay when due all taxes, assessments and governmental charges and levies upon it or its income, profits or Property, except those which are being contested in good faith by appropriate proceedings and with respect to which adequate reserves have been set aside in accordance with GAAP.

6.7 <u>Insurance</u>.

The Borrower will, and will cause each Significant Subsidiary to, maintain (either in such Borrower's name or in such Significant Subsidiary's own name) with financially sound and reputable insurance companies that are not Affiliates of the Borrower or its Subsidiaries (other than any captive insurance company) insurance on all their respective Properties and business against loss or damage of the kinds customarily insured against by Persons engaged in the same or similar business, of such types and in such amounts as are customarily carried under similar circumstances by such other Persons, and the Borrower will furnish to the Lenders upon request by the Administrative Agent, full information presented in reasonable detail as to the insurance carried. Such insurance may be subject to co-insurance, deductibility or similar clauses which, in effect, result in self-insurance of certain losses; provided that such self-insurance is in accord with the customary industry practices for Persons in the same or similar businesses and adequate insurance reserves are maintained in connection with such self-insurance to the extent required by GAAP.

6.8 Compliance with Laws.

compliance therewith is contested in good faith by appropriate proceedings or except where the failure to comply could not reasonably be expected to have a Material Adverse Effect.

6.9 <u>Maintenance of Properties; Books of Record.</u>

The Borrower will, and will cause each Significant Subsidiary to, (i) do all things necessary to maintain, preserve, protect and keep its material Property in good repair, working order and condition, ordinary wear and tear excepted, and make all necessary and proper repairs, renewals and replacements so that its business carried on in connection therewith may be properly conducted at all times and (ii) keep proper books of record and account, in which full and correct entries shall be made of all material financial transactions and the assets and business of the Borrower and each Significant Subsidiary in accordance with GAAP; provided that nothing in this Section 6.9 shall prevent the Borrower or any Si gnificant Subsidiary from selling, abandoning, disposing of, or otherwise discontinuing the operation or maintenance of any of its Property (including the Capital Stock of any Subsidiary that is not a Significant Subsidiary) or equipment the retention of which in the good faith judgment of the Borrower or such Significant Subsidiary is inadvisable or unnecessary to the Borrower and its Subsidiaries, taken as a whole, or the failure to could not reasonably be expected to have a Material Adverse Effect.

6.10 <u>Inspection</u>.

The Borrower will, and if a Default or Unmatured Default exists, will cause each Significant Subsidiary to, permit the Administrative Agent and the Lenders, by their respective representatives and agents designated by the Required Lenders, upon not less than five (5) Business Days notice, to inspect any of the Property, books and financial records of such Person, to examine and make copies of the books of accounts and other financial records of such Person, and to discuss the affairs, finances and accounts of such Person with, and to be advised as to the same by, such Person's officers at such reasonable times and intervals as the Administrative Agent or any Lender may designate, subject, however, in all cases to the imposition of such conditions as the Borrower and such Significant Subsidiaries shall deem necessary based on reason able considerations of safety and security; provided, however, that neither the Borrower nor any of its Significant Subsidiaries shall be required to disclose to the Administrative Agent, and any Lender, or any representatives thereof any information which is the subject of attorney-client privilege or attorney work-product privilege properly asserted by the applicable Person to prevent the loss of such privilege in connection with such information or which is prevented from disclosure pursuant to a confidentiality agreement with third parties. After the occurrence and during the continuance of a Default, any such inspection shall be at the Borrower's expense; at all other times, the Borrower shall not be liable to pay the expenses of the Administrative Agent or any Lender in connection with such inspections.

6.11 <u>Consolidations, Mergers and Sale of Assets.</u>

The Borrower will not, nor will it permit any Significant Subsidiary to, sell, lease, transfer, or otherwise dispose of all or substantially all of its assets (whether by a single transaction or a number of related transactions and whether at one time or over a period of time) or consolidate with or merge into any Person or permit any Person to merge into it, except

61

- (a) (i) A Wholly-Owned Subsidiary may be merged into the Borrower, and (ii) the Borrower may be merged into Great Plains provided that Great Plains shall expressly assume, in a writing reasonably satisfactory to the Administrative Agent, the due and punctual payment of the Obligations and the due and punctual performance of and compliance with all of the terms of this Agreement and the other Loan Documents to be performed or complied with by the Borrower.
- (b) Any Significant Subsidiary may sell all or substantially all of its assets to, or consolidate or merge into, another Significant Subsidiary; provided that, immediately before and after such merger, consolidation or sale, no Default or Unmatured Default shall exist.
 - (c) the Borrower may sell or transfer accounts receivable, pursuant to one or more securitization transactions.
- (d) The Borrower may sell all or substantially all of its assets to, or consolidate with or merge into, any other Person, or permit another Person to merge into it; provided that (i) the surviving Person, if such surviving Person is not the Borrower, or the transferee Person in the case of a sale of all or substantially all of the Borrower's assets (A) shall be a Person organized and existing under the laws of the United States of America or a state thereof or the District of Columbia, (B) shall expressly assume, in a writing reasonably satisfactory to the Administrative Agent, the due and punctual payment of the Obligations and the due and punctual performance of and compliance with all of the terms of this Agreement and the other Loan Documents to be performed or complied with by the Borrower and (C) shall deliver all documents required to be delivered pursuant to Sections 4.1(c)(ii), <a href="mailto:(c)(iii), <a href="mailto:(c)(iiii), <a href="mailto:(c)(iiii), <a href="mailto:(c)(iiii), <a href="mailto:(c)(iiii), <a href="mailto:(c)(i

Notwithstanding the foregoing, the Borrower and its Consolidated Subsidiaries will not convey, transfer, lease or otherwise dispose of (whether in one transaction or a series of transactions, but excluding (i) sales of inventory in the ordinary course of business, (ii) transactions permitted by clauses (a) through (d) above, (iii) transfers by the Borrower of assets related to, or ownership interests in, Iatan 2 to co-owners of Iatan 2 pursuant to the co-ownership, co-operating or other similar agreements of the co-owners of Iatan 2 and (iv) sales of the capital stock or assets of KLT Gas Inc. and Subsidiaries thereof) in

6.12 Liens.

The Borrower will not, nor will it permit any Significant Subsidiary to, create, incur, or suffer to exist any Lien in, of or on the Property of the Borrower or any of its Significant Subsidiaries, except:

- (a) Liens for taxes, assessments or governmental charges or levies on its Property if the same shall not at the time be delinquent or thereafter can be paid without penalty, or are being contested in good faith and by appropriate proceedings and for which adequate reserves in accordance with GAAP shall have been set aside on its books.
- (b) Liens imposed by law, such as carriers', warehousemen's, mechanics' and landlords' liens and other similar liens arising in the ordinary course of business which secure payment of obligations not more than sixty (60) days past due or which are being contested in good faith by appropriate proceedings and for which adequate reserves shall have been set aside on its books.
- (c) Liens arising out of pledges or deposits in the ordinary course of business under worker's compensation laws, unemployment insurance, old age pensions, or other social security or retirement benefits, or similar legislation, other than any Lien imposed under ERISA.
- Liens incidental to the normal conduct of the Borrower or any Significant Subsidiary or the ownership or leasing of its Property or the conduct of the ordinary course of its business, including (i) zoning restrictions, easements, building restrictions, rights of way, reservations, restrictions on the use of real property and such other encumbrances or charges against real property as are of a nature generally existing with respect to properties of a similar character and which are not substantial in amount and do not in any material way affect the marketability of the same, (ii) rights of lessees and lessors under leases, (iii) rights of collecting banks having rights of setoff, revocation, refund or chargeback with respect to money or instruments of the Borrower or any Significant Subsidiary on deposit with or in the possession of such banks, (iv) Liens or deposits to secure the performance of statutory obligations, tenders, bids, contracts, leases, progress payments, performance or return-of-money bonds, surety and appeal bonds, performance or other similar bonds, letters of credit, or other obligations of a similar nature incurred in the ordinary course of business, (v) Liens required by any contract or statute in order to permit the Borrower or Significant Subsidiary to perform any contract or subcontract made by it with or pursuant to the requirements of a governmental entity, in each case which are not incurred in connection with the borrowing of money, the obtaining of advances of credit or the payment of the deferred purchase price of Property and which do not in the aggregate impair the use of Property in the operation of the business of the Borrower and its Significant Subsidiaries taken as a whole and (vi) rights reserved to or vested in any Governmental Author ity by the terms of any right, power, franchise, grant, license or permit, or by any Laws, to terminate such right, power, franchise, license or permit or to purchase, condemn, expropriate or

63

recapture or to designate a purchaser of any of the Property of the Borrower or any of its Significant Subsidiaries.

- (e) Liens arising under the General Mortgage Indenture and Deed of Trust Dated December 1, 1986 from the Borrower to UMB, N.A.
- (f) Liens on Property of the Borrower existing on the date hereof and any renewal or extension thereof; <u>provided</u> that the Property covered thereby is not increased and any renewal or extension of the obligations secured or benefited thereby is permitted by this Agreement.
 - (g) Judgment Liens which secure payment of legal obligations that would not constitute a Default under Section 7.9.
- (h) Liens on Property acquired by the Borrower or a Significant Subsidiary after the date hereof, existing on such Property at the time of acquisition thereof (and not created in anticipation thereof); <u>provided</u> that in any such case no such Lien shall extend to or cover any other Property of the Borrower or such Significant Subsidiary, as the case may be.
- (i) Liens on the Property, revenues and/or assets of any Person that exist at the time such Person becomes a Significant Subsidiary and the continuation of such Liens in connection with any refinancing or restructuring of the obligations secured by such Liens.
- (j) Liens on Property securing Indebtedness incurred or assumed at the time of, or within twelve (12) months after, the acquisition of such Property for the purpose of financing all or any part of the cost of acquiring such Property; <u>provided</u> that (i) such Lien attaches to such Property concurrently with or within twelve (12) months after the acquisition thereof, (ii) such Lien attaches solely to the Property so acquired in such transaction and (iii) the principal amount of the Indebtedness secured thereby does not exceed the cost or fair market value determined at the date of incurrence, whichever is lower, of the Property being acquired on the date of acquisition.
- (k) Liens on any improvements to Property securing Indebtedness incurred to provide funds for all or part of the cost of such improvements in a principal amount not exceeding the cost of construction of such improvements and incurred within twelve (12) months after completion of such improvements or construction, <u>provided</u> that such Liens do not extend to or cover any property of the Borrower or any Significant Subsidiary other than such improvements.
- (l) Liens to government entities granted to secure pollution control or industrial revenue bond financings, which Liens in each financing transaction cover only Property the acquisition or construction of which was financed by such financings and Property related thereto.

- (m) Liens on or over gas, oil, coal, fissionable material, or other fuel or fuel products as security for any obligations incurred by such Person (or any special purpose entity formed by such Person) for the sole purpose of financing the acquisition or storage of such fuel or fuel products or, with respect to nuclear fuel, the processing, reprocessing, sorting, storage and disposal thereof.
- (n) Liens on (including Liens arising out of the sale of) accounts receivable and/or contracts which will give rise to accounts receivable of the Borrower; and other Liens on (including Liens arising out of the sale of) accounts receivable and/or contracts which will give rise to accounts receivable of the Borrower or any Significant Subsidiary in an aggregate amount not at any time exceeding \$10,000,000.
 - (o) Liens on Property or assets of a Significant Subsidiary securing obligations owing to the Borrower or any Significant Subsidiary.
 - (p) Liens on Property of the Borrower arising in connection with utility co-ownership, co-operating and similar agreements that are consistent with the utilities business and ancillary operations.
- (q) Liens on assets held by entities which are deemed Subsidiaries solely due to the application of <u>clause (c)</u> of the definition of "Subsidiary" and which are required to be included in the Borrower's consolidated financial statements solely as a result of the application of ASC Topic 810, as it may be amended or supplemented.
 - (r) Liens securing Swap Contracts permitted to be incurred under this Agreement.
- (s) Liens securing any extension, renewal, replacement or refinancing of Indebtedness secured by any Lien referred to in the foregoing clauses (h), (i), (j), (k), (l), (m) and (p); provided that (A) such new Lien shall be limited to all or part of the same Property that secured the original Lien (plus improvements on such Property) and (B) the amount secured by such Lien at such time is not increased to any amount greater than the amount outstanding at the time of such renewal, replacement or refinancing.
- (t) Liens in favor of the Issuers or the Swing Line Lender on cash collateral securing the obligations of a Defaulting Lender to fund risk participations hereunder.
- (u) Liens which would otherwise not be permitted by <u>clauses (a)</u> through <u>(t)</u> securing additional Indebtedness of the Borrower or a Significant Subsidiary; <u>provided</u> that after giving effect thereto the aggregate unpaid principal amount of Indebtedness (including Capitalized Lease Obligations) of the Borrower and its Significant Subsidiaries (including prepayment premiums and penalties) secured by Liens permitted by this <u>clause (u)</u> shall not exceed the

greater of (a) \$50,000,000 and (b) ten percent (10%) of Consolidated Tangible Net Worth.

(v) Liens on the stock or assets of Subsidiaries created pursuant to "rate reduction" bonds, for the payment of which legislatively authorized charges are imposed on customers.

6.13 Affiliates.

Except to the extent required by applicable Law or regulation, the Borrower will not, and will not permit any Subsidiary to, enter into any transaction (including the purchase or sale of any Property or service) with, or make any payment or transfer to, any Affiliate except in the ordinary course of business and pursuant to the reasonable requirements of the Borrower's or such Subsidiary's business and upon fair and reasonable terms no less favorable to the Borrower or such Subsidiary than the Borrower or such Subsidiary would obtain in a comparable arms-length transaction.

6.14 ERISA.

The Borrower will not, nor will it permit any Significant Subsidiary to, (i) voluntarily terminate any Plan, so as to result in any material liability of the Borrower or any Significant Subsidiary to the PBGC or (ii) enter into any transaction that the Borrower or such Significant Subsidiary knows or should know is a Prohibited Transaction (as defined in Section 4975 of the Code and in Section 406 of ERISA) involving any Plan which results in any liability of the Borrower or any Significant Subsidiary that could reasonably be expected, individually or in the aggregate, to cause a Material Adverse Effect or (iii) cause any occurrence of any Reportable Event which results in any liability of the Borrower or any Significant Subsidiary to the PBGC that could reasonably be expected, individually or in the aggregate, to cause a Material Adverse Effect or (iv) allow or suffer to exist any other event or condition known to the Borrower which results in any material liability of the Borrower or any Significant Subsidiary to the PBGC.

6.15 <u>Total Indebtedness to Total Capitalization</u>.

The Borrower shall at all times cause the ratio of (i) Total Indebtedness to (ii) Total Capitalization to be less than or equal to 0.65 to 1.0.

6.16 Restrictions on Subsidiary Dividends.

The Borrower will not, nor will it permit any Significant Subsidiary to, be a party to any agreement prohibiting or restricting the ability of such Significant Subsidiary to declare or pay dividends to the Borrower; <u>provided</u>, that the foregoing provisions of this <u>Section 6.16</u> shall not prohibit the Borrower or any Significant Subsidiary from entering into any debt instrument containing a total debt to capitalization covenant.

6.17 <u>Organization Documents</u>.

The Borrower and its Subsidiaries shall not amend, modify or change their Organization Documents in a manner materially adverse to the Lenders.

ARTICLE VII

DEFAULTS

The occurrence of any one or more of the following events shall constitute a Default:

- 7.1 Any representation or warranty made or deemed made by or on behalf of the Borrower to the Lenders or the Administrative Agent under or in connection with this Agreement, any Loan, or any certificate or information delivered in connection with this Agreement or any other Loan Document shall prove to have been incorrect in any material respect when made or deemed made.
- 7.2 (a) Nonpayment of principal of any Loan when due, nonpayment of any Reimbursement Obligations within one (1) Business Day after the same becomes due, or (b) nonpayment of interest upon any Loan or of any fee or other obligation under any of the Loan Documents within five (5) Business Days after the same becomes due.
- **7.3** The breach by the Borrower of any of the terms or provisions of <u>Section 6.3</u>, <u>6.10</u> (with respect to the Borrower and its Significant Subsidiaries only), <u>6.11</u>, <u>6.15</u> or <u>6.16</u>.
- **7.4** The breach by the Borrower of any of the terms or provisions of <u>Section 6.12</u> which is not remedied within thirty (30) days after the earlier of (a) an Authorized Officer of the Borrower becoming aware of such breach and (b) receipt by the Borrower of written notice from the Administrative Agent or any Lender.
- 7.5 The breach by the Borrower (other than a breach which constitutes a Default under another Section of this <u>Article VII</u>) of any of the terms or provisions of this Agreement which is not remedied within thirty (30) days after the earlier of (a) an Authorized Officer of the Borrower becoming aware of such breach and (b) receipt by the Borrower of written notice from the Administrative Agent or any Lender; <u>provided</u> that if such breach is capable of cure but (i) cannot be c ured by payment of money and (ii) cannot be cured by diligent efforts within such thirty (30)-day period, but such diligent efforts shall be properly commenced within such thirty (30)-day period and the Borrower is diligently pursuing, and shall continue to pursue diligently, remedy of such failure, the cure period shall be extended for an additional ninety (90) days, but in no event beyond the Facility Termination Date.
- **7.6** Failure of the Borrower or any of its Significant Subsidiaries to pay when due any Indebtedness aggregating in excess of \$50,000,000 ("<u>Material Indebtedness</u>"); or the default by the Borrower or any of its Significant Subsidiaries in the performance of any term, provision or condition contained in any agreement under which any such Material Indebtedness was created

67

or is governed, or any other event shall occur or condition exist, the effect of which default or event is to cause, or to permit the holder or holders of such Material Indebtedness to cause, such Material Indebtedness to become due prior to its stated maturity; or any Material Indebtedness of the Borrower or any of its Significant Subsidiaries shall be declared to be due and payable or required to be prepaid or repurchased (other than by a regularly scheduled payment) prior to the stated maturity thereof; or the Borrower or any of its Significant Subsidiaries shall not pay, or admit in writing its inability to pay, its debts generally as they become due.

- 7.7 The Borrower or any of its Significant Subsidiaries shall (a) have an order for relief entered with respect to it under the Federal bankruptcy laws as now or hereafter in effect, (b) make an assignment for the benefit of creditors, (c) apply for, seek, consent to, or acquiesce in, the appointment of a receiver, custodian, trustee, examiner, liquidator or similar official for it or any Substantial Portion of its Property, (d) institute any proceeding seeking an order for relief under the Federal bankruptcy laws as now or hereafter in effect or seeking to adjudicate it a bankrupt or insolvent, or seeking di ssolution, winding up, liquidation, reorganization, arrangement, adjustment or composition of it or its debts under any law relating to bankruptcy, insolvency or reorganization or relief of debtors or fail to file an answer or other pleading denying the material allegations of any such proceeding filed against it, (e) take any corporate, partnership or limited liability company action to authorize or effect any of the foregoing actions set forth in this Section 7.7 or (f) fail to contest in good faith any appointment or proceeding described in Section 7.8.
- **7.8** Without the application, approval or consent of the Borrower or any of its Subsidiaries, a receiver, trustee, examiner, liquidator or similar official shall be appointed for the Borrower or any of its Subsidiaries or any Substantial Portion of its Property, or a proceeding described in Section 7.7(d) shall be instituted against the Borrower or any of its Subsidiaries and such appointment continues undischarged or such proceeding continues undismissed or unstayed for a period of sixty (60) consecutive days.
- 7.9 The Borrower or any of its Significant Subsidiaries shall fail within thirty (30) days to pay, bond or otherwise discharge (a) any judgment or order for the payment of money not covered by insurance in excess of \$50,000,000 (either singly or in the aggregate with other such judgments) or (b) any non-monetary final judgment that has, or could reasonably be expected to have, a Material Adverse Effect, in either case which is not stayed on appeal or otherwise being appropriately contested in good faith.
 - **7.10** A Change of Control shall occur.

- **7.11** A Reportable Event shall have occurred which results in liability of the Borrower or any Significant Subsidiary in excess of \$50,000,000 and, thirty (30) days after written notice thereof shall have been given to the Borrower by the Administrative Agent or any Lender, the amount of such liability shall still exceed \$50,000,000.
- **7.12** Any authorization or approval or other action by any Governmental Authority or regulatory body required for the execution, delivery or performance of this Agreement or any other Loan Document by the Borrower shall fail to have been obtained at the time required or be

terminated, revoked or rescinded or shall otherwise no longer be in full force and effect, and such occurrence shall (i) materially adversely affect the enforceability of the Loan Documents against the Borrower and (ii) to the extent that such occurrence can be cured, shall continue for five (5) days (unless being contested in good faith by the Borrower and the pendency of such contest shall suspend such adverse effect).

- **7.13** Great Plains shall fail to own, directly or indirectly, all of the outstanding stock of the Borrower which, in the absence of any contingency, has the right to vote in an election of directors of the Borrower.
- **7.14** Any Loan Document, at any time after its execution and delivery and for any reason other than as expressly permitted hereunder or thereunder or satisfaction in full of all the Obligations, ceases to be in full force and effect in any material respect; or the Borrower contests in any manner the validity or enforceability of any Loan Document; or the Borrower denies that it has any or further liability or obligation under any Loan Document, or purports to revoke, terminate or rescind any Loan Document.

ARTICLE VIII

ACCELERATION, WAIVERS, AMENDMENTS AND REMEDIES

8.1 Acceleration; Letter of Credit Account.

(a) If any Default described in Section 7.7 or 7.8 occurs with respect to the Borrower, the obligations of the Lenders to make Loans hereunder and the obligation and power of the Issuers to issue Letters of Credit and the Swing Line Lender to make Swing Line Loans shall automatically terminate and the Obligations shall immediately become due and payable without any election or action on the part of the Administrative Agent, any Lender or any Issuer and the Borrower will be and become thereby unconditionally obligated, without any further notice, act or demand, to pay to the Administrative Agent an amount in immediately available funds, which funds shall be held in the LC Collateral Accounts, equal to the excess of (i) the amount of Letter of Credit Obligations at such time over (ii) the amount on deposit in the LC Collateral Account at such time which is free and clear of all rights and claims of third parties and has not been applied against the Obligations (such difference, the "Collateral Shortfall Amount"). If any other Default occurs, the Administrative Agent may with the consent, or shall at the request, of the Required Lenders, (x) terminate or suspend the obligations of the Lenders to make Loans hereunder and the obligation and power of the Issuers to issue Letters of Credit, or declare the Obligations to be due and payable, or both, whereupon the Obligations shall become immediately due and payable, without presentment, demand, protest or notice of any kind, all of which the Borrower hereby expressly waives, and (y) upon notice to the Borrower and i n addition to the continuing right to demand payment of all amounts payable under this Agreement, make demand on the Borrower to pay, and the Borrower will, forthwith upon such demand and without any further notice or act, pay to the Administrative Agent in immediately available funds the Collateral Shortfall Amount, which funds shall be deposited in the LC Collateral Accounts.

69

If (a) within thirty (30) days after acceleration of the maturity of the Obligations or termination of the obligations of the Lenders to make Loans hereunder as a result of any Default (other than any Default as described in Section 7.7 or 7.8 with respect to the Borrower) and (b) before any judgment or decree for the payment of the Obligations due shall have been obtained or entered, the Required Lenders (in their sole discretion) shall so direct, the Administrative Agent shall, by notice to the Borrower, rescind and annul such acceleration and/or termination.

8.2 <u>Amendments</u>.

Subject to the provisions of this <u>Article VIII</u>, the Required Lenders (or the Administrative Agent with the consent in writing of the Required Lenders) and the Borrower may enter into agreements supplemental hereto for the purpose of adding or modifying any provisions to the Loan Documents or changing in any manner the rights of the Lenders or the Borrower hereunder or waiving any Default hereunder; <u>provided</u> that no such supplemental agreement shall:

- (a) Extend the final maturity of any Loan or the expiry date of any Letter of Credit to a date after the Facility Termination Date or forgive all or any portion of the principal amount thereof, or reduce the rate or extend the time of payment of interest (other than default interest) or fees thereon, without the consent of each Lender directly affected thereby;
- (b) Reduce the percentage specified in the definition of Required Lenders, without the consent of each Lender directly affected thereby;
- (c) Increase the amount of the Commitment of any Lender without the consent of such Lender (except as provided for in <u>Section 2.6</u>), or extend the Facility Termination Date (except as provided for in <u>Section 2.21</u>), reduce the amount or extend the payment date for, the mandatory

payments required under <u>Section 2.2</u> or permit the Borrower to assign its rights under this Agreement, without the consent of the Administrative Agent, the Issuers, the Swing Line Lender, and each Lender directly affected thereby;

- (d) Amend this <u>Section 8.2</u> without the consent of each Lender directly affected thereby;
- (e) Release any funds from the LC Collateral Accounts, except to the extent such release is expressly permitted hereunder without the consent of each Lender directly affected thereby; or
- (f) Alter the pro rata sharing of payments, the order of application of payments required thereby or the pro rata treatment of the Lenders without the consent of each Lender directly affected thereby.

Notwithstanding the foregoing, any amendment, waiver, or consent which by its terms requires the consent of all Lenders or each affected Lender may be effected with the consent of

70

the applicable Lenders other than the Defaulting Lenders except that (x) the Commitment of any Defaulting Lender may not be increased or extended without the consent of each Lender and (y) any waiver, amendment or modification requiring the consent of all Lenders or each affected Lender that (i) by its terms affects any Defaulting Lender more adversely than other affected Lenders, (ii) reduces the principal of or the rate of interest for Loans of such Defaulting Lender, or fees or other amounts payable hereunder or under any other Loan Document to such Defaulting Lender or (iii) extends the time of the payment of interest or fees owing to such Defaulting Lender shall require the consent of such Defaulting Lender. No amendment of any provision of this Agreement relating to the Administrative Agent shall be effective without the written consent of such Issuer and no amendment of any provision of this Agreement relating to the Swing Line Lender shall be effective without the written consent of the Swing Line Lender. The Administrative Agent may waive payment of the fee required under Section 12.1(b) without obtaining the consent of any other party to this Agreement.

8.3 Preservation of Rights.

No delay or omission of the Lenders, the Swing Line Lender, the Issuers or the Administrative Agent to exercise any right under the Loan Documents shall impair such right or be construed to be a waiver of any Default or an acquiescence therein, and the making of a Credit Extension notwithstanding the existence of a Default or the inability of the Borrower to satisfy the conditions precedent to such Credit Extension shall not constitute any waiver or acquiescence. Any single or partial exercise of any such right shall not preclude other or further exercise thereof or the exercise of any other right, and no waiver, amendment or other variation of the terms, conditions or provisions of the Loan Documents whatsoever shall be valid unless in writing signed by the Lenders required pursuant to Section 8.2, and then only to the extent in such writing specifically set forth. All remedies contained in the Loan Documents or by law afforded shall be cumulative and all shall be available to the Administrative Agent, the Swing Line Lender, the Lenders and the Issuers until the Obligations have been paid in full.

ARTICLE IX

GENERAL PROVISIONS

9.1 <u>Survival of Representations</u>.

All representations and warranties of the Borrower contained in this Agreement shall survive the making of the Credit Extensions herein contemplated.

9.2 <u>Governmental Regulation</u>.

Anything contained in this Agreement to the contrary notwithstanding, no Lender shall be obligated to extend credit to the Borrower in violation of any limitation or prohibition provided by any applicable statute or regulation.

71

9.3 <u>Headings</u>.

Section headings in the Loan Documents are for convenience of reference only, and shall not govern the interpretation of any of the provisions of the Loan Documents.

9.4 Entire Agreement.

THIS AGREEMENT AND THE OTHER LOAN DOCUMENTS REPRESENT THE FINAL AGREEMENT AMONG THE PARTIES AND MAY NOT BE CONTRADICTED BY EVIDENCE OF PRIOR, CONTEMPORANEOUS, OR SUBSEQUENT ORAL AGREEMENTS OF THE PARTIES. THERE ARE NO UNWRITTEN ORAL AGREEMENTS AMONG THE PARTIES.

9.5 <u>Several Obligations; Benefits of this Agreement.</u>

The respective obligations of the Lenders hereunder are several and not joint and no Lender shall be the partner or agent of any other (except to the extent to which the Administrative Agent is authorized to act as such). The failure of any Lender to perform any of its obligations hereunder shall not relieve any other Lender from any of its obligations hereunder. This Agreement shall not be construed so as to confer any right or benefit upon any Person other than the parties to this Agreement and their respective successors and assigns; <u>provided</u> that the parties hereto expressly agree that each Arranger shall enjoy the benefits of the provisions of <u>Sections 9.6</u>, <u>9.10</u> and <u>10.7</u> to the extent specifically set forth therein and shall have the right to enforce such provisions on its own behalf and in its own name to the same extent as if it were a party to this Agreement.

9.6 <u>Expenses; Indemnification</u>.

(a) The Borrower shall reimburse (i) the Agents and the Arrangers for any reasonable costs and expenses (including the reasonable fees and charges of outside counsel for the Agents) paid or incurred by the Agents or the Arrangers in connection with the preparation, negotiation, execution, delivery, syndication, distribution (including via the internet), review, amendment, modification, and administration of the Loan Documents and (ii) the Issuers for all reasonable out-of-pocket expenses incurred by the Issuers in connection with the issuance, amendment, renewal or extension of any Letter of Credit or any demand for payment thereunder. The Borrower also agrees to reimburse each Agent, each Arranger, the Swing Line Lender, each Lender and each Issuer for any reasonable costs, internal charges and expenses (including the reasonable fees and charges of attorneys for such Agent, such Arranger, the Swing Line Lender, such Lender or such Issuer) paid or incurred by either Agent, either Arranger, the Swing Line Lender, any Lender or any Issuer in connection with the collection and enforcement, attempted enforcement, and preservation of rights and remedies under, any of the Loan Documents (including all such costs and expenses incurred during any "workout" or restructuring in respect of the Obligations and during any legal proceeding).

72

- (b) The Borrower hereby further agrees to indemnify each Agent, each Arranger, the Swing Line Lender, each Lender, each Issuer, their respective affiliates and the directors, partners, officers, employees, agents and advisors of the foregoing against all losses, claims, damages, penalties, judgments, liabilities and expenses (including all reasonable expenses of litigation or preparation therefor whether or not either Agent, either Arranger, the Swing Line Lender, any Lender or any Issuer or any affiliate is a party thereto) which any of them may pay or incur arising out of or relating to this Agreement, the other Loan Documents, the transactions contemplated hereby or the direct or indirect application or proposed application of the proceeds of any Credit Extension hereund er except to the extent that they are determined in a final non-appealable judgment by a court of competent jurisdiction to have resulted from the gross negligence or willful misconduct of the party seeking indemnification. In the case of any investigation, litigation or proceeding to which the indemnity in this Section applies, such indemnity shall be effective whether or not such investigation, litigation or proceeding is brought by a third party, by the Borrower or by any affiliate of the Borrower. The obligations of the Borrower under this Section 9.6 shall survive the payment of the Obligations and termination of this Agreement.
- (c) To the extent that the Borrower for any reason fails to indefeasibly pay any amount required under <u>subsection (a)</u> or (<u>b</u>) of this Section to be paid by it to the Administrative Agent (or any sub-agent thereof), the Swing Line Lender, any Issuer or any Related Party of any of the foregoing (and without limiting its obligation to do so), each Lender severally agrees to pay to the Administrative Agent (or any such sub-agent), the Swing Line Lender, such Issuer or such Related Party, as the case may be, such Lender's Pro Rata Share (determined as of the time that the applicable unreimbursed expense or indemnity payment is sought but without giving effect to the reallocation of Pro Rata Shares contemplated by <u>Section 2.23(a)(iv)</u>) of such unpaid amount, <u>provided</u> that the unreimbursed expense or indemnified loss, claim, damage, liability or related expense, as the case may be, was incurred by or asserted against the Administrative Agent (or any such sub-agent), the Swing Line Lender or any Issuer in its capacity as such, or against any Related Party of any of the foregoing acting for the Administrative Agent (or any such sub-agent), the Swing Line Lender or any Issuer in connection with such capacity.
- (d) Notwithstanding any provision to the contrary, this <u>Section 9.6</u> shall not apply with respect to any matters, liabilities, or obligations relating to Taxes; all such matters are governed exclusively by <u>Section 3.5</u>.

9.7 <u>Numbers of Documents</u>.

All statements, notices, closing documents, and requests hereunder shall be furnished to the Administrative Agent with sufficient counterparts so that the Administrative Agent may furnish one to each of the Lenders.

9.8 Accounting.

Except as provided to the contrary herein, all accounting terms used herein shall be interpreted and all accounting determinations hereunder shall be made in accordance with GAAP.

73

9.9 <u>Severability of Provisions</u>.

Any provision in any Loan Document that is held to be inoperative, unenforceable, or invalid in any jurisdiction shall, as to that jurisdiction, be inoperative, unenforceable, or invalid without affecting the remaining provisions in that jurisdiction or the operation, enforceability, or validity of that provision in any other jurisdiction, and to this end the provisions of all Loan Documents are declared to be severable. Without limiting the foregoing provisions of this Section 9.9, if and to the extent that the enforceability of any provisions in this Agreement relating to Defaulting Lenders shall be limited by Debtor Relief Laws, as determined in good faith by the Administrative Agent, the Issuers or the Swing Line Lender, as applicable, then such provisions shall be deemed to be in effect only to the extent not so limited.

9.10 Nonliability of Lenders.

The relationship between the Borrower on the one hand and the Lenders, the Swing Line Lender, the Issuers and the Agents on the other hand shall be solely that of borrower and lender. None of either Agent, either Arranger, the Swing Line Lender, any Lender or any Issuer shall have any fiduciary responsibilities to the Borrower. None of either Agent, either Arranger, the Swing Line Lender, any Lender or any Issuer undertakes any responsibility to the Borrower to review or inform the Borrower of any matter in connection with any phase of the Borrower's business or operations. The Borrower agrees that none of either Agent, either Arranger, the Swing Line Lender, any Lender or any Issuer shall have liability to the Borrower (whether sounding in tort, contract or otherwise) for losses suffered by the Bor rower in connection with, arising out of, or in any way related to, the transactions contemplated and the relationship established by the Loan Documents, or any act, omission or event occurring in connection therewith, unless it is determined in a final non-appealable judgment by a court of competent jurisdiction that such losses resulted from the gross negligence or willful misconduct of the party from which recovery is sought. None of either Agent, either Arranger, the Swing Line Lender, any Lender, any Issuer or any Related Party of any of the foregoing Persons shall have any liability with respect to, and the Borrower hereby waives, releases and agrees not to sue for, any special, indirect, consequential or punitive damages suffered by the Borrower in connection with, arising out of, or in any way related to the Loan Documents or the transactions contemplated thereby. None of either Agent, either Arranger, the Swing Line Lender, any Lender, any Issuer or any Related Party of any of the foregoing Persons shall be liable for any damages arising from the use by unintended recipients of any information or other materials distributed by it through telecommunications, electronic or other information transmission syste

9.11 <u>Limited Disclosure</u>.

(a) Neither the Administrative Agent nor any Lender may disclose to any Person any Specified Information (as defined below) except (i) to its, and its Affiliates', officers, employees, agents, accountants, legal counsel, advisors and other representatives who have a need to know

74

such Specified Information or any direct, indirect, actual or prospective counterparty (and its advisor) to any swap, derivative or securitization transaction related to the Obligations or to any credit insurance provider relating to the Obligations (it being understood that the Persons to whom such disclosure is made will be informed of the confidential nature of such Specified Information and instructed to keep such Specified Information confidential, and shall agree to do so) or (ii) with the Borrower's prior consent. "Specified Information" means information that the Borrower furnishes to the Administrative Agent or any Lender that is designated in writing as confidential, but does not include any such information that is or becomes generally available to the public or that is or becomes available to the Administrative Agent or such Lender from a source other than the Borrower.

(b) The provisions of <u>clause (a)</u> above shall not apply to Specified Information (i) that is a matter of general public knowledge or has heretofore been or is hereafter published in any source generally available to the public, (ii) that is required to be disclosed by law, regulation or subpoena, judicial order or other legal process, (iii) that is requested by any regulatory body (including any self-regulatory authority) with jurisdiction over the Administrative Agent or any Lender, or (iv) that is disclosed (A) to legal counsel, accountants and other professional advisors to such Lender, (B) in connection with the exercise of any right or remedy hereunder or under any Note or any suit or other litigation or proceeding relating to this Agreement or any Note, (C) to a rating agency if required by such agency in connection with a rating relating to Credit Extensions hereunder or (D) to assignees or participants or potential assignees or participants who agree to be bound by the provisions of this <u>Section 9.11</u>.

9.12 <u>USA PATRIOT ACT NOTIFICATION</u>.

The following notification is provided to the Borrower pursuant to Section 326 of the USA Patriot Act of 2001, 31 U.S.C. Section 5318: IMPORTANT INFORMATION ABOUT PROCEDURES FOR OPENING A NEW ACCOUNT. To help the government fight the funding of terrorism and money laundering activities, Federal law requires all financial institutions to obtain, verify and record information that identifies each person or entity that opens an account, including any deposit account, treasury management account, loan, other extension of credit or other financial services product. What this means for the Borrower: When the Borrower opens an account, the Lenders will ask for the Borrower's name, tax identification number, business address and other information that will allow the Administrative Agent and the Lenders to identify the Borrower. The Administrative Agent and the Lenders may also ask to see the Borrower's legal organizational documents or other identifying documents.

9.13 Nonreliance.

Each Lender hereby represents that it is not relying on or looking to any margin stock (as defined in Regulation U of the FRB) for the repayment of the Loans provided for herein.

9.14 <u>No Advisory or Fiduciary Responsibility.</u>

In connection with all aspects of each transaction contemplated hereby, the Borrower acknowledges and agrees, and acknowledges its Affiliates' understanding, that: (a) the credit

75

facility provided for hereunder and any related arranging or other services in connection therewith (including in connection with any amendment, waiver or other modification hereof or of any other Loan Document) are an arm's-length commercial transaction between the Borrower and its Affiliates, on the one hand, and the Agents and the Arrangers, on the other hand, and the Borrower is capable of evaluating and understanding and understands and accepts the terms, risks and conditions of the transactions contemplated hereby and by the other Loan Documents (including any amendment, waiver or other modification hereof or thereof); (b) in connection with the process leading to such transaction, each Agent and Arranger is and has been acting solely as a principal and is not the financial advisor, agent or fiduciary, for the Borrower or any of its Affiliates, stockholders, creditors or employees or any other Person;

(c) no Agent or Arranger has assumed or will assume an advisory, agency or fiduciary responsibility in favor of the Borrower with respect to any of the transactions contemplated hereby or the process leading thereto, including with respect to any amendment, waiver or other modification hereof or of any other Loan Document (irrespective of whether any Agent or Arranger has advised or is currently advising the Borrower or any of its Affiliates on other matters) and no Agent or Arranger has any obligation to the Borrower or any of its Affiliates with respect to the transactions contemplated hereby except those obligations expressly set forth herein and in the other Loan Documents; (d) the Agents and the Arrangers and their respective Affiliates may be engaged in a broad range of transactions that involve interests that differ from those of the Borrower and its Affiliates, and no Agent or Arranger has any obligation to disclose any of s uch interests by virtue of any advisory, agency or fiduciary relationship; and (e) the Agents and the Arrangers have not provided and will not provide any legal, accounting, regulatory or tax advice with respect to any of the transactions contemplated hereby (including any amendment, waiver or other modification hereof or of any other Loan Document) and the Borrower has consulted its own legal, accounting, regulatory and tax advisors to the extent it has deemed appropriate. The Borrower hereby waives and releases, to the fullest extent permitted by law, any claims that it may have against any Agent and any Arranger with respect to any breach or alleged breach of agency or fiduciary duty.

ARTICLE X

THE ADMINISTRATIVE AGENT

10.1 Appointment and Authority.

Each of the Lenders and the Issuers hereby irrevocably appoints Bank of America to act on its behalf as the Administrative Agent hereunder and under the other Loan Documents and authorizes the Administrative Agent to take such actions on its behalf and to exercise such powers as are delegated to the Administrative Agent by the terms hereof or thereof, together with such actions and powers as are reasonably incidental thereto. The provisions of this Article are solely for the benefit of the Administrative Agent, the Lenders and the Issuers, and the Borrower shall have no rights as a third party beneficiary of any of such provisions.

76

10.2 Rights as a Lender.

The Person serving as the Administrative Agent hereunder shall have the same rights and powers in its capacity as a Lender as any other Lender and may exercise the same as though it were not the Administrative Agent and the term "Lender" or "Lenders" shall, unless otherwise expressly indicated or unless the context otherwise requires, include the Person serving as the Administrative Agent hereunder in its individual capacity. Such Person and its Affiliates may accept deposits from, lend money to, act as the financial advisor or in any other advisory capacity for and generally engage in any kind of business with the Borrower or any Subsidiary or other Affiliate thereof as if such Person were not the Administrative Agent hereunder and without any duty to account therefor to the Lenders.

10.3 <u>Exculpatory Provisions</u>.

The Administrative Agent shall not have any duties or obligations except those expressly set forth herein and in the other Loan Documents. Without limiting the generality of the foregoing, the Administrative Agent:

- (a) shall not be subject to any fiduciary or other implied duties, regardless of whether a Default or Unmatured Default has occurred and is continuing;
- (b) shall not have any duty to take any discretionary action or exercise any discretionary powers, except discretionary rights and powers expressly contemplated hereby or by the other Loan Documents that the Administrative Agent is required to exercise as directed in writing by the Required Lenders (or such other number or percentage of the Lenders as shall be expressly provided for herein or in the other Loan Documents), <u>provided</u> that the Administrative Agent shall not be required to take any action that, in its opinion or the opinion of its counsel, may expose the Administrative Agent to liability or that is contrary to any Loan Document or applicable law; and
- (c) shall not, except as expressly set forth herein and in the other Loan Documents, have any duty to disclose, and shall not be liable for the failure to disclose, any information relating to the Borrower or any of its Affiliates that is communicated to or obtained by the Person serving as the Administrative Agent or any of its Affiliates in any capacity.

The Administrative Agent shall not be liable for any action taken or not taken by it (i) with the consent or at the request of the Required Lenders (or such other number or percentage of the Lenders as shall be necessary, or as the Administrative Agent shall believe in good faith shall be necessary, under the circumstances as provided in Sections 8.1 and 8.2) or (ii) in the absence of its own gross negligence or willful misconduct. The Administrative Agent shall be deemed not to have knowledge of any Default or Unmatured Default unless and until notice describing such Default or Unmatured Default is given to the Administrative Agent by the Borrower, a Lender or an Issuer.

The Administrative Agent shall not be responsible for or have any duty to ascertain or inquire into (i) any statement, warranty or representation made in or in connection with this

77

Agreement or any other Loan Document, (ii) the contents of any certificate, report or other document delivered hereunder or thereunder or in connection herewith or therewith, (iii) the performance or observance of any of the covenants, agreements or other terms or conditions set forth herein or therein or the

occurrence of any Default or Unmatured Default, (iv) the validity, enforceability, effectiveness or genuineness of this Agreement, any other Loan Document or any other agreement, instrument or document or (v) the satisfaction of any condition set forth in <u>Article IV</u> or elsewhere herein, other than to confirm receipt of items expressly required to be delivered to the Administrative Agent.

10.4 Reliance by Administrative Agent.

The Administrative Agent shall be entitled to rely upon, and shall not incur any liability for relying upon, any notice, request, certificate, consent, statement, instrument, document or other writing (including any electronic message, Internet or intranet website posting or other distribution) believed by it to be genuine and to have been signed, sent or otherwise authenticated by the proper Person. The Administrative Agent also may rely upon any statement made to it orally or by telephone and believed by it to have been made by the proper Person, and shall not incur any liability for relying thereon. In determining compliance with any condition hereunder to the making of a Loan, or the issuance of a Letter of Credit, that by its terms must be fulfilled to the satisfaction of a Lender or an Issuer, the Administrative Agent may presume that such condition is satisfactory to such Lender or such Issuer unless the Administrative Agent shall have received notice to the contrary from such Lender or such Issuer prior to the making of such Loan or the issuance of such Letter of Credit. The Administrative Agent may consult with legal counsel (who may be counsel for the Borrower), independent accountants and other experts selected by it, and shall not be liable for any action taken or not taken by it in accordance with the advice of any such counsel, accountants or experts.

10.5 <u>Delegation of Duties</u>.

The Administrative Agent may perform any and all of its duties and exercise its rights and powers hereunder or under any other Loan Document by or through any one or more sub-agents appointed by the Administrative Agent. The Administrative Agent and any such sub-agent may perform any and all of its duties and exercise its rights and powers by or through their respective Related Parties. The exculpatory provisions of this Article shall apply to any such sub-agent and to the Related Parties of the Administrative Agent and any such sub-agent, and shall apply to their respective activities in connection with the syndication of the credit facilities provided for herein as well as activities as Administrative Agent.

10.6 Resignation of Administrative Agent.

The Administrative Agent may, upon giving five (5) Business Days prior written notice to the Borrower, resign at any time by giving notice of its resignation to the Lenders, the Issuers and the Borrower. Upon receipt of any such notice of resignation, the Borrower shall have the right, with the consent of the Required Lenders, such consent not to be unreasonably withheld, conditioned or delayed, to appoint a successor, which successor shall be a Lender. If no such successor shall have been so appointed by the Borrower and shall have accepted such appointment within thirty (30) days after the retiring Administrative Agent gives notice of its

78

resignation, then the retiring Administrative Agent may on behalf of the Lenders and the Issuers, appoint a successor Administrative Agent from among the Lenders meeting the qualifications set forth above; provided that if the Administrative Agent shall notify the Borrower and the Lenders that no qualifying Person has accepted such appointment, then such resignation shall nonetheless become effective in accordance with such notice and (1) the retiring Administrative Agent shall be discharged from its duties and obligations hereunder and under the other Loan Documents and (2) all payments, communications and determinations provided to be made by, to or through the Administrative Agent shall instead be made by or to each Lender and each Issuer directly, until such ti me as the Borrower appoints a successor Administrative Agent as provided for above in this Section. Upon the acceptance of a successor's appointment as Administrative Agent hereunder, such successor shall succeed to and become vested with all of the rights, powers, privileges and duties of the retiring (or retired) Administrative Agent, and the retiring Administrative Agent shall be discharged from all of its duties and obligations hereunder or under the other Loan Documents (if not already discharged therefrom as provided above in this Section). The fees payable by the Borrower to a successor Administrative Agent shall be the same as those payable to its predecessor unless otherwise agreed between the Borrower and such successor. After the retiring Administrative Agent's resignation hereunder and under the other Loan Documents, the provisions of this Article and Section 9.6 shall continue in effect for the benefit of such retiring Administrative Agent, its sub-agents and their respective Related Parties in respect of any actions taken or omitted to be taken by any of them while the retiring Administrative Agent was acting as Administrative Agent.

Any resignation by Bank of America as Administrative Agent pursuant to this Section shall also constitute its resignation as an Issuer and as the Swing Line Lender. Upon the acceptance of a successor's appointment as Administrative Agent hereunder, (a) such successor shall succeed to and become vested with all of the rights, powers, privileges and duties of the retiring Issuer and the retiring Swing Line Lender, (b) the retiring Issuer and the retiring Swing Line Lender shall be discharged from all of their respective duties and obligations hereunder or under the other Loan Documents, and (c) the successor Issuer shall issue letters of credit in substitution for the Letters of Credit, if any, outstanding at the time of such succession or make other arrangements satisfactory to the retiring Issuer to effectively assu me the obligations of the retiring Issuer with respect to such Letters of Credit.

10.7 Non-Reliance on Administrative Agent and Other Lenders.

Each Lender, the Swing Line Lender and each Issuer acknowledges that it has, independently and without reliance upon the Administrative Agent or any other Lender or any of their Related Parties and based on such documents and information as it has deemed appropriate, made its own credit analysis and decision to enter into this Agreement. Each Lender, the Swing Line Lender and each Issuer also acknowledges that it will, independently and without reliance upon the Administrative Agent or any other Lender or any of their Related Parties and based on such documents and information as it shall from time to time deem appropriate, continue to make its own decisions in taking or not taking action under or based upon this Agreement, any other Loan Document or any related agreement or any document furnished hereunder or thereunder.

Anything herein to the contrary notwithstanding, none of the Bookrunners or Arrangers listed on the cover page hereof shall have any powers, duties or responsibilities under this Agreement or any of the other Loan Documents, except in its capacity, as applicable, as the Administrative Agent, the Swing Line Lender, a Lender or an Issuer hereunder.

10.9 Administrative Agent May File Proofs of Claim.

In case of the pendency of any proceeding under any Debtor Relief Law or other judicial proceeding relative to the Borrower, the Administrative Agent (irrespective of whether the principal of any Loan or Letter of Credit Obligation shall then be due and payable as herein expressed or by declaration or otherwise and irrespective of whether the Administrative Agent shall have made any demand on any Borrower) shall be entitled and empowered, by intervention in such proceeding or otherwise

- (a) to file and prove a claim for the whole amount of the principal and interest owing and unpaid in respect of the Loans, Letter of Credit Obligations and all other Obligations that are owing and unpaid and to file such other documents as may be necessary or advisable in order to have the claims of the Lenders, the Swing Line Lender, the Issuers and the Administrative Agent (including any claim for the reasonable compensation, expenses, disbursements and advances of the Lenders, the Swing Line Lender, the Issuers and the Administrative Agent and their respective agents and counsel and all other amounts due the Lenders, the Issuers and the Administrative Agent under Sections 2.5, 2.19(d), and 9.6) allowed in such judicial proceeding; and
 - (b) to collect and receive any monies or other property payable or deliverable on any such claims and to distribute the same;

and any custodian, receiver, assignee, trustee, liquidator, sequestrator or other similar official in any such judicial proceeding is hereby authorized by each Lender, the Swing Line Lender and each Issuer to make such payments to the Administrative Agent and, in the event that the Administrative Agent shall consent to the making of such payments directly to the Swing Line Lender, the Lenders and the Issuers, to pay to the Administrative Agent any amount due for the reasonable compensation, expenses, disbursements and advances of the Administrative Agent and its agents and counsel, and any other amounts due the Administrative Agent under Sections 2.5 and 9.6.

Nothing contained herein shall be deemed to authorize the Administrative Agent to authorize or consent to or accept or adopt on behalf of any Lender, the Swing Line Lender or any Issuer any plan of reorganization, arrangement, adjustment or composition affecting the Obligations or the rights of any Lender or to authorize the Administrative Agent to vote in respect of the claim of any Lender in any such proceeding.

80

ARTICLE XI

SETOFF; RATABLE PAYMENTS; PAYMENTS SET ASIDE

11.1 <u>Setoff</u>.

In addition to, and without limitation of, any rights of the Lenders under applicable law, if the Borrower becomes insolvent, however evidenced, or any Default occurs, after obtaining the prior written consent of the Administrative Agent, any and all deposits (including all account balances, whether provisional or final and whether or not collected or available) and any other Indebtedness at any time held or owing by any Lender, any Issuer, the Swing Line Lender or any such Affiliate to or for the credit or account of the Borrower may be offset and applied toward the payment of the Obligations owing to such Lender, the Issuer or the Swing Line Lender whether or not the Obligations, or any part hereof, shall then be due; provided, that in the event that any Defaulting Lender shall exercise any such right of setoff, (x) all amounts so set off shall be paid over immediately to the Administrative Agent for further application in accordance with the provisions of Section 2.23 and, pending such payment, shall be segregated by such Defaulting Lender from its other funds and deemed held in trust for the benefit of the Administrative Agent and the Lenders, and (y) the Defaulting Lender shall provide promptly to the Administrative Agent a statement describing in reasonable detail the Obligations owing to such Defaulting Lender as to which it exercised such right of setoff. Each Lender, each Issuer and the Swing Line Lender agree to notify the Borrower and the Administrative Agent promptly after any such setoff and application; provided that the failure to give such notice shall not affect the validity of such setoff and application.

11.2 Ratable Payments.

If any Lender, whether by setoff or otherwise, has payment made to it upon its Outstanding Credit Exposure (other than payments received pursuant to Section 3.1, 3.2, 3.4 or 3.5 and payments made to any Issuer in respect of Reimbursement Obligations so long as the Lenders have not funded their participations therein) in a greater proportion than that received by any other Lender, such Lender agrees, promptly upon demand, to purchase a portion of the Aggregate Outstanding Credit Exposure held by the other Lenders so that after such purchase each Lender will hold its Pro Rata Sha re of the Aggregate Outstanding Credit Exposure. If any Lender, whether in connection with setoff or amounts which might be subject to setoff or otherwise, receives collateral or other protection for its Obligations or such amounts which may be subject to setoff, such Lender agrees, promptly upon demand, to take such action necessary such that all Lenders share in the benefits of such collateral ratably in accordance with their respective Pro Rata Shares. In case any such payment is disturbed by legal process, or otherwise, appropriate further adjustments shall be made. For the avoidance of doubt, the provisions of this Section shall not be construed to apply to (A) any payment made by or on behalf of the Borrower pursuant to and in accordance with the express terms of this Agreement (including the application of funds arising from the existence of a Defaulting Lender), (B) the application of Cash Collateral provided for in Section 2.22, or (C) any payment obtained by a Lender as consideration for the assignment of or sale of a participation in any of its Loans or subparticipations in Letter of Credit Obligations or Swing Line Loans to any assignee or

participant, other than an assignment to the Borrower or any Subsidiary thereof (as to which the provisions of this Section shall apply).

11.3 Payments Set Aside.

To the extent that any payment by or on behalf of the Borrower is made to the Administrative Agent, any Issuer or any Lender, or the Administrative Agent, any Issuer or any Lender exercises its right of setoff, and such payment or the proceeds of such setoff or any part thereof is subsequently invalidated, declared to be fraudulent or preferential, set aside or required (including pursuant to any settlement entered into by the Administrative Agent, such Issuer or such Lender in its discretion) to be repaid to a trustee, receiver or any other party, in connection with any proceeding under any Debtor Relief Law or otherwise, then (a) to the extent of such recovery, the obligation or part thereof originally intended to be satisfied shall be revived and continued in full force and effect as if such payment had not been made or such setoff had not occurred, and (b) each Lender and each Issuer severally agrees to pay to the Administrative Agent upon demand its applicable share (without duplication) of any amount so recovered from or repaid by the Administrative Agent, plus interest thereon from the date of such demand to the date such payment is made at a rate per annum equal to the Federal Funds Rate from time to time in effect. The obligations of the Lenders and the Issuers under clause (b) of the preceding sentence shall survive the payment in full of the Obligations and the termination of this Agreement.

ARTICLE XII

BENEFIT OF AGREEMENT; ASSIGNMENTS; PARTICIPATIONS

12.1 <u>Successors and Assigns</u>.

(a) The provisions of this Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns permitted hereby, except that the Borrower may not assign or otherwise transfer any of its rights or obligations hereunder without the prior written consent of the Administrative Agent, the Swing Line Lender and each Lender and no Lender may assign or otherwise transfer any of its rights or obligations hereunder except (i) to an Eligible Assignee in accordance with the provisions of subsection (b) of this Section, (ii) by way of participation in accordance with the provisions of subsection (d) of this Section, or (iii) by way of pledge or assignment of a security interest subject to the restrictions of subsection (f) of this Section (and any other attempted assignment or transfer by any party hereto shall be null and void). Nothing in this Agreement, expressed or implied, shall be construed to confer upon any Person (other than the parties hereto, their respective successors and assigns permitted hereby, Participants to the extent provided in subsection (d) of this Section and, to the extent expressly contemplated hereby, the Related Parties of each of the Administrative Agent, the Issuers and the Lenders) any legal or equitable right, remedy or claim under or by reason of this Agreement.

82

- (b) Any Lender may at any time assign to one or more Eligible Assignees all or a portion of its rights and obligations under this Agreement (including all or a portion of its Commitment and the Loans (including for purposes of this <u>subsection (b)</u> participations in Letter of Credit Obligations) at the time owing to it); <u>provided</u> that
 - (i) except in the case of an assignment of the entire remaining amount of the assigning Lender's Commitment and the Loans at the time owing to it or in the case of an assignment to a Lender or an Affiliate of a Lender or an Approved Fund with respect to a Lender, the aggregate amount of the Commitment (which for this purpose includes Loans outstanding thereunder) or, if the Commitment is not then in effect, the principal outstanding balance of the Loans of the assigning Lender subject to each such assignment, determined as of the date the Assignment Agreement with respect to such assignment is delivered to the Administrative Agent or, if "Trade Date" is specified in the Assignment Agreement, as of the Trade Date, shall not be less than \$5,000,000 unless each of the Administrative Agent and, so long as no Default has occurred and is continuing, the Borrower otherwise consents (each such consent not to be unreasonably withheld or delayed); provided, however, that concurrent assignments to members of an Assignee Group and concurrent assignments from members of an Assignee Group to a single Eligible Assignee (or to an Eligible Assignee and members of its Assignee Group) will be treated as a single assignment for purposes of determining whether such minimum amount has been met; provided, further, that the Borrower shall be deemed to have consented to any such assignment unless it shall object thereto by written notice to the Administrative Agent within ten (10) Business Days after having received notice thereof.
 - (ii) each partial assignment shall be made as an assignment of a proportionate part of all the assigning Lender's rights and obligations under this Agreement with respect to the Loans or the Commitment assigned;
 - (iii) any assignment of a Commitment must be approved by the Administrative Agent, the Swing Line Lender and the Issuers (each such approval not to be unreasonably withheld or delayed) unless the Person that is the proposed assignee is itself a Lender, an Affiliate of such Lender or an Approved Fund with respect to such Lender;
 - (iv) (A) such Lender shall at the same time enter into an Assignment Agreement (as defined in the Great Plains Credit Agreement) with the same Eligible Assignee(s) in an amount representing an equal proportion of such Lender's Commitment (as defined in the Great Plains Credit Agreement) under the Great Plains Credit Agreement and (B) such Lender shall at the same time enter into an Assignment Agreement (as defined in the KCPL GMO Credit Agreement) with the same Eligible Assignee(s) in an amount representing an equal proportion of such Lender's Commitment (as defined in the KCPL GMO Credit Agreement) under the KCPL GMO Credit Agreement;
 - (v) the parties to each assignment shall execute and deliver to the Administrative Agent an Assignment Agreement, together with a processing and recordation fee of \$3,500; <u>provided</u>, <u>however</u>, that (A) such processing and recordation

fee shall not apply to assignments to Affiliates of the assigning Lender, (B) such processing and recordation fee shall, with respect to a particular assignment, be payable only once by such assigning Lender in connection with the concurrent assignments of Commitments of such assigning Lender required by Section 12.1(b)(iv) and (C) the Administrative Agent may, in its sole discretion, elect to waive such processing and recordation fee in the case of any assignment. The Eligible Assignee, if it shall not be a Lender, shall deliver to the Administrative Agent an Administrative Questionnaire;

- (vi) No such assignment shall be made (A) to the Borrower or any of the Borrower's Affiliates or Subsidiaries, (B) to any Defaulting Lender or any of its Subsidiaries, or any Person who upon becoming a Lender hereunder would constitute any of the foregoing Persons described in this <u>clause (B)</u>, or (C) to a natural person; and
- (vii) In connection with any assignment of rights and obligations of any Defaulting Lender hereunder, no such assignment shall be effective unless and until, in addition to the other conditions thereto set forth herein, the parties to the assignment shall make such additional payments to the Administrative Agent in an aggregate amount sufficient, upon distribution thereof as appropriate (which may be outright payment, purchases by the assignee of participations or subparticipations, or other compensating actions, including funding, with the consent of the Borrower and the Administrative Agent, the applicable pro rata share of Loans previously requested but not funded by the Defaulting Lender, to each of which the applicable assignee and assignor hereby irrevocably consent), to (x) pay and satisfy in full all payment liabilities then owed by such Defaulting Lender to the Administrative Agent or any Lender hereunder (and interest accrued thereon) and (y) acquire (and fund as appropriate) its full pro rata share of all Loans and participations in Letters of Credit and Swing Line Loans in accordance with its Pro Rata Share. Notwithstanding the foregoing, in the event that any assignment of rights and obligations of any Defaulting Lender hereunder shall become effective under applicable Law without compliance with the provisions of this paragraph, then the assignee of such interest shall be deemed to be a Defaulting Lender for all purposes of this Agreement until such compliance occurs.

Subject to acceptance and recording thereof by the Administrative Agent pursuant to <u>subsection</u> (c) of this Section, from and after the effective date specified in each Assignment Agreement, the Eligible Assignee thereunder shall be a party to this Agreement and, to the extent of the interest assigned by such Assignment Agreement, have the rights and obligations of a Lender under this Agreement, and the assigning Lender thereunder shall, to the extent of the interest assigned by such Assignment Agreement, be released from its obligations under this Agreement (and, in the case of an Assignment Agreement covering all of the assigning Lender's rights and obligations under this Agreement, such Lender shall cease to be a party hereto) but shall continue to be entitled to the benefits of Sections 3.1, 3.4, 3.5, and 9.6 with respect to facts and circumstances occurring prior to the effective date of such assignment. Upon request, the Borrower (at its expense) shall execute and deliver a Note to the assignee Lender. Any assignment or transfer by a Lender of rights or obligations under this Agreement that does not comply with this subsection shall be treated for purposes of this Agreement as a sale by such

84

Lender of a participation in such rights and obligations in accordance with <u>subsection (d)</u> of this Section.

- (c) The Administrative Agent, acting solely for this purpose as an agent of the Borrower, shall maintain at the Administrative Agent's Office a copy of each Assignment Agreement delivered to it and a register for the recordation of the names and addresses of the Lenders, and the Commitments of, and principal amounts of the Loans and Letter of Credit Obligations owing to, each Lender pursuant to the terms hereof from time to time (the "Register"). The entries in the Register shall be conclusive, and the Borrower, the Administrative Agent and the Lenders may treat each Person whose name is recorded in the Register pursuant to the terms hereof as a Lender hereunder for all purposes of this Agreement, notwithstanding notice to the contrary. In addition, the Administrative Agent shall maintain on the Register information regarding the designation, and revocation of designation, of any Lender as a Defaulting Lender. The Register shall be available for inspection by each of the Borrower and the Issuers at any reasonable time and from time to time upon reasonable prior notice. In addition, at any time that a request for a consent for a material or substantive change to the Loan Documents is pending, any Lender may request and receive from the Administrative Agent a copy of the Register.
- (d) Any Lender may at any time, without the consent of, or notice to, the Borrower or the Administrative Agent, sell participations to any Person (other than a natural person, a Defaulting Lender, or the Borrower or any of the Borrower's Affiliates or Subsidiaries) (each, a "Participant") in all or a portion of such Lender's rights and/or obligations under this Agreement (including all or a portion of its Commitment and/or the Loans (including such Lender's participations in Letter of Credit Obligations) owing to it); provided that (i) such Lender's obligations under this Agreement shall remain unchanged, (ii) such Lender shall remain solely responsible to the other parties hereto for the performance of such obligations and (iii) the Borrower, the Administrative Agent, the Lenders and the Issuers shall continue to deal solely and directly with such Lender in connection with such Lender's rights and obligations under this Agreement.

Any agreement or instrument pursuant to which a Lender sells such a participation shall provide that such Lender shall retain the sole right to enforce this Agreement and to approve any amendment, modification or waiver of any provision of this Agreement; provided that such agreement or instrument may provide that such Lender will not, without the consent of the Participant, agree to any amendment, waiver or other modification described in the proviso to Section 8.2 that affects such Participant. Subject to subsection (e) of this Section, the Borrower agrees that each Participant shall be entitled to the benefits of Sections 3.1, 3.4 and 3.5 to the same extent as if it were a Lender and had acquired its interest by assignment pursuant to subsection (b) of this Section. To the extent permitted by law, each Participant also shall be entitled to the benefits of Section 11.1 as though it were a Lender, provided such Participant agrees to be subject to Section 11.2 as though it were a Lender.

(e) A Participant shall not be entitled to receive any greater payment under <u>Section 3.1</u> or <u>3.5</u> than the applicable Lender would have been entitled to receive with respect to the participation sold to such Participant, unless the sale of the participation to such Participant is

made with the Borrower's prior written consent. A Participant that would be a Foreign Lender if it were a Lender shall not be entitled to the benefits of Section 3.5 unless the Borrower is notified of the participant of such Participant and such Participant agrees, for the benefit of the Borrower, to comply with Section 3.5(e) as though it were a Lender.

- (f) Any Lender may at any time pledge or assign a security interest in all or any portion of its rights under this Agreement (including under its Note, if any) to secure obligations of such Lender, including any pledge or assignment to secure obligations to a Federal Reserve Bank; <u>provided</u> that no such pledge or assignment shall release such Lender from any of its obligations hereunder or substitute any such pledgee or assignee for such Lender as a party hereto.
- (g) The words "execution," "signed," "signature," and words of like import in any Assignment Agreement shall be deemed to include electronic signatures or the keeping of records in electronic form, each of which shall be of the same legal effect, validity or enforceability as a manually executed signature or the use of a paper-based recordkeeping system, as the case may be, to the extent and as provided for in any applicable law, including the Federal Electronic Signatures in Global and National Commerce Act, the New York State Electronic Signatures and Records Act, or any other similar state laws based on the Uniform Electronic Transactions Act.
- (h) Notwithstanding anything to the contrary contained herein, if at any time Bank of America assigns all of its Commitment and Loans pursuant to subsection (b) above, Bank of America may, upon thirty (30) days' notice to the Borrower and the Lenders, resign as an Issuer. In the event of any such resignation as an Issuer, the Borrower shall be entitled to appoint from among the Lenders another Issuer hereunder who agrees to such appointment; provided, however, that no failure by the Borrower to appoint any such Issuer, or the failure of any Lender to accept su ch appointment, shall affect the resignation of Bank of America as an Issuer. If Bank of America resigns as an Issuer, it shall retain all the rights, powers, privileges and duties of an Issuer hereunder with respect to all Letters of Credit outstanding as of the effective date of its resignation as Issuer and all Letter of Credit Obligations with respect thereto (including the right to require the Lenders to fund risk participations pursuant to Section 2.19(e)).

12.2 Replacement of Lenders.

If (i) any Lender requests compensation under Section 3.1 or Section 3.2, (ii) suspends the availability of Eurodollar Loans under Section 3.3, (iii) the Borrower is required to pay any additional amount to any Lender or any Governmental Authority for the account of any Lender pursuant to Section 3.4 or Section 3.5, (iv) any Lender is a Non-Extending Lender pursuant to Section 2.21(b) or (v) any Lender is a Defaulting Lender, then the Borrower may, at its sole expense and effort, upon notice to such Lender and the Administrative Agent, require such Lender to assign and delegate, without recourse (in accordance with and subject to the restrictions contained in, and consents required by, Section 12.1(b)), all of its interests, rights and obligations under this Agreement and the related Loan Documents to an Eligible Assignee that shall assume such obligations (which Eligible Assignee may be another Lender, if a Lender accepts such assignment), provided that:

86

- (a) the Borrower shall have paid to the Administrative Agent the assignment fee specified in Section 12.1(b);
- (b) such Lender shall have received payment of an amount equal to the outstanding principal of its Loans and participations, accrued interest thereon, accrued fees and all other amounts payable to it hereunder and under the other Loan Documents (including any amounts payable under Section 3.4) from such Eligible Assignee (to the extent of such outstanding principal and accrued interest and fees) or the Borrower (in the case of all other amounts);
- (c) in the case of any such assignment resulting from a claim for compensation under <u>Section 3.1</u> or payments required to be made pursuant to <u>Section 3.4</u>, such assignment will result in a reduction in such compensation or payments thereafter; and
 - (d) such assignment does not conflict with applicable Laws.

A Lender shall not be required to make any such assignment or delegation if, prior thereto, as a result of a waiver by such Lender or otherwise, the circumstances entitling the Borrower to require such assignment and delegation cease to apply.

ARTICLE XIII

NOTICES

13.1 Notices.

Except as otherwise permitted by Section 2.14 with respect to borrowing notices, all notices, requests and other communications to any party hereunder shall be in writing (including electronic transmission, facsimile transmission or similar writing) and shall be given to such party: (i) in the case of the Borrower or the Administrative Agent, at its address or facsimile number set forth on Schedule IV, (ii) in the case of any Lender, at its address or facsimile number specified in its Administrative Questionnaire or (iii) in the case of any party, at such other address or facsimile number as such party may hereafter specify for the purpose by notice to the Administrative Agent and the Borrower in accordance with the provisions of this Section 13.1. Each such notice, request or other communication shall be effective (a) if given by facsimile transmission, when transmitted to the facsimile number specified in this Section and confirmation of receipt is received, (b) if given by mail, when received, or (c) if given by any other means, when delivered (or, in the case of electronic transmission, received) at the address specified in this Section; provided that notices to the Administrative Agent under Article II shall not be effective until received.

13.2 Change of Address.

The Borrower, the Administrative Agent and any Lender may each change the address for service of notice upon it by a notice in writing to the other parties hereto.

13.3 The Platform.

THE PLATFORM IS PROVIDED "AS IS" AND "AS AVAILABLE." THE AGENT PARTIES (AS DEFINED BELOW) DO NOT WARRANT THE ACCURACY OR COMPLETENESS OF THE BORROWER MATERIALS OR THE ADEQUACY OF THE PLATFORM, AND EXPRESSLY DISCLAIM LIABILITY FOR ERRORS IN OR OMISSIONS FROM THE BORROWER MATERIALS. NO WARRANTY OF ANY KIND, EXPRESS, IMPLIED OR STATUTORY, INCLUDING ANY WARRANTY OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE, NON INFRINGEMENT OF THIRD PARTY RIGHTS OR FREEDOM FROM VIRUSES OR OTHER CODE DEFECTS, IS MADE BY ANY AGENT PARTY IN CONNECTION WITH THE BORROWER MATERIALS OR THE PLATFORM. In no event shall the Administrative Agent or any of its Related Parties (collectively, the "Agent Parties") have any liability to Borrower, any Lender, any Issuer or any other Person for losses, claims, damages, liabilities or expenses of any kind (whether in tort, contract or otherwise) arising out of Borrower's or the Administrative Agent's transmission of Borrower Materials through the Internet, except to the extent that such losses, claims, damages, liabilities or expenses are determined by a court of competent jurisdiction by a final and nonappealable judgment to have resulted from the gross negligence or willful misconduct of such Agent Party; provided, however, that in no event shall any Agent Party have any liability to Borrower, any Lender, any Issuer or any other Person for indirect, special, incidental, consequential or punitive damages (as opposed to direct or actual damages).

ARTICLE XIV

COUNTERPARTS

This Agreement may be executed in any number of counterparts, all of which taken together shall constitute one agreement, and any of the parties hereto may execute this Agreement by signing any such counterpart. This Agreement shall be effective when it has been executed by the Borrower, the Administrative Agent and the Lenders and each party has notified the Administrative Agent by facsimile transmission or telephone that it has taken such action.

ARTICLE XV

OTHER AGENTS

No Lender identified on the cover page, the signature pages or otherwise in this Agreement, or in any document related hereto, as being the "Syndication Agent" or a "Co-Documentation Agent" shall have any right, power, obligation, liability, responsibility or

88

duty under this Agreement in such capacity other than those applicable to all Lenders. Each Lender acknowledges that it has not relied, and will not rely, on the Syndication Agent or any Co-Documentation Agent in deciding to enter into this Agreement or in taking or refraining from taking any action hereunder or pursuant hereto.

ARTICLE XVI

CHOICE OF LAW; CONSENT TO JURISDICTION; WAIVER OF JURY TRIAL.

16.1 <u>CHOICE OF LAW</u>.

THIS AGREEMENT SHALL BE CONSTRUED IN ACCORDANCE WITH THE INTERNAL LAWS (AND NOT THE LAW OF CONFLICTS) OF THE STATE OF NEW YORK, BUT GIVING EFFECT TO FEDERAL LAWS APPLICABLE TO NATIONAL BANKS.

16.2 <u>CONSENT TO JURISDICTION.</u>

THE BORROWER, THE ADMINISTRATIVE AGENT, THE SWING LINE LENDER, EACH LENDER AND EACH ISSUER HEREBY IRREVOCABLY SUBMIT TO THE NON-EXCLUSIVE JURISDICTION OF ANY UNITED STATES FEDERAL OR NEW YORK STATE COURT SITTING IN NEW YORK CITY, NEW YORK IN ANY ACTION OR PROCEEDING ARISING OUT OF OR RELATING TO THIS AGREEMENT AND THE BORROWER, THE ADMINISTRATIVE AGENT, THE SWING LINE LENDER, EACH LENDER AND EACH ISSUER HEREBY IRREVOCABLY AGREE THAT ALL CLAIMS IN RESPECT OF SUCH ACTION OR PROCEEDING MAY BE HEARD AND DETERMINED IN ANY SUCH COURT AND IRREVOCABLY WAIVE ANY OBJECTION THEY MAY NOW OR HEREAFTER HAVE AS TO THE VENUE OF ANY SUCH SUIT, ACTION OR PROCEEDING BROUGHT IN SUCH A COURT OR THAT SUCH COURT IS AN INCONVENIENT FORUM. NOTHING HEREIN SHALL LIMIT THE RIGHT OF ANY PARTY TO BRING PROCEEDINGS AGAINST THE OTHER PARTIES IN THE COURTS OF ANY OTHER JURISDICTION. ANY JUDICIAL PROCEEDING INVOLVING, DIRECTLY OR INDIRECTLY, ANY MATTER IN ANY WAY ARISING OUT OF, RELATED TO, OR CONNECTED WITH THIS AGREEMENT SHALL BE BROUGHT ONLY IN A COURT IN NEW YORK CITY, NEW YORK.

16.3 <u>WAIVER OF JURY TRIAL</u>.

TRIAL BY JURY IN ANY JUDICIAL PROC	EEDING INV AY ARISING C	ENT, THE SWING LINE LENDER, EACH LENDER AND EACH ISSUER HEREBY WAIVE OLVING, DIRECTLY OR INDIRECTLY, ANY MATTER (WHETHER SOUNDING IN TORT, DUT OF, RELATED TO, OR CONNECTED WITH ANY LOAN DOCUMENT OR THE
		89
		ARTICLE XVII
	TERMINA	ATION OF EXISTING CREDIT FACILITY
Facility) hereby waive any advance notice req	uirement for tendit Facility and	ility (and which constitute "Required Lenders" under and as defined in the Existing Credit rminating the commitments under the Existing Credit Facility, and the Borrower and the the commitments thereunder shall be terminated on the date hereof (except for any provisions
		90
IN WITNESS WHEREOF, the Bondate first above written.		ders, the Swing Line Lender, the Issuers and the Agents have executed this Agreement as of the CITY POWER & LIGHT COMPANY, ower /s/ Michael W. Cline Michael W. Cline Vice President – Investor Relations and Treasurer and Corporate Secretary
		AMERICA, N.A., rative Agent /s/ Kimberly Williams Kimberly Williams Vice President
	BANK OF A	AMERICA, N.A., , as Swing Line Lender and as a Lender

By: /s/ Kevin P. Bertelsen Name: Kevin P. Bertelsen Title: Senior Vice President as an Issuer and as a Lender

By: /s/ Bradford Joyce
Name: Bradford Joyce
Title: Authorized Signatory

WELLS FARGO BANK, NATIONAL ASSOCIATION, as Syndication Agent, as an Issuer and as a Lender

By: /s/ Allison Newman Name: Allison Newman Title: Vice President

UNION BANK, N.A., as Syndication Agent and as a Lender

By: /s/ Susan K. Johnson
Name: Susan K. Johnson
Title: Vice President

JPMORGAN CHASE BANK, N.A., as a Lender

By: /s/ Nancy R. Barwig
Name: Nancy R. Barwig
Title: Senior Vice President

BNP PARIBAS, as a Lender

By: /s/ Pasquale A. Perraglia IV Name: Pasquale A. Perraglia IV

Title: Vice President

By: /s/ Denis O'Meara Name: Denis O'Meara Title: Managing Director

THE BANK OF NOVA SCOTIA, as a Lender

By: /s/ Thane Rattew
Name: Thane Rattew
Title: Managing Director

THE ROYAL BANK OF SCOTLAND PLC, as a Lender

By: /s/ Tyler J. McCarthy Name: Tyler J. McCarthy

Title: Director

BARCLAYS BANK PLC, as a Lender

By: /s/ David Barton
Name: David Barton
Title: Director

U.S. BANK NATIONAL ASSOCIATION as a Lender

By: /s/ Eric J. Cosgrove Name: Eric J. Cosgrove Title: Vice President

THE BANK OF NEW YORK MELLON, as a Lender

By: /s/ Hussam S. Alsahlani Name: Hussam S. Alsahlani Title: Senior Associate

KEYBANK NATIONAL ASSOCIATION, as a Lender

By: /s/ Keven D. Smith
Name: Keven D. Smith
Title: Senior Vice President

SUNTRUST BANK, as a Lender

By: /s/ Andrew Johnson
Name: Andrew Johnson
Title: Director

DEUTSCHE BANK AG NEW YORK BRANCH, as a Lender

By: /s/ Frederick Laird
Name: Frederick Laird
Title: Managing Director

By: /s/ Heidi Sandquist Name: Heidi Sandquist Title: Director

UMB BANK, N.A., as a Lender

By: /s/ Robert P. Elbert
Name: Robert P. Elbert
Title: Senior Vice President

COMMERCE BANK, N.A., as a Lender

By: /s/ Aaron M. Siders Name: Aaron M. Siders

Title: Commercial Banking Officer

SCHEDULE I

COMMITMENTS

Lender	Commitment
Bank of America, N.A.	\$57,600,000.00
Wells Fargo Bank, National Association	\$57,600,000.00
The Bank of Tokyo-Mitsubishi UFJ, Ltd.	\$28,800,000.00
Union Bank, N.A.	\$28,800,000.00
JPMorgan Chase Bank, N.A.	\$40,800,000.00
BNP Paribas	\$40,800,000.00
The Bank of Nova Scotia	\$40,800,000.00
The Royal Bank of Scotland plc	\$40,800,000.00
Barclays Bank PLC	\$40,800,000.00
U.S. Bank National Association	\$40,800,000.00
The Bank of New York Mellon	\$36,000,000.00
KeyBank National Association	\$36,000,000.00
SunTrust Bank	\$36,000,000.00
Deutsche Bank AG New York Branch	\$36,000,000.00
UMB Bank, n.a.	\$24,000,000.00
Commerce Bank, N.A.	\$14,400,000.00
Total	\$600,000,000.00

LETTER OF CREDIT COMMITMENTS

Issuer	Letter of Credit Commitment
Bank of America, N.A.	\$25,000,000.00
Wells Fargo Bank, National Association	\$25,000,000.00
The Bank of Tokyo-Mitsubishi UFJ, Ltd.	\$25,000,000.00
Total	\$75,000,000.00

SCHEDULE II

EXISTING LETTERS OF CREDIT

Issuer	Beneficiary	Amount	Expiry Date	Letter of Credit Number
Part A				
JPMorgan Chase Bank, N.A.	Southwest Power Pool	\$ 3,461,805.00	05/05/11	CPCS-286557
JPMorgan Chase Bank, N.A.	ACE American Insurance Company	\$ 8,476,467.00	11/09/10	CPCS-292173
Part B				
Bank of America, N.A.	Missouri Transportation	\$ 5,122,649.00	05/07/11	T00000068044449
Bank of America, N.A.	NECA/IBEW Family Med Care Plan	\$ 3,860,000.00	04/03/11	T00000068046279
Bank of America, N.A.	Southwest Power Pool	\$ 1,762,500.00	05/11/11	T00000068050227
Bank of America, N.A.	Southwest Power Pool	\$ 1,762,500.00	05/11/11	T00000068050228

SCHEDULE III

PRICING SCHEDULE

	>A-/A3	>BBB+/Baa1	>BBB/Baa2	>BBB-/Baa3	>BB+/Ba1	<bb ba2<="" th=""></bb>
Pricing	LEVEL I STATUS	LEVEL II STATUS	LEVEL III STATUS	LEVEL IV STATUS	LEVEL V Status	LEVEL VI STATUS
Applicable Margin for Eurodollar Advances and Letter of Credit Fee Rate	1.75%	2.00%	2.25%	2.75%	3.00%	3.25%
Commitment Fee Rate	0.225%	0.30%	0.375%	0.50%	0.625%	0.625%
Applicable Margin for Floating Rate Advances	0.75%	1.00%	1.25%	1.75%	2.00%	2.25%

"Level I Status" exists at any date if, on such date, the Borrower's Moody's Rating is A3 or better or the Borrower's S&P Rating is A- or better.

"Level II Status" exists at any date if, on such date, (i) the Borrower has not qualified for Level I Status and (ii) the Borrower's Moody's Rating is Baa1 or better *or* the Borrower's S&P Rating is BBB+ or better.

"Level III Status" exists at any date if, on such date, (i) the Borrower has not qualified for Level I Status or Level II Status and (ii) the Borrower's Moody's Rating is Baa2 or better *or* the Borrower's S&P Rating is BBB or better.

"Level IV Status" exists at any date if, on such date, (i) the Borrower has not qualified for Level I Status, Level II Status or Level III Status and (ii) the Borrower's Moody's Rating is Baa3 or better *or* the Borrower's S&P Rating is BBB- or better.

"Level V Status" exists at any date if, on such date, (i) the Borrower has not qualified for Level I Status, Level II Status, Level III Status or Level IV Status and (ii) the Borrower's Moody's Rating is BB+ or better *or* the Borrower's S&P Rating is Ba1 or better.

"Level VI Status" exists at any date if, on such date, the Borrower has not qualified for Level I Status, Level II Status, Level III Status, Level IV Status or Level V Status.

"Moody's Rating" means, at any time, the rating issued by Moody's and then in effect with respect to the Borrower's senior unsecured long-term debt securities without third-party credit enhancement (or, if there is no such debt outstanding, the Borrower's issuer rating issued by Moody's then in effect for the Borrower).

"S&P Rating" means, at any time, the rating issued by S&P and then in effect with respect to the Borrower's senior unsecured long-term debt securities without third-party credit

enhancement (or, if there is no such debt outstanding, the Borrower's issuer rating issued by S&P then in effect for the Borrower).

"Status" means Level I Status, Level II Status, Level III Status, Level IV Status, Level V Status or Level VI Status.

The Applicable Margin for Eurodollar Advances, the Letter of Credit Fee Rate, the Applicable Margin for Floating Rate Advances and the Commitment Fee Rate shall be determined in accordance with the foregoing pricing grid based on the Borrower's Status as determined from its then-current Moody's and S&P Ratings. The credit rating in effect on any date for purposes of this Schedule is that in effect at the close of business on such date. If at any time the Borrower ceases to have a Moody's Rating or an S&P Rating, Level VI Status shall exist.

Notwithstanding the foregoing, (a) if the Borrower is split-rated and the ratings differential is one level, the higher rating will apply; and (b) if the Borrower is split-rated and the ratings differential is two levels or more, the intermediate rating at the midpoint will apply. If there is no midpoint, the higher of the two intermediate ratings will apply.

SCHEDULE IV

1200 Main Street

Kansas City, MO 64105

Attention: Michael W. Cline, Vice-President – Investor Relations, Treasurer and Corporate Secretary

Telephone: (816) 556-2622 Facsimile: (816) 556-2992

Electronic Mail: michael.cline@kcpl.com

ADMINISTRATIVE AGENT:

Administrative Agent's Office

(for payments and Requests for Credit Extensions):

Bank of America, N.A. Credit Services Rep. II Mail Code: TX1-492-14-12 Bank of America Plaza 901 Main Street Dallas, TX 75202-3714

Attention: Karen Puente Telephone: 214-209-4108 Telecopier: 214-290-8378

Electronic Mail: Karen.r.puente@baml.com

Bank of America New York, NY Account No.: 1292000883 Account Name: Credit Services

Ref: Great Plains ABA# 026009593

Other Notices as Administrative Agent:

Bank of America, N.A. Agency Management Bank of America Plaza 101 S. Tryon Street Charlotte, NC 28255-0001 Mail Code: NC1-002-15-36

Attention: Kelly Weaver
Telephone: (980) 387-5452
Facsimile: (704) 208-2871

Electronic Mail: kelly.weaver@baml.com

ISSUERS:

Bank of America, N.A. Trade Operations 1000 West Temple Street, 7th Floor Mail Code: CA9-705-07-05 Los Angeles, CA 90012 1514

Attention: Sandra Leon
Telephone: (213) 580-8369
Telecopier: (213) 457-8841
Electronic Mail: sandra.leon@baml.com

The Bank of Tokyo-Mitsubishi UFJ, Ltd. 1251 Avenue of the Americas, 12th Floor New York, NY 10020-1104

Attention: Jaya Angara Attention: Antonina Bondi Telephone: (201) 413-8843 Telephone: (201) 413-8823 Telecopier: (201) 413-8817 Telecopier: (201) 413-8817 Electronic Mail: jangara@us.mufg.jp Electronic Mail: abondi@us.mufg.jp

Wells Fargo Bank, N 301 S. College Stree Charlotte, NC 28288	t, 15th Floor
Attention: Telephone: Telecopier: Electronic Mail:	Elaine Shue (704) 715-3133 (704) 715-0358 Elaine.Shue@wachovia.com
	EXHIBIT A
	COMPLIANCE CERTIFICATE
To: The Lende Credit Agreement D	ers parties to the escribed Below
from time to time, "Lenders"), Union E	liance Certificate is furnished pursuant to the Credit Agreement dated as of August 9, 2010 (as amended, modified, renewed or extended the "Agreement") among Kansas City Power & Light Company (the "Borrower"), the lenders from time to time party thereto (the Bank, N.A. and Wells Fargo Bank, National Association, as Syndication Agents, and Bank of America, N.A., as Administrative Agent for otherwise defined herein, capitalized terms used in this Compliance Certificate have the meanings ascribed thereto in the Agreement.
THE UNDI	ERSIGNED HEREBY CERTIFIES THAT:
1. I a	m the duly electedof the Borrower;
transactions and cor	ave reviewed the terms of the Agreement and I have made, or have caused to be made under my supervision, an appropriate review of the nditions of the Borrower and, to the extent relevant to the Agreement, its Subsidiaries during the accounting period covered by the attached as <u>Schedule I</u> ;
constitutes a Default Certificate, except as	The examinations described in paragraph 2 did not disclose, and I have no knowledge of, the existence of any condition or event which or Unmatured Default during or at the end of the accounting period covered by the attached financial statements or as of the date of this set forth on Schedule II which lists, in detail, the nature of the condition or event, the period during which it has existed and the action has taken, is taking, or proposes to take with respect to each such condition or event; and
4. Ser	t forth on <u>Schedule III</u> are true and correct computations of the financial ratios and restrictions set forth in Section 6.15 of the Agreement.
The foregoiday of,	ing certifications, together with the financial statements delivered with this Certificate in support hereof, are made and delivered this _
	KANSAS CITY POWER & LIGHT COMPANY
Title:	By: Print Name:

SCHEDULE I

TO COMPLIANCE CERTIFICATE

Reports and Deliveries Currently Due

SCHEDULE II

TO COMPLIANCE CERTIFICATE

Conditions or Events which Constitute a Default or Unmatured Default

SCHEDULE III

TO COMPLIANCE CERTIFICATE

Computation of Financial Covenants

EXHIBIT B

ASSIGNMENT AGREEMENT

This Assignment and Assumption (the "Assignment and Assumption") is dated as of the Effective Date set forth below and is entered into by and between [Insert name of Assignor] (the "Assignor") and [Insert name of Assignee] (the "Assignee"). Capitalized terms used but not defined herein shall have the respective meanings given to them in the Credit Agreement identified below (as amended, the "Credit Agreement"), receipt of a copy of which is here by acknowledged by the Assignee. The Terms and Conditions set forth in Annex 1 attached hereto (the "Terms and Conditions") are hereby agreed to and incorporated herein by reference and made a part of this Assignment and Assumption as if set forth herein in full.

For an agreed consideration, the Assignor hereby irrevocably sells and assigns to the Assignee, and the Assignee hereby irrevocably purchases and assumes from the Assignor, subject to and in accordance with the Terms and Conditions and the Credit Agreement, as of the Effective Date inserted by the Administrative Agent as contemplated below, the interest in and to all of the Assignor's rights and obligations in its capacity as a Lender under the Credit Agreement and any other documents or instruments delivered pursuant thereto that represents the amount and percentage interest identified below of all of the Assignor's outstanding rights and obligations under the respective facilities identified below (including any letters of credit, swing line loans and guaranties included in such facilities and, to the extent permitted to be assigned under applicable law, all claims (including contract claims, tort claims, malpractice claims, statutory claims and all other claims at law or in equity), suits, causes of action and any other right of the Assignor against any Person whether known or unknown arising under or in connection with the Credit Agreement, any other documents or instruments delivered pursuant thereto or the loan transactions governed thereby) (the "Assigned Interest"). Such sale and assignment is without recourse to the Assignor and, except as expressly provided in this Assignment and Assumption, without representation or warranty by the Assignor.

	_	
2.	Assignee:	
3.	Borrower:	Kansas City Power & Light Company
4.	Administrative Age	nt: Bank of America, N.A., as the Administrative Agent under the Credit Agreement.
5.	Credit Agreement:	Credit Agreement dated as of August 9, 2010 among the Borrower, the Lenders from time to time party thereto and Bank of America, N.A., as Administrative Agent.

Assigned Interest:

Assignor:

6.

1.

Aggregate Amount of Commitment/Loans for all Lenders*	Amount of Commitment/Loans Assigned*	Percentage Assigned of Commitment/Loans ¹
\$		%
\$		%

7.	Trade Date:	<u>2</u>
	ive Date: [TO BE INSERTED BY A INISTRATIVE AGENT.]	, 20 GENT AND WHICH SHALL BE THE EFFECTIVE DATE OF RECORDATION OF TRANSFER BY THE
	The terms set forth in this	Assignment and Assumption are hereby agreed to:
		ASSIGNOR [NAME OF ASSIGNOR]
		By: Title:
		ASSIGNEE [NAME OF ASSIGNEE]
		By: Title:
	ented to and] ³ Accepted: K OF AMERICA, N.A., as Ac	lministrative Agent
By: Title:		
* Amo	ount to be adjusted by the cou	nterparties to take into account any payments or prepayments made between the Trade Date and the Effective Date.
		a percentage of the Commitment/Loans of all Lenders thereunder.
		amounts is to be determined as of the Trade Date.
³ To b	e added only if the consent of	the Administrative Agent is required by the terms of the Credit Agreement.
	ented to:]4	
KANS	SAS CITY POWER & LIGHT	Γ COMPANY
By: Title:		
[Cons	ented to:] ⁵	
[ISSU	ERS]	
By: _ Title:		
[Cons	ented to:] ⁶	
[SWII	NG LINE LENDER]	

____%

\$

\$

Title:
⁴ To be added only if the consent of the Borrower is required by the terms of the Credit Agreement.
⁵ To be added only if the consent of the Issuers is required by the terms of the Credit Agreement.
⁶ To be added only if the consent of the Swing Line Lender is required by the terms of the Credit Agreement.
ANNEX 1 TERMS AND CONDITIONS FOR ASSIGNMENT AND ASSUMPTION
1. Representations and Warranties.
1.1 <u>Assignor</u> . The Assignor (a) represents and warrants that (i) it is the legal and beneficial owner of the Assigned Interest, (ii) the Assigned Interest is free and clear of any lien, encumbrance or other adverse claim and (iii) it has full power and authority, and has taken all action necessary, to execute and deliver this Assignment and to consummate the transactions contemplated hereby; and (b) assumes no responsibility with respect to (i) any statements, warranties or representations made in or in connection with the Credit Agreement or any other Loan Document, (ii) the execution, legality, validity, enforceability, genuineness, sufficiency or value of the Loan Documents or any collateral th ereunder, (iii) the financial condition of the Borrower, any of its Subsidiaries or Affiliates or any other Person obligated in respect of any Loan Document or (iv) the performance or observance by the Borrower, any of its Subsidiaries or Affiliates or any other Person of any of their respective obligations under any Loan Document.
Assignee. The Assignee (a) represents and warrants that (i) it has full power and authority, and has taken all action necessary, to execute and deliver this Assignment and to consummate the transactions contemplated hereby and to become a Lender under the Credit Agreement, (ii) it meets all requirements of an Eligible Assignee under the Credit Agreement (subject to receipt of such consents as may be required under the Credit Agreement), (iii) from and after the Effective Date, it shall be bound by the provisions of the Credit Agreement as a Lender thereunder and, to the extent of the Assigned Interest, shall have the obligations of a Lender thereunder, (iv) it has received a copy of the Credit t Agreement, together with copies of the most recent financial statements delivered pursuant to Section 6.1(a) and (b) thereof, as applicable, and such other documents and information as it has deemed appropriate to make its own credit analysis and decision to enter into this Assignment and to purchase the Assigned Interest on the basis of which it has made such analysis and decision independently and without reliance on the Administrative Agent or any other Lender and (v) if it is a Foreign Lender, attached hereto is any documentation required to be delivered by it pursuant to the terms of the Credit Agreement, duly completed and executed by the Assignee; and (b) agrees that (i) it will, independently and without reliance on the Administrative Agent, the Assignor or any other Lender, and based on such documents and information as it shall deem appropriate at the time, continue to make its own credit decisions in taking or not taking action under the Loan Documents, and (ii) it will perform in accordance with their terms all of the obligations which by the terms of the Loan Documents are required to be performed by it as a Lender.
2. <u>Payments</u> . From and after the Effective Date, the Administrative Agent shall make all payments in respect of the Assigned Interest (including payments of principal, interest, fees and other amounts) to the Assignor for amounts which have accrued to but excluding the
Effective Date and to the assignee for amounts which have accrued from and after the Effective Date. 3. General Provisions. This Assignment shall be binding upon, and inure to the benefit of, the parties hereto and their respective successors and assigns. This Assignment may be executed in any number of counterparts, which together shall constitute one instrument. Delivery of an executed counterpart of a signature page of this Assignment by telecopy shall be effective as delivery of a manually executed counterpart of this Assignment. This Assignment shall be governed by, and construed in accordance with, the law of the State of New York.

as Administrative Agent (the "Administrative Agent") under the Credit Agreement described below.

Encility Identification Number(s)

Re: Credit Agreement dated as of August 9, 2010 (as the same may be amended or modified, the "Credit Agreement"), among Kansas City Power & Light Company (the "Borrower"), the lenders from time to time party thereto, Union Bank, N.A. and Wells Fargo Bank, National Association, as Syndication Agents, and the Administrative Agent. Capitalized terms used herein and not otherwise defined herein shall have the meanings assigned thereto in the Credit Agreement.

The Administrative Agent is specifically authorized and directed to act upon the following standing money transfer instructions with respect to the proceeds of Credit Extensions or other extensions of credit from time to time until receipt by the Administrative Agent of a specific written revocation of such instructions by the Borrower; provided that the Administrative Agent may otherwise transfer funds as hereafter directed in writing by the Borrower in accordance with Section 13.1 of the Credit Agreement or based on any telephonic notice made in accordance with Section 2.14 of the Credit Agreement.

racinty identification (vulnoer(s)	_
Customer/Account Name	_
Transfer Funds To	
For Account No.	
Reference/Attention To	
Authorized Officer (Customer Representative)	Date
(Please Print)	Signature
Bank Officer Name	Date
(Please Print)	Signature
(Deliver Con	oleted Form to Credit Support Staff For Immediate Processing)
	EWHIDIT D
	EXHIBIT D
	REVOLVING NOTE
	, 20
(the "Le	COMPANY, a Missouri corporation (the "Borrower"), promises to pay to the order of der") the aggregate unpaid principal amount of all Loans made by the Lender to the Borrower pursuant
), in immediately available funds at the Administrative Agent's Office, together with interest on the edates set forth in the Agreement. The Borrower shall pay the principal of and accrued and unpaid Date.

The Lender shall, and is hereby authorized to, record on a schedule attached hereto, or to otherwise record in accordance with its usual practice, the date and amount of each Loan and the date and amount of each principal payment hereunder.

This Revolving Note is one of the Revolving Notes issued pursuant to, and is entitled to the benefits of, the Credit Agreement dated as of August 9, 2010 (which, as it may be amended or modified and in effect from time to time, is herein called the "Agreement"), among the Borrower, the lenders from time to time party thereto, including the Lender, Union Bank, N.A. and Wells Fargo Bank, National Association, as Syndication Agents, and Bank of America, N.A., as Administrative Agent, to which Agreement reference is hereby made for a statement of the terms and conditions governing this Revolving Note, including the terms and conditions under which this Revolving Note may be prepaid or its maturity date accelerated. Capitalized terms used herein and not otherwise defined herein are used with the meanings attributed t o them in the Agreement.

THIS REVOLVING NOTE SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF NEW YORK.

[Remainder of page intentionally left blank]

KANSAS CITY POV	ER & LIGHT COMPANY
Title:	By: Print Name:
	EXHIBIT E
	SWING LINE NOTE
	, 20_
accordance with the provision the Swing Line Lender to the from time to time, the "Credi National Association, as Synd otherwise defined herein have The Borrower promis principal amount is paid in full made to the Administrative Agany amount is not paid in full of the second secon	ED, the undersigned (the "Borrower"), hereby promises to pay to BANK OF AMERICA, N.A. (the "Swing Line Lender"), of the Credit Agreement (as hereinafter defined), the principal amount of each Swing Line Loan from time to time made be borrower under that certain Credit Agreement dated as of August 9, 2010 (as amended, modified, supplemented or extended Agreement") among the Borrower, the Lenders from time to time party thereto, Union Bank, N.A. and Wells Fargo Bankation Agents, and Bank of America, N.A., as Administrative Agent and Swing Line Lender. Capitalized terms used but note meanings provided in the Credit Agreement. It is to pay interest on the unpaid principal amount of each Swing Line Loan from the date of such Swing Line Loan until such at such interest rates and at such times as provided in the Credit Agreement. All payments of principal and interest shall be not for the account of the Swing Line Lender in dollars in immediately available funds at the Administrative Agent's Office. Then due hereunder, such unpaid amount shall bear interest, to be paid upon demand, from the due date thereof until the date well as after judgment) computed at the per annum rate set forth in the Credit Agreement.
This Swing Line Not in part subject to the terms ar Agreement, all amounts then r in the Credit Agreement. Swi Lender in the ordinary course	is the Swing Line Note referred to in the Credit Agreement, is entitled to the benefits thereof and may be prepaid in whole conditions provided therein. Upon the occurrence and continuation of one or more of the Defaults specified in the Credit nation of unique of the Defaults specified in the Credit nation of unique of the Defaults specified in the Credit nation of unique of the Defaults specified in the Credit nation of the Defaults specified
The Borrower, for it dishonor and nonpayment of the	elf, its successors and assigns, hereby waives diligence, presentment, protest and demand and notice of protest, deman s Swing Line Note.
THIS SWING LINE YORK.	OTE SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF NE
	[Remainder of page intentionally left blank]
	KANSAS CITY POWER & LIGHT COMPANY
Title:	By: Print Name:



Published CUSIP Number: 48243EAC9

CREDIT AGREEMENT

Dated as of August 9, 2010

among

KCP&L GREATER MISSOURI OPERATIONS COMPANY, as the Borrower,

GREAT PLAINS ENERGY INCORPORATED, as the Guarantor,

CERTAIN LENDERS,

BANK OF AMERICA, N.A., as Administrative Agent

and

UNION BANK, N.A. and WELLS FARGO BANK, NATIONAL ASSOCIATION, as Syndication Agents

THE ROYAL BANK OF SCOTLAND PLC and BNP PARIBAS, as Documentation Agents

BANC OF AMERICA SECURITIES LLC,
UNION BANK, N.A.
and
WELLS FARGO SECURITIES, LLC,
as Joint Lead Arrangers and Joint Book Managers

TABLE OF CONTENTS

		Page
ARTICLE	I <u>DEFINITIONS</u>	
1.1	<u>Definitions</u> .	1
1.2	Accounting Principles.	20
1.3	Letter of Credit Amounts.	20
1.4	<u>Times of Day</u> .	20
ARTICLE	II THE CREDITS	
2.1	Commitment.	20
2.2	Required Payments; Termination.	21
2.3	Ratable Loans.	21
2.4	Types of Advances; Minimum Amount.	21
2.5	Commitment Fee.	21
2.6	<u>Changes in Aggregate Commitment</u> .	21
2.7	Optional Prepayments.	24
2.8	Method of Selecting Types and Interest Periods for New Advances.	25
2.9	Conversion and Continuation of Outstanding Advances.	26
2.10	<u>Interest Rates</u> .	26
2.11	Rates Applicable After Default.	27
2.12	Method of Payment.	27
2.13	Noteless Agreement; Evidence of Indebtedness.	28

2.14	<u>Telephonic Notices.</u>	28
2.15	<u>Interest Payment Dates</u> ; <u>Interest and Fee Basis</u> .	29
2.16	Notification of Advances, Interest Rates, Prepayments and Commitment Reductions.	29
2.17	<u>Lending Installations</u> .	29
2.18	Non-Receipt of Funds by the Administrative Agent.	30
2.19	<u>Letters of Credit.</u>	30
2.20	Swing Line Loans.	37
2.21	Extension of Facility Termination Date.	40
2.22	Cash Collateral.	42
2.23	<u>Defaulting Lenders</u> .	43
	YIELD PROTECTION; TAXES	
3.1	Yield Protection.	45
3.2	<u>Changes in Capital Adequacy Regulations.</u>	46
3.3	<u>Availability of Types of Advances.</u>	46
3.4	<u>Funding Indemnification</u> .	46
3.5	<u>Taxes</u> .	47
3.6	<u>Lender Statements; Survival of Indemnity.</u>	50
ARTICLE IV		
4.1	<u>The Guaranty.</u>	51
4.2	Obligations Unconditional.	51
4.3	Reinstatement.	52
4.4	Certain Additional Waivers.	52
4.5	Remedies.	53
4.6	<u>Guarantee of Payment; Continuing Guarantee</u> .	53
	i	
	1	
ARTICLE V C	CONDITIONS PRECEDENT	
5.1	Initial Credit Extension.	53
5.2	Each Credit Extension.	54
	REPRESENTATIONS AND WARRANTIES	٥.
6.1	Existence and Standing.	55
6.2	Authorization and Validity.	55
6.3	No Conflict; Government Consent.	56
6.4	Financial Statements.	56
6.5	Material Adverse Change.	56
6.6	Taxes.	57
6.7	<u>Litigation; etc.</u>	57
6.8	ERISA.	57
6.9	Accuracy of Information.	57
6.10	Regulation U.	58
6.11	Material Agreements.	58
6.12	Compliance With Laws.	58
6.13	Ownership of Properties.	58
6.14	Plan Assets; Prohibited Transactions.	59
6.15	Environmental Matters.	59
6.16	Investment Company Act.	59
6.17	Pari Passu Indebtedness.	59
6.18	Solvency.	59
6.19	No Defaults.	60
ARTICLE VII	COVENANTS	
7.1	<u>Financial Reporting and Notices.</u>	60
7.2	Permits, Etc.	62
7.3	<u>Use of Proceeds</u> .	62
7.4	Notice of Default.	63
7.5	Conduct of Business.	63
7.6	<u>Taxes</u> .	63
7.7	<u>Insurance</u> .	63
7.8	Compliance with Laws.	64
7.9	Maintenance of Properties; Books of Record.	64
7.10	<u>Inspection</u> .	64
7.11	Consolidations, Mergers and Sale of Assets.	65
7.12	Liens.	66
7.13	Affiliates.	70
7.14	ERISA.	70
7.15	Total Indebtedness to Total Capitalization.	70
7.16	Restrictions on Subsidiary Dividends.	70
7.17	Organization Documents.	70
ARTICLE VIII		
	ACCELERATION, WAIVERS, AMENDMENTS AND REMEDIES	
9.1	Acceleration; Letter of Credit Account.	73
9.2	Amendments.	74
9.3	<u>Preservation of Rights.</u>	75

ARTICLE X G	ENERAL PROVISIONS	
10.1	Survival of Representations.	75
10.2	Governmental Regulation.	75
10.3	Headings.	75
		76
10.5	Several Obligations; Benefits of this Agreement.	76
10.6	Expenses; Indemnification.	76
10.7	Numbers of Documents.	77
10.8	Accounting.	77
10.9	Severability of Provisions.	77
10.10	Nonliability of Lenders.	78
10.11	Limited Disclosure.	78
		79
	Nonreliance.	79
10.14	No Advisory or Fiduciary Responsibility.	79
	THE ADMINISTRATIVE AGENT	
	Appointment and Authority.	80
	Rights as a Lender.	80
	Exculpatory Provisions.	81
	Reliance by Administrative Agent.	82
	Delegation of Duties.	82
	Resignation of Administrative Agent.	82
	Non-Reliance on Administrative Agent and Other Lenders.	83
	No Other Duties, Etc.	83
	Administrative Agent May File Proofs of Claim.	84
	SETOFF; RATABLE PAYMENTS; PAYMENTS SET ASIDE	_
		84
	Ratable Payments.	85
	Payments Set Aside.	85
	BENEFIT OF AGREEMENT; ASSIGNMENTS; PARTICIPATIONS	
	Successors and Assigns.	86
	Replacement of Lenders.	90
ARTICLE XIV		
		91
		91
		91
	<u>COUNTERPARTS</u>	
	OTHER AGENTS	
	CHOICE OF LAW; CONSENT TO JURISDICTION; WAIVER OF JURY TRIAL.	
	CHOICE OF LAW.	92
	CONSENT TO JURISDICTION.	93
	WAIVER OF JURY TRIAL.	93
	TERMINATION OF EXISTING CREDIT FACILITY	_

iii

SCHEDULES

T	Commitments	and I offer of	Crodit	Commitment
1	Commitments	and Letter of	· Crean	Commitments

II **Existing Letters of Credit**

Pricing Schedule III

IV Certain Addresses for Notices

EXHIBITS

_	_		
Λ	Form of	Compliance	Cortificato
А	LOIIII OI	Computative	Ceruncate

- Form of Assignment and Assumption В C
 - Form of Wire Transfer Instructions
 - D Form of Revolving Note
 - E Form of Swing Line Note

CREDIT AGREEMENT

This Credit Agreement dated as of August 9, 2010 is among KCP&L Greater Missouri Operations Company, a Delaware corporation, Great Plains Energy Incorporated, a Missouri corporation, the Lenders, Union Bank, N.A. and Wells Fargo Bank, National Association, as Syndication Agents, and Bank of America, N.A., as Administrative Agent. The parties hereto agree as follows:

ARTICLE I

DEFINITIONS

1.1 <u>Definitions</u>.

As used in this Agreement, the following terms have the following meanings (such meanings to be equally applicable to both the singular and plural forms of such terms):

- "Additional Commitment Lender" is defined in Section 2.21(d).
- "Administrative Agent" means Bank of America in its capacity as administrative agent under any of the Loan Documents, or any successor administrative agent.
 - "Administrative Agent's Fee Letter" means certain fee letter dated June 2, 2010 among Bank of America, the Guarantor, the Borrower and KCPL.
- "<u>Administrative Agent's Office</u>" means the Administrative Agent's address and, as appropriate, account as set forth on <u>Schedule IV</u> or such other address or account as the Administrative Agent may from time to time notify to the Borrower and the Lenders.
 - "Administrative Questionnaire" means an Administrative Questionnaire in a form supplied by the Administrative Agent.
- "Advance" means a borrowing hereunder (or conversion or continuation thereof) consisting of the aggregate amount of the several Loans, including the Swing Line Loans, made on the same Borrowing Date (or date of conversion or continuation) by the Lenders to the Borrower of the same Type and, in the case of Eurodollar Advances, for the same Interest Period.
- "Affiliate" of any Person means any other Person directly or indirectly controlled by or under common control with such Person. For purposes of this definition, "control" (including, with correlative meanings, the terms "controlled by" and "under common control with") shall mean the possession, directly or indirectly, of the power to direct or cause the direction of the management or policies of a Person, whether through the ownership of voting securities or by contract or otherwise.
 - "Agent Parties" is defined in Section 14.3.

- "Agents" means, collectively, the Administrative Agent and the Syndication Agents, and "Agent" means any one of them.
- "Aggregate Commitment" means the aggregate of the Commitments of all Lenders, as changed from time to time pursuant to the terms hereof. The amount of the Aggregate Commitment in effect as of the Closing Date is FOUR HUNDRED AND FIFTY MILLION DOLLARS (\$450,000,000).
 - "Aggregate Outstanding Credit Exposure" means, at any time, the aggregate of the Outstanding Credit Exposure of all Lenders.
 - "Agreement" means this credit agreement, as it may be amended or modified and in effect from time to time.
- "Alternate Base Rate" means for any day a fluctuating rate per annum equal to the highest of (a) the Federal Funds Rate plus 1/2 of 1%, (b) the rate of interest in effect for such day as publicly announced from time to time by Bank of America as its "prime rate" and (c) Eurodollar Base Rate plus 1.0%. The "prime rate" is a rate set by Bank of America based upon various factors including Bank of America's costs and desired return, general economic conditions and other factors, and is used as a reference point for pricing some loans, which may be priced at, above, or below such announced rate. Any change in such prime rate announced by Bank of America shall take effect at the opening of business on the& #160;day specified in the public announcement of such change.
- "Applicable Margin" means, with respect to Advances of any Type at any time, the percentage rate per annum which is applicable at such time with respect to Advances of such Type as set forth in the Pricing Schedule.
- "Approved Fund" means any Fund that is administered or managed by (a) a Lender, (b) an Affiliate of a Lender or (c) an entity or an Affiliate of an entity that administers or manages a Lender.
 - "Approving Lenders" is defined in Section 2.21(e).
 - "Arrangers" means BAS, Union Bank and Wells Fargo Securities, LLC and "Arranger" means any one of them.
 - "Article" means an article of this Agreement unless another document is specifically referenced.
- "Assignment Agreement" means an assignment and assumption entered into by a Lender and an Eligible Assignee (with the consent of any party whose consent is required by Section 13.1(b)), and accepted by the Administrative Agent, in substantially the form of Exhibit B or any other form approved by the Administrative Agent.

"Attributable Indebtedness" means, on any date, (a) in respect of any Capitalized Lease Obligation of any Person, the capitalized amount thereof that would appear on a balance sheet of such Person prepared as of such date in accordance with GAAP, and (b) in respect of any Synthetic Lease Obligation, the capitalized amount of the remaining lease payments under the relevant lease that would appear on a balance sheet of such Person prepared as of such date in accordance with GAAP if such lease were accounted for as a Capitalized Lease.

"Authorized Officer" means any of the Chief Executive Officer, the President, any Vice President, the Chief Financial Officer or the Treasurer of the Borrower or the Guarantor, as applicable, in each case acting singly.

"Bank of America" means Bank of America, N.A. in its individual capacity and its successors.

"BAS" means Banc of America Securities LLC.

"Borrower" means KCP&L Greater Missouri Operations Company, a Delaware corporation, and its permitted successors and assigns.

"Borrower Materials" is defined in Section 7.1.

"Borrowing Date" means a date on which an Advance is made hereunder.

"Borrowing Notice" is defined in Section 2.8.

"BTMU" means The Bank of Tokyo-Mitsubishi UFJ, Ltd., in its individual capacity and its successors.

"Business Day" means any day other than a Saturday, Sunday or other day on which commercial banks are authorized to close under the Laws of, or are in fact closed in, New York City, New York and, if such day relates to any Eurodollar Loan or any Floating Rate Loan bearing interest at a rate based on the Eurodollar Rate, means any such day that is also a London Banking Day.

"Capitalized Lease" of a Person means any lease of Property by such Person as lessee which would be capitalized on a balance sheet of such Person prepared in accordance with GAAP.

"Capitalized Lease Obligations" of a Person means the amount of the obligations of such Person under Capitalized Leases which would be shown as a liability on a balance sheet of such Person prepared in accordance with GAAP.

"Cash Collateralize" means to pledge and deposit with or deliver to the Administrative Agent, for the benefit of the Administrative Agent, the Issuers or the Swing Line Lender (as applicable) and the Lenders, as collateral for Letter of Credit Obligations, Obligations in respect of Swing Line Loans, or obligations of Lenders to fund participations in respect of either thereof

3

(as the context may require), cash or deposit account balances or, if the applicable Issuer or the Swing Line Lender benefitting from such collateral shall agree in its sole discretion, other credit support, in each case pursuant to documentation in form and substance reasonably satisfactory to (a) the Administrative Agent and (b) the applicable Issuer or the Swing Line Lender (as applicable). "Cash Collateral" shall have a meaning correlative to the foregoing and shall include the proceeds of such cash collateral and other credit support.

"Change" is defined in Section 3.2.

"Change in Law" means the occurrence, after the date of this Agreement, of any of the following: (a) the adoption or taking effect of any law, rule, regulation or treaty, (b) any change in any law, rule, regulation or treaty or in the administration, interpretation or application thereof by any Governmental Authority or (c) the making or issuance of any request, guideline or directive (whether or not having the force of law) by any Governmental Authority.

"Change of Control" means an event or series of events by which:

- (i) any "person" or "group" (as such terms are used in Sections 13(d) and 14(d) of the Securities Exchange Act of 1934, but excluding any employee benefit plan of the Guarantor or its Subsidiaries, or any Person acting in its capacity as trustee, agent or other fiduciary or administrator of any such plan) becomes the "beneficial owner" (as defined in Rules 13d-3 and 13d-5 under the Securities Exchange Act of 1934), directly or indirectly, of thirty-three and one-third percent (33 1/3%) or more of the "voting equity interests" (meaning for this purpose the power under ordinary circumstances to vote for the election of members of the board of directors) of the Guarantor; or
- (ii) during any period of twelve (12) consecutive months (or such lesser period of time as shall have elapsed since the formation of the Guarantor), a majority of the members of the board of directors or other equivalent governing body of the Guarantor ceases to be composed of individuals (x) who were members of that board or equivalent governing body on the first day of such period, (y) whose election or nomination to that board or equivalent governing body was approved by individuals referred to in $\underline{\text{clause}(\underline{x})}$ above constituting at the time of such election or nomination at least a majority of that board or equivalent governing body or (z) whose election or nomination to that board or other equivalent governing body was approved by individuals referred to in $\underline{\text{clauses}(\underline{x})}$ and $\underline{(y)}$ above constituting at the time of such election or nomination at least a majority of that board or equivalent governing body.

"Closing Date" means August 9, 2010.

"Code" means the Internal Revenue Code of 1986, as amended, reformed or otherwise modified from time to time.

"Collateral Shortfall Amount" is defined in Section 9.1.

4

"Commitment" means, for each Lender, the obligation of such Lender to make Loans and to participate in Letters of Credit and Swing Line Loans in an aggregate amount not exceeding the amount set forth on Schedule I hereto or as set forth in any Assignment Agreement relating to any assignment that has become effective pursuant to Section 13.1(b), as such amount may be modified from time to time pursuant to the terms hereof.

"Commitment Fee Rate" means, at any time, the percentage rate per annum at which commitment fees are accruing at such time as set forth in the Pricing Schedule.

"Consolidated Guarantor Net Income" means, for any period, for the Guarantor and its Consolidated Subsidiaries, the net income of the Guarantor and its Consolidated Subsidiaries from continuing operations, excluding extraordinary items for that period.

"Consolidated Net Income" means, for any period, for the Borrower and its Consolidated Subsidiaries, the net income of the Borrower and its Consolidated Subsidiaries from continuing operations, excluding extraordinary items for that period.

"<u>Consolidated Subsidiaries</u>" means all Subsidiaries of the Borrower or the Guarantor, as applicable, that are (or should be) included when preparing the consolidated financial statements of the Borrower or the Guarantor, as applicable.

"Consolidated Tangible Net Worth" means, as of any date of determination, for the Borrower and its Consolidated Subsidiaries, Shareholders' Equity of the Borrower and its Consolidated Subsidiaries on that date minus the Intangible Assets of the Borrower and its Consolidated Subsidiaries on that date.

"Contingent Obligation" of a Person means any agreement, undertaking or arrangement by which such Person assumes, guarantees, endorses, contingently agrees to purchase or provide funds for the payment of, or otherwise becomes or is contingently liable upon, the obligation or liability of any other Person, or agrees to maintain the net worth or working capital or other financial condition of any other Person, or otherwise assures any creditor of such other Person against loss.

"Controlled Group" means all members of a controlled group of corporations or other business entities and all trades or businesses (whether or not incorporated) under common control which, together with the Borrower or the Guarantor, are treated as a single employer under Section 414 of the Code.

"Conversion/Continuation Notice" is defined in Section 2.9.

"Credit Extension" means the making of an Advance or the issuance of a Letter of Credit.

"<u>Debtor Relief Law</u>" means the Bankruptcy Code of the United States, and all other liquidation, conservatorship, bankruptcy, assignment for the benefit of creditors, moratorium, rearrangement, receivership, insolvency, reorganization, or similar debtor relief Laws of the

5

United States or other applicable jurisdictions from time to time in effect and affecting the rights of creditors generally.

"Default" means an event described in Article VIII.

"Defaulting Lender" means, subject to Section 2.23(b), any Lender, as reasonably determined by the Administrative Agent, that, at any time during the term of this Agreement, (a) has failed to perform any of its funding obligations hereunder, including in respect of its Loans or participations in respect of Letters of Credit or Swing Line Loans, within three (3) Business Days of the date required to be funded by it hereunder, unless such obligation is the subject of a good faith dispute, (b) has notified the Borrower, the Administrative Agent or any Lender in writing, or has otherwise indicated through a public statement, that it does not intend to comply with any of its funding obligations under this Agreement or generally under other agreements in which it commits to extend credit, (c) has failed, within five (5) Business Days after written request by the Administrative Agent (based on the reasonable belief that such Lender may not fulfill any of its funding obligations hereunder or after any reasonable request by the Borrower to do so), to confirm in a manner reasonably satisfactory to the Administrative Agent and the Borrower that it will comply with its funding obligations, unless such obligation is the subject of a good faith dispute, provided that any such Lender shall cease to be a Defaulting Lender under this clause (c) upon receipt of such confirmation by the Administrative Agent in a manner reasonably satisfactory to the Administrative Agent, (d) otherwise failed to pay over to the Administrative Agent or any other Lender any other amount (other than a de minimis amount) required to be paid by it hereunder within three (3) Business Days of the date when due, unless the subject of a good-faith dispute, or (e) has, or has a direct or indirect parent company that has, (i) become the subject of a proceeding under any Debtor Relief Law, (ii) had a receiver, conservator, trustee, administrator, assignee for the benefit of creditors or similar Person charged with reorganization or liquidation of its business or a custodian appointed for it or for any substantial part of its assets, or (iii) taken any action in furtherance of, or indicated its consent to, approval of or acquiescence in any such proceeding or appointment; provided that a Lender shall not be a Defaulting Lender solely by virtue of the ownership or acquisition of any equity interest in that Lender or any direct or indirect parent company thereof, or the exercise of control over such Lender or any direct or indirect parent company thereof controlling such Lender, by a Governmental Authority.

"Eligible Assignee" means (a) a Lender; (b) an Affiliate of a Lender; (c) an Approved Fund; and (d) any other Person (other than a natural person) approved by (i) the Administrative Agent, the Swing Line Lender and the Issuers, and (ii) unless a Default has occurred and is continuing, the Borrower (each such approval not to be unreasonably withheld or delayed); provided that notwithstanding the foregoing, "Eligible Assignee" shall not include the Borrower or the Guarantor or any of the Borrower's or the Guarantor's Affiliates or Subsidiaries.

"Environmental Laws" means any and all federal, state, local and foreign statutes, laws, judicial decisions, regulations, ordinances, rules, judgments, orders, decrees, plans, injunctions, permits, concessions, grants, franchises, licenses, agreements and other governmental restrictions

6

relating to (a) the protection of the environment, (b) the effect of the environment on human health, (c) emissions, discharges or releases of pollutants, contaminants, hazardous substances or wastes into surface water, ground water or land, or (d) the manufacture, processing, distribution, use, treatment, storage, disposal, transport or handling of pollutants, contaminants, hazardous substances or wastes or the clean-up or other remediation thereof.

"Equity-Linked Securities" means (a) all securities issued by the Guarantor (including without limitation the May 2009 issuance of 5.75 million equity units by the Guarantor) or any Subsidiary that contain two distinct components: (i) medium-term debt and (ii) a forward contract for the issuance of common stock of the Guarantor or such Subsidiary prior to the maturity of, and in an amount not less than, such debt, including the securities commonly referred to by the tradenames "FELINE PRIDES", "PEPS", "HITS" and "DECS" and generally referred to as "equity units"; provided that such securities shall not contain any provision permitting them to be put to the Guarantor or any Subsidiary prior to the settlement of the related purchase contract and (b) all other securities issued by the Guarantor or any Subsidiary that are similar to those described in clause (a).

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended from time to time, and any rule or regulation issued thereunder.

"Eurodollar Advance" means an Advance which bears interest at the applicable Eurodollar Rate.

"Eurodollar Rate" means:

(a) for any Interest Period with respect to a Eurodollar Loan, a rate per annum (rounded to the nearest multiple of 1/16 of 1%) determined by the Administrative Agent pursuant to the following formula:

Eurodollar Rate = $\frac{Eurodollar Base Rate}{1.00 - Eurodollar Reserve Percentage}$

and (b) for any day with respect to a Floating Rate Loan bearing interest at a rate based on the Eurodollar Base Rate, a rate per annum (rounded to the nearest multiple of 1/16 of 1%) determined by the Administrative Agent pursuant to the following formula:

 $Eurodollar\ Rate \ = \underbrace{\frac{Eurodollar\ Base\ Rate}{1.00-Eurodollar\ Reserve\ Percentage}}.$

"Eurodollar Base Rate" means:

(a) for such Interest Period with respect to any Eurodollar Loan, the rate per annum equal to (i) the British Bankers Association LIBOR Rate ("BBA LIBOR"), as published by Reuters (or such other commercially available source providing quotations of BBA LIBOR as may be designated by the Administrative Agent from time to time) at approximately 11:00 a.m., London time, two (2) London Banking Days prior to the commencement of such Interest Period, for

7

Dollar deposits (for delivery on the first day of such Interest Period) with a term equivalent to such Interest Period or (ii) if such published rate is not available at such time for any reason, the rate determined by the Administrative Agent to be the rate at which deposits in Dollars for delivery on the first day of such Interest Period in same day funds in the approximate amount of the Eurodollar Loan being made, continued or converted and with a term equivalent to such Interest Period would be offered by Bank of America's London Branch to major banks in the London interbank eurodollar market at their request at approximately 11:00 a.m. (London time) two (2) London Banking Days prior to the commencement of such Interest Period and

(b) for any interest rate calculation with respect to any Floating Rate Loan, the rate per annum equal to (i) BBA LIBOR, at approximately 11:00 a.m., London time determined two (2) London Banking Days prior to such date for Dollar deposits being delivered in the London interbank market for a term of one (1) month commencing that day or (ii) if such published rate is not available at such time for any reason, the rate per annum determined by the Administrative Agent to be the rate at which deposits in Dollars for delivery on the date of determination in same day funds in the approximate amount of the Floating Rate Loan being made or maintained and with a term equal to one (1) month would be offered by Bank of America's London Branch to major banks in the London interbank Eurodollar market at their request at the date and time of det ermination.

"Eurodollar Reserve Percentage" means the reserve percentage (expressed as a decimal, carried out to five decimal places) in effect on such day, whether or not applicable to any Lender, under regulations issued from time to time by the FRB for determining the maximum reserve requirement (including any emergency, supplemental or other marginal reserve requirement) with respect to Eurocurrency funding (currently referred to as "Eurocurrency liabilities"). The Eurodollar Rate for each outstanding Eurodollar Loan or Floating Rate Loan bearing interest at a rate based on the Eurodollar Rate shall be adjusted automatically as of the effective date of any change in the Eurodollar Reserve Percentage.

"Eurodollar Loan" means a Loan which bears interest at the applicable Eurodollar Rate.

"Excluded Taxes" means, with respect to the Administrative Agent, any Lender, or any other recipient of any payment to be made by or on account of any Obligation of the Borrower hereunder or under any other Loan Document, (a) taxes imposed on or measured by its overall net income (however denominated), and franchise taxes imposed on it (in lieu of net income taxes), by the jurisdiction (or any political subdivision thereof) under the laws of which such recipient is organized or in which its principal office is located or, in the case of any Lender, in which its applicable Lending Installation is located, (b) any branch profits taxes imposed by the United States or any similar tax imposed by any other jurisdiction, (c) any backup withholding tax that is required by the Code to be withheld from amounts payable to a Lender, (d) in the case of a Foreign Lender (other than an assignee pursuant to a request by the Borrower under Section 13.2), any United States withholding tax that (i) is imposed on amounts payable to such Foreign Lender pursuant to the laws in force at the time such Foreign Lender becomes a party hereto (or designates a new Lending Installation) or (ii) is attributable to such Foreign Lender's failure or inability (other than solely as a result of a Change in Law occurring after such Foreign Lender becomes a party to this Agreement or designates a new Lending Installation) to comply with

8

<u>Section 3.5(e)</u>, except to the extent that such Foreign Lender (or its assignor, if any) was entitled, immediately prior to the time of designation of a new Lending Installation (or assignment), to receive additional amounts from the Borrower with respect to such withholding tax pursuant to <u>Section 3.5(a)(ii)</u>, and (e) any taxes imposed with respect to the requirements of FATCA.

"Exhibit" refers to an exhibit to this Agreement, unless another document is specifically referenced.

"Existing Credit Facility" means the credit agreement among the Borrower, the Guarantor, Bank of America, N.A., as administrative agent and the other lenders party thereto, dated as of September 23, 2008, as amended or modified from time to time.

"Existing Letters of Credit" means those letters of credit identified on Schedule II.

"Facility Termination Date" means (a) the later of (i) August 9, 2013 and (ii) with respect to some or all of the Lenders if the facility termination date is extended pursuant to Section 2.21, such extended facility termination date or (b) any earlier date on which the Aggregate Commitment is reduced to zero or otherwise terminated pursuant to the terms hereof.

"FATCA" means sections 1471 through 1474 of the Code and any regulations that are issued thereunder or official governmental interpretations thereof.

"Federal Funds Rate" means, for any day, the rate per annum equal to the weighted average of the rates on overnight Federal funds transactions with members of the Federal Reserve System arranged by Federal funds brokers on such day, as published by the Federal Reserve Bank of New York on the Business Day next succeeding such day; provided that (a) if such day is not a Business Day, the Federal Funds Rate for such day shall be such rate on such transactions on the next preceding Business Day as so published on the next succeeding Business Day, and (b) if no such rate is so published on such next succeeding Business Day, the Federal Funds Rate for such day shall be the average rate (rounded upward, i f necessary, to a whole multiple of 1/100 of 1%) charged to Bank of America on such day on such transactions as determined by the Administrative Agent.

"Fee Letter" means that certain fee letter dated June 2, 2010 among the Agents, the Arrangers, the Guarantor, the Borrower and KCPL.

"Floating Rate Advance" means an Advance which bears interest at the Alternate Base Rate.

"Floating Rate Loan" means a Loan which bears interest at the Alternate Base Rate.

"Foreign Lender" means any Lender that is not a "United States person" within the meaning of Section 7701(a)(30) of the Code.

"FRB" means the Board of Governors of the Federal Reserve System.

9

"Fronting Exposure" means, at any time there is a Defaulting Lender, (a) with respect to an Issuer, such Defaulting Lender's Pro Rata Share of the outstanding Letter of Credit Obligations other than Letter of Credit Obligations as to which such Defaulting Lender's participation obligation has been reallocated to other Lenders or Cash Collateralized in accordance with the terms hereof, and (b) with respect to the Swing Line Lender, such Defaulting Lender's Pro Rata Share of Swing Line Loans other than Swing Line Loans as to which such Defaulting Lender's participation obligation has been reallocated to other Lenders or Cash Collateralized in accordance with the terms hereof.

"Fund" means any Person (other than a natural person) that is (or will be) engaged in making, purchasing, holding or otherwise investing in commercial loans and similar extensions of credit in the ordinary course of its activities.

"GAAP" means generally accepted accounting principles set forth from time to time in the opinions and pronouncements of the Accounting Principles Board and the American Institute of Certified Public Accountants and statements of the Financial Accounting Standards Board.

"GMO-GPE Re-Transfer" is defined in Section 2.6(c)(ii).

"GMO-GPE Transfer" is defined in Section 2.6(c)(i).

"Governmental Authority" means the government of the United States or any other nation, or of any political subdivision thereof, whether state or local, and any agency, authority, instrumentality, regulatory body, court, central bank or other entity exercising executive, legislative, judicial, taxing, regulatory or administrative powers or functions of or pertaining to government.

"GPE-GMO Re-Transfer" is defined in Section 2.6(d)(ii).

"GPE-GMO Transfer" is defined in Section 2.6(d)(i).

"Great Plains Credit Agreement" means that certain Credit Agreement dated as of the Closing Date among the Guarantor, the financial institutions party thereto, Union Bank and Wells Fargo, as syndication agents and Bank of America, N.A., as administrative agent, as amended or modified from time to time.

"Guarantor" means Great Plains Energy Incorporated, a Missouri corporation.

"Guarantor Shareholders' Equity" means, as of any date of determination for the Guarantor and its Consolidated Subsidiaries on a consolidated basis, shareholders' equity as of that date determined in accordance with GAAP.

"Honor Date" is defined in Section 2.19(f).

"including" means "including without limiting the generality of the following".

10

"Indebtedness" means, as to any Person at a particular time, all of the following, without duplication, to the extent recourse may be had to the assets or properties of such Person in respect thereof: (a) all obligations of such Person for borrowed money and all obligations of such Person evidenced by bonds, debentures, notes, loan agreements or other similar instruments; (b) any direct or contingent obligations of such Person in the aggregate in excess of \$2,000,000 arising under letters of credit (including standby and commercial), banker's acceptances, bank guaranties, surety bonds and similar instruments; (c) net obligations of such Person under Swap Contracts; (d) all obligations of such Person to pay the deferred purchase price of property or service s (except trade accounts payable arising, and accrued expenses incurred, in the ordinary course of business), and indebtedness (excluding prepaid interest thereon) secured by a Lien on property owned or being purchased by such Person (including indebtedness arising under conditional sales or other title retention agreements), whether or not such indebtedness shall have been assumed by such Person or is limited in recourse; (e) Capitalized Lease Obligations and Synthetic Lease Obligations of such Person; and (f) all Contingent Obligations of such Person in respect of any of the foregoing.

For all purposes hereof, the Indebtedness of any Person shall include the Indebtedness of any partnership or joint venture in which such Person is a general partner or a joint venturer, unless such Indebtedness is non-recourse to such Person. It is understood and agreed that Indebtedness (including Contingent Obligations) shall not include any obligations of any Person with respect to (i) subordinated, deferrable interest debt securities, and any related securities issued by a trust or other special purpose entity in connection therewith, or any similar securities that are classified at the time of issuance, as possessing a minimum of "intermediate equity content" by S&P and "Basket C equity credit" by Moody's (or the equivalent classification then in effect by such agencies), as long as the m aturity date of such debt is subsequent to the Facility Termination Date; provided that the amount of mandatory principal amortization or defeasance of such debt prior to the Facility Termination Date shall be included in this definition of Indebtedness; (ii) Equity-Linked Securities until the mandatory redemption date therefor, provided that the principal amount of all outstanding Equity-Linked Securities in excess of twenty percent (20%) of Total Guarantor Capitalization shall constitute Indebtedness of the Guarantor; or (iii) utility "rate reduction" bonds, for the payment of which legislatively authorized charges are imposed on customers. The amount of any Capitalized Lease Obligation or Synthetic Lease Obligation as of any date shall be deemed to be the amount of Attributable Indebtedness in respect thereof as of such date.

"Indemnified Taxes" means Taxes other than Excluded Taxes.

"Intangible Assets" means, assets that are considered to be intangible assets under GAAP, including, but not limited to, customer lists, goodwill, computer software, copyrights, trade names, trademarks, patents, franchises and licenses.

"Interest Period" means, with respect to a Eurodollar Advance, a period of one (1), two (2), three (3) or six (6) months commencing on a Business Day selected by the Borrower pursuant to this Agreement. Such Interest Period shall end on the day which corresponds numerically to such date one (1), two (2), three (3) or six (6) months thereafter; provided that if there is no such numerically corresponding day in such next, second, third or sixth succeeding

11

month, such Interest Period shall end on the last Business Day of such next, second, third or sixth succeeding month. If an Interest Period would otherwise end on a day which is not a Business Day, such Interest Period shall end on the next succeeding Business Day; <u>provided</u> that if said next succeeding Business Day falls in a new calendar month, such Interest Period shall end on the immediately preceding Business Day.

"ISP" means, with respect to any Letter of Credit, the "International Standby Practices 1998" published by the Institute of International Banking Law & Practice, Inc. (or such later version thereof as may be in effect at the time of issuance of such Letter of Credit).

"Issuer" means each of Bank of America, Wells Fargo, BTMU and any other Lender selected by the Borrower with the approval of the Administrative Agent (except with respect to any selection for which such approval is not required pursuant to Section 2.19(n)), in each case, in its capacity as an issuer of Letters of Credit hereunder and includes, without limitation, any Lender appointed by the Borrower (with the consent of the Administrative Agent, except with respect to any selection for which such approval is not required pursuant to Section 2.19(n)) as such by notice to the Lenders as a

replacement for any Issuer who at the time of su ch appointment is a Defaulting Lender. For the avoidance of doubt, any Lender (other than Bank of America, Wells Fargo and BTMU) approached to become an issuer of Letters of Credit may elect or decline, in its sole discretion, to become an issuer.

"Issuer Documents" means with respect to any Letter of Credit, any Letter of Credit Application and any other document, agreement and instrument entered into by the applicable Issuer and the Borrower or in favor of the applicable Issuer and relating to such Letter of Credit.

"KCPL" means Kansas City Power & Light Company, a Missouri corporation.

"KCPL Credit Agreement" means that certain Credit Agreement dated as of the Closing Date among KCPL, the financial institutions party thereto, Union Bank and Wells Fargo, as syndication agents and Bank of America, N.A., as administrative agent, as amended or modified from time to time.

"<u>Laws</u>" means, collectively, all international, foreign, federal, state and local statutes, treaties, rules, guidelines, regulations, ordinances, codes and administrative or judicial precedents or authorities, including the interpretation or administration thereof by any Governmental Authority charged with the enforcement, interpretation or administration thereof, and all applicable administrative orders, directed duties, requests, licenses, authorizations and permits of, and agreements with, any Governmental Authority, in each case as in effect from time to time.

"LC Collateral Accounts" is defined in Section 2.22(b).

"<u>Lenders</u>" means the lending institutions listed on the signature pages of this Agreement and their respective successors and assigns, and any other Person that shall become a party hereto pursuant to an Assignment Agreement or a joinder agreement provided in <u>Section 2.6(b)(iv)</u>, and as the context requires, includes the Swing Line Lender.

12

"<u>Lending Installation</u>" means, with respect to a Lender or the Administrative Agent, the office, branch, subsidiary or affiliate of such Lender or the Administrative Agent listed on the signature pages hereof or on a Schedule or otherwise selected by such Lender or the Administrative Agent pursuant to <u>Section 2.17</u>.

"Letter of Credit" means any standby letter of credit issued pursuant to Section 2.19 and any Existing Letter of Credit.

"Letter of Credit Application" is defined in Section 2.19(c).

"<u>Letter of Credit Commitment</u>" means, for each Issuer, the obligation of such Issuer to issue Letters of Credit in an aggregate amount not exceeding the amount set forth on <u>Schedule I</u> hereto or as set forth in any commitment agreement entered into by any Lender appointed as an Issuer by the Borrower in accordance with the terms hereof.

"Letter of Credit Fee" is defined in Section 2.19(d).

"Letter of Credit Fee Rate" means, at any time, the percentage rate per annum applicable to Letter of Credit Fees at such time as set forth in the Pricing Schedule.

"<u>Letter of Credit Obligations</u>" means, at any time, the sum, without duplication, of (i) the aggregate undrawn stated amount of all outstanding Letters of Credit at such time plus (ii) the aggregate unpaid amount of all Reimbursement Obligations at such time. For purposes of computing the amount available to be drawn under any Letter of Credit, the amount of such Letter of Credit shall be determined in accordance with <u>Section 1.3</u>. For all purposes of this Agreement, if on any date of determination a Letter of Credit has expired by its terms but any amount may still be drawn thereunder by reason of the operation of Rule 3.14 of the ISP, such Letter of Credit shall be deemed to be "outstanding" in the amount so remaining available to be drawn.

"Letter of Credit Payment Date" is defined in Section 2.19(e).

"<u>Letter of Credit Sublimit</u>" means an amount equal to the lesser of (a) \$50,000,000 and (b) the Aggregate Commitments. The Letter of Credit Sublimit is part of, and not in addition to, the Aggregate Commitments.

"<u>Lien</u>" means any lien (statutory or other), mortgage, pledge, hypothecation, assignment, deposit arrangement, encumbrance or preference, priority or other security agreement or preferential arrangement of any kind or nature whatsoever (including the interest of a vendor or lessor under any conditional sale, Capitalized Lease or other title retention agreement).

"Loan" means, with respect to a Lender, such Lender's loans made pursuant to <u>Article II</u> (or any conversion or continuation thereof), including the Swing Line Loans.

13

"<u>Loan Documents</u>" means this Agreement, each Note, any agreement creating or perfecting rights in Cash Collateral pursuant to the provisions of <u>Section 2.22</u>, each Letter of Credit, each Issuer Document, the Administrative Agent's Fee Letter and the Fee Letter.

"London Banking Day." means any day on which dealings in Dollar deposits are conducted by and between banks in the London interbank eurodollar market.

"Material Adverse Effect" means a material adverse effect on (i) the business, Property, financial condition, or results of operations of the Borrower and its Subsidiaries taken as a whole, (ii) the business, Property, financial condition, or results of operations of the Guarantor and its Subsidiaries taken as a whole, (iii) the ability of the Borrower or the Guarantor to perform their respective obligations under the Loan Documents or (iv) the validity or enforceability of any of the Loan Documents or the rights or remedies of the Agents, the Lenders or the Issuers thereunder.

"Material Indebtedness" is defined in Section 8.6.

"Modification" and "Modify" are defined in Section 2.19(a).

"Moody's" means Moody's Investors Service, Inc and any successor thereto.

"Multiemployer Plan" means "multiemployer plan" as defined in Section 3(37) of ERISA to which the Borrower or the Guarantor or any member of the Controlled Group is obligated to make contributions.

"Non-Extending Lender" is defined in Section 2.21(b).

"Note" or "Notes" means the Revolving Notes and/or the Swing Line Note, as applicable.

"Notice Date" is defined in Section 2.21(b).

"Obligations" means all unpaid principal of and accrued and unpaid interest on the Loans, all Reimbursement Obligations and accrued and unpaid interest thereon, all accrued and unpaid fees and all expenses, reimbursements, indemnities and other obligations of the Borrower or the Guarantor to any Lender, any Issuer, either Agent or any indemnified party arising under any Loan Document, including interest and fees after commencement of, by or against the Borrower or the Guarantor of any proceeding under any the federal bankruptcy laws or any comparable provisions of any applicable state law naming such Person as the debtor in such proceeding, regardless of whether such interest and fees are allowed claims in such proceeding. The foregoing shall also include all obligations under any Swap Contract between the Borrower and any Lender or Affiliate of a Lender that is permitted to be incurred hereunder.

"Organization Documents" means, (a) with respect to any corporation, the certificate or articles of incorporation and the bylaws (or equivalent or comparable constitutive documents with respect to any non-U.S. jurisdiction); (b) with respect to any limited liability company, the certificate or articles of formation or organization and operating agreement; and (c) with respect

14

to any partnership, joint venture, trust or other form of business entity, the partnership, joint venture or other applicable agreement of formation or organization and any agreement, instrument, filing or notice with respect thereto filed in connection with its formation or organization with the applicable Governmental Authority in the jurisdiction of its formation or organization and, if applicable, any certificate or articles of formation or organization of such entity.

"Other Taxes" means all present or future stamp or documentary taxes or any other excise or property taxes, charges or similar levies (other than Excluded Taxes) arising from any payment made hereunder or under any other Loan Document or from the execution, delivery or enforcement of, or otherwise with respect to, this Agreement or any other Loan Document.

"Outstanding Credit Exposure" means, as to any Lender at any time, the sum of (a) the aggregate principal amount of its Loans outstanding at such time, plus (b) its Pro Rata Share of the Letter of Credit Obligations at such time, plus (c) its Pro Rata Share of Swing Line Loans at such time.

"Participant" is defined in Section 13.1(d).

"Payment Date" means the last Business Day of each March, June, September and December.

"PBGC" means the Pension Benefit Guaranty Corporation, or any successor thereto.

"<u>Person</u>" means any natural person, corporation, firm, joint venture, partnership, limited liability company, association, enterprise, trust or other entity or organization, or any government or political subdivision or any agency, department or instrumentality thereof.

"<u>Plan</u>" means an employee pension benefit plan which is covered by Title IV of ERISA or subject to the minimum funding standards under Section 412 or Section 430 of the Code as to which the Borrower or any member of the Controlled Group may have any liability.

"Platform" is defined in Section 7.1.

"Pricing Schedule" means Schedule III attached hereto identified as such.

"<u>Project Finance Subsidiary</u>" means any Subsidiary that meets the following requirements: (a) it is primarily engaged, directly or indirectly, in the ownership, operation and/or financing of independent power production and related facilities and assets; and (b) neither the Guarantor, the Borrower nor any other Subsidiary (other than another Project Finance Subsidiary) has any liability, contingent or otherwise, for the Indebtedness or other obligations of such Subsidiary (other than non-recourse liability resulting from the pledge of stock of such Subsidiary).

"Property" of a Person means any and all property, whether real, personal, tangible, intangible, or mixed, of such Person, or other assets owned or leased by such Person.

"Pro Rata Share" means, with respect to any Lender on any date of determination, the percentage which the amount of such Lender's Commitment is of the Aggregate Commitment (or, if the Commitments have terminated, which such Lender's Outstanding Credit Exposure is of the Aggregate Outstanding Credit Exposure) as of such date, subject to adjustment as provided in Section 2.23. For purposes of determining liability for any indemnity obligation under Section 10.6(c), each Lender's Pro Rata Share shall be determined as of the date the applicable Issuer, the Swing Line Lender or the Administrative Agent notifies the Lenders of such indemnity obligation (or, if such notice is given after termination of this Agreement, as of the date of such termination).

"Public Lender" is defined in Section 7.1.

"Register" is defined in Section 13.1(c).

"Regulation U" means Regulation U of the FRB as from time to time in effect and any successor or other regulation or official interpretation of the FRB relating to the extension of credit by banks for the purpose of purchasing or carrying margin stocks applicable to member banks of the Federal Reserve System.

"Reimbursement Obligations" means, at any time, the aggregate of all obligations of the Borrower then outstanding under Section 2.19 to reimburse (i) the Issuers for amounts paid by the Issuers in respect of any one or more drawings under Letters of Credit and/or (ii) the Lenders for amounts paid by the Lenders to reimburse the Issuers pursuant to Section 2.19.

"Related Parties" means, with respect to any Person, such Person's Affiliates and the partners, directors, officers, employees, agents, trustees and advisors of such Person and of such Person's Affiliates.

"Reportable Event" means a reportable event as defined in Section 4043 of ERISA and the regulations issued under such section, with respect to a Plan, excluding, however, such events as to which the PBGC has by regulation waived the requirement of Section 4043(a) of ERISA that it be notified within thirty (30) days of the occurrence of such event; provided that a failure to meet the minimum funding standard of Section 412 or Section 430 of the Code and of Section 302 of ERISA shall be a Reportable Event regardless of the issuance of any such waiver of the notice requirement in accordance with either Section 4043(a) of ERISA or Section 412(d) of the Code.

"Required Lenders" means Lenders in the aggregate having more than fifty percent (50%) of the Aggregate Commitment or, if the Aggregate Commitment has been terminated, Lenders in the aggregate holding more than fifty percent (50%) of the Aggregate Outstanding Credit Exposure. The unfunded Commitments of, and the outstanding Loans, Letter of Credit Obligations and participations therein held or deemed held by, any Defaulting Lender shall be excluded for purposes of making a determination of Required Lenders.

"Revolving Note" is defined in Section 2.13(d).

16

"S&P" means Standard and Poor's Ratings Services, a division of The McGraw Hill Companies, Inc and any successor thereto.

"Schedule" refers to a specific schedule to this Agreement, unless another document is specifically referenced.

"SEC" means the Securities and Exchange Commission or any other U.S. federal governmental authority succeeding to any or all of the functions of the Securities and Exchange Commission.

"Section" means a numbered section of this Agreement, unless another document is specifically referenced.

"Shareholders' Equity" means, as of any date of determination for the Borrower and its Consolidated Subsidiaries on a consolidated basis, shareholders' equity as of that date determined in accordance with GAAP.

"Significant Subsidiary" means, at any time, the Borrower and each other Subsidiary which (a) as of the date of determination, owns consolidated assets equal to or greater than fifteen percent (15%) of the consolidated assets of the Guarantor and its Subsidiaries or (b) which had consolidated net income from continuing operations (excluding extraordinary items) during the four (4) most recently ended fiscal quarters equal to or greater than fifteen percent (15%) of Consolidated Guarantor Net Income (excluding extraordinary items) during such period.

"Single Employer Plan" means a Plan, other than a Multiemployer Plan, maintained by the Guarantor or any member of the Controlled Group for employees of the Guarantor or any member of the Controlled Group.

"Specified Information" is defined in Section 10.11(a).

"Subsidiary" of a Person means (a) any corporation more than fifty percent (50%) of the outstanding securities having ordinary voting power of which shall at the time be owned or controlled, directly or indirectly, by such Person or by one or more of its Subsidiaries or by such Person and one or more of its Subsidiaries, (b) any partnership, limited liability company, association, joint venture or similar business organization more than fifty percent (50%) of the ownership interests having ordinary voting power of which shall at the time be so owned or controlled; or (c) any other Person the operations and/or financial results of which are required to be consolidated with those of such first Person in accordance with GAAP. Unless otherwise expressly s tated, all references herein to a "Subsidiary" shall mean a Subsidiary of the Guarantor.

"Substantial Portion" means, with respect to the Property of the Guarantor and its Subsidiaries, Property which represents more than fifteen percent (15%) of the consolidated assets of the Guarantor and its Consolidated Subsidiaries as would be shown in the consolidated financial statements of the Guarantor and its Consolidated Subsidiaries as at the beginning of the

17

twelve-month period ending with the month in which such determination is made, or of the Consolidated Guarantor Net Income (excluding extraordinary items) of the Guarantor and its Consolidated Subsidiaries as reflected in the financial statements referred to above.

"Swap Contract" means (a) any and all rate swap transactions, basis swaps, credit derivative transactions, forward rate transactions, commodity swaps, commodity options, forward commodity contracts, equity or equity index swaps or options, bond or bond price or bond index swaps or options or forward bond or forward bond price or forward bond index transactions, interest rate options, forward foreign exchange transactions, cap transaction, floor transactions, culrency swap transactions, cross-currency rate swap transactions, currency options, spot contracts, or any other similar transactions or any combination of any of the foregoing (including any options to enter into any of the foregoing), whether or not any such transaction is governed by or subject to any master agreement, and (b) any and all transactions of any kind, and the related confirmations, which are subject to the terms and conditions of, or governed by, any form of master agreement published by the International Swaps and Derivatives Association, Inc., any International Foreign Exchange Master Agreement, or any other master agreement (any such master agreement, together with any related schedules, a "Master Agreement"), including any such obligations or liabilities under any Master Agreement.

"Swing Line Lender" means Bank of America in its capacity as provider of Swing Line Loans, or any successor swing line lender hereunder.

"Swing Line Loan" is defined in Section 2.20(a).

"Swing Line Note" is defined in Section 2.13(d).

"Swing Line Loan Notice" is defined in Section 2.20(b).

"Swing Line Sublimit" means an amount equal to the lesser of (a) \$25,000,000 and (b) the Aggregate Commitments. The Swing Line Sublimit is part of, and not in addition to, the Aggregate Commitments.

"Syndication Agents" means Union Bank and Wells Fargo, each in its capacity as syndication agent hereunder, and not in its individual capacity as a Lender, and any successor thereto.

"Synthetic Lease Obligation" means the monetary obligation of a Person under (a) a so-called synthetic or tax retention lease, pursuant to which notwithstanding the off-balance sheet treatment of the lease obligation the assets are deemed owned by the lessee for U.S. federal income tax purposes, or (b) an agreement for the use or possession of property creating obligations that do not appear on the balance sheet of such Person but which, upon the insolvency or bankruptcy of such Person, would be characterized as the indebtedness of such Person (without regard to accounting treatment).

18

"<u>Taxes</u>" means all present or future taxes, levies, imposts, duties, deductions, withholdings (including backup withholding), assessments, fees or other charges imposed by any Governmental Authority, including any interest, additions to tax or penalties applicable thereto.

"<u>'34 Act Reports</u>" means the periodic reports of the Borrower or the Guarantor filed with the SEC on Forms 10-K, 10-Q and 8-K (or any successor forms thereto).

"<u>Total Capitalization</u>" means Total Indebtedness of the Borrower and its Consolidated Subsidiaries plus the sum of (a) Shareholders' Equity (without giving effect to the application of ASC Topic 815) and (b) to the extent not otherwise included in Total Indebtedness or Shareholders' Equity, preferred and preference stock and securities of the Borrower and its Subsidiaries included in a consolidated balance sheet of the Borrower and its Consolidated Subsidiaries in accordance with GAAP.

"Total Guarantor Capitalization" means Total Guarantor Indebtedness of the Guarantor and its Consolidated Subsidiaries plus the sum of (a) Guarantor Shareholders' Equity (without giving effect to the application of ASC Topic 815) and (b) to the extent not otherwise included in Total Guarantor Indebtedness or Guarantor Shareholders' Equity, preferred and preference stock and securities of the Guarantor and its Subsidiaries included in a consolidated balance sheet of the Guarantor and its Consolidated Subsidiaries in accordance with GAAP.

"<u>Total Guarantor Indebtedness</u>" means all Indebtedness of the Guarantor and its Consolidated Subsidiaries on a consolidated basis (and without duplication) but without giving effect to the application of ASC Topic 860 with respect to transfers of accounts receivable by KCPL, the Borrower or one or more of their respective Subsidiaries to a non-Subsidiary, excluding (a) Indebtedness arising under Swap Contracts entered into in the ordinary course of business to hedge bona fide transactions and business risks and not for speculation, (b) Indebtedness of Project Finance Subsidiaries, and (c) Indebtedness of KLT Investments Inc. incurred in connection with the acquisition and maintenance of its interests (whether direct or indirect) in low income housing projects.

"<u>Total Indebtedness</u>" means all Indebtedness of the Borrower and its Consolidated Subsidiaries on a consolidated basis (and without duplication) but without giving effect to the application of ASC Topic 860 with respect to transfers of accounts receivable by the Borrower or one or more of its Subsidiaries to a non-Subsidiary, excluding Indebtedness arising under Swap Contracts entered into in the ordinary course of business to hedge bona fide transactions and business risks and not for speculation.

"Type" means, with respect to any Advance, its nature as a Floating Rate Advance or a Eurodollar Advance.

"Union Bank" means Union Bank, N.A. in its individual capacity and its successors.

"Unmatured Default" means an event which but for the lapse of time or the giving of notice, or both, would constitute a Default.

19

"Unreimbursed Amount" is defined in Section 2.19(f).

"Wells Fargo" means Wells Fargo Bank, National Association in its individual capacity and its successors.

"Wholly-Owned Subsidiary." of a Person means (i) any Subsidiary all of the outstanding voting securities of which (except directors' qualifying shares) shall at the time be owned or controlled, directly or indirectly, by such Person or one or more Wholly-Owned Subsidiaries of such Person, or by such Person and one or more Wholly-Owned Subsidiaries of such Person, or (ii) any partnership, limited liability company, association, joint venture or similar business organization one hundred percent (100%) of the ownership interests having ordinary voting power of which (except directors' qualifying shares) shall at the time be so owned or controlled.

1.2 Accounting Principles.

Unless the context otherwise clearly requires, all accounting terms not expressly defined herein shall be construed, and all financial computations required under this Agreement shall be made, in accordance with GAAP, consistently applied; provided that if the Borrower notifies the Administrative Agent that the Borrower wishes to amend any covenant in Article VII to eliminate the effect of any change in GAAP on the operation of such covenant (or if the Administrative Agent notifies the Borrower that the Required Lenders wish to amend any covenant in Article VII for such purpose), then the Borrower's compliance with such covenant shall be de termined on the basis of GAAP in effect immediately before the relevant change in GAAP became effective, until either such notice is withdrawn or such covenant is amended in a manner satisfactory to the Borrower and the Required Lenders.

1.3 Letter of Credit Amounts.

Unless otherwise specified herein, the amount of a Letter of Credit at any time shall be deemed to be the stated amount of such Letter of Credit in effect at such time; <u>provided</u> that with respect to any Letter of Credit that, by its terms or the terms of any Issuer Document related thereto, provides for one or more automatic increases in the stated amount thereof, the amount of such Letter of Credit shall be deemed to be the maximum stated amount of such Letter of Credit after giving effect to all such increases, whether or not such maximum stated amount is in effect at such time.

1.4 <u>Times of Day</u>.

Unless otherwise specified, all references herein to times of day shall be references to Eastern time (daylight or standard, as applicable).

20

ARTICLE II

THE CREDITS

2.1 Commitment.

From and including the date of this Agreement and prior to the Facility Termination Date, subject to the terms and conditions set forth in this Agreement, (a) each Lender severally agrees to make Loans to the Borrower from time to time in amounts not to exceed in the aggregate at any one time outstanding the amount of its Commitment, and (b) each Issuer agrees to issue Letters of Credit for the account of the Borrower from time to time in amounts not to exceed in the aggregate at any one time outstanding the amount of its Letter of Credit Commitment (and each Lender severally agrees to participate in each such Letter of Credit as more fully set forth in Section 2.19); provided (i) that the Aggr egate Outstanding Credit Exposure shall not at any time exceed the Aggregate Commitment; (ii) the Outstanding Credit Exposure of any Lender shall not at any time exceed the amount of such Lender's Commitment; and (iii) the outstanding amount of Letter of Credit Obligations shall not exceed the Letter of Credit Sublimit. Subject to the terms of this Agreement, the Borrower may borrow, repay and reborrow Loans at any time prior to the Facility Termination Date. The Commitments shall expire on the Facility Termination Date.

2.2 Required Payments; Termination.

The Borrower shall (a) repay the principal amount of all Advances made to it on the Facility Termination Date and (b) deposit into the LC Collateral Accounts on the Facility Termination Date an amount in immediately available funds equal to the aggregate stated amount of all Letters of Credit that will remain outstanding after the Facility Termination Date.

2.3 Ratable Loans.

Each Advance hereunder shall consist of Loans made from the several Lenders ratably in proportion to their respective Pro Rata Shares.

2.4 <u>Types of Advances; Minimum Amount.</u>

The Advances may be Floating Rate Advances or Eurodollar Advances, or a combination thereof, selected by the Borrower in accordance with Sections 2.8 and 2.9. Each Eurodollar Advance shall be in the amount of \$5,000,000 or a higher integral multiple of \$1,000,000, and each Floating Rate Advance shall be in the amount of \$1,000,000 or an integral multiple thereof.

2.5 Commitment Fee.

The Borrower agrees to pay to the Administrative Agent for the account of each Lender which is not a Defaulting Lender in accordance with its Pro Rata Share, a commitment fee equal to the product of (i) the Commitment Fee Rate <u>times</u> (ii) the actual daily amount by which the Aggregate Commitments exceed the sum of (x) the outstanding principal amount of Loans (other

21

than Swing Line Loans) and (y) the outstanding amount of Letter of Credit Obligations, subject to adjustment as provided in <u>Section 2.23</u>. The Commitment Fee shall be due and payable quarterly in arrears on each Payment Date, commencing with the first Payment Date to occur after the Closing Date, and on the Facility Termination Date. The Commitment Fee shall be calculated quarterly in arrears. For purposes of clarification, Swing Line Loans shall not be considered outstanding for purposes of determining the unused portion of the Aggregate Commitments.

2.6 <u>Changes in Aggregate Commitment.</u>

- (a) <u>Optional Commitment Reductions</u>. The Borrower may at any time and from time to time, without penalty, permanently reduce the Aggregate Commitment in whole or in part ratably among the Lenders (according to their respective Pro Rata Shares) in a minimum principal amount of \$5,000,000 and integral multiples of \$1,000,000 in excess thereof, upon at least three (3) Business Days' prior written notice to the Administrative Agent, which notice shall specify the amount of any such reduction; <u>provided</u> that the amount of the Aggregate Commitment may not be reduced below the Aggregate Outstanding Credit Exposure. All accrued commitment fees shall be payable on the effective date of any termination of the obligations of the Lenders to make Loans hereunder.
- (b) <u>Optional Increase of Commitments</u>. The Borrower may at any time and from time to time, upon five (5) days' prior written notice to the Administrative Agent, increase the Commitments (but not the Letter of Credit Sublimit or the Swing Line Sublimit) with additional Commitments from any existing Lender with a Commitment or new Commitments from any other Person selected by the Borrower and reasonably acceptable to the Administrative Agent, the Swing Line Lender and the Issuers; <u>provided</u> that:
 - (i) any such increase shall be in a minimum principal amount of \$10,000,000 and in integral multiples of \$1,000,000 in excess thereof;
 - (ii) no Default or Unmatured Default shall exist and be continuing at the time of any such increase;
 - (iii) no existing Lender shall be under any obligation to increase its Commitment and any such decision whether to increase its Commitment shall be in such Lender's sole and absolute discretion;
 - (iv) (A) any new Lender shall join this Agreement by executing such joinder documents required by the Administrative Agent and/or (B) any existing Lender electing to increase its Commitment shall have executed a commitment agreement reasonably satisfactory to the Administrative Agent;
 - (v) as a condition precedent to such increase, the Borrower shall deliver to the Administrative Agent (A) a certificate of each of the Borrower, the Guarantor and KCPL dated as of the date of such increase (in sufficient copies for each Lender) signed by an Authorized Officer of such Person certifying and

22

attaching the resolutions adopted by such Person approving or consenting to such increase, and (B) a certificate of the Borrower certifying that, before and after giving effect to such increase, (x) the representations and warranties contained in <u>Article VI</u> and the other Loan Documents are true and correct in all material respects on and as of the date of such increase, except to the extent that such representations and warranties specifically refer to an earlier date, in which case they are true and correct in all material respects as of such earlier date, and (y) no Default or Unmatured Default exists as of such date; and

(vi) after giving effect to the increase in the Commitments (but without giving effect to any increase in the Commitments pursuant to Sections 2.6(c) or (d)), the Aggregate Commitment shall not be greater than SIX HUNDRED AND FIFTY MILLION DOLLARS (\$650,000,000).

The Borrower shall prepay any Loans owing by it and outstanding on the date of any such increase with the proceeds required to be advanced by any Lender increasing its, or any Person providing a new, Commitment under this Section (and pay any additional amounts required pursuant to Article III) to the extent necessary to keep the outstanding Loans ratable with any revised Commitments arising from any nonratable increase in the Commitments under this Section.

(c) <u>Transfer and Re-Transfer of the Commitments to the Guarantor</u>. (i) Subject to Section 4.2 of the Great Plains Credit Agreement, the Borrower and the Guarantor may, by joint election in a written notice to the Administrative Agent (which shall promptly provide a copy of such notice to the

Lenders) and the "Administrative Agent" under the Great Plains Credit Agreement, transfer from time to time up to TWO HUNDRED MILLION DOLLARS (\$200,000,000) of the unused Commitments to the Commitments (as such term is defined in the Great Plains Credit Agreement) under the Great Plains Credit Agreement (any such reduction, a "GMO-GPE Transfer").

- (ii) Subject to Section 5.2, the Borrower and the Guarantor may, by joint election in a written notice to the Administrative Agent (which shall promptly provide a copy of such notice to the Lenders) and the "Administrative Agent" under the Great Plains Credit Agreement, re-transfer from time to time any portion (up to \$200,000,000) of the unused Commitments (as such term is defined in the Great Plains Credit Agreement) previously transferred from this Agreement to the Great Plains Credit Agreement pursuant to subclause (c)(i) above back to the Commitments hereunder (any such addition, a "GMO-GPE Re-Transfer").
- (iii) On the effective date of a GMO-GPE Transfer, which shall be specified in the notice delivered pursuant to Section 2.6(c)(i) and which shall not be less than five (5) Business Days subsequent to the date of giving of such notice, then subject to the satisfaction of the conditions precedent specified in Section 4.2 of the Great Plains Credit Agreement, (i) the Commitments hereunder shall be ratably decreased by the aggregate amount specified in such notice and (ii) the aggregate amount of the "Commitments" under and as defined in the Great Plains Credit Agreement shall be ratably increased by such amount. Such GMO-GPE Transfer</di>

23

and the consequent decreases and increases shall be irrevocable subject, however, to subsequent permissible GMO-GPE Re-Transfers in accordance with the terms hereof.

- (iv) On the effective date of a GMO-GPE Re-Transfer, which shall be specified in the notice delivered pursuant to Section 2.6(c)(ii) and which shall not be less than five (5) Business Days subsequent to the date of giving of such notice, then subject to the satisfaction of the conditions precedent specified in Section 5.2, (i) the Commitments hereunder shall be ratably increased by the aggregate amount specified in such notice and (ii) the aggregate amount of the "Commitments" under and as defined in the Great Plains Credit Agreement shall be ratably decreased by such amount. Such GMO-GPE Re-Transfer and the consequent decreases and increases shall be irrevocable, subject, however, to subsequent permissible GMO-GPE Transfers in accordance with the terms hereof.
- (d) <u>Transfer and Re-Transfer of the "Commitments" of the Guarantor to the Borrower</u>. (i) Subject to <u>Section 5.2</u>, the Borrower and the Guarantor may, by joint election in a written notice to the Administrative Agent (which shall promptly provide a copy of such notice to the Lenders) and the "Administrative Agent" under the Great Plains Credit Agreement, transfer from time to time up to TWO HUNDRED MILLION DOLLARS (\$200,000,000) of the unused Commitments (as such term is defined in the Great Plains Credit Agreement) under the Great Plains Credit Agreement to the Commitments hereunder (any such addition, a "GPE-GMO Transfer").
- (ii) Subject to Section 4.2 of the Great Plains Credit Agreement, the Borrower and the Guarantor may, by joint election in a written notice to the Administrative Agent (which shall promptly provide a copy of such notice to the Lenders) and the "Administrative Agent" under the Great Plains Credit Agreement, re-transfer from time to time any portion (up to \$200,000,000) of the unused Commitments previously transferred from the Great Plains Credit Agreement to this Agreement pursuant to <u>subclause (d)(i)</u> above back to the Commitments (as such term is defined in the Great Plains Credit Agreement) under the Great Plains Credit Agreement (any such decrease, a "<u>GPE-GMO Re-Transfer</u>").
- (iii) On the effective date of a GPE-GMO Transfer, which shall be specified in the notice delivered pursuant to Section 2.6(d)(i) and which shall not be less than five (5) Business Days subsequent to the date of giving of such notice, then subject to (A) the satisfaction of the conditions precedent specified in Section 5.2 and (B) if necessary in the reasonable discretion of the Administrative Agent, receipt by the Administrative Agent of a certificate of each of the Borrower and the Guarantor dated as of the date of such GPE-GMO Transfer (in sufficient copies for each Lender) signed by an Authori zed Officer of the Borrower or the Guarantor certifying and attaching resolutions adopted by the Borrower approving or consenting to borrowings up to at least the amount of the Commitments after giving effect to such GPE-GMO Transfer, (i) the Commitments hereunder shall be ratably increased by the aggregate amount specified in such notice and (ii) the aggregate amount of the "Commitments" under and as defined in the Great Plains Credit Agreement shall be ratably decreased by such amount. Such GPE-GMO Transfer and the consequent increases and decreases shall be irrevocable subject, however, to subsequent permissible GPE-GMO Re-Transfers in accordance with the terms hereof.

24

(iv) On the effective date of a GPE-GMO Re-Transfer, which shall be specified in the notice delivered pursuant to Section 2.6(d)(ii) and which shall not be less than five (5) Business Days subsequent to the date of giving of such notice, then subject to the satisfaction of the conditions precedent specified in Section 4.2 of the Great Plains Credit Agreement, (i) the "Commitments" under and as defined in the Great Plains Credit Agreement shall be ratably increased by the aggregate amount specified in such notice and (ii) the aggregate amount of the Commitments hereunder shall be ratably decreased by such amount. Such GPE-GMO Re-Transfer and the consequent increases and decreases shall be irrevocable, subject, however, to subsequent permissible GPE-GMO Transfers in accordance with the terms hereof.

Notwithstanding the foregoing, the parties acknowledge and agree that after giving effect to any increase in the Commitments pursuant to any GMO-GPE Transfer or GMO-GPE Re-Transfer contemplated in subclauses (i) and (ii) of Section 2.6(c) and any GPE-GMO Transfer or GPE-GMO Re-Transfer contemplated in subclauses (i) and (ii) of Section 2.6(d), (a) the aggregate Commitments hereunder shall not exceed \$650,000,000, (b) the aggregate Commitments under and as defined in the Great Plains Credit Agreement shall not exceed \$400,000,000 and (c) the aggregate commitments under this Agreement, the Great Plains Credit Agreement and the KCPL GMO Credit Agreement shall not exceed \$1,250,000,000; provided, however, that after giving effect to such GMO-GPE Transfers, GMO-GPE Re-Transfers, GPE-GMO Transfers, GPE-GMO Re-Transfers, Section 2.6(b) and Section 2.6(b) of the Great Plains Credit Agreement, (x) the aggregate Commitments hereunder shall not exceed \$850,000,000, (y) the aggregate Commitments under and as defined in

the Great Plains Credit Agreement shall not exceed \$600,000,000 and (z) the aggregate commitments under this Agreement, the Great Plains Credit Agreement and the KCPL Credit Agreement shall not exceed \$1,850,000,000.

2.7 Optional Prepayments.

- (a) The Borrower may at any time and from time to time, in whole or in part, prepay Floating Rate Advances upon one (1) Business Day's prior written notice to the Administrative Agent, without penalty or premium. Each partial prepayment of Floating Rate Advances shall be in an aggregate amount of \$1,000,000 or an integral multiple thereof. Each such notice shall specify the date and amount of such prepayment, and the payment amount specified in such notice shall be due and payable on the date specified therein; <u>provided</u>, <u>however</u>, subject to <u>Section 3.4</u>, t hat a notice of prepayment may state that such notice is conditioned upon the effectiveness of other credit facilities or the closing of a capital markets transaction, in which case such notice may be revoked by the Borrower (by notice to the Administrative Agent on or prior to the specified prepayment date) if such condition is not satisfied.
- (b) The Borrower may at any time and from time to time, in whole or in part, prepay Eurodollar Advances (subject to the payment of any funding indemnification amounts required by Section 3.4) upon three (3) Business Days' prior written notice to the Administrative Agent, without penalty or premium. Each partial prepayment of Eurodollar Advances shall be in an aggregate amount of \$5,000,000 or a higher integral multiple of \$1,000,000. Each such notice shall specify the date and amount of such prepayment, and the payment amount specified in such notice shall be due and payable on the date specified therein; provided, however, subject to

25

<u>Section 3.4</u>, that a notice of prepayment may state that such notice is conditioned upon the effectiveness of other credit facilities or the closing of a capital markets transaction, in which case such notice may be revoked by the Borrower (by notice to the Administrative Agent on or prior to the specified prepayment date) if such condition is not satisfied. Any notice to prepay Eurodollar Advances shall indicate the Interest Period for such Eurodollar Advances.

- (c) Subject to Section 2.23, all prepayments of Advances shall be applied ratably to the Loans of the Lenders in accordance with their respective Pro Rata Shares.
- (d) The Borrower may at any time and from time to time prepay Swing Line Loans upon one (1) Business Day's prior notice to the Swing Line Lender (with a copy to the Administrative Agent), without penalty or premium; provided that (i) such notice must be received by the Swing Line Lender and the Administrative Agent not later than 1:00 p.m. on the date of the prepayment, and (ii) any such prepayment shall be in a minimum principal amount of \$500,000 or a whole multiple of \$100,000 in excess thereof (or, if less, the entire principal thereof then outstanding). Each such notice shall specify the date and amount of such prepayment. If such notice is given by the Borrower, the Borrower's hall make such prepayment and the payment amount specified in such notice shall be due and payable on the date specified therein; provided, however, subject to Section 3.4, that a notice of prepayment may state that such notice is conditioned upon the effectiveness of other credit facilities or the closing of a capital markets transaction, in which case such notice may be revoked by the Borrower (by notice to the Administrative Agent on or prior to the specified prepayment date) if such condition is not satisfied.

2.8 <u>Method of Selecting Types and Interest Periods for New Advances.</u>

The Borrower shall select the Type of Advance and, in the case of each Eurodollar Advance, the Interest Period applicable thereto from time to time. The Borrower shall give the Administrative Agent irrevocable notice (a "Borrowing Notice") not later than noon on the Borrowing Date of each Floating Rate Advance and not later than noon three (3) Business Days before the Borrowing Date for each Eurodollar Advance, specifying:

- (i) the Borrowing Date, which shall be a Business Day, of such Advance,
- (ii) the aggregate amount of such Advance,
- (iii) the Type of Advance selected, and
- (iv) in the case of each Eurodollar Advance, the Interest Period applicable thereto.

Not later than 1:00 p.m. on each Borrowing Date, each Lender shall make available its Loan or Loans in funds immediately available to the Administrative Agent at its address specified pursuant to <u>Article XIV</u>. The Administrative Agent will make the funds so received from the Lenders available to the Borrower at the Administrative Agent's aforesaid address.

26

2.9 <u>Conversion and Continuation of Outstanding Advances.</u>

Floating Rate Advances shall continue as Floating Rate Advances unless and until such Floating Rate Advances are converted into Eurodollar Advances pursuant to this Section 2.9 or are repaid in accordance with Section 2.7. Each Eurodollar Advance shall continue as a Eurodollar Advance until the end of the then applicable Interest Period therefor, at which time such Eurodollar Advance shall be automatically converted into a Floating Rate Advance unless (x) such Eurodollar Advance is or was repaid in accordance with Section 2.7 or (y) the Borrower shall have given the Administrative Agent a Conversion/Continuation Notice (as defined below) requesting that, at the end of such Interest Period, such Eurodollar Advance continue as a Eurodollar Advance for the same or another Interest Period. Subject to the terms of Section 2.4, the Borrower may elect from time to time to convert all or any part of a Floating Rate Advance into a Eurodollar Advance. The Borrower shall give the Administrative Agent irrevocable notice (a "Conversion/Continuation Notice") of each conversion of a Floating Rate Advance into a Eurodollar Advance or continuation of a Eurodollar Advance not later than 11:00 a.m. at least three (3) Business Days prior to the date of the requested conversion or continuation, specifying:

- (i) the requested date, which shall be a Business Day, of such conversion or continuation,
- (ii) the aggregate amount and Type of the Advance which is to be converted or continued, and
- (iii) the amount of such Advance which is to be converted into or continued as a Eurodollar Advance and the duration of the Interest Period applicable thereto.

2.10 <u>Interest Rates</u>.

Each Floating Rate Advance shall bear interest on the outstanding principal amount thereof, for each day from and including the date such Advance is made or is automatically converted from a Eurodollar Advance into a Floating Rate Advance pursuant to Section 2.9, to but excluding the date it is paid or is converted into a Eurodollar Advance pursuant to Section 2.9 hereof, at a rate per annum equal to the Alternate Base Rate plus the Applicable Margin for such day. Changes in the rate of interest on that portion of any Advance maintained as a Floating Rate Advance will take effect simultaneously with each change in the Alternate Base Rate. Each Eurodollar Advance shall bear inter est on the outstanding principal amount thereof from and including the first day of the Interest Period applicable thereto to (but not including) the last day of such Interest Period at the Eurodollar Rate plus the Applicable Margin based upon the Borrower's selections under Sections 2.8 and 2.9 and otherwise in accordance with the terms hereof. No Interest Period may end after the Facility Termination Date. Each Swing Line Loan shall bear interest on the outstanding principal amount thereof from the applicable borrowing date at a rate per annum equal to the Alternate Base Rate plus the Applicable Margin for each day from and including the date such Swing Line Loan is made to but excluding the date it is paid. Changes in the rate of interest on that portion of any Swing Line Loan will take effect simultaneously with each change in the Alternate Base Rate.

27

2.11 Rates Applicable After Default.

Notwithstanding anything to the contrary contained in Section 2.8 or 2.9, during the continuance of a Default or Unmatured Default the Required Lenders may, at their option, by notice to the Borrower (which notice may be revoked at the option of the Required Lenders notwithstanding any provision of Section 9.2 requiring unanimous consent of the Lenders to changes in interest rates), declare that no Advance may be made as, converted into or continued as a Eurodollar Advance. During the continuance of a Default the Required Lenders may, at their option, by notice to the Borrower (which notice may be revoked at the option of the Required Lenders notwithstanding any provision of Section 9.2 requiring unanimous consent of the Lenders to changes in interest rates), declare that (i) each Eurodollar Advance shall bear interest for the remainder of the applicable Interest Period at the rate otherwise applicable to such Interest Period plus two percent (2%) per annum, (ii) each Floating Rate Advance shall bear interest at a rate per annum equal to the Alternate Base Rate in effect from time to time plus the Applicable Margin plus two percent (2%) per annum, (iii) the Letter of Credit Fee Rate shall be increased by two percent (2%) per annum and (iv) each Swing Line Loan shall bear interest at a rate per annum equal to the Alternate Base Rate in effect from time to time plus the Applicable Margin plus two percent (2%) per annum; provided that, during the continuance of a Default under Sections 8.7 or 8.8, the interest rates set forth in clauses (ii), (iii) and (iv) above and the increase in the Letter of Credit Fee Rate set forth in clause (iii) above shall be applicable to all applicable Credit Extensions without any election or action on the part of the Administrative Agent or any Lender.

2.12 <u>Method of Payment</u>.

Except to the extent that any Tax is required to be withheld or deducted under applicable Law, but subject to the provisions of Article III, all payments of the Obligations hereunder shall be made, without setoff, deduction, or counterclaim, in immediately available funds to the Administrative Agent at the Administrative Agent among the Lenders in accordance with their respective Pro Rata Shares. Each payment delivered to the Administrative Agent for the account of any Lender shall be delivered promptly by the Administrative Agent to such Lender in the same type of funds that the Administrative Agent received at the Administrative Agent's Office from such Lender.

2.13 <u>Noteless Agreement; Evidence of Indebtedness.</u>

- (a) Each Lender shall maintain in accordance with its usual practice an account or accounts evidencing the indebtedness of the Borrower to such Lender resulting from each Loan made by such Lender from time to time, including the amounts of principal and interest payable and paid to such Lender from time to time hereunder.
- (b) The Administrative Agent shall also maintain accounts in which it will record (i) the amount of each Loan (including any Swing Line Loan) made hereunder, the Type thereof and the Interest Period with respect thereto, (ii) the amount of any principal or interest due and payable or to become due and payable from the Borrower to each Lender hereunder, (iii) the

28

original stated amount of each Letter of Credit and the amount of Letter of Credit Obligations outstanding at any time and (iv) the amount of any sum received by the Administrative Agent hereunder from the Borrower and each Lender's share thereof.

- (c) The entries maintained in the accounts maintained pursuant to <u>clauses (a)</u> and <u>(b)</u> above shall be *prima facie* evidence of the existence and amounts of the Obligations therein recorded; <u>provided</u> that the failure of the Administrative Agent or any Lender to maintain such accounts or any error therein shall not in any manner affect the obligation of the Borrower to repay the Obligations in accordance with their terms.
- (d) Any Lender may request that its Loans be evidenced by a promissory note substantially in the form of <u>Exhibit D</u> (a "<u>Revolving Note</u>"). In such event, the Borrower shall prepare, execute and deliver to such Lender a Note payable to the order of such Lender. Thereafter, the Loans evidenced by

such Note and interest thereon shall at all times (including after any assignment pursuant to <u>Section 13.1(b)</u>) be represented by one or more Notes payable to the order of the payee named therein or any assignee pursuant to <u>Section 13.1(b)</u>, except to the extent that any such Lender or assignee subsequently returns any such Note for cancellation and requests that such Loans once again be evidenced as described in <u>clauses (a)</u> and <u>(b)</u> above. The Swing Line Loans shall be evidenced by a promissory note substantially in the form of <u>Exhibit E</u> (the "<u>Swing Line Note</u>").

2.14 <u>Telephonic Notices</u>.

The Borrower hereby authorizes the Lenders and the Administrative Agent to extend, convert or continue Advances, effect selections of Types of Advances and to transfer funds based on telephonic notices made by any person or persons the Administrative Agent or any Lender in good faith believes to be acting on behalf of the Borrower. The Borrower agrees to deliver promptly to the Administrative Agent a written confirmation, if such confirmation is requested by the Administrative Agent or any Lender, of each telephonic notice signed by an Authorized Officer. If the written confirmation differs in any material respect from the action taken by the Administrative Agent and the Lenders, the records of the Administrative Agent and the Lenders shall govern absent manifest error.

2.15 <u>Interest Payment Dates; Interest and Fee Basis</u>.

Interest accrued on each Floating Rate Advance and each Swing Line Loan shall be payable on each Payment Date, commencing with the first such date to occur after the date hereof, and at maturity. Interest accrued on each Eurodollar Advance shall be payable on the last day of its applicable Interest Period, on any date on which such Eurodollar Advance is prepaid, whether by acceleration or otherwise, and at maturity. Interest accrued on each Eurodollar Advance having an Interest Period longer than three (3) months shall also be payable on the last day of each three (3)-month interval during such Interest Period. All computations of interest for Floating Rate Loans and Swing Line Loans (including Floating Rate Loans determined by reference to the Eurodollar Rate) shall be made on the basis of a year of 365 or 366 days, as the case may be, and actual days elapsed. All other computations of interest and fees shall be calculated for actual days elapsed on the basis of a 360-day year. Interest shall be payable for

29

the day an Advance is made but not for the day of any payment on the amount paid if payment is received prior to 1:00 p.m. at the place of payment (it being understood that the Administrative Agent shall be deemed to have received a payment prior to 1:00 p.m. if (x) the Borrower has provided the Administrative Agent with evidence satisfactory to the Administrative Agent that the Borrower has initiated a wire transfer of such payment prior to such time and (y) the Administrative Agent actually receives such payment on the same Business Day on which such wire transfer was initiated). If any payment of principal of or interest on an Advance shall become due on a day which is not a Business Day, such payment shall be made on the next succeeding Business Day and, in the case of a principal payment, such extension of time s hall be included in computing interest in connection with such payment.

2.16 <u>Notification of Advances, Interest Rates, Prepayments and Commitment Reductions.</u>

Promptly after receipt thereof, the Administrative Agent will notify each Lender of the contents of each Aggregate Commitment reduction notice, Borrowing Notice, Conversion/Continuation Notice, and repayment notice received by it hereunder. The Administrative Agent will notify each Lender of the interest rate applicable to each Eurodollar Advance promptly upon determination of such interest rate and will give each Lender prompt notice of each change in the Alternate Base Rate. The Administrative Agent will also promptly notify each Lender of any increase or reduction of the Aggregate Commitments pursuant to the terms hereof. The Administrative Agent will also promptly notify the Borrower of the interest rate applicable to each Eurodollar Advance and Floating Rate Advance promptly upon determination of such interest rates and will give the Borrower prompt notice of each change in the Alternate Base Rate.

2.17 <u>Lending Installations</u>.

Each Lender may book its Loans at any Lending Installation selected by such Lender and may change its Lending Installation from time to time. All terms of this Agreement shall apply to any such Lending Installation and the Loans and any Notes issued hereunder shall be deemed held by each Lender for the benefit of such Lending Installation. Each Lender may, by written notice to the Administrative Agent and the Borrower in accordance with <u>Article XIV</u>, designate replacement or additional Lending Installations through which Loans will be made by it and for whose account Loan payments are to be made; <u>provided</u>, <u>however</u>, that such designation shall be permitted by applicable Laws and that such designation shall not result in the imposition of increased costs or obligations on the Borrower.

2.18 <u>Non-Receipt of Funds by the Administrative Agent.</u>

Unless a Lender notifies the Administrative Agent, prior to the proposed date of any Eurodollar Advance (or, in the case of any Floating Rate Advance, prior to noon on the date of such Advance), that such Lender will not make available to the Administrative Agent such Lender's share of such Advance, the Administrative Agent may assume that such payment has been made. Unless the Borrower notifies the Administrative Agent, prior to the date on which it is scheduled to make a payment to the Administrative Agent of principal, interest or fees for the

30

account of the Lenders, that it does not intend to make such payment, the Administrative Agent may assume that such payment has been made. The Administrative Agent may, but shall not be obligated to, make the amount of such payment available to the intended recipient in reliance upon such assumption. If such Lender or the Borrower, as the case may be, has not in fact made such payment to the Administrative Agent, the recipient of such payment shall, on demand by the Administrative Agent, repay to the Administrative Agent the amount so made available together with interest thereon in respect of each day during the period commencing on the date such amount was so made available by the Administrative Agent until the date the Administrative Agent recovers such amount at a rate per annum equal to (x) in the case of payment by a Lender, the Federal Funds Rate for such day or (y) in the case of payment by the Borrower, the interest rate applicable to the relevant Loan.

2.19 Letters of Credit.

- (a) <u>Issuance</u>. Each Issuer hereby agrees, on the terms and conditions set forth in this Agreement and in reliance upon the agreements of the Lenders set forth in this <u>Section 2.19</u>, to issue Letters of Credit and to extend, increase, decrease or otherwise modify Letters of Credit ("<u>Modify</u>," and each such action a "<u>Modification</u>") in amounts not to exceed in the aggregate at any one time outstanding the amount of its Letter of Credit Commitment from time to time on any Business Day from a nd including the date of this Agreement and prior to the Facility Termination Date upon the request of the Borrower; <u>provided</u> that immediately after each such Letter of Credit is issued or Modified, (x) the Aggregate Outstanding Credit Exposure shall not exceed the Aggregate Commitment and (y) the outstanding amount of Letters of Credit Obligations shall not exceed the Letter of Credit Sublimit. No Letter of Credit shall have an expiry date later than the date that is five (5) days prior to the scheduled Facility Termination Date. All Existing Letters of Credit shall be deemed to have been issued pursuant hereto, and from and after the Closing Date shall be subject to and governed by the terms and conditions hereof.
- (b) <u>Participations</u>. Upon the issuance or Modification by any Issuer of a Letter of Credit in accordance with this <u>Section 2.19</u>, such Issuer shall be deemed, without further action by any Person, to have unconditionally and irrevocably sold to each Lender, and each Lender shall be deemed, without further action by any Person, to have unconditionally and irrevocably purchased from such Issuer, a participation in such Letter of Credit (and each Modification thereof) and the related Letter of Credit Obligations in proportion to its Pro Rata Share.
- (c) Notice. Subject to Section 2.19(a), the Borrower shall give the applicable Issuer and the Administrative Agent notice prior to 11:00 a.m. at least three (3) Business Days (or such lesser period of time as the Administrative Agent and such Issuer may agree in their sole discretion) prior to the proposed date of issuance or Modification of each Letter of Credit, specifying, in form and detail reasonably satisfactory to the applicable Issuer, (A) the proposed issuance date of the requested Letter of Credit (which shall be a Business Day) and the expiry date of such Letter of Credit; (B) the amount thereof; (C) the name and a ddress of the beneficiary thereof; (D) the documents to be presented by such beneficiary in case of any drawing thereunder; (E) the full text of any certificate to be presented by such beneficiary in case of any drawing thereunder; (F) the purpose and nature of the requested Letter of Credit; and (G) such

31

other matters as the applicable Issuer may reasonably require. In the case of a request for a Modification of any outstanding Letter of Credit, such notice shall specify in form and detail reasonably satisfactory to the applicable Issuer (A) the Letter of Credit to be amended; (B) the proposed date of amendment thereof (which shall be a Business Day); (C) the nature of the proposed amendment; and (D) such other matters as the applicable Issuer may reasonably require. Upon receipt of such notice, the applicable Issuer shall promptly notify the Administrative Agent, and the Administrative Agent shall promptly notify each Lender, of the contents thereof and of the amount of such Lender's participation in such proposed Letter of Credit. The issuance or Modification by an Issuer of any Letter of Credit shall, in addition to the conditions precedent set forth in Article V (the satisfaction of which such Issuer shall have no duty to ascertain, it being understood, however, that such Issuer shall not issue any Letter of Credit if it has received written notice from the Borrower, the Administrative Agent or any Lender one (1) day prior to the proposed date of issuance, that any such condition precedent has not been satisfied), be subject to the conditions precedent that such Letter of Credit shall be satisfactory to such Issuer and that an Authorized Officer of the Borrower shall have reasonably requested (each a "Letter of Credit Application"). Promptly a fter its delivery of any Letter of Credit or any amendment to a Letter of Credit to an advising bank with respect thereto or to the beneficiary thereof, the applicable Issuer will also deliver to the Borrower and the Administrative Agent a true and complete copy of such Letter of Credit or amendment. In the event of any conflict between the terms of this Agreement and the terms of any Letter of Credit Application, the terms of this Agreement shall control.

- (d) Letter of Credit Fees. Upon the issuance of each Letter of Credit and until termination, cancellation or expiration of such Letter of Credit, the Borrower agrees to pay to the Administrative Agent, for the account of the Lenders ratably in accordance with their respective Pro Rata Shares, with respect to such Letter of Credit, a letter of credit fee (the "Letter of Credit Fee") at a per annum rate equal to the Letter of Credit Fee Rate in effect from time to time multiplied by the daily maximum amount available under such Letter of Credit, such fee to be payable in arrears on each Payment Date, on the Facility Terminati on Date and, if applicable, thereafter on demand; provided, however, any Letter of Credit Fees otherwise payable for the account of a Defaulting Lender with respect to any Letter of Credit as to which such Defaulting Lender has not provided Cash Collateral satisfactory to the applicable Issuer pursuant to this Section 2.19 shall be payable, to the maximum extent permitted by applicable Law, to the other Lenders in accordance with the upward adjustments in their respective Pro Rata Shares allocable to such Letter of Credit pursuant to Section 2.23(a)(iv), with the balance of such fee, if any, payable to the applicable Issuer for its own account. The Borrower shall also pay to each Issuer for its own account (x) a fronting fee of 0.20% per annum on the aggregate maximum stated amount for each Letter of Credit issued by such Issuer and currently outstanding, with such fee to be payable in arrears on the first Business Day following the end of each March, June, September and December, and (y) documentary and processing charges in connection with the issuance or Modification of and draws under Letters of Credit in accordance with such Issuer's standard schedule for such charges as in effect from time to time.
- (e) <u>Administration; Reimbursement by Lenders</u>. Upon receipt from the beneficiary of any Letter of Credit of any demand for payment under such Letter of Credit, the applicable

32

Issuer shall notify the Administrative Agent and the Borrower and the Administrative Agent shall promptly notify each Lender of the amount to be paid by such Issuer as a result of such demand and the proposed payment date (the "Letter of Credit Payment Date"). After honoring any demand for payment under any Letter of Credit, the applicable Issuer shall, upon the request of the Borrower, send to the Borrower the beneficiary statement or an accurate copy of such beneficiary statement pursuant to which the applicable beneficiary requested payment. The responsibility of any Issuer to the Borrower and each Lender shall be only to determine that the documents delivered under each Letter of Credit issued by such Issuer in connection with a demand f or payment are in conformity in all material respects with such Letter of Credit. Each Issuer shall endeavor to exercise the same care in its issuance and administration of Letters of Credit as it does with respect to letters of credit in which no participations are granted, it being understood that in the absence of any gross

negligence or willful misconduct by such Issuer, each Lender shall be unconditionally and irrevocably obligated, without regard to the occurrence of any Default or any condition precedent whatsoever, to reimburse (and the Administrative Agent may apply Cash Collateral for this purpose) such Issuer on demand for (i) such Lender's Pro Rata Share of the amount of each payment made by such Issuer under each Letter of Credit to the extent such amount is not reimbursed by the Borrower pursuant to Section 2.19(f) below, plus, without limiting the other provisions of this Agreement, (ii) interest on the fore going amount, for each day from the date of the applicable payment by such Issuer to the date on which such Issuer is reimbursed by such Lender for its Pro Rata Share thereof, at a rate per annum equal to the greater of the Federal Funds Rate and a rate determined by the applicable Issuer in accordance with banking industry rules on interbank compensation, plus (iii) any administrative, processing or similar fees customarily charged by the applicable Issuer in connection with the foregoing.

(f) Reimbursement by Borrower. (i) The Borrower shall be irrevocably and unconditionally obligated to reimburse each Issuer through the Administrative Agent, for any amount to be paid by such Issuer upon any drawing under any Letter of Credit without presentment, demand, protest or other formalities of any kind, not later than (i) the applicable Letter of Credit Payment Date, if the Borrower shall have received such notice on or prior to 11:00 a.m. on such Letter of Credit Payment Date or (ii) on the Business Day immediately following the applicable Letter of Credit Payment Date, if the Borrower shall have received such notice after 11:00 a.m. (either of such dates, the "Honor Date"); provided that the Borrower shall not be precluded from asserting any claim for direct (but not consequential) damages suffered by the Borrower which the Borrower proves were caused by (i) the willful misconduct or gross negligence of such Issuer in determining whether a request presented under any Letter of Credit complied with the terms of such Letter of Credit or (ii) such Issuer's failure to pay under any Letter of Credit after the presentation to it of a request strictly complying with the terms and conditions of such Letter of Credit. The Administrative Agent will pay to each Lender ratably in accordance with its Pro Rata Share all amounts received by it from the Borrower for application in payment, in whole or in part, of the Reimbursement Obligation in respect of any Letter of Credit, but only to the extent such Lender made payment to the applicable Issuer in respect of such Letter of Credit pursuant to Section 2.19(e). If the Borrower fails to so reimburse any Issuer for any amounts paid by any Issuer by such applicable time, the Administrative Agent shall promptly notify each Lender of the Honor Date, the amount of the unreimbursed drawing (the "Unreimbursed Amount"), and the amount of such Lender's Pro Rata Share thereof. In such event, the Borrower shall be deemed to have requested a Floating Rate Advance to be disburse

33

on the Honor Date in an amount equal to the Unreimbursed Amount and, so long as no Unmatured Default or Default has occurred and is continuing, such disbursement shall be deemed to occur automatically without further act and without regard to the minimum and multiples specified in Section 2.4 for the principal amount of Floating Rate Advance but subject to the conditions set forth in Section 5.2 (other than delivery by the Borrower of a Borrowing Notice) and provided that, after giving effect to such Floating Rate Advance, the Aggregate Outstanding Credit Ex posure shall not exceed the Aggregate Commitments. To the extent that any amounts paid by an Issuer are not reimbursed by the Borrower pursuant to the terms of this Section 2.19(f), all such amounts shall bear interest, payable on demand, for each day until paid at a rate per annum equal to the sum of two percent (2%) plus the rate applicable to Floating Rate Advances. Any notice given by any Issuer or the Administrative Agent pursuant to this Section 2.19(f)(i) may be given by telephone if immediately confirmed in writing; provided that the lack of such an immediate confirmation shall not affect the conclusiveness or binding effect of such notice. Until each Lender funds its Floating Rate Loan to reimburse the applicable Issuer for any amount drawn under any Letter of Credit, interes t in respect of such Lender's Pro Rata Share of such amount shall be solely for the account of such Issuer. Each Lender shall upon any notice pursuant to this Section 2.19(f) make funds available (and the Administrative Agent may apply Cash Collateral provided for this purpose) for the account of the applicable Issuer at the Administrative Agent's Office in an amount equal to its Pro Rata Share of the Unreimbursed Amount not later than 1:00 p.m. on the Business Day specified in such notice by the Administrative Agent. The Administrative Agent shall remit the funds so received to the applicable Issuer.

(ii) Each Lender's obligation to make Floating Rate Loans to reimburse an Issuer for amounts drawn under Letters of Credit, as contemplated by this Section 2.19(f), shall be absolute and unconditional and shall not be affected by any circumstance, including (A) any setoff, counterclaim, recoupment, defense or other right which such Lender may have against the applicable Issuer, the Borrower or any other Person for any reason whatsoever; (B) the occurrence or continuance of a Default, or (C) any other occurrence, event or condition, whether or not similar to any of the foregoing; provided, however, that each Lender's obligation to make Floating Rate Loans is subject to the conditions set forth in Section 5.2 (other than delivery by the Borrower of a Borrowing Notice).

If any payment received by the Administrative Agent for the account of any Issuer pursuant to Section 2.19(f) is required to be returned under any of the circumstances described in Section 12.3 (including pursuant to any settlement entered into by an Issuer in its discretion), each Lender shall pay to the Administrative Agent for the account of the applicable Issuer its Pro Rata Share thereof on demand of the Administrative Agent, plus interest thereon from the date of such demand to the date such amount is returned by such Lender, at a rate per annum equal to the Federal Funds Rate from time to time in effect. The obligations of the Lenders under this paragraph shall survive the payment in full of the Obligations and the termination of this Agreement.

(g) <u>Obligations Absolute</u>. Except to the extent that any Tax is required to be withheld or deducted under applicable Law, but subject to the provisions of <u>Article III</u>, the Borrower's obligations under this <u>Section 2.19</u> shall be absolute, unconditional and irrevocable under any

34

and all circumstances and irrespective of any setoff, counterclaim or defense to payment which the Borrower or any Subsidiary may have or have had against any Issuer, any Lender or any beneficiary or transferee of a Letter of Credit. The Borrower further agrees with the Issuers and the Lenders that neither any Issuer nor any Lender shall be responsible for, and the Borrower's Reimbursement Obligation in respect of any Letter of Credit shall not be affected by, among other things, (i) the validity, enforceability or genuineness of documents or of any endorsements thereon, even if such documents should in fact prove to be in any or all respects invalid, fraudulent or forged or any statement therein untrue or inaccurate, (ii) any dispute between or among the Borrower, any of its Affiliates, the beneficiary of any Letter of Credit or any financing institution or other party to whom any Letter of Credit may be transferred or any claims or defenses whatsoever of the Borrower or of any of its Affiliates against the beneficiary of any Letter of Credit or any such transferee, (iii) any payment by an Issuer under such Letter of Credit to any Person purporting to be a trustee in bankruptcy, debtor-in-possession, assignee for the benefit of creditors, liquidator, receiver or other representative of or successor to any beneficiary or any transferee of such Letter of Credit, including any arising in

connection with any proceeding under any Debtor Relief Law or (v) any other circumstance or happening whatsoever, whether or not similar to any of the foregoing, including any other circumstance that might otherwise constitute a defense available to, or a discharge of, the Borrower or any Subsidiary. No Issuer shall be liable for any error, omission, interruption or delay in transmission, dispatch or delivery of any message or advice, however transmitted, in connection with any Letter of Credit other than, with respect to any Issuer, any such error, omission, interruption or delay in transmission, dispatch or delivery of any message or advice resulting from the gross negligence or willful misconduct of such Issuer. The Borrower agrees that any action taken or omitted by any Issuer or any Lender under or in connection with any Letter of Credit and the related drafts and documents, if done without gross negligence or willful misconduct, shall be binding upon the Borrower and shall not put any Issuer or any Lender under any liability to the Borrower. The Borrower shall promptly examine a copy of each Letter of Credit and each amendment thereto that is delivered to it and, in the event of any claim of noncompliance with the Borrower's instructions or other irregularity, the Borrower will immediately notify the applicable Issuer. The Borrower shall be conclusively deemed to have waived any such claim against the applicable Issuer and its correspondents unless such notice is given as aforesaid. Nothing in this Section 2.19(g) is intended to limit the right of the Borrower to make a claim against any Issuer for damages as contemplated by the proviso to the first sentence of Section 2.19(f).

(h) Actions of the Issuers. Each Issuer shall be entitled to rely, and shall be fully protected in relying, upon any Letter of Credit, draft, writing, resolution, notice, consent, certificate, affidavit, letter, cablegram, telegram, facsimile, telex or teletype message, statement, order or other document believed by it to be genuine and correct and to have been signed, sent or made by the proper Person or Persons, and upon advice and statements of legal counsel, independent accountants and other experts selected by such Issuer. Each Lender and the Borrower agree that, in paying any drawing under a Letter of Credit, the Issuers shall not have any responsibility to obtain any document (other than any sight draft, certificates and documents expressly required by such Letter of Credit) or to ascertain or inquire as to the validity or accuracy of any such document or the authority of the Person executing or delivering any such

35

document. Each Issuer shall be fully justified in failing or refusing to take any action under this Agreement unless it shall first have received such advice or concurrence of the Required Lenders as it reasonably deems appropriate or it shall first be indemnified to its reasonable satisfaction by the Lenders against any and all liability and expense which may be incurred by it by reason of taking or continuing to take any such action. Notwithstanding any other provision of this Section 2.19, each Issuer shall in all cases be fully protected in acting, or in refraining from acting, under this Agreement in accordance with a request of the Required Lenders, and such request and any action taken or failure to act pursuant thereto shall be binding upon the Lenders and any future holder of a participation in any Letter of Credit issued by such Issuer. The Borrower hereby assumes all risks of the acts or omissions of any beneficiary or transferee with respect to its use of any Letter of Credit; provided, however, that this assumption is not intended to, and shall not, preclude the Borrower's pursuing such rights and remedies as it may have against the beneficiary or transferee at law or under any other agreement. None of the Issuers, the Administrative Agent, any of their respective Related Parties nor any correspondent, participant or assignee of any of the Issuers shall be liable or responsible for any of the matters described in clauses (i) through (v) of Section 2.19(g) except, with respect to any Issuer, for any such matters that are determined by a court of competent jurisdiction by final and non-appealable judgment to have resulted from the gross negligence or willful misconduct of such Person. In furtherance and not in limitation of the foregoing, the Issuers may accept documents that appear on their face to be in order, without responsibility for further investigation, regardless of any notice or information to the contrary, and the Issuers shall not be responsible for the validity or sufficiency of any inst

(i) Indemnification. The Borrower agrees to indemnify and hold harmless each Lender, each Issuer and the Administrative Agent, and their respective directors, officers, agents and employees, from and against any and all claims and damages, losses, liabilities, costs or expenses which such Person may incur (or which may be claimed against such Person by any other Person whatsoever) by reason of or in connection with the issuance, execution and delivery or transfer of or payment or failure to pay under any Letter of Credit or any actual or proposed use of any Letter of Credit, including any claims, damages, losses, liabilities, costs or expenses which any Issuer may incur by reason of or in connection with (i) the failure of any other Lender to fulfill or comply with its obligations to such Issuer hereunder (but nothing herein contained shall affect any right the Borrower may have against any defaulting Lender) or (ii) by reason of or on account of such Issuer issuing any Letter of Credit which specifies that the term "Beneficiary" therein includes any successor by operation of law of the named Beneficiary, but which Letter of Credit does not require that any drawing by any such successor Beneficiary be accompanied by a copy of a legal document, satisfactory to such Issuer, evidencing the appointment of such successor Beneficiary; provided that the Borrower shall not be required to indemnify any Person for any claims, damages, losses, liabilities, costs or expenses to the extent, but only to the extent, caused by (x) the willful misconduct or gross negligence of any Issuer in determining whether a request presented under any Let ter of Credit issued by such Issuer complied with the terms of such Letter of Credit or (y) any Issuer's failure to pay under any Letter of Credit issued by it after the presentation to it of a request strictly complying with the

36

terms and conditions of such Letter of Credit. Nothing in this <u>Section 2.19(i)</u> is intended to limit the obligations of the Borrower under any other provision of this Agreement.

- (j) The Issuers' Obligation to Issue Letters of Credit. No Issuer shall be under any obligation to issue any Letter of Credit if:
- (i) any order, judgment or decree of any Governmental Authority or arbitrator shall by its terms purport to enjoin or restrain such Issuer from issuing such Letter of Credit, or any law applicable to such Issuer or any request or directive (whether or not having the force of law) from any Governmental Authority with jurisdiction over such Issuer shall prohibit, or request that such Issuer refrain from, the issuance of letters of credit generally or such Letter of Credit in particular or shall impose upon such Issuer with respect to such Letter of Credit any restriction, reserve or capital requirement (for which such Issuer is not otherwise compensated hereunder) not in effect on the Closing Date, or shall impose upon such Issuer any unreimbursed loss, cost or expense which was not applicable on the Closing Date and which such Issuer in good faith reasonably deems material to it; provided, however, that in the event a Lender participating in the Letters of Credit is not affected by any such restriction, requirement or imposition, and is able to issue such Letter of Credit and expressly agrees in its sole discretion to issue such Letter of Credit, such Lender, subject to the consent of the Administrative Agent, such consent not to be unreasonably withheld, conditioned or delayed, shall be added as an Issuer

pursuant to <u>Section 2.19(n)</u> and shall issue such Letter of Credit and shall be deemed the Issuer with regard to such Letter of Credit for all purposes of this Agreement;

- (ii) the issuance of such Letter of Credit would violate one or more policies of such Issuer applicable to letters of credit generally;
- (iii) except as otherwise agreed by the Administrative Agent and the applicable Issuer, such Letter of Credit is in an initial stated amount less than \$250,000;
 - (iv) such Letter of Credit is to be denominated in a currency other than Dollars; or
- (v) any Lender is at that time a Defaulting Lender, unless each Issuer has entered into arrangements, including the delivery of Cash Collateral, satisfactory to the Issuers (in their sole discretion) with the Borrower or such Lender to eliminate the Issuers' actual or potential Fronting Exposure (after giving effect to Section 2.23(a)(iv)) with respect to the Defaulting Lender arising from either the Letter of Credit then proposed to be issued or that Letter of Credit and all other Letter of Credit Obligations as to which the Issuers have actual or potential Fronting Exposure, as it may elect in their sole discretion.
- (k) <u>Rights as a Lender</u>. In its capacity as a Lender, each Issuer shall have the same rights and obligations as any other Lender.
- (l) <u>Letters of Credit Issued for Subsidiaries</u>. Notwithstanding that a Letter of Credit issued or outstanding hereunder is in support of any obligations of, or is for the account of, a

37

Subsidiary, the Borrower shall be obligated to reimburse the Issuers hereunder for any and all drawings under such Letter of Credit. The Borrower hereby acknowledges that the issuance of Letters of Credit for the account of Subsidiaries inures to the benefit of the Borrower, and that the Borrower's business derives substantial benefits from the businesses of such Subsidiaries.

- (m) <u>Applicability of ISP</u>. Unless otherwise expressly agreed by each Issuer and the Borrower when a Letter of Credit is issued, the rules of the ISP shall apply to each Letter of Credit.
- (n) Replacement or Addition of Issuer. Any Issuer may be replaced or added at any time by written agreement among the Borrower, the Administrative Agent (unless, in the case of the replacement of any Issuer, the successor Issuer is a Lender and, if applicable, such agreement not to be unreasonably withheld, conditioned or delayed) and the successor or additional Issuer, as applicable; provided that such new Issuer shall be assigned all or a portion, as applicable, of the Letter of Credit Commitment of another Issuer. The Administrative Agent shall notify the Lenders of any such replacement or addition, as applicable, of any Issuer. Where any Issuer is replaced, at the time such replacement shall become effective, the Borrower shall pay all unpaid fees accrued for account of the replaced Issuer. Furthermore, from and after the effective date of such replacement, the successor Issuer, shall have all the rights and obligations of the replaced Issuer under this Agreement with respect to Letters of Credit to be issued thereafter. References herein to the term "Issuer" shall be deemed to refer to any successor or additional Issuer, as applicable, or to any previous Issuer, or to any successor or additional Issuer, as applicable, and all previous Issuers, as the context shall require. After the replacement of any Issuer hereunder, the replaced Issuer shall remain a party hereto and shall continue to have all the rights and obligations of an Issuer under this Agreement with respect to Letters of Credit issued by it prior to such re placement, but shall not be required to issue additional Letters of Credit.

2.20 <u>Swing Line Loans</u>.

(a) Swing Line Facility. From and including the date of this Agreement and prior to the Facility Termination Date, subject to the terms and conditions set forth in this Agreement, the Swing Line Lender, in reliance upon the agreements of the other Lenders set forth in this Section 2.20, shall, unless (i) any Lender at such time is a Defaulting Lender and (ii) the Swing Line Lender has not entered into arrangements satisfactory to it with the Borrower or such Defaulting Lender to eliminate the Swing Line Lender's risk with respect to such Defaulting Lender, in which case the Swing Line Lender may in its discretion, make loans (each such loan, a "Swing Line Loan") to the Borrower in Dollars from time to time in an aggregate amount not to exceed at any time outstanding the amount of the Swing Line Sublimit; provided, however, that after giving effect to any Swing Line Loan, (i) the Aggregate Outstanding Credit Exposure shall not exceed the Aggregate Commitments, and (ii) the Outstanding Credit Exposure of any Lender (other than the Swing Line Lender) shall not exceed such Lender's Commitment, and provided, further, that the Borrower shall not use the proceeds of any Swing Line Loan to refinance any outstanding Swing Line Loan. Within the foregoing limits, and subject to the other terms and conditions hereof, the Borrower may borrow Swing Line Loans under this Section 2.20, prepay Swing Line Loans under Section 2.7, and reborrow Swing Line Loans under this Section 2.20. Each Swing Line Loan shall bear interest at the Alternate Base Rate plus the Applicable Margin.

38

Immediately upon the making of a Swing Line Loan, each Lender shall be deemed to, and hereby irrevocably and unconditionally agrees to, purchase from the Swing Line Lender a risk participation in such Swing Line Loan in an amount equal to such Lender's Pro Rata Share times the amount of such Swing Line Loan.

(b) <u>Borrowing Procedures</u>. Each Swing Line Loan shall be made upon the Borrower's irrevocable notice to the Swing Line Lender and the Administrative Agent, which may be given by telephone. Each such notice must be received by the Swing Line Lender and the Administrative Agent not later than 1:00 p.m. on the requested borrowing date, and shall specify (i) the amount to be borrowed, which shall be a minimum principal amount of \$500,000 and integral multiples of \$100,000 in excess thereof, and (ii) the requested borrowing date, which shall be a Business Day. Each such telephonic

notice must be confirmed promptly by delivery to the Swing Line Lender and the Administrative A gent of a written notice (the "Swing Line Loan Notice"). Promptly after receipt by the Swing Line Lender of any telephonic Swing Line Loan Notice, the Swing Line Lender will confirm with the Administrative Agent (by telephone or in writing) that the Administrative Agent has also received such Swing Line Loan Notice and, if not, the Swing Line Lender will notify the Administrative Agent (by telephone or in writing) of the contents thereof. Unless the Swing Line Lender has received notice (by telephone or in writing) from the Administrative Agent (including at the request of any Lender) prior to 2:00 p.m. on the date of the proposed Advance of Swing Line Loans (A) directing the Swing Line Lender not to make such Swing Line Loan as a result of the limitations set forth in the first proviso to the first sentence of Section 2.20(a), or (B) that one or more of the applicable conditions specified in Article V is not then satisfied, then, subject to the terms and conditions hereof, the Swing Line Lender will, not later than 3:00 p.m. on the borrowing date specified in such Swing Line Loan Notice, make the amount of its Swing Line Loan available to the Borrower by means of a credit to the general deposit account of the Borrower with the Swing Line Lender or such other account specified by the Borrower to the Swing Line Lender.

(c) <u>Refinancing of Swing Line Loans</u>.

(i) The Swing Line Lender at any time in its sole and absolute discretion may request, on behalf of the Borrower (which hereby irrevocably requests and authorizes the Swing Line Lender to so request on its behalf), that each Lender make a Floating Rate Loan in an amount equal to such Lender's Pro Rata Share of the amount of Swing Line Loans then outstanding. Such request shall be made in writing (which written request shall be deemed to be a Borrowing Notice for purposes hereof) and in accordance with the requirements of Section 2.8 without regard to the minimum and multiples specified in Section 2.4 and provided that, after giving effect to such Advance, the Aggregate Outstanding Credit Exposure shall not exceed the Aggregate Commitments. The Swing Line Lender shall furnish the Borrower with a copy of the applicable Loan notice promptly after delivering such notice to the Administrative Agent. Each Lender shall make an amount equal to its Pro Rata Share of the amount specified in such Loan notice available to the Administrative Agent in immediately available funds (and the Administrative Agent may apply Cash Collateral available with respect to the applicable Swing Line Loan) for the account of the Swing Line Lender at the Administrative Agent's Office not later than 1:00 p.m. on the day specified in such Loan notice,

39

whereupon, subject to <u>Section 2.20(c)(ii)</u>, each Lender that so makes funds available shall be deemed to have made a Floating Rate Loan to the Borrower in such amount. The Administrative Agent shall remit the funds so received to the Swing Line Lender.

- (ii) If for any reason any Swing Line Loan cannot be refinanced by such a Advance of Loans in accordance with <u>Section 2.20(c)(i)</u>, the request for Floating Rate Loans submitted by the Swing Line Lender as set forth herein shall be deemed to be a request by the Swing Line Lender that each of the Lenders fund its risk participation in the relevant Swing Line Loan and each Lender's payment to the Administrative Agent for the account of the Swing Line Lender pursuant to <u>Section 2.20(c)(i)</u> shall be deemed payment in respect of such participation.
- (iii) If any Lender fails to make available to the Administrative Agent for the account of the Swing Line Lender any amount required to be paid by such Lender pursuant to the foregoing provisions of this Section 2.20(c) by the time specified in Section 2.20(c)(i), the Swing Line Lender shall be entitled to recover from such Lender (acting through the Administrative Agent), on demand, such amount with interest thereon for the period from the date such payment is required to the date on which such payment is immediately available to the Swing Line Lender at a rate per annum equal to the greater of the Federal Funds Rate and a rate determined by the Swing Line Lender in accordance with banking industry rules on interbank compensation. A certificate of the Swing Line Lender submitted to any Lender (through the Administrative Agent) with respect to any amounts owing under this clause (iii) shall be conclusive absent manifest error.
- (iv) Each Lender's obligation to make Loans or to purchase and fund risk participations in Swing Line Loans pursuant to this Section 2.20(c) shall be absolute and unconditional and shall not be affected by any circumstance, including (A) any setoff, counterclaim, recoupment, defense or other right that such Lender may have against the Swing Line Lender, the Borrower or any other Person for any reason whatsoever, (B) the occurrence or continuance of a Default or Unmatured Default, or (C) any other occurrence, event or condition, whether or not similar to any of the foregoing; provided, however, that each Lender's obligation to make Loans pursuant to this Section 2.20(c) is subject to the conditions set forth in Section 5.2. No such purchase or funding of risk participations shall relieve or otherwise impair the obligation of the Borrower to repay Swing Line Loans, together with interest as provided herein.

(d) <u>Repayment of Participations.</u>

(i) At any time after any Lender has purchased and funded a risk participation in a Swing Line Loan, if the Swing Line Lender receives any payment on account of such Swing Line Loan, the Swing Line Lender will distribute to such Lender its Pro Rata Share of such payment (appropriately adjusted, in the case of interest payments, to reflect the period of time during which such Lender's risk participation was funded) in the same funds as those received by the Swing Line Lender.

40

(ii) If any payment received by the Swing Line Lender in respect of principal or interest on any Swing Line Loan is required to be returned by the Swing Line Lender under any of the circumstances described in Section 12.3 (including pursuant to any settlement entered into by the Swing Line Lender in its discretion), each Lender shall pay to the Swing Line Lender its Pro Rata Share thereof on demand of the Administrative Agent, plus interest thereon from the date of such demand to the date such amount is returned, at a rate per annum equal to the Federal Funds Rate. The Administrative Agent will make such demand upon the request of the Swing Line Lender. The obligations of the Lenders under this clause shall survive the payment in full of the Obligations and the termination of this Agreement.

- (e) <u>Interest for Account of Swing Line Lender</u>. The Swing Line Lender shall be responsible for invoicing the Borrower for interest on the Swing Line Loans. Until each Lender funds its Floating Rate Loans or risk participation pursuant to this <u>Section 2.20</u> to refinance such Lender's Pro Rata Share of any Swing Line Loan, interest in respect of such Applicable Percentage shall be solely for the account of the Swing Line Lender.
- (f) <u>Payments Directly to Swing Line Lender</u>. The Borrower shall make all payments of principal and interest in respect of the Swing Line Loans directly to the Swing Line Lender.

2.21 <u>Extension of Facility Termination Date</u>.

- (a) Request for Extension. The Borrower may by written notice to the Administrative Agent (who shall promptly notify the Lenders) given not more than sixty (60) days and not less than forty-five (45) days prior to any anniversary of the Closing Date, request that each Lender extend the Facility Termination Date for an additional one (1) year from the then existing Facility Termination Date; provided, that the Borrower shall only be permitted to exercise this extension option two (2) times during the term of the Agreement.
- (b) <u>Lenders Election to Extend</u>. Each Lender, acting in its sole and individual discretion, shall, by notice to the Administrative Agent given not later than fifteen (15) days following the receipt of notice of such request from the Administrative Agent (the "<u>Notice Date</u>"), advise the Administrative Agent in writing whether or not such Lender agrees to such extension (and each Lender that determines not to so extend its Facility Termination Date (a "<u>Non-Extending Lender</u>") shall notify the Administrative Agent of such fact promptly after such determination (but in any event no later than the Notice Date) and any Lender that does not so advise the Administrative Agent on or before the Notice Date shall be deemed to be a Non-Extending Lender. The election of any Lender to agree to such extension shall not obligate any other Lender to so agree.
- (c) <u>Notification by Administrative Agent</u>. The Administrative Agent shall notify the Borrower of each Lender's determination under this Section promptly and in any event no later than the date fifteen (15) days after the Notice Date (or, if such date is not a Business Day, on the next preceding Business Day).

41

- (d) <u>Additional Commitment Lenders</u>. The Borrower shall have the right on or before the applicable anniversary of the Closing Date to replace each Non-Extending Lender with, and add as "Lenders" under this Agreement in place thereof, one or more Eligible Assignees (each, an "<u>Additional Commitment Lender</u>") as provided in <u>Section 13.2</u>, each of which Additional Commitment Lenders shall have entered into an Assignment Agreement pursuant to which such Additional Commitment Lender shall, undertake a Commitment (and, if any such Additional Commitment Lender is already a Lender, its Commitment shall be in addition to such Lender's Commitment hereunder on such date) and shall be a "Lender" for all purposes of this Agreement.
- (e) Minimum Extension Requirement. If all of the Lenders agree to any such request for extension of the Facility Termination Date then the Facility Termination for all Lenders shall be extended for the additional one (1) year, as applicable. If there exists any Non-Extending Lenders then the Borrower shall (i) withdraw its extension request and the Facility Termination Date will remain unchanged or (ii) provided that the Required Lenders (but for the avoidance of doubt, not including any Additional Commitment Lenders) have agreed to the extension request (such Lenders agreeing to such extension, the "Approving Lenders & #8221;) no later than fifteen (15) days prior to such anniversary of the Closing Date, then the Borrower may extend the Facility Termination Date solely as to the Approving Lenders and the Additional Commitment Lenders with a reduced amount of Aggregate Commitments during such extension period equal to the aggregate Commitments of the Approving Lenders and the Additional Commitment Lenders so long as the aggregate amount of the Commitments of the Approving Lenders and the Additional Commitment Lenders); it being understood that (A) the Facility Termination Date relating to any Non-Extending Lenders not replaced by an Additional Commitment Lender shall not be extended and the repayment of all obligations owed to them and the termination of their Commitments shall occur on the already existing Facility Termination Date and (B) the Facility Termination Date relating to the Approving Lenders and the Additional Commitment Lenders shall be extended for an additional year, as applicable.
- (f) <u>Conditions to Effectiveness of Extensions</u>. Notwithstanding the foregoing, any extension of the Facility Termination Date pursuant to this Section shall not be effective with respect to any Lender unless:
 - (i) on the date of such extension the conditions for a Credit Extension provided in <u>Section 5.2(a)</u>, <u>Section 5.2(b)</u> and <u>Section 5.2(c)</u> shall be satisfied; and
 - (ii) on any Facility Termination Date, the Borrower shall prepay any Loans outstanding on such date (and pay any additional amounts required pursuant to Section 3.4) to the extent necessary to keep outstanding Loans ratable with any revised Pro Rata Shares of the respective Lenders effective as of such date.

2.22 <u>Cash Collateral</u>.

(a) <u>Certain Credit Support Events</u>. Upon the request of the Administrative Agent or any of the Issuers (i) if any Issuer has honored any full or partial drawing request under any Letter of Credit and such drawing has not been reimbursed by the Borrower on the Letter of

42

Credit Payment Date or (ii) if, as of the Facility Termination Date, any Letter of Credit Obligation for any reason remains outstanding and partially or wholly drawn, the Borrower shall, in each case, immediately Cash Collateralize the then outstanding amount of all Letter of Credit Obligations. At any time that there shall exist a Defaulting Lender, immediately upon the request of the Administrative Agent, the Issuers or the Swing Line Lender, the Borrower shall

deliver to the Administrative Agent Cash Collateral in an amount sufficient to cover all Fronting Exposure, if any (after giving effect to Section 2.23(a)(iv) and any Cash Collateral provided by the Defaulting Lender).

- (b) Grant of Security Interest. All Cash Collateral (other than credit support not constituting funds subject to deposit) shall be maintained in blocked, interest bearing deposit accounts at Bank of America (the "LC Collateral Accounts"). The Borrower, and to the extent provided by any Lender, such Lender, hereby grants to (and subjects to the control of) the Administrative Agent, for the benefit of the Administrative Agent, the Issuers and the Lenders (including the Swing Line Lender), and agrees to maintain, a first priority security interest in all such cash, deposit accounts and all balances therein, and all other property so provided as collateral pursuant hereto, and in all proceeds of the foregoing, all as security for the obligations to which such Cash Collateral may be applied pursuant to Section 2.22(c). If at any time the Administrative Agent determines that Cash Collateral is less than the applicable Fronting Exposure and other obligations secured thereby, the Borrower or the relevant Defaulting Lender will, promptly upon written demand by the Administrative Agent, pay or provide to the Administrative Agent additional Cash Collateral in an amount sufficient to eliminate such deficiency.
- (c) <u>Application</u>. Notwithstanding anything to the contrary contained in this Agreement, Cash Collateral provided under any of this <u>Section 2.22</u> or <u>Sections 2.19</u>, <u>2.20</u>, <u>2.20</u>, <u>2.23</u> or <u>9.1</u> in respect of Letters of Credit or Swing Line Loans shall be held and applied to the satisfaction of the specific Letter of Credit Obligations, Swing Lin e Loans, obligations to fund participations therein (including, as to Cash Collateral provided by a Defaulting Lender, any interest accrued on such obligation) and other obligations for which the Cash Collateral was so provided, prior to any other application of such property as may be provided for herein.
- (d) Release. Cash Collateral (or the appropriate portion thereof) provided to reduce Fronting Exposure or other obligations, together with all interest accrued thereon, shall be released promptly following (i) the elimination of the applicable Fronting Exposure or other obligations giving rise thereto (including by the termination of Defaulting Lender status of the applicable Lender (or, as appropriate, its assignee following compliance with Section 13.1(b) (v))) or (ii) the Administrative Agent's good faith determination that there exists excess Cash Collateral; provided, however, (x) that Cash Collateral furnished by or on behalf of the Borrower shall not be released during the continuance of an Unmatured Default or a Default, and (y) the Person providing Cash Collateral and the Issuers or Swing Line Lender, as applicable, may agree, in their sole discretion, that Cash Collateral shall not be released but instead held to support future anticipated Fronting Exposure or other obligations.

43

2.23 <u>Defaulting Lenders</u>.

- (a) <u>Adjustments</u>. Notwithstanding anything to the contrary contained in this Agreement, if any Lender becomes a Defaulting Lender, then, until such time as that Lender is no longer a Defaulting Lender, to the extent permitted by applicable Law:
 - (i) <u>Waivers and Amendments</u>. That Defaulting Lender's right to approve or disapprove any amendment, waiver or consent with respect to this Agreement shall be restricted as set forth in <u>Section 9.2</u>.
 - Reallocation of Payments. Any payment of principal, interest, fees or other amounts received by the Administrative Agent for the account of that Defaulting Lender under the Loan Documents (whether voluntary or mandatory, at maturity, pursuant to Article IX or otherwise, and including any amounts made available to the Administrative Agent by that Defaulting Lender pursuant to Section 12.1), shall be applied at such time or times as may be reasonably determined by the Administrative Agent as follows: first, to the payment of any amounts owing by that Defaulting Lender to the Administrative Agent hereunder; second, to the payment on a pro rata basis of any amounts owing by that Defaulting Lender to the Issuers or Swing Line Lender hereunder; third, if so reasonably determined by the Administrative Agent or requested by the Issuers or Swing Line Lender, to be held as Cash Collateral for future funding obligations of that Defaulting Lender of any participation in any Swing Line Loan or Letter of Credit; fourth, as the Borrower may request (so long as no Default exists), to the funding of any Loan in respect of which that Defaulting Lender has failed to fund its portion thereof as required by this Agreement, as determined by the Administrative Agent; fifth, if so reasonably determined by the Administrative Agent and the Borrower, to be held in a non-interest bearing deposit account and released in order to satisfy obligations of that Defaulting Lender to fund Loans under this Agreement; sixth, to the payment of any amounts owing to the Lenders, the Issuers or Swing Line Lender as a result of any judgment of a court of competent jurisdiction obtained by any Lender, the Issuers or Swing Line Lender against that Defaulting Lender as a result of that Defaulting Lender's breach of its obligations under this Agreement; seventh, so long as no Default exists, to the payment of any amounts owing to the Borrower as a result of any judgment of a court of competent jurisdiction obtained by the Borrower against that Defaulting Lender as a result of that Defaulting Lender's breach of its obligations under this Agreement; and eighth, to that Defaulting Lender or as otherwise directed by a court of competent jurisdiction; provided that if (x) such payment is a payment of the principal amount of any Loans or a reimbursement of any payment on a Letter of Credit, in respect of which that Defaulting Lender has not fully funded its appropriate share and (y) such Loans or reimbursement of payment on a Letter of Credit were made at a time when the conditions set forth in Section 5.2 were satisfied or waived, such payment shall be applied solely to pay the Loans of, and such Letter of Credit reimbursements owed to, all non-Defaulting Lenders on a pro rata basis prior to being applied to the payment of any Loans of, or such Letter of Credit reimbursements owed to, that Defaulting Lender. Any pay ments, prepayments or other amounts paid or payable to a Defaulting Lender that are applied (or held) to pay amounts owed by a Defaulting Lender or to post Cash Collateral pursuant to

44

this Section 2.23(a)(ii) shall be deemed paid to and redirected by that Defaulting Lender, and each Lender irrevocably consents hereto.

(iii) <u>Certain Fees.</u> That Defaulting Lender (x) shall not be entitled to receive any commitment fee pursuant to <u>Section 2.5</u> for any period during which that Lender is a Defaulting Lender (and the Borrower shall not be required to pay any such fee that otherwise would have been required to have been paid to that Defaulting Lender) and (y) shall be limited in its right to receive Letter of Credit Fees as provided in <u>Section 2.19(d)</u>.

- (iv) Reallocation of Pro Rata Shares to Reduce Fronting Exposure. During any period in which there is a Defaulting Lender, for purposes of computing the amount of the obligation of each non-Defaulting Lender to acquire, refinance or fund participations in Letters of Credit or Swing Line Loans pursuant to Sections 2.19 and 2.20, the "Pro Rata Share" of each non-Defaulting Lender shall be computed without giving effect to the Commitment of that Defaulting Lender; provided, that, (i) each such reallocation shall be given effect only if, at the date the applicable Lender becomes a Defaulting Lender, no Default exists; and (ii) the aggregate obligation of each non-Defaulting Lender to acquire, refinance or fund participations in Letters of Credit and Swing Line Loans shall not exceed the positive difference, if any, of (1) the Commitment of that non-Defaulting Lender minus (2) the Outstanding Credit Exposure of that Lender.
- (b) <u>Defaulting Lender Cure</u>. If the Borrower, the Administrative Agent, the Swing Line Lender and the Issuers agree in writing in their reasonable discretion that a Defaulting Lender should no longer be deemed to be a Defaulting Lender, the Administrative Agent will so notify the parties hereto, whereupon as of the effective date specified in such notice and subject to any conditions set forth therein (which may include arrangements with respect to any Cash Collateral), that Lender will, to the extent applicable, purchase that portion of outstanding Loans of the other Lenders or take such other actions as the Administrative Agent may determine to be necessary to cause the Loans and funded and unfunded participations in Letters of Credit and Swing Line Loans to be held on a pro rata basis by the Lenders in accordance with their Pro Rata Shares (without giving effect to Section 2.23(a)(iv)), whereupon that Lender will cease to be a Defaulting Lender; provided that no adjustments will be made retroactively with respect to fees accrued or payments made by or on behalf of the Borrower while that Lender was a Defaulting Lender; and provided, further, that except to the extent otherwise expressly agreed by the affected parties, no change hereunder from Defaulting Lender to Lender will constitute a waiver or release of any claim of any party hereunder arising from that Lender's having been a Defaulti ng Lender.

ARTICLE III

YIELD PROTECTION; TAXES

3.1 Yield Protection.

If, on or after the date of this Agreement, the adoption of any law or any governmental or quasi-governmental rule, regulation, policy, guideline or directive (whether or not having the

45

force of law), or any change which occurs after the date of this Agreement in the interpretation or administration thereof by any governmental or quasi-governmental authority, central bank or comparable agency charged with the interpretation or administration thereof, or compliance by any Lender, any applicable Lending Installation or any Issuer with any request or directive (whether or not having the force of law) of any such authority, central bank or comparable agency imposed after the date of this Agreement:

- (i) subjects any Lender to any tax of any kind whatsoever with respect to this Agreement, any Letter of Credit, any participation in a Letter of Credit or any Eurodollar Loan made by it, or change the basis of taxation of payments to such Lender in respect thereof (except for Indemnified Taxes or Other Taxes covered by Section 3.5 and the imposition of, or any change in the rate of, any Excluded Tax), or
- (ii) imposes or increases or deems applicable any reserve, assessment, insurance charge, special deposit or similar requirement against assets of, deposits with or for the account of, or credit extended by, any Lender, any applicable Lending Installation or any Issuer (other than reserves and assessments taken into account in determining the interest rate applicable to Eurodollar Advances), or
- (iii) imposes any other condition the result of which is to increase the cost to any Lender, any applicable Lending Installation or any Issuer of making, funding or maintaining its Eurodollar Loans or of issuing or participating in Letters of Credit or reduces any amount receivable by any Lender, any applicable Lending Installation or any Issuer in connection with its Eurodollar Loans or Letters of Credit, or requires any Lender, any applicable Lending Installation or any Issuer to make any payment calculated by reference to the amount of Eurodollar Loans or Letters of Credit held or interest received by it, by an amount deemed material by such Lender or such Issuer, as the case may be,

and the result of any of the foregoing is to increase the cost to such Lender, the applicable Lending Installation or such Issuer of making or maintaining its Eurodollar Loans, Letters of Credit or Commitment or to reduce the return received by such Lender, the applicable Lending Installation or such Issuer in connection with such Eurodollar Loans, Letters of Credit or Commitment, then, within fifteen (15) Business Days of written demand by such Lender or such Issuer, the Borrower shall pay such Lender or such Issuer such additional amount or amounts as will compensate such Lender or such Issuer for such increased cost or reduction in amount received.

3.2 Changes in Capital Adequacy Regulations.

If a Lender or an Issuer determines the amount of capital required or expected to be maintained by such Lender, any Lending Installation of such Lender, such Issuer or any corporation controlling such Lender or such Issuer is increased as a result of a Change, then, within thirty (30) days of written demand by such Lender or such Issuer, the Borrower shall pay such Lender or such Issuer the amount necessary to compensate for any shortfall in the rate of return on the portion of such increased capital which such Lender or such Issuer determines is

46

attributable to this Agreement, its Outstanding Credit Exposure or its Commitment to make Loans or to issue or participate in Letters of Credit hereunder (after taking into account such Lender's policies as to capital adequacy). "Change" means (i) any change after the date of this Agreement in (or in the interpretation of) the Risk-Based Capital Guidelines or (ii) any adoption of or change in (or any change in the interpretation of) any other law, governmental

or quasi-governmental rule, regulation, policy, guideline, interpretation, or directive (whether or not having the force of law) after the date of this Agreement which affects the amount of capital required or expected to be maintained by any Lender, any Lending Installation, any Issuer or any corporation controlling any Lender or any Issuer. "Risk-Based Capital Guidelines" means (x) the risk-based capital guidelines in effect in the United States on the date of this Agreement, including transition rules, and (y) the corresponding capital regulations promulgated by regulatory authorities outside the United States implementing the July 1988 report of the Basle Committee on Banking Regulation and Supervisory Practices Entitled "International Convergence of Capital Measurements and Capital Standards," including transition rules, and any amendments to such regulations adopted prior to the date of this Agreement.

3.3 **Availability of Types of Advances.**

If (i) any Lender determines that maintenance of its Eurodollar Loans at a suitable Lending Installation would violate any applicable law, rule, regulation, or directive, whether or not having the force of law, enacted or imposed after the date of this Agreement, (ii) the Required Lenders determine that (a) deposits of a type and maturity appropriate to match fund Eurodollar Advances are not available or (b) the interest rate applicable to a Type of Advance does not accurately reflect the cost of making or maintaining such Advance or (iii) the Administrative Agent determines that adequate and reasonable means do not exist for determining the Eurodollar Base Rate, then the Administrative Agent shall suspend the availability of the affected Type of Advance and, in the case of clause (i), require any affected Eurodollar Advances to be repaid or converted to Floating Rate Advances, subject to the payment of any funding indemnification amounts required by Section 3.4.

3.4 <u>Funding Indemnification</u>.

If any conversion, prepayment or payment of a Eurodollar Advance occurs on a date which is not the last day of the applicable Interest Period, whether because of acceleration, prepayment or otherwise, or a Eurodollar Advance is not made, paid, continued or converted on the date or in the amount specified by the Borrower for any reason other than default by the Lenders, the Borrower will, within fifteen (15) Business Days of written demand by any Lender, indemnify such Lender for any loss or cost incurred by it resulting therefrom, including any loss or cost in liquidating or employing deposits acquired to fund or maintain such Eurodollar Advance; <u>provided</u> that each such Lender shall have delivered to the Borrower a certificate as to the amount of such loss or cost, setting forth in reasonable detail the calculation thereof, which certificate shall be conclusive in the absence of manifest error.

47

3.5 <u>Taxes</u>.

- (a) <u>Payments Free of Taxes; Obligation to Withhold; Payments on Account of Taxes</u>. (i) Any and all payments by or on account of any Obligation of the Borrower hereunder or under any other Loan Document shall to the extent permitted by applicable Laws be made free and clear of and without reduction or withholding for any Taxes. If, however, applicable Laws require the Borrower or the Administrative Agent to withhold or deduct any Tax, such Tax shall be withheld or deducted in accordance with such Laws as determined by the Borrower or the Administrative Agent, as the case may be, upon the basis of the information and documentation to be delivered pursuant to subsection (e) below.
- (ii) If the Borrower or the Administrative Agent shall be required by the Code to withhold or deduct any Taxes, including both United States Federal backup withholding and withholding taxes, from any payment, then (A) the Borrower or the Administrative Agent, as the case may be, shall withhold or make such deductions as are determined by the Borrower or the Administrative Agent, as the case may be, to be required based upon the information and documentation it has received pursuant to subsection(e) below, (B) the Borrower or the Administrative Agent, as the case may be, shall timely pay the full amount withheld or deducted to the relevant Governmental Authority in accordance with the Code, and (C) to the extent that the withholding or deduction is made on account of Indemnified Taxes or Other Taxes, the sum payable by the Borrower shall be increased as necessary so that after any required withholding or the making of all required deductions (including deductions applicable to additional sums payable under this Section) the Administrative Agent or Lender, as the case may be, receives an amount equal to the sum it would have received had no such withholding or deduction of Indemnified Taxes or Other Taxes been made.
- (b) <u>Payment of Other Taxes by the Borrower</u>. Without limiting the provisions of <u>subsection (a)</u> above, the Borrower shall timely pay any Other Taxes to the relevant Governmental Authority in accordance with applicable Laws.
- (c) Tax Indemnifications. (i) Without limiting the provisions of subsection (a) or (b) above, the Borrower shall, and does hereby, indemnify the Administrative Agent and each Lender, and shall make payment in respect thereof within ten (10) days after demand therefor, for the full amount of any Indemnified Taxes or Other Taxes (including Indemnified Taxes or Other Taxes imposed or asserted on or attributable to amounts payable under this Section but excluding Indemnified Taxes or Other Taxes paid by the Borrower to the Administrative Agent pursuant to Section 3.5(a)(ii)(C)) withheld or deducted by the Borrower or the Administrative Agent or paid by the Administrative Agent or such Lender, as the case may be, and any penalties, interest and reasonable expenses arising therefrom or with respect thereto, whether or not such Indemnified Taxes or Other Taxes were correctly or legally imposed or asserted by the relevant Governmental Authority. The Borrower shall also, and does hereby, indemnify the Administrative Agent, and shall make payment in respect thereof within ten (10) days after demand therefor, for any amount which a Lender for any reason fails to pay indefeasibly to the Administrative Agent as required by clause (ii) of this subsection. A certificate as to the amount of any such payment or liability delivered to the Borrower by a Lender (with a copy to the Administrative Agent), or by the Administrative Agent on its own behalf or on behalf of a Lender, shall be conclusive absent manifest error.

48

(ii) Without limiting the provisions of <u>subsection (a)</u> or (b) above, each Lender shall, and does hereby, indemnify the Borrower and the Administrative Agent, and shall make payment in respect thereof within ten (10) days after demand therefor, against any and all Taxes and any and all related losses, claims, liabilities, penalties, interest and expenses (including the fees, charges and disbursements of any counsel for the Borrower or the Administrative Agent) incurred by or asserted against the Borrower or the Administrative Agent by any Governmental Authority as a result of the failure by such Lender to deliver, or as a result of the inaccuracy, inadequacy or deficiency of, any documentation required to be

delivered by such Lender to the Borrower or the Administrative Agent pursuant to <u>subsection (e)</u>. Each Lender hereby authorizes the Administrative Agent to set off and apply any and all amounts at any time owing to such Lender under this Agreement or any other Loan Document against any amount due to the Administrative Agent under this <u>clause (ii)</u>. The agreements in this <u>clause (ii)</u> shall survive the resignation and/or replacement of the Administrative Agent, any assignment of rights by, or the replacement of, a Lender, the termination of the Aggregate Commitments and the repayment, satisfaction or discharge of all other Obligations.

- (d) <u>Evidence of Payments</u>. Upon request by the Borrower or the Administrative Agent, as the case may be, after any payment of Taxes by the Borrower or the Administrative Agent to a Governmental Authority as provided in this <u>Section 3.5</u>, the Borrower shall deliver to the Administrative Agent or the Administrative Agent shall deliver to the Borrower, as the case may be, the original or a certified copy of a receipt issued by such Governmental Authority evidencing such payment or other evidence of such payment reasonably satisfactory to the Borrower or the Administrative Agent, as the case may be.
- (e) <u>Status of Lenders; Tax Documentation</u>. (i) Each Lender shall deliver to the Borrower and to the Administrative Agent, at the time or times prescribed by applicable Laws or when reasonably requested by the Borrower or the Administrative Agent, such properly completed and executed documentation prescribed by applicable Laws or by the taxing authorities of any jurisdiction and such other reasonably requested information as will permit the Borrower or the Administrative Agent, as the case may be, to determine (A) whether or not payments made hereunder or under any other Loan Document are subject to Taxes, (B) if applicable, the required rate of withholding or deduction, and (C) such Lender's entitlement to any available exemption from, or reduction of, applicable Taxes in respect of all payments to be made to such Lender by the Borrower pursuant to this Agreement or otherwise to establish such Lender's status for withholding tax purposes in the applicable jurisdiction.
 - (ii) Without limiting the generality of the foregoing, if the Borrower is resident for tax purposes in the United States,
 - (A) any Lender that is a "United States person" within the meaning of Section 7701(a)(30) of the Code shall deliver to the Borrower and the Administrative Agent executed originals of Internal Revenue Service Form W-9 or such other documentation or information prescribed by applicable Laws or reasonably requested by the Borrower or the Administrative Agent as will enable the Borrower or the Administrative Agent, as the case may be, to determine whether or not such Lender is subject to backup withholding or information reporting requirements; and

49

- (B) each Foreign Lender that is entitled under the Code or any applicable treaty to an exemption from or reduction of withholding tax with respect to payments hereunder or under any other Loan Document shall deliver to the Borrower and the Administrative Agent (in such number of copies as shall be requested by the recipient) on or prior to the date on which such Foreign Lender becomes a Lender under this Agreement (and from time to time thereafter upon the request of the Borrower or the Administrative Agent, but only if such Foreign Lender is legally entitled to do so), whichever of the following is applicable:
 - (I) duly completed and executed originals of Internal Revenue Service Form W-8BEN claiming eligibility for benefits of an income tax treaty to which the United States is a party,
 - (II) duly completed and executed originals of Internal Revenue Service Form W-8ECI,
 - (III) duly completed and executed originals of Internal Revenue Service Form W-8IMY and all required supporting documentation,
 - (IV) in the case of a Foreign Lender claiming the benefits of the exemption for portfolio interest under section 881(c) of the Code, (x) a certificate to the effect that such Foreign Lender is not (A) a "bank" within the meaning of section 881(c)(3)(A) of the Code, (B) a "10 percent shareholder" of the Borrower within the meaning of section 881(c)(3)(B) of the Code, or (C) a "controlled foreign corporation" described in section 881(c)(3)(C) of the Code and (y) duly completed and executed originals of Internal Revenue Service Form W-8BEN, or
 - (V) duly completed and executed originals of any other form prescribed by applicable Laws as a basis for claiming exemption from or a reduction in United States Federal withholding tax together with such supplementary documentation as may be prescribed by applicable Laws to permit the Borrower or the Administrative Agent to determine the withholding or deduction required to be made.
- (iii) Each Lender shall promptly (A) notify the Borrower and the Administrative Agent of any change in circumstances which would modify or render invalid any claimed exemption or reduction, and (B) take such steps as shall not be materially disadvantageous to it, in the reasonable judgment of such Lender, and as may be reasonably necessary (including the re-designation of its lending office) to avoid any requirement of applicable Laws of any jurisdiction that the Borrower or the Administrative Agent make any withholding or deduction for Taxes from amounts payable to such Lender.
- (iv) Each Foreign Lender shall provide such documentation or information prescribed by applicable Laws or reasonably requested by the Borrower or the Administrative Agent as will enable the Borrower or the Administrative Agent, as the

(f) Treatment of Certain Refunds. Unless required by applicable Laws, at no time shall the Administrative Agent have any obligation to file for or otherwise pursue on behalf of a Lender, or have any obligation to pay to any Lender, any refund of Taxes withheld or deducted from funds paid for the account of such Lender. If the Administrative Agent or any Lender determines, in its sole discretion, that it has received a refund of any Indemnified Taxes or Other Taxes as to which it has been indemnified by the Borrower or with respect to which the Borrower has paid additional amounts pursuant to this Section, it shall pay to the Borrower an amount equal to such refund (but only to the extent of indemnity payments made, or additional amounts paid, by the Borrower under this Section with respect to the Indemnified Taxes or Other Taxes giving rise to such refund), net of all out-of-pocket expenses incurred by the Administrative Agent or such Lender and without interest (other than any interest paid by the relevant Governmental Authority with respect to such refund), provided that the Borrower, upon the request of the Administrative Agent or such Lender in the event the Administrative Agent or such Lender is required to repay such refund to such Governmental Authority. This subsection shall not be construed to require the Borrower, the Administrative Agent or any Lender to make available its tax returns (or any other information relatin g to its Taxes that it deems confidential) to the Borrower or any other Person.

3.6 <u>Lender Statements; Survival of Indemnity.</u>

To the extent reasonably possible and upon the request of the Borrower, each Lender or Issuer shall designate an alternate Lending Installation to the extent it would reduce any liability of the Borrower to such Lender or Issuer under Sections 3.1, 3.2 and 3.5 or to avoid the unavailability of Eurodollar Advances under Section 3.3, so long as such designation is not, in the judgment of such Lender or Issuer, disadvantageous to such Lender or Issuer. Each Lender or each Issuer, as applicable, shall deliver a written statement of such Lender or such Issuer to the Borrower (with a copy to the Administrative Agent) as to any amount due under Section 3.1, 3.2, 3.4 or 3.5 and containing the other information contemplated by such Sections. Such written statement shall set forth in reasonable detail the calculations upon which such Lender or such Issuer determined such amount and shall be final, conclusive and binding on the Borrower in the absence of manifest error. Determination of amounts payable under such Sections in connection with a Eurodollar Loan shall be calculated as though each Lender funded its Eurodollar Loan through the purchase of a deposit of the type and maturity corresponding to the deposit used as a reference in determining the Eurodollar Rate applicable to such Loan , whether in fact that is the case or not. Unless otherwise provided herein, the amount specified in the written statement of any Lender or any Issuer shall be payable on demand after receipt by the Borrower of such written statement. The obligations of the Borrower under Sections 3.1, 3.2, 3.4 and 3.5 shall survive payment of the Obligations and termination of this Agreement.

51

ARTICLE IV

GUARANTY

4.1 <u>The Guaranty</u>.

The Guarantor hereby guarantees to each Lender, each Issuer, each Affiliate of a Lender that enters into a Swap Contract with the Borrower, and the Administrative Agent as hereinafter provided, as primary obligor and not as surety, the prompt payment of the Obligations in full when due (whether at stated maturity, as a mandatory prepayment, by acceleration, as a mandatory cash collateralization or otherwise) strictly in accordance with the terms thereof. The Guarantor hereby further agrees that if any of the Obligations are not paid in full when due (whether at stated maturity, as a mandatory prepayment, by acceleration, as a mandatory cash collateralization or otherwise), the Guarantor will promptly pay the same, without any demand or notice whatsoever, and that in the case of any extension of time of payment or renewal of any of the Obligations, the same will be promptly paid in full when due (whether at extended maturity, as a mandatory prepayment, by acceleration, as a mandatory cash collateralization or otherwise) in accordance with the terms of such extension or renewal.

Notwithstanding any provision to the contrary contained herein or in any other of the Loan Documents or Swap Contracts, the obligations of the Guarantor under this Agreement and the other Loan Documents and the Swap Contracts shall be limited to an aggregate amount equal to the largest amount that would not render such obligations subject to avoidance under the federal bankruptcy laws or any comparable provisions of any applicable state law.

4.2 Obligations Unconditional.

The obligations of the Guarantor under Section 4.1 are absolute and unconditional, irrespective of the value, genuineness, validity, regularity or enforceability of any of the Loan Documents, Swap Contracts, or any other agreement or instrument referred to therein, or any substitution, release, impairment or exchange of any other guarantee of or security for any of the Obligations, and, to the fullest extent permitted by applicable law, irrespective of any law or regulation or other circumstance whatsoever which might otherwise constitute a legal or equitable discharge or defense of a surety or guarantor, it being the intent of this Section 4.2 that the obligations of the Guarantor hereunder shall be absolute and unconditional under any and all circumstances. The Guarantor agrees that it shall have no right of subrogation, indemnity, reimbursement or contribution against the Borrower for amounts paid under this Article IV until such time as the Obligations (other than contingent indemnification obligations that survive the termination of this Agreement) have been paid in full and the Commitments have expired or terminated. Without limiting the generality of the foregoing, it is agreed that, to the fullest extent permitted by law, the occurrence of any one or more of the following shall not alter or impair the liability of the Guarantor hereunder, which shall remain absolute and unconditional as described above:

52

(a) at any time or from time to time, without notice to the Guarantor, the time for any performance of or compliance with any of the Obligations shall be extended, or such performance or compliance shall be waived;

- (b) any of the acts mentioned in any of the provisions of any of the Loan Documents, any Swap Contract between the Borrower and any Lender, or any Affiliate of a Lender, or any other agreement or instrument referred to in the Loan Documents or such Swap Contracts shall be done or omitted;
- (c) the maturity of any of the Obligations shall be accelerated, or any of the Obligations shall be modified, supplemented or amended in any respect, or any right under any of the Loan Documents, any Swap Contract between any Loan Party and any Lender, or any Affiliate of a Lender, or any other agreement or instrument referred to in the Loan Documents or such Swap Contracts shall be waived or any other guarantee of any of the Obligations or any security therefor shall be released, impaired or exchanged in whole or in part or otherwise dealt with; or
- (d) any of the Obligations shall be determined to be void or voidable (including, without limitation, for the benefit of any creditor of the Guarantor) or shall be subordinated to the claims of any Person (including, without limitation, any creditor of the Guarantor).

With respect to its obligations hereunder, the Guarantor hereby expressly waives diligence, presentment, demand of payment, protest and all notices whatsoever, and any requirement that the Administrative Agent or any Lender exhaust any right, power or remedy or proceed against any Person under any of the Loan Documents, any Swap Contract between any Loan Party and any Lender, or any Affiliate of a Lender, or any other agreement or instrument referred to in the Loan Documents or such Swap Contracts or against any other Person under any other guarantee of, or security for, any of the Obligations.

4.3 Reinstatement.

The obligations of the Guarantor under this Article IV shall be automatically reinstated if and to the extent that for any reason any payment by or on behalf of any Person in respect of the Obligations is rescinded or must be otherwise restored by any holder of any of the Obligations, whether as a result of any proceedings in bankruptcy or reorganization or otherwise, and the Guarantor agrees that it will indemnify the Administrative Agent, the Swing Line Lender, the Issuers and each Lender on demand for all reasonable costs and expenses (including, without limitation, the fees, charges and disbursements of counsel) incurred by the Administrative Agent, the Swing Line Lender, the Issuers or such Lender in connection with such rescission or restoration, including any such costs and expenses incurred in defending against any claim alleging that such payment constituted a preference, fraudulent transfer or similar payment under any bankruptcy, insolvency or similar law.

4.4 <u>Certain Additional Waivers</u>.

The Guarantor agrees that it shall have no right of recourse to security for the Obligations, except through the exercise of rights of subrogation pursuant to Section 4.2.

53

4.5 Remedies.

The Guarantor agrees that, to the fullest extent permitted by law, as between the Guarantor, on the one hand, and the Administrative Agent and the Lenders, on the other hand, the Obligations may be declared to be forthwith due and payable as provided in <u>Section 9.1</u> (and shall be deemed to have become automatically due and payable in the circumstances provided in said <u>Section 9.1</u>) for purposes of <u>Section 4.1</u> notwithstanding any stay, injunction or other prohibition preventing such declaration (or preventing the Obligations from becoming automatically due and payable) as against any other Person and that, in the event of such declaration (or the Obligations being deemed to have become automatically due and payable), the Obligations (whether or not due and payable by any other Person) shall forthwith become due and payable by the Guarantors for purposes of <u>Section 4.1</u>.

4.6 **Guarantee of Payment; Continuing Guarantee.**

The guarantee in this <u>Article IV</u> is a guaranty of payment and not of collection, is a continuing guarantee, and shall apply to all Obligations whenever arising.

ARTICLE V

CONDITIONS PRECEDENT

5.1 <u>Initial Credit Extension</u>.

This Agreement shall become effective upon and the obligation of the Lenders and the Issuers to make the initial Credit Extension hereunder is subject to the Administrative Agent's receipt of (a) all fees required to be paid to the Administrative Agent, the Arrangers and the Lenders on the date hereof (including the reasonable fees and expenses of counsel to the Administrative Agent for which reasonably detailed invoices have been presented on or prior to the date hereof), (b) evidence that, prior to or concurrently with the effectiveness of this Agreement, all obligations under the Existing Credit Facility have been paid in full and all commitments to lend thereunder have been terminated, and (c) all of the following, in form and substance satisfactory to each Agent and each Lender, and in sufficient copies for each Lender:

- (i) Copies of the articles or certificate of incorporation of the Borrower and the Guarantor, together with all amendments, certified by the Secretary or an Assistant Secretary of the Borrower and the Guarantor, and a certificate of good standing, certified by the appropriate governmental officer in the jurisdiction of incorporation of the Borrower and the Guarantor, as well as any other information that any Lender may request that is required by Section 326 of the USA PATRIOT ACT or necessary for the Administrative Agent or any Lender to verify the identity of the Borrower or the Guarantor as required by Section 326 of the USA PATRIOT ACT.
- (ii) Copies, certified by the Secretary or an Assistant Secretary of each of the Borrower and the Guarantor, of the by-laws of the Borrower or the Guarantor, as

applicable, and the Board of Directors' resolutions and of resolutions or actions of any other body authorizing the execution of the Loan Documents to which the Borrower or the Guarantor is a party.

- (iii) An incumbency certificate, executed by the Secretary or an Assistant Secretary of each of the Borrower and the Guarantor, which shall identify by name and title and bear the signatures of the Authorized Officers and any other officers of the Borrower and the Guarantor authorized to sign the Loan Documents to which the Borrower or the Guarantor is a party, upon which certificate the Administrative Agent and the Lenders shall be entitled to rely until informed of any change in writing by the Borrower or the Guarantor.
- (iv) A certificate, signed by an Authorized Officer of the Borrower, stating that (A) on the date hereof, no Default or Unmatured Default has occurred and is continuing and (B) the representations and warranties contained in <u>Article VI</u> are true and correct in all material respects as of the date hereof.
- (v) A written opinion of counsel to the Borrower and the Guarantor, addressed to the Administrative Agent and the Lenders in a form reasonably satisfactory to the Administrative Agent and its counsel.
- (vi) Executed counterparts of this Agreement executed by the Borrower, the Guarantor, the Administrative Agent, each Issuer and each Lender.
- (vii) Any Revolving Notes requested by a Lender pursuant to <u>Section 2.13</u> payable to the order of each such requesting Lender and the Swing Line Note.
 - (viii) If the initial Credit Extension will be the issuance of a Letter of Credit, a properly completed Letter of Credit Application.
 - (ix) Evidence of the effectiveness of the Great Plains Credit Agreement and the KCPL Credit Agreement.
- (x) Written money transfer instructions, in substantially the form of <u>Exhibit C</u>, addressed to the Administrative Agent and signed by an Authorized Officer of the Borrower who has executed and delivered an incumbency certificate in accordance with the terms hereof, together with such other related money transfer authorizations as the Administrative Agent may have reasonably requested.
 - (xi) Such other documents as any Lender or its counsel may have reasonably requested.

5.2 <u>Each Credit Extension</u>.

The Lenders shall not be required to make any Credit Extension (other than a Credit Extension that, after giving effect thereto and to the application of the proceeds thereof, does not

55

increase the aggregate amount of outstanding Credit Extensions) or increase their Commitments pursuant to any GMO-GPE Re-Transfer or GPE-GMO Transfer, unless on the date of such Credit Extension, GMO-GPE Re-Transfer or GPE-GMO Transfer:

- (a) No Default or Unmatured Default exists or would result from such Credit Extension.
- (b) The representations and warranties contained in <u>Article VI</u> are true and correct in all material respects as of the date of such Credit Extension except to the extent any such representation or warranty is stated to relate solely to an earlier date, in which case such representation or warranty shall have been true and correct in all material respects on and as of such earlier date; <u>provided</u> that this <u>clause (b)</u> shall not apply to the representations and warranties set forth in <u>Section 6.5</u> (as it rel ates to <u>clause (i)</u>, <u>(ii)</u> or <u>(iii)</u> of the definition of "Material Adverse Effect"), <u>clause (a)</u> of the first sentence of <u>Section 6.7</u> and the second sentence of <u>Section 6.7</u> with respect to any borrowing hereunder which is not part of the Initial Credit Extension.
- (c) With respect to a Credit Extension, no authorization or approval of any Governmental Authority, other than any such authorizations and approvals as have already been obtained and remain in full force and effect, will be required for the making of such Credit Extension.

Each delivery of a Borrowing Notice and each request for the issuance of a Letter of Credit shall constitute a representation and warranty by the Borrower that the conditions contained in <u>Sections 5.2(a)</u>, <u>(b)</u> and <u>(c)</u> have been satisfied.

ARTICLE VI

REPRESENTATIONS AND WARRANTIES

Each of the Borrower and the Guarantor represents and warrants to the Lenders, with respect to itself and, to the extent referenced, its respective Subsidiaries or Significant Subsidiaries, that:

6.1 <u>Existence and Standing.</u>

Each of the Borrower, the Guarantor and their respective Significant Subsidiaries (a) is a corporation, partnership (in the case of Subsidiaries only) or limited liability company duly and properly incorporated or organized, as the case may be, validly existing and (to the extent such concept applies to such entity) in good standing under the laws of its jurisdiction of incorporation or organization and (b) has all corporate power and authority and all governmental licenses, authorizations, consents and approvals required to carry on its business as now conducted, except where the failure to do so could not be reasonably expected to have a Material Adverse Effect.

56

6.2 <u>Authorization and Validity</u>.

Each of the Borrower and the Guarantor has the requisite corporate power and authority and legal right to execute and deliver the Loan Documents and to perform its obligations thereunder. The execution and delivery by the Borrower and the Guarantor of the Loan Documents and the performance of their respective obligations thereunder have been duly authorized by all necessary corporate action, and the Loan Documents constitute legal, valid and binding obligations of each of the Borrower and the Guarantor, as applicable, enforceable against each of the Borrower and the Guarantor in accordance with their terms, except as enforceability may be limited by bankruptcy, moratorium, reorganization or similar laws affecting the enforcement of creditors' rights generally and by general principles of equity (regardless of whether suc h enforceability is considered in a proceeding in equity or law).

6.3 <u>No Conflict; Government Consent.</u>

Neither the execution and delivery by the Borrower or the Guarantor of the Loan Documents, nor the consummation of the transactions therein contemplated, nor compliance with the provisions thereof will (a) violate (i) in any material respect any law, statute, rule, regulation, order, writ, judgment, injunction, decree or award binding on the Borrower or the Guarantor, (ii) the Borrower's or the Guarantor's certificate of incorporation or by-laws or (iii) the provisions of any indenture, instrument or agreement to which the Borrower or the Guarantor is a party or is subject, or by which it, or its Property, is bound, except to the extent any such violation could not reasonably be expected to have a Material Adverse Effect, (b) conflict with or constitute a default under any such indenture, instrument or other agreement in a ma nner which could reasonably be expected to have a Material Adverse Effect, or (c) result in, or require, the creation or imposition of any Lien in, of or on the Property of the Borrower or the Guarantor pursuant to the terms of any such indenture, instrument or other agreement. No order, consent, adjudication, approval, license, authorization, or validation of, or filing, recording or registration with, or exemption by, or other action in respect of any governmental or public body or authority, or any subdivision thereof, which has not been obtained by the Borrower or the Guarantor, is required to be obtained by the Borrower or the Guarantor in connection with the execution and delivery of the Loan Documents, the borrowings under this Agreement, the payment and performance by the Borrower or the Guarantor of the Obligations or the legality, validity, binding effect or enforceability of any of the Loan Documents excepting, however, (a) '34 Act Reports due after the date hereof disclosing the Loan Documents and related information as required pursuant to applicable '34 Act Report filing requirements, and (b) Borrower's short-term borrowing authorization pursuant to the Federal Power Act at t

6.4 Financial Statements.

The December 31, 2009 and June 30, 2010 consolidated and consolidating financial statements of the Guarantor and its Subsidiaries heretofore delivered to the Lenders were prepared in accordance with GAAP and fairly present the consolidated and consolidating financial condition and results of operations of the Guarantor and its Subsidiaries at such dates

57

and the consolidated and consolidating results of their operations for the periods then ended subject, in the case of the June 30, 2010 financial statements, to normal year-end adjustments.

6.5 <u>Material Adverse Change</u>.

Since December 31, 2009 and except as disclosed in the Borrower's, if any, and the Guarantor's '34 Act Reports, there has been no change in the business, Property, financial condition or results of operations of (i) the Guarantor and its Subsidiaries, taken as a whole, or (ii) the Borrower and its Subsidiaries, taken as a whole, which, in either case, could reasonably be expected to have a Material Adverse Effect.

6.6 <u>Taxes</u>.

The Borrower, the Guarantor and their respective Significant Subsidiaries have filed all United States federal income tax returns and all other material tax returns which are required to be filed by them and have paid all taxes due and payable pursuant to said returns or pursuant to any assessment received by the Borrower, the Guarantor or any of their respective Significant Subsidiaries, except such taxes, if any, as are being contested in good faith and as to which adequate reserves have been provided in accordance with GAAP and as to which no Lien exists or except to the extent that the failure to pay any such tax, or to file any such tax return, could not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect. No tax liens have been filed and no material claims are being asserted agains t the Borrower, the Guarantor or any Significant Subsidiary with respect to any such taxes except to the extent that such tax liens or claims could not reasonably be expected to have a Material Adverse Effect.

6.7 <u>Litigation; etc.</u>

Except as disclosed in the Borrower's, if any, and the Guarantor's '34 Act Reports, there is no litigation, arbitration, governmental investigation, proceeding or inquiry pending or, to the knowledge of any of their officers, threatened against or affecting the Borrower, the Guarantor or any of their respective Subsidiaries in which there is a reasonable possibility of an adverse decision and which (a) could reasonably be expected to have a Material Adverse Effect or (b) seeks to prevent, enjoin or delay the making of any Credit Extension. Other than any liability incident to any litigation, arbitration or

proceeding which could not reasonably be expected to have a Material Adverse Effect, each of the Borrower and the Guarantor has no material contingent obligations not provided for or disclosed in the financial statements referred to in <u>Section 6.4</u>.

6.8 ERISA.

Each of the Borrower and the Guarantor and each other member of the Controlled Group has fulfilled its obligations under the minimum funding standards of ERISA and the Code with respect to each Plan and is in compliance with the presently applicable provisions of ERISA and the Code with respect to each Plan, except to the extent that noncompliance, individually or in the aggregate, has not resulted in and could not reasonably be expected to result in a Material Adverse Effect. Neither the Borrower, nor the Guarantor nor any other member of the

58

Controlled Group has (i) failed to make any required contribution or payment to any Plan or Multiemployer Plan, or made any amendment to any Plan which has resulted or could result in the imposition of a Lien or the posting of a bond or other security under ERISA or the Code or (ii) incurred any liability under Title IV of ERISA other than a liability to the PBGC for premiums under Section 4007 of ERISA.

6.9 Accuracy of Information.

No information, exhibit or report furnished by the Borrower, the Guarantor or any of their respective Subsidiaries to the Administrative Agent, any Issuer or any Lender in connection with the negotiation of, or compliance with, the Loan Documents contained, as of the date of delivery thereof and taken as a whole, any untrue statement of a material fact or, when considered together with all reports filed with the SEC and furnished or deemed furnished pursuant to the terms hereof, omitted to state any material fact necessary to make the statements contained therein, in light of the circumstances under which they were made, not misleading; provided that, with respect to projected financial information, forecasts and other forward-looking information, each of the Borrower and the Guarantor represents only that such information was prepared in good faith based upon assumptions and estimates developed by management of the Borrower or the Guarantor, as applicable, in good faith and believed to be reasonable at the time (it being understood that such information is not a guarantee of future performance and that actual results during the period or periods covered by such information may materially differ from the projected results therein).

6.10 Regulation U.

Neither the Borrower nor the Guarantor is engaged, nor will engage, principally or as one of its important activities, in the business of purchasing or carrying margin stock (as defined in Regulation U), or extending credit for the purpose of purchasing or carrying margin stock. Margin stock constitutes less than twenty-five percent (25%) of the value of those assets of the Borrower and its Subsidiaries which are subject to any limitation on sale, pledge or other restriction hereunder, and constitutes less than twenty-five percent (25%) of the value of those assets of the Guarantor and its Subsidiaries which are subject to any limitation on sale, pledge or other restriction hereunder.

6.11 <u>Material Agreements</u>.

Neither the Borrower, nor the Guarantor nor any Subsidiary is a party to any agreement or instrument or subject to any charter or other corporate restriction which is reasonably likely to have a Material Adverse Effect. Neither the Borrower, nor the Guarantor nor any Subsidiary is in default in the performance, observance or fulfillment of any of the obligations, covenants or conditions contained in any agreement to which it is a party, which default could reasonably be expected to have a Material Adverse Effect.

59

6.12 <u>Compliance With Laws</u>.

The Borrower, the Guarantor and their respective Subsidiaries have complied with all applicable statutes, rules, regulations, orders and restrictions of any domestic or foreign government or any instrumentality or agency thereof having jurisdiction over the conduct of their respective businesses or the ownership of their respective Property except for any failure to comply with any of the foregoing which could not reasonably be expected to have a Material Adverse Effect.

6.13 Ownership of Properties.

On the date of this Agreement, the Borrower, the Guarantor and their respective Significant Subsidiaries have good title, free of all Liens other than those permitted by Section 7.12, to all of the Property and assets reflected as owned by the Guarantor and its Significant Subsidiaries in the Guarantor's most recent consolidated financial statements provided to the Administrative Agent.

6.14 <u>Plan Assets; Prohibited Transactions.</u>

To each of the Borrower's and the Guarantor's knowledge, (i) neither the Borrower nor the Guarantor is an entity deemed to hold "plan assets" within the meaning of 29 C.F.R. § 2510.3-101 of another entity's employee benefit plan (as defined in Section 3(3) of ERISA) which is subject to Title I of ERISA or any plan (within the meaning of Section 4975 of the Code), and (ii) neither the execution of this Agreement nor the making of Loans hereunder gives rise to a prohibited transaction within the meaning of Section 406 of ERISA or Section 4975 of the Code.

6.15 <u>Environmental Matters</u>.

Except as set forth in the Borrower's, if any, and the Guarantor's '34 Act Reports, there are no known risks and liabilities accruing to the Borrower, the Guarantor or any of their respective Subsidiaries due to Environmental Laws that could reasonably be expected to have a Material Adverse Effect.

6.16 <u>Investment Company Act.</u>

Neither the Borrower, nor the Guarantor nor any Subsidiary is or is required to be registered as an "investment company" within the meaning of the Investment Company Act of 1940, as amended.

6.17 Pari Passu Indebtedness.

The Indebtedness under the Loan Documents ranks at least <u>pari passu</u> with all other unsecured Indebtedness of the Borrower. The Guarantor's obligations under Article IV rank at least <u>pari passu</u> with all other unsecured Indebtedness of the Guarantor.

60

6.18 Solvency.

As of the date hereof and after giving effect to the consummation of the transactions contemplated by the Loan Documents, the Borrower, the Guarantor and each Significant Subsidiary is solvent. For purposes of the preceding sentence, solvent means (a) the fair saleable value (on a going concern basis) of the Borrower's assets, the Guarantor's assets or a Significant Subsidiary's assets, as applicable, exceed its liabilities, contingent or otherwise, fairly valued, (b) such Person will be able to pay its debts as they become due and (c) such Person will not be left with unreasonably small capital as is necessary to satisfy all of its current and reasonably anticipated obligations giving due consideration to the prevailing practice in the industry in which such Person is engaged. In computing the amount of contingent liabilities at any time, it is intended that such liabilities will be computed at the amount which, in light of all the facts and circumstances existing at such time, represents the amount that can reasonably be expected to become an actual or matured liability. Neither the Borrower nor the Guarantor is entering into the Loan Documents with the actual intent to hinder, delay or defraud its current or future creditors, nor does the Borrower or the Guarantor intend to or believe that it will incur, as a result of entering into this Agreement and the other Loan Documents, debts beyond its ability to repay.

6.19 No Defaults.

No Unmatured Default or Default has occurred and is continuing.

ARTICLE VII

COVENANTS

During the term of this Agreement, unless the Required Lenders shall otherwise consent in writing, each of the Borrower and the Guarantor covenants and agrees with respect to itself and, to the extent referenced, its respective Subsidiaries or Significant Subsidiaries, as follows:

7.1 Financial Reporting and Notices.

The Borrower and the Guarantor will maintain, for itself and each Subsidiary, a system of accounting established and administered in accordance with generally accepted accounting principles, and (subject to the penultimate paragraph of this <u>Section 7.1</u>) furnish to the Administrative Agent for distribution to the Lenders:

(a) Within ninety (90) days after the close of each of the Guarantor's fiscal years, (i) an unqualified audit report certified by an independent registered public accounting firm which is a member of the "Big Four," prepared in accordance with GAAP on a consolidated basis for the Guarantor and its Consolidated Subsidiaries, including balance sheets as of the end of such period and related statements of income, common shareholders' equity and cash flows, accompanied by any management letter prepared by said accountants and (ii) an unaudited consolidating balance sheet for the Guarantor and its Subsidiaries as of the end of such period and related consolidating

61

income statements, certified by the Chief Accounting Officer or Chief Financial Officer of the Guarantor.

- (b) Within forty-five (45) days after the close of the first three (3) quarterly periods of each of the Guarantor's fiscal years, (i) for the Guarantor and its Consolidated Subsidiaries, either (A) consolidated unaudited balance sheets as at the close of each such period and consolidated and consolidating profit and loss and reconciliation of surplus statements and a statement of cash flows for the period from the beginning of such fiscal year to the end of such quarter, all certified by the Chief Accounting Officer or Chief Financial Officer of the Guarantor or (B) if the Guarantor is then a "registrant" within the meaning of Rule 1-01 of Regulation S-X of the SEC and required to file a report on Form 10-Q with the SEC, a copy of the Guarantor and its Subsidiaries as at the close of each such period, all certified by the Chief Accounting Officer or Chief Financial Officer of the Guarantor or (B) if the Borrower is then a "registrant" within the meaning of Rule 1-01 of Regulation S-X of the SEC and required to file a report on Form 10-Q with the SEC, a copy of the Borrower's report on Form 10-Q for such quarterly period.
- (c) Together with the financial statements required under <u>Sections 7.1(a)</u> and (b), a compliance certificate in substantially the form of <u>Exhibit A</u> signed by the Chief Accounting Officer or Chief Financial Officer of the Borrower setting forth calculations of the financial covenants

contained in <u>Article VII</u> and stating that no Default or Unmatured Default exists, or if any Default or Unmatured Default exists, stating the nature and status thereof.

- (d) As soon as possible and in any event within thirty (30) days after the Borrower, the Guarantor or any member of the Controlled Group knows that any Reportable Event has occurred with respect to any Plan that could reasonably be expected to result in liability in an aggregate amount in excess of \$50,000,000, a statement, signed by the Chief Accounting Officer or Chief Financial Officer of the Borrower or the Guarantor, as applicable, describing said Reportable Event and the action which the Borrower, the Guarantor or member of the Controlled Group proposes to take with respect thereto.
- (e) Promptly after receipt of notice by the Borrower, the Guarantor or any member of the Controlled Group of the PBGC's intention to terminate any Plan or to have a trustee appointed to administer any Plan, a copy of such notice.
- (f) Promptly upon the furnishing thereof to the shareholders of the Borrower or the Guarantor, copies of all financial statements, reports and proxy statements so furnished.
- (g) Promptly upon the filing thereof, copies of all registration statements and annual, quarterly, monthly or other regular reports which the Borrower or the Guarantor files with the SEC.

62

- (h) Promptly after the public announcement of, or promptly after receiving a written notice of, any change (whether an increase or decrease) by Moody's or S&P in the senior unsecured debt rating of the Borrower or the Guarantor.
- (i) Such other information (including non-financial information) as the Administrative Agent or any Lender may from time to time reasonably request.

The statements and reports required to be furnished by the Borrower or the Guarantor pursuant to <u>clauses (a)</u>, (<u>b</u>), (<u>f</u>) and (<u>g</u>) above shall be deemed furnished for such purpose upon becoming publicly available on the SEC's EDGAR web page.

The Borrower and the Guarantor hereby acknowledge that (a) the Administrative Agent and/or BAS will make available to the Lenders and Issuers materials and/or information provided by or on behalf of the Borrower or the Guarantor hereunder (collectively, "Borrower Materials") by posting the Borrower Materials on IntraLinks or another similar restricted access electronic system (the "Platform") and (b) certain of the Lenders may be "public-side" Lenders (i.e., Lenders that do not wish to receive material non-public information with respect to the Borrower or the Guarantor or their respective securities) (each, a "Public Lender"). Each of the Borrower and the Guarantor hereby agrees that (w) all Borrower Materials that are to be made available to Public Lenders shall be clearly and conspicuously marked "PUBLIC" which, at a minimum, shall mean that the word "PUBLIC" shall appear prominently on the first page thereof; (x) by marking Borrower Materials "PUBLIC," the Borrower and the Guarantor shall be deemed to have authorized the Agents, the Arrangers, the Issuers and the Lenders to treat such Borrower Materials as not containing any material non-public information with respect to the Borrower or the Guarantor or their respective securities for purposes of United States Federal and state securities laws (provided, however, that to the extent such Borrower Materials constitute Specified Information, they shall be treated as set forth in Section 10.11(a)); (y) all Borrower Materials marked "PUBLIC" are permitted to be made available through a portion of the Platform designated "Public Investor;" and (z) the Administrative Agent and BAS shall be entitled to treat any Borrower Materials that are not marked "PUBLIC" as being suitable only for posting on a portion of the Platform not designated "Public Investor." Notwithstanding the foregoing, neither the Borrower nor the Guarantor shall be under any obligation to mark any Borrower Materials "PUBLIC."

7.2 <u>Permits, Etc.</u>

The Borrower and the Guarantor will, and will cause their respective Significant Subsidiaries to, take all reasonable action to maintain all rights, privileges, permits, licenses and franchises necessary or desirable in the normal conduct of its business, except to the extent failure to do so could not reasonably be expected to have a Material Adverse Effect; and preserve or renew all of its registered patents, trademarks, trade names and service marks, the non-preservation of which could reasonably be expected to have a Material Adverse Effect.

63

7.3 <u>Use of Proceeds</u>.

The Borrower will use the proceeds of the Credit Extensions (i) to repay the Existing Credit Facility and (ii) for general corporate purposes, including the refinancing of any commercial paper program of the Borrower and its Subsidiaries and use for working capital, capital expenditures and other lawful purposes of the Borrower and its Subsidiaries. The Borrower will not use any of the proceeds of the Credit Extensions to purchase or carry any margin stock (as defined in Regulation U) or to extend credit for the purpose of purchasing or carrying margin stock; provided that the Borrower may repurchase its own stock so long as such stock is not reissued. The Guarantor will not permit margin stock to constitute twenty-five percent (25%) or more of the value of those assets of the Guarantor and its Subsidiaries which are subject to any limitation on sale, pledge or other restriction hereunder. The Borrower will not permit margin stock to constitute twenty-five percent (25%) or more of the value of those assets of the Borrower and its Subsidiaries which are subject to any limitation on sale, pledge or other restriction hereunder.

7.4 Notice of Default.

The Borrower and the Guarantor will give prompt notice in writing to the Administrative Agent and the Lenders of the occurrence of any Default or Unmatured Default and of any other development, financial or otherwise, which could reasonably be expected to have a Material Adverse Effect.

7.5 <u>Conduct of Business</u>.

Except as otherwise permitted under this Agreement, the Borrower and the Guarantor will, and will cause their respective Significant Subsidiaries to, carry on and conduct its business in substantially the same manner and in substantially the same general fields of enterprise as it is presently conducted and do all things necessary to remain duly incorporated or organized, validly existing and (to the extent such concept applies to such entity) in good standing as a domestic corporation, partnership or limited liability company in its jurisdiction of incorporation or organization, as the case may be, and except to the extent the failure to do so could not reasonably be expected to have a Material Adverse Effect, keep in full force and effect its respective rights, privileges and franchises necessary or desirable in the normal conduct of b usiness; provided, however, that nothing in this Section 7.5 shall prohibit the termination of the corporate existence of a Subsidiary of the Borrower or of the Guarantor or a component of its business if the Borrower or the Guarantor, as applicable, determines in good faith that such termination is in the best interest of the Borrower or the Guarantor, as applicable, and is not materially disadvantageous to the Lenders.

7.6 <u>Taxes</u>.

The Borrower and the Guarantor will, and will cause their respective Significant Subsidiaries to, timely file United States federal and applicable foreign, state and local tax returns required by law and pay when due all taxes, assessments and governmental charges and levies upon it or its income, profits or Property, except those which are being contested in good

64

faith by appropriate proceedings and with respect to which adequate reserves have been set aside in accordance with GAAP.

7.7 <u>Insurance</u>.

The Borrower and the Guarantor will, and will cause their respective Significant Subsidiaries to, maintain (either in such Borrower's or Guarantor's name or in such Significant Subsidiary's own name) with financially sound and reputable insurance companies that are not Affiliates of the Borrower, the Guarantor or their respective Subsidiaries (other than any captive insurance company) insurance on all their respective Properties and business against loss or damage of the kinds customarily insured against by Persons engaged in the same or similar business, of such types and in such amounts as are customarily carried under similar circumstances by such other Persons, and the Borrower and the Guarantor will furnish to the Lenders upon request by the Administrative Agent, full information presented in reasonable detail as t o the insurance carried. Such insurance may be subject to co-insurance, deductibility or similar clauses which, in effect, result in self-insurance of certain losses; provided that such self-insurance is in accord with the customary industry practices for Persons in the same or similar businesses and adequate insurance reserves are maintained in connection with such self-insurance to the extent required by GAAP.

7.8 <u>Compliance with Laws</u>.

The Borrower and the Guarantor will, and will cause their respective Significant Subsidiaries to, comply with all laws (including, without limitation, Environmental Laws), rules, regulations, orders, writs, judgments, injunctions, decrees or awards to which it may be subject, except where the necessity of compliance therewith is contested in good faith by appropriate proceedings or except where the failure to comply could not reasonably be expected to have a Material Adverse Effect.

7.9 <u>Maintenance of Properties; Books of Record.</u>

The Borrower and the Guarantor will, and will cause their respective Significant Subsidiaries to, (i) do all things necessary to maintain, preserve, protect and keep its material Property in good repair, working order and condition, ordinary wear and tear excepted, and make all necessary and proper repairs, renewals and replacements so that its business carried on in connection therewith may be properly conducted at all times and (ii) keep proper books of record and account, in which full and correct entries shall be made of all material financial transactions and the assets and business of the Borrower and the Guarantor and each Significant Subsidiary in accordance with GAAP; provided that nothing in this Section 60;7.9 shall prevent the Borrower, the Guarantor or any Significant Subsidiary from selling, abandoning, disposing of, or otherwise discontinuing the operation or maintenance of any of its Property (including the Capital Stock of any Subsidiary that is not a Significant Subsidiary) or equipment the retention of which in the good faith judgment of the Borrower, the Guarantor or such Significant Subsidiary is inadvisable or unnecessary to the Borrower, the Guarantor and their respective Subsidiaries, taken as a whole, or the failure to could not reasonably be expected to have a Material Adverse Effect.

65

7.10 <u>Inspection</u>.

The Borrower and the Guarantor will, and if a Default or Unmatured Default exists, will cause each of their respective Significant Subsidiaries to, permit the Administrative Agent and the Lenders, by their respective representatives and agents designated by the Required Lenders, upon not less than five (5) Business Days notice, to inspect any of the Property, books and financial records of such Person, to examine and make copies of the books of accounts and other financial records of such Person, and to discuss the affairs, finances and accounts of such Person with, and to be advised as to the same by, such Person's officers at such reasonable times and intervals as the Administrative Agent or any Lender may designate, subject, however, in all cases to the imposition of such conditions as the Borrower, the Guarantor and such Significant Subsidiaries shall deem necessary based on reasonable considerations of safety and security; provided, however, that neither the Borrower, the Guarantor nor any of their respective Significant Subsidiaries shall be required to disclose to the Administrative Agent, and any Lender, or any representatives thereof any information which is the subject of attorney-client privilege or attorney work-product privilege properly asserted by the applicable Person to prevent the loss of such privilege in connection with such information or which is prevented from disclosure pursuant to a confidentiality agreement with third parties. After the occurrence and during the continuance of a Default, any

such inspection shall be at the Borrower's or the Guarantor's expense, as applicable; at all other times, the Borrower and the Guarantor shall not be liable to pay the ex penses of the Administrative Agent or any Lender in connection with such inspections.

7.11 Consolidations, Mergers and Sale of Assets.

The Borrower and the Guarantor will not, nor will they permit any of their respective Significant Subsidiaries (other than any Project Finance Subsidiary) to, sell, lease, transfer, or otherwise dispose of all or substantially all of its assets (whether by a single transaction or a number of related transactions and whether at one time or over a period of time) or consolidate with or merge into any Person or permit any Person to merge into it, except

- (a) (i) A Wholly-Owned Subsidiary may be merged into the Borrower, and (ii) a Wholly-Owned Subsidiary (including, without limitation, the Borrower) may be merged into the Guarantor <u>provided</u> that, if the Borrower is merged into the Guarantor, the Guarantor shall expressly assume, in a writing reasonably satisfactory to the Administrative Agent, the due and punctual payment of the Obligations and the due and punctual performance of and compliance with all of the terms of this Agreement and the other Loan Documents to be performed or complied with by the Borrower.
- (b) Any Significant Subsidiary may sell all or substantially all of its assets to, or consolidate or merge into, another Significant Subsidiary; <u>provided</u> that, immediately before and after such merger, consolidation or sale, no Default or Unmatured Default shall exist.
 - (c) the Borrower or KCPL may sell or transfer accounts receivable, in each case pursuant to one or more securitization transactions.

66

Notwithstanding the foregoing, the Borrower, the Guarantor and their respective Consolidated Subsidiaries (excluding Project Finance Subsidiaries) will not convey, transfer, lease or otherwise dispose of (whether in one transaction or a series of transactions, but excluding (i) sales of inventory in the ordinary course of business, (ii) transactions permitted by clauses (a) through (d) above, (iii) transfers by KCPL or the Borrower of assets related to, or ownership interests in, Iatan 2 to co-owners of Iatan 2 pursuant to the co-ownership, co-operating or other similar agreements of the co-owners of Iatan 2 and (iv) sales of the capital stock or assets of KLT Gas Inc. and Subsidiaries thereof) in the aggregate within any twelve (12)-month period, more than twenty percent (20%) of the aggregate book value of the assets of the Guarantor and its Consolidated Subsidiaries (excluding Project Finance Subsidiaries) as calculated as of the end of the most recent fiscal quarter.

67

7.12 <u>Liens</u>.

The Borrower and the Guarantor will not, nor will they permit any of their respective Significant Subsidiaries (other than any Project Finance Subsidiary) to, create, incur, or suffer to exist any Lien in, of or on the Property of the Borrower, the Guarantor or any of their respective Significant Subsidiaries (other than any Project Finance Subsidiary), except:

- (a) Liens for taxes, assessments or governmental charges or levies on its Property if the same shall not at the time be delinquent or thereafter can be paid without penalty, or are being contested in good faith and by appropriate proceedings and for which adequate reserves in accordance with GAAP shall have been set aside on its books.
- (b) Liens imposed by law, such as carriers', warehousemen's, mechanics' and landlords' liens and other similar liens arising in the ordinary course of business which secure payment of obligations not more than sixty (60) days past due or which are being contested in good faith

by appropriate proceedings and for which adequate reserves shall have been set aside on its books.

- (c) Liens arising out of pledges or deposits in the ordinary course of business under worker's compensation laws, unemployment insurance, old age pensions, or other social security or retirement benefits, or similar legislation, other than any Lien imposed under ERISA.
- (d) Liens incidental to the normal conduct of the Borrower, the Guarantor or any of their respective Significant Subsidiaries or the ownership or leasing of its Property or the conduct of the ordinary course of its business, including (i) zoning restrictions, easements, building restrictions, rights of way, reservations, restrictions on the use of real property and such other encumbrances or charges against real property as are of a nature generally existing with respect to properties of a similar character and which are not substantial in amount and do not in any material way affect the marketability of the same, (ii) rights of lessees and lessors under leases, (iii) rights of collecting banks having rights of setoff, revocation, refund or chargeback with respect to money or instruments of the Borrower, the Guarantor or any of their respective Significant Subsidiaries on deposit with or in the possession of such banks, (iv) Liens or deposits to secure the performance of statutory obligations, tenders, bids, contracts, leases, progress payments, performance or return-of-money bonds, surety and appeal bonds, performance or other similar bonds, letters of credit, or other obligations of a similar nature incurred in the ordinary course of business, (v) Liens required by any contract or statute in order to permit the Borrower, the Guarantor or any of their respective Significant Subsidiaries to perform any contract or subcontract made by it with or pursuant to the requirements of a governmental entity, in each case which are not incurred in connection with the borrowing of money, the obtaining of advances of credit or the payment of the deferred purchase price of Property and which do not in the aggregate impair the use of Property in the operation of the business of the Borrower , the Guarantor and their respective Significant Subsidiaries taken as a whole and (vi) rights reserved to or vested in any Governmental Authority by

68

the terms of any right, power, franchise, grant, license or permit, or by any Laws, to terminate such right, power, franchise, license or permit or to purchase, condemn, expropriate or recapture or to designate a purchaser of any of the Property of the Borrower, the Guarantor or any of their respective Significant Subsidiaries.

- (e) Liens arising under the General Mortgage Indenture and Deed of Trust Dated December 1, 1986 from KCPL to UMB, N.A.
- (f) Liens on Property of the Borrower, the Guarantor or KCPL existing on the date hereof and any renewal or extension thereof; <u>provided</u> that the Property covered thereby is not increased and any renewal or extension of the obligations secured or benefited thereby is permitted by this Agreement.
 - (g) Judgment Liens which secure payment of legal obligations that would not constitute a Default under Section 8.9.
- (h) Liens on Property acquired by the Borrower, the Guarantor or a Significant Subsidiary of either of the Borrower or the Guarantor after the date hereof, existing on such Property at the time of acquisition thereof (and not created in anticipation thereof); <u>provided</u> that in any such case no such Lien shall extend to or cover any other Property of the Borrower, the Guarantor or such Significant Subsidiary, as the case may be.
- (i) Liens on the Property, revenues and/or assets of any Person that exist at the time such Person becomes a Significant Subsidiary and the continuation of such Liens in connection with any refinancing or restructuring of the obligations secured by such Liens.
- (j) Liens on Property securing Indebtedness incurred or assumed at the time of, or within twelve (12) months after, the acquisition of such Property for the purpose of financing all or any part of the cost of acquiring such Property; provided that (i) such Lien attaches to such Property concurrently with or within twelve (12) months after the acquisition thereof, (ii) such Lien attaches solely to the Property so acquired in such transaction and (iii) the principal amount of the Indebtedness secured thereby does not exceed the cost or fair market value determined at the date of incurrence, whichever is lower, of the Property being acquired on the date of acquisition.
- (k) Liens on any improvements to Property securing Indebtedness incurred to provide funds for all or part of the cost of such improvements in a principal amount not exceeding the cost of construction of such improvements and incurred within twelve (12) months after completion of such improvements or construction, <u>provided</u> that such Liens do not extend to or cover any property of the Borrower, the Guarantor or any Significant Subsidiary other than such improvements.
- (l) Liens to government entities granted to secure pollution control or industrial revenue bond financings, which Liens in each financing transaction cover only

69

Property the acquisition or construction of which was financed by such financings and Property related thereto.

- (m) Liens on or over gas, oil, coal, fissionable material, or other fuel or fuel products as security for any obligations incurred by such Person (or any special purpose entity formed by such Person) for the sole purpose of financing the acquisition or storage of such fuel or fuel products or, with respect to nuclear fuel, the processing, reprocessing, sorting, storage and disposal thereof.
- (n) Liens on (including Liens arising out of the sale of) accounts receivable and/or contracts which will give rise to accounts receivable of the Borrower or KCPL; and other Liens on (including Liens arising out of the sale of) accounts receivable and/or contracts which will give rise to accounts receivable of the Borrower or any other Significant Subsidiary of the Guarantor in an aggregate amount not at any time exceeding \$10,000,000.

- (o) Liens on Property or assets of a Significant Subsidiary securing obligations owing to the Borrower, the Guarantor or any Significant Subsidiary (other than a Project Finance Subsidiary).
- (p) Liens on the stock or other equity interests of any Project Finance Subsidiary to secure obligations of such Project Finance Subsidiary (provided that the agreement under which any such Lien is created shall expressly state that it is non-recourse to the pledgor).
 - (q) Liens on Property of KCPL and the Borrower arising in connection with utility co-ownership, co-operating and similar agreements that are consistent with the utilities business and ancillary operations.
 - (r) Liens on assets held by entities which are deemed Subsidiaries solely due to the application of clause (c) of the definition of "Subsidiary" and which are required to be included in the Borrower's or the Guarantor's consolidated financial statements solely as a result of the application of ASC Topic 810, as it may be amended or supplemented.
 - (s) Liens securing Swap Contracts permitted to be incurred under this Agreement.
 - Liens securing any extension, renewal, replacement or refinancing of Indebtedness secured by any Lien referred to in the foregoing $\underline{\text{clauses (h), (i), (j), (k), (l), (m)}}$ and (q); provided that (A) such new Lien shall be limited to all or part of the same Property that secured the original Lien (plus improvements on such Property) and (B) the amount secured by such Lien at such time is not increased to any amount greater than the amount outstanding at the time of such renewal, replacement or refinancing.

70

- (u) Liens in favor of the Issuers or the Swing Line Lender on cash collateral securing the obligations of a Defaulting Lender to fund risk participations hereunder.
- (v) Liens which would otherwise not be permitted by <u>clauses (a)</u> through <u>(u)</u> securing additional Indebtedness of the Borrower, the Guarantor or a Significant Subsidiary (other than a Project Finance Subsidiary); <u>provided</u> that after giving effect thereto the aggregate unpaid principal amount of Indebtedness (including Capitalized Lease Obligations) of the Guarantor and its Significant Subsidiaries (other than any Project Finance Subsidiary) (including prepayment premiums and penalties) secured by Liens permitted by this <u>clause (v)</u> shall not exceed the greater of (a) \$50,000,000 and (b) ten percent (10%) of Consolidated Tangible Net Worth.
- (w) Liens on the stock or assets of Subsidiaries created pursuant to "rate reduction" bonds, for the payment of which legislatively authorized charges are imposed on customers.

7.13 Affiliates.

Except to the extent required by applicable Law or regulation with respect to transactions among the Guarantor and its Subsidiaries (excluding any Project Finance Subsidiary), the Borrower and the Guarantor will not, and will not permit any Subsidiary (other than any Project Finance Subsidiary) to, enter into any transaction (including the purchase or sale of any Property or service) with, or make any payment or transfer to, any Affiliate except in the ordinary course of business and pursuant to the reasonable requirements of the Guarantor's or such Subsidiary's business and upon fair and reasonable terms no less favorable to the Guarantor or such Subsidiary than the Guarantor or such Subsidiary would obtain in a comparable arms-length transaction.

7.14 <u>ERISA</u>.

The Borrower and the Guarantor will not, nor will they permit any Significant Subsidiary to, (i) voluntarily terminate any Plan, so as to result in any material liability of the Borrower, the Guarantor or any Significant Subsidiary to the PBGC or (ii) enter into any transaction that the Borrower, the Guarantor or such Significant Subsidiary knows or should know is a Prohibited Transaction (as defined in Section 4975 of the Code and in Section 406 of ERISA) involving any Plan which results in any liability of the Borrower, the Guarantor or any Significant Subsidiary that could reasonably be expected, individually or in the aggregate, to cause a Material Adverse Effect or (iii) cause any occurrence of any Reportable Event which results in any liability of the Borrower, the Guarantor or any Significant Subsidiary to the PBGC that could reasonably be expected, individually or in the aggregate, to cause a Material Adverse Effect or (iv) allow or suffer to exist any other event or condition known to the Borrower which results in any material liability of the Borrower, the Guarantor or any Significant Subsidiary to the PBGC.

71

7.15 <u>Total Indebtedness to Total Capitalization</u>.

The Borrower shall at all times cause the ratio of (i) Total Indebtedness to (ii) Total Capitalization to be less than or equal to 0.65 to 1.0.

7.16 <u>Restrictions on Subsidiary Dividends.</u>

The Borrower and the Guarantor will not, nor will they permit any Significant Subsidiary (other than any Project Finance Subsidiary) to, be a party to any agreement prohibiting or restricting the ability of such Significant Subsidiary to declare or pay dividends to the Borrower or the Guarantor; provided, that the foregoing provisions of this Section 7.16 shall not prohibit the Guarantor or any Significant Subsidiary from entering into any debt instrument containing a total debt to capitalization covenant.

7.17 <u>Organization Documents</u>.

The Borrower, the Guarantor and their respective Subsidiaries shall not amend, modify or change their Organization Documents in a manner materially adverse to the Lenders.

ARTICLE VIII

DEFAULTS

The occurrence of any one or more of the following events shall constitute a Default:

- **8.1** Any representation or warranty made or deemed made by or on behalf of the Borrower or the Guarantor to the Lenders or the Administrative Agent under or in connection with this Agreement, any Loan, or any certificate or information delivered in connection with this Agreement or any other Loan Document shall prove to have been incorrect in any material respect when made or deemed made.
- **8.2** (a) Nonpayment of principal of any Loan when due, nonpayment of any Reimbursement Obligations within one (1) Business Day after the same becomes due, or (b) nonpayment of interest upon any Loan or of any fee or other obligation under any of the Loan Documents within five (5) Business Days after the same becomes due.
- **8.3** The breach by the Borrower or the Guarantor of any of the terms or provisions of <u>Section 7.3</u>, <u>7.10</u> (with respect to the Borrower, the Guarantor and their respective Significant Subsidiaries only), <u>7.11</u>, <u>7.15</u> or <u>7.16</u>.
- **8.4** The breach by the Borrower or the Guarantor of any of the terms or provisions of Section 7.12 which is not remedied within thirty (30) days after the earlier of (a) an Authorized Officer of the Borrower or the Guarantor becoming aware of such breach and (b) receipt by the Borrower or the Guarantor of written notice from the Administrative Agent or any Lender.

72

- **8.5** The breach by the Borrower or the Guarantor (other than a breach which constitutes a Default under another Section of this Article VIII) of any of the terms or provisions of this Agreement which is not remedied within thirty (30) days after the earlier of (a) an Authorized Officer of the Borrower or the Guarantor becoming aware of such breach and (b) receipt by the Borrower or the Guarantor of written notice from the Administrative Agent or any Lender; provided that i f such breach is capable of cure but (i) cannot be cured by payment of money and (ii) cannot be cured by diligent efforts within such thirty (30)-day period, but such diligent efforts shall be properly commenced within such thirty (30)-day period and the Borrower or the Guarantor is diligently pursuing, and shall continue to pursue diligently, remedy of such failure, the cure period shall be extended for an additional ninety (90) days, but in no event beyond the Facility Termination Date.
- **8.6** Failure of the Borrower, the Guarantor or any of their respective Significant Subsidiaries to pay when due any Indebtedness aggregating in excess of \$50,000,000 ("Material Indebtedness"); or the default by the Borrower, the Guarantor or any of their respective Significant Subsidiaries in the performance of any term, provision or condition contained in any agreement under which any such Material Indebtedness was created or is governed, or any other event shall occur or condition exist, the effect of which default or event is to cause, or to permit the holder or holders of such Material Indebtedness to cause, such Material Indebtedness to become due prior to its stated maturity; or any Material Indebtedness of the Borrower, the Guarantor or any of their respective Significant Subsidiaries shall be declared to be due and payable or required to be prepaid or repurchased (other than by a regularly scheduled payment) prior to the stated maturity thereof; or the Borrower, the Guarantor or any of their respective Significant Subsidiaries shall not pay, or admit in writing its inability to pay, its debts generally as they become due.
- **8.7** The Borrower, the Guarantor or any of their respective Significant Subsidiaries shall (a) have an order for relief entered with respect to it under the Federal bankruptcy laws as now or hereafter in effect, (b) make an assignment for the benefit of creditors, (c) apply for, seek, consent to, or acquiesce in, the appointment of a receiver, custodian, trustee, examiner, liquidator or similar official for it or any Substantial Portion of its Property, (d) institute any proceeding seeking an order for relief under the Federal bankruptcy laws as now or hereafter in effect or seeking to adjudicate it a bankrupt or insolvent, or seeking dissolution, winding up, liquidation, reorganization, arrangement, adjustment or composition of it or its debts under any law relating to bankruptcy, insolvency or reorganization or relief of debtors or fail to file an answer or other pleading denying the material allegations of any such proceeding filed against it, (e) take any corporate, partnership or limited liability company action to authorize or effect any of the foregoing actions set forth in this Section 8.7 or (f) fail to contest in good faith any appointment or proceeding described in Section 8.8.
- **8.8** Without the application, approval or consent of the Borrower, the Guarantor or any of their respective Subsidiaries, a receiver, trustee, examiner, liquidator or similar official shall be appointed for the Borrower, the Guarantor or any of their respective Subsidiaries or any Substantial Portion of its Property, or a proceeding described in <u>Section 8.7(d)</u> shall be instituted against the Borrower, the Guarantor or any of their respective Subsidiaries and such

73

appointment continues undischarged or such proceeding continues undismissed or unstayed for a period of sixty (60) consecutive days.

8.9 The Borrower, the Guarantor or any of their respective Significant Subsidiaries shall fail within thirty (30) days to pay, bond or otherwise discharge (a) any judgment or order for the payment of money not covered by insurance in excess of \$50,000,000 (either singly or in the aggregate with other such judgments) or (b) any non-monetary final judgment that has, or could reasonably be expected to have, a Material Adverse Effect, in either case which is not stayed on appeal or otherwise being appropriately contested in good faith.

- **8.10** A Change of Control shall occur.
- **8.11** A Reportable Event shall have occurred which results in liability of the Borrower or the Guarantor or any of their respective Significant Subsidiaries in excess of \$50,000,000 and, thirty (30) days after written notice thereof shall have been given to the Borrower or the Guarantor by the Administrative Agent or any Lender, the amount of such liability shall still exceed \$50,000,000.
- **8.12** Any authorization or approval or other action by any Governmental Authority or regulatory body required for the execution, delivery or performance of this Agreement or any other Loan Document by the Borrower or the Guarantor shall fail to have been obtained at the time required or be terminated, revoked or rescinded or shall otherwise no longer be in full force and effect, and such occurrence shall (i) materially adversely affect the enforceability of the Loan Documents against the Borrower or the Guarantor and (ii) to the extent that such occurrence can be cured, shall continue for five (5) days (unless being contested in good faith by the Borrower or the Guarantor and the pendency of such contest shall suspend such adverse effect).
- **8.13** The Guarantor shall fail to own, directly or indirectly, all of the outstanding stock of the Borrower which, in the absence of any contingency, has the right to vote in an election of directors of the Borrower.
- **8.14** Any Loan Document, at any time after its execution and delivery and for any reason other than as expressly permitted hereunder or thereunder or satisfaction in full of all the Obligations, ceases to be in full force and effect in any material respect; or the Borrower or the Guarantor contests in any manner the validity or enforceability of any Loan Document; or the Borrower or the Guarantor denies that it has any or further liability or obligation under any Loan Document, or purports to revoke, terminate or rescind any Loan Document.

74

ARTICLE IX

ACCELERATION, WAIVERS, AMENDMENTS AND REMEDIES

9.1 <u>Acceleration; Letter of Credit Account.</u>

(a) If any Default described in Section 8.7 or 8.8 occurs with respect to the Borrower or the Guarantor, the obligations of the Lenders to make Loans hereunder and the obligation and power of the Issuers to issue Letters of Credit and the Swing Line Lender to make Swing Line Loans shall automatically terminate and the Obligations shall immediately become due and payable without any election or action on the part of the Administrative Agent, any Lender or any Issuer and the Borrower and the Guarantor will be and become thereby unconditionally obligated, without any further notice, act or demand, to pay to the Administrative Agent an amount in immediately available funds, which funds shall be held in the LC Collateral Accounts, equal to the excess of (i) the amount of Letter of Credit Obligations at such time over (ii) the amount on deposit in the LC Collateral Account at such time which is free and clear of all rights and claims of third parties and has not been applied against the Obligations (such difference, the "Collateral Shortfall Amount"). If any other Default occurs, the Administrative Agent may with the consent, or shall at the request, of the Required Lenders, (x) terminate or suspend the obligations of the Lenders to make Loans hereunder and the obligation and power of the Issuers to issue Letters of Credit, or declare the Obligations to be due and payable, or both, whereupon the Obligations shall become immediately due and payable, without presentment, demand, protest or notice of any kind, all of which each of the Borrower and the Guarantor hereb y expressly waives, and (y) upon notice to the Borrower or the Guarantor and in addition to the continuing right to demand payment of all amounts payable under this Agreement, make demand on the Borrower to pay, and the Borrower will, forthwith upon such demand and without any further notice or act, pay to the Administrative Agent in immediately available funds the Collateral Shortfall Amount, which funds shall be deposited in the LC Collateral Accounts.

If (a) within thirty (30) days after acceleration of the maturity of the Obligations or termination of the obligations of the Lenders to make Loans hereunder as a result of any Default (other than any Default as described in <u>Section 8.7</u> or <u>8.8</u> with respect to the Borrower or the Guarantor) and (b) before any judgment or decree for the payment of the Obligations due shall have been obtained or entered, the Required Lenders (in their sole discretion) shall so direct, the Administrative Agent shall, by notice to the Borrower, rescind and annul such acceleration and/or termination.

9.2 Amendments.

Subject to the provisions of this <u>Article IX</u>, the Required Lenders (or the Administrative Agent with the consent in writing of the Required Lenders) and the Borrower and the Guarantor may enter into agreements supplemental hereto for the purpose of adding or modifying any provisions to the Loan Documents or changing in any manner the rights of the Lenders or the Borrower or the Guarantor hereunder or waiving any Default hereunder; <u>provided</u> that no such supplemental agreement shall:

75

- (a) Extend the final maturity of any Loan or the expiry date of any Letter of Credit to a date after the Facility Termination Date or forgive all or any portion of the principal amount thereof, or reduce the rate or extend the time of payment of interest (other than default interest) or fees thereon, without the consent of each Lender directly affected thereby;
- (b) Reduce the percentage specified in the definition of Required Lenders, without the consent of each Lender directly affected thereby:
- (c) Increase the amount of the Commitment of any Lender without the consent of such Lender (except as provided for in Section 2.6), or extend the Facility Termination Date (except as provided for in Section 2.21), reduce the amount or extend the payment date for, the mandatory payments required under Section 2.2 or permit the Borrower to assign its rights under this Agreement, without the consent of the Administrative Agent, the Issuers, the Swing Line Lender, and each Lender directly affected thereby;

- (d) Amend this <u>Section 9.2</u> without the consent of each Lender directly affected thereby;
- (e) Release any funds from the LC Collateral Accounts, except to the extent such release is expressly permitted hereunder without the consent of each Lender directly affected thereby;
 - (f) Release the Guarantor from its obligations under this Agreement without the consent of each Lender; or
- (g) Alter the pro rata sharing of payments, the order of application of payments required thereby or the pro rata treatment of the Lenders without the consent of each Lender directly affected thereby.

Notwithstanding the foregoing, any amendment, waiver, or consent which by its terms requires the consent of all Lenders or each affected Lender may be effected with the consent of the applicable Lenders other than the Defaulting Lenders except that (x) the Commitment of any Defaulting Lender may not be increased or extended without the consent of each Lender and (y) any waiver, amendment or modification requiring the consent of all Lenders or each affected Lender that (i) by its terms affects any Defaulting Lender more adversely than other affected Lenders, (ii) reduces the principal of or the rate of interest for Loans of such Defaulting Lender, or fees or other amounts payable hereunder or under any other Loan Document to such Defaulting Lender or (iii) extends the time of the payment of interest or fees owing to such Defaulting Lender shall require the consent of such Defaulting Lender. No amendment of any provision of this Agreement relating to the Administrative Agent shall be effective without the written consent of such Issuer and no amendment of any provision of this Agreement relating to the Swing Line Lender shall be effective without the written consent of the Swing Line Lender. The Administrative Agent may

76

waive payment of the fee required under <u>Section 13.1(b)</u> without obtaining the consent of any other party to this Agreement.

9.3 Preservation of Rights.

No delay or omission of the Lenders, the Swing Line Lender, the Issuers or the Administrative Agent to exercise any right under the Loan Documents shall impair such right or be construed to be a waiver of any Default or an acquiescence therein, and the making of a Credit Extension notwithstanding the existence of a Default or the inability of the Borrower to satisfy the conditions precedent to such Credit Extension shall not constitute any waiver or acquiescence. Any single or partial exercise of any such right shall not preclude other or further exercise thereof or the exercise of any other right, and no waiver, amendment or other variation of the terms, conditions or provisions of the Loan Documents whatsoever shall be valid unless in writing signed by the Lenders required pursuant to Section 9.2, and then only to the extent in such writing specifically set forth. All remedies contained in the Loan Documents or by law afforded shall be cumulative and all shall be available to the Administrative Agent, the Swing Line Lender, the Lenders and the Issuers until the Obligations have been paid in full.

ARTICLE X

GENERAL PROVISIONS

10.1 <u>Survival of Representations.</u>

All representations and warranties of the Borrower and the Guarantor contained in this Agreement shall survive the making of the Credit Extensions herein contemplated.

10.2 <u>Governmental Regulation</u>.

Anything contained in this Agreement to the contrary notwithstanding, no Lender shall be obligated to extend credit to the Borrower in violation of any limitation or prohibition provided by any applicable statute or regulation.

10.3 Headings.

Section headings in the Loan Documents are for convenience of reference only, and shall not govern the interpretation of any of the provisions of the Loan Documents.

10.4 Entire Agreement.

THIS AGREEMENT AND THE OTHER LOAN DOCUMENTS REPRESENT THE FINAL AGREEMENT AMONG THE PARTIES AND MAY NOT BE CONTRADICTED BY EVIDENCE OF PRIOR, CONTEMPORANEOUS, OR SUBSEQUENT ORAL AGREEMENTS OF THE PARTIES. THERE ARE NO UNWRITTEN ORAL AGREEMENTS AMONG THE PARTIES.

The respective obligations of the Lenders hereunder are several and not joint and no Lender shall be the partner or agent of any other (except to the extent to which the Administrative Agent is authorized to act as such). The failure of any Lender to perform any of its obligations hereunder shall not relieve any other Lender from any of its obligations hereunder. This Agreement shall not be construed so as to confer any right or benefit upon any Person other than the parties to this Agreement and their respective successors and assigns; <u>provided</u> that the parties hereto expressly agree that each Arranger shall enjoy the benefits of the provisions of <u>Sections 10.6</u>, <u>10.10</u> and <u>11.7</u> to the extent specifically set forth therein and shall have the right to enforce such provisions on its own behalf and in its own name to the same extent as if it were a party to this Agreement.

10.6 <u>Expenses; Indemnification</u>.

- (a) Each of the Borrower and the Guarantor shall reimburse (i) the Agents and the Arrangers for any reasonable costs and expenses (including the reasonable fees and charges of outside counsel for the Agents) paid or incurred by the Agents or the Arrangers in connection with the preparation, negotiation, execution, delivery, syndication, distribution (including via the internet), review, amendment, modification, and administration of the Loan Documents and (ii) the Issuers for all reasonable out-of-pocket expenses incurred by the Issuers in connection with the issuance, amendment, renewal or extension of any Letter of Credit or any demand for payment thereunder. Each of the Borrower and the Guarantor also agrees to reimburse each Agent, each Arranger, the Swing Li ne Lender, each Lender and each Issuer for any reasonable costs, internal charges and expenses (including the reasonable fees and charges of attorneys for such Agent, such Arranger, the Swing Line Lender, such Lender or such Issuer) paid or incurred by either Agent, either Arranger, the Swing Line Lender, any Lender or any Issuer in connection with the collection and enforcement, attempted enforcement, and preservation of rights and remedies under, any of the Loan Documents (including all such costs and expenses incurred during any "workout" or restructuring in respect of the Obligations and during any legal proceeding).
- (b) Each of the Borrower and the Guarantor hereby further agrees to indemnify each Agent, each Arranger, the Swing Line Lender, each Lender, each Issuer, their respective affiliates and the directors, partners, officers, employees, agents and advisors of the foregoing against all losses, claims, damages, penalties, judgments, liabilities and expenses (including all reasonable expenses of litigation or preparation therefor whether or not either Agent, either Arranger, the Swing Line Lender, any Lender or any Issuer or any affiliate is a party thereto) which any of them may pay or incur arising out of or relating to this Agreement, the other Loan Documents, the transactions contemplated hereby or the direct or indirect application or proposed application of the proceeds of any Credit Extension hereunder except to the extent that they are determined in a final non-appealable judgment by a court of competent jurisdiction to have resulted from the gross negligence or willful misconduct of the party seeking indemnification. In the case of any investigation, litigation or proceeding to which the indemnity

78

in this Section applies, such indemnity shall be effective whether or not such investigation, litigation or proceeding is brought by a third party, by the Borrower or the Guarantor or by any affiliate of the Borrower or the Guarantor. The obligations of the Borrower and the Guarantor under this Section 10.6 shall survive the payment of the Obligations and termination of this Agreement.

- (c) To the extent that the Borrower or the Guarantor for any reason fails to indefeasibly pay any amount required under <u>subsection (a)</u> or (b) of this Section to be paid by it to the Administrative Agent (or any sub-agent thereof), the Swing Line Lender, any Issuer or any Related Party of any of the foregoing (without limiting its obligation to do so), each Lender severally agrees to pay to the Administrative Agent (or any such sub-agent), the Swing Line Lender, such Issuer or such Related Party, as the case may be, such Lender's Pro Rata Share (determined as of the time that the applicable unreimbursed expense or indemnity payment is sought but without giving effect to the reallocation of Pro Rata Shares contemplated by <u>Section 2.23(a)(iv)</u>) of such unpaid amount, <u>provided</u> that the unreimbursed expense or indemnified loss, claim, damage, liability or related expense, as the case may be, was incurred by or asserted against the Administrative Agent (or any such sub-agent), the Swing Line Lender or any Issuer in its capacity as such, or against any Related Party of any of the foregoing acting for the Administrative Agent (or any such sub-agent), the Swing Line Lender or any Issuer in connection with such capacity.
- (d) Notwithstanding any provision to the contrary, this <u>Section 10.6</u> shall not apply with respect to any matters, liabilities, or obligations relating to Taxes; all such matters are governed exclusively by <u>Section 3.5</u>.

10.7 <u>Numbers of Documents</u>.

All statements, notices, closing documents, and requests hereunder shall be furnished to the Administrative Agent with sufficient counterparts so that the Administrative Agent may furnish one to each of the Lenders.

10.8 Accounting.

Except as provided to the contrary herein, all accounting terms used herein shall be interpreted and all accounting determinations hereunder shall be made in accordance with GAAP.

10.9 <u>Severability of Provisions</u>.

Any provision in any Loan Document that is held to be inoperative, unenforceable, or invalid in any jurisdiction shall, as to that jurisdiction, be inoperative, unenforceable, or invalid without affecting the remaining provisions in that jurisdiction or the operation, enforceability, or validity of that provision in any other jurisdiction, and to this end the provisions of all Loan Documents are declared to be severable. Without limiting the foregoing provisions of this Section 10.9, if and to the extent that the enforceability of any provisions in this Agreement relating to Defaulting Lenders shall be limited by Debtor Relief Laws, as determined in good

faith by the Administrative Agent, the Issuers or the Swing Line Lender, as applicable, then such provisions shall be deemed to be in effect only to the extent not so limited.

10.10 Nonliability of Lenders.

The relationship between the Borrower and the Guarantor on the one hand and the Lenders, the Swing Line Lender, the Issuers and the Agents on the other hand shall be solely that of borrower/guarantor and lender. None of either Agent, either Arranger, the Swing Line Lender, any Lender or any Issuer shall have any fiduciary responsibilities to the Borrower or the Guarantor. None of either Agent, either Arranger, the Swing Line Lender, any Lender or any Issuer undertakes any responsibility to the Borrower or the Guarantor to review or inform the Borrower or the Guarantor of any matter in connection with any phase of the Borrower's business or operations. Each of the Borrower and the Guarantor agrees that none of either Agent, either Arranger, the Swing Line Lender, any Lender or any Issuer shall have liability to the Borrower or the Guarantor (whether sounding in tort, contract or otherwise) for losses suffered by the Borrower or the Guarantor in connection with, arising out of, or in any way related to, the transactions contemplated and the relationship established by the Loan Documents, or any act, omission or event occurring in connection therewith, unless it is determined in a final non-appealable judgment by a court of competent jurisdiction that such losses resulted from the gross negligence or willful misconduct of the party from which recovery is sought. None of either Agent, either Arranger, the Swing Line Lender, any Lender, any Issuer or any Related Party of any of the foregoing Persons shall have any liability with respect to, and each of the Borrower and the Guarantor hereby waives, releases and agrees not to sue for, any special, indirect, consequential or punitive damages suffered by the Borrower or the Guarantor in connection with, arising out of, or in any way related to the Lo an Documents or the transactions contemplated thereby. None of either Agent, either Arranger, the Swing Line Lender, any Lender, any Issuer or any Related Party of any of the foregoing Persons shall be liable for any damages arising from the use by unintended recipients of any information or other materials distributed by it through telecommunications, electronic or other information transmission systems in connection with this Agreement or the other Loan Documents or the transactions contemplated hereby or thereby except to the extent such receives such information due to the gross negligence of the party from which recovery is sought.

10.11 <u>Limited Disclosure</u>.

(a) Neither the Administrative Agent nor any Lender may disclose to any Person any Specified Information (as defined below) except (i) to its, and its Affiliates', officers, employees, agents, accountants, legal counsel, advisors and other representatives and to any direct or indirect, actual or prospective counterparty (or such counterparty's professional advisor) under any Swap Contract related to Loans outstanding under this Agreement who have a need to know such Specified Information or any direct, indirect, actual or prospective counterparty (and its advisor) to any swap, derivative or securitization transaction related to the Obligations or to any credit insurance provider relating to the Obligations (it being understood that the Persons to whom such disclosure is made will be informed of the confidential nature of such Specified Information and instructed to keep such Specified Information confidential, and shall agree to do so) or (ii) with the Borrower's prior consent. "Specified Information" means information that

80

the Borrower furnishes to the Administrative Agent or any Lender that is designated in writing as confidential, but does not include any such information that is or becomes generally available to the public or that is or becomes available to the Administrative Agent or such Lender from a source other than the Borrower.

(b) The provisions of <u>clause (a)</u> above shall not apply to Specified Information (i) that is a matter of general public knowledge or has heretofore been or is hereafter published in any source generally available to the public, (ii) that is required to be disclosed by law, regulation or subpoena, judicial order or other legal process, (iii) that is requested by any regulatory body (including any self-regulatory authority) with jurisdiction over the Administrative Agent or any Lender, or (iv) that is disclosed (A) to legal counsel, accountants and other professional advisors to such Lender, (B) in connection with the exercise of any right or remedy hereunder or under any Note or any suit or other litigation or proceeding relating to this Agreement or any Note, (C) to a rating agency if required by such agency in connection with a rating relating to Credit Extensions hereunder or (D) to assignees or participants or potential assignees or participants who agree to be bound by the provisions of this <u>Section 10.11</u>.

10.12 <u>USA PATRIOT ACT NOTIFICATION</u>.

The following notification is provided to the Borrower and the Guarantor pursuant to Section 326 of the USA Patriot Act of 2001, 31 U.S.C. Section 5318: IMPORTANT INFORMATION ABOUT PROCEDURES FOR OPENING A NEW ACCOUNT. To help the government fight the funding of terrorism and money laundering activities, Federal law requires all financial institutions to obtain, verify and record information that identifies each person or entity that opens an account, including any deposit account, treasury management account, loan, other extension of credit or other financial services product. What this means for the Borrower and the Guarantor: When the Borrower or the Guarantor opens an account, the Lenders will ask for the Borrower's or the Guarantor's, as applicable, name, tax identific ation number, business address and other information that will allow the Administrative Agent and the Lenders to identify the Borrower or the Guarantor, as applicable. The Administrative Agent and the Lenders may also ask to see the Borrower's or the Guarantor's legal organizational documents or other identifying documents.

10.13 Nonreliance.

Each Lender hereby represents that it is not relying on or looking to any margin stock (as defined in Regulation U of the FRB) for the repayment of the Loans provided for herein.

10.14 <u>No Advisory or Fiduciary Responsibility.</u>

In connection with all aspects of each transaction contemplated hereby, each of the Borrower and the Guarantor acknowledges and agrees, and acknowledges its Affiliates' understanding, that: (a) the credit facility provided for hereunder and any related arranging or other services in connection therewith (including in connection with any amendment, waiver or other modification hereof or of any other Loan Document) are an arm's-length commercial transaction between the Borrower and the Guarantor and their respective Affiliates, on the one

hand, and the Agents and the Arrangers, on the other hand, and the Borrower and the Guarantor are capable of evaluating and understanding and understand and accept the terms, risks and conditions of the transactions contemplated hereby and by the other Loan Documents (including any amendment, waiver or other modification hereof or thereof); (b) in connection with the process leading to such transaction, each Agent and Arranger is and has been acting solely as a principal and is not the financial advisor, agent or fiduciary, for the Borrower or the Guarantor or any of their respective Affiliates, stockholders, creditors or employees or any other Person; (c) no Agent or Arranger has assumed or will assume an advisory, agency or fiduciary responsibility in favor of the Borrower or the Guarantor with respect to any of the transactions contemp lated hereby or the process leading thereto, including with respect to any amendment, waiver or other modification hereof or of any other Loan Document (irrespective of whether any Agent or Arranger has advised or is currently advising the Borrower or the Guarantor or any of their respective Affiliates on other matters) and no Agent or Arranger has any obligation to the Borrower or the Guarantor or any of their respective Affiliates with respect to the transactions contemplated hereby except those obligations expressly set forth herein and in the other Loan Documents; (d) the Agents and the Arrangers and their respective Affiliates may be engaged in a broad range of transactions that involve interests that differ from those of the Borrower and the Guarantor and its Affiliates, and no Agent or Arranger has any obligation to disclose any of such interests by virtue of any advisory, agency or fiduciary relationship; and (e) the Agents and the Arrangers have not provided and will not provide any legal, accountin g, regulatory or tax advice with respect to any of the transactions contemplated hereby (including any amendment, waiver or other modification hereof or of any other Loan Document) and the Borrower has consulted its own legal, accounting, regulatory and tax advisors to the extent it has deemed appropriate. Each of the Borrower and the Guarantor hereby waives and releases, to the fullest extent permitted by law, any claims that it may have against any Agent and any Arranger with respect to any breach or alleged breach of agency or fiduciary duty.

ARTICLE XI

THE ADMINISTRATIVE AGENT

11.1 **Appointment and Authority.**

Each of the Lenders and the Issuers hereby irrevocably appoints Bank of America to act on its behalf as the Administrative Agent hereunder and under the other Loan Documents and authorizes the Administrative Agent to take such actions on its behalf and to exercise such powers as are delegated to the Administrative Agent by the terms hereof or thereof, together with such actions and powers as are reasonably incidental thereto. The provisions of this Article are solely for the benefit of the Administrative Agent, the Lenders and the Issuers, and the Borrower and the Guarantor shall have no rights as a third party beneficiary of any of such provisions.

11.2 Rights as a Lender.

The Person serving as the Administrative Agent hereunder shall have the same rights and powers in its capacity as a Lender as any other Lender and may exercise the same as though it were not the Administrative Agent and the term "Lender" or "Lenders" shall, unless otherwise

82

expressly indicated or unless the context otherwise requires, include the Person serving as the Administrative Agent hereunder in its individual capacity. Such Person and its Affiliates may accept deposits from, lend money to, act as the financial advisor or in any other advisory capacity for and generally engage in any kind of business with the Borrower or the Guarantor or any Subsidiary or other Affiliate thereof as if such Person were not the Administrative Agent hereunder and without any duty to account therefor to the Lenders.

11.3 <u>Exculpatory Provisions</u>.

The Administrative Agent shall not have any duties or obligations except those expressly set forth herein and in the other Loan Documents. Without limiting the generality of the foregoing, the Administrative Agent:

- (a) shall not be subject to any fiduciary or other implied duties, regardless of whether a Default or Unmatured Default has occurred and is continuing;
- (b) shall not have any duty to take any discretionary action or exercise any discretionary powers, except discretionary rights and powers expressly contemplated hereby or by the other Loan Documents that the Administrative Agent is required to exercise as directed in writing by the Required Lenders (or such other number or percentage of the Lenders as shall be expressly provided for herein or in the other Loan Documents), <u>provided</u> that the Administrative Agent shall not be required to take any action that, in its opinion or the opinion of its counsel, may expose the Administrative Agent to liability or that is contrary to any Loan Document or applicable law; and
- (c) shall not, except as expressly set forth herein and in the other Loan Documents, have any duty to disclose, and shall not be liable for the failure to disclose, any information relating to the Borrower or the Guarantor or any of its Affiliates that is communicated to or obtained by the Person serving as the Administrative Agent or any of its Affiliates in any capacity.

The Administrative Agent shall not be liable for any action taken or not taken by it (i) with the consent or at the request of the Required Lenders (or such other number or percentage of the Lenders as shall be necessary, or as the Administrative Agent shall believe in good faith shall be necessary, under the circumstances as provided in Sections 9.1 and 9.2) or (ii) in the absence of its own gross negligence or willful misconduct. The Administrative Agent shall be deemed not to have knowledge of any Default or Unmatured Default unless and until notice describing such Default or Unmatured Default is given to the Administrative Agent by the Borrower, the Guarantor, a Lender or an Issuer.

The Administrative Agent shall not be responsible for or have any duty to ascertain or inquire into (i) any statement, warranty or representation made in or in connection with this Agreement or any other Loan Document, (ii) the contents of any certificate, report or other document delivered hereunder or

thereunder or in connection herewith or therewith, (iii) the performance or observance of any of the covenants, agreements or other terms or conditions set forth herein or therein or the occurrence of any Default or Unmatured Default, (iv) the validity, enforceability, effectiveness or genuineness of this Agreement, any other Loan Document or any

83

other agreement, instrument or document or (v) the satisfaction of any condition set forth in <u>Article V</u> or elsewhere herein, other than to confirm receipt of items expressly required to be delivered to the Administrative Agent.

11.4 Reliance by Administrative Agent.

The Administrative Agent shall be entitled to rely upon, and shall not incur any liability for relying upon, any notice, request, certificate, consent, statement, instrument, document or other writing (including any electronic message, Internet or intranet website posting or other distribution) believed by it to be genuine and to have been signed, sent or otherwise authenticated by the proper Person. The Administrative Agent also may rely upon any statement made to it orally or by telephone and believed by it to have been made by the proper Person, and shall not incur any liability for relying thereon. In determining compliance with any condition hereunder to the making of a Loan, or the issuance of a Letter of Credit, that by its terms must be fulfilled to the satisfaction of a Lender or an Issuer, the Administrative Agent may presume that such condition is satisfactory to such Lender or such Issuer unless the Administrative Agent shall have received notice to the contrary from such Lender or such Issuer prior to the making of such Loan or the issuance of such Letter of Credit. The Administrative Agent may consult with legal counsel (who may be counsel for the Borrower or the Guarantor), independent accountants and other experts selected by it, and shall not be liable for any action taken or not taken by it in accordance with the advice of any such counsel, accountants or experts.

11.5 Delegation of Duties.

The Administrative Agent may perform any and all of its duties and exercise its rights and powers hereunder or under any other Loan Document by or through any one or more sub-agents appointed by the Administrative Agent. The Administrative Agent and any such sub-agent may perform any and all of its duties and exercise its rights and powers by or through their respective Related Parties. The exculpatory provisions of this Article shall apply to any such sub-agent and to the Related Parties of the Administrative Agent and any such sub-agent, and shall apply to their respective activities in connection with the syndication of the credit facilities provided for herein as well as activities as Administrative Agent.

11.6 Resignation of Administrative Agent.

The Administrative Agent may, upon giving five (5) Business Days prior written notice to the Borrower, resign at any time by giving notice of its resignation to the Lenders, the Issuers and the Borrower. Upon receipt of any such notice of resignation, the Borrower shall have the right, with the consent of the Required Lenders, such consent not to be unreasonably withheld, conditioned or delayed, to appoint a successor, which successor shall be a Lender. If no such successor shall have been so appointed by the Borrower and shall have accepted such appointment within thirty (30) days after the retiring Administrative Agent gives notice of its resignation, then the retiring Administrative Agent may on behalf of the Lenders and the Issuers, appoint a successor Administrative Agent from among the Lenders meeting the qualifications set forth above; provided that if the Administrative Agent shall notify the Borrower and the Lenders that no qualifying Person has accepted such appointment, then such resignation shall nonetheless become effective in accordance with such notice and (1) the retiring Administrative Agent shall

84

be discharged from its duties and obligations hereunder and under the other Loan Documents and (2) all payments, communications and determinations provided to be made by, to or through the Administrative Agent shall instead be made by or to each Lender and each Issuer directly, until such time as the Borrower appoints a successor Administrative Agent as provided for above in this Section. Upon the acceptance of a successor's appointment as Administrative Agent hereunder, such successor shall succeed to and become vested with all of the rights, powers, privileges and duties of the retiring (or retired) Administrative Agent, and the retiring Administrative Agent shall be discharged from all of its duties and obligations hereunder or under the other Loan Documents (if not already discharged therefrom as provided above in this Section). The fees payable by the Borrower to a successor Administrative Agent shall be the same as those payable to its predecessor unless otherwise agreed between the Borrower and such successor. After the retiring Administrative Agent's resignation hereunder and under the other Loan Documents, the provisions of this Article and Section 10.6 shall continue in effect for the benefit of such retiring Administrative Agent, its sub-agents and their respective Related Parties in respect of any actions taken or omitted to be taken by any of them while the retiring Administrative Agent was acting as Administrative Agent.

Any resignation by Bank of America as Administrative Agent pursuant to this Section shall also constitute its resignation as an Issuer and as the Swing Line Lender. Upon the acceptance of a successor's appointment as Administrative Agent hereunder, (a) such successor shall succeed to and become vested with all of the rights, powers, privileges and duties of the retiring Issuer and the retiring Swing Line Lender, (b) the retiring Issuer and the retiring Swing Line Lender shall be discharged from all of their respective duties and obligations hereunder or under the other Loan Documents, and (c) the successor Issuer shall issue letters of credit in substitution for the Letters of Credit, if any, outstanding at the time of such succession or make other arrangements satisfactory to the retiring Issuer to effectively assu me the obligations of the retiring Issuer with respect to such Letters of Credit.

11.7 <u>Non-Reliance on Administrative Agent and Other Lenders.</u>

Each Lender, the Swing Line Lender and each Issuer acknowledges that it has, independently and without reliance upon the Administrative Agent or any other Lender or any of their Related Parties and based on such documents and information as it has deemed appropriate, made its own credit analysis and decision to enter into this Agreement. Each Lender, the Swing Line Lender and each Issuer also acknowledges that it will, independently and without reliance upon the Administrative Agent or any other Lender or any of their Related Parties and based on such documents and information as it shall from time

to time deem appropriate, continue to make its own decisions in taking or not taking action under or based upon this Agreement, any other Loan Document or any related agreement or any document furnished hereunder or thereunder.

11.8 No Other Duties, Etc.

Anything herein to the contrary notwithstanding, none of the Bookrunners or Arrangers listed on the cover page hereof shall have any powers, duties or responsibilities under this Agreement or any of the other Loan Documents, except in its capacity, as applicable, as the Administrative Agent, the Swing Line Lender, a Lender or an Issuer hereunder.

85

11.9 Administrative Agent May File Proofs of Claim.

In case of the pendency of any proceeding under any Debtor Relief Law or other judicial proceeding relative to the Borrower or the Guarantor, the Administrative Agent (irrespective of whether the principal of any Loan or Letter of Credit Obligation shall then be due and payable as herein expressed or by declaration or otherwise and irrespective of whether the Administrative Agent shall have made any demand on the Borrower or the Guarantor) shall be entitled and empowered, by intervention in such proceeding or otherwise

- (a) to file and prove a claim for the whole amount of the principal and interest owing and unpaid in respect of the Loans, Letter of Credit Obligations and all other Obligations that are owing and unpaid and to file such other documents as may be necessary or advisable in order to have the claims of the Lenders, the Swing Line Lender, the Issuers and the Administrative Agent (including any claim for the reasonable compensation, expenses, disbursements and advances of the Lenders, the Swing Line Lender, the Issuers and the Administrative Agent and their respective agents and counsel and all other amounts due the Lenders, the Issuers and the Administrative Agent under Sections 2.5, 2.19(d), and 10.6) allowed in such judicial proceeding; and
 - (b) to collect and receive any monies or other property payable or deliverable on any such claims and to distribute the same;

and any custodian, receiver, assignee, trustee, liquidator, sequestrator or other similar official in any such judicial proceeding is hereby authorized by each Lender, the Swing Line Lender and each Issuer to make such payments to the Administrative Agent and, in the event that the Administrative Agent shall consent to the making of such payments directly to the Swing Line Lender, the Lenders and the Issuers, to pay to the Administrative Agent any amount due for the reasonable compensation, expenses, disbursements and advances of the Administrative Agent and its agents and counsel, and any other amounts due the Administrative Agent under Sections 2.5 and 10.6.

Nothing contained herein shall be deemed to authorize the Administrative Agent to authorize or consent to or accept or adopt on behalf of any Lender, the Swing Line Lender or any Issuer any plan of reorganization, arrangement, adjustment or composition affecting the Obligations or the rights of any Lender or to authorize the Administrative Agent to vote in respect of the claim of any Lender in any such proceeding.

ARTICLE XII

SETOFF; RATABLE PAYMENTS; PAYMENTS SET ASIDE

12.1 Setoff.

In addition to, and without limitation of, any rights of the Lenders under applicable law, if the Borrower or the Guarantor becomes insolvent, however evidenced, or any Default occurs, after obtaining the prior written consent of the Administrative Agent, any and all deposits

86

(including all account balances, whether provisional or final and whether or not collected or available) and any other Indebtedness at any time held or owing by any Lender, any Issuer, the Swing Line Lender or any such Affiliate to or for the credit or account of the Borrower or the Guarantor may be offset and applied toward the payment of the obligations of the Borrower or the Guarantor owing to such Lender, the Issuer or the Swing Line Lender under this Agreement or any other Loan Document whether or not such obligations, or any part hereof, shall then be due; provided, that in the event that any Defaulting Lender shall exercise any such right of setoff, (x) all amounts so set off (excluding any amount set off in satisfaction of any Swap Contract) shall be paid over immedi ately to the Administrative Agent for further application in accordance with the provisions of Section 2.23 and, pending such payment, shall be segregated by such Defaulting Lender from its other funds and deemed held in trust for the benefit of the Administrative Agent and the Lenders, and (y) the Defaulting Lender shall provide promptly to the Administrative Agent a statement describing in reasonable detail the obligations (other than obligations in respect of any Swap Contract) owing to such Defaulting Lender as to which it exercised such right of setoff. Each Lender, each Issuer and the Swing Line Lender agree to notify the Borrower and the Administrative Agent promptly after any such setoff and application; provided that the failure to give such notice shall not affect the validity of such setoff and application.

12.2 Ratable Payments.

If any Lender, whether by setoff or otherwise, has payment made to it upon its Outstanding Credit Exposure (other than payments received pursuant to Section 3.1, 3.2, 3.4 or 3.5 and payments made to any Issuer in respect of Reimbursement Obligations so long as the Lenders have not funded their participations therein) in a greater proportion than that received by any other Lender, such Lender agrees, promptly upon demand, to purchase a portion of the Aggregate Outstanding Credit Exposure held by the other Lenders so that after such purchase each Lender will hold its Pro Rata Sha re of the Aggregate Outstanding Credit Exposure. If any Lender, whether in connection with setoff or amounts which might be subject to setoff or otherwise, receives collateral

or other protection for its Obligations or such amounts which may be subject to setoff, such Lender agrees, promptly upon demand, to take such action necessary such that all Lenders share in the benefits of such collateral ratably in accordance with their respective Pro Rata Shares. In case any such payment is disturbed by legal process, or otherwise, appropriate further adjustments shall be made. For the avoidance of doubt, the provisions of this Section shall not be construed to apply to (A) any payment made by or on behalf of the Borrower pursuant to and in accordance with the express terms of this Agreement (including the application of funds arising from the existence of a Defaulting Lender), (B) the application of Cash Collateral provided for in Section 2.22, or (C) any payment obtained by a Lender as consideration for the assignment of or sale of a participation in any of its Loans or subparticipations in Letter of Credit Obligations or Swing Line Loans to any assignee or participant, other than an assignment to the Borrower, the Guarantor or any Subsidiary thereof (as to which the provisions of this Section shall apply).

87

12.3 Payments Set Aside.

To the extent that any payment by or on behalf of the Borrower is made to the Administrative Agent, any Issuer or any Lender, or the Administrative Agent, any Issuer or any Lender exercises its right of setoff, and such payment or the proceeds of such setoff or any part thereof is subsequently invalidated, declared to be fraudulent or preferential, set aside or required (including pursuant to any settlement entered into by the Administrative Agent, such Issuer or such Lender in its discretion) to be repaid to a trustee, receiver or any other party, in connection with any proceeding under any Debtor Relief Law or otherwise, then (a) to the extent of such recovery, the obligation or part thereof originally intended to be satisfied shall be revived and continued in full force and effect as if such payment had not been made or such setoff had not occurred, and (b) each Lender and each Issuer severally agrees to pay to the Administrative Agent upon demand its applicable share (without duplication) of any amount so recovered from or repaid by the Administrative Agent, plus interest thereon from the date of such demand to the date such payment is made at a rate per annum equal to the Federal Funds Rate from time to time in effect. The obligations of the Lenders and the Issuers under clause (b) of the preceding sentence shall survive the payment in full of the Obligations and the termination of this Agreement.

ARTICLE XIII

BENEFIT OF AGREEMENT; ASSIGNMENTS; PARTICIPATIONS

13.1 Successors and Assigns.

- (a) The provisions of this Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns permitted hereby, except that the Borrower may not assign or otherwise transfer any of its rights or obligations hereunder without the prior written consent of the Administrative Agent, the Swing Line Lender and each Lender and no Lender may assign or otherwise transfer any of its rights or obligations hereunder except (i) to an Eligible Assignee in accordance with the provisions of subsection (b) of this Section, (ii) by way of participation in accordance with the provisions of subsection (d) of the is Section, or (iii) by way of pledge or assignment of a security interest subject to the restrictions of subsection (f) of this Section (and any other attempted assignment or transfer by any party hereto shall be null and void). Nothing in this Agreement, expressed or implied, shall be construed to confer upon any Person (other than the parties hereto, their respective successors and assigns permitted hereby, Participants to the extent provided in subsection (d) of this Section and, to the extent expressly contemplated hereby, the Related Parties of each of the Administrative Agent, the Issuers and the Lenders) any legal or equitable right, remedy or claim under or by reason of this Agreement.
- (b) Any Lender may at any time assign to one or more Eligible Assignees all or a portion of its rights and obligations under this Agreement (including all or a portion of its Commitment and the Loans (including for purposes of this <u>subsection (b)</u> participations in Letter of Credit Obligations) at the time owing to it); <u>provided</u> that

88

- (i) except in the case of an assignment of the entire remaining amount of the assigning Lender's Commitment and the Loans at the time owing to it or in the case of an assignment to a Lender or an Affiliate of a Lender or an Approved Fund with respect to a Lender, the aggregate amount of the Commitment (which for this purpose includes Loans outstanding thereunder) or, if the Commitment is not then in effect, the principal outstanding balance of the Loans of the assigning Lender subject to each such assignment, determined as of the date the Assignment Agreement with respect to such assignment is delivered to the Administrative Agent or, if "Trade Date" is specified in the Assignment Agreement, as of the Trade Date, shall not be less than \$5,000,000 unless each of the Administrative Agent and, so long as no Default has occurred and is continuing, the Borrower otherwise consents (each such consent not to be unreasonably withheld or delayed); provided, however, that concurrent assignments to members of an Assignee Group and concurrent assignments from members of an Assignee Group to a single Eligible Assignee (or to an Eligible Assignee and members of its Assignee Group) will be treated as a single assignment for purposes of determining whether such minimum amount has been met; provided, further, that the Borrower shall be deemed to have consented to any such assignment unless it shall object thereto by written notice to the Administrative Agent within ten (10) Business Days after having received notice thereof.
- (ii) each partial assignment shall be made as an assignment of a proportionate part of all the assigning Lender's rights and obligations under this Agreement with respect to the Loans or the Commitment assigned;
- (iii) any assignment of a Commitment must be approved by the Administrative Agent, the Swing Line Lender and the Issuers (each such approval not to be unreasonably withheld or delayed) unless the Person that is the proposed assignee is itself a Lender, an Affiliate of such Lender or an Approved Fund with respect to such Lender;

- (iv) (A) such Lender shall at the same time enter into an Assignment Agreement (as defined in the KCPL Credit Agreement) with the same Eligible Assignee(s) in an amount representing an equal proportion of such Lender's Commitment (as defined in the KCPL Credit Agreement) under the KCPL Credit Agreement and (B) such Lender shall at the same time enter into an Assignment Agreement (as defined in the Great Plains Credit Agreement) with the same Eligible Assignee(s) in an amount representing an equal proportion of such Lender's Commitment (as defined in the Great Plains Credit Agreement) under the Great Plains Credit Agreement;
- (v) the parties to each assignment shall execute and deliver to the Administrative Agent an Assignment Agreement, together with a processing and recordation fee of \$3,500; provided, however, that (A) such processing and recordation fee shall not apply to assignments to Affiliates of the assigning Lender, (B) such processing and recordation fee shall, with respect to a particular assignment, be payable only once by such assigning Lender in connection with the concurrent assignments of Commitments of such assigning Lender required by Section 13.1(b)(iv) and (C) the Administrative Agent may, in its sole discretion, elect to waive such processing and

89

recordation fee in the case of any assignment. The Eligible Assignee, if it shall not be a Lender, shall deliver to the Administrative Agent an Administrative Questionnaire;

- (vi) No such assignment shall be made (A) to the Borrower, the Guarantor or any of the Borrower's or the Guarantor's Affiliates or Subsidiaries, (B) to any Defaulting Lender or any of its Subsidiaries, or any Person who upon becoming a Lender hereunder would constitute any of the foregoing Persons described in this <u>clause (B)</u>, or (C) to a natural person; and
- (vii) In connection with any assignment of rights and obligations of any Defaulting Lender hereunder, no such assignment shall be effective unless and until, in addition to the other conditions thereto set forth herein, the parties to the assignment shall make such additional payments to the Administrative Agent in an aggregate amount sufficient, upon distribution thereof as appropriate (which may be outright payment, purchases by the assignee of participations or subparticipations, or other compensating actions, including funding, with the consent of the Borrower and the Administrative Agent, the applicable pro rata share of Loans previously requested but not funded by the Defaulting Lender, to each of which the applicable assignee and assignor hereby irrevocably consent), to (x) pay and satisfy in full all payment liabilities then owed by such Defaulting Lender to the Administrative Agent or any Lender hereunder (and interest accrued thereon) and (y) acquire (and fund as appropriate) its full pro rata share of all Loans and participations in Letters of Credit and Swing Line Loans in accordance with its Pro Rata Share. Notwithstanding the foregoing, in the event that any assignment of rights and obligations of any Defaulting Lender hereunder shall become effective under applicable Law without compliance with the provisions of this paragraph, then the assignee of such interest shall be deemed to be a Defaulting Lender for all purposes of this Agreement until such compliance occurs.

Subject to acceptance and recording thereof by the Administrative Agent pursuant to <u>subsection</u> (c) of this Section, from and after the effective date specified in each Assignment Agreement, the Eligible Assignee thereunder shall be a party to this Agreement and, to the extent of the interest assigned by such Assignment Agreement, have the rights and obligations of a Lender under this Agreement, and the assigning Lender thereunder shall, to the extent of the interest assigned by such Assignment Agreement, be released from its obligations under this Agreement (and, in the case of an Assignment Agreement covering all of the assigning Lender's rights and obligations under this Agreement, such Lender shall cease to be a party hereto) but shall continue to be entitled to the benefits of Sections 3.1, 3.4, 3.5, and 10.6 with respect to facts and circumstances occurring prior to the effective date of such assignment. Upon request, the Borrower (at its expense) shall execute and deliver a Note to the assignee Lender. Any assignment or transfer by a Lender of rights or obligations under this Agreement that does not comply with this subsection shall be treated for purposes of this Agreement as a sale by such Lender of a participation in such rights and obligations in accordance with <u>subsection (d)</u> of this Section.

(c) The Administrative Agent, acting solely for this purpose as an agent of the Borrower, shall maintain at the Administrative Agent's Office a copy of each Assignment

90

Agreement delivered to it and a register for the recordation of the names and addresses of the Lenders, and the Commitments of, and principal amounts of the Loans and Letter of Credit Obligations owing to, each Lender pursuant to the terms hereof from time to time (the "Register"). The entries in the Register shall be conclusive, and the Borrower, the Administrative Agent and the Lenders may treat each Person whose name is recorded in the Register pursuant to the terms hereof as a Lender hereunder for all purposes of this Agreement, notwithstanding notice to the contrary. In addition, the Administrative Agent shall maintain on the Register information regarding the designation, and revocation of designation, of any Lender as a Defaulting Lender. & #160;The Register shall be available for inspection by each of the Borrower and the Issuers at any reasonable time and from time to time upon reasonable prior notice. In addition, at any time that a request for a consent for a material or substantive change to the Loan Documents is pending, any Lender may request and receive from the Administrative Agent a copy of the Register.

(d) Any Lender may at any time, without the consent of, or notice to, the Borrower or the Administrative Agent, sell participations to any Person (other than a natural person, a Defaulting Lender, or the Borrower or the Guarantor or any of the Borrower's or the Guarantor's Affiliates or Subsidiaries) (each, a "Participant") in all or a portion of such Lender's rights and/or obligations under this Agreement (including all or a portion of its Commitment and/or the Loans (including such Lender's participations in Letter of Credit Obligations) owing to it); provided that (i) such Lender's obligations under this Agreement shall remain unchanged, (ii) such Lender shall remain solely responsible to the other parties hereto for the performance of such obligations and (iii) the Borrower, the Administrative Agent, the Lenders and the Issuers shall continue to deal solely and directly with such Lender in connection with such Lender's rights and obligations under this Agreement.

Any agreement or instrument pursuant to which a Lender sells such a participation shall provide that such Lender shall retain the sole right to enforce this Agreement and to approve any amendment, modification or waiver of any provision of this Agreement; provided that such agreement or instrument may provide that such Lender will not, without the consent of the Participant, agree to any amendment, waiver or other modification described in

the proviso to Section 9.2 that affects such Participant. Subject to subsection (e) of this Section, the Borrower agrees that each Participant shall be entitled to the benefits of Sections 3.1, 3.4 and 3.5 to the same extent as if it were a Lender and had acquired its interest by assignment pursuant to subsection (b) of this Section. To the extent permitted by law, each Participant also shall be entitled to the benefits of Section 12.1 as though it were a Lender, provided such Participant agrees to be subject to Section 12.2 as though it were a Lender.

(e) A Participant shall not be entitled to receive any greater payment under <u>Section 3.1</u> or <u>3.5</u> than the applicable Lender would have been entitled to receive with respect to the participation sold to such Participant, unless the sale of the participation to such Participant is made with the Borrower's prior written consent. A Participant that would be a Foreign Lender if it were a Lender shall not be entitled to the benefits of <u>Section 3.5</u> unless the Borrower is notified of the participation so ld to such Participant and such Participant agrees, for the benefit of the Borrower, to comply with <u>Section 3.5(e)</u> as though it were a Lender.

91

- (f) Any Lender may at any time pledge or assign a security interest in all or any portion of its rights under this Agreement (including under its Note, if any) to secure obligations of such Lender, including any pledge or assignment to secure obligations to a Federal Reserve Bank; <u>provided</u> that no such pledge or assignment shall release such Lender from any of its obligations hereunder or substitute any such pledgee or assignee for such Lender as a party hereto.
- (g) The words "execution," "signed," "signature," and words of like import in any Assignment Agreement shall be deemed to include electronic signatures or the keeping of records in electronic form, each of which shall be of the same legal effect, validity or enforceability as a manually executed signature or the use of a paper-based recordkeeping system, as the case may be, to the extent and as provided for in any applicable law, including the Federal Electronic Signatures in Global and National Commerce Act, the New York State Electronic Signatures and Records Act, or any other similar state laws based on the Uniform Electronic Transactions Act.
- (h) Notwithstanding anything to the contrary contained herein, if at any time Bank of America assigns all of its Commitment and Loans pursuant to <u>subsection (b)</u> above, Bank of America may, upon thirty (30) days' notice to the Borrower and the Lenders, resign as an Issuer. In the event of any such resignation as an Issuer, the Borrower shall be entitled to appoint from among the Lenders another Issuer hereunder who agrees to such appointment; <u>provided</u>, <u>however</u>, that no failure by the Borrower to appoint any such Issuer, or the failure of any Lender to accept suc h appointment, shall affect the resignation of Bank of America as an Issuer. If Bank of America resigns as an Issuer, it shall retain all the rights, powers, privileges and duties of an Issuer hereunder with respect to all Letters of Credit outstanding as of the effective date of its resignation as Issuer and all Letter of Credit Obligations with respect thereto (including the right to require the Lenders to fund risk participations pursuant to <u>Section 2.19(e)</u>).

13.2 Replacement of Lenders.

If (i) any Lender requests compensation under Section 3.1 or Section 3.2, (ii) suspends the availability of Eurodollar Loans under Section 3.3, (iii) the Borrower is required to pay any additional amount to any Lender or any Governmental Authority for the account of any Lender pursuant to Section 3.4 or Section 3.5, (iv) any Lender is a Non-Extending Lender pursuant to Section 2.21(b) or (v) any Lender is a Defaulting Lender, then the Borrower may, at its sole expense and effort, upon notice to such Lender and the Administrative Agent, require such Lender to assign and delegate, without recourse (in accordance with and subject to the restrictions contained in, and consents required by, Section 13.1(b)), all of its interests, rights and obligations under this Agreement and the related Loan Documents to an Eligible Assignee that shall assume such obligations (which Eligible Assignee may be another Lender, if a Lender accepts such assignment), provided that:

(a) the Borrower shall have paid to the Administrative Agent the assignment fee specified in Section 13.1(b);

92

- (b) such Lender shall have received payment of an amount equal to the outstanding principal of its Loans and participations, accrued interest thereon, accrued fees and all other amounts payable to it hereunder and under the other Loan Documents (including any amounts payable under Section 3.4) from such Eligible Assignee (to the extent of such outstanding principal and accrued interest and fees) or the Borrower (in the case of all other amounts);
- (c) in the case of any such assignment resulting from a claim for compensation under <u>Section 3.1</u> or payments required to be made pursuant to <u>Section 3.4</u>, such assignment will result in a reduction in such compensation or payments thereafter; and
 - (d) such assignment does not conflict with applicable Laws.

A Lender shall not be required to make any such assignment or delegation if, prior thereto, as a result of a waiver by such Lender or otherwise, the circumstances entitling the Borrower to require such assignment and delegation cease to apply.

ARTICLE XIV

NOTICES

Except as otherwise permitted by Section 2.14 with respect to borrowing notices, all notices, requests and other communications to any party hereunder shall be in writing (including electronic transmission, facsimile transmission or similar writing) and shall be given to such party: (i) in the case of the Borrower, the Guarantor or the Administrative Agent, at its address or facsimile number set forth on Schedule IV, (ii) in the case of any Lender, at its address or facsimile number specified in its Administrative Questionnaire or (iii) in the case of any party, at such other address or facsimile number as such party may hereafter specify for the purpose by notice to the Administrative Agent and the Borro wer in accordance with the provisions of this Section 14.1. Each such notice, request or other communication shall be effective (a) if given by facsimile transmission, when transmitted to the facsimile number specified in this Section and confirmation of receipt is received, (b) if given by mail, when received, or (c) if given by any other means, when delivered (or, in the case of electronic transmission, received) at the address specified in this Section; provided that notices to the Administrative Agent under Article II shall not be effective until received.

14.2 <u>Change of Address</u>.

The Borrower, the Administrative Agent and any Lender may each change the address for service of notice upon it by a notice in writing to the other parties hereto.

93

14.3 The Platform.

THE PLATFORM IS PROVIDED "AS IS" AND "AS AVAILABLE." THE AGENT PARTIES (AS DEFINED BELOW) DO NOT WARRANT THE ACCURACY OR COMPLETENESS OF THE BORROWER MATERIALS OR THE ADEQUACY OF THE PLATFORM, AND EXPRESSLY DISCLAIM LIABILITY FOR ERRORS IN OR OMISSIONS FROM THE BORROWER MATERIALS. NO WARRANTY OF ANY KIND, EXPRESS, IMPLIED OR STATUTORY, INCLUDING ANY WARRANTY OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE, NON INFRINGEMENT OF THIRD PARTY RIGHTS OR FREEDOM FROM VIRUSES OR OTHER CODE DEFECTS, IS MADE BY ANY AGENT PARTY IN CONNECTION WITH THE BORROWER MATERIALS OR THE PLATFORM. In no event shall the Administrative Agent or any of its Related Parties (collectively, the "Agent Parties") have any liability to Borrower, any Lender, any Issuer or any other Person for losses, claims, damages, liabilities or expenses of any kind (whether in tort, contract or otherwise) arising out of Borrower's or the Administrative Agent's transmission of Borrower Materials through the Internet, except to the extent that such losses, claims, damages, liabilities or expenses are determined by a court of competent jurisdiction by a final and nonappealable judgment to have resulted from the gross negligence or willful misconduct of such Agent Party; provided, however, that in no event shall any Agent Party have any liability to Borrower, any Lender, any Issuer or any other Person for indirect, special, incidental, consequential or punitive damages (as opposed to direct or actual damages).

ARTICLE XV

COUNTERPARTS

This Agreement may be executed in any number of counterparts, all of which taken together shall constitute one agreement, and any of the parties hereto may execute this Agreement by signing any such counterpart. This Agreement shall be effective when it has been executed by the Borrower, the Guarantor, the Administrative Agent and the Lenders and each party has notified the Administrative Agent by facsimile transmission or telephone that it has taken such action.

ARTICLE XVI

OTHER AGENTS

No Lender identified on the cover page, the signature pages or otherwise in this Agreement, or in any document related hereto, as being the "Syndication Agent" or a "Co-Documentation Agent" shall have any right, power, obligation, liability, responsibility or duty under this Agreement in such capacity other than those applicable to all Lenders. Each Lender acknowledges that it has not relied, and will not rely, on the Syndication Agent or any Co-Documentation Agent in deciding to enter into this Agreement or in taking or refraining from taking any action hereunder or pursuant hereto.

94

ARTICLE XVII

CHOICE OF LAW; CONSENT TO JURISDICTION; WAIVER OF JURY TRIAL.

17.1 CHOICE OF LAW.

THIS AGREEMENT SHALL BE CONSTRUED IN ACCORDANCE WITH THE INTERNAL LAWS (AND NOT THE LAW OF CONFLICTS) OF THE STATE OF NEW YORK, BUT GIVING EFFECT TO FEDERAL LAWS APPLICABLE TO NATIONAL BANKS.

17.2 <u>CONSENT TO JURISDICTION</u>.

THE BORROWER, THE GUARANTOR, THE ADMINISTRATIVE AGENT, THE SWING LINE LENDER, EACH LENDER AND EACH ISSUER HEREBY IRREVOCABLY SUBMIT TO THE NON-EXCLUSIVE JURISDICTION OF ANY UNITED STATES FEDERAL OR NEW YORK STATE COURT SITTING IN NEW YORK CITY, NEW YORK IN ANY ACTION OR PROCEEDING ARISING OUT OF OR RELATING TO THIS AGREEMENT AND THE BORROWER, THE GUARANTOR, THE ADMINISTRATIVE AGENT, THE SWING LINE LENDER, EACH LENDER AND EACH ISSUER HEREBY IRREVOCABLY AGREE THAT ALL CLAIMS IN RESPECT OF SUCH ACTION OR PROCEEDING MAY BE HEARD AND DETERMINED IN ANY SUCH COURT AND IRREVOCABLY WAIVE ANY OBJECTION THEY MAY NOW OR HEREAFTER HAVE AS TO THE VENUE OF ANY SUCH SUIT, ACTION OR PROCEEDING BROUGHT IN SUCH A COURT OR THAT SUCH COURT IS AN INCONVENIENT FORUM. NOTHING HEREIN SHALL LIMIT THE RIGHT OF ANY PARTY TO BRING PROCEEDINGS AGAINST THE OTHER PARTIES IN THE COURTS OF ANY OTHER JURISDICTION. ANY JUDICIAL PROCEEDING INVOLVING, DIRECTLY OR INDIRECTLY, ANY MATTER IN ANY WAY ARISING OUT OF, RELATED TO, OR CONNECTED WITH THIS AGREEMENT SHALL BE BROUGHT ONLY IN A COURT IN NEW YORK CITY, NEW YORK.

17.3 <u>WAIVER OF JURY TRIAL</u>.

THE BORROWER, THE GUARANTOR, THE ADMINISTRATIVE AGENT, THE SWING LINE LENDER, EACH LENDER AND EACH ISSUER HEREBY WAIVE TRIAL BY JURY IN ANY JUDICIAL PROCEEDING INVOLVING, DIRECTLY OR INDIRECTLY, ANY MATTER (WHETHER SOUNDING IN TORT, CONTRACT OR OTHERWISE) IN ANY WAY ARISING OUT OF, RELATED TO, OR CONNECTED WITH ANY LOAN DOCUMENT OR THE RELATIONSHIP ESTABLISHED THEREUNDER.

95

ARTICLE XVIII

TERMINATION OF EXISTING CREDIT FACILITY

Lenders which are parties to the Existing Credit Facility (and which constitute "Required Lenders" under and as defined in the Existing Credit Facility) hereby waive any advance notice requirement for terminating the commitments under the Existing Credit Facility, and the Borrower and the applicable Lenders agree that the Existing Credit Facility and the commitments thereunder shall be terminated on the date hereof (except for any provisions thereof which by their terms survive termination thereof).

96

IN WITNESS WHEREOF, the Borrower, the Guarantor, the Lenders, the Swing Line Lender, the Issuers and the Agents have executed this Agreement as of the date first above written.

KCP&L GREATER MISSOURI OPERATIONS COMPANY, as the Borrower

By: /s/ Michael W. Cline Name: Michael W. Cline

Title: Vice President – Investor Relations and

Treasurer and Corporate Secretary

GREAT PLAINS ENERGY INCORPORATED, as the Guarantor

By: /s/ Michael W. Cline Name: Michael W. Cline

Title: Vice President – Investor Relations and

Treasurer and Corporate Secretary

BANK OF AMERICA, N.A., as Administrative Agent

By: /s/ Kimberly Williams
Name: Kimberly Williams
Title: Vice President

BANK OF AMERICA, N.A., as an Issuer, as Swing Line Lender and as a Lender

By: /s/ Kevin P. Bertelsen
Name: Kevin P. Bertelsen
Title: Senior Vice President

THE BANK OF TOKYO-MITSUBISHI UFJ, LTD., as an Issuer and as a Lender

By: /s/ Bradford Joyce Name: Bradford Joyce Title: Authorized Signatory

WELLS FARGO BANK, NATIONAL ASSOCIATION, as Syndication Agent, as an Issuer and as a Lender

By: /s/ Allison Newman Name: Allison Newman Title: Vice President

UNION BANK, N.A., as Syndication Agent and as a Lender

By: /s/ Susan K. Johnson Name: Susan K. Johnson Title: Vice President JPMORGAN CHASE BANK, N.A., as a Lender

By: /s/ Nancy R. Barwig
Name: Nancy R. Barwig
Title: Senior Vice President

BNP PARIBAS, as a Lender

By: /s/ Pasquale A. Perraglia IV Name: Pasquale A. Perraglia IV

Title: Vice President

By: /s/ Denis O'Meara Name: Denis O'Meara Title: Managing Director

THE BANK OF NOVA SCOTIA, as a Lender

By: /s/ Thane Rattew
Name: Thane Rattew
Title: Managing Director

THE ROYAL BANK OF SCOTLAND PLC, as a Lender

By: /s/ Tyler J. McCarthy Name: Tyler J. McCarthy

Title: Director

BARCLAYS BANK PLC, as a Lender

By: /s/ David Barton
Name: David Barton
Title: Director

U.S. BANK NATIONAL ASSOCIATION

as a Lender

By: /s/ Eric J. Cosgrove Name: Eric J. Cosgrove Title: Vice President

THE BANK OF NEW YORK MELLON,

as a Lender

By: /s/ Hussam S. Alsahlani Name: Hussam S. Alsahlani Title: Senior Associate

KEYBANK NATIONAL ASSOCIATION,

as a Lender

By: /s/ Keven D. Smith Name: Keven D. Smith Title: Senior Vice President

SUNTRUST BANK, as a Lender

By: /s/ Andrew Johnson Name: Andrew Johnson

Title: Director

DEUTSCHE BANK AG NEW YORK BRANCH, as a Lender

By: /s/ Frederick Laird Name: Frederick Laird Managing Director Title:

/s/ Heidi Sandquist By: Name: Heidi Sandquist Director

Title:

UMB BANK, N.A., as a Lender

By: /s/ Robert P. Elbert Name: Robert P. Elbert Title: Senior Vice President COMMERCE BANK, N.A., as a Lender

By: /s/ Aaron M. Siders Name: Aaron M. Siders

Title: Commercial Banking Officer

SCHEDULE I

COMMITMENTS

Lender	Commitment
Bank of America, N.A.	\$43,200,000.00
Wells Fargo Bank, National Association	\$43,200,000.00
The Bank of Tokyo-Mitsubishi UFJ, Ltd.	\$21,600,000.00
Union Bank, N.A.	\$21,600,000.00
JPMorgan Chase Bank, N.A.	\$30,600,000.00
BNP Paribas	\$30,600,000.00
The Bank of Nova Scotia	\$30,600,000.00
The Royal Bank of Scotland plc	\$30,600,000.00
Barclays Bank PLC	\$30,600,000.00
U.S. Bank National Association	\$30,600,000.00
The Bank of New York Mellon	\$27,000,000.00
KeyBank National Association	\$27,000,000.00
SunTrust Bank	\$27,000,000.00
Deutsche Bank AG New York Branch	\$27,000,000.00
UMB Bank, n.a.	\$18,000,000.00
Commerce Bank, N.A.	\$10,800,000.00
Total	\$450,000,000.00

LETTER OF CREDIT COMMITMENTS

Issuer	Letter of Credit Commitment
Bank of America, N.A.	\$16,666,666.67
Wells Fargo Bank, National Association	\$16,666,666.67
The Bank of Tokyo-Mitsubishi UFJ, Ltd.	\$16,666,666.66
Total	\$50,000,000.00

SCHEDULE II

EXISTING LETTERS OF CREDIT

Issuer	Beneficiary	Amount	Expiry Date	Letter of Credit Number
Bank of America, N.A.	Union Bank, N.A.	\$ 5,375,000.00	09/17/11	T00000068033545
Bank of America, N.A.	Union Bank, N.A.	\$ 7,408,000.00	09/17/11	T00000068033546
Bank of America, N.A.	National Union Fire	\$ 377,700.00	08/31/10	T00000068045558

SCHEDULE III

PRICING SCHEDULE

	>A-/A3	>BBB+/Baa1	>BBB/Baa2	>BBB-/Baa3	>BB+/Ba1	<bb ba2<="" th=""></bb>
Pricing	LEVEL I STATUS	LEVEL II STATUS	LEVEL III STATUS	LEVEL IV STATUS	Level V Status	LEVEL VI STATUS
Applicable Margin for Eurodollar Advances and Letter of Credit Fee Rate	1.75%	2.00%	2.25%	2.75%	3.00%	3.25%
Commitment Fee Rate	0.225%	0.30%	0.375%	0.50%	0.625%	0.625%
Applicable Margin for Floating Rate Advances	0.75%	1.00%	1.25%	1.75%	2.00%	2.25%

"Level I Status" exists at any date if, on such date, the Guarantor's Moody's Rating is A3 or better *or* the Guarantor's S&P Rating is A- or better.

"Level II Status" exists at any date if, on such date, (i) the Guarantor has not qualified for Level I Status and (ii) the Guarantor's Moody's Rating is Baa1 or better *or* the Guarantor's S&P Rating is BBB+ or better.

"Level III Status" exists at any date if, on such date, (i) the Guarantor has not qualified for Level I Status or Level II Status and (ii) the Guarantor's Moody's Rating is Baa2 or better *or* the Guarantor's S&P Rating is BBB or better.

"Level IV Status" exists at any date if, on such date, (i) the Guarantor has not qualified for Level I Status, Level II Status or Level III Status and (ii) the Guarantor's Moody's Rating is Baa3 or better *or* the Guarantor's S&P Rating is BBB- or better.

"Level V Status" exists at any date if, on such date, (i) the Guarantor has not qualified for Level I Status, Level II Status, Level III Status or Level IV Status and (ii) the Guarantor's Moody's Rating is BB+ or better *or* the Guarantor's S&P Rating is Ba1 or better.

"Level VI Status" exists at any date if, on such date, the Guarantor has not qualified for Level I Status, Level II Status, Level III Status, Level IV Status or Level V Status.

"Moody's Rating" means, at any time, the rating issued by Moody's and then in effect with respect to the Guarantor's senior unsecured long-term debt securities without third-party credit enhancement (or, if there is no such debt outstanding, the Guarantor's issuer rating issued by Moody's then in effect for the Guarantor).

"S&P Rating" means, at any time, the rating issued by S&P and then in effect with respect to the Guarantor's senior unsecured long-term debt securities without third-party credit enhancement (or, if there is no such debt outstanding, the Guarantor's issuer rating issued by S&P then in effect for the Guarantor).

"Status" means Level I Status, Level II Status, Level III Status, Level IV Status, Level V Status or Level VI Status.

The Applicable Margin for Eurodollar Advances, the Letter of Credit Fee Rate, the Applicable Margin for Floating Rate Advances and the Commitment Fee Rate shall be determined in accordance with the foregoing pricing grid based on the Guarantor's Status as determined from its then-current Moody's and S&P Ratings. The credit rating in effect on any date for purposes of this Schedule is that in effect at the close of business on such date. If at any time the Guarantor ceases to have a Moody's Rating or an S&P Rating, Level VI Status shall exist.

Notwithstanding the foregoing, (a) if the Guarantor is split-rated and the ratings differential is one level, the higher rating will apply; and (b) if the Guarantor is split-rated and the ratings differential is two levels or more, the intermediate rating at the midpoint will apply. If there is no midpoint, the higher of the two intermediate ratings will apply.

SCHEDULE IV

CERTAIN ADDRESSES FOR NOTICES

BORROWER AND GUARANTOR:

1200 Main Street

Kansas City, MO 64105

Attention: Michael W. Cline, Vice-President – Investor Relations, Treasurer and Corporate Secretary

Telephone: (816) 556-2622 Facsimile: (816) 556-2992 Electronic Mail: michael.cline@kcpl.com

ADMINISTRATIVE AGENT:

Administrative Agent's Office

(for payments and Requests for Credit Extensions):

Bank of America, N.A. Credit Services Rep. II Mail Code: TX1-492-14-12 Bank of America Plaza 901 Main Street

Dallas, TX 75202-3714

Attention: Karen Puente Telephone: 214-209-4108 Telecopier: 214-290-8378

Electronic Mail: Karen.r.puente@baml.com

Bank of America New York, NY Account No.: 1292000883 Account Name: Credit Services

Ref: Great Plains ABA# 026009593

Other Notices as Administrative Agent:

Bank of America, N.A. Agency Management Bank of America Plaza 101 S. Tryon Street Charlotte, NC 28255-0001

Mail Code: NC1-002-15-36

Attention: Kelly Weaver Telephone: (980) 387-5452 (704) 208-2871 Facsimile:

Electronic Mail: kelly.weaver@baml.com

ISSUERS:

Bank of America, N.A. **Trade Operations** 1000 West Temple Street, 7^{th} Floor Mail Code: CA9-705-07-05 Los Angeles, CA 90012 1514

Attention: Sandra Leon Telephone: (213) 580-8369 Telecopier: (213) 457-8841 Electronic Mail: sandra.leon@baml.com

The Bank of Tokyo-Mitsubishi UFJ, Ltd. 1251 Avenue of the Americas, 12th Floor

New York, NY 10020-1104

Attention: Jaya Angara Telephone: (201) 413-8843 Telecopier: (201) 413-8817 Electronic Mail: jangara@us.mufg.jp Attention:

Telephone:

Telecopier:

Electronic Mail: abondi@us.mufg.jp

Antonina Bondi

(201) 413-8823 (201) 413-8817

Wells Fargo Bank, National Association 301 S. College Street, 15th Floor Charlotte, NC 28288 MAC: D1053-150

Attention: Elaine Shue Telephone: (704) 715-3133 (704) 715-0358 Telecopier: Electronic Mail: Elaine.Shue@wachovia.com

EXHIBIT A

COMPLIANCE CERTIFICATE

To: The Lenders parties to the Credit Agreement Described Below

This Compliance Certificate is furnished pursuant to the Credit Agreement dated as of August 9, 2010 (as amended, modified, renewed or extended from time to time, the "Agreement") among KCP&L Greater Missouri Operations Company (the "Borrower"), Great Plains Energy Incorporated (the "Guarantor"), the lenders from time to time party thereto (the "Lenders"), Union Bank, N.A. and Wells Fargo Bank, National Association, as Syndication Agents, and Bank of America, N.A., as Administrative Agent for the Lenders. Unless otherwise defined herein, capitalized terms used in this Compliance Certificate have the meanings ascribed thereto in the Agreement.

5 · · · · · · · · · · · · · · · · · · ·
THE UNDERSIGNED HEREBY CERTIFIES THAT:
1. I am the duly electedof the Borrower;
2. I have reviewed the terms of the Agreement and I have made, or have caused to be made under my supervision, an appropriate review of the transactions and conditions of the Borrower and, to the extent relevant to the Agreement, its Subsidiaries during the accounting period covered by the financial statements attached as Schedule I;
3. The examinations described in paragraph 2 did not disclose, and I have no knowledge of, the existence of any condition or event which constitutes a Default or Unmatured Default during or at the end of the accounting period covered by the attached financial statements or as of the date of the Certificate, except as set forth on Schedule II which lists, in detail, the nature of the condition or event, the period during which it has existed and the action which the Borrower has taken, is taking, or proposes to take with respect to each such condition or event; and
4. Set forth on <u>Schedule III</u> are true and correct computations of the financial ratios and restrictions set forth in Section 7.15 of the Agreemen
The foregoing certifications, together with the financial statements delivered with this Certificate in support hereof, are made and delivered this day of,
KCP&L GREATER MISSOURI OPERATIONS COMPANY
By: Print Name: Title:
SCHEDULE I
TO COMPLIANCE CERTIFICATE
Reports and Deliveries Currently Due
SCHEDULE II
TO COMPLIANCE CERTIFICATE

Conditions or Events which Constitute a Default or Unmatured Default

TO COMPLIANCE CERTIFICATE

Computation of Financial Covenants

EXHIBIT B

ASSIGNMENT AGREEMENT

This Assignment and Assumption (the "Assignment and Assumption") is dated as of the Effective Date set forth below and is entered into by and between [*Insert name of Assignor*] (the "Assignor") and [*Insert name of Assignee*] (the "Assignee"). Capitalized terms used but not defined herein shall have the respective meanings given to them in the Credit Agreement identified below (as amended, the "Credit Agreement"), receipt of a copy of which is here by acknowledged by the Assignee. The Terms and Conditions set forth in Annex 1 attached hereto (the "Terms and Conditions") are hereby agreed to and incorporated herein by reference and made a part of this Assignment and Assumption as if set forth herein in full.

For an agreed consideration, the Assignor hereby irrevocably sells and assigns to the Assignee, and the Assignee hereby irrevocably purchases and assumes from the Assignor, subject to and in accordance with the Terms and Conditions and the Credit Agreement, as of the Effective Date inserted by the Administrative Agent as contemplated below, the interest in and to all of the Assignor's rights and obligations in its capacity as a Lender under the Credit Agreement and any other documents or instruments delivered pursuant thereto that represents the amount and percentage interest identified below of all of the Assignor's outstanding rights and obligations under the respective facilities identified below (including any letters of credit, swing line loans and guaranties included in such facilities and, to the extent permitted to be assigned under applicable law, all claims (including contract claims, tort claims, malpractice claims, statutory claims and all other claims at law or in equity), suits, causes of action and any other right of the Assignor against any Person whether known or unknown arising under or in connection with the Credit Agreement, any other documents or instruments delivered pursuant thereto or the loan transactions governed thereby) (the "Assigned Interest"). Such sale and assignment is without recourse to the Assignor and, except as expressly provided in this Assignment and Assumption, without representation or warranty by the Assignor.

l.	Assignor:
----	-----------

2. Assignee:

8. Borrower: KCP&L Greater Missouri Operations Company

4. Administrative Agent: Bank of America, N.A., as the Administrative Agent under the Credit Agreement.

5. Credit Agreement: Credit Agreement dated as of August 9, 2010 among

the Borrower, Great Plains Energy Incorporated (the "Guarantor"), the Lenders from time to time party thereto and Bank of America, N.A., as Administrative Agent.

6. Assigned Interest:

Aggregate Amount of Commitment/Loans for	or Amount of Commitment	t/Loans Assigned* Percentage Assigned of Commitment/Loans ¹
all Lenders [*]		
\$	\$	%
\$	\$	%
\$	\$	%

7	Trade Date:	2
/ .	HAUE DAIE.	

Effective Date: , 20

[TO BE INSERTED BY AGENT AND WHICH SHALL BE THE EFFECTIVE DATE OF RECORDATION OF TRANSFER BY THE ADMINISTRATIVE AGENT.]

The terms set forth in this Assignment and Assumption are hereby agreed to:

ASSIGNOR [NAME OF ASSIGNOR]

Ву:			
Dy			
Title			

ASSIGNEE [NAME OF ASSIGNEE] By: _ Title: [Consented to and] ³ Accepted: BANK OF AMERICA, N.A., as Administrative Agent By: Title: * Amount to be adjusted by the counterparties to take into account any payments or prepayments made between the Trade Date and the Effective Date. ¹ Set forth, to at least 9 decimals, as a percentage of the Commitment/Loans of all Lenders thereunder. ² Insert if satisfaction of minimum amounts is to be determined as of the Trade Date. ³ To be added only if the consent of the Administrative Agent is required by the terms of the Credit Agreement. [Consented to:]4 KCP&L GREATER MISSOURI OPERATIONS COMPANY Title: [Consented to:]⁵ [ISSUERS] By: Title: [Consented to:]6 [SWING LINE LENDER] By: Title:

⁴ To be added only if the consent of the Borrower is required by the terms of the Credit Agreement.

⁵ To be added only if the consent of the Issuers is required by the terms of the Credit Agreement.

⁶ To be added only if the consent of the Swing Line Lender is required by the terms of the Credit Agreement.

ASSIGNMENT AND ASSUMPTION

Representations and Warranties.

- Assignor. The Assignor (a) represents and warrants that (i) it is the legal and beneficial owner of the Assigned Interest, (ii) the Assigned Interest is free and clear of any lien, encumbrance or other adverse claim and (iii) it has full power and authority, and has taken all action necessary, to execute and deliver this Assignment and to consummate the transactions contemplated hereby; and (b) assumes no responsibility with respect to (i) any statements, warranties or representations made in or in connection with the Credit Agreement or any other Loan Document, (ii) the execution, legality, validity, enforceability, genuineness, sufficiency or value of the Loan Documents or any collateral th ereunder, (iii) the financial condition of the Borrower, any of its Subsidiaries or Affiliates or any other Person obligated in respect of any Loan Document or (iv) the performance or observance by the Borrower, any of its Subsidiaries or Affiliates or any other Person of any of their respective obligations under any Loan Document.
- Assignee. The Assignee (a) represents and warrants that (i) it has full power and authority, and has taken all action necessary, to execute and deliver this Assignment and to consummate the transactions contemplated hereby and to become a Lender under the Credit Agreement, (ii) it meets all requirements of an Eligible Assignee under the Credit Agreement (subject to receipt of such consents as may be required under the Credit Agreement), (iii) from and after the Effective Date, it shall be bound by the provisions of the Credit Agreement as a Lender thereunder and, to the extent of the Assigned Interest, shall have the obligations of a Lender thereunder, (iv) it has received a copy of the Credit Agreement, together with copies of the most recent financial statements delivered pursuant to Section 7.1(a) and (b) thereof, as applicable, and such other documents and information as it has deemed appropriate to make its own credit analysis and decision to enter into this Assignment and to purchase the Assigned Interest on the basis of which it has made such analysis and decision independently and without reliance on the Administrative Agent or any other Lender and (v) if it is a Foreign Lender, attached hereto is any documentation required to be delivered by it pursuant to the terms of the Credit Agreement, duly completed and executed by the Assignee; and (b) agrees that (i) it will, independently and without reliance on the Administrative Agent, the Assignor or any other Lender, and based on such documents and information as it shall deem appropriate at the time, continue to make its own credit decisions in taking or not taking action under the Loan Documents, and (ii) it will perform in accordance with their terms all of the obligations which by the terms of the Loan Documents are required to be performed by it as a Lender.

2. <u>Pa</u>	<u>yments</u> . From and	after the Effective Date,	the Administrative Age	ent shall make all payments	in respect of the Assigned Intere
(including payments of	principal, interest, fe	ees and other amounts) to	the Assignor for amoun	ts which have accrued to but	excluding the

Effective Date and to the assignee for amounts which have accrued from and after the Effective Date.

3. <u>General Provisions</u>. This Assignment shall be binding upon, and inure to the benefit of, the parties hereto and their respective successors and assigns. This Assignment may be executed in any number of counterparts, which together shall constitute one instrument. Delivery of an executed counterpart of a signature page of this Assignment by telecopy shall be effective as delivery of a manually executed counterpart of this Assignment. This Assignment shall be governed by, and construed in accordance with, the law of the State of New York.

EXHIBIT C

LOAN/CREDIT RELATED MONEY TRANSFER INSTRUCTION

To: Bank of America, N.A., as Administrative Agent (the "Administrative Agent") under the Credit Agreement described below.

Re: Credit Agreement dated as of August 9, 2010 (as the same may be amended or modified, the "Credit Agreement"), among KCP&L Greater Missouri Operations Company (the "Borrower"), Great Plains Energy Incorporated (the "Guarantor"), the lenders from time to time party thereto, Union Bank, N.A. and Wells Fargo Bank, National Association, as Syndication Agents, and the Administrative Agent. Capitalized terms used herein and not otherwise defined herein shall have the meanings assigned thereto in the Credit Agreement.

The Administrative Agent is specifically authorized and directed to act upon the following standing money transfer instructions with respect to the proceeds of Credit Extensions or other extensions of credit from time to time until receipt by the Administrative Agent of a specific written revocation of such instructions by the Borrower; provided that the Administrative Agent may otherwise transfer funds as hereafter directed in writing by the Borrower in accordance with Section 14.1 of the Credit Agreement or based on any telephonic notice made in accordance with Section 2.14 of the Credit Agreement.

Facility Identification Number(s)	
Customer/Account Name	-
Transfer Funds To	
For Account No.	
Reference/Attention To	

Authorized Officer (Customer Representative)	Date
(Please Print)	Signature
Bank Officer Name	Date
(Please Print)	Signature
(Deliver O	Completed Form to Credit Support Staff For Immediate Processing)
	EXHIBIT D
	REVOLVING NOTE
	, 20
to <u>Article II</u> of the Agreement (as hereinafter de	RATIONS COMPANY, a Delaware corporation (the "Borrower"), promises to pay to the order of "Lender") the aggregate unpaid principal amount of all Loans made by the Lender to the Borrower pursuant fined), in immediately available funds at the Administrative Agent's Office, together with interest on the on the dates set forth in the Agreement. The Borrower shall pay the principal of and accrued and unpaid action Date.
The Lender shall, and is hereby authorized date and amount of each Loan and the date and am	ed to, record on a schedule attached hereto, or to otherwise record in accordance with its usual practice, the nount of each principal payment hereunder.
2010 (which, as it may be amended or modified an Incorporated (the "Guarantor"), the lenders from Association, as Syndication Agents, and Bank of the terms and conditions governing this Revolving	ving Notes issued pursuant to, and is entitled to the benefits of, the Credit Agreement dated as of August 9, nd in effect from time to time, is herein called the "Agreement"), among the Borrower, Great Plains Energy n time to time party thereto, including the Lender, Union Bank, N.A. and Wells Fargo Bank, National America, N.A., as Administrative Agent, to which Agreement reference is hereby made for a statement of g Note, including the terms and conditions under which this Revolving Note may be prepaid or its maturity and not otherwise defined herein are used with the meanings attributed to them in the Agreement.
•	GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF NEW
	[Remainder of page intentionally left blank]
KCP&L GREATER MISSOURI OPERA	TIONS COMPANY
	By: Print Name:
Title:	

, 20
FOR VALUE RECEIVED, the undersigned (the "Borrower"), hereby promises to pay to BANK OF AMERICA, N.A. (the "Swing Line Lender"), in accordance with the provisions of the Credit Agreement (as hereinafter defined), the principal amount of each Swing Line Loan from time to time made by the Swing Line Lender to the Borrower under that certain Credit Agreement dated as of August 9, 2010 (as amended, modified, supplemented or extended from time to time, the "Credit Agreement") among the Borrower, Great Plains Energy Incorporated (the "Guarantor"), the Lenders from time to time party thereto, Union Bank, N.A. and Wells Fargo Bank, National Association, as Syndication Agents, and Bank of America, N.A., as Administrative Agent and Swing Line Lender. Capitalized terms used but not otherwise defined herein have the meanings provided in the Credit Agreement.
The Borrower promises to pay interest on the unpaid principal amount of each Swing Line Loan from the date of such Swing Line Loan until such principal amount is paid in full, at such interest rates and at such times as provided in the Credit Agreement. All payments of principal and interest shall be made to the Administrative Agent for the account of the Swing Line Lender in dollars in immediately available funds at the Administrative Agent's Office. I any amount is not paid in full when due hereunder, such unpaid amount shall bear interest, to be paid upon demand, from the due date thereof until the date of actual payment (and before as well as after judgment) computed at the per annum rate set forth in the Credit Agreement.
This Swing Line Note is the Swing Line Note referred to in the Credit Agreement, is entitled to the benefits thereof and may be prepaid in whole or in part subject to the terms and conditions provided therein. Upon the occurrence and continuation of one or more of the Defaults specified in the Credit Agreement, all amounts then remaining unpaid on this Swing Line Note shall become, or may be declared to be, immediately due and payable all as provided in the Credit Agreement. Swing Line Loans made by the Swing Line Lender shall be evidenced by one or more loan accounts or records maintained by the Lender in the ordinary course of business. The Swing Line Lender may also attach schedules to this Swing Line Note and endorse thereon the date, amoun and maturity of its Swing Line Loans and payments with respect there to.
The Borrower, for itself, its successors and assigns, hereby waives diligence, presentment, protest and demand and notice of protest, demand dishonor and nonpayment of this Swing Line Note.
THIS SWING LINE NOTE SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF NEW YORK.
[Remainder of page intentionally left blank]
KCP&L GREATER MISSOURI OPERATIONS COMPANY
By: Print Name: Title:

CHAR1\1170499v7

GREAT PLAINS ENERGY INCORPORATED

COMPUTATION OF RATIO OF EARNINGS TO FIXED CHARGES

Year to Date September 30

	•	2010	2009	2008		2007	2006	2005
				(m	illions))		
Income from continuing operations Add	\$	216.7	\$ 151.9	\$ 119.7	\$	120.9	\$ 136.7	\$ 142.9
Equity investment loss		0.9	0.4	1.3		2.0	1.9	0.4
Income subtotal		217.6	152.3	121.0		122.9	138.6	143.3
Add								
Taxes on income		112.1	29.5	63.8		44.9	60.3	22.2
Kansas City earnings tax		0.5	0.4	0.3		0.5	0.5	0.5
Total taxes on income		112.6	29.9	64.1		45.4	60.8	22.7
Interest on value of leased property		4.6	6.5	3.6		3.9	4.1	6.2
Interest on long-term debt		162.9	203.6	126.2		74.1	62.6	64.3
Interest on short-term debt		7.4	10.3	18.2		26.4	9.2	4.5
Other interest expense and amortization (a)		(4.7)	4.7	(1.4)		5.8	3.9	4.3
Total fixed charges		170.2	225.1	146.6		110.2	79.8	79.3
Earnings before taxes on								
income and fixed charges	\$	500.4	\$ 407.3	\$ 331.7	\$	278.5	\$ 279.2	\$ 245.3
Ratio of earnings to fixed charges		2.94	1.81	2.26		2.53	3.50	3.09

⁽a) On January 1, 2007, Great Plains Energy elected to make an accounting policy change to recognize interest related to uncertain tax positions in interest expense.

KANSAS CITY POWER & LIGHT COMPANY

COMPUTATION OF RATIO OF EARNINGS TO FIXED CHARGES

Year to Date September 30

	36	ptember 3	U										
		2010		2009		2008		2007		2006		2005	
						(n	nillions	5)					
Income from continuing operations	\$	160.0	\$	128.9	\$	125.2	\$	156.7	\$	149.3	\$	151.5	
Add													
Taxes on income		83.6		46.9		59.8		59.3		70.3		48.0	
Kansas City earnings tax		0.5		0.2		0.5		0.5		0.5		0.5	
Total taxes on income		84.1		47.1		60.3		59.8		70.8		48.5	_
Interest on value of leased property		4.3		6.0		3.3		3.9		4.1		6.2	
Interest on long-term debt		88.4		110.4		79.3		54.5		55.4		56.7	
Interest on short-term debt		2.6		5.3		15.2		20.3		8.0		3.1	
Other interest expense and amortization (a)		(5.9)		0.3		1.4		6.8		3.2		3.6	_
Total fixed charges		89.4		122.0		99.2		85.5		70.7		69.6	
Earnings before taxes on income and fixed charges	\$	333.5	\$	298.0	\$	284.7	\$	302.0	\$	290.8	\$	269.6	
income and fixed charges	D	555.5	Ą	230.0	ψ	204./	φ	302.0	Φ	230.0	ψ	209.0	_
Ratio of earnings to fixed charges		3.73		2.44		2.87		3.53		4.11		3.87	

⁽a) On January 1, 2007, Kansas City Power & Light Company elected to make an accounting policy change to recognize interest related to uncertain tax positions in interest expense.

I, Michael J. Chesser, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of Great Plains Energy Incorporated;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(f) and 15d-15(f)) for the registrant and have:
 - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: October 28, 2010 /s/ Michael J. Chesser

Michael J. Chesser Chairman of the Board and Chief Executive Officer

I, James C. Shay, certify that:

- I have reviewed this quarterly report on Form 10-Q of Great Plains Energy Incorporated;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: October 28, 2010 /s/ James C. Shay

James C. Shay
Senior Vice President – Finance and Strategic Development and Chief
Financial Officer

I, Michael J. Chesser, certify that:

- I have reviewed this quarterly report on Form 10-Q of Kansas City Power & Light Company;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: October 28, 2010 /s/ Michael J. Chesser

Michael J. Chesser Chairman of the Board and Chief Executive Officer

I, James C. Shay, certify that:

- I have reviewed this quarterly report on Form 10-Q of Kansas City Power & Light Company;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the 3. financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our (b) supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most (d) recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to 5. the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal (b) control over financial reporting.

October 28, 2010 Date: /s/ James C. Shay

James C. Shav

Senior Vice President – Finance and Strategic Development and Chief

Financial Officer

Certification of CEO and CFO Pursuant to 18 U.S.C. Section 1350, as Adopted Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002

In connection with the Quarterly Report on Form 10-Q of Great Plains Energy Incorporated (the "Company") for the quarterly period ended September 30, 2010, as filed with the Securities and Exchange Commission on the date hereof (the "Report"), Michael J. Chesser, as Chairman of the Board and Chief Executive Officer of the Company, and James C. Shay, as Senior Vice President - Finance and Strategic Development and Chief Financial Officer of the Company, each hereby certifies, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that, to the best of his knowledge:

- (1) The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

/s/ Michael J. Chesser

Name: Michael J. Chesser

Title: Chairman of the Board and Chief Executive Officer

Date: October 28, 2010

/s/ James C. Shay

Name: James C. Shay

Title: Senior Vice President – Finance and Strategic Development and Chief Financial Officer

Date: October 28, 2010

Certification of CEO and CFO Pursuant to 18 U.S.C. Section 1350, as Adopted Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002

In connection with the Quarterly Report on Form 10-Q of Kansas City Power & Light Company (the "Company") for the quarterly period ended September 30, 2010, as filed with the Securities and Exchange Commission on the date hereof (the "Report"), Michael J. Chesser, as Chairman of the Board and Chief Executive Officer of the Company, and James C. Shay, as Senior Vice President – Finance and Strategic Development and Chief Financial Officer of the Company, each hereby certifies, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that, to the best of his knowledge:

- (1) The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

/s/ Michael J. Chesser

Name: Michael J. Chesser

Title: Chairman of the Board and Chief Executive Officer

Date: October 28, 2010

/s/ James C. Shay

Name: James C. Shay

Title: Senior Vice President - Finance and Strategic Development and Chief Financial Officer

Date: October 28, 2010